UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D. C. 20549

FORM 10-Q

□ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT
 □ OF 1934

For the quarterly period ended September 30, 2024

OR

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from

to

Commission File Number 001-35707

LIBERTY MEDIA CORPORATION

(Exact name of Registrant as specified in its charter)

State of Delaware

37-1699499 (I.R.S. Employer

(State or other jurisdiction of incorporation or organization)

Identification No.)

12300 Liberty Boulevard Englewood, Colorado

80112

(Address of principal executive offices)

(Zip Code)

Emerging Growth Company

Smaller Reporting Company □

Registrant's telephone number, including area code: **(720) 875-5400** Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol	Name of each exchange on which registered
Series A Liberty Formula One Common Stock	FWONA	The Nasdaq Stock Market LLC
Series C Liberty Formula One Common Stock	FWONK	The Nasdaq Stock Market LLC
Series A Liberty Live Common Stock	LLYVA	The Nasdaq Stock Market LLC
Series C Liberty Live Common Stock	LLYVK	The Nasdaq Stock Market LLC

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T ($\S232.405$ of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes ∞ No \square

Indicate by check mark whether the Registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Non-accelerated Filer

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Accelerated Filer □

 Series A
 Series B
 Series C

 Liberty Formula One common stock
 23,985,441
 2,434,102
 222,286,598

 Liberty Live common stock
 25,563,295
 2,541,341
 63,633,585

Table of Contents

Table of Contents

Part I – Financial Information

Item 1. Financial Statements	
LIBERTY MEDIA CORPORATION AND SUBSIDIARIES Condensed Consolidated Balance Sheets (unaudited)	I-3
LIBERTY MEDIA CORPORATION AND SUBSIDIARIES Condensed Consolidated Statements of Operations (unaudited)	I-5
LIBERTY MEDIA CORPORATION AND SUBSIDIARIES Condensed Consolidated Statements of Comprehensive Earnings (Loss)	I-7
(unaudited)	
LIBERTY MEDIA CORPORATION AND SUBSIDIARIES Condensed Consolidated Statements of Cash Flows (unaudited)	I-8
LIBERTY MEDIA CORPORATION AND SUBSIDIARIES Condensed Consolidated Statements of Equity (unaudited)	I-10
LIBERTY MEDIA CORPORATION AND SUBSIDIARIES Notes to Condensed Consolidated Financial Statements (unaudited)	I-12
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	I-34
Item 3. Quantitative and Qualitative Disclosures about Market Risk	I-46
Item 4. Controls and Procedures	I-47
Part II — Other Information	
Item 1. Legal Proceedings	II-1
Item 1A. Risk Factors	II-1
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	II-10
Item 5. Other Information	II-10
Item 6. Exhibits	II-11
SIGNATURES	II-12

Condensed Consolidated Balance Sheets

(unaudited)

		September 30, 2024	December 31, 2023
		amounts in mill	ions
Assets			
Current assets:			
Cash and cash equivalents	\$	3,054	1,713
Trade and other receivables, net		122	123
Other current assets		503	180
Current assets of discontinued operations (note 2)		<u> </u>	1,361
Total current assets		3,679	3,377
Investments in affiliates, accounted for using the equity method (note 7)		483	374
Property and equipment, at cost		1,010	973
Accumulated depreciation		(183)	(135)
·		827	838
Goodwill		4,192	3,956
Intangible assets subject to amortization, net		2,757	2,858
Other assets		1,270	1,384
Noncurrent assets of discontinued operations (note 2)		1,270	28,540
Total assets	\$	12 208	
Total assets	Þ	13,208	41,327
Liabilities and Equity			
Current liabilities:			
Accounts payable and accrued liabilities	\$	404	474
Current portion of debt, including \$55 million and \$69 million measured at			
fair value, respectively (note 8)		76	106
Deferred revenue		756	247
Other current liabilities		37	40
Current liabilities of discontinued operations (note 2)		_	3,876
Total current liabilities		1,273	4,743
Long-term debt, including \$1,902 million and \$1,728 million measured at fair			
value, respectively (note 8)		4,289	4,117
Other liabilities		187	188
Noncurrent liabilities of discontinued operations (note 2)		_	12,834
Total liabilities	\$	5,749	21,882

(Continued)

Condensed Consolidated Balance Sheets (Continued)

(unaudited)

	September 30, 2024	December 31, 2023
	amounts in mi	illions,
	except share a	nounts
Stockholders' equity:		
Preferred stock, \$.01 par value. Authorized 50,000,000 shares; no shares issued	\$ —	_
Series A Liberty Formula One common stock, \$.01 par value. Authorized		
500,000,000 shares; issued and outstanding 23,985,441 shares at September 30, 2024		
and 23,981,960 shares at December 31, 2023 (note 3)	_	_
Series A Liberty Live common stock, \$.01 par value. Authorized 521,400,000 shares;		
issued and outstanding 25,561,195 shares at September 30, 2024 and 25,558,577		
shares at December 31, 2023 (note 3)		_
Series A Liberty SiriusXM common stock, \$.01 par value. Authorized 2,000,000,000		
shares at December 31, 2023; issued and outstanding 98,134,522 shares at December	NA	1
31, 2023 (note 3) Series B Liberty Formula One common stock, \$.01 par value. Authorized 18,750,000	NA	1
shares; issued and outstanding 2,434,102 shares at September 30, 2024 and 2,437,583		
shares at December 31, 2023 (note 3)		
Series B Liberty Live common stock, \$.01 par value. Authorized 19,552,500 shares;		_
issued and outstanding 2,543,441 shares at September 30, 2024 and 2,546,146 shares		
at December 31, 2023 (note 3)	_	_
Series B Liberty SiriusXM common stock, \$.01 par value. Authorized 75,000,000		
shares at December 31, 2023; issued and outstanding 9,761,336 shares at December		
31, 2023 (note 3)	NA	_
Series C Liberty Formula One common stock, \$.01 par value. Authorized		
500,000,000 shares; issued and outstanding 222,236,393 shares at September 30,		
2024 and 208,196,119 shares at December 31, 2023 (note 3)	2	2
Series C Liberty Live common stock, \$.01 par value. Authorized 521,400,000 shares;		
issued and outstanding 63,631,315 shares at September 30, 2024 and 63,589,030		
shares at December 31, 2023 (note 3)	1	1
Series C Liberty SiriusXM common stock, \$.01 par value. Authorized 2,000,000,000		
shares at December 31, 2023; issued and outstanding 218,692,718 shares at		
December 31, 2023 (note 3)	NA	2
Additional paid-in capital	_	1,317
Accumulated other comprehensive earnings (loss), net of taxes	(87)	12
Retained earnings	7,520	15,061
Total stockholders' equity	7,436	16,396
Noncontrolling interests in equity of subsidiaries	23	3,049
Total equity	7,459	19,445
Commitments and contingencies (note 9)		
Total liabilities and equity	\$ 13,208	41,327

Condensed Consolidated Statements of Operations

(unaudited)

	T	hree montl Septemb		Nine mont Septemb	
		2024	2023	2024	2023
		,	amounts in	millions.	
		e:	xcept per sha	are amounts	
Revenue:			• •		
Formula 1 revenue	\$	848	887	2,251	1,992
Other revenue		63	49	235	350
Total revenue		911	936	2,486	2,342
Operating costs and expenses:					
Cost of Formula 1 revenue (exclusive of depreciation shown separately below)		554	615	1,472	1,340
Other cost of sales		41	_	161	_
Other operating expense		2	38	8	274
Selling, general and administrative, including stock-based compensation (note 4):		116	93	302	297
Impairment and acquisition costs		3	_	23	1
Depreciation and amortization		88	89	263	291
		804	835	2,229	2,203
Operating income (loss)		107	101	257	139
Other income (expense):					
Interest expense		(62)	(62)	(184)	(188)
Share of earnings (losses) of affiliates, net (note 7)		116	143	175	228
Realized and unrealized gains (losses) on financial instruments, net (note 6)		(55)	39	11	(45)
Unrealized gains (losses) on intergroup interests			(4)	_	(68)
Other, net		29	(19)	76	12
		28	97	78	(61)
Earnings (loss) from continuing operations before income taxes		135	198	335	78
Income tax (expense) benefit		(3)	(46)	(41)	(37)
Net earnings (loss) from continuing operations	_	132	152	294	41
Net earnings (loss) from discontinued operations (note 2)		(3,002)	291	(2,412)	757
Net earnings (loss)		(2,870)	443	(2,118)	798
Less net earnings (loss) attributable to the noncontrolling interests		(502)	58	(410)	145
Net earnings (loss) attributable to Liberty stockholders	\$	(2,368)	385	(1,708)	653
	_				
Net earnings (loss) from continuing operations attributable to Liberty stockholders:					
Liberty Formula One common stock	\$	117	118	218	125
Liberty Live common stock	•	15	(19)	76	(19)
Liberty SiriusXM common stock		_	74	_	45
Liberty Braves common stock		NA	(21)	NA	(109)
Net earnings (loss) from discontinued operations attributable to Liberty stockholders:					()
Liberty SiriusXM common stock		(2,500)	233	(2,002)	611
,	\$	(2,368)	385	(1,708)	653
	Ě	. ,,			

(Continued)

Condensed Consolidated Statements of Operations (Continued)

(unaudited)

	Three months		Nine month Septembe	
	 2024	2023	2024	2023
Basic net earnings (loss) from continuing operations attributable to Liberty				_
stockholders per common share (notes 3 and 5):				
Series A, B and C Liberty Formula One common stock	\$ 0.48	0.50	0.92	0.53
Series A, B and C Liberty Live common stock	\$ 0.16	(0.21)	0.83	(0.21)
Series A, B and C Liberty SiriusXM common stock	\$ _	0.23	_	0.14
Series A, B and C Liberty Braves common stock	\$ NA	(0.40)	NA	(2.06)
Basic net earnings (loss) from discontinued operations attributable to Liberty				
stockholders per common share (notes 3 and 5):				
Series A, B and C Liberty SiriusXM common stock	\$ (7.65)	0.71	(6.12)	1.87
Diluted net earnings (loss) from continuing operations attributable to Liberty				
stockholders per common share (notes 3 and 5):				
Series A, B and C Liberty Formula One common stock	\$ 0.48	0.39	0.90	0.37
Series A, B and C Liberty Live common stock	\$ 0.16	(0.21)	0.83	(0.21)
Series A, B and C Liberty SiriusXM common stock	\$ _	0.21	_	0.13
Series A, B and C Liberty Braves common stock	\$ NA	(0.40)	NA	(2.06)
Diluted net earnings (loss) from discontinued operations attributable to Liberty				
stockholders per common share (notes 3 and 5):				
Series A, B and C Liberty SiriusXM common stock	\$ (7.65)	0.68	(6.16)	1.76

Condensed Consolidated Statements of Comprehensive Earnings (Loss)

(unaudited)

	Three month September		Nine month Septemb	
	2024	2023	2024	2023
		amounts in		
Net earnings (loss)	\$ (2,870)	443	(2,118)	798
Other comprehensive earnings (loss), net of taxes:				
Foreign currency translation adjustments	8	(4)	6	4
Credit risk on fair value debt instruments gains (losses)	(45)	49	(50)	30
Unrealized holding gains (losses) arising during the period	_	1	_	_
Share of other comprehensive earnings (loss) of equity affiliates	(13)	(11)	(48)	59
Recognition of previously unrealized (gains) losses on debt	_	21	_	(13)
Other comprehensive earnings (loss) from continuing operations	(50)	56	(92)	80
Other comprehensive earnings (loss) from discontinued operations	(11)	(22)	(40)	(52)
Comprehensive earnings (loss)	(2,931)	477	(2,250)	826
Less comprehensive earnings (loss) attributable to the noncontrolling				
interests	(502)	56	(410)	145
Comprehensive earnings (loss) attributable to Liberty stockholders	\$ (2,429)	421	(1,840)	681
Comprehensive earnings (loss) from continuing operations attributable to				
Liberty stockholders:				
Liberty Formula One common stock	\$ 121	132	217	125
Liberty Live common stock	(39)	41	(15)	41
Liberty SiriusXM common stock	_	56	_	65
Liberty Braves common stock	NA	(21)	NA	(109)
Comprehensive earnings (loss) from discontinued operations attributable to				
Liberty stockholders:				
Liberty SiriusXM common stock	(2,511)	213	(2,042)	559
	\$ (2,429)	421	(1,840)	681

Condensed Consolidated Statements of Cash Flows

(unaudited)

		Nine months e	
		September 3	2023
	-	amounts in mi	
Cash flows from operating activities:		amounts in im	mons
Net earnings (loss)	\$	(2,118)	798
Adjustments to reconcile net earnings (loss) to net cash provided by operating activities:			
(Earnings) loss from discontinued operations		2,412	(757)
Depreciation and amortization		263	291
Stock-based compensation		27	22
Share of (earnings) loss of affiliates, net		(175)	(228)
Realized and unrealized (gains) losses on financial instruments, net		(11)	45
Unrealized (gains) losses on intergroup interests, net			68
Deferred income tax expense (benefit)		44	24
Intergroup tax allocation		(98)	(137)
Intergroup tax (payments) receipts		131	90
Other, net		21	50
Changes in operating assets and liabilities			
Current and other assets		(71)	(166)
Payables and other liabilities		151	480
Net cash provided (used) by operating activities		576	580
Cash flows from investing activities:			
Investments in equity method affiliates and debt and equity securities		(10)	(173)
Cash proceeds from dispositions		107	111
Cash (paid) received for acquisitions, net of cash acquired		(205)	_
Capital expended for property and equipment, including internal-use software and website			
development		(52)	(343)
Other investing activities, net		(12)	(20)
Net cash provided (used) by investing activities	,	(172)	(425)
Cash flows from financing activities:			
Borrowings of debt		644	1,165
Repayments of debt		(683)	(1,002)
Issuance of Series C Liberty Formula One common stock		939	_
Settlement of intergroup interests		_	(273)
Atlanta Braves Holdings, Inc. Split-Off		_	(188)
Other financing activities, net		43	22
Net cash provided (used) by financing activities		943	(276)
Effect of foreign exchange rate changes on cash, cash equivalents and restricted cash		1	
Net cash provided (used) by discontinued operations:	,		
Cash provided (used) by operating activities		879	1,301
Cash provided (used) by investing activities		(709)	(565)
Cash provided (used) by financing activities		(485)	(771)
Net cash provided (used) by discontinued operations		(315)	(35)
Net increase (decrease) in cash, cash equivalents and restricted cash		1,033	(156)
Cash, cash equivalents and restricted cash at beginning of period		2,028	2,276
Cash, cash equivalents and restricted cash at end of period	\$	3,061	2,120

Table of Contents

The following table reconciles cash and cash equivalents and restricted cash reported in our condensed consolidated balance sheets to the total amount presented in our condensed consolidated statements of cash flows:

	Sep	2024	December 31, 2023
		amounts in m	nillions
Cash and cash equivalents	\$	3,054	1,713
Restricted cash included in other current assets		7	_
Cash and cash equivalents included in current assets of discontinued operations		_	306
Restricted cash included in noncurrent assets of discontinued operations		_	9
Total cash and cash equivalents and restricted cash at end of period	\$	3,061	2,028

Condensed Consolidated Statements of Equity

(unaudited)

							S	tockholder	s' equity							
		erred		ty Formul			Liberty Liv			erty Sirius		Additional Paid-in	Accumulated other comprehensive		Noncontrolling interest in equity of	Total
	- 51	ock	Series A	Series B	Series C	Series A	Series B	Series C		series B its in millio		Capital	earnings (loss)	earnings	subsidiaries	equity
Balance at January 1, 2024	\$	_	_	_	2	_	_	1	1	— —	2	1,317	12	15,061	3,049	19,445
Net earnings (loss)	-	_	_	_	_	_	_	_	_	_	_		_	(1,708)	(410)	(2,118)
Other comprehensive earnings (loss)		_	_	_	_	_	_	_	_	_	_	_	(130)	`	(2)	(132)
Liberty SiriusXM Holdings Inc. Split- Off		_	_	_	_	_	_	_	(1)	_	(2)	(8,187)	31	_	(2,641)	(10,800)
Issuance of Series C Liberty Formula One common stock		_	_	_	_	_	_	_	_	_	_	939	_	_	_	939
Stock-based compensation		_	_	_	_	_	_	_	_	_	_	151	_	_	24	175
Withholding taxes on net share settlements of stock-based																
compensation		_	_	_	_	_	_	_	_	_	_	(47)	_	_	(52)	(47)
Dividends paid by subsidiary		_		_			_		_				_		(52)	(52)
Reclassification to additional paid-in capital		_	_	_	_	_	_	_	_	_	_	5,832	_	(5,832)	_	_
Other, net												(5)		(1)	55	49
Balance at September 30, 2024	\$				2			1	NA	NA	NA		(87)	7,520	23	7,459

	Pref	ferred	Liber	rty Formul	a One	-	S Liberty Liv	tockholder		erty Sirius	XM	Additional Paid-in	Accumulated other comprehensive	Retained	Noncontrolling interest in equity of	Total
		ock	Series A	Series B		Series A	Series B	Series C	Series A	Series B	Series C	Capital	earnings (loss)	earnings	subsidiaries	equity
									amoun	ts in millio	ons					
Balance at June 30, 2024	\$	_	_	_	2	_	_	1	1	_	2	1,387	(57)	15,724	3,147	20,207
Net earnings (loss)		_	_					_	_	_			_	(2,368)	(502)	(2,870)
Other comprehensive earnings (loss)		_	_	_	_	_	_	_	_	_	_	_	(61)	_	_	(61)
Liberty SiriusXM Holdings Inc. Split- Off		_	_	_	_	_	_	_	(1)	_	(2)	(8,187)	31	_	(2,641)	(10,800)
Issuance of Series C Liberty Formula One common stock		_	_	_	_	_	_	_	_	_		939	_	_	``_	939
Stock-based compensation		_	_	_			_	_	_	_		48	_	_	7	55
Withholding taxes on net share settlements of stock-based																
compensation		_	_	_	_	_	_	_	_	_	_	(16)	_	_	(10)	(16)
Dividends paid by subsidiary Reclassification to additional paid-in		_							_			_			(18)	(18)
capital		_	_	_	_	_	_	_	_	_	_	5,832	_	(5,832)	_	_
Other, net												(3)		(4)	30	23
Balance at September 30, 2024	\$				2	_=		1	NA	NA	NA		(87)	7,520	23	7,459

Condensed Consolidated Statements of Equity

(unaudited)

								Stock	cholders' eq	mitv								
	Preferred	Libor	rty Formul	a One		Liberty Live			erty SiriusX			iberty Brav	ros.	Additional Paid-in	Accumulated other comprehensive	Retained	Noncontrolling interest in	Total
	Stock	Series A					Series CSe		Series B	Series CS	eries A		Series C	Capital	earnings (loss)	earnings	equity of subsidiaries	equity
Balance at									an	nounts in	millions							
January 1, 2023	\$ —	_	_	2	NA	NA	NA	1	_	2	_	_	_	1,408	(39)	14,589	3,163	19,126
Net earnings (loss)	_	_	_	_	_	_	_	_	_	_		_	_	_	_	653	145	798
Other comprehensive																055	145	770
earnings (loss)	_	_	_	_	_	_	_	_	_	_	_	_	_	_	28	_	_	28
Stock-based compensation	_	_	_	_	_	_	_	_	_	_		_	_	161	_	_	25	186
Withholding taxes on net share settlements of stock-based																	20	
compensation Shares	-	_	_	=	_	=	_	_	_	_	_	=	_	(61)	_	_	_	(61
repurchased by subsidiary	_	_	_	_	_	_	_	_	_	_	_	_	_	46	_	_	(320)	(274
Shares issued by subsidiary	_	_	_	_	_	_	_	_	_	_	_	_	_	(50)	_	_	54	4
Dividends paid by subsidiary																	(48)	(48
Atlanta Braves Holdings, Inc.															_		(40)	(40
Split-Off Formula One	_	_	_	_	_	_	_	_	_	_	_	_	_	(173)	1	_	(11)	(183
Distribution	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	(289)	_	(289
Reclassification Other, net						_	1	_	_	_	_	_	_	(1) (32)	_	_	(16)	(4)
Balance at September 30,																		
023	<u>\$</u>			2			1	1		2	NA	NA	NA	1,298	(10)	14,953	2,992	19,239
								a										
	Df	Tiboo	4. F	- 0		T ile andra T inc			xholders' eq			th auto Dura		Additional	Accumulated other	D-4-in-d	Noncontrolling interest in	T-4-1
	Preferred Stock	Liber Series A	rty Formul Series B	a One Series CS		Liberty Live Series B	e Series CSe	Libe	erty SiriusX Series B	CM Series CS	eries A	iberty Brav		Additional Paid-in Capital		Retained earnings	Noncontrolling interest in equity of subsidiaries	Total equity
Balance at June	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A			Paid-in Capital	other comprehensive earnings (loss)	earnings	interest in equity of subsidiaries	equity
0, 2023								Libe	erty SiriusX Series B	CM Series CS	eries A			Paid-in	other comprehensive		interest in equity of	equit
0, 2023 Net earnings (loss)	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A			Paid-in Capital	other comprehensive earnings (loss)	earnings	interest in equity of subsidiaries	equit
0, 2023 Net earnings (loss) Other comprehensive	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A			Paid-in Capital	other comprehensive earnings (loss) (47)	14,857	interest in equity of subsidiaries 2,995	19,279 443
0, 2023 Net earnings (loss) Other comprehensive earnings (loss) Stock-based	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A			Paid-in Capital	other comprehensive earnings (loss) (47) — 36	14,857	interest in equity of subsidiaries 2,995 58 (2)	19,27 44
0, 2023 Net earnings (loss) Other comprehensive earnings (loss) Stock-based compensation Withholding taxes on net share settlements of stock-based	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A			Paid-in Capital	other comprehensive earnings (loss) (47)	14,857	interest in equity of subsidiaries 2,995	19,275 445 346
0, 2023 Net earnings (loss) Other comprehensive earnings (loss) Stock-based compensation Withholding taxes on net share settlements of stock-based compensation Stock-based	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A			Paid-in Capital	other comprehensive earnings (loss) (47) — 36	14,857	interest in equity of subsidiaries 2,995 58 (2)	19,27: 44. 3.
0, 2023 Net earnings (loss) Other comprehensive earnings (loss) Stock-based compensation Withholding taxes on net share settlements of stock-based compensation Shares repurchased by subsidiary	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A			Paid-in Capital	other comprehensive earnings (loss) (47) — 36	14,857	interest in equity of subsidiaries 2,995 58 (2)	19,279 443 3-69
0, 2023 Net earnings (loss) Other comprehensive earnings (loss) Stock-based compensation Withholding taxes on net share settlements of stock-based compensation Shares repurchased by subsidiary Shares issued by subsidiary	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A			1,469	other comprehensive earnings (loss) (47) — 36	14,857	interest in equity of subsidiaries 2,995 58 (2) 9	19,279 443 34 69 (33
0, 2023 Net earnings (loss) Other comprehensive earnings (loss) Stock-based compensation Withholding taxes on net share settlements of stock-based compensation Shares repurchased by subsidiary Shares issued by subsidiary Dividends paid	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A			Paid-in Capital 1,469 — 60 (33)	other comprehensive earnings (loss) (47) — 36	14,857	interest in equity of subsidiaries 2,995 58 (2) 9 (78)	19,279 44: 3-69 (3:
0, 2023 Net earnings (loss) Other comprehensive earnings (loss) Stock-based compensation Withholding taxes on net share settlements of stock-based compensation Shares repurchased by subsidiary Shares issued by subsidiary Atlanta Braves Holdings, Inc. Split-Off	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A		Series C	Paid-in Capital 1,469 — 60 (33) 6 (32) —	other comprehensive earnings (loss) (47) — 36	14,857 385 — — — — —	2,995 58 (2) 9 (78) 36 (15)	19,279 443 34 69 (33)
0, 2023 Net earnings (loss) Other comprehensive earnings (loss) Stock-based compensation Withholding taxes on net share settlements of stock-based compensation Shares repurchased by subsidiary Shares issued by subsidiary Shares issued by subsidiary Atlanta Braves Holdings, Inc. Split-Off Formula One	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A		Series C	Paid-in Capital 1,469 — 60 (33) 6 (32)	other comprehensive earnings (loss) (47) 36 1	14,857 385	interest in equity of subsidiaries 2,995 58 (2) 9 (78) 36	equity 19,275 443 34 65 (33 (72 4 (15) (183
(loss) Other comprehensive earnings (loss) Stock-based compensation Withholding taxes on net share compensation Shares repurchased by subsidiary Shares issued by subsidiary Dividends paid by subsidiary Atlanta Braves Holdings, Inc. Split-Off Formula One Distribution Reclassification	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A		Series C	Paid-in Capital 1,469 — 60 (33) 6 (32) — (173) — (1)	other comprehensive earnings (loss) (47) 36	14,857 385 — — — — —	2,995 58 (2) 9 (78) 36 (15)	equity 19,275 443 344 655 (333 (722 4 (155) (1833 (286)
10, 2023 Net earnings (loss) Other comprehensive earnings (loss) Stock-based compensation Withholding taxes on net share compensation Shares repurchased by subsidiary Shares issued by subsidiary Dividends paid by subsidiary Atlanta Braves Holdings, Inc. Spili-Off Formula One Distribution	Stock			Series CS	Series A	Series B	Series CSe	Libe	erty SiriusX Series B	KM Series CS nounts in	eries A		Series C	Paid-in Capital 1,469 — 60 (33) 6 (32) — (173)	other comprehensive earnings (loss) (47) 36 1	14,857 385	2,995 58 (2) 9 (78) 36 (15)	equity 19,275 443 34 65 (33) (72 4 (15)

Notes to Condensed Consolidated Financial Statements

(unaudited)

(1) Basis of Presentation

The accompanying condensed consolidated financial statements include all the accounts of Liberty Media Corporation and its controlled subsidiaries ("Liberty," the "Company," "we," "us," or "our" unless the context otherwise requires). All significant intercompany accounts and transactions have been eliminated.

Liberty, through its ownership of interests in subsidiaries and other companies, is primarily engaged in the media and entertainment industries primarily in North America and the United Kingdom. Liberty's most significant subsidiary is Delta Topco Limited (the parent company of Formula 1). Our most significant investment accounted for under the equity method is Live Nation Entertainment, Inc. ("Live Nation").

Braves Holdings, LLC ("Braves Holdings") was a subsidiary of the Company until the Atlanta Braves Holdings Split-Off (as defined in note 3) on July 18, 2023. Braves Holdings is not presented as a discontinued operation in the Company's condensed consolidated financial statements as the Atlanta Braves Holdings Split-Off did not represent a strategic shift that had a major effect on the Company's operations and financial results.

Sirius XM Holdings Inc. ("Sirius XM Holdings") was a subsidiary of the Company until the Liberty Sirius XM Holdings Split-Off (as defined in note 2) on September 9, 2024. Liberty Sirius XM Holdings Inc. ("Liberty Sirius XM Holdings"), which includes Sirius XM Holdings, is presented as a discontinued operation in the Company's condensed consolidated financial statements. See note 2 for details of the Liberty Sirius XM Holdings Split-Off.

The accompanying (a) condensed consolidated balance sheet as of December 31, 2023, which has been derived from audited financial statements, and (b) the interim unaudited condensed consolidated financial statements have been prepared in accordance with U.S. generally accepted accounting principles ("GAAP") for interim financial information and the instructions to Form 10-Q and Article 10 of Regulation S-X as promulgated by the Securities and Exchange Commission. Accordingly, they do not include all of the information and footnotes required by GAAP for complete financial statements. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation of the results for such periods have been included. The results of operations for any interim period are not necessarily indicative of results for the full year. Additionally, certain prior period amounts have been reclassified for comparability with current period presentation. These condensed consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto contained in Liberty's Annual Report on Form 10-K for the year ended December 31, 2023.

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities at the date of the financial statements and the reported amounts of revenue and expenses during the reporting period. Actual results could differ from those estimates. The Company considers (i) fair value measurement of non-financial instruments and (ii) accounting for income taxes to be its most significant estimates.

Liberty holds investments that are accounted for using the equity method. Liberty does not control the decision making process or business management practices of these affiliates. Accordingly, Liberty relies on management of these affiliates to provide it with accurate financial information prepared in accordance with GAAP that the Company uses in the application of the equity method. In addition, Liberty relies on audit reports that are provided by the affiliates' independent auditors on the financial statements of such affiliates. The Company is not aware, however, of any errors in or possible misstatements of the financial information provided by its equity affiliates that would have a material effect on Liberty's condensed consolidated financial statements.

On January 2, 2024, the Company purchased QuintEvents, LLC ("QuintEvents") for total consideration of approximately \$277 million, comprised of \$205 million of cash, net of cash acquired of \$66 million, and a \$6 million settlement of a pre-existing condition. The Company recorded \$235 million of goodwill, \$113 million of intangible assets

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

subject to amortization, net and \$121 million of deferred revenue as a result of the acquisition. The acquisition price allocation is preliminary and subject to revision as of September 30, 2024.

On March 29, 2024, the Company agreed, subject to certain conditions, to acquire approximately 86% of the equity interests in Dorna Sports, S.L., ("Dorna") for a purchase price of approximately €3.0 billion, to be funded with cash. The Company entered into foreign currency forward contracts related to the acquisition.

On August 22, 2024, the Company issued approximately 12.2 million shares of Series C Liberty Formula One common stock at an offering price of \$77.50 per share, resulting in gross proceeds of approximately \$949 million. The Company expects to use the net proceeds of the offering to partially fund the acquisition of Dorna and for general corporate purposes.

Liberty has entered into certain agreements with Qurate Retail, Inc. ("Qurate Retail"), Liberty TripAdvisor Holdings, Inc. ("TripCo"), Liberty Broadband Corporation ("Liberty Broadband") and Atlanta Braves Holdings, Inc. ("Atlanta Braves Holdings"), all of which are separate publicly traded companies, in order to govern relationships between the companies. None of these entities has any stock ownership, beneficial or otherwise, in any of the others. These agreements include Reorganization Agreements (in the case of Qurate Retail, Liberty Broadband and Atlanta Braves Holdings only), Services Agreements, Facilities Sharing Agreements, Tax Sharing Agreements (in the case of Liberty Broadband and Atlanta Braves Holdings only) and Aircraft Time Sharing Agreements. In addition, as a result of certain corporate transactions, Liberty and Qurate Retail may have obligations to each other for certain tax related matters. Effective August 31, 2024, the Facilities Sharing Agreement and the Aircraft Time Sharing Agreement with Atlanta Braves Holdings was terminated and members of Liberty management that served as officers of Atlanta Braves Holdings stepped down from their positions with Atlanta Braves Holdings (with limited exceptions), even though they may continue to provide services on an as-needed basis.

The Reorganization Agreements provide for, among other things, provisions governing the relationships between Liberty and each of Qurate Retail, Liberty Broadband and Atlanta Braves Holdings, including certain cross-indemnities. Pursuant to the Services Agreements, Liberty provides Qurate Retail, TripCo, Liberty Broadband and Atlanta Braves Holdings with general and administrative services including legal, tax, accounting, treasury, information technology, cybersecurity and investor relations support. Qurate Retail, TripCo, Liberty Broadband and Atlanta Braves Holdings reimburse Liberty for direct, out-of-pocket expenses incurred by Liberty in providing these services and, in the case of Qurate Retail, Qurate Retail's allocable portion of costs associated with any shared services or personnel based on an estimated percentage of time spent providing services to Qurate Retail. TripCo, Liberty Broadband and Atlanta Braves Holdings reimburse Liberty for shared services and personnel based on a flat fee. Under the Facilities Sharing Agreements, Liberty shares office space and related amenities at its corporate headquarters with Qurate Retail, TripCo, Liberty Broadband and, until August 31, 2024, Atlanta Braves Holdings. Under these various agreements, approximately \$7 million and \$12 million of these allocated expenses were reimbursed to Liberty during the three months ended September 30, 2024 and 2023, respectively, and approximately \$18 million and \$20 million were reimbursed to Liberty during the nine months ended September 30, 2024 and 2023, respectively.

In connection with Liberty's employment arrangement with Gregory B. Maffei, Liberty's President and Chief Executive Officer, pursuant to the Services Agreements between Liberty and each of TripCo, Liberty Broadband, Qurate Retail and Atlanta Braves Holdings (collectively, the "Service Companies"), components of Mr. Maffei's compensation are either paid directly to him by each Service Company or reimbursed to Liberty, in each case, based on allocations among Liberty and the Service Companies set forth in the respective services agreement, which are subject to adjustment on an annual basis and upon the occurrence of certain events. Effective August 31, 2024, upon the effectiveness of Mr. Maffei's resignation as an officer of Atlanta Braves Holdings, Mr. Maffei no longer receives compensation from Atlanta Braves Holdings.

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

Seasonality

Formula 1 recognizes the majority of its revenue and expenses in connection with World Championship race events ("Events") that take place in different countries around the world throughout the year. The Events in the past have generally taken place between March and December each year. As a result, the revenue and expenses recognized by Formula 1 are generally lower during the first quarter as compared to the rest of the quarters throughout the year.

QuintEvents' revenue is seasonal around its largest events, which are generally during the second and fourth quarters.

Braves Holdings revenue was seasonal, for the periods included herein, with the majority of revenue recognized during the second and third quarters which aligns with the baseball season.

(2) Discontinued Operations

On September 9, 2024, Liberty completed the split-off of its wholly owned subsidiary, Liberty Sirius XM Holdings (the "Liberty Sirius XM Holdings Split-Off"). The Liberty Sirius XM Holdings Split-Off was accomplished through the redemption by the Company of each outstanding share of Liberty Sirius XM common stock in exchange for 0.8375 of a share of Liberty Sirius XM Holdings common stock, with cash paid in lieu of fractional shares. Liberty Sirius XM Holdings is comprised of the businesses, assets and liabilities attributed to the Liberty SiriusXM Group immediately prior to the Liberty Sirius XM Holdings Split-Off. The Liberty Sirius XM Holdings Split-Off was intended to be tax-free to holders of Liberty SiriusXM common stock (except with respect to cash received in lieu of fractional shares).

Following the Liberty Sirius XM Holdings Split-Off, on September 9, 2024, a wholly owned subsidiary of Liberty Sirius XM Holdings merged with and into Sirius XM Holdings, with Sirius XM Holdings surviving the merger as a wholly owned subsidiary of Liberty Sirius XM Holdings (the "Merger" and, together with the Liberty Sirius XM Holdings Split-Off, the "Transactions"). As a result of the Transactions, Liberty Sirius XM Holdings is an independent public company separate from Liberty.

As disclosed in note 1, Liberty Sirius XM Holdings is presented as a discontinued operation in the Company's condensed consolidated financial statements as the Liberty Sirius XM Holdings Split-Off represents a strategic shift that had a major effect on the Company's operations and financial results.

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

The following table presents a reconciliation of the carrying amounts of the major classes of assets and liabilities of discontinued operations to the total assets and liabilities of discontinued operations as presented in the condensed consolidated balance sheet.

		ts in millions
Assets	u	
Current assets	\$	1,361
Investments in affiliates, accounted for using the equity method		715
Property and equipment, net		1,245
Intangible assets not subject to amortization		25,051
Intangible assets subject to amortization, net		1,014
Other assets		515
Total assets	\$	29,901
Liabilities		
Accounts payable and accrued liabilities	\$	1,536
Current portion of debt		1,074
Other current liabilities		1,266
Long-term debt		10,063
Deferred income tax liabilities		2,245
Other liabilities		526
Total liabilities	\$	16,710

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

The following table provided details about the major classes of line items constituting earnings (loss) from discontinued operations, net of tax as presented in the condensed consolidated statements of operations.

	Three months ended			Nine months ended		
		September	30,	Septembe	er 30,	
		2024	2023	2024	2023	
			amounts in m	illions		
Revenue	\$	1,664	2,271	6,004	6,665	
Cost of Sirius XM Holdings services (exclusive of depreciation shown						
separately below)		792	1,057	2,852	3,144	
Operating expense		122	159	461	504	
Selling, general and administrative		297	376	1,030	1,148	
Impairment, restructuring and acquisition costs		3,322	6	3,339	56	
Depreciation and amortization		110	145	421	460	
		4,643	1,743	8,103	5,312	
Operating income (loss)		(2,979)	528	(2,099)	1,353	
Other income (expense):						
Interest expense		(94)	(134)	(349)	(404)	
Other, net		8	(24)	122	10	
		(86)	(158)	(227)	(394)	
Earnings (loss) from discontinued operations before income taxes		(3,065)	370	(2,326)	959	
Income tax (expense) benefit		63	(79)	(86)	(202)	
Net earnings (loss) from discontinued operations		(3,002)	291	(2,412)	757	
Less net earnings (loss) from discontinued operations attributable to the						
noncontrolling interests		(502)	58	(410)	146	
Net earnings (loss) from discontinued operations attributable to Liberty						
stockholders	\$	(2,500)	233	(2,002)	611	

(3) Tracking Stocks

A tracking stock is a type of common stock that the issuing company intends to reflect or "track" the economic performance of a particular business or "group," rather than the economic performance of the company as a whole.

On July 18, 2023, the Company completed the split-off of its wholly owned subsidiary, Atlanta Braves Holdings (the "Atlanta Braves Holdings Split-Off"). The Atlanta Braves Holdings Split-Off was accomplished by a redemption by the Company of each outstanding share of Liberty Braves common stock in exchange for one share of the corresponding series of Atlanta Braves Holdings common stock. Atlanta Braves Holdings is comprised of the businesses, assets and liabilities attributed to the Liberty Braves Group (the "Braves Group") immediately prior to the Atlanta Braves Holdings Split-Off, except for the intergroup interests in the Braves Group attributed to the Liberty SiriusXM Group and Liberty Formula One Group (the "Formula One Group"), which were settled and extinguished in connection with the Atlanta Braves Holdings Split-Off.

On August 3, 2023, the Company reclassified its then-outstanding shares of common stock into three new tracking stocks — Liberty SiriusXM common stock, Liberty Formula One common stock and Liberty Live common stock, and, in

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

connection therewith, provided for the attribution of the businesses, assets and liabilities of the Company's remaining tracking stock groups among its newly created Liberty SiriusXM Group, Formula One Group and Liberty Live Group (the "Reclassification"). As a result of the Reclassification, each then-outstanding share of Liberty SiriusXM common stock was reclassified into one share of the corresponding series of new Liberty SiriusXM common stock and 0.2500 of a share of the corresponding series of Liberty Live common stock and each outstanding share of Liberty Formula One common stock was reclassified into one share of the corresponding series of new Liberty Formula One common stock and 0.0428 of a share of the corresponding series of Liberty Live common stock.

Each of the Atlanta Braves Holdings Split-Off and the Reclassification were intended to be tax-free to stockholders of the Company, except with respect to the receipt of cash in lieu of fractional shares. In July 2024, the IRS completed its review of the Reclassification and notified the Company that it agreed with the nontaxable characterization of the transaction. In September 2024, the IRS completed its review of the Atlanta Braves Holdings Split-Off and notified the Company that it agreed with the nontaxable characterization of the transaction. The Atlanta Braves Holdings Split-Off and the Reclassification are reflected in the Company's consolidated financial statements on a prospective basis.

While the Formula One Group and the Liberty Live Group have separate collections of businesses, assets and liabilities attributed to them, no group is a separate legal entity and therefore cannot own assets, issue securities or enter into legally binding agreements. Holders of tracking stock have no direct claim to the group's stock or assets and therefore, do not own, by virtue of their ownership of a Liberty tracking stock, any equity or voting interest in a public company, such as Live Nation, in which Liberty holds an interest that is attributed to a Liberty tracking stock group, the Liberty Live Group. Holders of tracking stock are also not represented by separate boards of directors. Instead, holders of tracking stock are stockholders of the parent corporation, with a single board of directors and subject to all of the risks and liabilities of the parent corporation.

The Liberty Formula One common stock is intended to track and reflect the separate economic performance of the businesses, assets and liabilities attributed to the Formula One Group, which as of September 30, 2024, include Liberty's interests in Formula 1 and QuintEvents, cash and Liberty's 2.25% Convertible Senior Notes due 2027. As of September 30, 2024, the Formula One Group had cash and cash equivalents of approximately \$2,666 million, which included \$1,449 million of subsidiary cash.

The Liberty Live common stock is intended to track and reflect the separate economic performance of the businesses, assets and liabilities attributed to the Liberty Live Group. As of September 30, 2024, the Liberty Live Group is comprised of Liberty's interest in Live Nation, cash, other minority investments, Liberty's 0.5% Exchangeable Senior Debentures due 2050, Liberty's 2.375% Exchangeable Senior Debentures due 2053 and an undrawn margin loan.

Prior to the Liberty Sirius XM Holdings Split-Off, the Liberty SiriusXM common stock was intended to track and reflect the separate economic performance of the businesses, assets and liabilities attributed to the Liberty SiriusXM Group. At the time of the Liberty Sirius XM Holdings Split-Off, the Liberty SiriusXM Group was comprised of Liberty's interest in Sirius XM Holdings, corporate cash, Liberty's 3.75% Convertible Senior Notes due 2028, Liberty's 2.75% Exchangeable Senior Debentures due 2049 and a margin loan obligation incurred by a wholly-owned special purpose subsidiary of Liberty. As disclosed in note 1, Liberty Sirius XM Holdings is presented as a discontinued operation in the Company's condensed consolidated financial statements. Prior to the Reclassification, Liberty's interest in Live Nation, Liberty's 0.5% Exchangeable Senior Debentures due 2050 and a margin loan secured by shares of Live Nation (the "Live Nation Margin Loan") were attributed to the Liberty SiriusXM Group and are presented as continuing operations in the Company's condensed consolidated financial statements.

Prior to the Atlanta Braves Holdings Split-Off, the Liberty Braves common stock was intended to track and reflect the separate economic performance of the businesses, assets and liabilities attributed to the Braves Group. The Braves Group was comprised primarily of Braves Holdings, which indirectly owns the Atlanta Braves Major League Baseball

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

Club ("ANLBC"), certain assets and liabilities associated with ANLBC's stadium (the "Stadium") and a mixed-use development around the Stadium that features retail, office, hotel and entertainment opportunities and corporate cash.

See Exhibit 99.1 to this Quarterly Report on Form 10-Q for unaudited attributed financial information for Liberty's tracking stock groups.

(4) Stock-Based Compensation

Liberty grants, to certain of its directors, employees and employees of its subsidiaries, restricted stock, restricted stock units ("RSUs") and stock options to purchase shares of its common stock (collectively, "Awards"). The Company measures the cost of employee services received in exchange for an equity classified Award (such as stock options and restricted stock) based on the grant-date fair value ("GDFV") of the Award, and recognizes that cost over the period during which the employee is required to provide service (usually the vesting period of the Award). The Company measures the cost of employee services received in exchange for a liability classified Award based on the current fair value of the Award, and remeasures the fair value of the Award at each reporting date. Stock-based compensation expense, included in selling, general and administrative expense in the accompanying condensed consolidated statements of operations, was \$7 million and \$6 million for the three months ended September 30, 2024 and 2023, respectively, and \$27 million and \$22 million for the nine months ended September 30, 2024 and 2023, respectively.

Liberty—Grants of Awards

Options granted during the nine months ended September 30, 2024 are summarized as follows:

	Nine Months Ended			
	September 30, 2024			
	Options granted (000's)	Weighted average GDFV		
Series C Liberty Formula One common stock, Liberty employees and directors (1)	4	\$	30.41	
Series C Liberty Formula One common stock, subsidiary employees (2)	83	\$	29.77	
Series C Liberty Live common stock, Liberty employees and directors (1)	1	\$	14.92	
Series C Liberty Live common stock, Liberty CEO (3)	70	\$	16.07	

N:-- M--4- F-J-J

- (1) Grants vest equally over three years.
- (2) Grants mainly vest in equal installments over 2024.
- (3) Grant cliff vests on December 31, 2024. Grant was made in connection with the CEO's employment agreement.

The Company did not grant any options to purchase Series A or Series B Liberty Formula One or Liberty Live common stock during the nine months ended September 30, 2024.

Also during the nine months ended September 30, 2024, the Company granted 88 thousand performance-based RSUs of Series C common stock of Liberty Formula One to our CEO. The RSUs had a GDFV of \$72.05 per share and cliff vest one year from the month of grant, subject to the satisfaction of certain performance objectives. Performance objectives, which are subjective, are considered in determining the timing and amount of compensation expense recognized. When the satisfaction of the performance objectives becomes probable, the Company records compensation expense. The probability of satisfying the performance objectives is assessed at the end of each reporting period.

Liberty calculates the GDFV for all of its equity classified options and the subsequent remeasurement of its liability classified options using the Black-Scholes Model. Liberty estimates the expected term of the options based on historical

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

exercise and forfeiture data. The volatility used in the calculation for options is based on the historical volatility of Liberty common stock and, when available, the implied volatility of publicly traded Liberty options. Liberty uses a zero-dividend rate and the risk-free rate for Treasury Bonds with a term similar to that of the subject options.

Liberty—Outstanding Awards

The following tables present the number and weighted average exercise price ("WAEP") of options to purchase Liberty common stock granted to certain officers, employees and directors of the Company, as well as the weighted average remaining life and aggregate intrinsic value of the options.

Liberty Formula One

	Series C						
	Liberty options (000's)		WAEP	Weighted average remaining life	in	gregate trinsic value villions)	
Outstanding at January 1, 2024	6,599	\$	37.62				
Granted	87	\$	72.53				
Exercised	(1,904)	\$	34.13				
Forfeited/Cancelled	_	\$	_				
Outstanding at September 30, 2024	4,782	\$	39.64	2.7 years	\$	181	
Exercisable at September 30, 2024	3,904	\$	37.37	2.4 years	\$	156	

Sories C

Liberty Live

	Series C							
	Liberty options (000's)		WAEP	Weighted average remaining life	int v	gregate trinsic alue illions)		
Outstanding at January 1, 2024	1,652	\$	42.36					
Granted	71	\$	40.00					
Exercised	(264)	\$	38.95					
Forfeited/Cancelled	(10)	\$	42.29					
Outstanding at September 30, 2024	1,449	\$	42.86	3.0 years	\$	12		
Exercisable at September 30, 2024	1,085	\$	43.77	2.4 years	\$	8		

As of September 30, 2024, there were no outstanding Series A or Series B options to purchase shares of Series A or Series B Liberty Formula One or Liberty Live common stock.

As of September 30, 2024, the total unrecognized compensation cost related to unvested Awards was approximately \$15 million. Such amount will be recognized in the Company's condensed consolidated statements of operations over a weighted average period of approximately 2.1 years.

As of September 30, 2024, Liberty reserved 4.8 million shares and 1.4 million shares of Series C common stock of Liberty Formula One and Liberty Live, respectively, for issuance under exercise privileges of outstanding stock options.

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

(5) Earnings Attributable to Liberty Media Corporation Stockholders Per Common Share

Basic earnings (loss) per common share ("EPS") is computed by dividing net earnings (loss) by the weighted average number of common shares outstanding ("WASO") for the period. Diluted EPS presents the dilutive effect on a per share basis of potential common shares as if they had been converted at the beginning of the periods presented, including any necessary adjustments to earnings (loss) attributable to shareholders.

Excluded from diluted EPS for the three and nine months ended September 30, 2024 are approximately 6 million and 4 million potentially dilutive shares of Series A and Series C Liberty Formula One common stock, respectively, 1 million potentially dilutive shares of Series A and Series C Liberty Live common stock, and 51 million and 18 million potentially dilutive shares of Series A and Series C Liberty SiriusXM common stock, respectively, because their inclusion would be antidilutive. Excluded from diluted EPS for the three and nine months ended September 30, 2023 are approximately 1 million and 3 million potentially dilutive shares of Series A and Series C Liberty Formula One common stock, respectively, 2 million potentially dilutive shares of Series A and C Liberty Live common stock, 24 million and 25 million potentially dilutive shares of Series A and Series C Liberty SiriusXM common stock, respectively, and 2 million and 7 million potentially dilutive shares of Series A and Series C Liberty Braves common stock, respectively, primarily due to warrants issued in connection with a bond hedge transaction (prior to settlement) and shares of Series A Liberty Braves common stock and Series A Liberty Formula One common stock underlying the intergroup interests (prior to settlement / extinguishment), because their inclusion would be antidilutive.

Series A, Series B and Series C Liberty Formula One Common Stock

The basic and diluted EPS calculations are based on the following WASO.

	Three months ended	September 30,	Nine months ended September 30,				
	2024 2023		2024	2023			
		numbers of shares in millions					
Basic WASO	242	235	237	234			
Potentially dilutive shares (a)	2	9	5	7			
Diluted WASO (b)	244	244	242	241			

- (a) Potentially dilutive shares are excluded from the computation of diluted EPS during periods in which losses are reported since the result would be antidilutive.
- (b) The Liberty SiriusXM Group's intergroup interest in the Formula One Group was settled and extinguished on July 12, 2023. The intergroup interest was a quasi-equity interest which was not represented by outstanding shares of common stock; rather, the Liberty SiriusXM Group had an attributed value in the Formula One Group which was generally stated in terms of a number of shares of stock issuable to the Liberty SiriusXM Group with respect to its interest in the Formula One Group. Each reporting period, the notional shares representing the intergroup interest were marked to fair value. As the notional shares underlying the intergroup interest were not represented by outstanding shares of common stock, such shares had not been officially designated Series A, B or C Liberty Formula One common stock. However, Liberty assumed that the notional shares (if and when issued) would be comprised of Series A Liberty Formula One common stock was underlying the 1.375% Cash Convertible Senior Notes due 2023. Therefore, the market price of Series A Liberty Formula One common stock was used for the quarterly mark-to-market adjustment through the unaudited attributed condensed consolidated statements of operations. The notional shares representing the intergroup interest had no impact on the basic WASO. However, if dilutive, the notional shares representing the intergroup interest were included in the diluted WASO as if the shares had been issued and outstanding during the period. For periods in which share settlement of the intergroup interest was dilutive, an adjustment was also made to the numerator in the diluted

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

earnings per share calculation for the unrealized gain or loss incurred from marking the intergroup interest to fair value during the period.

For periods in which share settlement of the 2.25% Convertible Senior Notes due 2027, which may be settled in shares of Series C Liberty Formula One common stock, is dilutive, the numerator adjustment includes a reversal of the interest expense and the unrealized gain or loss recorded on the instrument during the period, net of tax where appropriate. Additionally, an adjustment was also made to the numerator for a hypothetical mark-to-market adjustment on the shares of Series A Liberty Formula One common stock underlying the warrants in periods in which cash settlement would have been more dilutive than share settlement.

		Three months ended	September 30,	Nine months ended September 30,		
	2024		2023	2024	2023	
			amounts in			
Basic earnings (loss) attributable to Liberty						
Formula One stockholders	\$	117	118	218	125	
Adjustments		_	(23)	_	(37)	
Diluted earnings (loss) attributable to Liberty						
Formula One stockholders	\$	117	95	218	88	

Series A, Series B and Series C Liberty Live Common Stock

The basic and diluted EPS calculations are based on the following WASO.

	Three months ended September 30, 2024			August 4, 2023 to September 30, 2023
		numbers of sha	res in millions	
Basic WASO	92	92	92	92
Potentially dilutive shares (a)	_	_	_	_
Diluted WASO	92	92	92	92

⁽a) Potentially dilutive shares are excluded from the computation of diluted EPS during periods in which losses are reported since the result would be antidilutive.

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

Series A, Series B and Series C Liberty SiriusXM Common Stock

The basic and diluted EPS calculations are based on the following WASO.

	July 1, 2024 to September 9, 2024	Three months ended September 30, 2023 September 9, 2024		• •		Nine months ended September 30, 2023
		numbers of shar				
Basic WASO	327	327	327	327		
Potentially dilutive shares (a)		25	13	21		
Diluted WASO (b)	327	352	340	348		

⁽a) Potentially dilutive shares are excluded from the computation of EPS during periods in which net losses are reported since the result would be antidilutive.

(b) For periods in which share settlement of the 2.125% Exchangeable Senior Debentures due 2048 and 2.75% Exchangeable Senior Debentures due 2049, which could have been settled in shares of Series C Liberty SiriusXM common stock, and 3.75% Convertible Senior Notes due 2028, which could have been settled in shares of Series A Liberty SiriusXM common stock, are dilutive, the numerator adjustment includes a reversal of the interest expense and the unrealized gain or loss recorded on the instruments during the period, net of tax where appropriate. The settlement of the 2.125% Exchangeable Senior Debentures due 2048 changed to solely cash, pursuant to a supplemental indenture entered into during February 2023. Accordingly, the impact of share settlement of the 2.125% Exchangeable Senior Debentures due 2048 was considered for purposes of calculating diluted WASO prior to the execution of the supplemental indenture.

Additionally, a hypothetical mark-to-market adjustment on the shares of Series A Liberty SiriusXM common stock underlying the warrants was included in the numerator adjustment in periods in which cash settlement of the warrants would have been more dilutive than share settlement.

	July 1, 2024 to September 9, 2024		Three months ended September 30, 2023	January 1, 2024 to September 9, 2024	Nine months ended September 30, 2023
			amounts in	millions	
Basic earnings (loss) from discontinued					
operations attributable to Liberty SiriusXM					
stockholders	\$	(2,500)	233	(2,002)	611
Adjustments		_	6	(93)	1
Diluted earnings (loss) from discontinued					
operations attributable to Liberty SiriusXM					
stockholders	\$	(2,500)	239	(2,095)	612

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

Series A, Series B and Series C Liberty Braves Common Stock

The basic and diluted EPS calculations are based on the following WASO.

	Three months ended September 30, 2024	July 1, 2023 to Nine months ended July 18, 2023 September 30, 2024		January 1, 2023 to July 18, 2023			
	numbers of shares in millions						
Basic WASO	NA	53	NA	53			
Potentially dilutive shares (a)	NA	_	NA	1			
Diluted WASO	NA	53	NA	54			

⁽a) Potentially dilutive shares are excluded from the computation of diluted EPS during periods in which losses are reported since the result would be antidilutive.

(6) Assets and Liabilities Measured at Fair Value

For assets and liabilities required to be reported at fair value, GAAP provides a hierarchy that prioritizes inputs to valuation techniques used to measure fair value into three broad levels. Level 1 inputs are quoted market prices in active markets for identical assets or liabilities that the reporting entity has the ability to access at the measurement date. Level 2 inputs are inputs, other than quoted market prices included within Level 1, that are observable for the asset or liability, either directly or indirectly. Level 3 inputs are unobservable inputs for the asset or liability. Liberty does not have any assets or liabilities required to be measured at fair value considered to be Level 3.

Liberty's assets and liabilities measured at fair value are as follows:

			Value Measureme eptember 30, 202			ents at 23	
Description	_	<u>Total</u>	Quoted prices in active markets for identical assets (Level 1)	Significant other observable inputs (Level 2)	Total	Quoted prices in active markets for identical assets (Level 1)	Significant other observable inputs (Level 2)
Cash equivalents	P	2,677	2,677	amounts in	1.053	1,053	
•	Ф	2,077	2,077		,	,	
Debt and equity securities	\$	_	_	_	113	113	_
Financial instrument assets	\$	185	86	99	88	64	24
Debt	\$	1,957	_	1,957	1,797	_	1,797

The majority of Liberty's Level 2 financial instruments are debt related instruments and derivative instruments, which include foreign currency forward contracts and interest rate swaps. These assets and liabilities are not always traded publicly or not considered to be traded on "active markets," as defined in GAAP. The fair values for such instruments are derived from a typical model using observable market data as the significant inputs or a trading price of a similar asset or liability is utilized. Accordingly, those debt securities, financial instruments and debt or debt related instruments are reported in the foregoing table as Level 2 fair value. As of December 31, 2023, debt and equity securities included in the table above are included in the Other assets line item in the condensed consolidated balance sheet. As of September 30, 2024, \$95 million and \$90 million of financial instrument assets included in the table above are included in the Other current assets line items, respectively, in the condensed consolidated balance sheet. As of December 31,

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

2023, financial instrument assets included in the table above are included in the Other assets line item in the condensed consolidated balance sheet.

Realized and Unrealized Gains (Losses) on Financial Instruments, net

Realized and unrealized gains (losses) on financial instruments, net is comprised of changes in the fair value of the following:

	 Three months ended September 30,		Nine months ended September 30,		
	 2024 202		2024	2023	
		amounts in m	millions		
Debt and equity securities	\$ _	3	16	19	
Foreign currency forward contracts	85	_	93	_	
Debt measured at fair value (a)	(106)	8	(109)	(132)	
Other	(34)	28	11	68	
	\$ (55)	39	11	(45)	

⁽a) The Company elected to account for its exchangeable senior debentures and convertible notes (as described in note 8) using the fair value option. Changes in the fair value of the exchangeable senior debentures and convertible notes recognized in the condensed consolidated statements of operations are due to market factors primarily driven by changes in the fair value of the underlying shares into which the debt is exchangeable. The Company isolates the portion of the unrealized gain (loss) attributable to changes in the instrument specific credit risk and recognizes such amount in other comprehensive earnings (loss). The change in the fair value of the exchangeable senior debentures and convertible notes attributable to changes in the instrument specific credit risk was a loss of \$56 million and gain of \$56 million for the three months ended September 30, 2024 and 2023, respectively, and a loss of \$64 million and gain of \$38 million for the nine months ended September 30, 2024 and 2023, respectively. The cumulative change since issuance was a gain of \$70 million as of September 30, 2024, net of the recognition of previously unrecognized gains and losses.

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

(7) Investments in Affiliates Accounted for Using the Equity Method

Liberty has various investments accounted for using the equity method. The following table includes the Company's carrying amount and percentage ownership of the more significant investments in affiliates at September 30, 2024 and the carrying amount at December 31, 2023:

	September 30, 2024					December 31, 2023	
	Percentage ownership		Fair Value (Level 1) dollar amo	a	arrying mount millions	Carrying amount	
Formula One Group							
Other	various		NA	\$	35	41	
Total Formula One Group					35	41	
Liberty Live Group							
Live Nation	30 %	\$	7,625		422	307	
Other			NA		26	26	
Total Liberty Live Group					448	333	
Consolidated Liberty				\$	483	374	

The following table presents the Company's share of earnings (losses) of affiliates:

	· ·	Three months September		Nine months ended September 30,		
	2	024	2023	2024	2023	
Formanda Ona Crayon			amounts in m	illions		
Formula One Group			_			
Other (a)	\$	(1)	2	(6)	(1)	
Total Formula One Group		(1)	2	(6)	(1)	
Liberty Live Group						
Live Nation (a)		117	90	180	90	
Other (a)				1	<u> </u>	
Total Liberty Live Group		117	90	181	90	
Liberty SiriusXM Group						
Live Nation (a)		NA	50	NA	127	
Total Liberty Sirius XM Group		NA	50	NA	127	
Braves Group						
Other		NA	1	NA	12	
Total Braves Group		NA	1	NA	12	
Consolidated Liberty	\$	116	143	175	228	

⁽a) Liberty's interests in Live Nation and certain other equity affiliates were reattributed to the Liberty Live Group effective August 3, 2023.

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

Live Nation

Live Nation is considered the world's leading live entertainment company and seeks to innovate and enhance the live entertainment experience for artists and fans before, during and after the show. See note 8 for details regarding the number and fair value of shares pledged as collateral as of September 30, 2024 pursuant to the Live Nation Margin Loan.

Summarized financial information for Live Nation is as follows:

Balance Sheets

	 September 30, 2024	December 31, 2023		
	amounts in millions			
Current assets	\$ 9,775	9,533		
Property, plant and equipment, net	2,376	2,101		
Intangible assets	1,428	1,539		
Goodwill	2,670	2,691		
Investments in affiliates	515	448		
Other assets	2,983	2,718		
Total assets	\$ 19,747	19,030		
Current liabilities	\$ 9,681	9,984		
Long-term debt, net	5,673	5,459		
Other liabilities	2,439	2,175		
Redeemable noncontrolling interests	1,024	860		
Equity	 930	552		
Total liabilities and equity	\$ 19,747	19,030		

Statements of Operations

	Three months ended September 30,			Nine months ended September 30,		
		2024			2023	
			amounts in m	illions		
Revenue	\$	7,651	8,155	17,474	16,907	
Operating expenses:						
Direct operating expenses		5,780	6,298	12,840	12,590	
Selling, general and administrative expenses		1,005	974	2,913	2,533	
Depreciation and amortization		137	131	407	382	
Other operating expenses		89	98	250	236	
		7,011	7,501	16,410	15,741	
Operating income (loss)		640	654	1,064	1,166	
Interest expense		(88)	(86)	(249)	(257)	
Other income (expense), net		34	64	226	147	
Earnings (loss) before income taxes		586	632	1,041	1,056	
Income tax (expense) benefit		(70)	(51)	(192)	(127)	
Net earnings (loss)		516	581	849	929	
Less net earnings (loss) attributable to noncontrolling interests		64	60	154	139	
Net earnings (loss) attributable to Live Nation stockholders	\$	452	521	695	790	

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

(8) Long-Term Debt

Debt is summarized as follows:

	Outstanding	value	
	Principal September 30, 2024	September 30, 2024	December 31, 2023
		amounts in millions	
Formula One Group			
Corporate level notes and loans:			
2.25% Convertible Senior Notes due 2027 (1)	475	520	480
Other	54	54	58
Subsidiary notes and loans:			
Formula 1 Senior Loan Facilities	2,384	2,360	2,377
Deferred financing costs		(6)	(9)
Total Formula One Group	2,913	2,928	2,906
Liberty Live Group			
Corporate level notes and loans:			
0.5% Exchangeable Senior Debentures due 2050 (1)	50	55	69
2.375% Exchangeable Senior Debentures due 2053 (1)	1,150	1,382	1,248
Live Nation Margin Loan	_	_	_
Total Liberty Live Group	1,200	1,437	1,317
Total debt	\$ 4,113	4,365	4,223
Debt classified as current		(76)	(106)
Total long-term debt		\$ 4,289	4,117

(1) Measured at fair value

2.25% Convertible Senior Notes due 2027

On August 12, 2022, Liberty issued \$475 million convertible notes at an interest rate of 2.25% per annum, which, at Liberty's election, are convertible into cash, shares of Series C Liberty Formula One common stock or a combination of cash and shares of Series C Liberty Formula One common stock and mature on August 15, 2027. As of September 30, 2024, the conversion rate for the notes is approximately 12.0505 shares of Series C Liberty Formula One common stock per \$1,000 principal amount of notes, equivalent to a conversion price of approximately \$82.98 per share of Series C Liberty Formula One common stock. The notes are attributed to the Formula One Group. Liberty has elected to account for the notes using the fair value option. See note 6 for information related to unrealized gains (losses) on debt measured at fair value.

0.5% Exchangeable Senior Debentures due 2050

In November 2020, Liberty closed a private offering of approximately \$920 million aggregate principal amount of its 0.5% exchangeable senior debentures due 2050 (the "0.5% Exchangeable Senior Debentures due 2050"). Upon an exchange of debentures, pursuant to a supplemental indenture entered into in July 2024, Liberty delivered solely cash to satisfy its exchange obligations. The number of shares of Live Nation common stock attributable to a debenture represented an initial exchange price of approximately \$90.10 per share. Initially, a total of approximately 10 million shares of Live Nation common stock were attributable to the debentures. Interest was payable quarterly on March 1, June 1, September 1 and December 1 of each year. Holders of the debentures had the right to require Liberty to purchase their debentures on

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

September 1, 2024. Accordingly, the 0.5% Exchangeable Senior Debentures due 2050 are classified as a current liability in the condensed consolidated balance sheet. In August 2024, Liberty issued a redemption notice for all of its 0.5% Exchangeable Senior Debentures due 2050. Any debentures that were not so purchased or properly surrendered for exchange were redeemed in full on September 1, 2024. Settlement of any debentures properly surrendered for exchange was completed in October 2024. The redemption and purchase price generally equaled 100% of the adjusted principal amount of the debentures plus accrued and unpaid interest to (but excluding) the redemption or purchase date, plus any final period distribution. As of September 30, 2024, approximately 0.6 million shares of Live Nation common stock are attributable to the debentures. Liberty elected to account for the debentures using the fair value option. See note 6 for information related to unrealized gains (losses) on debt measured at fair value. On August 3, 2023, in connection with the Reclassification, as described in note 3, the debentures were reattributed from the Liberty SiriusXM Group to the Liberty Live Group.

2.375% Exchangeable Senior Debentures due 2053

In September 2023, Liberty closed a private offering of approximately \$1.15 billion aggregate principal amount of its 2.375% exchangeable senior debentures due 2053"). Upon an exchange of debentures, Liberty, at its option, may deliver Live Nation common stock, cash or a combination of Live Nation common stock and/or cash. The number of shares of Live Nation common stock attributable to a debenture represents an initial exchange price of approximately \$104.91 per share. A total of approximately 11 million shares of Live Nation common stock are attributable to the debentures. Interest is payable quarterly in arrears on March 31, June 30, September 30 and December 31 of each year. The debentures may be redeemed by Liberty, in whole or in part, on or after September 30, 2028. Holders of the debentures also have the right to require Liberty to purchase their debentures on September 30, 2028. The redemption and purchase price will generally equal 100% of the adjusted principal amount of the debentures plus accrued and unpaid interest to the redemption date, plus any final period distribution. The debentures are attributed to the Liberty Live Group. Liberty has elected to account for the debentures using the fair value option. See note 6 for information related to unrealized gains (losses) on debt measured at fair value.

Live Nation Margin Loan

On May 9, 2022, the Live Nation Margin Loan agreement was amended, replacing a delayed draw term loan with a \$400 million revolving line of credit, changing the interest rate to the Adjusted Term Secured Overnight Financing Rate ("SOFR") plus Term SOFR Adjustment (0.1%) plus 2.0% and extending the maturity to May 9, 2025. On September 5, 2023, the Live Nation Margin Loan agreement was amended to, among other things, extend the maturity date to September 9, 2026 and change the interest rate to Term SOFR plus 2.0%. The undrawn portion carries a commitment fee of 0.50% per annum. Interest on the margin loan is payable on the last business day of each calendar quarter. As of September 30, 2024, availability under the Live Nation Margin Loan was \$400 million. As of September 30, 2024, 9.0 million shares of the Company's Live Nation common stock with a value of \$982 million were pledged as collateral to the loan. The Live Nation Margin Loan contains various affirmative and negative covenants that restrict the activities of the borrower. The loan agreement does not include any financial covenants. On August 3, 2023, in connection with the Reclassification, as described in note 3, the Live Nation Margin Loan was reattributed from the Liberty SiriusXM Group to the Liberty Live Group.

Formula 1 Senior Loan Facilities

On November 23, 2022, Formula 1 refinanced its previous Term Loan B and revolving credit facility with a new \$725 million first lien Term Loan A, a refinanced \$1.7 billion Term Loan B and a new \$500 million revolving credit facility. On September 19, 2024, Formula 1 refinanced the Term Loan B with a new \$1.7 billion Term Loan B and extended the maturities of the approximately \$689 million Term Loan A and the \$500 million revolving credit facility (collectively, the

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

"Senior Loan Facilities"). The Term Loan A and revolving credit facility mature on September 30, 2029 and the Term Loan B matures on September 30, 2031. As of September 30, 2024, there were no outstanding borrowings under the \$500 million revolving credit facility. The margin for the Term Loan B, originally set at 3.25%, stepped down to 3.00% effective May 5, 2023, after a certain leverage test was met as of March 31, 2023. Formula 1 repriced the Term Loan B on October 4, 2023, reducing the margin to 2.25%. On September 19, 2024, the margin for the Term Loan B was reduced to 2.0%, with the potential to permanently step down to 1.75% if a certain leverage test is met on or after the earlier of the acquisition of Dorna or the termination of the Dorna acquisition. The margin for the Term Loan A and revolving credit facility is between 1.50% and 2.25% depending on leverage ratios, amongst other things, and was fixed at 1.75% for the first year and reduced to 1.5% effective November 24, 2023. The reference rate for the Term Loan A, Term Loan B and dollar borrowings under the revolving credit facility is Term SOFR. The weighted average interest rate on the Senior Loan Facilities was approximately 6.46% as of September 30, 2024. The Senior Loan Facilities remain non-recourse to Liberty. The Senior Loan Facilities are secured by share pledges and floating charges over Formula 1's primary operating companies with certain cross guarantees. Additionally, in order to manage the interest rate risk of its \$2.4 billion Senior Loan Facilities, Formula 1 had \$2.2 billion of interest rate swaps as of September 30, 2024, with a termination date in September 2031 and an early termination date in September 2029, at the option of the counterparty.

In connection with the September 19, 2024 refinancing, Formula 1 also marketed an incremental \$850 million of Term Loan B funding, which is in addition to an incremental \$150 million of commitments to the newly extended Term Loan A obtained in April 2024 (collectively, the "Incremental Term Loans"). The Incremental Term loans will be used to fund a portion of the Dorna acquisition, as described in note 1. The funding of the Incremental Term Loans are conditioned upon the scheduled consummation of the Dorna acquisition.

Debt Covenants

The Formula 1 Senior Loan Facilities contain certain financial covenants, including a leverage ratio. Additionally, Formula 1's debt and other borrowings contain certain non-financial covenants.

Fair Value of Debt

Due to the variable rate nature of the Live Nation Margin Loan and other debt, the Company believes that the carrying amount approximates fair value at September 30, 2024.

(9) Commitments and Contingencies

Guarantees

In connection with agreements for the sale of assets by the Company or its subsidiaries, the Company may retain liabilities that relate to events occurring prior to its sale, such as tax, environmental, litigation and employment matters. The Company generally indemnifies the purchaser in the event that a third party asserts a claim against the purchaser that relates to a liability retained by the Company. These types of indemnification obligations may extend for a number of years. The Company is unable to estimate the maximum potential liability for these types of indemnification obligations as the sale agreements may not specify a maximum amount and the amounts are dependent upon the outcome of future contingent events, the nature and likelihood of which cannot be determined at this time. Historically, the Company has not made any significant indemnification payments under such agreements and no amount has been accrued in the accompanying condensed consolidated financial statements with respect to these indemnification guarantees.

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

Litigation

The Company has contingent liabilities related to legal and tax proceedings and other matters arising in the ordinary course of business. Although it is reasonably possible the Company may incur losses upon conclusion of such matters, an estimate of any loss or range of loss cannot be made. In the opinion of management, it is expected that amounts, if any, which may be required to satisfy such contingencies will not be material in relation to the accompanying condensed consolidated financial statements.

(10) Information About Liberty's Operating Segments

The Company, through its ownership interests in subsidiaries and other companies, is primarily engaged in the media and entertainment industries. The Company identifies its reportable segments as (A) those consolidated subsidiaries that represent 10% or more of its consolidated annual revenue, annual Adjusted OIBDA (as defined below) or total assets and (B) those equity method affiliates whose share of earnings represent 10% or more of the Company's annual pre-tax earnings.

Liberty's chief operating decision maker evaluates performance and makes decisions about allocating resources to the Company's reportable segments based on financial measures such as revenue and Adjusted OIBDA (as defined below). In addition, the Company reviews nonfinancial measures such as subscriber growth, churn and penetration.

For the nine months ended September 30, 2024, the Company identified Formula 1 as a reportable segment. Formula 1 is a global motorsports business that holds exclusive commercial rights with respect to the World Championship, an annual, approximately ninemonth long, motor race-based competition in which teams compete for the Constructors' Championship and drivers compete for the Drivers' Championship. The World Championship takes place on various circuits with a varying number of Events taking place in different countries around the world each season. Formula 1 is responsible for the commercial exploitation and development of the World Championship as well as various aspects of its management and administration.

As of December 31, 2023, Live Nation met the Company's reportable segment threshold for equity method affiliates. See note 7 for segment disclosures related to Live Nation.

The Company's reportable segments are strategic business units that offer different products and services. They are managed separately because each segment requires different technologies, differing revenue sources and marketing strategies. The significant accounting policies of the segments are the same as those described in the Company's summary of significant policies in the Company's annual financial statements filed on Form 10-K.

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

Performance Measures

The following table disaggregates revenue by segment and by source:

	Three months September		Nine months ended September 30,	
	2024 2023		2024	2023
		amounts in mil	lions	
Formula One Group				
Formula 1:				
Primary	\$ 758	790	1,960	1,722
Other	103	97	325	270
Corporate and other	70	_	255	_
Intergroup elimination	(20)	_	(54)	_
Total Formula One Group	911	887	2,486	1,992
Braves Group				
Corporate and other:				
Baseball	NA	46	NA	318
Mixed-Use Development	NA	3	NA	32
Total Braves Group	NA	49	NA	350
Consolidated Liberty	\$ 911	936	2,486	2,342

Our subsidiaries' customers generally pay for services in advance of the performance obligation and therefore these prepayments are recorded as deferred revenue. The deferred revenue is recognized as revenue in our unaudited condensed consolidated statement of operations as the services are provided. The opening and closing balances for our deferred revenue related to Formula 1 and other subsidiaries for the nine months ended September 30, 2024 were approximately \$247 million and \$756 million, respectively. The primary cause for the increase related to the receipt of cash from our customers in advance of satisfying our performance obligations.

Significant portions of the transaction prices for Formula 1 and other subsidiaries are related to undelivered performance obligations that are under contractual arrangements that extend beyond one year. The Company anticipates recognizing revenue from the delivery of such performance obligations of approximately \$673 million for the remainder of 2024, \$2,593 million in 2025, \$2,130 million in 2026, \$5,880 million in 2027 through 2031, and \$1,870 million thereafter. We have not included any amounts in the undelivered performance obligations amounts for those performance obligations that relate to a contract with an original expected duration of one year or less.

For segment reporting purposes, the Company defines Adjusted OIBDA as revenue less operating expenses, and selling, general and administrative expenses excluding all stock-based compensation, separately reported litigation settlements and restructuring, acquisition and impairment charges. The Company believes this measure is an important indicator of the operational strength and performance of its businesses, by identifying those items that are not directly a reflection of each business' performance or indicative of ongoing business trends. In addition, this measure allows management to view operating results and perform analytical comparisons and benchmarking between businesses and identify strategies to improve performance. This measure of performance excludes depreciation and amortization, stock-based compensation, separately reported litigation settlements and restructuring, acquisition and impairment charges that are included in the measurement of operating income pursuant to GAAP. Accordingly, Adjusted OIBDA should be considered in addition to, but not as a substitute for, operating income, net income, cash flow provided by operating

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

activities and other measures of financial performance prepared in accordance with GAAP. The Company generally accounts for intersegment sales and transfers as if the sales or transfers were to third parties, that is, at current prices.

Adjusted OIBDA is summarized as follows:

	Three months ended September 30,			Nine months ended September 30,		
	2	024	2023	2024	2023	
			amounts in mi	llions		
Formula One Group						
Formula 1	\$	221	215	589	487	
Corporate and other		(14)	(18)	(15)	(44)	
Total Formula One Group		207	197	574	443	
Liberty Live Group						
Corporate and other		(2)	(4)	(4)	(4)	
Total Liberty Live Group		(2)	(4)	(4)	(4)	
Braves Group						
Corporate and other		NA	3	NA	14	
Total Braves Group		NA	3	NA	14	
Consolidated Liberty	\$	205	196	570	453	

Other Information

	September 30, 2024		
	Total assets	Investments in affiliates	
	amounts in millions		
Formula One Group			
Formula 1	\$ 9,324	1	
Corporate and other	2,798	34	
Intergroup elimination	(142)	_	
Total Formula One Group	11,980	35	
Liberty Live Group			
Corporate and other	1,255	448	
Total Liberty Live Group	1,255	448	
Elimination	(27)		
Consolidated Liberty	\$ 13,208	483	
· · · · · · · · · · · · · · · · · · ·			

Notes to Condensed Consolidated Financial Statements (Continued)

(unaudited)

The following table provides a reconciliation of Adjusted OIBDA to Operating income (loss) and Earnings (loss) from continuing operations before income taxes:

	Three months ended September 30,			Nine months ended September 30,	
		2024	2023	2024	2023
			amounts in mil	lions	
Adjusted OIBDA	\$	205	196	570	453
Impairment and acquisition costs		(3)	_	(23)	(1)
Stock-based compensation		(7)	(6)	(27)	(22)
Depreciation and amortization		(88)	(89)	(263)	(291)
Operating income (loss)		107	101	257	139
Interest expense		(62)	(62)	(184)	(188)
Share of earnings (losses) of affiliates, net		116	143	175	228
Realized and unrealized gains (losses) on financial instruments, net		(55)	39	11	(45)
Unrealized gains (losses) on intergroup interests		_	(4)	_	(68)
Other, net		29	(19)	76	12
Earnings (loss) from continuing operations before income taxes	\$	135	198	335	78

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Certain statements in this Quarterly Report on Form 10-Q constitute forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995, including statements regarding our projected sources and uses of cash; fluctuations in interest rates and stock prices; the anticipated non-material impact of certain contingent liabilities related to legal and tax proceedings; and other matters arising in the ordinary course of business. Where, in any forward-looking statement, we express an expectation or belief as to future results or events, such expectation or belief is expressed in good faith and believed to have a reasonable basis, but there can be no assurance that the expectation or belief will result or be achieved or accomplished. The following include some but not all of the factors (as they relate to our consolidated subsidiaries and equity affiliates) that could cause actual results or events to differ materially from those anticipated:

- the historical financial information of the Liberty Formula One Group (the "Formula One Group") and the Liberty Live Group may not necessarily reflect their results had they been separate companies;
- our ability to obtain additional financing on acceptable terms and cash in amounts sufficient to service debt and other financial obligations;
- our and our subsidiaries' indebtedness could adversely affect operations and could limit the ability of our subsidiaries to react to changes in the economy or our industry;
- the success of businesses attributed to each of our tracking stock groups and their popularity with audiences;
- our ability to realize the benefits of acquisitions or other strategic investments;
- the impact of weak and uncertain economic conditions on consumer demand for products, services and events offered by our businesses attributed to our tracking stock groups;
- our overlapping directors and management with Qurate Retail, Inc. ("Qurate Retail"), Liberty Broadband Corporation ("Liberty Broadband") and Liberty TripAdvisor Holdings, Inc. ("TripCo");
- the outcome of pending or future litigation;
- the operational risks of our subsidiaries and business affiliates with operations outside of the United States ("U.S.");
- our ability to use net operating loss, disallowed business interest and tax credit carryforwards to reduce future tax payments;
- the degradation, failure or misuse of our information systems;
- the ability of our subsidiaries and business affiliates to comply with government regulations, including, without limitation, competition laws and adverse outcomes from regulatory proceedings;
- the regulatory and competitive environment of the industries in which we, and the entities in which we have interests, operate;
- changes in the nature of key strategic relationships with partners, vendors and joint venturers;
- the impact of a future pandemic and other public health related risks and events, such as COVID-19, on our customers, vendors and businesses generally;
- reliance on intellectual property and the ability to protect intellectual property;
- · reliance on third parties;
- the ability to attract and retain qualified personnel;
- the impact of our equity method investment in Live Nation Entertainment, Inc. ("Live Nation") on our net earnings and the net earnings of the Liberty Live Group;
- termination of or changes in any of the agreements, commitments or policies Formula 1 relies on to operate and the limitations such agreements, commitments and policies impose on Formula 1;
- challenges by tax authorities in the jurisdictions where Formula 1 operates;
- changes in tax laws that affect Formula 1 and the Formula One Group;
- the ability of Formula 1 to expand into new markets;

- changes in laws and regulations and/or their interpretations related to advertising, media rights and the environment;
- the relationship between the United Kingdom ("U.K.") and the European Union ("E.U.") following Brexit;
- the establishment of rival motorsports events or other circumstances that impact the competitive position of Formula 1;
- the impact of cancelations or postponements of events or accidents or terrorist attacks during events;
- changes in consumer viewing habits and the emergence of new content distribution platforms;
- fluctuations in currencies against the U.S. dollar;
- the risks associated with the Company as a whole and our use of tracking stock groups, even if a holder does not own shares of common stock of all of our groups;
- market confusion that results from misunderstandings about our capital structure;
- market price of our tracking stocks may be volatile;
- we may not pay dividends equally to our tracking stocks or at all;
- our directors' or officers' equity ownership may create the appearance of conflicts of interest;
- geopolitical incidents, accidents, terrorist acts, international conflicts, natural disasters, including the effects of climate change, or
 other events that cause one or more events to be cancelled or postponed, are not covered by insurance, or cause reputational
 damage to our subsidiaries and business affiliates; and
- challenges related to assessing the future prospects of tracking stock groups based on past performance.

For additional risk factors, please see Part II, Item 1A. Risk Factors of this Quarterly Report on Form 10-Q and Part I, Item 1A. Risk Factors of our Annual Report on Form 10-K for the year ended December 31, 2023. Any forward-looking statements and such risks, uncertainties and other factors speak only as of the date of this Quarterly Report, and we expressly disclaim any obligation or undertaking to disseminate any updates or revisions to any forward-looking statement contained herein, to reflect any change in our expectations with regard thereto, or any other change in events, conditions or circumstances on which any such statement is based.

The following discussion and analysis provides information concerning our results of operations and financial condition. This discussion should be read in conjunction with our accompanying condensed consolidated financial statements and the notes thereto and our Annual Report on Form 10-K for the year ended December 31, 2023.

The information contained herein relates to Liberty Media Corporation and its controlled subsidiaries ("Liberty," the "Company," "we," "us," or "our" unless the context otherwise requires).

Overview

We own controlling and non-controlling interests in a broad range of media and entertainment companies. Formula 1, our most significant operating subsidiary, is a wholly-owned consolidated subsidiary and is also a reportable segment. Formula 1 is a global motorsports business that holds exclusive commercial rights with respect to the World Championship, an annual, approximately nine-month long, motor race-based competition in which teams compete for the Constructors' Championship and drivers compete for the Drivers' Championship. The World Championship takes place on various circuits with a varying number of events ("Events") taking place in different countries around the world each season. Formula 1 is responsible for the commercial exploitation and development of the World Championship as well as various aspects of its management and administration.

We hold an ownership interest in Live Nation, which is accounted for as an equity method investment as of September 30, 2024. Live Nation is considered the world's leading live entertainment company. As of December 31, 2023, Live Nation met the Company's reportable segment threshold for equity method affiliates.

Our "Corporate and Other" category includes our consolidated subsidiary QuintEvents, LLC ("QuintEvents"), corporate expenses and investments and related financial instruments in public companies, which are accounted for at their

respective fair market values. Braves Holdings, LLC ("Braves Holdings"), a consolidated subsidiary, was included in "Corporate and Other" prior to the Atlanta Braves Holdings Split-Off (defined below).

A tracking stock is a type of common stock that the issuing company intends to reflect or "track" the economic performance of a particular business or "group," rather than the economic performance of the company as a whole.

On July 18, 2023, the Company completed the split-off (the "Atlanta Braves Holdings Split-Off") of its wholly owned subsidiary, Atlanta Braves Holdings, Inc. ("Atlanta Braves Holdings"). The Atlanta Braves Holdings Split-Off was accomplished by a redemption by the Company of each outstanding share of Liberty Braves common stock in exchange for one share of the corresponding series of Atlanta Braves Holdings common stock. Atlanta Braves Holdings is comprised of the businesses, assets and liabilities attributed to the Liberty Braves Group (the "Braves Group") immediately prior to the Atlanta Braves Holdings Split-Off, except for the intergroup interests in the Braves Group attributed to the Liberty SiriusXM Group and Formula One Group, which were settled and extinguished in connection with the Atlanta Braves Holdings Split-Off.

On August 3, 2023, the Company reclassified its then-outstanding shares of common stock into three new tracking stocks — Liberty SiriusXM common stock, Liberty Formula One common stock and Liberty Live common stock, and, in connection therewith, provided for the attribution of the businesses, assets and liabilities of the Company's remaining tracking stock groups among its newly created Liberty SiriusXM Group, Formula One Group and Liberty Live Group (the "Reclassification"). As a result of the Reclassification, each thenoutstanding share of Liberty SiriusXM common stock was reclassified into one share of the corresponding series of new Liberty SiriusXM common stock and 0.2500 of a share of the corresponding series of Liberty Live common stock and each outstanding share of Liberty Formula One common stock and 0.0428 of a share of the corresponding series of Liberty Live common stock.

Each of the Atlanta Braves Holdings Split-Off and the Reclassification were intended to be tax-free to stockholders of the Company, except with respect to the receipt of cash in lieu of fractional shares. In July 2024, the IRS completed its review of the Reclassification and notified the Company that it agreed with the nontaxable characterization of the transaction. In September 2024, the IRS completed its review of the Atlanta Braves Holdings Split-Off and notified the Company that it agreed with the nontaxable characterization of the transaction. The Atlanta Braves Holdings Split-Off and the Reclassification are reflected in the Company's consolidated financial statements on a prospective basis.

On January 2, 2024, the Company purchased QuintEvents, LLC ("QuintEvents") for total consideration of approximately \$277 million, comprised of \$205 million of cash, net of cash acquired of \$66 million, and a \$6 million settlement of a pre-existing condition.

On September 9, 2024, Liberty completed the split-off (the "Liberty Sirius XM Holdings Split-Off") of its wholly owned subsidiary, Liberty Sirius XM Holdings Inc. ("Liberty Sirius XM Holdings"). The Liberty Sirius XM Holdings Split-Off was accomplished through the redemption by the Company of each outstanding share of Liberty Sirius XM common stock in exchange for 0.8375 of a share of Liberty Sirius XM Holdings common stock, with cash paid in lieu of fractional shares. Liberty Sirius XM Holdings is comprised of the businesses, assets and liabilities attributed to the Liberty Sirius XM Group immediately prior to the Liberty Sirius XM Holdings Split-Off. The Liberty Sirius XM Holdings Split-Off was intended to be tax-free to holders of Liberty Sirius XM common stock (except with respect to cash received in lieu of fractional shares).

Following the Liberty Sirius XM Holdings Split-Off, on September 9, 2024, a wholly owned subsidiary of Liberty Sirius XM Holdings merged with and into Sirius XM Holdings Inc. ("Sirius XM Holdings"), with Sirius XM Holdings surviving the merger as a wholly owned subsidiary of Liberty Sirius XM Holdings (the "Merger" and, together with the Liberty Sirius XM Holdings Split-Off, the "Transactions"). As a result of the Transactions, Liberty Sirius XM Holdings is an independent public company separate from Liberty.

While the Formula One Group and the Liberty Live Group have separate collections of businesses, assets and liabilities attributed to them, no group is a separate legal entity and therefore cannot own assets, issue securities or enter into legally binding agreements. Holders of tracking stock have no direct claim to the group's stock or assets and therefore, do not own,

by virtue of their ownership of a Liberty tracking stock, any equity or voting interest in a public company, such as Live Nation, in which Liberty holds an interest that is attributed to a Liberty tracking stock group, the Liberty Live Group. Holders of tracking stock are also not represented by separate boards of directors. Instead, holders of tracking stock are stockholders of the parent corporation, with a single board of directors and subject to all of the risks and liabilities of the parent corporation.

As of September 30, 2024, the Formula One Group is primarily comprised of Liberty's interests in Formula 1 and QuintEvents, cash and Liberty's 2.25% Convertible Senior Notes due 2027. The Formula One Group had cash and cash equivalents of approximately \$2,666 million as of September 30, 2024, which included \$1,449 million of subsidiary cash.

As of September 30, 2024, the Liberty Live Group is primarily comprised of Liberty's interest in Live Nation, cash, other minority investments, Liberty's 0.5% Exchangeable Senior Debentures due 2050, Liberty's 2.375% Exchangeable Senior Debentures due 2053 and an undrawn margin loan.

Prior to the Atlanta Braves Holdings Split-Off, the Braves Group was primarily comprised of Braves Holdings, which indirectly owns the Atlanta Braves Major League Baseball Club ("ANLBC"), certain assets and liabilities associated with ANLBC's stadium (the "Stadium") and a mixed-use development around the Stadium that features retail, office, hotel and entertainment opportunities and corporate cash.

Prior to the Liberty Sirius XM Holdings Split-Off, the Liberty SiriusXM common stock was intended to track and reflect the separate economic performance of the businesses, assets and liabilities attributed to the Liberty SiriusXM Group. At the time of the Liberty Sirius XM Holdings Split-Off, the Liberty SiriusXM Group was comprised of Liberty's interest in Sirius XM Holdings, corporate cash, Liberty's 3.75% Convertible Senior Notes due 2028, Liberty's 2.75% Exchangeable Senior Debentures due 2049 and a margin loan obligation incurred by a wholly-owned special purpose subsidiary of Liberty. Liberty Sirius XM Holdings is presented as a discontinued operation in the accompanying condensed consolidated financial statements. Prior to the Reclassification, Liberty's interest in Live Nation, Liberty's 0.5% Exchangeable Senior Debentures due 2050 and a margin loan secured by shares of Live Nation were attributed to the Liberty SiriusXM Group and are presented as continuing operations in the accompanying condensed consolidated financial statements.

Results of Operations—Consolidated

General. Provided in the tables below is information regarding our consolidated operating results and other income and expense, as well as information regarding the contribution to those items from our reportable segments. The "Corporate and other" category consists of those assets or businesses which do not qualify as a separate reportable segment. For a more detailed discussion and analysis of the financial results of our principal reportable segments see "Results of Operations—Businesses" below.

Braves Holdings was a subsidiary of the Company until the Atlanta Braves Holdings Split-Off on July 18, 2023. Braves Holdings is not presented as a discontinued operation in the Company's consolidated financial statements as the Atlanta Braves Holdings Split-Off did not represent a strategic shift that had a major effect on the Company's operations and financial results.

Consolidated Operating Results

		Three months ended September 30,			Nine months ended September 30,		
		2024	2023	2024	2023		
Revenue			amounts in mi	llions			
Formula One Group							
Formula 1	\$	861	887	2,285	1,992		
Corporate and other	Ψ	70	—	255	1,772		
Intergroup elimination		(20)	_	(54)	_		
Total Formula One Group		911	887	2,486	1,992		
Braves Group		711	007	2,.00	1,222		
Corporate and other		NA	49	NA	350		
Total Braves Group		NA	49	NA	350		
Consolidated Liberty	\$	911	936	2,486	2,342		
Operating Income (Loss)							
Formula One Group							
Formula 1		146	132	366	239		
Corporate and other		(36)	(25)	(102)	(64)		
Total Formula One Group		110	107	264	175		
Liberty Live Group	_	110	107	204	173		
Corporate and other		(3)	(5)	(7)	(5)		
Total Liberty Live Group		(3)	(5)	(7)	(5)		
Braves Group		(3)	(3)	(/)	(3)		
Corporate and other		NA	(1)	NA	(31)		
Total Braves Group		NA	(1)	NA	(31)		
Consolidated Liberty	\$	107	101	257	139		
Consolidated Liberty	Ψ	107	101	237	137		
Adjusted OIBDA							
Formula One Group							
Formula 1		221	215	589	487		
Corporate and other		(14)	(18)	(15)	(44)		
Total Formula One Group		207	197	574	443		
Liberty Live Group							
Corporate and other		(2)	(4)	(4)	(4)		
Total Liberty Live Group		(2)	(4)	(4)	(4)		
Braves Group							
Corporate and other		NA	3	NA	14		
Total Braves Group		NA	3	NA	14		
Consolidated Liberty	<u>\$</u>	205	196	570	453		

Revenue. Our consolidated revenue decreased \$25 million for the three months ended September 30, 2024, as compared to the corresponding period in the prior year, driven by a decrease in Braves Holdings revenue due to the Atlanta Braves Holdings Split-Off in 2023, a decrease in Formula 1 revenue and intergroup eliminations, partially offset by revenue from QuintEvents, which was acquired in January 2024. Our consolidated revenue increased \$144 million for the nine months ended September 30, 2024, as compared to the corresponding period in the prior year, driven by an increase in Formula 1 revenue and revenue from QuintEvents, partially offset by a decrease in Braves Holdings revenue due to the Atlanta Braves Holdings Split-Off in 2023 and intergroup eliminations. See "Results of Operations—Businesses" below for a more complete discussion of Formula 1's results of operations.

Operating income (loss). Our consolidated operating income increased \$6 million for the three months ended September 30, 2024, as compared to the corresponding period in the prior year, primarily driven by a \$14 million increase

Table of Contents

in Formula 1 operating results, partially offset by QuintEvents operating results, which was acquired in January 2024. Our consolidated operating income increased \$118 million for the nine months ended September 30, 2024, as compared to the corresponding period in the prior year, primarily driven by an increase in Formula 1 operating results and the Atlanta Braves Holdings Split-Off in 2023, partially offset by QuintEvents operating results. See "Results of Operations—Businesses" below for a more complete discussion of Formula 1's results of operations.

Stock-based compensation. Stock-based compensation includes compensation related to options, stock appreciation rights, restricted stock awards, restricted stock units, performance-based restricted stock units and other stock-based awards granted to officers, employees, nonemployee directors and employees of our subsidiaries. We recorded \$27 million and \$22 million of stock-based compensation expense for the nine months ended September 30, 2024 and 2023, respectively. As of September 30, 2024, the total unrecognized compensation cost related to unvested Liberty equity awards was approximately \$15 million. Such amount will be recognized in our condensed consolidated statements of operations over a weighted average period of approximately 2.1 years.

Acquisition costs. The Company recorded \$3 million and \$23 million of costs related to corporate acquisitions during the three and nine months ended September 30, 2024, respectively.

Adjusted OIBDA. To provide investors with additional information regarding our financial results, we also disclose Adjusted OIBDA, which is a non-GAAP (as defined below) financial measure. We define Adjusted OIBDA as operating income (loss) plus depreciation and amortization, stock-based compensation, separately reported litigation settlements, restructuring, acquisition and impairment charges. Our chief operating decision maker and management team use this measure of performance in conjunction with other measures to evaluate our businesses and make decisions about allocating resources among our businesses. We believe this is an important indicator of the operational strength and performance of our businesses by identifying those items that are not directly a reflection of each business' performance or indicative of ongoing business trends. In addition, this measure allows us to view operating results, perform analytical comparisons and benchmarking between businesses and identify strategies to improve performance. Accordingly, Adjusted OIBDA should be considered in addition to, but not as a substitute for, operating income, net income, cash flow provided by operating activities and other measures of financial performance prepared in accordance with U.S. generally accepted accounting principles ("GAAP"). The following table provides a reconciliation of Operating income (loss) to Adjusted OIBDA:

		Three months ended September 30,			ıs ended er 30,		
		2024		2024 2023		2024	2023
			nillions				
Operating income (loss)	\$	107	101	257	139		
Depreciation and amortization		88	89	263	291		
Stock-based compensation		7	6	27	22		
Impairment and acquisition costs		3	_	23	1		
Adjusted OIBDA	\$	205	196	570	453		

Consolidated Adjusted OIBDA increased \$9 million and \$117 million for the three and nine months ended September 30, 2024, respectively, as compared to the corresponding periods in the prior year, primarily due to increases in Formula 1 Adjusted OIBDA, partially offset by the Atlanta Braves Holdings Split-Off in 2023. See "Results of Operations—Businesses" below for a more complete discussion of Formula 1's results of operations.

Other Income and Expense

Components of Other Income (Expense) are presented in the table below.

	Three months ended September 30,			Nine months ended September 30,		
		2024	2023	2024	2023	
Interest expense			amounts in m	illions		
Formula One Group	\$	(54)	(56)	(162)	(161)	
Liberty Live Group	Ψ	(8)	(2)	(22)	(2)	
Liberty SiriusXM Group		_	(1)	(<u></u>)	(4)	
Braves Group		NA	(3)	NA	(21)	
Consolidated Liberty	\$	(62)	(62)	(184)	(188)	
Share of earnings (losses) of affiliates, net						
Formula One Group	\$	(1)	2	(6)	(1)	
Liberty Live Group		117	90	181	90	
Liberty SiriusXM Group			50		127	
Braves Group		NA	1	NA	12	
Consolidated Liberty	\$	116	143	175	228	
Realized and unrealized gains (losses) on financial instruments, net						
Formula One Group	\$	39	66	86	83	
Liberty Live Group		(94)	(72)	(75)	(72)	
Liberty SiriusXM Group		_	45	<u> </u>	(59)	
Braves Group		NA	_	NA	3	
Consolidated Liberty	\$	(55)	39	11	(45)	
Unrealized gains (losses) on intergroup interests						
Formula One Group	\$		16		15	
Braves Group	Ф	_	(20)	_	(83)	
Consolidated Liberty	\$		(4)		(68)	
Consolidated Liberty	D		(4)		(08)	
Other, net						
Formula One Group	\$	21	14	56	48	
Liberty Live Group		8	(35)	20	(35)	
Liberty SiriusXM Group		_	1	_	(6)	
Braves Group		NA	<u> </u>	NA	5	
Consolidated Liberty	\$	29	(19)	76	12	
	\$	28	97	78	(61)	
	4				(01)	

Interest expense. Consolidated interest expense was flat and decreased \$4 million for the three and nine months ended September 30, 2024, respectively, as compared to the corresponding periods in the prior year. Interest expense for the Liberty Live Group increased due to an increase in the average amount of debt outstanding. Certain debt was reattributed from the Liberty SiriusXM Group to the Liberty Live Group effective August 3, 2023. The interest related to such debt is reflected in interest expense for the Liberty SiriusXM Group prior to the Reclassification and in interest expense for the Liberty Live Group following the Reclassification. Interest expense for the Braves Group decreased due to the Atlanta Braves Holdings Split-Off.

Share of earnings (losses) of affiliates, net. The following table presents our share of earnings (losses) of affiliates:

	1	Three months ended September 30,			s ended er 30,
		2024	2023 amounts in 1	2024 millions	2023
Formula One Group			amounts in i	iiiioiis	
Other (a)	\$	(1)	2	(6)	(1)
Total Formula One Group		(1)	2	(6)	(1)
Liberty Live Group					
Live Nation (a)		117	90	180	90
Other (a)		_	_	1	_
Total Liberty Live Group		117	90	181	90
Liberty SiriusXM Group					
Live Nation (a)		NA	50	NA	127
Total Liberty Sirius XM Group		NA	50	NA	127
Braves Group					
Other		NA	1	NA	12
Total Braves Group		NA	1	NA	12
Consolidated Liberty	\$	116	143	175	228

⁽a) Liberty's interests in Live Nation and certain other equity affiliates were reattributed to the Liberty Live Group effective August 3 2023

Realized and unrealized gains (losses) on financial instruments, net. Realized and unrealized gains (losses) on financial instruments are comprised of changes in the fair value of the following:

	Three months ended September 30,			Nine months ended September 30,		
	2024 2023 amounts in 1			2024 millions	2023	
Debt and equity securities	\$	_	3	16	19	
Foreign currency forward contracts		85	_	93	_	
Debt measured at fair value		(106)	8	(109)	(132)	
Other		(34)	28	11	68	
	\$	(55)	39	11	(45)	

The changes in unrealized gains (losses) on debt and equity securities are due to market factors primarily driven by changes in the fair value of the stock underlying these financial instruments. Changes in unrealized gains (losses) on foreign currency forward contracts are driven by changes in foreign currency exchange rates. Changes in unrealized gains (losses) on debt measured at fair value are due to market factors primarily driven by changes in the fair value of the underlying shares into which the debt is exchangeable. Other realized and unrealized gains (losses) are primarily driven by changes in the fair value of Formula 1's interest rate swaps and the realized gains (losses) on Formula 1's interest rate swaps.

Unrealized gains (losses) on intergroup interests. Unrealized gains (losses) on intergroup interests are driven by changes in the fair value of notional shares representing the intergroup interests. The intergroup interests were settled and extinguished during the nine months ended September 30, 2023.

Other, net. Other, net income increased \$48 million and \$64 million for the three and nine months ended September 30, 2024, respectively, as compared to the corresponding periods in the prior year due to decreases in losses on early extinguishment of debt, increases in foreign exchange gains and increases in interest and dividend income. The increase

during the nine months ended September 30, 2024 was also driven by tax related expense pursuant to a tax sharing agreement with Qurate Retail, Inc. recorded during the nine months ended September 30, 2023.

Income taxes. During the three and nine months ended September 30, 2024, we had earnings from continuing operations before income taxes of \$135 million and \$335 million, respectively, and income tax expense of \$3 million and \$41 million, respectively. During the three and nine months ended September 30, 2023, we had earnings from continuing operations before income taxes of \$198 million and \$78 million, respectively, and income tax expense of \$46 million and \$37 million, respectively. For the three months ended September 30, 2024, the Company recognized tax expense less than the expected federal tax rate of 21% primarily due to certain gains that are not taxable and earnings in foreign jurisdictions taxed at rates lower than the 21% U.S. federal tax rate. For the nine months ended September 30, 2024, the Company recognized tax expense less than the expected federal tax rate of 21% due to tax benefits related to deductible stock-based compensation, certain gains that are not taxable and earnings in foreign jurisdictions taxed at rates lower than the 21% U.S. federal tax rate. For the three months ended September 30, 2023, the Company recognized tax expense greater than the expected federal tax rate of 21% primarily due to the impact of a reduction to deferred tax assets for certain federal tax credit carryforwards, partially offset by certain gains that are not taxable. For the nine months ended September 30, 2023, the Company recognized tax expense greater than the expected federal tax rate of 21% primarily due to certain losses that are not deductible for tax purposes.

Net earnings (loss) from continuing operations. We had net earnings from continuing operations of \$132 million and net earnings from continuing operations of \$294 million during the three and nine months ended September 30, 2024, respectively, and net earnings from continuing operations of \$152 million and \$41 million for the three months and nine months ended September 30, 2023, respectively. The changes were the result of the above-described fluctuations in our revenue, expenses and other gains and losses.

Material Changes in Financial Condition

As of September 30, 2024, substantially all of our cash and cash equivalents were invested in U.S. Treasury securities, other government securities or government guaranteed funds, AAA rated money market funds and other highly rated financial and corporate debt instruments.

The following are potential sources of liquidity: available cash balances, cash generated by the operating activities of our subsidiaries (to the extent such cash exceeds the working capital needs of the subsidiaries and is not otherwise restricted), proceeds from net asset sales, monetization of our public investment portfolio (including derivatives), debt borrowings and equity issuances, and dividend and interest receipts.

Liberty does not have a debt rating.

As of September 30, 2024, Liberty's cash and cash equivalents were as follows:

	Equ	and Cash nivalents s in millions
Formula One Group		
Formula 1	\$	1,383
Corporate and other		1,283
Total Formula One Group	\$	2,666
Liberty Live Group		
Corporate and other	\$	388
Total Liberty Live Group	\$	388

Cash held by Formula 1 is accessible by Liberty, except when a restricted payment ("RP") test imposed by the first lien term loan and the revolving credit facility at Formula 1 is not met. Pursuant to the RP test, Liberty does not have

unlimited access to Formula 1's cash when the leverage ratio (defined as net debt divided by covenant earnings before interest, tax, depreciation and amortization for the trailing twelve months) exceeds a certain threshold. During the nine months ended September 30, 2024, Formula 1 distributed \$150 million to Liberty and the RP test was met, pro forma for such distribution. If distributions are made in the future, the RP test, pro forma for such distributions, would have to be met. As of September 30, 2024, Liberty had \$400 million available under Liberty's margin loan secured by shares of Live Nation. Liberty believes that it currently has appropriate legal structures in place to repatriate foreign cash as tax efficiently as possible and meet the business needs of the Company.

The Company and Formula 1 are in compliance with their debt covenants as of September 30, 2024.

	Nine months ended September 30,		
		2023	
Cash Flow Information		amounts in m	illions
Formula One Group cash provided (used) by operating activities	\$	587	554
Liberty Live Group cash provided (used) by operating activities		(11)	(3)
Liberty SiriusXM Group cash provided (used) by operating activities		_	(3)
Braves Group cash provided (used) by operating activities		NA	32
Net cash provided (used) by operating activities	\$	576	580
Formula One Group cash provided (used) by investing activities	\$	(277)	(391)
Liberty Live Group cash provided (used) by investing activities		105	1
Braves Group cash provided (used) by investing activities		NA	(35)
Net cash provided (used) by investing activities	\$	(172)	(425)
Formula One Group cash provided (used) by financing activities	\$	954	(426)
Liberty Live Group cash provided (used) by financing activities		(11)	317
Liberty SiriusXM Group cash provided (used) by financing activities		_	3
Braves Group cash provided (used) by financing activities		NA	(170)
Net cash provided (used) by financing activities	\$	943	(276)

Liberty's primary uses of cash during the nine months ended September 30, 2024 (excluding cash used by Formula 1) were \$205 million for acquisitions, net of cash acquired, and \$117 million for debt repayments, which were primarily funded by cash on hand and proceeds from dispositions. In addition, the Company generated approximately \$939 million of net proceeds from the issuance of approximately 12.2 million shares of Series C Liberty Formula One common stock during the nine months ended September 30, 2024.

During the nine months ended September 30, 2024, Formula 1's primary use of cash was \$50 million of capital expenditures, funded by cash from operations.

The projected uses of Liberty's cash (excluding Formula 1's uses of cash) are primarily capital expenditures, the investment in existing or new businesses, including the acquisition of Dorna Sports, S.L., debt service and the potential buyback of common stock under the approved share buyback program. Liberty expects to fund its projected uses of cash with cash on hand, borrowing capacity under margin loans and outstanding or new debt instruments, or distributions from operating subsidiaries. Liberty may be required to make net payments of income tax liabilities to settle items under discussion with tax authorities.

Formula 1's uses of cash are expected to be capital expenditures, debt service payments and operating expenses. Liberty expects Formula 1 to fund its projected uses of cash with cash on hand and cash provided by operations.

We believe that the available sources of liquidity are sufficient to cover our projected future uses of cash.

Results of Operations—Businesses

Formula 1. Formula 1 is a global motorsports business that holds exclusive commercial rights with respect to the World Championship, an annual, approximately nine-month long, motor race-based competition in which teams compete

for the Constructors' Championship and drivers compete for the Drivers' Championship. The World Championship takes place on various circuits throughout the world. Formula 1 derives its primary revenue from the commercial exploitation and development of the World Championship through a combination of race promotion, media rights and sponsorship arrangements. A significant majority of the race promotion, media rights and sponsorship contracts specify payments in advance and annual increases in the fees payable over the course of the contracts. The 2024 World Championship calendar is scheduled to have 24 Events. The 2023 World Championship calendar was originally scheduled to have 23 Events. However, following the cancellation of the Emilia-Romagna Grand Prix at Imola due to severe flooding in the region, and confirmation that it would not be replaced in May 2023, 22 Events took place, with income and cost recognition adjusted to reflect the revised 22 Event calendar during the second quarter of 2023.

During the three months ended September 30, 2024, seven Events took place, compared to eight Events in the corresponding period in the prior year. During the nine months ended September 30, 2024, 18 Events took place, compared to 16 Events in the corresponding period in the prior year.

Following the acquisition of QuintEvents, Formula 1's results include intergroup revenue that is eliminated in consolidation.

Formula 1's operating results were as follows:

	Three months ended September 30,			Nine months ended September 30,			
		2024 2				2024	2023
			amounts in m	illions			
Primary Formula 1 revenue	\$	758	790	1,960	1,722		
Other Formula 1 revenue		103	97	325	270		
Total Formula 1 revenue		861	887	2,285	1,992		
Operating expenses:							
Cost of Formula 1 revenue		(561)	(615)	(1,492)	(1,340)		
Selling, general and administrative expenses (excluding stock-							
based compensation)		(79)	(57)	(204)	(165)		
Adjusted OIBDA		221	215	589	487		
Stock-based compensation		(1)	(1)	(2)	(2)		
Depreciation and amortization		(74)	(82)	(221)	(246)		
Operating income (loss)	\$	146	132	366	239		
Number of Events		7	8	18	16		

Primary Formula 1 revenue is derived from the commercial exploitation and development of the World Championship through a combination of race promotion fees (earned from granting the rights to host, stage and promote each Event on the World Championship calendar, fees from certain race promoters to license additional commercial rights from Formula 1 to secure Formula 2, Formula 3 and F1 Academy races at their Events, technical service fees from promoters to support the origination of program footage and ticketing revenue from Formula 1's direct promotion of the Las Vegas Grand Prix), media rights fees (earned from licensing the right to broadcast Events and Formula 2 and Formula 3 races on television and other platforms, F1 TV subscriptions and other related services, the origination of program footage, footage from Formula 1's archives and the licensing of radio broadcast and other ancillary media rights) and sponsorship fees (earned from the sale of World Championship and Event-related advertising and sponsorship rights and the servicing of such rights, rights to advertise on Formula 1's digital platforms and at non-Championship related events).

Primary Formula 1 revenue decreased \$32 million during the three months ended September 30, 2024, as compared to the corresponding period in the prior year, driven by the calendar variance, with one less Event held. Media rights revenue decreased during the three months ended September 30, 2024 due to the impact of the calendar variance on revenue recognition, partially offset by contractual increases in fees and continued growth in F1 TV subscription revenue. Sponsorship revenue decreased during the three months ended September 30, 2024 due to the impact of the calendar variance on revenue recognition and the different mix of Events held. Race promotion revenue increased during the three months ended September 30, 2024 due to the different mix of Events held and contractual increases in fees.

Primary Formula 1 revenue increased \$238 million during the nine months ended September 30, 2024, as compared to the corresponding period in the prior year, driven by the calendar variance, with two more Events held. Media rights revenue increased during the nine months ended September 30, 2024 due to the impact of the calendar variance on revenue recognition, contractual increases in fees and continued growth in F1 TV subscription revenue. Sponsorship revenue increased during the nine months ended September 30, 2024 driven by two additional Events in the period and fees from new sponsors. Race promotion revenue increased during the nine months ended September 30, 2024 driven by two more Events, contractual increases in fees and growing Sprint and F1 Academy fees.

Other Formula 1 revenue is generated from miscellaneous and ancillary sources primarily related to the sale of tickets to the Formula One Paddock Club at most Events, facilitating the shipment of cars and equipment to and from Events outside of Europe, the sale of hospitality and experiences at the Las Vegas Grand Prix, the operation of the Formula 2, Formula 3 and the new F1 Academy series, other licensing opportunities, various television production activities and other ancillary operations. Other Formula 1 revenue increased \$6 million during the three months ended September 30, 2024, as compared to the corresponding period in the prior year, driven by growth in licensing revenue and revenue generated from third-party events at the Grand Prix Plaza site in Las Vegas, partially offset by decreases in revenue from hospitality and experiences due to one less Event held. Other revenue increased \$55 million during the nine months ended September 30, 2024, as compared to the corresponding period in the prior year, driven by the impact of two more Events, resulting in higher hospitality and freight income. The increase during the nine months ended September 30, 2024 was also driven by the sale of new Formula 2 cars and associated parts at the beginning of the new Formula 2 vehicle cycle, higher licensing revenue and revenue generated from third-party events at the Grand Prix Plaza site in Las Vegas.

Cost of Formula 1 revenue

	Three months	ended	Nine months ended September 30,	
	 September	r 30,		
	 2024 2023		2024	2023
		amounts in m	nillions	
Team payments	\$ (371)	(432)	(969)	(888)
Other costs of Formula 1 revenue	(190)	(183)	(523)	(452)
Cost of Formula 1 revenue	\$ (561)	(615)	(1,492)	(1,340)

Cost of Formula 1 revenue decreased \$54 million and increased \$152 million during the three and nine months ended September 30, 2024, respectively, as compared to the corresponding periods in the prior year.

Team payments are recognized on a pro-rata basis across the Events of the World Championship calendar. Team payments decreased during the three months ended September 30, 2024 due to the pro rata recognition of expected team payments, with one less Event held in the current period. Team payments increased during the nine months ended September 30, 2024 due to the pro rata recognition of expected increased team payments.

Other costs of Formula 1 revenue are largely variable in nature and relate to both primary and other Formula 1 revenue. On an annual basis, the largest components of other costs of Formula 1 revenue are costs related to promoting, organizing and delivering the Las Vegas Grand Prix, hospitality costs, which are principally related to catering and other aspects of the production and delivery of hospitality offerings at the Las Vegas Grand Prix and the Paddock Club at other Events, and costs incurred in the provision and sale of freight, travel and logistical services. Other costs of Formula 1 revenue also include sponsorship and digital product sales' commissions, circuit rights' fees payable under various agreements with race promoters to acquire certain commercial rights at Events, including the right to sell advertising, hospitality and support race opportunities, annual Federation Internationale de l'Automobile ("FIA") regulatory fees, Formula 2 and Formula 3 cars, parts and maintenance services, costs related to the F1 Academy series, television production and post-production services, advertising production services and digital and social media activities. Other costs increased \$7 million during the three months ended September 30, 2024, as compared to the corresponding period in the prior year, primarily due to Las Vegas Grand Prix related lease costs, higher digital, F1 Academy and commissions and partner servicing costs associated with increased Primary Formula 1 revenue streams, partially offset by the impact of one less Event during the period, leading to lower F1A regulatory, technical, hospitality and travel costs. Other costs increased \$71 million during the nine months ended September 30, 2024, as compared to the corresponding period in the prior year, primarily due to higher commissions and partner servicing costs associated with increased Primary Formula 1 revenue streams, higher F1A

regulatory, digital, technical, hospitality, freight and travel costs related to two more Events in the current year period, higher costs of the F1 Academy series and Las Vegas Grand Prix related lease costs. The increase for the nine months ended September 30, 2024 was also driven by the costs of supplying the new Formula 2 cars and associated parts at the start of the new vehicle cycle.

Selling, general and administrative expenses include personnel costs, legal, professional and other advisory fees, bad debt expense, rental expense, information technology costs, insurance premiums, maintenance and utility costs and other general office administration costs. Selling, general and administrative expenses increased \$22 million and \$39 million during the three and nine months ended September 30, 2024, respectively, as compared to the corresponding periods in the prior year, primarily due to higher personnel, information technology, property, marketing, legal and other professional fee costs.

Stock-based compensation was flat during the both the three and nine months ended September 30, 2024, as compared to the corresponding periods in the prior year.

Depreciation and amortization includes depreciation of property and equipment and amortization of intangible assets. Depreciation and amortization decreased \$8 million and \$25 million during the three and nine months ended September 30, 2024, respectively, as compared to the corresponding periods in the prior year primarily due to a decrease in amortization expense related to certain intangible assets acquired in the acquisition of Formula 1 by Liberty.

Item 3. Quantitative and Qualitative Disclosures about Market Risk

We are exposed to market risk in the normal course of business due to our ongoing investing and financial activities. Market risk refers to the risk of loss arising from adverse changes in stock prices and interest rates. The risk of loss can be assessed from the perspective of adverse changes in fair values, cash flows and future earnings. We have established policies, procedures and internal processes governing our management of market risks and the use of financial instruments to manage our exposure to such risks.

We are exposed to changes in interest rates primarily as a result of our borrowing and investment activities, which include investments in fixed and floating rate debt instruments and borrowings used to maintain liquidity and to fund business operations. The nature and amount of our long-term and short-term debt are expected to vary as a result of future requirements, market conditions and other factors. We manage our exposure to interest rates by maintaining what we believe is an appropriate mix of fixed and variable rate debt. We believe this best protects us from interest rate risk. We have achieved this mix by (i) issuing fixed rate debt that we believe has a low stated interest rate and significant term to maturity, (ii) issuing variable rate debt with appropriate maturities and interest rates and (iii) entering into interest rate swap arrangements when we deem appropriate. As of September 30, 2024, our debt is comprised of the following amounts:

	variable rate debt			rixea i	ate debt
	rincipal amount	Weighted avg interest rate	Princi amou		Weighted avg interest rate
	 	dollar amounts	in millions	s	
Formula One Group	\$ 184	6.5 %	\$ 2	2,729	4.4 %
Liberty Live Group	NA	NA	\$ 1	,200	2.3 %

The Company is exposed to changes in stock prices primarily as a result of our significant holdings in publicly traded securities. We continually monitor changes in stock markets, in general, and changes in the stock prices of our holdings, specifically. We believe that changes in stock prices can be expected to vary as a result of general market conditions, technological changes, specific industry changes and other factors. We periodically use equity collars and other financial instruments to manage market risk associated with certain investment positions. These instruments are recorded at fair value based on option pricing models and other appropriate methods.

Additionally, our stock in Live Nation (one of our equity method affiliates), a publicly traded security, is not reflected at fair value in our balance sheet. This security is also subject to market risk that is not directly reflected in our condensed consolidated statement of operations, and had the market price of such security been 10% lower at September 30, 2024 the aggregate value of such security would have been \$763 million lower.

Item 4. Controls and Procedures

In accordance with Rules 13a-15 and 15d-15 under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), the Company carried out an evaluation, under the supervision and with the participation of management, including its chief executive officer and principal accounting and financial officer (the "Executives"), of the effectiveness of its disclosure controls and procedures as of the end of the period covered by this Quarterly Report. Based on that evaluation, the Executives concluded that the Company's disclosure controls and procedures were effective as of September 30, 2024 to provide reasonable assurance that information required to be disclosed in its reports filed or submitted under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the Securities and Exchange Commission's rules and forms.

There has been no change in the Company's internal control over financial reporting that occurred during the three months ended September 30, 2024 that has materially affected, or is reasonably likely to materially affect, its internal control over financial reporting.

PART II—OTHER INFORMATION

Item 1. Legal Proceedings

Our Annual Report on Form 10-K for the year ended December 31, 2023 includes "Legal Proceedings" under Item 3 of Part I. Refer to note 9 in the accompanying notes to the condensed consolidated financial statements for changes in the legal proceedings described in the Form 10-K.

Item 1A. Risk Factors

Except as discussed below, there have been no material changes in our risk factors from those disclosed in Part 1, Item 1A of our Annual Report on Form 10-K for the year ended December 31, 2023 (the "2023 Form 10-K"), which Risk Factors are incorporated by reference into this Quarterly Report on Form 10-Q.

As a result of the completion of the Liberty Sirius XM Holdings Split-Off, all of the risk factors under the headings "Risks Relating to the Liberty Sirius XM Group" and "Risks Relating to the proposed Liberty Sirius XM Holdings Split-Off and Merger" are no longer applicable as they relate to the Liberty Sirius XM Group. In addition, certain of the risk factors under the headings "Risks Relating to Our Company, as a Whole" and "Risks Relating to the Ownership of Our Common Stock Due to Our Tracking Stock Capitalization" are hereby replaced in their entirety as set forth below.

Risks Relating to our Company, as a Whole

The historical financial information of the Formula One Group and the Liberty Live Group included in this Quarterly Report on Form 10-Q may not necessarily reflect their results had they been separate companies.

One of the reasons for the creation of a tracking stock is to permit equity investors to apply more specific criteria in valuing the shares of a particular group, such as comparisons of earnings multiples with those of other companies in the same business sector. In valuing shares of Liberty Formula One common stock and Liberty Live common stock, investors should recognize that the historical financial information of the Formula One Group and the Liberty Live Group has been extracted from our consolidated financial statements and may not necessarily reflect what the Formula One Group's and the Liberty Live Group's results of operations, financial condition and cash flows would have been had each of these groups been separate, stand-alone entities pursuing independent strategies during the periods presented.

We may have future capital needs and may not be able to obtain additional financing on acceptable terms.

As of September 30, 2024, we had outstanding corporate-level indebtedness in the principal amount of \$1.7 billion. At September 30, 2024, our only wholly owned consolidated subsidiary is Formula 1. Our ability to access the cash flow of Formula 1 is subject to covenant restrictions set forth in the debt instruments of certain subsidiaries of Delta Topco, the parent company of Formula 1. Accordingly, our ability to obtain significant financing in the future, on favorable terms or at all, may be limited. If debt financing is not available to us in the future, we may obtain liquidity through the sale or monetization of our debt or equity securities, or we may issue equity securities. If additional funds are raised through the issuance of equity securities, our stockholders may experience significant dilution. If we are unable to obtain sufficient liquidity in the future, we may be unable to develop our businesses properly, complete acquisitions or otherwise take advantage of business opportunities or respond to competitive pressures, any of which could have a material adverse effect on our business, financial condition and results of operations and those attributed to our groups.

A substantial portion of our consolidated debt is held above the operating subsidiary level, and we could be unable in the future to obtain cash in amounts sufficient to service that debt and our other financial obligations.

As of September 30, 2024, we had approximately \$1.7 billion principal amount of corporate-level debt outstanding, consisting of \$475 million outstanding under our 2.25% Convertible Senior Notes due 2027, \$50 million outstanding under our 0.5% Exchangeable Senior Debentures due 2050, \$1.15 billion outstanding under our 2.375% Exchangeable Senior Debentures due 2053 and \$54 million of other obligations. Our ability to meet our financial obligations will depend on our ability to access cash. Our primary sources of cash include our available cash balances, dividends and interest from our

investments, monetization of our public investment portfolio and proceeds from asset sales. Further, our ability to receive dividends or payments or advances from our businesses depends on their individual operating results, any statutory, regulatory or contractual restrictions to which they may be or may become subject and the terms of their own indebtedness, including Formula 1's subsidiary debt. The agreements governing such indebtedness restrict sales of assets and prohibit or limit the payment of dividends or the making of distributions, loans or advances to stockholders, non-wholly owned subsidiaries or our partners. We generally do not receive cash, in the form of dividends, loans, advances or otherwise, from any of our subsidiaries or business affiliates.

The success of businesses attributed to each of our tracking stock groups, in part, depends on their popularity with audiences, which is difficult to predict.

Live entertainment events, including sporting events, are inherently risky businesses because the revenue derived from these businesses depends primarily upon their popularity with public audiences, which is difficult to predict. The commercial success of live entertainment depends upon the quality and acceptance of competing programs, the availability of alternative forms of entertainment and leisure time activities, general economic conditions and other tangible and intangible factors, many of which are difficult to predict. In the case of sponsorship agreements, audience size is an important factor when rates are negotiated. Audience size is an important factor when determining ticket pricing for live entertainment events and the value of broadcast rights. Consequently, low public acceptance of the services and events provided by companies such as Formula 1 and Live Nation could hurt the ability of these companies to maintain or grow revenue, which would adversely impact the financial performance of the groups to which these companies are attributed.

Our businesses attributed to the Formula One Group and the Liberty Live Group, such as Formula 1 and Live Nation, may not realize the benefits of acquisitions or other strategic investments and initiatives.

Our business strategy and that of our subsidiaries and business affiliates, including Formula 1 and Live Nation, may include selective acquisitions, other strategic investments and initiatives that allow them to expand their business. The success of any acquisition depends upon effective integration and management of acquired businesses and assets into the acquirer's operations, which is subject to risks and uncertainties, including the realization of the growth potential, any anticipated synergies and cost savings, the ability to retain and attract personnel, the diversion of management's attention from other business concerns and undisclosed or potential legal liabilities of acquired businesses or assets.

Weak and uncertain economic conditions may reduce consumer demand for products, services and events offered by our businesses attributed to each of our groups.

A weak or uncertain economy in the U.S. or, in the case of the Formula One Group, abroad, could adversely affect demand for our products, services and events. A substantial portion of our revenue is derived from discretionary spending by individuals, which typically falls during times of economic recession or instability. A reduction in discretionary spending could adversely affect revenue through reduced live-entertainment and sporting event expenditures. Accordingly, the ability of our businesses attributed to each of our groups to increase or maintain revenue and earnings could be adversely affected to the extent that relevant economic environments remain weak or decline further. In addition, inflation, which has significantly risen, may increase operational costs, including labor costs, and continued increases in interest rates in response to concerns about inflation may have the effect of further increasing economic uncertainty and heightening these risks. We currently are unable to predict the extent of any of these potential adverse effects.

Our Company has overlapping directors and management with Qurate Retail, Liberty Broadband and TripCo, which may lead to conflicting interests.

As a result of transactions between 2011 and 2014 that resulted in the separate corporate existence of our Company, Qurate Retail, Liberty Broadband and TripCo, all or most of the executive officers of Liberty also serve as executive officers of Qurate Retail, Liberty Broadband and TripCo, and there are overlapping directors at each of Qurate Retail, Liberty Broadband and TripCo. Our executive officers and members of the Board of Directors have fiduciary duties to our stockholders. Likewise, any such persons who serve in similar capacities at Qurate Retail, Liberty Broadband and TripCo have fiduciary duties to that company's stockholders. For example, there may be the potential for a conflict of interest when our Company, Qurate Retail, Liberty Broadband or TripCo pursues acquisitions and other business opportunities that

may be suitable for each of them. Therefore, such persons may have conflicts of interest or the appearance of conflicts of interest with respect to matters involving or affecting more than one of the companies to which they owe fiduciary duties. Moreover, most of our Company's directors and officers continue to own Qurate Retail, Liberty Broadband and/or TripCo stock and options to purchase stock in those companies. These ownership interests could create, or appear to create, potential conflicts of interest when the applicable individuals are faced with decisions that could have different implications for our Company, Qurate Retail, Liberty Broadband and/or TripCo. Any potential conflict that qualifies as a "related party transaction" (as defined in Item 404 of Regulation S-K under the Securities Act of 1933, as amended) is subject to review by an independent committee of the applicable issuer's board of directors in accordance with its corporate governance guidelines. Each of Liberty Broadband and TripCo has renounced its rights to certain business opportunities and its respective restated certificate of incorporation contains provisions deeming directors and officers not in breach of their fiduciary duties in certain cases for directing a corporate opportunity to another person or entity (including our Company, Qurate Retail, Liberty Broadband and TripCo) instead of such company. Other potential conflicts that arise will be addressed on a case-by-case basis, keeping in mind the applicable fiduciary duties owed by the executive officers and directors of each issuer. From time to time, we may enter into transactions with Qurate Retail, Liberty Broadband, TripCo and/or their respective subsidiaries or other affiliates. There can be no assurance that the terms of any such transactions will be as favorable to our Company, Qurate Retail, Liberty Broadband, TripCo or any of their respective subsidiaries or affiliates as would be the case where there is no overlapping officer or director.

We may be subject to significant tax liabilities related to the Liberty Sirius XM Holdings Split-Off.

In connection with the Liberty Sirius XM Holdings Split-Off, we received an opinion of our tax counsel to the effect that, for U.S. federal income tax purposes, the Liberty Sirius XM Holdings Split-Off will qualify as a generally tax-free transaction under Section 355, Section 368(a)(1)(D) and related provisions of the Internal Revenue Code of 1986, as amended (the "Code") to Liberty and to former holders of Liberty SiriusXM common stock. We did not obtain a private letter ruling from the Internal Revenue Service (the "IRS") regarding the U.S. federal income tax treatment of the Liberty Sirius XM Holdings Split-Off. Opinions of counsel are not binding on the IRS or the courts, and there can be no assurance that the IRS will not challenge the conclusions reached in such opinions or that a court would not sustain such a challenge. If it is determined that the Liberty Sirius XM Holdings Split-Off does not qualify under Section 355, Section 368(a)(1)(D) and related provisions of the Code, we and the former holders of Liberty SiriusXM common stock who received common stock of Liberty Sirius XM Holdings in the Liberty Sirius XM Holdings Split-Off could incur significant tax liabilities.

Even if the Liberty Sirius XM Holdings Split-Off otherwise qualifies under Section 355, Section 368(a)(1)(D), and related provisions of the Code, the Liberty Sirius XM Holdings Split-Off would result in a significant U.S. federal income tax liability to us (but not to former holders of Liberty SiriusXM common stock) under Section 355(e) of the Code if one or more persons acquire, directly or indirectly, a 50% or greater interest (measured by vote or value) in the stock of our Company or in the stock of Liberty Sirius XM Holdings (or any successor corporation) as part of a plan or series of related transactions that includes the Liberty Sirius XM Holdings Split-Off. The process for determining whether an acquisition is part of a plan under these rules is complex, inherently factual in nature, and subject to a comprehensive analysis of the facts and circumstances of the particular case. Notwithstanding the opinion of tax counsel described above, we or Liberty Sirius XM Holdings might inadvertently cause or permit a prohibited change in our or Liberty Sirius XM Holdings' ownership to occur, thereby triggering tax liability to us.

Prior to the Liberty Sirius XM Holdings Split-Off, we entered into a tax sharing agreement with Liberty Sirius XM Holdings. Under this agreement, our Company is generally responsible for taxes and losses resulting from the Liberty Sirius XM Holdings Split-Off; however, Liberty Sirius XM Holdings is required to indemnify us for any taxes and losses (other than any taxes or tax-related losses that result from Section 355(e) of the Code applying to the Liberty Sirius XM Holdings Split-Off as a result of the Liberty Sirius XM Holdings Split-Off being part of a plan (or series of related transactions) pursuant to which one or more persons acquire a 50-percent or greater interest (measured by vote or value) in the stock of our Company) resulting from the failure of the Liberty Sirius XM Holdings Split-Off to qualify as a generally tax-free transaction under Section 355, Section 368(a)(1)(D) and related provisions of the Code, to the extent such taxes or losses (i) result primarily from, individually or in the aggregate, the breach of certain covenants made by Liberty Sirius XM Holdings (applicable to actions or failures to act by Liberty Sirius XM Holdings and its subsidiaries following the completion of the Liberty Sirius XM Holdings Split-Off), (ii) result primarily from, individually or in the aggregate, the failure of certain representations made by Sirius XM Holdings in support of the opinion of our tax counsel regarding the

generally tax-free status of the Liberty Sirius XM Holdings Split-Off to be true and correct, or (iii) result from the application of Section 355(e) of the Code to the Liberty Sirius XM Holdings Split-Off as a result of the treatment of the Liberty Sirius XM Holdings Split-Off as part of a plan (or series of related transactions) pursuant to which one or more persons acquire, directly or indirectly, a 50% or greater interest (measured by vote or value) in the stock of Liberty Sirius XM Holdings (or any successor corporation), except, in the case of clauses (i) and (ii), if such taxes and losses result from an action required to be taken pursuant to the transaction agreements relating to the Liberty Sirius XM Holdings Split-Off. As the taxpaying entity, however, we are subject to the risk of non-payment by Liberty Sirius XM Holdings of its indemnification obligations under the tax sharing agreement. Cash for the payment of any taxes and losses resulting from the Liberty Sirius XM Holdings Split-Off which are not allocated to and paid by Liberty Sirius XM Holdings pursuant to our tax sharing agreement with Liberty Sirius XM Holdings generally would be drawn from funds attributed to the Liberty Live Group, except that cash for the payment of any taxes and losses (a) resulting primarily from a breach by us following the Liberty Sirius XM Holdings Split-Off of any contractual covenants made by us in connection with the Liberty Sirius XM Holdings Split-Off or (b) resulting from Section 355(e) of the Code applying to the Liberty Sirius XM Holdings Split-Off as a result of the Liberty Sirius XM Holdings Split-Off being part of a plan (or series of related transactions) pursuant to which one or more persons acquire a 50% or greater interest (measured by vote or value) in the stock of our Company shall, in each case, be drawn proportionately from funds attributed to the Formula One Group and the Liberty Live Group based upon the relative market capitalizations of the tracking stock of each group over the first three trading days following completion of the Liberty Sirius XM Holdings Split-Off, as determined in accordance with our tax sharing policies.

To preserve the tax-free treatment of the Liberty Sirius XM Holdings Split-Off, we may determine to forgo certain transactions that might have otherwise been advantageous to our Company, including certain asset dispositions or other strategic transactions for some period of time following the Liberty Sirius XM Holdings Split-Off. In addition, our potential tax liabilities related to the Liberty Sirius XM Holdings Split-Off might discourage, delay or prevent a change of control transaction for some period of time following the Liberty Sirius XM Holdings Split-Off.

Formula 1 and Live Nation have been, and may in the future be, materially impacted by a pandemic or epidemic, such as COVID-19.

Although Formula 1 and Live Nation saw a return to normal business operations, schedules and events following the COVID-19 pandemic, it is unclear whether and to what extent a future pandemic or epidemic will impact the use of and/or demand for the entertainment, events and services provided by these businesses and demand for sponsorship and advertising assets. If these businesses face cancelled events, closed venues and reduced attendance, as was the result of the COVID-19 pandemic, the impact may substantially decrease our revenue. For example, due to the revenue reductions caused by COVID-19 in 2020 and 2021, these businesses looked to reduce expenses, but should such impacts resume, the businesses may not be able to reduce expenses to the same degree as any decline in revenue, which may adversely affect our results of operations and cash flow.

In addition, our businesses are particularly sensitive to reductions in travel and discretionary consumer spending. We cannot predict the time period over which our businesses would be impacted by a future pandemic or epidemic. Over the long-term, a future pandemic or epidemic could impede economic activity in impacted regions or globally, causing a global recession, leading to a further decline in discretionary spending on sports and entertainment events and other leisure activities, which could result in long-term effects on our businesses.

For the reasons set forth above and other reasons that may come to light as a result of a future pandemic or epidemic, we cannot reasonably estimate the impact to our future revenue, results of operations, cash flows or financial condition, but such impacts have been, and may in the future be, significant and could have a material adverse effect on our business, revenue, results of operations, cash flows and financial condition.

Risks Relating to the Ownership of Our Common Stock Due to Our Tracking Stock Capitalization

The following risks relate to the ownership of our common stock due to our tracking stock capitalization.

Holders of Liberty Formula One common stock and Liberty Live common stock are common stockholders of our Company and, therefore, are subject to risks associated with an investment in our Company as a whole, even if a holder does not own shares of common stock of all of our groups.

Even though we have attributed, for financial reporting purposes, all of our consolidated assets, liabilities, revenue, expenses and cash flows among the Formula One Group and the Liberty Live Group in order to prepare the separate financial statement schedules for each of those groups, we will retain legal title to all of our assets and our tracking stock capitalization does not limit our legal responsibility, or that of our subsidiaries, for the liabilities included in any set of financial statement schedules. Holders of Liberty Formula One common stock and Liberty Live common stock do not have any legal rights related to specific assets attributed to their associated group and, in any liquidation, holders of Liberty Formula One common stock and holders of Liberty Live common stock will be entitled to receive a pro rata share of our available net assets based on their respective numbers of liquidation units.

Possible market confusion may result from holders of our tracking stocks mistakenly believing that they (i) directly own stock of a company that is attributed to one of our tracking stocks and (ii) have any equity or voting interests with respect to companies attributed to one of our tracking stocks.

Our company holds interests in various companies, including public companies, and these interests are attributed to our tracking stock groups. In particular, following the Reclassification, the Formula One Group is comprised of our subsidiary that owns the Formula 1 business and the Liberty Live Group is primarily comprised of our interest in Live Nation. Depending on the composition of the assets underlying our tracking stock groups from time to time, confusion in the marketplace may occur if holders of our tracking stock mistakenly believe they own stock of a company attributed to the applicable tracking stock group. As described above, holders of Liberty Formula One common stock and Liberty Live common stock do not have any legal rights related to specific assets attributed to their associated tracking stock group. Similarly, holders of these tracking stocks do not, by virtue of their ownership of our tracking stock, own any equity or voting interest in any company attributed to one of our tracking stock groups, including any public companies.

The market price of Liberty Formula One common stock and Liberty Live common stock may not reflect the performance of the businesses and assets attributed to the Formula One Group and the Liberty Live Group, respectively, as we intend.

We cannot assure you that the market price of the common stock related to a group will, in fact, reflect the performance of the group of businesses, assets and liabilities attributed to that group. Holders of Liberty Formula One common stock and Liberty Live common stock are common stockholders of our Company as a whole and, as such, are subject to all risks associated with an investment in our Company and all of our businesses, assets and liabilities. As a result, the market price of each tracking stock may, in part, reflect events that are intended to be reflected or tracked by a different tracking stock of our Company. In addition, investors may discount the value of the stock related to a group because it is part of a common enterprise rather than a stand-alone entity.

The market price of Liberty Formula One common stock and Liberty Live common stock may be volatile, could fluctuate substantially and could be affected by factors that do not affect traditional common stock.

The market prices of Liberty Formula One common stock and Liberty Live common stock and may be materially affected by, among other things:

- actual or anticipated fluctuations in a group's operating results or in the operating results of particular companies attributable to such group;
- potential acquisition activity by our Company (regardless of the group to which it is attributed) or the companies in which we invest:

- issuances of debt or equity securities to raise capital by our Company or the companies in which we invest and the manner in which that debt or the proceeds of an equity issuance are attributed to each of the groups;
- changes in financial estimates by securities analysts regarding Liberty Formula One common stock or Liberty Live common stock or the companies attributable to our tracking stock groups;
- the complex nature and the potential difficulties investors may have in understanding the terms of our three tracking stocks,
 as well as concerns regarding the possible effect of certain of those terms on an investment in our stocks; and
- general market conditions.

The market value of Liberty Formula One common stock or Liberty Live common stock could be adversely affected by events involving the assets and businesses attributed to one or more of the other groups.

Because we are the issuer of Liberty Formula One common stock and Liberty Live common stock, an adverse market reaction to events relating to the assets and businesses attributed to one of our groups, such as earnings announcements, announcements of new products or services or acquisitions or dispositions that the market does not view favorably, may cause an adverse market reaction in the common stock of the other groups. This could occur even if the triggering event is not material to us as a whole. Certain events may also have a greater impact on one group than the same triggering event would have on another group due to the asset composition of the affected group. In addition, the incurrence of significant indebtedness by us or any of our subsidiaries on behalf of one group, including indebtedness incurred or assumed in connection with acquisitions of or investments in businesses, could affect our credit rating and that of our subsidiaries and, therefore, could increase the borrowing costs of businesses attributable to our other groups or the borrowing costs of our Company as a whole.

We may not pay dividends equally or at all on Liberty Formula One common stock or Liberty Live common stock.

We do not presently intend to pay cash dividends on Liberty Formula One common stock or Liberty Live common stock for the foreseeable future. However, we have the right to pay dividends on the shares of common stock related to each group in equal or unequal amounts, and we may pay dividends on the shares of common stock related to one group and not pay dividends on shares of common stock related to another group. In addition, any dividends or distributions on, or repurchases of, shares relating to a group will reduce our assets legally available to be paid as dividends on the shares relating to another group.

Our tracking stock capital structure could create conflicts of interest, and our Board of Directors may make decisions that could adversely affect only some holders of our common stock.

Our tracking stock capital structure could give rise to occasions when the interests of holders of stock related to one group might diverge or appear to diverge from the interests of holders of stock related to one or both of the other groups. In addition, given the nature of their businesses, there may be inherent conflicts of interests between the Formula One Group and the Liberty Live Group. Our tracking stock groups are not separate entities and thus holders of Liberty Formula One common stock and Liberty Live common stock do not have the right to elect separate boards of directors. As a result, our Company's officers and directors owe fiduciary duties to our Company as a whole and all of our stockholders as opposed to only holders of a particular group. Decisions deemed to be in the best interest of our Company and all of our stockholders may not be in the best interest of a particular group or groups when considered independently. Examples include:

- decisions as to the terms of any business relationships that may be created between the tracking stock groups;
- the terms of any reattributions of assets between one or more groups;
- decisions as to the allocation of consideration among the holders of Liberty Formula One common stock and Liberty Live common stock, or among the series of stocks relating to our groups, to be received in connection with a merger involving our Company;
- decisions as to the allocation of corporate opportunities between the groups, especially where the opportunities might meet the strategic business objectives of more than one group;

- decisions as to operational and financial matters that could be considered detrimental to one or more groups but beneficial to another:
- decisions as to the conversion of shares of common stock of one group into shares of common stock of another;
- decisions regarding the creation of, and, if created, the subsequent increase or decrease of any inter-group interest that one group may own in another group;
- decisions as to the internal or external financing attributable to businesses or assets attributed to any of our groups;
- · decisions as to the dispositions of assets of any of our groups; and
- decisions as to the payment of dividends on the stock relating to any of our groups.

Our directors' or officers' equity ownership may create or appear to create conflicts of interest.

If directors or officers own disproportionate interests (in percentage or value terms) in Liberty Formula One common stock or Liberty Live common stock, that disparity could create or appear to create conflicts of interest when they are faced with decisions that could have different implications for the holders of Liberty Formula One common stock or Liberty Live common stock.

Other than pursuant to our management and allocation policies, we have not adopted any specific procedures for consideration of matters involving a divergence of interests among holders of shares of stock relating to our two groups, or among holders of different series of stock relating to a specific group.

Rather than develop additional specific procedures in advance, the Board of Directors intends to exercise its judgment from time to time, depending on the circumstances, as to how best to:

- obtain information regarding the divergence (or potential divergence) of interests;
- determine under what circumstances to seek the assistance of outside advisers;
- determine whether a committee of the Board of Directors should be appointed to address a specific matter and the appropriate members of that committee; and
- assess what is in our best interests and the best interests of all of our stockholders.

The Board of Directors believes the advantage of retaining flexibility in determining how to fulfill its responsibilities in any such circumstances as they may arise outweighs any perceived advantages of adopting additional specific procedures in advance.

Our Board of Directors may change the management and allocation policies to the detriment of one or more groups without stockholder approval.

The Board of Directors has adopted certain management and allocation policies to serve as guidelines in making decisions regarding the relationship between the Formula One Group and the Liberty Live Group with respect to matters such as tax liabilities and benefits, inter-group loans, inter-group interests, attribution of assets, financing alternatives, corporate opportunities and similar items. These policies also set forth the initial focuses and strategies of these groups and the initial attribution of our businesses, assets and liabilities among them. These policies are not included in the current Charter. The Board of Directors may at any time change or make exceptions to these policies. Because these policies relate to matters concerning the day-to-day management of our Company as opposed to significant corporate actions, such as a merger involving our Company or a sale of substantially all of our assets, no stockholder approval is required with respect to their adoption or amendment. A decision to change, or make exceptions to, these policies or adopt additional policies could disadvantage one or more groups while advantaging the other(s).

Stockholders will not vote on how to attribute consideration received in connection with a merger involving our Company among holders of Liberty Formula One common stock and Liberty Live common stock.

Our current Charter does not contain any provisions governing how consideration received in connection with a merger or consolidation involving our Company is to be attributed to holders of Liberty Formula One common stock and holders of Liberty Live common stock or to the holders of different series of stock, and none of the holders of Liberty Formula One common stock or Liberty Live common stock will have a separate class vote in the event of such a merger or consolidation. Consistent with applicable principles of Delaware law, the Board of Directors will seek to divide the type and amount of consideration received in a merger or consolidation involving our Company among holders of Liberty Formula One common stock and Liberty Live common stock in a fair manner. As the different ways the Board of Directors may divide the consideration between holders of stock relating to the different groups, and among holders of different series of a particular stock, might have materially different results, the consideration to be received by holders of Liberty Formula One common stock and Liberty Live common stock in any such merger or consolidation may be materially less valuable than the consideration they would have received if they had a separate class vote on such merger or consolidation.

We may dispose of assets of the Formula One Group or the Liberty Live Group without stockholder approval.

Delaware law requires stockholder approval only for a sale or other disposition of all or substantially all of the assets of our Company taken as a whole, and our current Charter does not require a separate class vote in the case of a sale of a significant amount of assets of any of our groups. As long as the assets attributed to the Formula One Group or the Liberty Live Group proposed to be disposed of represent less than substantially all of our assets, we may approve sales and other dispositions of any amount of the assets of such group without any stockholder approval.

If we dispose of all or substantially all of the assets attributed to any group (which means, for this purpose, assets representing 80% of the fair market value of the total assets of the disposing group, as determined by the Board of Directors), we would be required under the terms of our current Charter, if the disposition is not an exempt disposition under the terms of our current charter, to choose one or more of the following three alternatives:

- declare and pay a dividend on the disposing group's common stock;
- redeem shares of the disposing group's common stock in exchange for cash, securities or other property; and/or
- convert all or a portion of the disposing group's outstanding common stock into common stock of another group.

In this type of a transaction, holders of the disposing group's common stock may receive less value than the value that a third-party buyer might pay for all or substantially all of the assets of the disposing group.

The Board of Directors will decide, in its sole discretion, how to proceed and is not required to select the option that would result in the highest value to holders of any stock related to a particular group.

Holders of Liberty Formula One common stock or Liberty Live common stock may receive less consideration upon a sale of the assets attributed to that group than if that group were a separate company.

If the Formula One Group or the Liberty Live Group were a separate, independent company and its shares were acquired by another person, certain costs of that sale, including corporate level taxes, might not be payable in connection with that acquisition. As a result, stockholders of a separate, independent company with the same assets might receive a greater amount of proceeds than the holders of Liberty Formula One common stock or Liberty Live common stock would receive upon a sale of all or substantially all of the assets of the group to which their shares relate. In addition, we cannot assure you that in the event of such a sale the per share consideration to be paid to holders of Liberty Formula One common stock or Liberty Live common stock, as the case may be, will be equal to or more than the per share value of that share of stock prior to or after the announcement of a sale of all or substantially all of the assets of the applicable group. Further, there is no requirement that the consideration paid be tax-free to the holders of the shares of common stock related to that group. Accordingly, if we sell all or substantially all of the assets attributed to the Formula One Group or the Liberty Live Group, our stockholders could suffer a loss in the value of their investment in our stock.

In the event of a liquidation of Liberty, holders of Liberty Formula One common stock and Liberty Live common stock will not have a priority with respect to the assets attributed to the related tracking stock group remaining for distribution to stockholders.

Under our current Charter, upon Liberty's liquidation, dissolution or winding up, holders of Liberty Formula One common stock and Liberty Live common stock will be entitled to receive, in respect of their shares of such stock, their proportionate interest in all of Liberty's assets, if any, remaining for distribution to holders of common stock in proportion to their respective number of "liquidation units" per share, as previously determined in connection with the Reclassification. Hence, the assets to be distributed to a holder of any of our tracking stocks upon a liquidation, dissolution or winding up of Liberty will have nothing to do with the value of the assets attributed to the related tracking stock group or to changes in the relative value of Liberty Formula One common stock and Liberty Live common stock over time.

Holders of Liberty Formula One common stock and Liberty Live common stock vote together and have limited separate voting rights.

Holders of Series A and Series B Liberty Formula One common stock and Liberty Live common stock vote together as a single class, except in certain limited circumstances prescribed by our current Charter and under Delaware law. Each share of Series B common stock of each group has ten votes per share, and each share of Series A common stock of each group has one vote per share. Holders of Series C common stock of each group have no voting rights, other than those required under Delaware law. When holders of Liberty Formula One common stock and Liberty Live common stock vote together as a single class, holders having a majority of the votes are in a position to control the outcome of the vote even if the matter involves a conflict of interest among our stockholders or has a greater impact on one group than another.

Transactions in Liberty Formula One common stock and Liberty Live common stock by our insiders could depress the market price of those stocks.

Sales of, or hedging transactions such as collars relating to, shares of Liberty Formula One common stock or Liberty Live common stock by our Chairman of the Board of Directors, or any of our other directors or executive officers, could cause a perception in the marketplace that the stock price of the relevant shares has peaked or that adverse events or trends have occurred or may be occurring at our Company or the group to which the shares relates. This perception can result notwithstanding any personal financial motivation for these transactions. As a result, insider transactions could depress the market price for shares of the Liberty Formula One common stock or Liberty Live common stock.

Our capital structure, as well as the fact that the Formula One Group and the Liberty Live Group are not independent companies, may inhibit or prevent acquisition bids for the businesses attributed to the Formula One Group or the Liberty Live Group and may make it difficult for a third party to acquire us, even if doing so may be beneficial to our stockholders.

If the Formula One Group and the Liberty Live Group were separate independent companies, any person interested in acquiring the Formula One Group or the Liberty Live Group without negotiating with management could seek control of that group by obtaining control of its outstanding voting stock, by means of a tender offer or a proxy contest. Although we intend Liberty Formula One common stock and Liberty Live common stock to reflect the separate economic performance of the Formula One Group and the Liberty Live Group, respectively, those groups are not separate entities and a person interested in acquiring only one group without negotiation with our management could obtain control of that group only by obtaining control of a majority in voting power of all of the outstanding voting shares of our Company. The existence of shares of common stock, and different series of shares, relating to different groups could present complexities and in certain circumstances pose obstacles, financial and otherwise, to an acquiring person that are not present in companies that do not have a capital structure similar to ours.

Certain provisions of our current Charter and bylaws may discourage, delay or prevent a change in control of our Company that a stockholder may consider favorable. These provisions include:

• authorizing a capital structure with multiple series of common stock: a Series B common stock related to each group that entitles the holders to ten votes per share, a Series A common stock related to each group that

entitles the holder to one vote per share, and a Series C common stock related to each group that, except as otherwise required by Delaware law, entitles the holder to no voting rights;

- classifying the Board of Directors with staggered three-year terms, which may lengthen the time required to gain control of the Board of Directors;
- limiting who may call special meetings of stockholders;
- prohibiting stockholder action by written consent, thereby requiring all stockholder actions to be taken at a meeting of the stockholders;
- establishing advance notice requirements for nominations of candidates for election to the Board of Directors or for proposing matters that can be acted upon by stockholders at stockholder meetings;
- requiring stockholder approval by holders of at least 66%% of our aggregate voting power or the approval by at least 75% of the Board of Directors with respect to certain extraordinary matters, such as a merger or consolidation of our Company, a sale of all or substantially all of our assets or an amendment to our current Charter; and
- the existence of authorized and unissued stock, including "blank check" preferred stock, which could be issued by the Board
 of Directors to persons friendly to our then current management, thereby protecting the continuity of our management, or
 which could be used to dilute the stock ownership of persons seeking to obtain control of our Company.

Liberty's Chairman, John C. Malone, beneficially owns shares (based on outstanding share information as of September 30, 2024) representing the power to direct approximately 49% of the aggregate voting power in Liberty, due to his beneficial ownership of approximately 97% of the outstanding shares of each of the Series B Liberty Formula One common stock and the Series B Liberty Live common stock.

Item 2. Unregistered Sales of Equity Securities, Use of Proceeds, and Issuer Purchases of Equity Securities

Share Repurchase Programs

In November 2019, our board of directors authorized the repurchase of \$1 billion of the Company's common stock. In May 2022, our board of directors authorized the repurchase of an additional \$1 billion of the Company's common stock.

There were no repurchases of Liberty SiriusXM common stock, Liberty Formula One common stock or Liberty Live common stock during the three months ended September 30, 2024. As of September 30, 2024, approximately \$1.1 billion was available for future share repurchase under our share repurchase program.

During the three months ended September 30, 2024, no shares of Liberty Formula One common stock or Liberty Live common stock, no shares of Series A Liberty SiriusXM common stock and 2,450 shares of Series C Liberty SiriusXM common stock were surrendered by our officers and employees to pay withholding taxes and other deductions in connection with the vesting of their restricted stock, restricted stock units and options.

Item 5. Other Information

None of the Company's directors or officers adopted or terminated a Rule 10b5-1 trading arrangement or a non-Rule 10b5-1 trading arrangement during the Company's fiscal quarter ended September 30, 2024.

Item 6. Exhibits

(a) Exhibits

Listed below are the exhibits which are filed as a part of this Quarterly Report (according to the number assigned to them in Item 601 of Regulation S-K):

Exhibit No. Name 3.1 Amended and Restated Bylaws* 10.1 Amendment Agreement, dated September 19, 2024, by and among Formula One Management Limited, J.P. Morgan SE, as facility agent, and other financial institutions party thereto* 10.2 2nd Amendment Agreement, dated October 9, 2024, by and between Formula One Management Limited and J.P. Morgan SE, as facility agent* 10.3 Amended and Restated First Lien Facilities Agreement, dated November 23, 2022, by and among Formula One Management Limited, J.P. Morgan SE, as facility agent, NatWest Markets plc and other financial institutions party thereto, conformed to reflect amendments through October 9, 2024* 31.1 Rule 13a-14(a)/15d-14(a) Certification* 31.2 Rule 13a-14(a)/15d-14(a) Certification* 32 Section 1350 Certification** 99.1 Unaudited Attributed Financial Information for Tracking Stock Groups* 101.INS Inline XBRL Instance Document* - The instance document does not appear in the interactive data file because its XBRL tags are embedded within the inline XBRL document. 101.SCH Inline XBRL Taxonomy Extension Schema Document* 101.CAL Inline XBRL Taxonomy Calculation Linkbase Document* 101.LAB Inline XBRL Taxonomy Label Linkbase Document* 101.PRE Inline XBRL Taxonomy Presentation Linkbase Document* 101.DEF Inline XBRL Taxonomy Definition Document*

104 Cover Page Interactive Data File (formatted as Inline XBRL and contained in Exhibit 101)

^{*} Filed herewith

^{**} Furnished herewith

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

LIBERTY MEDIA CORPORATION

Date: November 12, 2024 By: /s/ GREGORY B. MAFFEI

Gregory B. Maffei

President and Chief Executive Officer

Date: November 12, 2024 By: /s/ BRIAN J. WENDLING

Brian J. Wendling

Chief Accounting Officer and Principal Financial Officer

LIBERTY MEDIA CORPORATION

A Delaware Corporation AMENDED AND RESTATED BYLAWS

ARTICLE I

STOCKHOLDERS

Section 1.1 Annual Meeting.

An annual meeting of stockholders for the purpose of electing directors and of transacting any other business properly brought before the meeting pursuant to these Bylaws shall be held each year at such date, time and place, either within or without the State of Delaware or, if so determined by the Board of Directors in its sole discretion, at no place (but rather by means of remote communication), as may be specified by the Board of Directors in the notice of meeting.

Section 1.2 <u>Special Meetings</u>.

Except as otherwise provided in the terms of any series of preferred stock or unless otherwise provided by law or by the Certificate of Incorporation, special meetings of stockholders of the Corporation, for the transaction of such business as may properly come before the meeting, may be called by the Secretary of the Corporation (the "Secretary") only (i) upon written request received by the Secretary at the principal executive offices of the Corporation by or on behalf of the holder or holders of record of outstanding shares of capital stock of the Corporation, representing collectively not less than 66 2/3% of the total voting power of the outstanding capital stock of the Corporation entitled to vote at such meeting or (ii) at the request of not less than 75% of the members of the Board of Directors then in office. Only such business may be transacted as is specified in the notice of the special meeting. The Board of Directors shall have the sole power to determine the time, date and place, either within or without the State of Delaware, or, if so determined by the Board of Directors in its sole discretion, at no place (but rather by means of remote communication), for any special meeting of stockholders (including those properly called by the Secretary in accordance with Section 1.2(i) hereof). Following such determination, it shall be the duty of the Secretary to cause notice to be given to the stockholders entitled to vote at such meeting that a meeting will be held at the time, date and place, if any, and in accordance with the record date determined by the Board of Directors.

Section 1.3 Record Date.

In order that the Corporation may determine the stockholders entitled to notice of any meeting of stockholders or any adjournment thereof, the Board of Directors may fix, in advance, a record date, which shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall not be more than sixty (60) calendar days nor less than ten (10) calendar days before the date of such meeting. If the Board of Directors so fixes a record date for determining the stockholders entitled to notice of any meeting of stockholders, such date shall be the record date for determining the stockholders entitled to vote at such meeting, unless the Board of Directors determines, at the time it fixes the record date for determining the stockholders entitled to notice of such meeting, that a later date on or before the date of the meeting shall be the record date for determining

stockholders entitled to vote at such meeting. In order that the Corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action, the Board of Directors may fix, in advance, a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall not be more than sixty (60) calendar days prior to such action. If no record date is fixed by the Board of Directors: (i) the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held, and (ii) the record date for determining stockholders for any other purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution relating thereto. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board of Directors may fix a new record date for the adjourned meeting in accordance with this Section 1.3.

Section 1.4 Notice of Meetings.

Notice of all stockholders meetings, stating the place, if any, date and hour thereof, as well as the record date for determining stockholders entitled to vote at such meeting (if such record date is different from the record date for determining stockholders entitled to notice of the meeting); the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting; and, in the case of a special meeting, the purpose or purposes for which the meeting is called, shall be delivered by the Corporation in accordance with Section 5.4 of these Bylaws, applicable law and applicable stock exchange rules and regulations by the Chairman of the Board, the President, any Vice President, the Secretary or an Assistant Secretary, to each stockholder entitled to notice of such meeting, unless otherwise provided by applicable law or the Certificate of Incorporation, at least ten (10) calendar days but not more than sixty (60) calendar days before the date of the meeting.

Section 1.5 <u>Notice of Stockholder Business.</u>

(a) <u>Annual Meetings of Stockholders</u>.

(1) At an annual meeting of the stockholders, only such business shall be conducted as shall have been properly brought before the meeting. To be properly brought before an annual meeting, nominations for persons for election to the Board of Directors and the proposal of business to be considered by the stockholders must be (i) specified in the notice of meeting (or any supplement thereto) given by or at the direction of the Board of Directors (or any duly authorized committee thereof), (ii) otherwise properly brought before the meeting by or at the direction of the Board of Directors (or any duly authorized committee thereof), or (iii) otherwise properly be requested to be brought before the meeting by a stockholder (x) who complies with the procedures set forth in this Section 1.5 and (y) who was a stockholder of record of the Corporation (and, with respect to any beneficial owner, if different, on whose behalf such business is proposed or such nomination or nominations made, only if such beneficial owner was the beneficial owner of shares of the Corporation) both at the time the notice provided for in Section 1.5(a)(2) is delivered to the Secretary and on the record date for the determination of stockholders entitled to vote at the meeting, and (z) who is entitled to vote at the meeting upon such election of directors or upon such business, as the case may be.

In addition to any other requirements under applicable law and the Corporation's **(2)** Certificate of Incorporation, for a nomination for election to the Board of Directors or the proposal of business to be properly requested to be brought before an annual meeting by a stockholder, the stockholder must have given timely notice thereof in proper written form to the Secretary and any such proposed business, other than the nominations of persons for election to the Board of Directors, must constitute a proper matter for stockholder action pursuant to the Certificate of Incorporation, these Bylaws, and applicable law. To be timely, a stockholder's notice must be received at the principal executive offices of the Corporation not less than ninety (90) calendar days nor more than one hundred twenty (120) calendar days prior to the first anniversary of the preceding year's annual meeting; provided, that, in the event that the date of the annual meeting is advanced by more than twenty (20) calendar days, or delayed by more than seventy (70) calendar days, from such anniversary date, notice by the stockholder to be timely must be so received not earlier than the one hundred twentieth (120th) day prior to such annual meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such annual meeting or the tenth (10th) day following the day on which notice of the date of the meeting was communicated to stockholders or public announcement (as defined below) of the date of the meeting was made, whichever occurs first, and provided further, that for purposes of the application of Rule 14a-4(c) under the Securities Exchange Act of 1934, as amended (the "Exchange Act") (or any successor provision), the date for notice specified in this paragraph (a)(2) shall be the earlier of the date calculated as hereinbefore provided or the date specified in paragraph (c)(1) of Rule 14a-4. In no event shall the public announcement of an adjournment or postponement of a meeting of stockholders commence a new time period (or extend any time period) for the giving of a stockholder notice as described herein.

To be in proper written form, such stockholder's notice to the Secretary must be submitted by a holder of record of stock entitled to vote on the nomination of directors of the Corporation and shall set forth in writing and describe in fair, accurate, and material detail (A) as to each person whom the stockholder proposes to nominate for election as a director (a "nominee") (i) the name, age, business and residence address, and principal occupation or employment of the nominee, (ii) all information relating to such nominee that is required to be disclosed in solicitations of proxies for election of directors in an election contest, or is otherwise required, in each case pursuant to and in accordance with Regulation 14A under the Exchange Act of 1934, and (iii) such nominee's written consent to being named in the proxy statement as a nominee, the accompanying proxy card and to serving as a director if elected; (B) as to any other business that the stockholder proposes to bring before the annual meeting, (i) a brief description of the business desired to be brought before the annual meeting and the reasons for conducting such business at the annual meeting, (ii) the text of the proposal or business (including the text of any resolutions proposed for consideration and, in the event that such business includes a proposal to amend the Bylaws of the Corporation, the language of the proposed amendment), and (iii) any material interest of the stockholder and beneficial owner, if any, on whose behalf the proposal is made, in such business; and (C) as to such stockholder giving notice and the beneficial owner or owners, if different, on whose behalf the nomination or proposal is made, and any affiliates or associates (each within the meaning of Rule 12b-2 under the Exchange Act) of such stockholder or beneficial owner (each a "Proposing Person") (i) the name and address, as they appear on the Corporation's books, of such Proposing Person, (ii) the class or series and number of shares of the capital stock of the Corporation that are, directly or indirectly, owned beneficially and of record (within the meaning of Rule 13d-3 under the Exchange Act) by such Proposing Person (provided that for purposes of this Section 1.5, such Proposing Person shall in all events be deemed to beneficially

own any shares of any class or series and number of shares of capital stock of the Corporation as to which such Proposing Person has a right to acquire beneficial ownership at any time in the future), (iii) a description of all agreements, arrangements or understandings between (or on behalf of) such Proposing Person and any other person or persons (including their names) pursuant to which the proposals or nominations are to be made by such stockholder, (iv) a representation by each Proposing Person who is a holder of record of stock of the Corporation (A) that the notice the Proposing Person is giving to the Secretary is being given on behalf of (x) such holder of record and/or (y) if different than such holder of record, one or more beneficial owners of stock of the Corporation held of record by such holder of record, (B) as to each such beneficial owner, the number of shares held of record by such holder of record that are beneficially owned by such beneficial owner, with documentary evidence of such beneficial ownership, and (C) that such holder of record is entitled to vote at such meeting and intends to appear in person or by proxy at the meeting to propose such business or nomination set forth in its notice, (v) a representation (I) whether any such Proposing Person or nominee has received any financial assistance, funding or other consideration from any other person in respect of the nomination (and the details thereof) (a "Stockholder Associated Person") and (II) whether and the extent to which any hedging, derivative or other transaction has been entered into with respect to the Corporation within the past twelve (12) months by, or is in effect with respect to, such Proposing Person, any person to be nominated by such Proposing Person or any Stockholder Associated Person, the effect or intent of which transaction is to mitigate loss to or manage risk or benefit of share price changes for, or to increase or decrease the voting power of, such stockholder, nominee or any such Stockholder Associated Person, (vi) a representation whether any Proposing Person intends or is part of a group that intends to (I) deliver a proxy statement and/or form of proxy to holders of at least the percentage of the Corporation's outstanding voting power required to approve or adopt the proposal or elect the nominee and/or (II) otherwise solicit proxies from stockholders in support of such proposal, (vii) a representation that no Proposing Person or nominee is subject to, nor will enter into, any voting or other agreement that has not been disclosed to the Corporation and that could limit or interfere with such nominee's ability to comply, if elected, with their fiduciary duties under applicable law, and (viii) any other information relating to such Proposing Person that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies in support of such proposal pursuant to Section 14 of the Exchange Act, and any rules and regulations promulgated thereunder. The foregoing notice requirements of this Section 1.5 shall not apply to any proposal made pursuant to Rule 14a-8 (or any successor thereof) promulgated under the Exchange Act. A proposal to be made pursuant to Rule 14a-8 (or any successor thereof) promulgated under the Exchange Act shall be deemed satisfied if the stockholder making such proposal complies with the provisions of Rule 14a-8 and has notified the Corporation of his or her intention to present a proposal at an annual meeting in compliance with Rule 14a-8 and such stockholder's proposal has been included in a proxy statement that has been prepared by the Corporation to solicit proxies for such annual meeting. The Corporation may require any proposed nominee to furnish such other information as it may reasonably require to determine (x) the eligibility of such proposed nominee to serve as a director of the Corporation and (y) whether the nominee would qualify as an "independent director" or "audit committee financial expert" under applicable law, securities exchange rule or regulation, or any publicly disclosed corporate governance guideline or committee charter of the Corporation.

(3) In addition to the other requirements of this Section 1.5, each nominee who a Proposing Person proposes to nominate for election or re-election as a director must deliver in writing (in accordance with the time periods prescribed for delivery of notice under this Section 1.5) to the Secretary at the principal executive offices of the Corporation a written

questionnaire completed and signed by such nominee (in the form provided by the Secretary upon written request of any stockholder of record within ten (10) days of such request) with respect to the background, qualifications, and independence of such nominee and the background of any other person or entity on whose behalf the nomination is being made.

- (4) Notwithstanding anything in paragraph (a)(2) of this Section 1.5 to the contrary, in the event that the number of directors to be elected to the Board of Directors of the Corporation at an annual meeting is increased and there is no public announcement by the Corporation naming all of the nominees for director or specifying the size of the increased Board of Directors at least one hundred (100) calendar days prior to the first anniversary date of the immediately preceding annual meeting, a stockholder's notice required by this Section 1.5 shall also be considered timely, but only with respect to nominees for any new positions created by such increase, if it shall be received by the Secretary at the principal executive offices of the Corporation not later than the close of business on the tenth (10) day following the day on which such public announcement is first made by the Corporation.
- law, if any stockholder or Proposing Person (i) provides notice pursuant to Rule 14a-19(b) under the Exchange Act with respect to any proposed nominee and (ii) subsequently fails to comply with the requirements of Rule 14a-19 under the Exchange Act (or fails to timely provide reasonable evidence sufficient to satisfy the Corporation that such stockholder has met the requirements of Rule 14a-19(a)(3) under the Exchange Act in accordance with the following sentence), then the nomination of each such proposed nominee shall be disregarded, notwithstanding that the nominee is included as a nominee in the Corporation's proxy statement, notice of meeting or other proxy materials for any annual meeting (or any supplement thereto) and notwithstanding that proxies or votes in respect of the election of such proposed nominees may have been received by the Corporation (which proxies and votes shall be disregarded). If any stockholder or Proposing Person provides notice pursuant to Rule 14a-19(b) promulgated under the Exchange Act, such stockholder shall deliver to the Corporation, no later than five business days prior to the date of the meeting and any adjournment or postponement thereof, reasonable evidence that it has met the requirements of Rule 14a-19(a) (3) promulgated under the Exchange Act.
- Special Meetings of Stockholders. Only such business shall be conducted at a special meeting of stockholders as shall have been brought before the meeting pursuant to the Corporation's notice of meeting. In the event the Corporation calls a special meeting of stockholders for the purpose of electing one or more directors to the Board of Directors, any such stockholder entitled to vote at such meeting who was a stockholder of record of the Corporation (and, with respect to any beneficial owner, if different, on whose behalf such nomination or nominations are made, only if such beneficial owner was the beneficial owner of shares of the Corporation) both at the time the notice provided for in paragraph (a)(2) of this Section 1.5 is delivered to the Secretary and on the record date for the determination of stockholders entitled to vote at the special meeting may nominate a person or persons (as the case may be) for election to such position(s) as specified in the Corporation's notice of meeting, if the stockholder's notice meeting the requirements of paragraph (a)(2) of this Section 1.5 (substituting special meeting for annual meeting as applicable) shall be received by the Secretary at the principal executive offices of the Corporation not earlier than the close of business on the one hundred twentieth (120th) day prior to such special meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such special meeting or the tenth (10th) day following the day on which public announcement is first made of the date of the special meeting and of the nominees proposed by the Board of Directors to

be elected at such meeting; provided, however, that a stockholder may nominate persons for election at a special meeting only to such directorship(s) as specified in the Corporation's notice of the meeting. In no event shall the public announcement of an adjournment or postponement of a special meeting commence a new time period (or extend any time period) for the giving of a stockholder's notice as described above.

<u>Updating and Supplementing of Stockholder Information</u>. A stockholder providing notice of nominations of persons for election to the Board of Directors at an annual or special meeting of stockholders or notice of business proposed to be brought before an annual meeting of stockholders shall further update and supplement such notice so that the information provided or required to be provided in such notice pursuant to paragraph (a)(2) of this Section 1.5 shall be true and correct both as of the record date for the determination of stockholders entitled to notice of the meeting and as of the date that is ten (10) business days before the meeting or any adjournment or postponement thereof, and such updated and supplemental information shall be delivered to, or mailed and received by, the Secretary at the principal executive offices of the Corporation (a) in the case of information that is required to be updated and supplemented to be true and correct as of the record date for the determination of stockholders entitled to notice of the meeting, not later than the later of five (5) business days after such record date or five (5) business days after the public announcement of such record date, and (b) in the case of information that is required to be updated and supplemented to be true and correct as of ten (10) business days before the meeting or any adjournment or postponement thereof, not later than eight (8) business days before the meeting or any adjournment or postponement thereof (or if not practicable to provide such updated and supplemental information not later than eight (8) business days before any adjournment or postponement, on the first practicable date before any such adjournment or postponement).

(b) General.

Only such persons who are nominated in accordance with the procedures set **(1)** forth in this Section 1.5 shall be eligible to be elected at an annual or special meeting of stockholders of the Corporation to serve as directors and only such business shall be conducted at a meeting of stockholders as shall have been brought before the meeting in accordance with the procedures set forth in this Section 1.5. Except as otherwise provided by law, the chairman of the meeting shall have the power and duty (i) to determine whether a nomination or any business proposed to be brought before the meeting was made or proposed, as the case may be, in accordance with the procedures set forth in this Section 1.5 (including whether the stockholder or beneficial owner, if any, on whose behalf the nomination or proposal is made solicited (or is part of a group which solicited) or did not so solicit, as the case may be, proxies in support of such stockholder's nominee or proposal in compliance with such stockholder's representation as required by clause (a)(2)(C)(vi) of this Section 1.5) and (ii) if any proposed nomination or proposed business was not made or proposed in compliance with this Section 1.5, to declare that such nomination shall be disregarded or that such proposed business shall not be transacted. Notwithstanding the foregoing provisions of this Section 1.5, if the stockholder (or a qualified representative of the stockholder) does not appear at the annual or special meeting of stockholders of the Corporation to present the nomination to the Board of Directors or to present the proposed business, such nomination shall be disregarded and such proposed business shall not be transacted, notwithstanding that proxies in respect of such vote may have been received by the Corporation. For purposes of this Section 1.5, to be considered a qualified representative of the stockholder, a person must be authorized by a writing executed by such stockholder or an electronic transmission delivered by such stockholder to act for such

stockholder as proxy at the meeting of stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, at the meeting of stockholders.

- (2) For purposes of this Section 1.5, (i) "public announcement" shall mean disclosure in a press release reported by a national news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to the Exchange Act, and (ii) "business day" shall mean any day, other than Saturday, Sunday and any day on which banks located in the State of New York are authorized or obligated by applicable law to close.
- (3) Notwithstanding the foregoing provisions of this Section 1.5, a stockholder shall also comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in this Section 1.5. Nothing in this Section 1.5 shall be deemed to affect any rights (i) of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act or (ii) of the holders of any series of preferred stock to elect directors pursuant to any applicable provisions of the Corporation's Certificate of Incorporation.

Section 1.6 Quorum.

Subject to the rights of the holders of any series of preferred stock and except as otherwise provided by law or in the Certificate of Incorporation or these Bylaws, at any meeting of stockholders, the holders of a majority in total voting power of the outstanding shares of stock entitled to vote at the meeting shall be present or represented by proxy in order to constitute a quorum for the transaction of any business. The chairman of the meeting shall have the power and duty to determine whether a quorum is present at any meeting of the stockholders. Shares of its own stock belonging to the Corporation or to another corporation, if a majority of the shares entitled to vote in the election of directors of such other corporation is held, directly or indirectly, by the Corporation, shall neither be entitled to vote nor be counted for quorum purposes; provided, however, that the foregoing shall not limit the right of the Corporation or any subsidiary of the Corporation to vote stock, including, but not limited to, its own stock, held by it in a fiduciary capacity. In the absence of a quorum, the chairman of the meeting may adjourn the meeting from time to time in the manner provided in Section 1.7 hereof until a quorum shall be present.

Section 1.7 Adjournment.

Any meeting of stockholders, annual or special, may be adjourned from time to time solely by the chairman of the meeting because of the absence of a quorum or for any other reason (including to address technical failures to convene or continue a meeting using remote communication) and to reconvene at the same or some other time, date and place, if any, or by means of remote communication. Notice need not be given of any such adjourned meeting if the time, date and place, if any, and the means of remote communications, if any, thereof are (i) announced at the meeting at which the adjournment is taken, (ii) displayed, during the time scheduled for the meeting, on the same electronic network used to enable stockholders and proxyholders to participate in the meeting by means of remote communication, or (iii) set forth in the notice of the meeting. The chairman of the meeting shall have full power and authority to adjourn a stockholder meeting in his sole discretion even over stockholder opposition to such adjournment. The stockholders present at a meeting shall not have the authority to adjourn the meeting. If the time, date and place, if any, thereof, and the means of remote

communication, if any, by which the stockholders and the proxy holders may be deemed to be present in person and vote at such adjourned meeting are announced at the meeting at which the adjournment is taken, displayed, during the time scheduled for the meeting, on the same electronic network used to enable stockholders and proxyholders to participate in the meeting by means of remote communication, or set forth in the notice of the meeting, and the adjournment is for less than thirty (30) calendar days, no notice need be given of any such adjourned meeting. If the adjournment is for more than thirty (30) calendar days or if after the adjournment a new record date for determining stockholders entitled to vote at the adjourned meeting is fixed for the adjourned meeting, then notice shall be given to each stockholder entitled to vote at the meeting. At the adjourned meeting, the stockholders may transact any business that might have been transacted at the original meeting.

Section 1.8 Organization.

The Chairman of the Board, or in the Chairman of the Board's absence or at the Chairman of the Board's direction, the President, or in the President's absence or at the President's direction, any officer of the Corporation, shall call to order meetings of stockholders and preside over and act as chairman of such meetings. The Board of Directors or, if the Board fails to act, the stockholders, may appoint any stockholder, director or officer of the Corporation to act as chairman of any meeting in the absence of the Chairman of the Board, the President and other officers. The date and time of the opening and closing of the polls for each matter upon which the stockholders will vote at a meeting shall be determined by the chairman of the meeting and announced at the meeting. The Board of Directors may adopt by resolution such rules and regulations for the conduct of the meeting of stockholders as it shall deem appropriate. Unless otherwise determined by the Board of Directors, the chairman of the meeting shall have the exclusive right to determine the order of business and to prescribe other such rules, regulations and procedures and shall have the authority in his discretion to regulate the conduct of any such meeting. Such rules, regulations or procedures, whether adopted by the Board of Directors or prescribed by the chairman of the meeting, may include, without limitation, the following: (i) rules and procedures for maintaining order at the meeting and the safety of those present; (ii) limitations on attendance at or participation in the meeting to stockholders of record of the Corporation, their duly authorized and constituted proxies or such other persons as the chairman of the meeting shall determine; (iii) restrictions on entry to the meeting after the time fixed for the commencement thereof; and (iv) limitations on the time allotted to questions or comments by participants. Unless and to the extent determined by the Board of Directors or the chairman of the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure.

The Secretary, or in the Secretary's absence, an Assistant Secretary, shall act as secretary of all meetings of stockholders, but, in the absence of the Secretary or the Assistant Secretary, the chairman of the meeting may appoint any other person to act as secretary of the meeting.

Section 1.9 <u>Postponement or Cancellation of Meeting.</u>

Any previously scheduled annual or special meeting of the stockholders may be postponed or canceled by resolution of the Board of Directors upon public notice given prior to the time previously scheduled for such meeting of stockholders.

Section 1.10 Voting.

Subject to the rights of the holders of any series of preferred stock and except as otherwise provided by law, the Certificate of Incorporation or these Bylaws and except for the election of directors, at any meeting duly called and held at which a quorum is present, the affirmative vote of a majority of the combined voting power of the outstanding shares present in person or represented by proxy at the meeting and entitled to vote on the subject matter shall be the act of the stockholders. Subject to the rights of the holders of any series of preferred stock, at any meeting duly called and held for the election of directors at which a quorum is present, directors shall be elected by a plurality of the combined voting power of the outstanding shares present in person or represented by proxy at the meeting and entitled to vote on the election of directors.

Section 1.11 List of Stockholders.

The Corporation shall prepare, no later than the tenth (10th) calendar day before each meeting of the stockholders, a complete list of the stockholders entitled to vote thereat, arranged in alphabetical order, and showing the address of each stockholder and the number of shares registered in the stockholder's name; provided, however, if the record date for determining the stockholders entitled to vote at the meeting is fewer than ten (10) calendar days before the meeting date, the list shall reflect the stockholders entitled to vote as of the tenth (10) calendar day before the meeting date. Nothing contained in this Section 1.11 shall require the Corporation to include electronic mail addresses or other electronic contact information on such list. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting for a period of ten (10) calendar days ending on the day before the meeting date: (i) on a reasonably accessible electronic network, provided that the information required to gain access to such list is provided with the notice of the meeting, or (ii) during ordinary business hours, at the principal place of business of the Corporation. If the Corporation determines to make the list available on an electronic network, the Corporation may take reasonable steps to ensure that such information is available only to stockholders of the Corporation. The stock ledger shall be the only evidence of the identity of the stockholders entitled to examine such list.

Section 1.12 Remote Communications.

For purposes of these Bylaws, if authorized by the Board of Directors in its sole discretion, and subject to such guidelines and procedures as the Board of Directors may adopt, stockholders and proxyholders may, by means of remote communication:

- (a) participate in a meeting of stockholders; and
- (b) be deemed present in person and vote at a meeting of stockholders whether such meeting is to be held at a designated place or solely by means of remote communication, provided that (i) the Corporation shall implement reasonable measures to verify that each person deemed present and permitted to vote at the meeting by means of remote communication is a stockholder or proxyholder, (ii) the Corporation shall implement reasonable measures to provide such stockholders and proxyholders a reasonable opportunity to participate in the meeting and to vote on matters submitted to the stockholders, including an opportunity to read or hear the proceedings of the meeting substantially concurrent with such proceedings, and (iii) if any stockholder or proxyholder votes or takes other action at the meeting by means of remote communication, a record of such vote or other action shall be maintained by the Corporation.

ARTICLE II

BOARD OF DIRECTORS

Section 2.1 Number and Term of Office.

- (a) Subject to any limitations set forth in the Certificate of Incorporation and to any provision of the Delaware General Corporation Law relating to the powers or rights conferred upon or reserved to the stockholders or the holders of any class or series of the issued and outstanding stock of the Corporation, the business and affairs of the Corporation shall be managed, and all corporate powers shall be exercised, by or under the direction of the Board of Directors. Subject to any rights of the holders of any series of preferred stock to elect additional directors, the Board of Directors shall be comprised of not less than three (3) members and the exact number will be fixed from time to time by the Board of Directors by resolution adopted by the affirmative vote of not less than 75% of the members of the Board of Directors then in office. Directors need not be stockholders of the Corporation. The Corporation shall nominate the persons holding the offices of Chairman of the Board and President for election as directors at any meeting at which such persons are subject to election as directors.
- (b) Except as otherwise fixed by the Certificate of Incorporation relating to the rights of the holders of any series of preferred stock to separately elect additional directors, which additional directors are not required to be classified pursuant to the terms of such series of preferred stock, the Board of Directors shall be divided into three (3) classes: Class I, Class II and Class III. Each class shall consist, as nearly as possible, of a number of directors equal to one-third (33 1/3%) of the then authorized number of members of the Board of Directors. The term of office of the initial Class I directors shall expire at the annual meeting of stockholders in 2014; the term of office of the initial Class III directors shall expire at the annual meeting of stockholders in 2015; and the term of office of the initial Class III directors shall expire at the annual meeting of stockholders in 2013. At each annual meeting of stockholders of the Corporation the successors of that class of directors whose term expires at that meeting shall be elected to hold office for a term expiring at the annual meeting of stockholders held in the third year following the year of their election. The directors of each class will serve until the earliest to occur of their death, resignation, removal or disqualification or the election and qualification of their respective successors.

Section 2.2 <u>Resignations</u>.

Any director of the Corporation, or any member of any committee, may resign at any time by giving notice in writing or by electronic transmission to the Board of Directors, the Chairman of the Board or the President or Secretary. Any such resignation shall take effect at the time specified therein or, if the time be not specified therein, then upon receipt thereof. The acceptance of such resignation shall not be necessary to make it effective unless otherwise stated therein.

Section 2.3 Removal of Directors.

Subject to the rights of the holders of any series of preferred stock, directors may be removed from office only for cause upon the affirmative vote of the holders of not less than a

majority of the total voting power of the then outstanding shares entitled to vote at an election of directors voting together as a single class.

Section 2.4 Newly Created Directorships and Vacancies.

Subject to the rights of the holders of any series of preferred stock, vacancies on the Board of Directors resulting from death, resignation, removal, disqualification or other cause, and newly created directorships resulting from any increase in the number of directors on the Board of Directors, shall be filled only by the affirmative vote of a majority of the remaining directors then in office (even though less than a quorum) or by the sole remaining director. Any director elected in accordance with the preceding sentence shall hold office for the remainder of the full term of the class of directors in which the vacancy occurred or to which the new directorship is apportioned, and until such director's successor shall have been elected and qualified. No decrease in the number of directors constituting the Board of Directors shall shorten the term of any incumbent director, except as may be provided in the terms of any series of preferred stock with respect to any additional director elected by the holders of such series of preferred stock. If at any time, by reason of death or resignation or other cause, the Corporation should have no directors in office, then any officer or any stockholder may call a special meeting of stockholders in the same manner that the Board of Directors may call such a meeting, and directors for the unexpired terms may be elected at such special meeting.

Section 2.5 <u>Meetings</u>.

Regular meetings of the Board of Directors shall be held on such dates and at such times and places, within or without the State of Delaware, as shall from time to time be determined by the Board of Directors, such determination to constitute the only notice of such regular meetings to which any director shall be entitled. In the absence of any such determination, such meeting shall be held, upon notice to each director in accordance with Section 2.6 of this Article II, at such times and places, within or without the State of Delaware, as shall be designated in the notice of meeting.

Special meetings of the Board of Directors shall be held at such times and places, if any, within or without the State of Delaware, as shall be designated in the notice of the meeting in accordance with Section 2.6 hereof. Special meetings of the Board of Directors may be called by the Chairman of the Board, and shall be called by the President or Secretary upon the written request of not less than 75% of the members of the Board of Directors then in office.

Section 2.6 <u>Notice of Meetings</u>.

The Secretary, or in his absence any other officer of the Corporation, shall give each director notice of the time and place of holding of any regular meetings (if required) or special meetings of the Board of Directors, in accordance with Section 5.4 of these Bylaws, by mail at least ten (10) calendar days before the meeting, or by courier service at least three (3) calendar days before the meeting, or by facsimile transmission, electronic mail or other electronic transmission, or personal service, in each case, at least twenty-four (24) hours before the meeting, unless notice is waived in accordance with Section 5.4 of these Bylaws. Unless otherwise stated in the notice thereof, any and all business may be transacted at any meeting without specification of such business in the notice.

Section 2.7 <u>Meetings by Conference Telephone or Other Communications.</u>

Members of the Board of Directors, or any committee thereof, may participate in a meeting of the Board of Directors or such committee by means of telephone conference or other communications equipment by means of which all persons participating in the meeting can hear each other and communicate with each other, and such participation in a meeting by such means shall constitute presence in person at such meeting.

Section 2.8 Quorum and Organization of Meetings.

A majority of the total number of members of the Board of Directors as constituted from time to time shall constitute a quorum for the transaction of business, but, if at any meeting of the Board of Directors (whether or not adjourned from a previous meeting) there shall be less than a quorum present, a majority of those present may adjourn the meeting to another time, date and place, and the meeting may be held as adjourned without further notice or waiver. Except as otherwise provided by law, the Certificate of Incorporation or these Bylaws, a majority of the directors present at any meeting at which a quorum is present may decide any question brought before such meeting. Meetings shall be presided over by the Chairman of the Board or in his absence by such other person as the directors may select. The Board of Directors shall keep written minutes of its meetings. The Secretary shall act as secretary of the meeting, but in his absence the chairman of the meeting may appoint any person to act as secretary of the meeting.

The Board may designate one or more committees, each committee to consist of one or more of the directors of the Corporation. The Board may designate one or more Directors as alternate members of any committee to replace absent or disqualified members at any meeting of such committee. If a member of a committee shall be absent from any meeting, or disqualified from voting thereat, the remaining member or members present and not disqualified from voting, whether or not such member or members constitute a quorum, may, by a unanimous vote, appoint another member of the Board of Directors to act at the meeting in place of any such absent or disqualified member. Any such committee, to the extent provided in a resolution of the Board of Directors passed as aforesaid, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be impressed on all papers that may require it, but no such committee shall have the power or authority of the Board of Directors in reference to (i) approving or adopting, or recommending to the stockholders, any action or matter expressly required by the laws of the State of Delaware to be submitted to the stockholders for approval or (ii) adopting, amending or repealing any Bylaw of the Corporation. Such committee or committees shall have such name or names as may be determined from time to time by resolution adopted by the Board of Directors. Unless otherwise specified in the resolution of the Board of Directors designating a committee, at all meetings of such committee a majority of the total number of members of the committee shall constitute a quorum for the transaction of business, and the vote of a majority of the members of the committee present at any meeting at which there is a quorum shall be the act of the committee. Each committee shall keep regular minutes of its meetings. Unless the Board of Directors otherwise provides, each committee designated by the Board of Directors may make, alter and repeal rules for the conduct of its business. In the absence of such rules each committee shall conduct its business in the same manner as the Board of Directors conducts its business pursuant to Article II of these Bylaws.

Section 2.9 Indemnification.

The Corporation shall indemnify members of the Board of Directors and officers of the Corporation and their respective heirs, personal representatives and successors in interest for or on account of any action performed on behalf of the Corporation, to the fullest extent permitted by the laws of the State of Delaware and the Corporation's Certificate of Incorporation, as now or hereafter in effect.

Section 2.10 <u>Indemnity Undertaking</u>.

To the extent not prohibited by law, the Corporation shall indemnify any person who is or was made, or threatened to be made, a party to any threatened, pending or completed action, suit or proceeding (a "**Proceeding**"), whether civil, criminal, administrative or investigative, including, without limitation, an action by or in the right of the Corporation to procure a judgment in its favor, by reason of the fact that such person, or a person of whom such person is the legal representative, is or was a director or officer of the Corporation, or is or was serving in any capacity at the request of the Corporation for any other corporation, partnership, limited liability company, joint venture, trust, employee benefit plan or other enterprises (an "**Other Entity**"), against judgments, fines, penalties, excise taxes, amounts paid in settlement and costs, charges and expenses (including attorneys' fees). Persons who are not directors or officers of the Corporation may be similarly indemnified in respect of service to the Corporation or to an Other Entity at the request of the Corporation to the extent the Board of Directors at any time specifies that such persons are entitled to the benefits of this Section 2.10. Except as otherwise provided in Section 2.12 hereof, the Corporation shall be required to indemnify a person in connection with a proceeding (or part thereof) commenced by such person only if the commencement of such proceeding (or part thereof) by the person was authorized by the Board of Directors.

Section 2.11 <u>Advancement of Expenses</u>.

The Corporation shall, from time to time, reimburse or advance to any director or officer or other person entitled to indemnification hereunder the funds necessary for payment of expenses, including attorneys' fees, incurred in connection with any Proceeding in advance of the final disposition of such Proceeding; provided, however, that, such expenses incurred by or on behalf of any director or officer or other person may be paid in advance of the final disposition of a Proceeding only upon receipt by the Corporation of an undertaking, by or on behalf of such director or officer or such person, to repay all amounts advanced if it shall ultimately be determined by final judicial decision from which there is no further right of appeal that such director, officer or other person is not entitled to be indemnified for such expenses. Except as otherwise provided in Section 2.12 hereof, the Corporation shall be required to reimburse or advance expenses incurred by a person in connection with a proceeding (or part thereof) commenced by such person only if the commencement of such proceeding (or part thereof) by the person was authorized by the Board of Directors.

Section 2.12 Claims.

If a claim for indemnification or advancement of expenses under this Article II is not paid in full within sixty (60) calendar days after a written claim therefor by the person seeking indemnification or reimbursement or advancement of expenses has been received by the Corporation, the person may file suit to recover the unpaid amount of such claim and, if successful, in whole or in part, shall be entitled to be paid the expense (including attorneys' fees) of prosecuting such claim to the fullest extent permitted by Delaware law. In any such action the Corporation shall have the burden of proving that the person seeking indemnification

or reimbursement or advancement of expenses is not entitled to the requested indemnification, reimbursement or advancement of expenses under applicable law.

Section 2.13 <u>Amendment, Modification or Repeal</u>.

Any amendment, modification or repeal of the foregoing provisions of this Article II shall not adversely affect any right or protection hereunder of any person entitled to indemnification under Section 2.9 hereof in respect of any act or omission occurring prior to the time of such repeal or modification.

Section 2.14 Executive Committee of the Board of Directors.

The Board of Directors, by the affirmative vote of not less than 75% of the members of the Board of Directors then in office, may designate an executive committee, all of whose members shall be directors, to manage and operate the affairs of the Corporation or particular properties or enterprises of the Corporation. Subject to the limitations of the law of the State of Delaware and the Certificate of Incorporation, such executive committee shall exercise all powers and authority of the Board of Directors in the management of the business and affairs of the Corporation including, but not limited to, the power and authority to authorize the issuance of shares of common or preferred stock. The executive committee shall keep minutes of its meetings and report to the Board of Directors not less often than quarterly on its activities and shall be responsible to the Board of Directors for the conduct of the enterprises and affairs entrusted to it. Regular meetings of the executive committee, of which no notice shall be necessary, shall be held at such time, dates and places, if any, as shall be fixed by resolution adopted by the executive committee. Special meetings of the executive committee shall be called at the request of the President or of any member of the executive committee, and shall be held upon such notice as is required by these Bylaws for special meetings of the Board of Directors, provided that oral notice by telephone or otherwise, or notice by electronic transmission shall be sufficient if received not later than the day immediately preceding the day of the meeting.

Section 2.15 Other Committees of the Board of Directors.

The Board of Directors may by resolution establish committees other than an executive committee and shall specify with particularity the powers and duties of any such committee. Subject to the limitations of the laws of the State of Delaware and the Certificate of Incorporation, any such committee shall exercise all powers and authority specifically granted to it by the Board of Directors, which powers may include the authority to authorize the issuance of shares of common or preferred stock. Such committees shall serve at the pleasure of the Board of Directors, keep minutes of their meetings and have such names as the Board of Directors by resolution may determine and shall be responsible to the Board of Directors for the conduct of the enterprises and affairs entrusted to them.

Section 2.16 <u>Directors' Compensation</u>.

Directors shall receive such compensation for attendance at any meetings of the Board and any expenses incidental to the performance of their duties as the Board of Directors shall determine by resolution. Such compensation may be in addition to any compensation received by the members of the Board of Directors in any other capacity.

Section 2.17 <u>Action Without Meeting</u>.

Nothing contained in these Bylaws shall be deemed to restrict the power of members of the Board of Directors or any committee designated by the Board of Directors to take any action required or permitted to be taken by them without a meeting; provided, however, that if such action is taken without a meeting by consent by electronic transmission or transmissions, such electronic transmissions must either set forth or be submitted with information from which it can be determined that the electronic transmission or transmissions were authorized by the director.

ARTICLE III

OFFICERS

Section 3.1 <u>Executive Officers</u>.

The Board of Directors shall elect from its own number, a Chairman of the Board and a President. The Board of Directors may also elect such Vice Presidents as in the opinion of the Board of Directors the business of the Corporation requires, a Treasurer and a Secretary, any of whom may or may not be directors. The Board of Directors may also elect, from time to time, such other or additional officers as in its opinion are desirable for the conduct of business of the Corporation and such officers shall hold office at the pleasure of the Board of Directors; provided, however, that the President shall not hold any other office except that of Chairman of the Board.

Section 3.2 <u>Powers and Duties of Officers.</u>

The Chairman of the Board shall have overall responsibility for the management and direction of the business and affairs of the Corporation and shall exercise such duties as customarily pertain to the office of Chairman of the Board and such other duties as may be prescribed from time to time by the Board of Directors. He shall be the senior officer of the Corporation and in case of the inability or failure of the President to perform his duties, he shall perform the duties of the President. He may appoint and terminate the appointment or election of officers, agents or employees other than those appointed or elected by the Board of Directors. He may sign, execute and deliver, in the name of the Corporation, powers of attorney, contracts, bonds and other obligations. The Chairman of the Board shall preside at all meetings of stockholders and of the Board of Directors at which he is present, and shall perform such other duties as may be prescribed from time to time by the Board of Directors or these Bylaws.

The President of the Corporation shall have such powers and perform such duties as customarily pertain to a chief executive officer and the office of a president, including, without limitation, being responsible for the active direction of the daily business of the Corporation, and shall exercise such other duties as may be prescribed from time to time by the Board of Directors. The President may sign, execute and deliver, in the name of the Corporation, powers of attorney, contracts, bonds and other obligations. In the absence or disability of the Chairman of the Board, the President shall perform the duties and exercise the powers of the Chairman of the Board.

Vice Presidents shall have such powers and perform such duties as may be assigned to them by the Chairman of the Board, the President, the executive committee, if any, or the

Board of Directors. A Vice President may sign and execute contracts and other obligations pertaining to the regular course of his duties which implement policies established by the Board of Directors.

Unless the Board of Directors otherwise declares by resolution, the Treasurer shall have general custody of all the funds and securities of the Corporation and general supervision of the collection and disbursement of funds of the Corporation. He shall endorse for collection on behalf of the Corporation checks, notes and other obligations, and shall deposit the same to the credit of the Corporation in such bank or banks or depository as the Board of Directors may designate. He may sign, with the Chairman of the Board, President or such other person or persons as may be designated for the purpose by the Board of Directors, all bills of exchange or promissory notes of the Corporation. He shall enter or cause to be entered regularly in the books of the Corporation a full and accurate account of all moneys received and paid by him on account of the Corporation, shall at all reasonable times exhibit his books and accounts to any director of the Corporation upon application at the office of the Corporation during business hours and, whenever required by the Board of Directors or the President, shall render a statement of his accounts. He shall perform such other duties as may be prescribed from time to time by the Board of Directors or by these Bylaws. He may be required to give bond for the faithful performance of his duties in such sum and with such surety as shall be approved by the Board of Directors. Any Assistant Treasurer shall, in the absence or disability of the Treasurer, perform the duties and exercise the powers of the Treasurer and shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.

The Secretary shall keep the minutes of all meetings of the stockholders and of the Board of Directors. The Secretary shall cause notice to be given of meetings of stockholders, of the Board of Directors, and of any committee appointed by the Board of Directors. He shall have custody of the corporate seal, minutes and records relating to the conduct and acts of the stockholders and Board of Directors, which shall, at all reasonable times, be open to the examination of any director. The Secretary or any Assistant Secretary may certify the record of proceedings of the meetings of the stockholders or of the Board of Directors or resolutions adopted at such meetings, may sign or attest certificates, statements or reports required to be filed with governmental bodies or officials, may sign acknowledgments of instruments, may give notices of meetings and shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.

Section 3.3 Bank Accounts.

In addition to such bank accounts as may be authorized in the usual manner by resolution of the Board of Directors, the Treasurer, with approval of the Chairman of the Board or the President, may authorize such bank accounts to be opened or maintained in the name and on behalf of the Corporation as he may deem necessary or appropriate, provided payments from such bank accounts are to be made upon and according to the check of the Corporation, which may be signed jointly or singularly by either the manual or facsimile signature or signatures of such officers or bonded employees of the Corporation as shall be specified in the written instructions of the Treasurer or Assistant Treasurer of the Corporation with the approval of the Chairman of the Board or the President of the Corporation.

Section 3.4 Proxies; Stock Transfers.

Unless otherwise provided in the Certificate of Incorporation or directed by the Board of Directors, the Chairman of the Board or the President or any Vice President or their designees shall have full power and authority on behalf of the Corporation to attend and to vote upon all matters and resolutions at any meeting of stockholders of any corporation in which this Corporation may hold stock, and may exercise on behalf of this Corporation any and all of the rights and powers incident to the ownership of such stock at any such meeting, whether regular or special, and at all adjournments thereof, and shall have power and authority to execute and deliver proxies and consents on behalf of this Corporation in connection with the exercise by this Corporation of the rights and powers incident to the ownership of such stock, with full power of substitution or revocation. Unless otherwise provided in the Certificate of Incorporation or directed by the Board of Directors, the Chairman of the Board or the President or any Vice President or their designees shall have full power and authority on behalf of the Corporation to transfer, sell or dispose of stock of any corporation in which this Corporation may hold stock.

ARTICLE IV

CAPITAL STOCK

Section 4.1 Shares.

The shares of the Corporation shall be represented by a certificate, provided that the Board of Directors may provide by resolution that some or all of any or all classes or series of the Corporation's stock shall be uncertificated. Certificates shall be signed by or in the name of the Corporation by any two authorized officers of the Corporation, and sealed with the seal of the Corporation. Such seal may be a facsimile, engraved or printed. Within a reasonable time after the issuance or transfer of uncertificated shares, the Corporation shall send to the registered owner thereof a notice, in writing or by electronic transmission, containing the information required to be set forth or stated on certificates pursuant to Sections 151, 156, 202(a) or 218(a) of the Delaware General Corporation Law or a statement that the Corporation will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative participating, optional or other special rights of each class of stock or series thereof and the qualification, limitations or restrictions of such preferences and/or rights.

Any of or all the signatures on a certificate may be facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such an officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if such officer, transfer agent or registrar had not ceased to hold such position at the time of its issuance.

Except as otherwise expressly provided by law, the rights and obligations of the holders of uncertificated shares and the rights and obligations of the holders of certificates representing stock of the same class and series shall be identical.

Section 4.2 Transfer of Shares.

(a) Upon surrender to the Corporation or the transfer agent of a certificate for shares duly endorsed or accompanied by proper evidence of succession, assignation or authority to transfer, it shall be the duty of the Corporation to issue a new certificate to the person entitled thereto, cancel the old certificate and record the transaction upon its books.

Upon receipt of proper transfer instructions from the registered owner of uncertificated shares such uncertificated shares shall be cancelled, and the issuance of new equivalent uncertificated shares or certificated shares shall be made to the person entitled thereto and the transaction shall be recorded upon the books of the Corporation.

(b) The person in whose name shares of stock stand on the books of the Corporation shall be deemed by the Corporation to be the owner thereof for all purposes, and the Corporation shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise provided by the laws of the State of Delaware.

Section 4.3 Lost Certificates.

The Board of Directors or any transfer agent of the Corporation may direct a new certificate or certificates or uncertificated shares representing stock of the Corporation to be issued in place of any certificate or certificates theretofore issued by the Corporation, alleged to have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming the certificate to be lost, stolen or destroyed. When authorizing such issue of a new certificate or certificates or uncertificated shares, the Board of Directors (or any transfer agent of the Corporation authorized to do so by a resolution of the Board of Directors) may, in its discretion and as a condition precedent to the issuance thereof, require the owner of such lost, stolen or destroyed certificate or certificates, or his legal representative, to give the Corporation a bond in such sum as the Board of Directors (or any transfer agent so authorized) shall direct to indemnify the Corporation and the transfer agent against any claim that may be made against the Corporation with respect to the certificate alleged to have been lost, stolen or destroyed or the issuance of such new certificates or uncertificated shares, and such requirement may be general or confined to specific instances.

Section 4.4 <u>Transfer Agent and Registrar.</u>

The Board of Directors may appoint one or more transfer agents and one or more registrars, and may require all certificates for shares to bear the manual or facsimile signature or signatures of any of them.

Section 4.5 Regulations.

The Board of Directors shall have power and authority to make all such rules and regulations as it may deem expedient concerning the issue, transfer, registration, cancellation and replacement of certificates representing stock of the Corporation or uncertificated shares, which rules and regulations shall comply in all respects with the rules and regulations of the transfer agent.

ARTICLE V

GENERAL PROVISIONS

Section 5.1 Offices.

The Corporation shall maintain a registered office in the State of Delaware as required by the laws of the State of Delaware. The Corporation may also have offices in such other places, either within or without the State of Delaware, as the Board of Directors may from time

to time designate or as the business of the Corporation may require.

Section 5.2 <u>Corporate Seal.</u>

The corporate seal shall have inscribed thereon the name of the Corporation, the year of its organization, and the words "Corporate Seal" and "Delaware."

Section 5.3 Fiscal Year.

The fiscal year of the Corporation shall be determined by resolution of the Board of Directors.

Section 5.4 Notices and Waivers Thereof.

Whenever any notice is required by the laws of the State of Delaware, the Certificate of Incorporation or these Bylaws to be given by the Corporation to any stockholder, director or officer, such notice, except as otherwise provided by law, may be given personally, by mail, by courier service, or by electronic transmission in accordance with applicable law. Any notice given by electronic mail shall be deemed to have been given when it shall have been directed to such stockholder's, director's or officer's electronic mail address as it appears on the records of the Corporation unless, in the case of a stockholder, such stockholder has notified the Corporation in writing or by electronic transmission of an objection to receiving notice by electronic mail or such notice is prohibited by Section 232(e) of the Delaware General Corporation Law, any notice given by mail shall be deemed to have been given when deposited in the United States mail with postage thereon prepaid directed to such stockholder, director, or officer, as the case may be, at such stockholder's, director's, or officer's, as the case may be, address as it appears in the records of the Corporation, and any notice given by courier service shall be deemed to have been given on the earlier of when such notice is received or left at such stockholder's, director's or officer's, as the case may be, address as it appears in the records of the Corporation. An affidavit of the Secretary or Assistant Secretary or of the transfer agent or other agent of the Corporation that the notice has been given by personal delivery, by mail, by courier service, or by a form of electronic transmission shall, in the absence of fraud, be prima facie evidence of the facts stated therein.

Whenever any notice is required to be given by law, the Certificate of Incorporation, or these Bylaws to the person entitled to such notice, a waiver thereof, in writing signed by the person, or by electronic transmission, whether before or after the meeting or the time stated therein, shall be deemed equivalent in all respects to such notice to the full extent permitted by law. If such waiver is given by electronic transmission, the electronic transmission must either set forth or be submitted with information from which it can be determined that the electronic transmission was authorized by the person waiving notice. In addition, notice of any meeting of the Board of Directors, or any committee thereof, need not be given to any director if such director shall sign the minutes of such meeting or attend the meeting, except that if such director attends a meeting for the express purpose of objecting at the beginning of the meeting to the transaction of any business because the meeting is not lawfully called or convened, then such director shall not be deemed to have waived notice of such meeting.

Section 5.5 <u>Saving Clause</u>.

These Bylaws are subject to the provisions of the Certificate of Incorporation and applicable law. In the event any provision of these Bylaws is inconsistent with the Certificate of Incorporation or the corporate laws of the State of Delaware, such provision shall be invalid to the extent only of such conflict, and such conflict shall not affect the validity of any other provision of these Bylaws.

Section 5.6 Amendments.

In furtherance and not in limitation of the powers conferred by the laws of the State of Delaware, the Board of Directors, by action taken by the affirmative vote of not less than 75% of the members of the Board of Directors then in office, is hereby expressly authorized and empowered to adopt, amend or repeal any provision of the Bylaws of this Corporation.

Subject to the rights of the holders of any series of preferred stock, these Bylaws may be adopted, amended or repealed by the affirmative vote of the holders of not less than 66 2/3% of the total voting power of the then outstanding capital stock of the Corporation entitled to vote thereon; provided, however, that this paragraph shall not apply to, and no vote of the stockholders of the Corporation shall be required to authorize, the adoption, amendment or repeal of any provision of the Bylaws by the Board of Directors in accordance with the preceding paragraph.

Section 5.7 <u>Gender/Number</u>.

As used in these Bylaws, the masculine, feminine, or neuter gender, and the singular and plural number, shall include the other whenever the context so indicates.

Section 5.8 <u>Electronic Transmission</u>.

For purposes of these Bylaws:

- (a) "**electronic transmission**" means any form of communication, not directly involving the physical transmission of paper, including the use of, or participation in, one or more electronic networks or databases (including one or more distributed electronic networks or databases), that creates a record that may be retained, retrieved, and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such recipient through an automated process;
- (b) "electronic mail" means an electronic transmission directed to a unique electronic mail address (which electronic mail shall be deemed to include any files attached thereto and any information hyperlinked to a website if such electronic mail includes the contact information of an officer or agent of the corporation who is available to assist with accessing such files and information); and
- (c) "electronic mail address" means destination, commonly expressed as a string of characters, consisting of a unique user name or mailbox (commonly referred to as the "local part" of the address) and a reference to an internet domain (commonly referred to as the "domain part" of the address), whether or not displayed, to which electronic mail can be sent or delivered.

AMENDMENT AGREEMENT

J.P. Morgan SE as Facility Agent

and

Others

Dated 19 September 2024

Table of Contents

		Page
<u>1.</u>	INTERPRETATION	<u>2</u>
<u>2.</u>	FACILITIES AGREEMENT AMENDMENTS	<u>3</u>
<u>3.</u>	CONFIRMATION	<u>3</u>
<u>4.</u>	REPRESENTATIONS AND WARRANTIES	
<u>5.</u>	STATUS OF DOCUMENTS	
<u>6.</u>	MISCELLANEOUS	
<u>7.</u>	GOVERNING LAW AND SUBMISSION TO JURISDICTION	
SC	CHEDULE 1 PARTIES	
SC	CHEDULE 2 AMENDED AND RESTATED FACILITIES AGREEMENT	

i

THIS AMENDMENT AGREEMENT (this "Agreement") is made on 19 September 2024

BETWEEN:

- (1) **FORMULA ONE MANAGEMENT LIMITED**, a company incorporated in England with registration number 01545332, ("**FOM**");
- (2) **J.P. MORGAN SE**, in its capacity as facility agent under the Facilities Agreement (as defined below) (the "Facility Agent");
- (3) THE ENTITIES listed in Part I of Schedule 1 as TLA Lenders (the "TLA Lenders"); and
- (4) THE ENTITIES listed in Part II of Schedule 1 as RCF Lenders (the "RCF Lenders")

WHEREAS:

- (A) The Parties wish to make certain amendments to the Facilities Agreement as set out in this Agreement.
- (B) FOM is entering into this Agreement for itself and as Obligors' Agent on behalf of the Obligors under and as defined in the Facilities Agreement.

THE PARTIES AGREE AS FOLLOWS:

1. INTERPRETATION

1.1 Definitions

Unless a contrary intention appears in this Agreement, any word or expression defined in the Facilities Agreement will have the same meaning when it is used in this Agreement.

In this Agreement:

"2024 Effective Date" has the meaning given to it in the Amendment Letter.

"Amended and Restated Facilities Agreement" means the Facilities Agreement as amended and restated in the form set out in Schedule 2.

"Amendment Conditions" means the following conditions:

- (i) the 2024 Effective Date has occurred (or will occur simultaneously with the occurrence of the Amendment Effective Date); and
- (ii) the Facility Agent is satisfied (acting reasonably) that the amendment and restatement of the Facilities Agreement in the form set out in Schedule 2 has been consented to by such Lenders as are required in order for such amendment and restatement to be effected by the Facility Agent in accordance with the Facilities Agreement.

- "Amendment Effective Date" means the date on which the Facility Agent confirms to FOM that it is satisfied that the Amendment Conditions are satisfied (acting reasonably).
- "Amendment Letter" means the amendment request letter from FOM provided to the Facility Agent and Goldman Sachs Bank USA dated on or about the date of this Agreement.
- "Facilities Agreement" means the facilities agreement dated 21 November 2006 between, amongst others, FOM and J.P. Morgan SE (as facility agent) and NatWest Markets plc (as security agent) as amended and/or restated from time to time.
- "Party" means a party to this Agreement.

1.2 Construction

Clause 1.2 (*Construction*) of the Facilities Agreement will apply to this Agreement as if set out in full in this Agreement mutatis mutandis.

2. FACILITIES AGREEMENT AMENDMENTS

With effect on the Amendment Effective Date (in each case without any requirement for any further action or notice):

- (a) the Facilities Agreement shall be amended and restated in the form set out in Schedule 2 and so that it shall be read and construed for all purposes as set out in Schedule 2; and
- (b) the rights and obligations of the parties to the Facilities Agreement shall be governed by the Amended and Restated Facilities Agreement.

3. CONFIRMATION

The Facility Agent shall promptly notify FOM when it is satisfied (acting reasonably) that the Amendment Conditions are satisfied.

4. REPRESENTATIONS AND WARRANTIES

FOM makes each of the representations and warranties in clause 24 of the Amended and Restated Facilities Agreement (other than clause 24.9) to each Finance Party on the Amendment Effective Date and acknowledges that each applicable Finance Party has agreed to the amendments effected by this Agreement and the Facility Agent has entered into this Agreement in full reliance on those representations and warranties.

5. STATUS OF DOCUMENTS

5.1 Continuing Effect

Except as varied by this Agreement and the Amendment Letter and the other documents entered into in connection with them, the Facilities Agreement and each other Finance Document remain in full force and effect unamended and unwaived and any reference in

any Finance Document to any such agreement or provision thereof will be construed as a reference to such agreement, or that provision, as amended by this Agreement.

5.2 Finance Document

This Agreement is designated as a "Finance Document" for the purposes of the Facilities Agreement.

6. MISCELLANEOUS

6.1 Invalidity

If any provision of this Agreement is or becomes invalid, illegal or unenforceable in any respect under any law, the validity, legality and enforceability of the remaining provisions shall not be affected or impaired in any way.

6.2 Counterparts

This Agreement may be executed in any number of counterparts and all of those counterparts taken together will be deemed to constitute one and the same instrument.

6.3 Third Party Rights

The Contracts (Rights of Third Parties) Act 1999 will apply to this Agreement for the benefit of all Finance Parties and Obligors, but no other person (other than the Parties, the Finance Parties and Obligors) shall have any rights under it. However, the consent of a person who is not a party to this Agreement is not required to amend or vary this Agreement.

6.4 Confirmations relating to the Facilities Agreement

FOM confirms agreement to the amendments pursuant to this Agreement for the purposes of the Facilities Agreement and as Obligors' Agent under and as defined in the Facilities Agreement.

6.5 Creditor Confirmations

Each TLA Lender and RCF Lender irrevocably consents to the amendment and restatement of the Facilities Agreement as set out in Schedule 2 and authorizes the Facility Agent to enter into this Agreement and any documents required in relation hereto.

The Facility Agent agrees to the terms of this Agreement for itself and each other Finance Party.

7. GOVERNING LAW AND SUBMISSION TO JURISDICTION

7.1 Governing Law

This Agreement and any non-contractual obligations arising out of or in connection with it are governed by English law.

7.2 Jurisdiction of English Courts

- (a) The courts of England have exclusive jurisdiction to settle any dispute arising out of or in connection with this Agreement (including a dispute regarding the existence, validity or termination of this Agreement) (a "Dispute").
- (b) The parties to this Agreement agree that the courts of England are the most appropriate and convenient courts to settle Disputes and accordingly no party will argue to the contrary.

IN WITNESS whereof this Agreement has been duly executed by each of the Parties on the date first above written.

SCHEDULE 1 PARTIES

Part I – TLA Lenders

Bank of America N.A., London Branch

BNP Paribas

CIBC Capital Markets (Europe) S.A.

Citibank, N.A.

Credit Agricole Corporate and Investment Bank

Goldman Sachs Bank USA

JPMorgan Chase Bank, N.A.

Mizuho Bank, Ltd.

Morgan Stanley Senior Funding, Inc.

MUFG Bank, Ltd.

National Westminster Bank plc

Société Générale

The Toronto-Dominion Bank, New York Branch

Truist Bank

Part II – RCF Lenders

Bank of America N.A., London Branch

CIBC Capital Markets (Europe) S.A.

BNP Paribas

Citibank, N.A.
Credit Agricole Corporate and Investment Bank
Goldman Sachs Lending Partners LLC
JPMorgan Chase Bank, N.A., London Branch
Mizuho Bank, Ltd.
Morgan Stanley Senior Funding, Inc.
MUFG Bank, Ltd.
National Westminster Bank plc
Royal Bank of Canada
Société Générale
The Toronto-Dominion Bank, New York Branch
Truist Bank
7

SCHEDULE 2 AMENDED AND RESTATED FACILITIES AGREEMENT

Signature pages to the Amendment Agreement

FOM	
FORMULA ONE MANAGEMENT LIMITED as Obligors' Agent acting by a director:	
[Signature Page to Ame	endment Agreement]

The Facility Agent			
J.P. MORGAN SE acting by an authorised signatory:			
	[Signature Page to An	nendment Agreement]	

The TLA Lenders

BANK OF AMERICA N.A., LONDON BRANCH acting by an authorised signatory:			
	[Signature Page to Amendment Agreement]		

BNP PARIBAS S.A. acting by an authorised signatory:	
	[Signature Page to Amendment Agreement]

TLA Lender CIBC CAPITAL MARKETS (EUROPE) S.A. acting by an authorised signatory:

IC:

[Signature Page to Amendment Agreement]

TLA Lender			
CITIBANK N.A. acting by an authorised signatory:			
	[Signature Page to An	nendment Agreement]	

CREDIT AGRICOLE CORPOR INVESTMENT BANK acting by an authorised signatory:	RATE AND
	[Signature Page to Amendment Agreement]

GOLDMAN SACHS BANK USA acting by an authorised signatory: [Signature Page to Amendment Agreement]

TLA Lender	
JPMORGAN CHASE BANK, N.A. acting by an authorised signatory:	
[Signature Page to	o Amendment Agreement]

MIZUHO BANK, LTD. acting by an authorised signatory: [Signature Page to Amendment Agreement]

MORGAN STANLEY SENIOR acting by an authorised signatory:	FUNDING, INC.		
	[Signature Page to Am	endment Agreement]	

MUFG BANK, LTD. acting by an authorised signatory:			
	_		
	[Signature Page to Amer	ndment Agreement]	

NATIONAL WESTMINSTER B acting by an authorised signatory:	ANK PLC
	[Signature Page to Amendment Agreement]

TLA Lender	
SOCIÉTÉ GÉNÉRALE S.A. acting by an authorised signatory:	
	[Signature Page to Amendment Agreement]

THE TORONTO-DOMINION E YORK BRANCH acting by an authorised signatory:	BANK, NEW
	[Signature Page to Amendment Agreement]

TRUIST BANK acting by an authorised signatory:	
	[Signature Page to Amendment Agreement]

The RCF Lenders

BANK OF AMERICA N.A., LOR acting by an authorised signatory:	NDON BRANCH		
	[Signature Page to Am	endment Agreement]	

BNP PARIBAS S.A. acting by an authorised signatory:	
	[Signature Page to Amendment Agreement]

RCF Lender

RCF Lender

CIBC CAPITAL MARKETS (El acting by an authorised signatory:	UROPE) S.A.
	[Signature Page to Amendment Agreement]

CITIBANK N.A. acting by an authorised signatory: [Signature Page to Amendment Agreement]

RCF Lender

CREDIT AGRICOLE CORPOR INVESTMENT BANK acting by an authorised signatory:	RATE AND
	[Signature Page to Amendment Agreement]

GOLDMAN SACHS BANK USA acting by an authorised signatory:	
[.	Signature Page to Amendment Agreement]

JPMORGAN CHASE BANK, N BRANCH acting by an authorised signatory:	.A., LONDON
	[Signature Page to Amendment Agreement]

MIZUHO BANK, LTD. acting by an authorised signatory: [Signature Page to Amendment Agreement]

MORGAN STANLEY SENIOR acting by an authorised signatory:	FUNDING, INC.		
	[Signature Page to Amendment A	tgreement]	

MUFG BANK, LTD., NEW YOR acting by an authorised signatory:			
	[Signature Page to Amer	ndment Agreement]	

ROYAL BANK OF CANADA acting by an authorised signatory: [Signature Page to Amendment Agreement]

SOCIÉTÉ GÉNÉRALE S.A. acting by an authorised signatory: [Signature Page to Amendment Agreement]

THE TORONTO-DOMINION E YORK BRANCH acting by an authorised signatory:	,
	[Signature Page to Amendment Agreement]

TRUIST BANK acting by an authorised signatory: [Signature Page to Amendment Agreement]

EXECUTION VERSION

2nd AMENDMENT AGREEMENT

J.P. Morgan SE as Facility Agent

and

Formula One Management Limited as Obligors' Agent

Dated 9 October 2024

Table of Contents

		Page
<u>1.</u>	<u>INTERPRETATION</u>	<u>i</u>
<u>2.</u>	FACILITIES AGREEMENT AMENDMENTS	<u>2</u>
<u>3.</u>	STATUS OF DOCUMENTS	<u>2</u>
<u>4.</u>	MISCELLANEOUS	<u>3</u>
<u>5.</u>	GOVERNING LAW AND SUBMISSION TO JURISDICTION	<u>3</u>

THIS 2nd AMENDMENT AGREEMENT (this "Agreement") is made on 9 October 2024

BETWEEN:

- (1) **FORMULA ONE MANAGEMENT LIMITED**, a company incorporated in England with registration number 01545332, ("**FOM**"); and
- (2) J.P. MORGAN SE, in its capacity as facility agent under the Facilities Agreement (as defined below) (the "Facility Agent").

WHEREAS:

- (A) The Facilities Agreement has been amended and restated on the date hereof and FOM and the Facility Agent wish to make a further amendment to the Facilities Agreement as set out in this Agreement pursuant to Clause 40.4 (*Technical Amendments*) of the Facilities Agreement.
- (B) FOM is entering into this Agreement for itself and as Obligors' Agent on behalf of the Obligors under and as defined in the Facilities Agreement.

THE PARTIES AGREE AS FOLLOWS:

1. INTERPRETATION

1.1 Definitions

Unless a contrary intention appears in this Agreement, any word or expression defined in the Facilities Agreement will have the same meaning when it is used in this Agreement.

In this Agreement:

"Facilities Agreement" means the facilities agreement dated 21 November 2006 between, amongst others, FOM and J.P. Morgan SE (as facility agent) and NatWest Markets plc (as security agent) as amended and restated on 19 September 2024.

i

"Party" means a party to this Agreement.

1.2 Construction

Clause 1.2 (*Construction*) of the Facilities Agreement will apply to this Agreement as if set out in full in this Agreement mutatis mutandis.

2. FACILITIES AGREEMENT AMENDMENTS

With effect on the date of this Agreement (in each case without any requirement for any further action or notice) Clause 29.3 (*Assignment or Transfer Fee*) of the Facilities Agreement shall be deleted and replaced with the following:

"29.3 Assignment or Transfer Fee

Unless the Facility Agent otherwise agrees and excluding an assignment or transfer:

- (a) to an Affiliate of a Lender;
- (b) to a Related Fund: or
- (c) made within the later of (x) 30 days of the date of the Dorna Acquisition and (y) the period separately provided in the master consent agreement between the 2024 Incremental Term Lender, the Facility Agent and the Obligors' Agent dated the 2024 Effective Date,

the New Lender shall, on the date upon which an assignment or transfer takes effect, pay to the Facility Agent (for its own account) a fee of \$3,500."

3. STATUS OF DOCUMENTS

3.1 Continuing Effect

Except as varied by this Agreement, the Facilities Agreement and each other Finance Document remain in full force and effect unamended and unwaived and any reference in any Finance Document to any such agreement or provision thereof will be construed as a reference to such agreement, or that provision, as amended by this Agreement.

3.2 Finance Document

This Agreement is designated as a "Finance Document" for the purposes of the Facilities Agreement.

4. MISCELLANEOUS

4.1 Invalidity

If any provision of this Agreement is or becomes invalid, illegal or unenforceable in any respect under any law, the validity, legality and enforceability of the remaining provisions shall not be affected or impaired in any way.

4.2 Counterparts

This Agreement may be executed in any number of counterparts and all of those counterparts taken together will be deemed to constitute one and the same instrument.

4.3 Third Party Rights

The Contracts (Rights of Third Parties) Act 1999 will apply to this Agreement for the benefit of all Finance Parties and Obligors, but no other person (other than the Parties, the Finance Parties and Obligors) shall have any rights under it. However, the consent of a person who is not a party to this Agreement is not required to amend or vary this Agreement.

4.4 Confirmations relating to the Facilities Agreement

FOM confirms agreement to the amendments pursuant to this Agreement for the purposes of the Facilities Agreement and as Obligors' Agent under and as defined in the Facilities Agreement.

4.5 Creditor Confirmations

The Facility Agent agrees to the terms of this Agreement for itself and each other Finance Party.

5. GOVERNING LAW AND SUBMISSION TO JURISDICTION

5.1 Governing Law

This Agreement and any non-contractual obligations arising out of or in connection with it are governed by English law.

5.2 Jurisdiction of English Courts

- (a) The courts of England have exclusive jurisdiction to settle any dispute arising out of or in connection with this Agreement (including a dispute regarding the existence, validity or termination of this Agreement) (a "**Dispute**").
- (b) The parties to this Agreement agree that the courts of England are the most appropriate and convenient courts to settle Disputes and accordingly no party will argue to the contrary.

IN WITNESS written.	whereof this	Agreement ha	s been dul	y executed by	each of the	Parties of	n the date	first above
			4					

Signature pages to the 2nd Amendment Agreement

FOM	
FORMULA ONE MANAGEMENT LIMITED as Obligors' Agent acting by a director:	
[Signature Page to 2 nd An	nendment Agreement]

The Facility Agent	
J.P. MORGAN SE acting by an authorised signatory:	
[Signature Page to 2 nd Ame	ndment Agreement]

Amended and Restated First Lien Facilities Agreement

J.P. MORGAN SE as Facility Agent,

GOLDMAN SACHS BANK USA

as Lead Left Arranger, Lead Bookrunner and Sole Syndication Agent with respect to Term Facility B,

JPMORGAN CHASE BANK, N.A.

as Lead Left Arranger and Lead Bookrunner with respect to Term Facility A and Facility RCF, and as Joint Lead Arranger and Joint Bookrunner with respect to Term Facility B,

GOLDMAN SACHS LENDING PARTNERS LLC

as Joint Lead Arranger, Joint Bookrunner and Co-Syndication Agent with respect to Term Facility A and Facility RCF,

BNP PARIBAS SECURITIES CORP.

as Joint Lead Arranger and Joint Bookrunner with respect to Term Facility B, Term Facility A and Facility RCF, and, solely with respect to Term Facility A and Facility RCF, Co-Syndication Agent,

BANK OF AMERICA, N.A., LONDON BRANCH, MIZUHO BANK, LTD.,

SOCIÉTÉ GÉNÉRALE,

TD SECURITIES (USA) LLC,

TRUIST BANK (solely as Co-Documentation Agent),

TRUIST SECURITIES, INC. (solely as Joint Lead Arranger and Joint Bookrunner),

CITIBANK N.A.,

CREDIT AGRICOLE CORPORATE AND INVESTMENT BANK,

MORGAN STANLEY SENIOR FUNDING, INC.,

MUFG BANK, LTD.,

CIBC CAPITAL MARKETS (EUROPE) S.A.,

BANCO SANTANDER, S.A., NEW YORK BRANCH,

SUMITOMO MITSUI BANKING CORPORATION (solely as Joint Lead Arranger and Joint Bookrunner),

ROYAL BANK OF CANADA,

each as Joint Lead Arranger and Joint Bookrunner with respect to Term Facility B, Term Facility A and Facility RCF, and, solely with respect to Term Facility A and Facility RCF, Co-Documentation Agent,

NATWEST MARKETS PLC

as Joint Lead Arranger and Joint Bookrunner with respect to Term Facility B,

NATIONAL WESTMINSTER BANK PLC

as Joint Lead Arranger, Joint Bookrunner and as Co-Documentation Agent with respect to Term Facility A and Facility RCF

and

NATWEST MARKETS PLC

as Security Agent

Dated 21 November 2006 as most recently amended and restated by an amendment agreement dated 19 September 2024

CONTENTS

CLAUSE		PAGE
1.	DEFINITIONS AND INTERPRETATION	1
2.	THE FACILITIES	
3.	PURPOSE	
4.	CONDITIONS OF UTILISATION	
5.	UTILISATION - LOANS	
6.	UTILISATION - LETTERS OF CREDIT	
7.	LETTERS OF CREDIT	
8.	OPTIONAL CURRENCIES	
9.	ANCILLARY FACILITIES	
10.	REPAYMENT	
11.	ILLEGALITY, VOLUNTARY PREPAYMENT AND CANCELLATION	
12.	MANDATORY PREPAYMENT	
13.	RESTRICTIONS	
14.	INTEREST	
15.	INTEREST PERIODS	
16.	CHANGES TO THE CALCULATION OF INTEREST	
10. 17.	FEES	
18.	TAX GROSS-UP AND INDEMNITIES	
	INCREASED COSTS	
19. 20.		
20.	OTHER INDEMNITIES	
	MITIGATION BY THE LENDERS	
22.	COSTS AND EXPENSES	
23.	GUARANTEE AND INDEMNITY	
24.	REPRESENTATIONS	
25.	INFORMATION UNDERTAKINGS	
26.	FINANCIAL DEFINITIONS AND FINANCIAL COVENANT	
27.	GENERAL UNDERTAKINGS	
28.	EVENTS OF DEFAULT	
29.	CHANGES TO THE LENDERS	
30.	CHANGES TO THE OBLIGORS	
31.	ROLE OF THE FACILITY AGENT, THE SECURITY AGENT, THE ARRANGERS, THE ISSUING BANK	
	OTHERS	
32.	CONDUCT OF BUSINESS BY THE FINANCE PARTIES	
33.	SHARING AMONG THE FINANCE PARTIES	
34.	PAYMENT MECHANICS	
35.	SET-OFF	
36.	NOTICES	
37.	CALCULATIONS AND CERTIFICATES	
38.	PARTIAL INVALIDITY	
39.	REMEDIES AND WAIVERS	
40.	AMENDMENTS AND WAIVERS	
41.	COUNTERPARTS	
42.	GOVERNING LAW	
43.	ENFORCEMENT	
44.	NET SHORT LENDERS	
45.	BAIL-IN	
	E 1	
	nal Parties	
	he Original Obligors	
	otal Commitments	
	CF Lenders	
	erm Facility A Lenders	
SCHEDUL	E 2	216

Conditions Precedent	
Part 1A - Conditions Precedent to Signing of the Agreement	216
Part 1B - Conditions Precedent to Initial Utilisation	
Part 2A - Conditions Subsequent	218
Part 2B – Conditions Subsequent	219
Part 2C - Conditions Subsequent	
Part 3 – Conditions Precedent Required to be Delivered by an Additional Obligor	221
Part 4A – transaction security documents	
Part 4B – Sixth Restatement Agreement Transaction Security Documents	227
1. ENGLISH LAW GOVERNED TRANSACTION SECURITY DOCUMENTS	227
2. JERSEY LAW GOVERNED TRANSACTION SECURITY DOCUMENTS	
3. LUXEMBOURG LAW GOVERNED TRANSACTION SECURITY DOCUMENTS	227
4. SWISS LAW GOVERNED TRANSACTION SECURITY DOCUMENTS	227
SCHEDULE 3	228
Requests	
Part 1A - Utilisation Request Loans	228
Part 1B - Utilisation Request Letters of Credit	229
Part 2 - Selection Notice	230
SCHEDULE 4	231
Form of Transfer Certificate	231
SCHEDULE 5	236
Form of Accession Letter	236
SCHEDULE 6	237
Form of Resignation Letter	237
SCHEDULE 7	
Form of Compliance Certificate	238
SCHEDULE 8	239
Agreed Security Principles	239
SCHEDULE 9	
Timetables	244
Part 1 - Loans	
Part 2 – Letters of Credit	
SCHEDULE 10	
Form of Letter of Credit	
SCHEDULE 11	
Incremental Facility Commitment Letter	
SCHEDULE 12	
List of Designated Entities	
SCHEDULE 13	
Form of Designated Entity Accession Agreement	
SCHEDULE 14	
Part 1 - Permitted Acquisition	
Part 2 – Permitted Disposal	
Part 3 – Permitted Financial Indebtedness	
Part 4 – Permitted Guarantees	
Part 5 – Permitted Joint Ventures	
Part 6 – Permitted Loans	
Part 7 – Permitted Security	
Part 8 – Cash Movement	
SCHEDULE 15	
Notices	
Part 1 – Obligors' Agent	
Part 2 – Facility Agent and Security Agent	255

THIS AGREEMENT is made on 21 November 2006 as most recently amended and restated by an amendment agreement dated 19 September 2024

BETWEEN:

- (1) DELTA TOPCO LIMITED (a company incorporated in Jersey with registered number 95136) (and any permitted successor, the "Parent");
- (2) **THE COMPANIES** listed in part 1 of schedule 1 (The Original Parties) as the original borrower and the RCF borrowers as at the 2022 Effective Date and as modified from time to time prior to the 2024 Effective Date (the "Original Borrowers");
- (3) **THE COMPANIES** listed in part 1 of schedule 1 (The Original Parties) as original guarantors as at the 2022 Effective Date and as modified from time to time prior to the 2024 Effective Date (the **"Original Guarantors"**);
- (4) THE FINANCIAL INSTITUTIONS who are "2024 Effective Date Original Lenders" (as defined below);
- (5) J.P. MORGAN SE as facility agent of the other Finance Parties (the "Facility Agent");
- (6) NATWEST MARKETS PLC as security agent of the other Secured Parties (the "Security Agent"); and
- (7) J.P. MORGAN CHASE BANK, N.A., LONDON BRANCH as Issuing Bank.

THE PARTIES AGREE AS FOLLOWS:

1. **DEFINITIONS AND INTERPRETATION**

1.1 Definitions

In this agreement:

"100 Year Agreements" means the various agreements involving FOA, FOWC, other members of the SLEC Group and FIA pursuant to which, among other things, FOAM grants to FOWC the right to undertake the Business for 100 years from 1 January 2011 comprising the Umbrella Agreement and five other agreements described as the commercial agreement, regulatory agreement, FIA transfer agreement, SLEC group asset transfer agreement and deed of release and waiver;

"2022 Amendment Agreement" means the amendment agreement designated as such by the Obligors' Agent and the Facility Agent;

"2022 Amendment Request" means the amendment request letter designated as such by the Obligors' Agent and the Facility Agent;

"2022 Effective Date" has the meaning given to it in the 2022 Amendment Request;

"2023 Amendment Letter" means the letter from the Obligor's Agent to, amongst others, the Facility Agent dated 4 October 2023 pursuant to which certain amendments, including to the definition of Margin, were made to this agreement;

"2023 Continued Term Facility B Commitment" means a "Continued Term Facility B Commitment" as defined in the 2023 Amendment Letter;

"2023 Continued Term Facility B Loan Participation" means a "Continued Term Facility B Loan Participation" as defined in the 2023 Amendment Letter;

- "2023 Effective Date " has the meaning given to it in the 2023 Amendment Letter;
- **"2023 Funded Term Facility B Commitments**" means the "Funded Term Facility B Commitments" as defined in the 2023 Amendment Letter in an amount as of the 2023 Effective Date equal to the aggregate outstanding 2023 Non-Continued Term Facility B Loan Participations;
- **"2023 Funded Term Facility B Loan Participation**" means a "Funded Term Facility B Loan Participation" as defined in the 2023 Amendment Letter;
- **"2023 Incremental Facility Commitment Letter"** means the Incremental Facility Commitment Letter dated 4 October 2023 pursuant to which 2023 Funded Term Facility B Commitments were made available by the 2023 Incremental Facility Lenders;
- "2023 Incremental Facility Lenders" means the 2023 Incremental Facility Lenders as defined in the 2023 Amendment Letter;
- **"2023 Non-Continued Term Facility B Loan Participation**" means a "Non-Continued Term Facility B Loan Participation" as defined in the 2023 Amendment Letter;
- **"2023 Term Facility B Refinancing Documents"** means, collectivelly, the 2023 Amendment Letter and the 2023 Incremental Facility Commitment Letter;
- **"2024 Amendment Agreement"** means the amendment agreement between amongst others the Obligors' Agent and the Facility Agent entered into on or about the 2024 Effective Date.
- "2024 Amendment Request" means the amendment request letter designated as such by the Obligors' Agent and the Facility Agent;
- "2024 Effective Date" has the meaning given to it in the 2024 Amendment Request;
- "2024 Effective Date Original Lender" means each Original RCF Lender, each Original Term Loan A Lender and each 2024 Incremental Term Lender;
- **"2024 Incremental Term Lenders"** each bank, financial institution or other entity listed in Part 5 of schedule 1 as a 2024 Incremental Term Lender;
- "2024 Incremental Facility Commitment Letter" means the Incremental Facility Commitment Letter dated on or about the 2024 Effective Date between the Facility Agent, Delta 2, the Obligors' Agent and the 2024 Incremental Term Lenders;
- "ABR", when used in reference to any Loan, refers to whether such Loan bear interest at a rate determined by reference to the Alternate Base Rate. All ABR Loans shall be denominated in Dollars;

"Acceptable Bank" means:

- (a) a bank or financial institution which has a rating for its short-term unsecured and non-credit enhanced debt obligations of A-2 or higher by S&P, F-2 or higher by Fitch or P-2 or higher by Moody's or a comparable rating from an internationally recognised credit rating agency;
- (b) (i) any bank or financial institution organized under the laws of the United States or any state thereof, the United Kingdom or the European Union and (ii) any branch of such bank managed and controlled by a bank referred to in the immediately preceding clause (i); or
- (c) any other bank or financial institution approved by the Facility Agent;

"Accession Letter" means a document substantially in the form set out in schedule 5 (Form of Accession Letter);

"Accounting Principles" means:

- (a) in the case of individual accounts of members of the Group, at the election of such Group member, the accounting standards generally accepted in the jurisdiction of incorporation of the member of the Group or International Financial Reporting Standards; and
- in the case of the consolidated accounts of the Group, at the election of the Obligors' Agent, UK generally accepted accounting principles, US generally accepted accounting principles or International Financial Reporting Standards;
- "Additional Borrower" means a company which becomes a Borrower in accordance with clause 30 (Changes to the Obligors);
- "Additional Debt" means Additional First Lien Debt, Additional Second Lien Debt and/or Additional Unsecured Debt (including in each case, Permitted Refinancing Indebtedness);
- "Additional Debt Maturity Exclusion" has the meaning given to it in clause 2.3(a) (Additional Debt);
- "Additional Debt Ratio Tests" has the meaning given to it in clause 2.3(a) (Additional Debt);
- "Additional First Lien Debt" has the meaning given to it in clause 2.3(a)(i) (Additional Debt);
- "Additional First Lien Debt Maturity Date" has the meaning given to it in clause 2.3(a) (Additional Debt);
- "Additional First Lien Debt Ratio Test" has the meaning given to it in clause 2.3(a) (Additional Debt);
- "Additional Guarantor" means a company which becomes a Guarantor in accordance with clause 30 (Changes to the Obligors);
- "Additional Non-First Lien Debt Maturity Date" has the meaning given to it in clause 2.3(a) (Additional Debt);
- "Additional Obligor" means an Additional Borrower or an Additional Guarantor;
- "Additional Second Lien Debt" has the meaning given to it in clause 2.3(a)(ii) (Additional Debt);
- "Additional Second Lien Debt Ratio Test" has the meaning given to it in clause 2.3(a) (Additional Debt);
- "Additional Unsecured Debt" has the meaning given to it in clause 2.3(a)(iii) (Additional Debt);
- "Additional Unsecured Debt Ratio Test" has the meaning given to it in clause 2.3(a) (Additional Debt);
- "Adjustment Date" has the meaning given to that term in the definition of Pricing Grid.
- "Affiliate" means, in relation to any Undertaking, a Subsidiary of that Undertaking or a Parent Undertaking of that Undertaking or any other Subsidiary of that Parent Undertaking;

- "Affiliated Lenders" has the meaning given to that term in clause 29.12 (Permitted Debt Purchase Transactions).
- "Affiliated Persons" means, with respect to any specified person, (a) such specified person's parents, spouse, siblings, descendants, step children, step grandchildren, nieces and nephews and their respective spouses, (b) the estate, legatees and devisees of such specified person and each of the persons referred to in clause (a) and, in the event of the incompetence or death of any of the persons described in clause (a), such person's executor, administrator, committee or other personal representative or similar fiduciary, (c) any trusts or private foundations created primarily for the benefit of, or controlled at the time of creation by, any of the persons described in the above clause (a) or (b) of this definition, or any trusts or private foundations created primarily for the benefit of any such trust or private foundation or for charitable purposes, and (d) any company, partnership, trust or other entity or investment vehicle controlled by any of the persons referred to in clause (a), (b), (c) or the holdings of which are for the primary benefit of any of such persons.
- "Agreed Security Principles" means the principles set out in schedule 8 (Agreed Security Principles);
- "Alternate Base Rate" means, for any day, a rate per annum equal to the greatest of (a) the Prime Rate in effect on such day, (b) the NYFRB Rate in effect on such day plus ½ of 1% and (c) the Term SOFR Rate for a one month Interest Period as published two U.S. Government Securities Business Days prior to such day (or if such day is not a Business Day, the immediately preceding Business Day); provided that for the purpose of this definition, the Term SOFR Rate for any day shall be based on the Term SOFR Reference Rate at approximately 5:00 a.m. Chicago time on such day (or any amended publication time for the Term SOFR Reference Rate, as specified by the CME Term SOFR Administrator in the Term SOFR Reference Rate methodology). Any change in the Alternate Base Rate due to a change in the Prime Rate, the NYFRB Rate or the Term SOFR Rate shall be effective from and including the effective date of such change in the Prime Rate, the NYFRB Rate or the Term SOFR Rate, respectively. If the Alternate Base Rate is being used as an alternate rate of interest pursuant to Clause 16.3 (for the avoidance of doubt, only until the Benchmark Replacement has been determined pursuant to Clause 16.3), then the Alternate Base Rate shall be the greater of clauses (a) and (b) above and shall be determined without reference to clause (c) above. For the avoidance of doubt, if the Alternate Base Rate as determined pursuant to the foregoing would be less than zero, such rate shall be deemed to be zero for purposes of this agreement.
- "Ancillary Commencement Date" means, in relation to an Ancillary Facility, the date on which that Ancillary Facility is first made available, which date shall be a Business Day within the Availability Period for the Revolving Facility from which that Ancillary Facility has been established;
- "Ancillary Commitment" means, in relation to an Ancillary Lender and an Ancillary Facility, the maximum Base Currency Amount which that Ancillary Lender has agreed (whether or not subject to satisfaction of conditions precedent) to make available from time to time under an Ancillary Facility and which has been authorised as such under clause 9 (Ancillary Facilities), to the extent that amount is not cancelled or reduced under this agreement or the Ancillary Documents relating to that Ancillary Facility;
- "Ancillary Document" means each document relating to or evidencing the terms of an Ancillary Facility;
- "Ancillary Facility" means any ancillary facility made available by an Ancillary Lender in accordance with clause 9 (Ancillary Facilities);
- "Ancillary Lender" means each Lender (or Affiliate of a Lender) which makes available an Ancillary Facility in accordance with clause 9 (Ancillary Facilities);

- "Ancillary Outstandings" means, at any time, in relation to an Ancillary Lender and an Ancillary Facility the aggregate of the equivalents (as calculated by that Ancillary Lender) in the Base Currency of the following amounts outstanding under that Ancillary Facility then in force:
 - (a) the principal amount outstanding under each overdraft facility or other on-demand short term loan facility (net of any credit balances on any account of any Borrower of an Ancillary Facility with the Ancillary Lender making available that Ancillary Facility to the extent that such credit balance is freely available to be set off by that Ancillary Lender against liabilities owed to it by that Borrower under that Ancillary Facility);
 - (b) the face amount of each guarantee, bond and letter of credit under that Ancillary Facility; and
 - (c) the amount fairly representing the aggregate outstanding exposure (excluding interest and similar charges) of that Ancillary Lender under each other type of accommodation provided under that Ancillary Facility,

in each case as determined by such Ancillary Lender acting reasonably in accordance with its normal banking practice and in accordance with the relevant Ancillary Document;

"Annual Financial Statements" means the financial statements for a Financial Year delivered pursuant to clause 25.1(a) (Financial Statements);

"Anti-Terrorism Law" means each of:

- (a) the Executive Order;
- (b) the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56 (commonly known as the "USA PATRIOT Act");
- (c) the Money Laundering Control Act of 1986, Public Law 99-570;
- (d) the International Emergency Economic Powers Act, 50 U.S.C. §§ 1701 et seq, the Trading with the Enemy Act, 50 U.S.C. App. §§ 1 et seq, any Executive Order or regulation promulgated thereunder and administered by OFAC; and
- (e) any similar law enacted in the United States of America after the Sixth Restatement Date;
- "Arrangers" means the Term Facility A Arrangers and the Term Facility B Arrangers as defined in the 2022 Amendment Request;
- "ASM" means Allsport Management SA a company incorporated in Switzerland with registered number CHE-103.211.396;
- "ASM Dissolution" means the dissolution and entry into liquidation of ASM (other than as a result of ASM being placed into insolvent liquidation), as evidenced by corresponding minutes of a shareholder meeting of ASM, provided furthermore that Formula One Management Limited confirms in writing to the Facility Agent that ASM is a dormant company and either owns no assets or assets of only negligible value;
- "Auditors" means (A) one of PwC, Ernst & Young, KPMG or Deloitte or (B) such other firm that is either (x) approved in advance by the Majority Lenders (such approval not to be unreasonably withheld or delayed) or (y) an independent public accountant of recognized standing;

"Authorisation" means an authorisation, consent, approval, resolution, licence, exemption, filing, notarisation or registration;

"Availability Period" means:

- (a) in relation to the Term Facility A, the 2022 Effective Date;
- (b) in relation to the Term Facility B, the 2022 Effective Date except for the 2023 Funded Term Facility B Commitments in which case it means the 2023 Effective Date and except for the Term Facility B Commitments established pursuant to the 2024 Incremental Facility Commitment Letter in which case it means the 2024 Effective Date;
- (c) in relation to Facility RCF, the period from and including the 2022 Effective Date to the applicable Termination Date;
- (d) in relation to each Incremental Facility, the availability period agreed with the relevant Incremental Facility Lenders in the relevant Incremental Facility Commitment Letter (or otherwise agreed by the relevant Borrower(s) and the relevant Incremental Facility Lender(s) from time to time);

"Available Amount Basket" means, on any date of calculation, for the period (taken as one period) from December 31, 2021 to the end of the Group's most recently ended Financial Quarter for which financial statements have been delivered pursuant to clause 25.1 (Financial Statements) as of such date, an amount equal to (without duplication):

- (a) 100% of Consolidated EBITDA of the Group less 1.3 times the Consolidated Interest Expense for the same period (which amount shall not be less than zero for any Financial Year and shall be deemed to be zero for any Financial Year in respect of which it would otherwise be a negative number) (this clause (a), the "EBITDA Builder Prong"), plus
- (b) (x) the aggregate amount of proceeds received by the Group from any New Shareholder Injections in the form of Qualified Equity or cash capital contributions after the 2022 Effective Date and (y) the fair market value of Cash Equivalent Investments, marketable securities or other property or assets, in each case received by the Group to the extent such Cash Equivalent Investments, marketable securities or other property or assets are contributed as a capital contribution to any Group Company or in return for any New Shareholder Injections in the form of Qualified Equity, plus
- (c) the proceeds of Financial Indebtedness and Disqualified Equity of the Group, which have been exchanged or converted into Qualified Equity of any Group Company together with the fair market value of any Cash Equivalent Investments and any property or assets received by any Group Company upon such exchange or conversion, <u>plus</u>
- (d) the aggregate of all cash dividends and distributions and the fair market value of any Cash Equivalent Investments and any property or assets received by members of the Group from Permitted Joint Ventures; plus
- (e) the proceeds received by the Group from sales of acquisitions, investments or loans made with the Available Amount Basket after the 2022 Effective Date (but in any event not to exceed the amount of such acquisition, investments or loan), <u>plus</u>
- (f) the aggregate of all cash and the fair market value of Cash Equivalent Investments and any property or assets received by the Group from returns, profits, distributions and similar amounts on acquisitions, investments or loans made with the Available Amount Basket after the 2022 Effective Date, <u>plus</u>

- (g) the acquisitions, investments or loans of any Group Company in any Unrestricted Subsidiary that has been redesignated as a Group Company or that has been merged or consolidated into a Group Company after the 2022 Effective Date (but in any event not to exceed the amount of such acquisition, investment or loan), plus
- (h) (i) the fair market value of the assets of any Unrestricted Subsidiary that have been transferred, conveyed or otherwise distributed to a Group Company to the extent the acquisition, investment or loan in such Unrestricted Subsidiary was made with the Available Amount Basket after the 2022 Effective Date and (ii) the proceeds of any Disposals of any Unrestricted Subsidiary (including the issuance or sale of equity interest thereof) received by any Group Company to the extent the acquisition, investment or loan in such Unrestricted Subsidiary was made with the Available Amount Basket after the 2022 Effective Date, plus
- (i) any Declined Amounts, plus
- (j) the greater of (1) \$300,000,000 (or its equivalent in other currencies) and (2) 60% of the then most recently reported Consolidated EBITDA, minus
- (k) the sum of all prior usages of the Available Amount Basket made after the 2022 Effective Date pursuant to clause (n) of Permitted Acquisition, clause (f) of Permitted Distribution, the definition of Permitted Joint Venture and clause (n) of the definition of Permitted Loan.

"Available Commitment" means, in relation to a Facility, a Lender's Commitment under that Facility minus (subject to clause 9.8 (Affiliates of Lenders as Ancillary Lenders) and as set out below):

- (a) the Base Currency Amount of its participation in any outstanding Utilisations under that Facility and, in the case of a Revolving Facility only, the Base Currency Amount of the aggregate of its Ancillary Commitments; and
- (b) in relation to any proposed Utilisation, the Base Currency Amount of its participation in any other Utilisations that are due to be made under that Facility on or before the proposed Utilisation Date and, in the case of a Revolving Facility only, the Base Currency Amount of its Ancillary Commitment in relation to any new Ancillary Facility that is due to be made available on or before the proposed Utilisation Date.

For the purposes of calculating a Lender's Available Commitment in relation to any proposed Utilisation under a Revolving Facility only, the following amounts shall not be deducted from a Lender's Commitment under that Facility:

- (i) that Lender's participation in any Revolving Facility Utilisations under that Facility that are due to be repaid or prepaid on or before the proposed Utilisation Date; and
- (ii) that Lender's (or its Affiliate's) Ancillary Commitments relating to that Facility to the extent that they are due to be reduced or cancelled on or before the proposed Utilisation Date;

"Available Facility" means, in relation to a Facility, the aggregate for the time being of each Lender's Available Commitment in respect of that Facility;

"Available Tenor" means, as of any date of determination and with respect to the then-current Benchmark, as applicable, any tenor for such Benchmark (or component thereof) or payment period for interest calculated with reference to such Benchmark (or component thereof), as applicable, that is or may be used for determining the length of an Interest Period for any term rate or otherwise, for determining any frequency of making payments of interest calculated pursuant to this agreement as of such date and not including, for the

avoidance of doubt, any tenor for such Benchmark that is then-removed from the definition of "Interest Period", Clause 14 (Interest) and/or Clause 15 (Interest Periods) pursuant to clause (d) of Clause 16.3 (Benchmark Replacement).

"Base Currency" means Dollars;

"Base Currency Amount" means:

- (a) in relation to a Utilisation, the amount specified in the Utilisation Request delivered by a Borrower for that Utilisation (or, if the amount requested is not denominated in the Base Currency, that amount converted into the Base Currency at the Facility Agent's Spot Rate of Exchange on the date which is three Business Days before the Utilisation Date or, if later, on the date the Facility Agent receives the Utilisation Request in accordance with the terms of this agreement) and, in the case of a Letter of Credit, as adjusted under clause 6.7 (Revaluation of Letters of Credit) at six-monthly intervals; and
- (b) in relation to an Ancillary Commitment, the amount specified as such in the notice delivered to the Facility Agent by Formula One Management Limited pursuant to clause 9.2 (Availability) (or, if the amount specified is not denominated in the Base Currency, that amount converted into the Base Currency at the Facility Agent's Spot Rate of Exchange on the date which is three Business Days before the Ancillary Commencement Date for that Ancillary Facility or, if later, the date the Facility Agent receives the notice of the Ancillary Commitment in accordance with the terms of this agreement).

as adjusted to reflect any repayment, prepayment, consolidation or division of a Utilisation, or (as the case may be) cancellation or reduction of an Ancillary Facility;

"Benchmark" means, initially, with respect to any (i) RFR Loan, the Daily Simple SOFR, (ii) with respect to any Term Benchmark Loan denominated in Dollars, the Term SOFR Rate, and (iii) with respect to any Term Benchmark Loan denominated in euros, EURIBOR; provided that if a Benchmark Transition Event, and the related Benchmark Replacement Date have occurred with respect to the then-current Benchmark for such currency, then "Benchmark" means the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate pursuant to sub-clause (a) of Clause 16.3 (Benchmark Replacement).

"Benchmark Replacement" means, for any Available Tenor, the first alternative set forth in the order below that can be determined by the Facility Agent for the applicable Benchmark Replacement Date, provided that, in the case of any Loan denominated in Euros, "Benchmark Replacement" shall mean the alternative set forth in (2) below:

- (1) the Daily Simple SOFR;
- (2) the sum of: (a) the alternate benchmark rate that has been selected by the Facility Agent and the Obligors' Agent as the replacement for the then-current Benchmark for the applicable corresponding tenor giving due consideration to (i) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (ii) any evolving or then-prevailing market convention for determining a benchmark rate as a replacement for the then-current Benchmark for syndicated credit facilities denominated in the relevant currency at such time in the United States and (b) the related Benchmark Replacement Adjustment;

If the Benchmark Replacement as determined pursuant to clause (1) or (2) above would be less than the applicable Floor, the Benchmark Replacement will be deemed to be the applicable Floor for the purposes of this agreement and the other Finance Documents.

"Benchmark Replacement Adjustment" means, with respect to any replacement of the then-current Benchmark with an Unadjusted Benchmark Replacement for any applicable Interest Period and Available Tenor for any setting of such Unadjusted Benchmark Replacement, the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by the Facility Agent and the relevant Borrower or the Obligors' Agent for the applicable corresponding tenor giving due consideration to (i) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body on the applicable Benchmark Replacement Date and/or (ii) any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement for syndicated credit facilities denominated in the applicable currency at such time.

"Benchmark Replacement Conforming Changes" means, with respect to any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Alternate Base Rate", the definition of "Business Day," the definition of "U.S. Government Securities Business Day," the definition of "Interest Period", Clause 14 (Interest) and Clause 15 (Interest Periods), timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, length of lookback periods, the applicability of breakage provisions, and other technical, administrative or operational matters) that the Facility Agent and the Obligors' Agent decide may be appropriate to reflect the adoption and implementation of such Benchmark and to permit the administration thereof by the Facility Agent in a manner substantially consistent with market practice (or, if the Facility Agent decides that adoption of any portion of such market practice is not administratively feasible or if the Facility Agent determines that no market practice for the administration of such Benchmark exists, in such other manner of administration as the Facility Agent and the Obligors' Agent decide are reasonably necessary in connection with the administration of this agreement and the other Finance Documents).

"Benchmark Replacement Date" means, with respect to any Benchmark, the earliest to occur of the following events with respect to such then-current Benchmark:

- (1) in the case of clause (1) or (2) of the definition of "Benchmark Transition Event," the later of (a) the date of the public statement or publication of information referenced therein and (b) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide all Available Tenors of such Benchmark (or such component thereof); or
- (2) in the case of clause (3) of the definition of "Benchmark Transition Event," the first date on which such Benchmark (or the published component used in the calculation thereof) has been determined and announced by the regulatory supervisor for the administrator of such Benchmark (or such component thereof) to be no longer representative; provided, that such non-representativeness will be determined by reference to the most recent statement or publication referenced in such clause (3) and even if any Available Tenor of such Benchmark (or such component thereof) continues to be provided on such date.

For the avoidance of doubt, (i) if the event giving rise to the Benchmark Replacement Date occurs on the same day as, but earlier than, the Reference Time in respect of any determination, the Benchmark Replacement Date will be deemed to have occurred prior to the Reference Time for such determination and (ii) the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (1) or (2) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

- "Benchmark Transition Event" means, with respect to any Benchmark, the occurrence of one or more of the following events with respect to such then-current Benchmark:
- (1) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely, provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof);
- (2) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof), the Federal Reserve Board, the NYFRB, the CME Term SOFR Administrator, the central bank for the relevant currency applicable to such Benchmark, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component), a resolution authority with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark (or such component), in each case, which states that the administrator of such Benchmark (or such component) has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof) permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof); or
- (3) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that all Available Tenors of such Benchmark (or such component thereof) are no longer, or as of a specified future date will no longer be, representative.

For the avoidance of doubt, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each thencurrent Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Unavailability Period" means, with respect to any Benchmark, the period (if any) (x) beginning at the time that a Benchmark Replacement Date pursuant to clauses (1) or (2) of that definition has occurred if, at such time, no Benchmark Replacement has replaced such then-current Benchmark for all purposes hereunder and under any Finance Document in accordance with Clauses 16.2 (Alternate Rate of Interest) and 16.3 (Benchmark Replacement) and (y) ending at the time that a Benchmark Replacement has replaced such then-current Benchmark for all purposes hereunder and under any Finance Document in accordance with Clauses 16.2 (Alternative Rate of Interest) and 16.3 (Benchmark Replacement).

"Bona Fide Debt Fund" means any Person (other than a Person who is separately identified by the Obligors' Agent as a Disqualified Lender pursuant to clause (x) of the definition of Disqualified Lender) that is primarily engaged in, or that advises, any bona fide debt fund or fixed income investor that is engaged in making, purchasing, holding or otherwise investing in commercial loans, bonds and similar extensions of credit or debt securities in the ordinary course of business and for which no personnel of a Disqualified Lender involved with such Person (A) makes investment decisions or (B) has access to non-public information relating to the Group or any person that forms part of the Group's business (including its Subsidiaries).

"Borrower" means an Original Borrower or an Additional Borrower unless it has ceased to be a Borrower in accordance with clause 30 (Changes to the Obligors) and, in respect of an Ancillary Facility only, any Group Company that becomes a borrower of that Ancillary Facility

with the approval of the relevant Lender pursuant to the provisions of clause 9.9 (Affiliates of Borrowers);

"Borrower Materials" has the meaning given to that term in clause 29.7 (Disclosure of Information).

"Break Costs" means (other than with respect to RFR Loans) the amount (if any) by which:

(a) the interest (excluding the Margin) which a Lender should have received for the period from the date of receipt of all or any part of its participation in a Loan or Unpaid Sum to the last day of the current Interest Period in respect of that Loan or Unpaid Sum, had the principal amount or Unpaid Sum received been paid on the last day of that Interest Period;

exceeds:

(b) the amount which that Lender would be able to obtain by placing an amount equal to the principal amount or Unpaid Sum received by it on deposit with a leading bank in the Relevant Interbank Market for a period starting on the Business Day following receipt or recovery and ending on the last day of the current Interest Period;

"Business" means the development, exploitation and administration of the F1 world championship in a commercial manner;

"Business Day" means a day (other than a Saturday or Sunday) on which banks are open for general business in London and New York; provided that:

- (a) in relation to RFR Loans, any such day is also an U.S. Government Securities Business Day; or
- (b) in relation to any date for payment or purchase of euro or calculation of EURIBOR, any such day is also a TARGET Day;

"Capital Expenditure" has the meaning given to that term in clause 26.1 (Financial Definitions);

"Cash Equivalent Investments" means (a) marketable debt securities (including commercial paper) with a maturity of six months or less and with a short term debt rating of at least A1+ granted by Standard & Poor's, a division of the McGraw-Hill Companies, Inc. or P-1 granted by Moody's Investors Services, Inc. at the date of acquisition or when received; (b) marketable direct obligations issued by, or unconditionally guaranteed by, the United States, the United Kingdom or issued by any agency thereof and backed by the full faith and credit of the United States or the United Kingdom, in each case maturing within one year from the date of acquisition or when received; (c) certificates of deposit, time deposits, eurodollar time deposits or overnight bank deposits having maturities of six months or less from the date of acquisition or receipt issued by any Lender or by any commercial bank organized under the laws of the United States or any state thereof or the United Kingdom; (d) repurchase obligations of any Lender or of any commercial bank satisfying the requirements of clause (c) of this definition, having a term of not more than 30 days, with respect to securities issued or fully guaranteed or insured by the United States government or the United Kingdom government; (e) securities with maturities of one year or less from the date of acquisition or receipt issued or fully guaranteed by any state, commonwealth or territory of the United States or the United Kingdom, by any political subdivision or taxing authority of any such state, commonwealth or territory or by any foreign government, the securities of which state, commonwealth, territory, political subdivision, taxing authority or foreign government (as the case may be) are rated at least A by Standard & Poor's or A by Moody's at the date of acquisition or receipt; (f) securities with maturities of six months or less from the date of acquisition or receipt backed by standby letters of credit issued by any Lender or any commercial bank satisfying the

requirements of clause (c) of this definition; (g) money market mutual or similar funds at least 95% of the assets of which constitute Cash Equivalent Investments of the kinds described in clauses (a) through (f) of this definition; (h) money market funds that (i) are rated AAA by Standard & Poor's or Aaa by Moody's at the date of acquisition or receipt and (ii) have portfolio assets of at least \$500,000,000; and (i) in the case of any Subsidiary that is not organized in the United States or the United Kingdom, (x) investments substantially comparable to any of the foregoing investments with respect to the country in which such Subsidiary is organized and (y) other short term investments utilized by such Subsidiaries in accordance with normal investment practices for cash management in such jurisdiction;

"CBR Loan" means a Loan that bears interest at a rate determined by reference to the Central Bank Rate.

"Central Bank Rate" means, for any Loan denominated in euro, (A) the greater of (i) one of the following three rates as may be selected by the Facility Agent in its reasonable discretion: (1) the fixed rate for the main refinancing operations of the European Central Bank (or any successor thereto), or, if that rate is not published, the minimum bid rate for the main refinancing operations of the European Central Bank (or any successor thereto), each as published by the European Central Bank (or any successor thereto) from time to time, (2) the rate for the marginal lending facility of the European Central Bank (or any successor thereto), as published by the European Central Bank (or any successor thereto) from time to time or (3) the rate for the deposit facility of the central banking system of the Participating Member States, as published by the European Central Bank (or any successor thereto) from time to time, and (ii) the applicable Floor; plus (B) the applicable Central Bank Rate Adjustment.

"Central Bank Rate Adjustment" means, for any day, for any Loan denominated in euro a rate equal to the difference (which may be a positive or negative value or zero) of (i) the average of the EURIBOR for the five most recent Business Days preceding such day for which the applicable Screen Rate was available (excluding, from such averaging, the highest and the lowest EURIBOR applicable during such period of five Business Days) minus (ii) the Central Bank Rate in respect of Euro in effect on the last Business Day in such period.

"Certified Synergies" means synergies which any financial officer of the Group determines in good faith as being factually supportable and reasonably achievable by the Group (including by any joint venture of the Group (whether accounted for on the financial statements of any such joint venture or the Group in an amount proportionate to the percentage of equity ownership of such joint venture that are beneficially owned by the Group) during the 24 month period following the date of any acquisition, investment, disposition, operating improvement, restructuring, reorganisation, cost savings initiative, relocation or similar transaction or initiative as a result of or in connection with such event;

"Change of Control" means, (i) at any time, a person or group of persons under common control (other than any Permitted Holder) gains the right to exercise more than 50 per cent of the voting rights exercisable in a general meeting of the Parent or (ii) at any time, the entire share capital of the Parent ceases to be owned directly by one or more persons who have entered into Transaction Security Documents in respect of the share capital of the Parent they own;

For these purposes and for purposes of this agreement,

- (i) persons are under common control if one person controls both of them or if one controls the other, where "control" has the meaning given to such term under section 450 of the Corporation Tax Act 2010;
- (ii) "LMC" means Liberty Media Corporation and any successor (by merger, consolidation, transfer or otherwise) to all or substantially all of its assets; and any subsequent successor (by merger, consolidation, transfer or

otherwise) to all or substantially all of a successor's assets, provided, that if a Transferee Parent becomes the beneficial owner of all or substantially all of the equity securities of the Parent then beneficially owned by LMC as to which LMC has dispositive power, the term "LMC" shall also mean such Transferee Parent and any successor (by merger, consolidation, transfer or otherwise) to all or substantially all of its assets. "Transferee Parent" for this purpose means, in the event of any transaction or series of related transactions involving the direct or indirect transfer (or relinquishment of control) by LMC of a person or persons (a "Transferred Person") that hold equity securities of the Parent beneficially owned by LMC, such Transferred Person or its successor in such transaction or any ultimate parent entity (within the meaning of the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended) of such Transferred Person or its successor if immediately after giving effect to such transaction or the last transaction in such series, voting securities representing at least a majority of the voting power of the outstanding voting securities of such Transferred Person, successor or ultimate parent entity are beneficially owned by any combination of LMC, persons who prior to such transaction were beneficial owners of a majority of, or a majority of the voting power of, the outstanding voting securities of LMC (or of any publicly traded class or series of voting securities of LMC designed to track the economic performance of a specified group of assets or businesses) or persons who are Control Persons as of the date of such transaction or the last transaction in such series. "Control Person" for this purpose means each of (a) the Chairman of the Board of LMC, (b) the President, Chief Executive Officer, Chief Financial Officer, Chief Legal Officer and Chief Corporate Development Officer of LMC, (c) any Senior Vice President of LMC, (d) each of the directors of LMC and (e) the respective Affiliated Persons of the persons referred to in clauses (a) through (d);

(iii) "Permitted Holder" means any one or more of:

- (A) LMC;
- (B) John C. Malone or Gregory B. Maffei (whether such persons are acting individually or in concert);
- (C) each of the respective Affiliated Persons of the persons referred to in clause (B);
- (D) any publicly traded person in which any of the persons referred to in clauses (B) and (C) (whether individually or together with the other persons in clauses (B) and (C)) is the largest beneficial owner of (x) the equity interests of such person or (y) the aggregate voting power of all the outstanding classes or series of the equity interests of such person then outstanding and normally entitled (without regard to the occurrence of any contingency) to vote in the election of directors;
- (E) any person a majority of the aggregate voting power of all the outstanding classes or series of the equity interests of which are directly or indirectly beneficially owned by any one or more of the persons referred to in clauses (A), (B), (C) or (D); and
- (F) any group consisting solely of persons described in clauses (A) through (E).

For purposes of the definition of "Permitted Holders", "person" and "group" have the meanings given to them for purposes of Sections 13(d) and 14(d) of the Exchange Act or any successor provisions, and the term "group" includes

any group acting for the purpose of acquiring, holding or disposing of securities within the meaning of Rule 13d-5(b)(1) under the Exchange Act, or any successor provision.

"Charged Property" means all of the assets of the Obligors which from time to time are, or are expressed to be, the subject of the Transaction Security;

"Class" when used in reference to any Loan or borrowing, refers to whether such Loan, or the Loans comprising such Utilisation, is a Revolving Facility Loan or a Term Loan;

"CME Term SOFR Administrator" means CME Group Benchmark Administration Limited as administrator of the forward-looking term Secured Overnight Financing Rate (SOFR) (or a successor administrator).

"Commitment" means a Term Facility A Commitment, a Term Facility B Commitment, a Revolving Facility Commitment or an Incremental Facility Commitment;

"Compliance Certificate" means a certificate substantially in the form set out in schedule 7 (Form of Compliance Certificate) in form and substance satisfactory to the Facility Agent (acting reasonably);

"Concorde Agreement" means the contractual arrangements with any member or members of the Group and F1 Teams taken as a whole dealing with (amongst other things) payments to F1 Teams for competing in the F1 Championships, as in effect from time to time;

"Consolidated EBITDA" has the meaning given to that term in clause 26.1 (Financial Definitions);

"Consolidated Interest Coverage Ratio" means on any date the ratio of (i) Consolidated EBITDA to (ii) Consolidated Interest Expense, in each case, for the most recently ended Testing Period on or prior to such date, all determined on a consolidated basis;

"Consolidated Total Net Debt" means the principal amount of all Financial Indebtedness of the Group less the aggregate amount at that time of Cash and Cash Equivalent Investments held by Group, less the amount of any outstandings under any Ancillary Facilities (other than any overdraft or working capital facility or any letter of credit or other guarantee facility covering monies borrowed by way of term debt, any overdraft, working capital facility or other monies borrowed (in each case to the extent of such borrowings)) and after making such adjustments any financial officer of the Group determines in good faith to be reasonable to reflect material timing differences in receipts of payments resulting from any change in race dates;

"Constitutional Documents" means the constitutional documents of Delta Topco;

"Control Person" has the meaning given to that term in the definition of LMC.

"CTA" means the Corporation Tax Act 2009.

"Daily Simple SOFR" means, for any day (a "SOFR Rate Day"), a rate per annum equal to SOFR for the day (such day "SOFR Determination Date") that is five (5) U.S. Government Securities Business Days prior to (i) if such SOFR Rate Day is a U.S. Government Securities Business Day, such SOFR Rate Day or (ii) if such SOFR Rate Day is not a U.S. Government Securities Business Day, the U.S. Government Securities Business Day immediately preceding such SOFR Rate Day, in each case, as such SOFR is published by the SOFR Administrator on the SOFR Administrator's Website; provided that if the Daily Simple SOFR as so determined would be less than the applicable Floor, such rate shall be deemed to be equal to the applicable Floor for the purposes of this agreement. Any change

in Daily Simple SOFR due to a change in SOFR shall be effective from and including the effective date of such change in SOFR without notice to the Borrower.

"Declined Amounts" has the meaning given to it in clause 12.2 (Disposal and Insurance Proceeds and Excess Cashflow);

"Declining Lender" has the meaning given to it in clause 12.2 (Disposal and Insurance Proceeds and Excess Cashflow);

"Deemed Date" has the meaning given to that term in clause 1.4 (Certain Calculations).

"Default" means an Event of Default or any event or circumstance specified in clause 28 (Events of Default) which would (with the expiry of a grace period, the giving of notice, the making of any determination under the Finance Documents or any combination of any of the foregoing) be an Event of Default;

"Defaulting Lender" means any Lender:

- (a) which has failed to make its participation in a Loan available or has notified the Facility Agent or the Parent that it will not make its participation in a Loan available by the Utilisation Date of that Loan in accordance with clause 5.4 (Lenders' Participation) or has failed to provide cash collateral (or has notified the Issuing Bank that it will not provide cash collateral) in accordance with clause 7.4 (Cash Collateral by Non-Acceptable L/C Lender);
- (b) which has otherwise rescinded or repudiated a Finance Document; or
- (c) with respect to which an Insolvency Event has occurred and is continuing,

unless, in the case of paragraph (a) above:

- (i) payment is made within three Business Days of its due date;
- (ii) its failure to pay is caused by:
 - (A) administrative or technical error; or
 - (B) a Disruption Event, and

payment is made within ten Business Days of its due date; or

(iii) the Lender is disputing in good faith whether it is contractually obliged to make the payment in question;

"Delegate" means any delegate, agent, attorney or co-trustee appointed by the Security Agent;

"Delta 2" means Delta 2 (Lux) Sàrl, a private limited liability company (société à responsabilité limitée) incorporated under the laws of the Grand Duchy of Luxembourg and with registered office at 9, rue de Bitbourg, L-1273, Luxembourg, the Grand Duchy of Luxembourg and registered with the Luxembourg trade and companies register (Registre de Commerce et des Sociétés de Luxembourg) with number B122129;

"Delta 3" means Delta 3 (UK) Limited a company incorporated in England and Wales with registered number 5988544:

"Delta Topco" means Delta Topco Limited, a company incorporated in Jersey, with registration number 95136;

"Designated Gross Amount" has the meaning given to that term in clause 9.2 (Availability);

"Designated Net Amount" has the meaning given to that term in clause 9.2 (Availability);

"Designated Non-Cash Consideration" means the fair market value of non-cash consideration received by any Group Company in connection any sale, lease, licence, transfer or other disposal made pursuant to clause (n) of the definition of Permitted Disposal that is designated as Designated Non-Cash Consideration by a financial officer of a Group Company less the amount of cash or Cash Equivalent Investments received in connection with a subsequent sale of or conversion of or collection on such Designated Non-Cash Consideration;

"Disposals Proceeds" has the meaning given to that term in clause 12.2 (Disposal and Insurance Proceeds and Excess Cashflow);

"Disqualified Equity" means, as of any date, any equity interest of any person that, by its terms (or by the terms of any security into which it is convertible or for which it is exchangeable at the option of the holder thereof), or upon the happening of any event, matures or is mandatorily redeemable, pursuant to a sinking fund obligation or otherwise, or is redeemable at the option of the holder thereof, in whole or in part, or requires or mandates cash payments or distributions, on or prior to the date that is 90 days after the latest Termination Date that existed at the time such equity interest was issued or incurred; provided, however, that any class of equity interests of such person that, by its terms, authorizes such person to satisfy in full its obligations with respect to the payment of dividends or upon maturity, redemption (pursuant to a sinking fund or otherwise) or repurchase thereof or otherwise by the delivery of equity interests that are not Disqualified Equity, and that is not convertible, puttable or exchangeable for Disqualified Equity or Financial Indebtedness, will not be deemed to be Disqualified Equity so long as such person satisfies its obligations with respect thereto by the delivery of equity interests that are not Disqualified Equity; provided, further, however, that an equity interest that would constitute Disqualified Equity solely because the holders thereof have the right to require such person to repurchase or redeem such equity interests upon the occurrence of (x) a disposal or change of control or (y) one or more other events shall not constitute Disqualified Equity if such person may repurchase or redeem any such equity interest in a manner that complies with clause 27.15 (Dividends and Share Redemptions). Notwithstanding the foregoing, any equity interests issued to any employee or to any plan for the benefit of employees of any Group Company or by any such plan to such employees shall not constitute Disqualified Equity solely because they may be required to be repurchased by a Group Company in order to satisfy applicable statutory or regulatory obligations or as a result of such employee's termination, death or disability.

"Disqualified Lender" shall mean, collectively, (x) those banks, financial institutions, other institutional lenders and investors and other persons designated by Obligors' Agent as a "Disqualified Lender" by written notice delivered to the Facility Agent on or prior to the 2022 Effective Date (and affiliates of such persons (A) identified in writing by the Obligors' Agent from time to time to the Facility Agent and/or (B) clearly identifiable as affiliates of such persons on the basis of such affiliates' names), (y) such other persons (and clearly identifiable affiliates of such persons (other than affiliates that are Bona Fide Debt Funds and that would not be a competitor but for being an affiliate of a competitor pursuant to this clause (y))) identified in writing by the Obligors' Agent to the Facility Agent from time to time to the extent such person is or becomes a competitor of the Group and (z) such other persons identified in writing by the Obligors' Agent to the Facility Agent from time to time at any time prior to the occurrence and continuation of an Event of Default under clause 28.1 (Non-Payment), clause 28.6 (Insolvency), clause 28.7 (Insolvency Proceedings) (other than clause 28.7(ii), clause 28.7(ii) and clause 28.7(ii)), clause 28.7(iii) and clause 28.7(iii) and clause 28.7(iii) and clause 28.7(iii) and clause 28.7(iii), subject to the consent of the Facility Agent (which consent shall not

be unreasonably withheld, delayed or conditioned); <u>provided</u> that (x) the foregoing shall not apply retroactively to disqualify any assignment, sub-participation or sub-contract to the extent such assignment, sub-participation or sub-contract was acquired by a party that was not a Disqualified Lender at the time of such assignment, sub-participation or sub-contract and (y) the execution by Obligors' Agent of an assignment with respect to a Disqualified Lender will not by itself result in such assignee no longer being considered a Disqualified Lender.

"Disruption Event" means either or both of:

- (a) a material disruption to those payment or communications systems or to those financial markets which are, in each case, required to operate in order for payments to be made in connection with the Facilities (or otherwise in order for the transactions contemplated by the Finance Documents to be carried out) which disruption is not caused by, and is beyond the control of, any of the Parties; or
- (b) the occurrence of any other event which results in a disruption (of a technical or systems-related nature) to the treasury or payments operations of a Party preventing that, or any other Party;
 - (i) from performing its payment obligations under the Finance Documents; or
 - (ii) from communicating with other Parties in accordance with the terms of the Finance Documents,

and which (in either such case) is not caused by, and is beyond the control of, the Party whose operations are disrupted;

"Dollars", "USD" or "\$" means the lawful currency of the United States of America;

"Dormant Subsidiary" means a Group Company which does not trade (for itself or as agent for any person) and does not own, legally or beneficially, assets (including without limitation Financial Indebtedness owed to it) which in aggregate have a value of \$50,000 or more or its equivalent in other currencies;

"Dorna Acquistion" has the meaning given to that term in the definition of Dorna Distribution.

"Dorna Distribution" means a dividend or distribution by any Group Company to any LMC Investor to help fund the acquisition (including the payment of fees and expenses related thereto) of approximately 86% of the equity interests of Dorna Sports, S.L., pursuant to a Share Purchase Agreement dated as of March 29, 2024 entered into by and among Dorna Sports, S.L., Libertad Especia, S.L.U., Liberty Media Corporation and the sellers thereto ("Dorna Acquisition"), and to consummate the transactions described therein.

"EBITDA Builder Prong" has the meaning given to that term in the definition of Available Amount Basket.

"Economic Sanctions Law" means any economic or financial sanctions administered by OFAC, the US State Department, the United Kingdom, the United Nations, the European Union or any member state thereof or Canada;

"Environmental Claim" means any claim, proceeding, formal notice or investigation by any person in respect of any Environmental Law;

"Environmental Law" means any applicable law or regulation which relates to:

(a) the pollution or protection of the environment;

- (b) harm to or the protection of human health;
- (c) the conditions of the workplace; or
- (d) any emission or substance capable of causing harm to any living organism or the environment;

"Environmental Permits" means any permit and other Authorisation and the filing of any notification, report or assessment required under any Environmental Law for the operation of the business of any Group Company conducted on or from the properties owned or used by any Group Company;

"Equity Documents" means the Constitutional Documents and any other document designated as an equity document by the Facility Agent and the Obligors' Agent;

"EURIBOR" means, in relation to any Loan in euro:

- (a) the applicable Screen Rate;
- (b) (if no Screen Rate is available for the Interest Period of that Loan) the Interpolated Screen Rate for that Loan; or
- (c) if no Screen Rate is available for the Interest Period of that Loan or no Screen Rate is available for the Interest Period of that Loan and it is not possible to calculate an Interpolated Screen Rate for that Loan, the rate that may be specified by the Facility Agent after consultation with the Obligor's Agent in another page or service displaying the relevant rate

as of (in the case of paragraphs (a) and (c) above) the Specified Time on the Quotation Day for euro for a period equal in length to the Interest Period for that Loan; provided that if the EURIBOR as so determined would be less than the applicable Floor, such rate shall be deemed to be equal to the applicable Floor for the purposes of this agreement;

"Euro", "EUR" or "€" means the single currency unit of the Participating Member States;

"Event of Default" means any event or circumstance specified as such in clause 28 (Events of Default);

"Excess Cashflow" has the meaning given to that term in clause 26.1 (Financial Definitions);

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Executive Order" means the executive Order No. 13224 of September 23, 2001 - Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten To Commit, or Support Terrorism;

"Existing Ancillary Facility" has the meaning given to it in the 2022 Amendment Agreement;

"Existing Ancillary Lender" has the meaning given to it in the 2022 Amendment Agreement;

"Expiry Date" means, for a Letter of Credit, the last day of its Term;

"Extension" has the meaning given to it in clause 2.6 (Extension Offers);

"Extension Amendment" has the meaning given to it in clause 2.6 (Extension Offers);

"Extended Revolving Commitment" has the meaning given to it in clause 2.6 (Extension Offers);

"Extended Revolving Loan" has the meaning given to it in clause 2.6 (Extension Offers);

"Extending Term Loan" has the meaning given to it in clause 2.6 (Extension Offers);

"Extending Lender" has the meaning given to it in clause 2.6 (Extension Offers);

"F1 Team" means a team which is competing or will compete in the Formula One Championship organised by the Group;

"Facility" means a Term Facility or a Revolving Facility;

"Facility Agent's Spot Rate of Exchange" means the first available of:

- (a) the Facility Agent's spot rate of exchange; or
- (b) the Thomson Reuters Rate of Exchange; or
- (c) any other commercially available spot rate of exchange selected by the Facility Agent,

for the purchase of the relevant currency with the Base Currency in the London foreign exchange market at or about 11:00 a.m. on a particular day;

"Facility Office" means:

- (a) in respect of a Lender or the Issuing Bank, the office or offices notified by that Lender or the Issuing Bank to the Facility Agent in writing on or before the date it becomes a Lender or the Issuing Bank (or, following that date, by not less than five Business Days' written notice) as the office or offices through which it will perform its obligations under this agreement; or
- (b) in respect of any other Finance Party, the office in the jurisdiction in which it is resident for tax purposes;

"Facility RCF" means the revolving credit facility made available on the 2022 Effective Date pursuant to clause 2.1(a)(iii) (The Facilities);

"FATCA" means:

- (a) sections 1471 to 1474 of the US Internal Revenue Code of 1986, as amended (the "IR Code") or any associated regulations; or
- (b) any law or regulation or administrative guidance enacted in any jurisdiction in connection with a bilateral or multilateral agreement between the government of the United States and such other jurisdiction relating to sections 1471 to 1474 of the IR Code;

"FATCA Agreement" has the meaning given to it in clause 18.1 (Definitions);

"FATCA Application Date" means:

(a) in relation to a 'withholdable payment' described in section 1473(1)(A)(i) of the IR Code (which includes payments of interest and certain other payments from sources within the US), 1 July 2014; or in relation to a 'passthru payment' described in section 1471(d)(7) of the IR Code not falling within paragraph (a) above, the first date from which such payment may become subject to a deduction or withholding required by FATCA;

"FATCA Compliant Party" means a Party (including, for this purpose, any Finance Party) payments to whom do not require a FATCA Deduction whether as a result of, without limitation, the status or jurisdiction of that Party or any other matter;

"FATCA Deduction" means a deduction or withholding from (or other payment to a governmental or taxation authority related to) a payment under a Finance Document pursuant to FATCA or a FATCA Agreement;

"Federal Funds Effective Rate" means, for any day, the rate calculated by the NYFRB based on such day's federal funds transactions by depositary institutions, as determined in such manner as shall be set forth on the NYFRB's Website from time to time, and published on the next succeeding Business Day by the NYFRB as the effective federal funds rate; provided that if the Federal Funds Effective Rate as so determined would be less than zero, such rate shall be deemed to be zero for the purposes of this agreement.

"Federal Reserve Board" means the Board of Governors of the Federal Reserve System of the United States of America.

"Fee Letter" means:

- (a) any Incremental Facility Commitment Fee Letter; and
- (b) any other agreement setting out fees payable to a Finance Party referred to in clause 17.4 (Fees Payable in Respect of Letters of Credit) or clause 17.5 (Interest, Commission and Fees on Ancillary Facilities) of this agreement or under any other Finance Document;

"FIA" means Federation Internationale de L'Automobile;

"FIAF" means Federation Internationale de L'Automobile France:

"FIAS" means Federation Internationale de L'Automobile Switzerland;

"Finance Document" means this agreement, the 2022 Amendment Agreement, the 2022 Amendment Request, the 2024 Amendment Agreement, the 2024 Amendment Request, any Incremental Facility Documents, any Accession Letter, any Ancillary Document, any Compliance Certificate, any Fee Letter, the Intercreditor Agreement, any Resignation Letter, any Selection Notice, any Transaction Security Document, any Utilisation Request and any other document designated as a "Finance Document" by Obligors' Agent and the Facility Agent;

"Finance Party" means the Facility Agent, an Arranger, the Security Agent, a Lender, the Issuing Bank and any Ancillary Lender;

"Financial Indebtedness" means (without double counting) any indebtedness in relation to or arising under or in connection with:

- (a) any money borrowed (including any overdraft);
- (b) any debenture, bond (other than a performance bond issued in the ordinary course by a Group Company), note or other similar instrument;
- (c) any amount raised by acceptance under any acceptance credit facility (or dematerialised equivalent);

- (d) any receivable sold or discounted (otherwise than on a non-recourse basis);
- (e) the purchase price of any asset or service to the extent payable by a Group Company after the time of sale or delivery to a Group Company, where the deferred payment is:
 - (i) arranged primarily as a method of raising finance; or
 - (ii) paid more than six months after the sale or delivery date;
- (f) the sale price of any asset or service to the extent paid to a Group Company before the time of sale or delivery by the Group Company liable to effect that sale or delivery, where the advance payment is arranged primarily as a method of raising finance;
- (g) any lease or hire purchase contract which would, in accordance with the Accounting Principles as applied as at the Sixth Restatement Date, be treated as a finance or capital lease;
- (h) for the purpose of clause 28.5 (Cross Default) only, Treasury Transactions;
- (i) any indemnity or counter-indemnity obligation in respect of a guarantee, bond, standby or documentary letter of credit or any other instrument issued by a bank or financial institution in respect of financial indebtedness of a member of the Group;
- (j) Disqualified Equity;
- (k) [reserved]; or
- (I) any guarantee of indebtedness of any person of a type referred to in paragraphs (a) to (k) (inclusive) above; provided that the amount of such indebtedness shall not exceed the amount that would be payable if such guarantee of indebtedness were to be enforced by a creditor;

and shall not include:

- (a) any amounts payable under any deferred compensation plans of any person relating to its, its parent entities or their subsidiaries' directors, management, employees or consultants;
- (b) any indebtedness owed between or among the Group;
- (c) any payment adjustments to which the seller may become entitled to the extent such payment is determined by a final closing balance sheet or such payment depends on the performance of the person or assets acquired; provided, however, that to the extent such payment becomes fixed and determined, the amount is paid within ninety (90) days thereafter;
- (d) any indebtedness that has been defeased pursuant to the deposit of cash or Cash Equivalent Investments (in any amount sufficient (in the good faith determination of Obligors' Agent) to satisfy all such indebtedness obligations at maturity or redemption, as applicable, and all payments of interest and premium, if any) in a trust or account created or pledged for the sole benefit of the holders of such indebtedness pursuant to the applicable terms of the instrument governing such indebtedness;
- (e) prepaid or deferred revenue;
- (f) prepayments or deposits received from clients or customers in the ordinary course of business; and

(g) all obligations, contingent or otherwise, of such person in respect of letters of guaranty provided in the ordinary course of business.

The amount of Financial Indebtedness of any person at any date shall be the outstanding balance at such date of all unconditional obligations (i.e., shall not take in account the fair value increases of such Financial Indebtedness).

"Financial Quarter" has the meaning given to that term in clause 26.1 (Financial Definitions);

"Financial Year" has the meaning given to that term in clause 26.1 (Financial Definitions);

"First Lien Finance Document" has the meaning given to it in the Intercreditor Agreement;

"First Lien Liabilities" has the meaning given to the term "First Lien Liabilities" in the Intercreditor Agreement;

"Fixed Amounts" has the meaning given to that term in clause 1.4 (Certain Calculations);

"Fitch" means Fitch, Inc.

"**Floor**" means the benchmark rate floor, if any, provided in this agreement (as of the 2022 Effective Date, the modification, amendment or renewal of this agreement or otherwise) with respect to the Term SOFR Rate, EURIBOR, the Daily Simple SOFR or the Central Bank Rate as applicable. For the avoidance of doubt the Floor for (x) each of Term SOFR Rate, Daily Simple SOFR, EURIBOR or the Central Bank Rate for the Facility RCF and Term Facility A shall be 0% and (y) each of Term SOFR Rate, Daily Simple SOFR or the Central Bank Rate for the Term Facility B shall be 0.50%.

"FOA" means Formula One Administration Limited, a company incorporated under the laws of England and Wales with registered number 3737094;

"FOAM" means Formula One Asset Management Limited, a company incorporated in England and Wales with registered number 4169896;

"FOL" means Formula One Licensing B.V., a company incorporated under the laws of the Netherlands with registered number 24198235;

"FOM" means Formula One Management Limited, a company incorporated under the laws of England and Wales with registered number 4169896;

"FOWC" means Formula One World Championship Limited, a company incorporated under the laws of England and Wales with registered number 4174493;

"Further Lender" means an entity specified as an Incremental Facility Lender in an Incremental Facility Commitment Letter which is not already a Lender;

"Global Anti Base Erosion Rules" means the laws of a jurisdiction which have been implemented in accordance with the Global Anti Base Erosion Model Rules (Pillar Two) published by the OECD in their Report "Tax Challenges Arising from the Digitalisation of the Economy - Global Anti-Base Erosion Model Rules (Pillar Two)" on 20 December 2021;

"Group" means the Parent and each of its Subsidiaries for the time being, excluding any Unrestricted Subsidiary;

"Group Company" means a member of the Group;

"Group Relief" means losses or other amounts eligible for surrender under Part 5 or Part 5A of the UK's Corporation Tax Act 2010.

"Guarantor" means an Original Guarantor or an Additional Guarantor, unless it has ceased to be a Guarantor in accordance with clause 30 (Changes to the Obligors);

"Hedging Agreement" has the meaning given to it in the Intercreditor Agreement;

"Hedge Counterparty" has the meaning given to it in the Intercreditor Agreement;

"IFRS" means international accounting standards within the meaning of IAS Regulation 1606/2002 to the extent applicable to the relevant financial statements;

"Incremental Facility" means an Incremental Term Facility and/or Incremental Revolving Facility (as the context requires);

"Incremental Facility Commitment" means:

- (a) in relation to any entity identified as an Incremental Facility Lender in an Incremental Facility Commitment Letter:
 - (i) the aggregate amount of Incremental Term Facility Commitments and/or Incremental RCF Commitments in the relevant currency or currencies of any Incremental Facility Commitment made available by it pursuant to clause 2.2 (Incremental Facilities) as specified in the relevant Incremental Facility Commitment Letter(s); and
 - (ii) the amount in the relevant currency or currencies of any Incremental Facility Commitment transferred to it under this agreement; and
- (b) in relation to any other Lender, the amount in the relevant currency or currencies of any Incremental Facility Commitment transferred to it under this agreement,

in each case, to the extent not cancelled, reduced or transferred by it under this agreement (or redesignated as a Term Facility A Commitment or Term Facility B Commitment pursuant to the 2022 Amendment Agreement, the 2023 Term Facility B Refinancing Documents or the 2024 Amendment Request);

"Incremental Facility Commitment Fee Letter" means each fee letter in respect of an Incremental Facility Commitment entered into between a Borrower and the relevant Incremental Facility Lender setting out fees referred to in clause 17.7 (Incremental Facility Commitment Fees);

"Incremental Facility Commitment Letter" means the notice delivered to the Facility Agent pursuant to clause 2.2(k) (Incremental Facilities) of this agreement a template of which is set out in schedule 11 (Incremental Facility Commitment Letter);

"Incremental Facility Currency" means a currency which is readily available in the amount anticipated to be required and freely convertible into the Base Currency in the Relevant Interbank Market at the time of the delivery of the relevant Incremental Facility Commitment Letter and as approved by each of the relevant Incremental Facility Lenders and the Facility Agent in the relevant Incremental Facility Commitment Letter;

"Incremental Facility Documents" means any Incremental Facility Commitment Fee Letter, any Incremental Facility Commitment Letter and any document related to an Incremental Facility designated as an "Incremental Facility Document" by the relevant Incremental Facility Lender(s) (or the Facility Agent on their behalf) and Obligors' Agent;

"Incremental Facility Lender" means:

- (a) any person specified as an Incremental Facility Lender in an Incremental Facility Commitment Letter; and
- (b) any person which has become a Party as a Lender under an Incremental Facility in accordance with clause 29 (Changes to the Lenders),

which in each case has not ceased to be a Party in accordance with the terms of this agreement;

"Incremental Facility Loan" means a loan made or to be made under an Incremental Facility or the principal amount outstanding for the time being of that loan;

"Incremental Facility Repayment Date" means each date specified as a repayment date in respect of an Incremental Facility Commitment in the relevant Incremental Facility Commitment Letter;

"Incremental First Lien Ratio Builder" has the meaning given to it in clause 2.2 (Incremental Facilities);

"Incremental Fixed Amount" has the meaning given to it in clause 2.2 (Incremental Facilities);

"Incremental RCF Commitments" has the meaning given to it in clause 2.2 (Incremental Facilities);

"Incremental Revolving Facility" has the meaning given to it in clause 2.2 (Incremental Facilities);

"Incremental Revolving Facility Loan" means a loan made or to be made under an Incremental Revolving Facility or the principal amount outstanding for the time being of that loan;

"Incremental Term Facility" has the meaning given to it in clause 2.2 (Incremental Facilities);

"Incremental Term Facility Commitments" has the meaning given to it in clause 2.2 (Incremental Facilities);

"Incremental Term Facility B Maturity Exclusion" has the meaning given to it in clause 2.2 (Incremental Facilities);

"Incremental Term Facility Loan" means a loan made or to be made under an Incremental Term Facility or the principal amount outstanding for the time being of that loan;

"Incurrence-Based Amounts" has the meaning given to that term in clause 1.4 (Certain Calculations).

"Independent Assets or Operations" means, with respect to the Parent Reporter, that such Parent Reporter's total assets, revenues, income from continuing operations before income taxes and cash flows from operating activities (excluding in each case amounts related to its investment in or loans to the Parent and its Subsidiaries), determined in accordance with Accounting Principles and as shown on the most recent balance sheet of the Parent Reporter, is more than 5.0% of the Parent Reporter's corresponding consolidated amount.

"Insolvency Event" in relation to a Finance Party means that the Finance Party:

(a) is dissolved (other than pursuant to a consolidation, amalgamation or merger);

- becomes insolvent or is unable to pay its debts or fails or admits in writing its inability generally to pay its debts as they become due;
- (c) makes a general assignment, arrangement or composition with or for the benefit of its creditors;
- (d) institutes or has instituted against it, by a regulator, supervisor or any similar official with primary insolvency, rehabilitative or regulatory jurisdiction over it in the jurisdiction of its incorporation or organisation or the jurisdiction of its head or home office, a proceeding seeking a judgment of insolvency or bankruptcy or any other relief under any bankruptcy or insolvency law or other similar law affecting creditors' rights, or a petition is presented for its winding-up or liquidation by it or such regulator, supervisor or similar official;
- (e) has instituted against it a proceeding seeking a judgment of insolvency or bankruptcy or any other relief under any bankruptcy or insolvency law or other similar law affecting creditors' rights, or a petition is presented for its winding-up or liquidation, and, in the case of any such proceeding or petition instituted or presented against it, such proceeding or petition is instituted or presented by a person or entity not described in paragraph (d) above and:
 - (i) results in a judgment of insolvency or bankruptcy or the entry of an order for relief or the making of an order for its winding-up or liquidation; or
 - (ii) is not dismissed, discharged, stayed or restrained in each case within 30 days of the institution or presentation thereof;
- (f) has exercised in respect of it one or more of the stabilisation powers pursuant to Part 1 of the Banking Act 2009 and/or has instituted against it a bank insolvency proceeding pursuant to Part 2 of the Banking Act 2009 or a bank administration proceeding pursuant to Part 3 of the Banking Act 2009;
- (g) has a resolution passed for its winding-up, official management or liquidation (other than pursuant to a consolidation, amalgamation or merger);
- (h) seeks or becomes subject to the appointment of an administrator, provisional liquidator, conservator, receiver, trustee, custodian or other similar official for it or for all or substantially all its assets;
- (i) has a secured party take possession of all or substantially all its assets or has a distress, execution, attachment, sequestration or other legal process levied, enforced or sued on or against all or substantially all its assets and such secured party maintains possession, or any such process is not dismissed, discharged, stayed or restrained, in each case within 30 days thereafter;
- (j) causes or is subject to any event with respect to it which, under the applicable laws of any jurisdiction, has an analogous effect to any of the events specified in paragraphs (a) to (i) (inclusive) above; or
- (k) takes any action in furtherance of, or indicating its consent to, approval of, or acquiescence in, any of the foregoing acts;

"Insolvent Liquidation" means in respect of a particular legal entity:

- (a) that legal entity being placed into insolvent liquidation; or
- (b) any insolvency, bankruptcy or other corporate procedure occurring in which there is a final finding by
 a competent court that the relevant legal entity is unable to pay its debts as they fall due or unable to
 pay any specific undisputed debt; or

(c) any other similar insolvency related event in relation to that legal entity which constitutes an event which would result in FIA being entitled to terminate one or more of the 100 Year Agreements or would result in the A Share in FOAM converting into a deferred share;

"Intellectual Property" means:

- any patents, trademarks, service marks, designs, business names, copyrights, design rights, moral rights, inventions, confidential information, knowhow and other intellectual property rights and interests, whether registered or unregistered; and
- (b) the benefit of all applications and rights to use such assets of each Group Company;

"Intercreditor Agreement" means the intercreditor agreement dated on 24 November 2006 (as amended) and made between (among others) the Parent, the financial institutions defined therein as first lien lenders and NatWest Markets plc as senior security agent as most recently amended pursuant to an amendment agreement on or about the 2022 Effective Date;

"Interest Payment Date" means (i) in the case of a Term Benchmark Loan, the last day of each Interest Period (and, if the Interest Period is longer than three Months, on the day falling at three Monthly intervals after the first day of the Interest Period), (ii) in the case of any RFR Loan, each date that is on the numerically corresponding day in each Month that is one Month after the date of utilisation of such Loan and on the applicable Termination Date and (iii) in the case of an ABR Loan, the last day of each March, June, September and December and the Termination Date;

"Interest Period" means, in relation to a Term Benchmark Loan, each period determined in accordance with clause 15 (Interest Periods) and, in relation to an Unpaid Sum, each period determined in accordance with clause 14.3 (Default Interest);

"Interpolated Screen Rate" means, in relation to EURIBOR for any Loan, the rate rounded to the same number of decimal places as the two relevant Screen Rates which results from interpolating on a linear basis between:

- (a) the applicable Screen Rate for the longest period (for which that Screen Rate is available) which is less than the Interest Period of that Loan; and
- (b) the applicable Screen Rate for the shortest period (for which that Screen Rate is available) which exceeds the Interest Period of that Loan,

each as of the Specified Time on the Quotation Day for the currency of that Loan.

"Investor Debt" means Financial Indebtedness owing by a Group Company to any Investor and/or any other person who is a shareholder (directly or indirectly) in Parent but excludes, for the avoidance of doubt, indebtedness under any First Lien Finance Document, Second Lien Finance Document, Additional Unsecured Debt or any other Permitted Financial Indebtedness (other than clause (b) of the definition of Permitted Financial Indebtedness);

"Investors" mean the LMC Investors and any of their subsequent successors or assigns or transferees;

"IR Code" has the meaning given to that term in the definition of FATCA;

"ISDA CDS Definitions" has the meaning given to that term in clause 44 (Net Short Lenders).

"Issuing Bank" means each Lender identified above as issuing bank and any other Lender which has agreed to the Obligors' Agent request to be an Issuing Bank pursuant to the

terms of this agreement and has notified the Facility Agent of such agreement (and if more than one Lender has so agreed, such Lenders shall be referred to whether acting individually or together as the "Issuing Bank") provided that, in respect of a Letter of Credit issued or to be issued pursuant to the terms of this agreement, the "Issuing Bank" shall be the Issuing Bank which has issued or agreed to issue that Letter of Credit;

"Joint Venture" means any joint venture entity, whether a company, unincorporated firm, undertaking, association, joint venture or partnership or any other entity and provided that a Majority Owned Group Company or company in which a Group Company has a Permitted Licensing Minority Investment shall not be a Joint Venture for the purposes of this agreement unless otherwise designated as a "Joint Venture" in writing by the Obligors' Agent to the Facility Agent;

"Joint Venture Investment" has the meaning given to that term in the definition of Permitted Joint Venture;

"L/C Proportion" means in relation to a Lender in respect of any Letter of Credit, the proportion (expressed as a percentage) borne by that Lender's Available Commitment to the relevant Available Facility immediately prior to the issue of that Letter of Credit, adjusted to reflect any assignment or transfer under this agreement to or by that Lender;

"Legal Reservations" means:

- (a) the principle that equitable remedies are remedies which may be granted or refused at the discretion of the court, the principle of reasonableness and fairness, the limitation of enforcement by laws relating to bankruptcy, insolvency, liquidation, reorganisation, court schemes, moratoria, administration and other laws generally affecting the rights of creditors and secured creditors;
- (b) the time barring of claims under applicable limitation laws and defences of acquiescence, set-off or counterclaim (including the English Limitation Acts) and the possibility that an undertaking to assume liability for or to indemnify a person against non-payment of UK stamp duty may be void;
- (c) the principle that in certain circumstances Security granted by way of fixed charge may be recharacterised as a floating charge or that Security purported to be constituted as an assignment may be re-characterised as a charge;
- (d) the principle that additional interest imposed pursuant to any relevant agreement may be held to be unenforceable on the grounds that it is a penalty and thus void;
- (e) the principle that an English court may not give effect to an indemnity for legal costs incurred by an unsuccessful litigant;
- (f) the principle that the creation or purported creation of Security over any contract or agreement which is subject to a prohibition on transfer, assignment or charging may be void, ineffective or invalid and may give rise to a breach of the contract or agreement over which Security has purportedly been created;
- (g) similar principles, rights and defences under the laws of any relevant jurisdiction;
- (h) the possibility that the Transaction Security in force prior to the Sixth Restatement Date may not extend to secure the Secured Obligations (or any part of the Secured Obligations) as defined in the Intercreditor Agreement notwithstanding the terms of clause 12 (Guarantee, security and confirmation Sixth Restatement Date) of the Sixth Restatement Agreement or any analogous provision or confirmation in any document entered into after such date by virtue of the legal principles concerning variation of guaranteed or secured obligations; and

(i) any other general principles which are set out as qualifications or reservations (however described) as to matters of law in the legal opinions delivered to a Finance Party under the Finance Documents;

"Lender" means:

- (a) any Original Lender;
- (b) any Incremental Facility Lender; and
- (c) any person which has become a Party in accordance with clause 29 (Changes to the Lenders),

which in each case has not ceased to be a Party in accordance with the terms of this agreement;

"Letter of Credit" means:

- (a) a letter of credit, substantially in the form set out in schedule 10 (Form of Letter of Credit) or in any other form requested by Obligors' Agent and agreed by the Facility Agent and the Issuing Bank; or
- (b) any guarantee, indemnity or other instrument in a form requested by a Borrower (or Obligors' Agent on its behalf) and agreed by the Facility Agent and the Issuing Bank;

"Limited Condition Transaction" means (a) any acquisition (including by means of a merger, amalgamation or consolidation) or investment by any Obligor which is not conditioned upon the availability of, or on obtaining, third party financing or in connection with which any fee or expense would be payable by any Obligor to the seller or target in the event financing to consummate the acquisition is not obtained as contemplated by the definitive acquisition agreement, (b) any Restricted Payment the irrevocable declaration of which occurs in advance of such payment and (c) any redemption, repurchase, defeasance, satisfaction and discharge or repayment of indebtedness requiring irrevocable notice in advance of such redemption, repurchase, defeasance, satisfaction and discharge or repayment;

"Listing" means the admission of or the grant of permission to deal in any part of equity securities of any Group Company (or Parent Undertaking of any Group Company other than Liberty Media Corporation) on any regulated stock exchange or any other sales or issue by way of flotation or public offering of any Group Company (or Parent Undertaking of any Group Company other than Liberty Media Corporation);

"LMC Investor" means any of Liberty Media Corporation and any of its Subsidiaries (other than the Parent and its respective Subsidiaries) and any other person who is a shareholder (directly or indirectly) in the Parent and any of their Subsidiaries (other than the Parent and its respective Subsidiaries and other than any shareholder of Liberty Media Corporation);

"Loan" means a Term Loan or a Revolving Facility Loan:

"Luxembourg Obligor" means an Obligor incorporated in Luxembourg;

"Majority Lenders" means:

- (a) (for the purposes of clause 40.1(a) (Required Consents) in the context of a waiver in relation to a proposed Utilisation of a Revolving Facility of the conditions in clause 4.2 (Further Conditions Precedent)), the Majority RCF Lenders; and
- (b) (in any other case), means, at any time, a Lender or Lenders whose Commitments aggregate more than 50 per cent of the Total Commitments (or, if the Total

Commitments have been reduced to zero, aggregated more than 50 per cent of the Total Commitments immediately prior to that reduction),

in each case subject to clause 40 (Amendments and waivers);

"Majority Owned Group Company" a Group Company which is majority owned but not wholly owned by another Group Company or Group Companies;

"Majority RCF Lenders" means a Lender or Lenders whose Revolving Facility Commitments aggregate more than 50 per cent of the Total Revolving Facility Commitments (or, if the Total Revolving Facility Commitments have been reduced to zero, whose Revolving Facility Commitments aggregated more than 50 per cent of the Total Revolving Facility Commitments immediately prior to that reduction) provided that for the avoidance of doubt, any Revolving Facility Commitments established pursuant to clause 2.2 (Incremental Facilities) shall be included as Revolving Facility Commitments for the purposes of this definition;

"Majority TLA/RCF Lenders" means (subject to clause 40 (Amendments and waivers)) a Lender or Lenders whose aggregate Revolving Facility Commitments and Term Facility A Commitments aggregate more than 50 per cent of the aggregate of the Total Revolving Facility Commitments and the Total Facility A Commitments (or, if the Total Revolving Facility Commitments and the Total Facility A Commitments have been reduced to zero, whose aggregate Revolving Facility Commitments and Term Facility A Commitments aggregate more than 50 per cent of the aggregate of the Total Revolving Facility Commitments and the Total Facility A Commitments immediately prior to that reduction)provided that for the avoidance of doubt, any Revolving Facility Commitments established pursuant to clause 2.2 (Incremental Facilities) shall be included as Revolving Facility Commitments for the purposes of this definition;

"Margin" means:

- (a) in relation to any Term Facility A Loan, (i) prior to the first Adjustment Date occurring after the one year anniversary of the 2022 Effective Date, (x) in the case of a Term Benchmark Term Facility A Loan, 1.75 per cent per annum and (y) in the case of an ABR Term Facility A Loan, 0.75 per cent per annum and (ii) on and after the first Adjustment Date occurring after the one year anniversary of the 2022 Effective Date, a per annum percentage determined in accordance with the Pricing Grid;
- (b) in relation to any Term Facility B Loan, (i) initially, on and after the 2024 Effective Date (x) in the case of a Term Benchmark Term Facility B Loan, 2.00 per cent per annum and (y) in the case of an ABR Term Facility B Loan, 1.00 per cent per annum and (ii) on and after the first date on which the Net First Lien Secured Leverage is equal to or less than 3.75:1.0 on or following the earlier of (a) the consummation of the Dorna Acquisition or (b) the termination of the Dorna Acquisition, as reflected in the Compliance Certificate delivered to the Facility Agent pursuant to clause 25.2 (Provision and Contents of Compliance Certificate), (x) in the case of a Term Benchmark Term Facility B Loan, 1.75 per cent per annum and (y) in the case of an ABR Term Facility B Loan, 0.75 per cent per annum (in each case with effect from the date such Compliance Certificate is delivered to the Facility Agent; provided that in the case of clause (ii)(b) such step down shall be in effect from the date of such termination if the Net First Lien Secured Leverage is equal to or less than 3.75:1.0 as reflected in the most recent Compliance Certificate delivered prior to such termination);
- (c) in relation to any RCF Loan, (i) prior to the first Adjustment Date occurring after the one year anniversary of the 2022 Effective Date, (x) in the case of a Term Benchmark RCF Loan, 1.75 per cent per annum and (y) in the case of an ABR RCF Loan, 0.75 per cent per annum and (ii) on and after the first Adjustment Date

- occurring after the one year anniversary of the 2022 Effective Date, a per annum percentage determined in accordance with the Pricing Grid;
- (d) in relation to any Incremental Facility Loan, the rate per annum specified in the relevant Incremental Facility Commitment Letter, subject to any margin adjustments specified therein pursuant to Clause 2.2 (Incremental Facilities);

"Material Adverse Effect" means any event or circumstance which (taking account of all relevant circumstances) is materially adverse to:

- (a) the business, operations, property or condition, financial or otherwise of the Group, taken as a whole, that results in a material impairment of the ability of the Obligors to perform any payment obligations hereunder; or
- (b) subject to the Legal Reservations and any Perfection Requirements which are not overdue, affects the validity or the enforceability of any of the Finance Documents in a manner which would be materially adverse to the interests of the Lenders under the Finance Documents taken as a whole, and if capable of remedy, is not remedied within 20 Business Days of Obligors' Agent becoming aware of the issue or being given notice of the issue by the Facility Agent;

"Material Company" means, at any time:

- (a) an Obligor; or
- (b) a Subsidiary of the Parent which has Consolidated EBITDA (determined for this purposes in respect of the relevant Subsidiary (and its Subsidiaries) rather than the Group as a whole) representing 5.00% or more of Consolidated EBITDA, as defined in clause 26.1 (Financial Definitions) of the Group, calculated on a consolidated basis as determined by reference to the most recent Compliance Certificate supplied by the Parent related to the most recent audited consolidated financial statements of the Group provided to the Facility Agent.

A report by the Auditors of the Parent that a Subsidiary is or is not a Material Company shall, in the absence of manifest error, be conclusive and binding on all Parties;

"Material Assets" means material Intellectual Property related to the Business;

"Material Contracts" means the Umbrella Agreement and the Concorde Agreement;

"Minority Shareholder" means any shareholder (other than a Group Company) of a Majority Owned Group Company and any of their Affiliates (other than a Group Company);

"Month" means a period starting on one day in a calendar month and ending on the numerically corresponding day in the next calendar month, except that:

- (a) (subject to paragraph (c) below) if the numerically corresponding day is not a Business Day, that period shall end on the next Business Day in that calendar month in which that period is to end if there is one, or if there is not, on the immediately preceding Business Day;
- (b) if there is no numerically corresponding day in the calendar month in which that period is to end, that period shall end on the last Business Day in that calendar month; and
- (c) if an Interest Period begins on the last Business Day of a calendar month, that Interest Period shall end on the last Business Day in the calendar month in which that Interest Period is to end.

The above exceptions will only apply to the last Month of any period; "Monthly" shall be construed accordingly;

"Moody's" means Moody's Investors Service Limited or any successor to its rating business;

"Net First Lien Secured Leverage" has the meaning given to it in clause 26.1 (Financial Definitions);

"Net Short Lender" has the meaning given to that term in clause 44 (Net Short Lenders).

"Net Short Representation" has the meaning given to that term in clause 44 (Net Short Lenders).

"Net Total Leverage" means the ratio of (i) Consolidated Total Net Debt as at the relevant Testing Date to (ii) Consolidated EBITDA for the most recently ended Testing Period on or prior to such Testing Date;

"Net Total Secured Leverage" has the meaning given to that term in clause 26.1 (Financial Definitions);

"New Shareholder Injections" has the meaning given to that term in clause 26.1 (Financial Definitions);

"Non-Acceptable L/C Lender" means a Lender under a Revolving Facility which:

- (a) is not an Acceptable Bank within the meaning of paragraph (a) of the definition of "Acceptable Bank" (other than a Lender which each Issuing Bank has agreed is acceptable to it notwithstanding that fact); or
- (b) is a Defaulting Lender; or
- (c) has failed to make (or has notified the Facility Agent that it will not make) a payment to be made by it under clause 7.3 (Indemnities) or clause 31.10 (Lenders' Indemnity to the Facility Agent) or any other payment to be made by it under the Finance Documents to or for the account of any other Finance Party in its capacity as Lender by the due date for payment unless the failure to pay falls within the description of any of those items set out at (i) to (iii) of the definition of "Defaulting Lender";

"Non-Obligor" means a Group Company which is not an Obligor;

"NYFRB" means the Federal Reserve Bank of New York.

"NYFRB Rate" means means, for any day, the greater of (a) the Federal Funds Effective Rate in effect on such day and (b) the Overnight Bank Funding Rate in effect on such day (or for any day that is not a Business Day, for the immediately preceding Business Day); provided that if none of such rates are published for any day that is a Business Day, the term "NYFRB Rate" means the rate for a federal funds transaction quoted at 11:00 a.m. (New York City time) on such day received by the Facility Agent from a federal funds broker of recognized standing selected by it; provided, further, that if any of the aforesaid rates as so determined be less than zero, such rate shall be deemed to be zero for purposes of this agreement.

"NYFRB's Website" means the website of the NYFRB at http://www.newyorkfed.org, or any successor source.

"Obligor" means a Borrower or a Guarantor (and, for the avoidance of doubt, does not include a shareholder in the Parent notwithstanding whether it grants Security to the Security Agent);

"Obligors' Agent" means initially Formula One Management Limited and, upon written notice to the Facility Agent, such other member of the Group that assumes in writing in form and substance reasonably satisfactory to the Facility Agent the role of Obligors' Agent hereunder, in each case, appointed to act on behalf of each Obligor in relation to the Finance Documents pursuant to clause 2.5 (Obligors' Agent);

"OFAC" means the Office of Foreign Assets Control of the US Department of the Treasury;

"Operating Expense Receivable" means any loan or other receivable outstanding from a direct or indirect shareholder of the Parent as a result of a payment permitted under paragraph (b) of the definition of "Permitted Payment";

"Optional Currency" means a currency (other than the Base Currency) which complies with the conditions set out in clause 4.3 (Conditions Relating to Optional Currencies);

"Original Lender" means a 2024 Effective Date Original Lender;

"Original Obligor" means an Original Borrower or an Original Guarantor;

"Original RCF Lender" means each bank, financial institution or other entity listed in Part 3 of schedule 1 as an RCF Lender;

"Original Term Facility A Lender" means each bank, financial institution or other entity listed in Part 4 of schedule 1 as an Term Facility A Lender;

"Overnight Bank Funding Rate" means, for any day, the rate comprised of both overnight federal funds and overnight eurodollar transactions denominated in Dollars by U.S.-managed banking offices of depository institutions, as such composite rate shall be determined by the NYFRB as set forth on the NYFRB's Website from time to time, and published on the next succeeding Business Day by the NYFRB as an overnight bank funding rate.

"Parent Reporter" means any person that is a direct or indirect parent entity (which may be organized as, among other things, a partnership) of Parent.

"Parent Reporter Option" has the meaning given to it in clause 25.1 (Financial Statements);

"Parent Share Charge" means the specific security agreement dated on or about the 2022 Effective Date between Liberty GR Acquisition Company Limited and the Security Agent in relation to shares in the Parent;

"Parent Undertaking" has the meaning given to that term in section 1162 of the Companies Act 2006;

"Participating Member State" means any member state of the European Community that has the euro as its lawful currency in accordance with legislation of the European Community relating to Economic and Monetary Union;

"Party" means a party to this agreement;

"Perfection Requirements" means any and all registrations, filings, notices and other actions and steps required to be made in any jurisdiction in order to perfect security created by the Transaction Security Documents or in order to achieve the relevant priority for such Security;

"Permitted Acquisition" means the following acquisitions or investments (other than investments in which a person becomes a creditor in respect of any Financial Indebtedness, which are separately permitted to the extent included under "Permitted Loans"):

- (a) an acquisition or investment in the ordinary course of its trading activities;
- (b) (i) an acquisition by a member of the Group of an asset sold, leased, transferred or otherwise disposed of by another member of the Group (including through an Unrestricted Subsidiary acting solely as an intermediary to the extent made in a substantially concurrent transaction) and (ii) an investment made by and among Group Companies (including through an Unrestricted Subsidiary to the extent such Unrestricted Subsidiary is acting solely as intermediary for such investment and such investment is substantially concurrently invested in a Group Company);
- (c) an acquisition of or investment in securities which are Cash Equivalent Investments;
- (d) the incorporation of a limited liability company or the purchase of shares in an off the shelf limited liability company which becomes a member of the Group;
- (e) an investment in or acquisition of a Permitted Joint Venture;
- (f) (i) an acquisition or investment pursuant to a Permitted Transaction, (ii) an acquisition or investment related to a Permitted Licensing Minority Investment, and (iii) Restricted Payments permitted by clause 27.15 (Dividends and Share Redemption) (to the extent constituting an investment or acquisition):
- (g) Capital Expenditure for the purposes of the business of the relevant Group Company;
- (h) an asset acquired by a Group Company which is a Non-Obligor from another Non-Obligor provided it is not otherwise restricted by this agreement, the Intercreditor Agreement or any other Finance Document;
- (i) an asset (other than shares) acquired (on terms no less favourable to the Obligor than arm's length terms) by an Obligor from a Non-Obligor and required by the Obligor for its business;
- (j) an acquisition by an Obligor from an Obligor provided that the relevant asset is, in the hands of the transferee, secured by legally valid and effective Security over the asset of the same type and level (if any) as the asset was subject prior to that acquisition and securing all or substantially all amounts outstanding under the Finance Documents, to the extent provided for in the Agreed Security Principles and (other than in respect of the acquisition of shares in members of the Group) clause 30.8 (Further Security);
- (k) an acquisition of aircraft to replace any aircraft owned by the Group on the 2022 Effective Date (or any replacement thereof) that, in the good faith judgment of the Parent, provides substantially similar utility to such aircraft being replaced as if such replaced aircraft was new;
- (I) any acquisition of shares in joint ventures or investments where the amount invested when aggregated with all other investments then outstanding permitted under this sub-clause (I) does not exceed in any Financial Year the greater of (i) \$2,500,000 (or its equivalent in other currencies) and (ii) 0.55% of the then most recently reported Consolidated EBITDA at the time of such acquisition or investment;
- (m) any acquisition or investment with the prior written consent of the Majority Lenders;
- (n) any acquisition or investment provided that:

- (i) as at the date such acquisition or investment is contractually committed, no Event of Default is continuing or would occur as a result of such acquisition or investment; and
- (ii) the aggregate consideration involved in respect of all acquisitions or outstanding investments made under this paragraph (n) does not exceed the sum of (x) the greater of (a) \$500,000,000 (or its equivalent in other currencies) and (b) 100% of the then most recently reported Consolidated EBITDA at the time of such acquisition or investment plus (y) any amount funded from the Available Amount Basket after the 2022 Effective Date;
- (o) any acquisition or investment using consideration consisting of equity interests of any Group Company (other than Disqualified Equity unless such Disqualified Equity is Permitted Financial Indebtedness), provided that:
 - (i) in the case of an acquisition of equity interests of a person, such person becomes a Group Company; or
 - (ii) in the case of an acquisition of assets other than equity interests, such assets are acquired by a Group Company; and
 - (iii) after giving pro forma effect to each such acquisition, no Default has occurred and is continuing; and
 - (iv) the Group shall have complied with all applicable Agreed Security Principles and clause 30.8 (Further Security), in each case, to the extent required to be complied with;
- (p) acquired investments in existence at the time such Person becomes a Group Company; provided such investment was not made in connection with or anticipation of such person becoming a Subsidiary;
- (q) acquisitions or investments by any Group Company pursuant to this clause (q), if the Group would otherwise be permitted to make a Restricted Payment under Clause 27.15 (Dividends and Share Redemptions) in such amount;
- (r) any acquisition or investment provided that:
 - no Default has occurred and is continuing when the acquisition is made or would arise as a result; and
 - (ii) Net First Lien Secured Leverage is equal to or less than 5.50:1 after giving pro forma effect to such acquisition or investment;
- (s) investments made in Unrestricted Subsidiaries in an aggregate amount, together with all other outstanding investments made by the Group under this clause (s), not to exceed (i) the greater of (x) \$100,000,000 (or its equivalent in other currencies) and (y) 20% of the then most recently reported Consolidated EBITDA at the time of such investments in any Fiscal Year and (ii) the greater of (x) \$210,000,000 (or its equivalent in other currencies) and (y) 42% of the then most recently reported Consolidated EBITDA at the time of such investments during the term of this agreement; provided that availability under this clause (s) shall be reduced by the aggregate amount of all outstanding loans made pursuant to clause (l) of the definition of Permitted Loans;
- acquisitions or investments in securities of trade creditors or customers received pursuant to any plan
 of reorganization or similar arrangement upon the bankruptcy or insolvency of such trade creditors or
 customers;

- (u) acquisitions or investments in stock, obligations or securities received in settlement of debts created in the ordinary course of business and owing to any Group Member or in satisfaction of judgments;
- (v) any investment existing on, or made pursuant to binding commitments existing on, the 2022 Effective Date and, solely to the extent in excess of \$5,000,000 (or its equivalent in other currencies), set forth on Schedule 14 part 1.

<u>provided</u> that no investment in an Unrestricted Subsidiary shall be made with Material Assets. It is understood and agreed that the outstanding amount of any Permitted Acquisition under this agreement shall be reduced by any returns on such Permitted Acquisition received by any Group Company without duplication of any returns on Permitted Acquisitions that increase the Available Amount Basket.

"Permitted Debt Change" means any matter referred to in clause 2.2(s) (Incremental Facilities) or 2.3(d) (Additional Debt);

"Permitted Disposal" means any Permitted Intercompany Debt Disposal, any Permitted Group Company Disposal and any sale, lease, licence, transfer or other disposal:

- (a) in the ordinary course of business of the disposing entity;
- (b) of any asset by a member of the Group (the "Disposing Company") to another member of the Group (the "Acquiring Company"), excluding any transfer of assets to FOAM (or a subsidiary of FOAM) which is not a Permitted Transaction;
- (c) of assets (other than Material Contracts) in exchange for other assets which in the opinion of the Disposing Company (acting reasonably) are comparable or superior as to type, value or quality;
- (d) of assets which are seized, expropriated or acquired by compulsory purchase by or by the order of any central or local government authority;
- (e) of surplus or obsolete assets (including aircraft) no longer required for the purposes of the business carried on by the relevant Group Company;
- (f) of Cash or Cash Equivalent Investments;
- (g) constituted by a licence of intellectual property rights which are not essential to the business of the Group as a whole but excluding any licence to FOAM unless it is a Permitted Transaction;
- (h) to a joint venture, if satisfying the criteria of a Permitted Joint Venture;
- (i) arising as a result of any Permitted Security or Permitted Transaction;
- (j) of assets which are simultaneously upon disposal leased back to the disposing entity where the higher of the market value or consideration receivable for such assets does not exceed the greater of (i) \$31,500,000 (or its equivalent in other currencies) and (ii) 6.30% of the then most recently reported Consolidated EBITDA at the time of such sale, lease, licence, transfer or disposal;
- (k) to which the prior written consent of the Majority Lenders has been obtained;
- (I) any disposal to another Group Company required or expressly permitted by the 100 Year Agreements;
- (m) arising as a result of a Permitted Share Issuances, Permitted Distributions and/or Permitted Acquisitions;

- (n) of any asset (other than Material Contracts) provided that (i) the consideration received for such asset shall be at least 75% cash and in an aggregate amount of at least equal to the fair market value thereof (provided, however, that (x) any Designated Non-Cash Consideration received by such Obligor having an aggregate fair market value, taken together with all other Designated Non-Cash Consideration received pursuant to this subclause (n) that is at that time outstanding, not in excess of the greater of (1) \$50,000,000 (or its equivalent in other currencies) and (2) 10% of the then most recently reported Consolidated EBITDA, calculated at the time of the receipt of the Designated Non-Cash Consideration (with the fair market value of each item of Designated Non-Cash Consideration being measured at the time received and without giving effect to subsequent changes in value), shall be deemed to be cash) and (ii) no Default shall have occurred and be continuing or would result therefrom;
- (o) of assets where the higher of the market value or consideration receivable does not exceed in any Financial Year the greater of (i) \$52,500,000 (subject to clause 27.31 (Basket Carry Forward)) (or its equivalent in other currencies) and (ii) 10.50% of the then most recently reported Consolidated EBITDA at the time of such sale, lease, licence, transfer or disposal;
- any sale of equity interests in, or Financial Indebtedness or other securities of, an Unrestricted Subsidiary (other than Unrestricted Subsidiaries substantially all of the assets of which are cash and Cash Equivalents);
- (q) existing on the 2022 Effective Date or that are subject to binding commitments existing on the 2022 Effective Date and, solely to the extent in excess of \$5,000,000 (or its equivalent in other currencies), set forth on Schedule 14 part 2;
- (r) so long as:
 - no Default has occurred and is continuing when such sale, lease, licence, transfer or other disposal is made or would arise as a result; and
 - (ii) Net First Lien Secured Leverage is equal to or less than 4.50:1 after giving pro forma effect to such sale, lease, licence, transfer or other disposal; and
- (s) of loans, debts or receivables owed to any Group Company either (x) in the ordinary course of business or (y) to the extent such amounts have been impaired or provided against in good faith by the management of such Group Company;

<u>provided</u> that notwithstanding the foregoing, in no event shall a Group Company dispose of, contribute or otherwise transfer to an Unrestricted Subsidiary any Material Asset.

"Permitted Distribution" means:

- (a) Restricted Payments (i) to the Parent or any of the Group Companies (ii) to Investors or any Minority Shareholder payable solely in Qualified Equity and/or (iii) declared or made by any Group Company to an Unrestricted Subsidiary acting solely as an intermediary to the extent the proceeds of which are substantially concurrently distributed or paid by such Unrestricted Subsidiary to a Group Company;
- (b) Restricted Payments provided that:
 - (i) no Default has occurred and is continuing when the payment is made or would arise as a result; and
 - (ii) Net First Lien Secured Leverage is equal to or less than 5.50:1 after giving pro forma effect to such payment;

- (c) Restricted Payments pursuant to and in accordance with stock option plans or other benefit plans for directors, management, employees or consultants of any Group Company (and its parent companies);
- (d) Restricted Payments in exchange for, or out of the proceeds of the substantially concurrent issuance and sale of, Qualified Equity;
- (e) Restricted Payments provided that the aggregate amount permitted under this paragraph (e) may not exceed the greater of (i) \$50,000,000 (or its equivalent in other currencies) and (ii) 10% of the most recently reported Consolidated EBITDA at the time of such Restricted Payment in aggregate per calendar year; provided that 100% of the unused amount in respect of this paragraph (e) may be carried forward to any succeeding calendar years and utilized in such calendar years;
- (f) Restricted Payments provided that:
 - (i) no Default is continuing or would occur immediately after the making of that payment; and
 - (ii) it is funded from the then Available Amount Basket; provided that usage of the EBITDA Builder Prong of the Available Amount Basket for this paragraph (f) shall only be permitted if Net Total Secured Leverage is equal to or less than 8.00:1 after giving pro forma effect to such payment;
- a direct or indirect payment made to any Investor that is a United States person (within the meaning (g) of Section 7701(a)(30) of the IR Code) to enable such Investor to pay US federal, state or local income, franchise or similar US Taxes owed by it (or by any consolidated, combined or similar tax group of which it is a member (for purposes of this definition, a "tax group")) or as compensation for any US federal, state or local income, franchise or similar US tax assets utilized by such Investor or its tax group, in either case, with respect to the income allocated to such Investor or tax group (or otherwise includible in such Investor's or Investor's tax group's annual income) attributable to such Investor's direct or indirect ownership of interests in the Parent or any of its subsidiaries; provided that the amount of Permitted Distributions under this clause (g) for an applicable taxable period shall not exceed the product of (i) the net taxable income of the Parent and its applicable subsidiaries (if the Parent or any such subsidiary is a disregarded entity for U.S. federal income tax purposes, treating such Person as a regarded entity for purposes of determining whether income is of the Parent or its subsidiaries) that would be required to be reported on tax returns of such Investor or such Investor's tax group for such taxable period for US federal income tax purposes (which amount, for the avoidance of doubt, shall include the amount of any "global intangible low-taxed income" (within the meaning of Section 951A of the IR Code) and the amount of any "subpart F income" (within the meaning of Section 952 of the IR Code), in each case, that would be required to be reported on tax returns of such Investor or such Investor's tax group for such taxable period, to the extent attributable to such Investor's direct or indirect ownership of interests in the Parent or any of its subsidiaries or any assets or operations of the foregoing (with such "global intangible low-taxed income" and "subpart F income" each to be treated as not less than zero for this purpose), with such modifications to such calculation (including by making such calculation on an entity by entity standalone basis) as such Investor reasonably may make in good faith to reflect any change in applicable Tax law or, with the consent of the Facility Agent (which consent shall not be unreasonably withheld or conditioned), to reflect a change in the organizational structure of the Group that, in either case, becomes effective after the 2022 Effective Date, and (ii) the highest combined marginal US federal, state and local tax rate for income, franchise and similar US taxes applicable to such Investor or such Investor's tax group (taking into account the character of

the taxable income in question and any preferential tax rates and the deductibility of state and local income taxes for US federal income tax purposes (and any applicable limitations thereon)); provided further that the amount of payments made in respect of an Unrestricted Subsidiary will be permitted only to the extent that cash or Cash Equivalent were paid by Unrestricted Subsidiaries to an Obligor for such purpose; provided further that nothing in this clause (g) shall require any Investor or its tax group to provide a copy of any tax return or any other tax information that it deems confidential;

- (h) Restricted Payments to any Minority Shareholder provided that (i) the aggregate amount permitted under this paragraph (i) may not exceed the greater of (A) \$52,500,000 (or its equivalent in other currencies) in aggregate in any Financial Year of the Parent and (B) 10.50% of the then most recently reported Consolidated EBITDA) at the time of such payment, dividend or other distribution and (ii) immediately after giving effect thereto on a pro forma basis, no Default shall have occurred and be continuing:
- any Group Company (other than Parent) may make Restricted Payments ratably with respect to their equity interests;
- (j) so long as no Default shall have occurred and be continuing at the time of such Restricted Payment or would arise as a result, immediately after giving pro forma effect to such Restricted Payment, any Restricted Payments to LMC and its subsidiaries to the extent necessary to pay principal and interest (including any related premiums, defeasance costs and fees and expenses in connection therewith) when due in respect of any Financial Indebtedness of LMC and/or its subsidiaries, provided that the amount of such Restricted Payment shall not exceed the sum of:
 - (i) an unlimited amount so long as after giving pro forma effect to such Restricted Payment, Net First Lien Secured Leverage is equal to or less than 6.5:1; plus
 - (ii) an amount not exceed the greater of (x) \$75,000,000 (or its equivalent in other currencies) and (y) 15% of the most recently reported Consolidated EBITDA at the time of such Restricted Payment in aggregate per calendar year; provided that 100% of the unused amount in respect of this clause (k)(ii) may be carried forward to any succeeding calendar years and utilized in such calendar years;
- (k) following the consummation of a Qualifying IPO, Restricted Payments in an amount in any Financial Year not to exceed 6.00% of the prior Financial Year's market capitalization of the Parent or such direct or indirect parent, so long as no Default shall have occurred and be continuing at the time of such Restricted Payment or would arise as a result;
- (I) any Restricted Payments made on or prior to the 2022 Effective Date; and
- (m) any Restricted Payments of equity interests of, or indebtedness owed to any Group Company by, Unrestricted Subsidiaries (other than Unrestricted Subsidiaries substantially all of the assets of which are cash and Cash Equivalents);
- (n) to the extent constituting Restricted Payments, Permitted Security and Permitted Acquisitions; and
- (o) any Restricted Payment made by any Obligor for Group Relief which has been surrendered to that Obligor, provided that, if the payment is to an entity that is not a member of the Group:

- (i) the payment does not exceed an amount equal to the rate of corporation tax applicable to that Obligor for the period multiplied by the amount of Group Relief surrendered to it;
- (ii) that Obligor makes the payment no earlier than the date on which, but for the surrender, it would otherwise have been liable to pay such tax or, if applicable, the date on which it would have been liable to pay the final instalment of the corporation tax liability for the accounting period in question; and
- (iii) the Group is no worse off as a result of such payment.

"Permitted Financial Indebtedness" means:

- (a) Financial Indebtedness arising under the First Lien Finance Documents (in the case of any Additional First Lien Finance Documents (as defined in the Intercreditor Agreement), as permitted by clause 2.3 (Additional Debt)), including in respect of any Incremental Facilities established pursuant to clause 2.2 (Incremental Facilities);
- (b) Financial Indebtedness arising in respect of Investor Debt (provided it is subordinated at least to the same extent as Investor Liabilities under and as defined in the Intercreditor Agreement or in the case of any Financial Indebtedness the benefits of, and/or creditors rights in relation to, which are being sold, transferred, distributed, assigned or disposed of pursuant to any Permitted Intercompany Debt Disposal, becomes so subordinated on the date of the relevant sale, transfer, distribution, assignment or disposal after taking into account all sales, transfers, distributions, assignments and other disposals occurring on the same date and subject always to the terms of this agreement and the Intercreditor Agreement;
- (c) unsecured overdraft or working capital facilities to the extent covered by a Letter of Credit or other letter of credit, quarantee or indemnity issued under an Ancillary Facility;
- (d) Financial Indebtedness arising under a foreign exchange transaction for spot or forward delivery entered into in connection with protection against fluctuation in currency rates where that foreign exchange exposure arises in the ordinary course of business or in respect of Utilisations made in Optional Currencies, but not a foreign exchange transaction for investment or speculative purposes;
- (e) Financial Indebtedness arising under (i) a Permitted Loan or (ii) a Permitted Guarantee or (iii) as permitted by clause 27.25 (Treasury Transactions);
- (f) Financial Indebtedness of any person acquired by a member of the Group after the Sixth Restatement Date which is incurred under arrangements in existence at the date of acquisition, but not incurred or increased or its maturity date extended in contemplation of, or since, that acquisition, and outstanding only for a period of three months following the date of acquisition;
- (g) Financial Indebtedness under finance or capital leases, provided that the aggregate capital value of all such items so leased under outstanding leases by members of the Group does not exceed the greater of (i) \$52,500,000 (or its equivalent in other currencies) and (ii) 10.50% of the then most recently reported Consolidated EBITDA at the time incurred, created or assumed;
- (h) Financial Indebtedness pursuant to a Permitted Security;
- (i) Financial Indebtedness owed by one Obligor to another Obligor;
- (j) Financial Indebtedness owed by one Non-Obligor to another Non-Obligor or an Obligor;

- (k) Financial Indebtedness owed by an Obligor to a Non-Obligor;
- (I) Financial Indebtedness owed by a Non-Obligor to an Obligor up to a maximum outstanding aggregate amount at the time of the then most recent incurrence or assumption equal to the greater of (i) \$26,250,000 (or its equivalent in other currencies) and (ii) 5.25% of the then most recently reported Consolidated EBITDA;
- (m) Permitted Refinancing Indebtedness;
- (n) Financial Indebtedness which is permitted under clause 2.3 (Additional Debt);
- (o) unsecured Financial Indebtedness in an amount (measured solely when incurred, created or assumed) that does not exceed the greater of (i) an amount such that the Net Total Leverage would not exceed (x) 8.25:1 or (y) solely in the case of any such unsecured Financial Indebtedness being incurred to finance a Permitted Acquisition, the Net Total Leverage immediately prior to the incurrence of such unsecured Financial Indebtedness and (ii) an amount such that the Consolidated Interest Coverage Ratio would not be less than 2.00:1, in each case, on a pro forma basis immediately after giving effect to such Financial Indebtedness and the use of the proceeds thereof;
- (p) Financial Indebtedness of any Group Company in an aggregate outstanding principal amount (such outstanding amount measured solely when incurred, created or assumed) not to exceed 100% of the amount of net cash proceeds received by the Parent from (i) the issuance or sale of common equity interests or (ii) any cash contribution to its common equity;
- (q) Financial Indebtedness the outstanding amount of which does not exceed a maximum aggregate amount at the time of the then most recent incurrence or assumption equal to the greater of (i) \$52,500,000 (or its equivalent in other currencies) in aggregate for the Group and (ii) 10.50% of the then most recently reported Consolidated EBITDA;
- (r) Financial Indebtedness of any Obligor arising in respect of Group Relief which has been surrendered to that Obligor; and
- (s) Financial Indebtedness of any Group Company to the extent outstanding on the 2022 Effective Date and, solely to the extent in excess of \$5,000,000 (or its equivalent in other currencies), set forth on Schedule 14 part 3.

"Permitted Group Company Disposal" means the disposal or winding up or dissolution and entry into liquidation of any Group Company (other than as a result of any Group Company being placed into insolvent liquidation, bankruptcy, suspension of payments or similar insolvency proceeding and other than the Parent) provided that (i) the requirements of paragraph (b) of the definition of "Permitted Transaction" are met (where applicable), (ii) at the time of such Permitted Group Company Disposal the relevant Group Company is not a shareholder of any other Group Company, (iii) the relevant Group Company is or has become a Dormant Subsidiary and (iv) the Permitted Group Company Disposal shall not materially adversely affect the value of the security of the Lenders under the Finance Documents in any respect;

"Permitted Group Company Release" means a release of the Transaction Security created or expressed to be created over the shares in a Group Company in connection with a transfer of such shares to another Group Company that is permitted under this agreement provided that contemporaneously with such release such shares will become subject to Transaction Security under a Transaction Security Document consistent with the Agreed Security Principles to which the acquiring Group Company is party;

"Permitted Guarantee" means:

- (a) the endorsement of negotiable instruments in the ordinary course of business;
- (b) any performance or similar bond guaranteeing performance by a member of the Group under any contract entered into in the ordinary course of business;
- (c) any guarantee of a joint venture satisfying the Permitted Joint Venture criteria;
- (d) any guarantee that is Permitted Financial Indebtedness;
- (e) any guarantee given in respect of the netting or set-off arrangements permitted pursuant to paragraph (b) of Permitted Security;
- (f) a guarantee by a member of the Group of the obligations of an Obligor;
- (g) a guarantee by a member of the Group of a Non-Obligor's obligations arising in the ordinary course of business;
- (h) any guarantee given by a person acquired pursuant to a Permitted Acquisition and existing at the time of that Permitted Acquisition provided such guarantee is discharged within three months of the Permitted Acquisition completing;
- any guarantee of a Treasury Transaction which is permitted in accordance with clause 27.25 (Treasury Transactions);
- (j) any guarantee with the prior written consent of the Majority Lenders;
- (k) guarantees given to professional advisers and consultants in the ordinary course of business;
- indemnities given in favour of directors and officers of any member of the Group in respect of their functions as such;
- (m) any guarantee of First Lien Liabilities granted or in effect on or about the 2022 Effective Date and/or permitted by the Intercreditor Agreement;
- (n) [Reserved];
- (o) any guarantee provided that the total aggregate amount permitted under this paragraph (o) may not exceed a maximum aggregate amount outstanding at the time of the then most recent incurrence equal to the greater of (i) \$26,250,000 (or its equivalent in other currencies) in aggregate for the Group and (ii) 5.25% of the then most recently reported Consolidated EBITDA; and
- (p) guarantees provided by any Group Company to the extent outstanding on the 2022 Effective Date and, solely to the extent in excess of \$5,000,000 (or its equivalent in other currencies), set forth on Schedule 14 part 4.

"Permitted Intercompany Debt Disposal" means:

- any sale, transfer, distribution, assignment or other disposal to any Permitted Intercompany Debt
 Transferee of the benefits of, and/or creditor's rights in relation to, any Financial Indebtedness or
 other indebtedness owed by or outstanding from any Group Company;
- (b) any sale, transfer, distribution, novation or other disposal to any Group Company of the obligations in relation to any Financial Indebtedness or other indebtedness owed by or outstanding from any Group Company (other than any Financial Indebtedness or other indebtedness owed or outstanding under any Finance Document) to any

- other Group Company (provided that no cash consideration is paid to any Non-Obligor for the purposes of any such disposals); and
- (c) any cancellation or conversion into equity (provided that such equity is issued to or otherwise owned by a Group Company or, in the case of equity of the Parent, a shareholder of the Parent) of any Financial Indebtedness or other indebtedness owed by or outstanding from any Group Company (other than any Financial Indebtedness or other indebtedness owed or outstanding under any Finance Document),

"Permitted Intercompany Debt Transferee" means any Obligor;

"Permitted Investor Release" means a release of the Security created or expressed to be created by an Investor over Investor Debt provided that the Parent or Obligors' Agent confirms to the Security Agent that there is no Investor Debt outstanding to such Investor from any Group Company at such time;

"Permitted Joint Venture" means (x) any Joint Ventures existing on the 2022 Effective Date and set forth on Schedule 14 part 5 and (y) any Joint Venture where:

- (a) the Joint Venture is incorporated, or established, and carries on its principal business in the United Kingdom, European Union, United States of America, China, Hong Kong or Singapore or the jurisdiction of incorporation of a Guarantor;
- (b) the Joint Venture is engaged in a business substantially the same or in business that is incidental, complementary, similar related or ancillary thereto (including reasonable extensions thereof) as that carried on by the Group; and
- (c) after the 2022 Effective Date, the aggregate outstanding amount (the **"Joint Venture Investment"**) (measured solely when such Joint Venture Investment is made) of:
 - all amounts subscribed for shares in, lent to, or invested in all such Joint Ventures by any Group Company (other than amounts subscribed for shares in, lent to, or invested in any Subsidiary of such Joint Venture by such Joint Venture);
 - (ii) the contingent liabilities of any Group Company under any guarantee given in respect of the liabilities of any such Joint Venture; and
 - (iii) the market value of any assets transferred by any Group Company to any such Joint Venture (not including assets transferred by such Joint Venture to any of its Subsidiaries to the extent already included in the calculation pursuant to this paragraph (iii)),

does not exceed the greater of (A) \$100,000,000 (or its equivalent in other currencies) and (B) 20% of the then most recently reported Consolidated EBITDA, in each case, plus any amount funded from the Available Amount Basket;

"**Permitted Jurisdiction**" means any member country of The Organisation for Economic Co-operation and Development (OECD) (or any successor thereto), as such list of member counties may be updated from time to time, and any other country as may be reasonably approved by the Facility Agent.

"Permitted Licencing Minority Investment" means a shareholding in a company or corporation (or analogous equity interest in another entity) of less than 50% of the total shares of that company or corporation (or analogous equity interest in another entity) that has been acquired by a Group Company in connection with a licensing transaction in the ordinary course of business;

"Permitted Loan" means:

- (a) a loan made in the ordinary course of business;
- (b) Financial Indebtedness which is referred to in the definition of, or otherwise constitutes, Permitted Financial Indebtedness (except the reference to Permitted Loan under paragraph (e)(i) of that definition);
- (c) a loan made to a joint venture satisfying the Permitted Joint Ventures criteria;
- any intra-Group loan made for the purposes of enabling an Obligor to meet its payment obligations under the Secured Debt Documents (as defined in the Intercreditor Agreement);
- (e) a loan made by an Obligor to another Obligor, a loan made by a member of the Group which is not an Obligor to another member of the Group, a loan permitted under paragraph (I) of the definition of Permitted Financial Indebtedness or a loan made by a member of the Group that is an Obligor to another member of the Group that is not an Obligor;
- (f) a loan made by a member of the Group to an employee or director of any member of the Group if the amount of that loan when aggregated with the amount of all loans to employees and directors by members of the Group then outstanding does not exceed at the time such loan is made the greater of (i) \$15,750,000 (or its equivalent in other currencies) and (ii) 3.15% of the then most recently reported Consolidated EBITDA;
- (g) any credit arising in connection with a disposal of an asset or assets with a market value of less than \$2,000,000 (or its equivalent in other currencies) entered into in accordance with clause 27.12 (Arm's Length Basis) below;
- (h) loans required to be made by mandatory provision of law;
- (i) a loan constituting a Permitted Payment, Permitted Acquisition or a Permitted Distribution (or which would so constitute if such loan was made as a Restricted Payment);
- (j) any loan so long as the aggregate amount of the Financial Indebtedness under any such loans outstanding does not exceed at the time such loan is made the greater of (i) \$52,500,000 (or its equivalent in other currencies) and (ii) 10.50% of the then most recently reported Consolidated EBITDA:
- (k) any loan provided that:
 - (i) no Default has occurred and is continuing when the loan is made; and
 - (ii) Net First Lien Secured Leverage is equal to or less than 5.50:1 after giving pro forma effect to such loan;
- (I) loans made to Unrestricted Subsidiaries in an aggregate amount, together with all other loans made by the Group under this clause (I) and still outstanding, not to exceed (i) the greater of (x) \$100,000,000 (or its equivalent in other currencies) and (y) 20% of the then most recently reported Consolidated EBITDA at the time of such loan; provided that availability under this clause (I) shall be reduced by the aggregate amount of all outstanding investments made pursuant to clause (s) of the definition of Permitted Acquisitions;

- (m) any loan existing on the 2022 Effective Date and, solely to the extent in excess of \$5,000,000 (or its equivalent in other currencies), set forth on Schedule 14 part 6; and
- (n) any loan provided that:
 - as at the date such loan, no Event of Default is continuing or would occur as a result of such loan; and
 - the aggregate amount of such loan at the time made shall not exceed the then Available Amount Basket;

It is understood and agreed that the outstanding amount of any Permitted Loan made pursuant to this agreement shall be reduced by any repayments or reductions of such Permitted Loans.

"Permitted Local Law Release" means a release of the English law governed Transaction Security created or expressed to be created over the shares in a Group Company that is not incorporated in England and Wales provided that contemporaneously with such release such shares will become subject to Transaction Security under a Transaction Security Document consistent with the Agreed Security Principles and governed by the law of the jurisdiction of incorporation of the relevant Group Company or another jurisdiction acceptable to the Security Agent acting reasonably;

"Permitted Parent Share Release" means a release of the Transaction Security created or expressed to be created over the shares in the Parent in connection with a transfer of such shares that is permitted under this agreement provided that contemporaneously with such release such shares will become subject to Transaction Security under a Transaction Security Document to which the acquiring shareholder is party consistent with the Agreed Security Principles and on substantially the same terms as the Parent Share Charge;

"Permitted Payment" means:

- (a) any payment under the First Lien Finance Documents or the Second Lien Finance Documents (as defined in the Intercreditor Agreement);
- (b) dividends by the Parent or other payments by any Group Company to fund reasonable legal, audit, professional and advisory costs and administrative costs incurred by any direct or indirect shareholder of the Parent;
- (c) the payment of monitoring fees and reasonable directors fees in a total amount not exceeding in any Financial Year the greater of (i) \$3,500,000 (or the equivalent in other currencies) and (ii) 0.70% of the then most recently reported Consolidated EBITDA at the time of payment;
- (d) the payment of salaries to directors and executives of any Group Company;
- (e) arms' length fees not to exceed the greater of (i) \$5,000,000 (or its equivalent in other currencies) and (ii) 10% of the then most recently reported Consolidated EBITDA in aggregate in any Financial Year for corporate finance advice relating to disposals or acquisitions permitted hereunder or approved by the Majority Lenders;
- (f) Restricted Payments provided that:
 - (i) no Default has occurred and is continuing when the payment is made or would arise as a result; and
 - (ii) Net First Lien Secured Leverage is equal to or less than 5.50:1 after giving pro forma effect to such payment;

- a payment permitted under (or made to facilitate a payment permitted under) the definition of Permitted Distribution or any other paragraph of this definition;
- (h) payments (other than dividends and/or other distributions in respect of the share capital and payments in respect of shareholder debt) in the ordinary course of business on arm's length terms (or better for the Group) as conducted with any affiliate of the Group;
- (i) payments used to purchase, redeem, retire, acquire, cancel or terminate equity interests of any Group Company or any parent company of the Parent, as the case may be, held by officers, directors or employees or former officers, directors or employees (or their transferees, estates or beneficiaries under their estates), upon their death, disability, retirement, severance or termination of employment or service; provided that the aggregate cash consideration paid for all such redemptions shall not exceed the greater of (x) \$10,000,000 (or its equivalent in other currencies) and (y) 2% of the then most recently reported Consolidated EBITDA during any calendar year (provided that 100% of the unused amount in respect of this clause (i) may be carried forward to any succeeding calendar years and utilized in such calendar years);
- (j) payments related to repurchases of equity interests deemed to occur upon the exercise of stock options (including stock options of parent companies) if the equity interests represent a portion of the exercise price thereof; and
- (k) payments by any Group Company used to purchase, redeem, retire, acquire, cancel or terminate equity interests in any of its Subsidiaries.

"Permitted Refinancing Indebtedness" has the meaning given to it in clause 2.3 (Additional Debt).

"Permitted Security" means:

- (a) any lien arising by operation of law and in the ordinary course of business;
- (b) any netting or set-off arrangement entered into by any member of the Group in the ordinary course
 of its banking arrangements for the purpose of netting debit and credit balances of member of the
 Group (including an Ancillary Facility which is an overdraft comprising more than one account);
- (c) any Security or Quasi-Security over or affecting any asset acquired by a member of the Group if:
 - (i) the Security or Quasi-Security was not created in contemplation of the acquisition of that asset by a member of the Group; and
 - (ii) the principal amount secured has not been increased in contemplation of or since the acquisition of that asset by a member of the Group;
- (d) any Security or Quasi-Security over or affecting any asset of any company which becomes a member of the Group, where the security or quasi-security is created prior to the date on which that company becomes a member of the Group if:
 - (i) the Security or Quasi-Security was not created in contemplation of the acquisition of that company; and
 - (ii) the principal amount secured has not increased in contemplation of or since the acquisition of that company;

- (e) any Security or Quasi-Security arising under any retention of title, hire purchase or conditional sale arrangement or arrangements having similar effect in respect of goods supplied to a member of the Group in the ordinary course of business and on the supplier's standard or usual terms and not arising as a result of any default or omission by any member of the Group;
- (f) any Security or Quasi-Security arising as a consequence of any finance lease permitted pursuant to paragraph (g) of the definition of "Permitted Financial Indebtedness";
- (g) any Security or Quasi-Security arising as a result of legal proceedings being contested by a member of the Group in good faith;
- (h) any Security or Quasi-Security over any rental deposits in respect of real estate leased or licensed by a member of the Group in respect of amounts representing not more than 24 months' rent or licence fee for that real estate;
- (i) any Security or Quasi-Security over documents of title and goods arising in the ordinary course of a documentary credit transaction entered into in the ordinary course of business;
- (j) any Security or Quasi-Security granted in respect of deferred duty or other customs arrangements or carriers', warehousemen's, mechanics', materialmen's repairmen's liens in the ordinary course of business or, if not in the ordinary course of business, where the outstanding aggregate value of the assets secured does not exceed at the then most recent time such Security or Quasi-Security is granted the greater of (i) \$15,750,000 (or its equivalent in other currencies) and (ii) 3.15% of the then most recently reported Consolidated EBITDA;
- (k) any Security or Quasi-Security over shares in joint ventures to secure obligations to the other joint venture partners which is required to be provided by the terms of the relevant joint venture agreement;
- (I) any Security or Quasi-Security granted in respect of the First Lien Finance Documents (including, any Additional First Lien Finance Documents, Permitted Refinancing Indebtedness and any Incremental Facilities, in each case, permitted under this agreement) and in respect of Second Lien Liabilities permitted to be incurred under this agreement (including, any Additional Second Lien Debt and Permitted Refinancing Indebtedness, in each case, permitted under this agreement) and/or otherwise in accordance with the Intercreditor Agreement;
- (m) any Security or Quasi-Security arising from Taxes not yet due and payable or from Taxes being contested in good faith by appropriate proceedings for which appropriate reserves have been established in accordance with Accounting Principles or to the extent the failure to establish such reserves would not reasonably be expected to result in a Material Adverse Effect;
- (n) any Security or Quasi-Security the outstanding principal amount of which does not exceed at the time such Security or Quasi-Security is granted the greater of (i) \$52,500,000 (or its equivalent in other currencies) and (ii) 10.50% of the then most recently reported Consolidated EBITDA;
- (o) any Security or Quasi-Security on any property or asset of any Group Company existing on the 2022
 Effective Date and, to the extent in excess of \$5,000,000 (or its equivalent in other currencies), set
 forth in Schedule 14 part 7;
- (p) any Quasi-Security (as defined in clause 27.10 (Negative Pledge)) to the extent such arrangements outstanding (measured solely when incurred, created or assumed) does not exceed of the greater of (i) \$100,000,000 (or its equivalent in other

currencies) and (ii) 20% of the then most recently reported Consolidated EBITDA; and

(q) any Security or Quasi Security given or arising under the Finance Documents.

"Permitted Share Issue" means an issue of:

- (a) ordinary shares by the Parent and which by their terms are not mandatorily redeemable;
- (b) shares by a Group Company (other than the Parent) to its immediate Parent Undertaking;
- (c) shares issued pursuant to a management or employee share scheme;
- (d) shares issued in connection with a Permitted Joint Venture or a Permitted Acquisition;
- (e) pursuant to a Permitted Transaction or to the extent constituting a Permitted Distribution or Permitted Disposal;
- (f) shares by Majority Owned Group Company issued to fund the operations of such Majority Owned Group Company and which by their terms are not mandatorily redeemable; or
- (g) shares issued on or prior to the 2022 Effective Date.

"Permitted Tax Residency Change" means any change in the tax residency of a Group Company that is not a Borrower to the United Kingdom;

"Permitted Transaction" means:

- any disposal required, Financial Indebtedness incurred, guarantee, indemnity or Security or Quasi-Security given, or other transaction arising, under the Finance Documents;
- (b) a reorganisation, merger, amalgamation, consolidation or transfer involving the business or assets of, or shares of (or other interests in), any member of the Group where all of the business, assets and shares of (or other interests in) the relevant member of the Group continue to be owned directly or indirectly by the Parent in the same or a greater percentage as prior to such reorganisation, save for:
 - (A) the shares of (or other interests in) any member of the Group which has been merged into another member of the Group or which has otherwise ceased to exist (including, for example, by way of the collapse of a solvent partnership or solvent winding up of a corporate entity) as a result of action otherwise permitted by this definition; or
 - (B) the business, assets and shares of (or other interests in) relevant members of the Group which cease to be owned:
 - (aa) as a result of a disposal or merger permitted under, but subject always to the terms of, this agreement; or
 - (bb) as a result of a cessation of business or solvent winding-up of a member of the Group (other than the Parent) in conjunction with a distribution of all or substantially all of its assets remaining after settlement of its liabilities to its immediate shareholder(s) or other persons directly holding partnership or other ownership interests in it; or

- (cc) as a result of a disposal of shares (or partnership or other ownership interests) in a member of the Group required to comply with applicable laws, provided that any such disposal is limited to the minimum amount required to comply with such applicable laws;
- (c) any transactions that were Permitted Transactions (as defined in this agreement immediately prior to the 2022 Effective Date) incurred prior to the 2022 Effective Date;
- (d) any reorganisation arising as a consequence of an obligation arising under this agreement;
- (e) any other re-organisation involving one or more members of the Group approved by the Majority Lenders:
- (f) a disposal of:
 - (i) Intellectual Property relating to the Business and related rights to exploit the same; and/or
 - (ii) any other Transfer Rights (as defined in the 100 Year Agreements),

in each case to FOAM or any of its Subsidiaries provided that the Group retains a licence or other right to use and exploit such Intellectual Property and related rights in connection with and for the purpose of the Business at least up to and including the latest Termination Date under this agreement;

- (g) the ASM Dissolution;
- a reorganisation, merger amalgamation, consolidation or transfer involving the business or assets of, (h) or shares of (or other interests in), the Parent; provided that (x)(a) Parent shall be the surviving entity thereof or (b) if the surviving entity is not Parent (such other person, the "Successor Parent"), (1) the Successor Parent shall be an entity organized or existing under the laws of the United States, any state or territory thereof or the District of Columbia or any Permitted Jurisdiction, (2) the Successor Parent shall expressly assume all the obligations of Parent under this agreement and the other Finance Documents pursuant to a supplement hereto or thereto in form reasonably satisfactory to the Facility Agent, (3) each other Obligor shall have by a supplement hereto confirmed that its guaranteed obligations under this agreement and the other Finance Documents shall apply to any Successor Parent's obligations, (3) the Successor Parent shall have delivered to the Facility Agent (i) a certificate of a financial officer stating that such merger or consolidation does not violate this agreement or any other Finance Document and (ii) if requested by the Facility Agent, an opinion of counsel to the effect that such merger or consolidation is in compliance with this agreement and covering such other matters as the Facility Agent may reasonably request (it being understood that if the foregoing are satisfied, the Successor Parent will succeed to, and be substituted for, the Parent under this agreement) and (4) the Successor Parent and any other Obligor or parent undertaking must give equivalent security over its assets and/or equity interests, to the extent required by the Agreed Security Principles and clause 30.8 (Further Security), (y) immediately after giving effect thereto, the Group shall be in compliance on a pro forma basis with Clause 26.2 (Financial Condition) as of the most recent Testing Period (whether or not at the time required to be tested) and no Default shall be continuing at the time of such transaction and (z) to the extent reasonably requested by the Facility Agent at least 10 Business Days prior to the consummation of such transaction, the Facility Agent shall have received at least three Business Days prior to such consummation all documentation and other

information in respect of the Parent required under applicable "know your customer" and anti-money laundering rules and regulations;

- (i) any Permitted Group Company Disposal; and
- (j) any waiver or cancellation or distribution to any Group Company or direct or indirect shareholder of the Parent of any Operating Expense Receivable.

provided that any such transaction shall not result in the transfer or assignment to an Unrestricted Subsidiary of any Material Asset.

"**Permitted Transaction Release**" means a Permitted Local Law Release, a Permitted Parent Share Release, a Permitted Investor Release and a Permitted Group Company Release;

"Prepayment Amount" has the meaning given to it in clause 12.2 (Disposal and Insurance Proceeds and Excess Cashflow);

"Prepayment Date" has the meaning given to it in clause 12.3 (Application of Mandatory Prepayments);

"Pricing Grid" means:

Net First Lien Secured Leverage	Commitment Fee Rate	Margin for Term Benchmark Term Facility A Loans and Term Benchmark RCF Loans	Margin for ABR Term Facility A Loans and ABR RCF Loans
>6.00:1.00	0.35%	2.25%	1.25%
$>5.00:1.00$ and $\leq 6.00:1.00$	0.30%	2.00%	1.00%
>4.00:1.00 and <u><</u> 5.00:1.00	0.25%	1.75%	0.75%
<u><</u> 4.00:1.00	0.25%	1.50%	0.50%

For the purposes of the Pricing Grid, changes in the Margin for Term Facility A Loans and RCF Loans and Commitment Fee Rate resulting from changes in the Net First Lien Secured Leverage shall become effective on the date (the "Adjustment Date") on which the Compliance Certificate is delivered to the Facility Agent pursuant to clause 25.2 (Provision and Contents of Compliance Certificate) and shall remain in effect until the next change to be effected pursuant to this paragraph. Notwithstanding the foregoing, if any Compliance Certificates referred to above are not delivered within the time periods specified in clause 25.2 (Provision and Contents of Compliance Certificate), then, until the date on which such Compliance Certificate is delivered, the highest rate set forth in each column of the Pricing Grid shall apply. In addition, at all times while an Event of Default shall have occurred and be continuing, the highest rate set forth in each column of the Pricing Grid shall apply. Each determination of the Net First Lien Secured Leverage pursuant to the Pricing Grid shall be made in a manner consistent with the determination thereof pursuant to clause 26.1 (Financial Definitions).

"Prime Rate" means the per annum rate of interest last quoted by The Wall Street Journal as the "Prime Rate" in the U.S. or, if The Wall Street Journal ceases to quote such rate, the per annum interest rate published by the Federal Reserve Board in Federal Reserve Statistical Release H.15 (519) (Selected Interest Rates) as the "bank prime loan" rate or, if such rate is no longer quoted therein, any similar rate quoted therein (as determined by the Facility Agent) or any similar release by the Federal Reserve Board (as determined by

the Facility Agent). Each change in the Prime Rate shall be effective from and including the date such change is publicly announced or quoted as being effective.

"Pro Rata Extension Offers" has the meaning given to it in clause 2.6 (Extension Offers);

"Public Lender" has the meaning given to that term in clause 29.7 (Disclosure of Information).

"Qualified Equity" means any equity interests of a Group Company other than Disqualified Equity.

"Qualifying IPO" means (x) the issuance and sale by the Parent or any parent entity of the Parent of its common equity interests in an underwritten primary public offering or (y) any transaction or event or series of related transactions or events whereby the equity interests of Parent or any parent entity of the Parent that are beneficially owned by Liberty Media Corporation (and any successor) are distributed, directly or indirectly (whether by redemption, dividend, share distribution, merger or otherwise) to the holders of one or more classes or series of the common stock of Liberty Media Corporation (or its successor) (such holders of one or more such classes or series, the "Holders") that are registered under Section 12(b) or 12(g) of the Exchange Act, or such equity interests are available to be acquired by Holders (including through any rights offering, exchange offer, exercise of subscription rights or other offer made available to Holders), whether voluntary or involuntary.

"Qualifying Lender" has the meaning given to that term in clause 18 (Tax Gross-Up and Indemnities);

"Quarter Date" means the last day of a Financial Quarter;

"Quarterly Information Pack" has the meaning given to it in clause 25.1 (Financial Statements);

"Quasi-Security" has the meaning given to that term in clause 27.10 (Negative Pledge);

"Quotation Day" means, in relation to any period or day for which an interest rate is to be determined:

- (a) (if the currency is euro) two TARGET Days before the first day of that period;
- (b) (for any Term Benchmark Loans that are in Dollars) two U.S. Government Securities Business Days before the first day of that period; or
- (c) (for any Daily Simple SOFR Loan that are in Dollars), five U.S. Government Securities Business Days prior to (i) if such SOFR Rate Day is a U.S. Government Securities Business Day, such SOFR Rate Day or (ii) if such SOFR Rate Day is not a U.S. Government Securities Business Day, the U.S. Government Securities Business Day immediately preceding such SOFR Rate Day,
- (d) (for any other currency), as determined by the Facility Agent and the Obligors' Agent in accordance with market practice in the Relevant Interbank Market;

"RCF Commitment" means:

- (a) in relation to an Original RCF Lender, the aggregate amount in the Base Currency set opposite its name in Part 3 of Schedule 1 and the amount of any other RCF Commitments transferred to it under this agreement;
- (b) in relation to any other Lender, the amount in the Base Currency of any RCF Commitment transferred to it under this agreement; and

(c) any Incremental RCF Commitments made available pursuant to clause 2.2 (Incremental Facilities) which are specified in the relevant Incremental Facility Commitment Letter to increase the commitments under Facility RCF,

to the extent not cancelled, reduced or transferred by it under this agreement;

"RCF Lender" means each Lender with an RCF Commitment;

"RCF Loan" means a loan made or to be made under Facility RCF or the principal amount outstanding for the time being of that loan;

"Receiver" means a receiver or receiver and manager or administrative receiver of the whole or any part of the Charged Property;

"Reference Time" with respect to any setting of the then-current Benchmark means (1) if such Benchmark is the Term SOFR Rate, 5:00 a.m. (Chicago time) on the day that is two U.S. Government Securities Business Days preceding the date of such setting, (2) if such Benchmark is EURIBOR, 11:00 a.m. Brussels time two TARGET Days preceding the date of such setting, (3) if the RFR for such Benchmark is Daily Simple SOFR, then four U.S. Government Securities Business Days prior to such setting or (4) if such Benchmark is none of the Term SOFR Rate or Daily Simple SOFR or EURIBOR, the time determined by the Facility Agent in its reasonable discretion with the Obligors' Agent.

"Refinanced Indebtedness" has the meaning given to it in clause 2.3(b) (Additional Debt);

"Refinancing Maturity Date" has the meaning given to it in clause 2.3(b) (Additional Debt);

"Refinancing Maturity Exclusion" has the meaning given to it in clause 2.3(b) (Additional Debt);

"Register" has the meaning given to that term in clause 29.2 (Conditions of Assignment or Transfer).

"Regulated Bank" means a bank that is (i) a U.S. depository institution the deposits of which are insured by the Federal Deposit Insurance Corporation; (ii) a corporation organized under section 25A of the U.S. Federal Reserve Act of 1913; (iii) a branch, agency or commercial lending company of a foreign bank operating pursuant to approval by and under the supervision of the Board under 12 CFR part 211; (iv) a non-U.S. branch of a foreign bank managed and controlled by a U.S. branch referred to in clause (iii); or (v) any other U.S. or non-U.S. depository institution or any branch, agency or similar office thereof supervised by a bank regulatory authority in any jurisdiction.

"Related Fund" in relation to a fund (the "First Fund"), means a fund which is managed or advised by the same investment manager or adviser as the First Fund or, if it is managed by a different investment manager or adviser, a fund whose investment manager or adviser is an Affiliate of the investment manager or adviser of the First Fund;

"Relevant Governmental Body" means (i) with respect to a Benchmark Replacement in respect of a Loan denominated in Dollars, the Federal Reserve Board and/or the NYFRB, the CME Term SOFR Administrator, as applicable, or a committee officially endorsed or convened by the Federal Reserve Board and/or the NYFRB or, in each case, any successor thereto, (ii) with respect to a Benchmark Replacement in respect of a Loan denominated in Euros, the European Central Bank, or a committee officially endorsed or convened by the European Central Bank or, in each case, any successor thereto, and (iii) with respect to a Benchmark Replacement in respect of a Loan denominated in any other currency, (a) the central bank for the currency in which such Benchmark Replacement is denominated or any central bank or other supervisor which is responsible for supervising either (1) such

Benchmark Replacement or (2) the administrator of such Benchmark Replacement or (b) any working group or committee officially endorsed or convened by (1) the central bank for the currency in which such Benchmark Replacement is denominated, (2) any central bank or other supervisor that is responsible for supervising either (A) such Benchmark Replacement or (B) the administrator of such Benchmark Replacement, (3) a group of those central banks or other supervisors or (4) the Financial Stability Board or any part thereof.

"Relevant Interbank Market" means in relation to euro, the European interbank market, in relation to Dollars, the New York interbank market and, in relation to any other currency, such other relevant interbank market for such currency;

"Relevant Jurisdiction" means, in relation to an Obligor:

- (a) its jurisdiction of incorporation;
- (b) any jurisdiction where any asset subject to or intended to be subject to the Transaction Security to be created by it is situated;
- (c) any jurisdiction where it conducts its business; and
- (d) the jurisdiction whose laws govern the perfection of any of the Transaction Security Documents entered into by it;

"Renewal Request" means a written notice delivered to the Facility Agent in accordance with clause 6.6 (Renewal of a Letter of Credit);

"Repayment Instalment" means an instalment for repayment referred to in clause 10.1 (Repayment of Term Loans);

"Repeating Representations" means each of the representations set out in clause 24.2 (Status) to clause 24.7 (Governing Law and Enforcement), clause 24.11 (No Default), clause 24.13 (Financial Statements) and clause 24.24 (Anti-Terrorism);

"Residual Entity" means an entity covered by the article 4.2 of the European Savings Directive 2003/48/EC, i.e. (i) that is not a legal person (except certain Swedish and Finnish entities which have legal personality but are nevertheless treated as not being separate legal persons); (ii) whose profits are not taxed under the general arrangements for business taxation; and (iii) which is not (or has not opted to be) a UCITS recognized in accordance with European Directive 85/611/EEC;

"Resignation Letter" means a letter substantially in the form set out in schedule 6 (Form of Resignation Letter);

"Restricted Party" means any person listed:

- (a) in the Annex to the Executive Order;
- (b) on the "Specially Designated Nationals and Blocked Persons" list maintained by OFAC; or
- (c) in any successor list to any of the foregoing;

"Restricted Payments" has the meaning given to that term in clause 27.15 (Dividends and Share Redemption).

"Restricted SLEC Company" has the meaning given to it in clause 27.30(e) (Termination Event Protection);

"Revolving Facility" means Facility RCF and/or any Incremental Revolving Facility;

"Revolving Facility Commitment" means a RCF Commitment or an Incremental RCF Commitment;

"Revolving Facility Loan" means a RCF Loan and/or Incremental Revolving Facility Loan or the principal amount outstanding for the time being of that Loan;

"Revolving Facility Utilisation" means a Revolving Facility Loan or a Letter of Credit;

"RFR Loan" means a Loan that bears interest at a rate based on the Daily Simple SOFR.

"Rollover Loan" means one or more Revolving Facility Loans under the same Revolving Facility:

- (a) made or to be made on the same day that:
 - (i) a maturing Revolving Facility Loan is due to be repaid; or
 - (ii) a demand by the Facility Agent pursuant to a drawing in respect of a Letter of Credit is due to be met:
- (b) the aggregate amount of which is equal to or less than the maturing Revolving Facility Loan or the relevant claim in respect of that Letter of Credit;
- (c) in the same currency as the maturing Revolving Facility Loan (unless it arose as a result of the operation of clause 8.2 (Unavailability of a Currency)) or the relevant claim in respect of that Letter of Credit; and
- (d) made or to be made to the same Borrower for the purpose of:
 - (i) refinancing that maturing Revolving Facility Loan; or
 - (ii) satisfying the relevant claim in respect of that Letter of Credit;

"Sanctioned Person" means any person:

- (a) designated on the list of Specially Designated Nationals and Blocked Persons maintained by OFAC, the Consolidated List of Financial Sanctions Targets or list of Investment Ban Targets, the Consolidated List of Persons, Groups and Entities Subject to EU Financial Sanctions maintained by the European Commission, the Consolidated List of Financial Sanctions Targets administered by HM Treasury in the United Kingdom, or any other similar list of targeted persons, entities, groups or bodies issued by the US, UK, UN, or EU (or any member state of the EU);
- (b) that is, or is part of, a government of any Sanctioned Territory;
- (c) incorporated or located within a Sanctioned Territory; or
- (d) otherwise targeted under any Economic Sanctions Law;
- (e) owned or controlled by any of the foregoing;

"Sanctioned Territory" means any country, state or territory:

(a) which is at the relevant time, subject to a general export, import, financial or investment restriction or embargo under any Economic Sanctions Law (including as of the 2022 Effective Date, but not limited to, the Crimea, Kherson and Zaporizhzhia regions of Ukraine, the so-called Donetsk People's Republic, the so-called Luhansk People's Republic, Cuba, Iran, North Korea, and Syria); or (b) to which it would be unlawful under or as a result of any such sanctions for a Lender to make loans to entities incorporated in such country, state or territory;

"Screen Rate" means:

- (a) in relation to a Term Benchmark Loan denominated in Dollars, the Term SOFR Reference Rate; and
- (b) in relation to a Term Benchmark Loan denominated in euros, the euro interbank offered rate administered by the European Money Markets Institute (or any other person which takes over the administration of that rate) for the relevant period displayed on page EURIBOR01 of the Thomson Reuters screen (or any replacement Thomson Reuters page which displays that rate) or on the appropriate page of such other information service which publishes that rate from time to time in place of Thomson Reuters as published at approximately 11:00 a.m. Brussels time two TARGET Days prior to the commencement of such Interest Period.

If such page or service ceases to be available, the Facility Agent may specify another page or service displaying the relevant rate after consultation with the Obligors' Agent;

"Second Lien Finance Document" has the meaning given to it in the Intercreditor Agreement;

"Second Lien Liabilities" has the meaning given to it in the Intercreditor Agreement;

"Secured Parties" means each Finance Party from time to time party to this agreement and any Receiver or Delegate;

"Security" means a mortgage, charge, pledge, lien or other security interest securing any obligation of any person or any other agreement or arrangement having a similar effect;

"Selection Notice" means a notice substantially in the form set out in part 2 of schedule 3 (Requests) given in accordance with clause 15 (Interest Periods) in relation to a Term Facility;

"Senior Distress Event" means a Distress Event (as defined in the Intercreditor Agreement) which arises as a result of action taken by one or more of the Finance Parties;

"Sixth Restatement Agreement" means the deed dated on or about 8 August 2014 pursuant to which (among other things) this agreement was amended and restated;

"Sixth Restatement Date" has the meaning given to it in the Sixth Restatement Agreement;

"Sixth Restatement English Law Debenture" means the English law governed debenture dated on or about the Sixth Restatement Date and delivered as a condition precedent to the Sixth Restatement Agreement;

"Sixth Restatement English Law Share Charge" means the English law governed share charge dated on or about the Sixth Restatement Date and delivered as a condition precedent to the Sixth Restatement Agreement;

"SLEC" means SLEC Holdings Limited, a company incorporated in Jersey with registered number 68316;

"SLEC Group" means SLEC and its Subsidiaries for the time being;

- "SLEC Share Charge" means the specific security agreement dated on or about the 2022 Effective Date between Alpha Prema UK Limited and the Security Agent in relation to shares in SLEC Holdings Limited;
- "SOFR" means a rate equal to the secured overnight financing rate as administered by the SOFR Administrator.
- "SOFR Administrator" means the NYFRB (or a successor administrator of the secured overnight financing rate).
- "SOFR Administrator's Website" means the NYFRB's website, currently at http://www.newyorkfed.org, or any successor source for the secured overnight financing rate identified as such by the SOFR Administrator from time to time.
- "SOFR Determination Date" has the meaning given to it in the definition of "Daily Simple SOFR".
- "SOFR Rate Day" has the meaning given to it in the definition of "Daily Simple SOFR".
- "Specified Time" means a time determined in accordance with schedule 9 (Timetables);
- **"Subsidiary"** means a subsidiary within the meaning of section 1159 of the Companies Act 2006 and a subsidiary undertaking within the meaning of section 1162 of the Companies Act 2006;
- "Successor Parent" has the meaning given to that term in the definition of Permitted Transaction.
- "Super Majority Lenders" means (subject to clause 40 (Amendments and waivers)) a Lender or Lenders whose Commitments are in aggregate equal to or more than 85 per cent of the Total Commitments (or, if the Total Commitments have been reduced to zero, aggregated equal to or more than 85 per cent of the Total Commitments immediately prior to that reduction);
- "Swiss Withholding Tax" means the tax imposed based on the Swiss Federal Act on Withholding Tax of 13 October 1965 (Bundesgesetz über die Verrechnungssteuer), as amended from time to time together with the related ordinances, regulations and guidelines;
- "S&P" means Standard & Poor's Rating Services, a division of The McGraw-Hill Companies, Inc. or any successor to its rating business;
- "TARGET" means Trans-European Automated Real-time Gross Settlement Express Transfer payment system;
- "TARGET Day" means any day on which TARGET is open for the settlement of payments in euro;
- "Tax" means any tax, levy, impost, duty or other charge or withholding of a similar nature (including any penalty or interest payable in connection with any failure to pay or any delay in paying any of the same);
- "Term" means each period determined under this agreement for which the Issuing Bank is under a liability under a Letter of Credit;
- "Term Benchmark" when used in reference to any Loan, refers to whether such Loan is bearing interest at a rate determined by reference to the Term SOFR Rate (other than pursuant to clause (c) of the definition of Alternate Base Rate) or EURIBOR.

"Term Facility" means Term Facility A, Term Facility B and each Incremental Term Facility;

"Term Facility A" means the Incremental Term Facility redesignated as such on the 2022 Effective Date pursuant to the 2022 Amendment Agreement;

"Term Facility A Commitment" means:

- (a) in relation to a Lender on the 2022 Effective Date, the aggregate amount in USD of:
 - (i) its Term Facility A Commitments redesignated as such on the 2022 Effective Date pursuant to the 2022 Amendment Agreement;
 - (ii) any Term Facility A Commitment transferred to it under this agreement; and
 - (iii) any Incremental Term Facility Commitments made available by that Lender pursuant to clause 2.2 (Incremental Facilities) which are specified in the relevant Incremental Facility Commitment Letter to increase the commitments under Term Facility A;
- (b) in relation to any other Lender, the aggregate amount in Dollars of:
 - (i) any Term Facility A Commitment transferred to it under this agreement; and
 - (ii) any Incremental Term Facility Commitments made available by that Lender pursuant to clause 2.2 (Incremental Facilities) which are specified in the relevant Incremental Facility Commitment Letter to increase the commitments under the then existing Term Facility A,

in each case, to the extent not cancelled, reduced or transferred by it under this agreement;

"Term Facility A Loan" means a loan made or to be made under Term Facility A or the principal amount outstanding for the time being of that loan and includes each loan redesignated as such on the 2022 Effective Date pursuant to the 2022 Amendment Agreement;

"Term Facility B" means the Base Currency term loan facility redesignated as such on the 2022 Effective Date pursuant to 2022 Amendment Agreement and as amended on the 2023 Effective Date pursuant to the 2023 Term Facility B Refinancing Documents and as amended on the 2024 Effective Date pursuant to the 2024 Incremental Facility Commitment Letter and the 2024 Amendment Agreement;

"Term Facility B Commitment" means:

- (a) in relation to a Lender on the 2024 Effective Date, the aggregate amount in USD of:
 - (i) its Term Facility B Commitments established pursuant to the 2024 Incremental Facility Commitment Letter;
 - (ii) any Term Facility B Commitment transferred to it under this agreement; and
 - (iii) any Incremental Term Facility Commitments made available by that Lender pursuant to clause 2.2 (Incremental Facilities) which are specified in the relevant Incremental Facility Commitment Letter to increase the commitments under Term Facility B other than pursuant to the 2024 Incremental Facility Commitment Letter;
- (b) in relation to any other Lender, the aggregate amount in USD of:
 - (i) any Term Facility B Commitment transferred to it under this agreement; and

(ii) any Incremental Term Facility Commitments made available by that Lender pursuant to clause 2.2 (Incremental Facilities) which are specified in the relevant Incremental Facility Commitment Letter to increase the commitments under the then existing Term Facility B,

in each case, to the extent not cancelled, reduced or transferred by it under this agreement;

"Term Facility B Loan" means a loan made or to be made under Term Facility B or the principal amount outstanding for the time being of that loan and includes each loan made pursuant to the 2024 Incremental Facility Commitment Letter;

"Term Loan" means a Term Facility A Loan, a Term Facility B Loan or any Incremental Facility Term Loan;

"Term SOFR Determination Day" has the meaning given to it in the definition of Term SOFR Reference Rate.

"Term SOFR Rate" means, with respect to any Term Benchmark Loan denominated in Dollars and for any tenor comparable to the applicable Interest Period, the Term SOFR Reference Rate at approximately 5:00 a.m., Chicago time, two U.S. Government Securities Business Days prior to the commencement of such tenor comparable to the applicable Interest Period, as such rate is published by the CME Term SOFR Administrator; provided that if the Term SOFR Rate as so determined would be less than the applicable Floor, such rate shall be deemed to be equal to the applicable Floor for the purposes of this agreement.

"Term SOFR Reference Rate" means, for any day and time (such day, the "Term SOFR Determination Day"), with respect to any Term Benchmark Loan denominated in Dollars and for any tenor comparable to the applicable Interest Period, the rate per annum published by the CME Term SOFR Administrator and identified by the Facility Agent as the forward-looking term rate based on SOFR. If by 5:00 pm (New York City time) on such Term SOFR Determination Day, the "Term SOFR Reference Rate" for the applicable tenor has not been published by the CME Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Rate has not occurred, then the Term SOFR Reference Rate for such Term SOFR Determination Day will be the Term SOFR Reference Rate as published in respect of the first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate was published by the CME Term SOFR Administrator, so long as such first preceding Business Day is not more than three (3) Business Days prior to such Term SOFR Determination Day.

"Termination Date" means:

- (a) in relation to each Term Facility A Loan, September 30, 2029;
- (b) in relation to each Term Facility B Loan, September 30, 2031;
- (c) in relation to Facility RCF and each RCF Loan, September 30, 2029; and
- (d) in relation to each Incremental Facility the termination date agreed between the Borrower and the relevant Lenders in the relevant Incremental Facility Commitment Letter;

"Testing Date" has the meaning given to that term in clause 26.1 (Financial Definitions);

"Testing Period" has the meaning given to that term in clause 26.1 (Financial Definitions);

"Total Commitments" means the aggregate of the Total Facility A Commitments, the Total Facility B Commitments, the Total RCF Commitments and the Total Incremental Facility Commitments;

"Total Facility A Commitments" means the aggregate of the Term Facility A Commitments;

"Total Facility B Commitments" means the aggregate of the Term Facility B Commitments;

"Total Incremental Facility Commitments" means the aggregate of the Incremental Facility Commitments, being zero as at the 2022 Effective Date;

"Total Incremental RCF Commitments" means the aggregate of the Incremental RCF Commitments;

"Total RCF Commitments" means the aggregate of the RCF Commitments, being \$500,000,000 as at the 2022 Effective Date;

"Total Revolving Facility Commitments" means the Total RCF Commitments and the aggregate of the Total Incremental RCF Commitments in respect of each Incremental Revolving Facility constituted pursuant to clause 2.2 (Incremental Facilities):

"Transaction Documents" means the First Lien Finance Documents, Second Lien Finance Documents and the Equity Documents;

"Transaction Security" means the Security created or expressed to be created in favour of the Security Agent pursuant to the Transaction Security Documents;

"Transaction Security Documents" means (in each case, subject to the proviso to this term as defined, and set out, in the Intercreditor Agreement):

- (a) each of the transaction security documents listed in part 4A of Schedule 2;
- (b) each of the transaction security documents set out in part 4B of schedule 2; and
- (c) any other document entered into by any shareholder of the Parent creating or expressed to create any Security over all or any part of the share capital of the Parent in respect of the obligations of any of the Obligors under any of the Finance Documents; and
- (d) any other document entered into by any Obligor creating or expressed to create any Security over all or any part of its assets in respect of the obligations of any of the Obligors under any of the Finance Documents:

"Transfer Certificate" means a certificate substantially in the form set out in schedule 4 (Form of Transfer Certificate) or any other form agreed between the Facility Agent and Obligors' Agent;

"Transfer Date" means, in relation to a transfer, the later of:

- (a) the proposed Transfer Date specified in the Transfer Certificate; and
- (b) the date on which the Facility Agent executes the Transfer Certificate;

"Transferee Parent" has the meaning given to that term in the definition of LMC.

"Transferee Person" has the meaning given to that term in the definition of LMC.

"Transformative Acquisition" shall mean any acquisition by the Group that is either (a) not permitted by the terms of the Finance Documents immediately prior to the consummation of such acquisition or (b) if permitted by the terms of the Finance Documents immediately prior to the consummation of such acquisition, would not provide the Group

with adequate flexibility under the Finance Documents for the continuation and/or expansion of their combined operations following such consummation, as reasonably determined by the Obligors' Agent acting in good faith.

"Treasury Transactions" means any derivative transaction entered into in connection with protection against or benefit from fluctuation in any rate or price;

"Type", when used in reference to any Loan or Utilisation, refers to whether the rate of interest on such Loan, or on the Loans comprising such Utilisation, is determined by reference to the Term SOFR Rate, the EURIBOR, the Daily Simple SOFR or the Alternate Base Rate.

"U.S. Government Securities Business Day" means any day except for (i) a Saturday, (ii) a Sunday or (iii) a day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States government securities.

"**UKEF Facility**" means any term loan facility provided in connection with an Export Development Guarantee provided by His Britannic Majesty's Secretary of State acting through the Export Credits Guarantee Department (operating as UK Export Finance) (or any successor scheme provided by any successor entity) and including both the relevant facility benefitting from such guarantee and the relevant facility provided in conjunction therewith that does not benefit from such guarantee.

"Umbrella Agreement" means the agreement dated 24 April 2001 between FIA and SLEC pursuant to which, among other things, the parties agreed to enter into the other 100 Year Agreements;

"Unadjusted Benchmark Replacement" means the applicable Benchmark Replacement excluding the related Benchmark Replacement Adjustment.

"Undertaking" has the meaning given to it in section 1161 of the Companies Act 2006 and shall include limited partnerships;

"Unpaid Sum" means any sum due and payable but unpaid by an Obligor under the Finance Documents;

"Unrestricted Subsidiary" means (a) any Subsidiary of the Parent that is designated as an Unrestricted Subsidiary by the Obligors' Agent after the 2022 Effective Date in a written notice to the Facility Agent and (b) any Subsidiary of any Subsidiary described in clause (a) above, provided, that, in each case, (i) all investments (other than guarantees) and loans in such Unrestricted Subsidiary at the time of designation are permitted in accordance with the relevant requirements of clause 27.6 (Acquisitions and Investments) and clause 27.13 (Permitted Loans) (it being understood that the aggregate fair market value of all outstanding investments and loans then owned and made by the Group in the Subsidiary so designated shall be deemed to be an investment or loan made as of the time of such designation), (ii) such Subsidiary does not own any Material Asset, (iii) at the time of such designation, no Event of Default shall have occurred and be continuing or would result therefrom and (iv) at the time of such designation, such Subsidiary shall not own any Equity Interests of, or own or hold any lien on any property of, the Group. It is understood that Unrestricted Subsidiaries shall be disregarded for the purposes of any calculation pursuant to this agreement relating to financial matters with respect to the Group and will not be subject to any of the restrictions, representation and warranties, affirmative or negative covenants or Event of Default provisions of this agreement or the Finance Documents.

If Obligors' Agent has designated a Subsidiary as an Unrestricted Subsidiary, Obligors' Agent may revoke such designation pursuant to a written notice to the Facility Agent so long as,

after giving pro forma effect to such revocation (x) no Event of Default shall be in existence and (y) all Security and Financial Indebtedness of such Unrestricted Subsidiary outstanding immediately following such revocation would, if incurred at such time, have been permitted to be incurred for all purposes of this agreement. Any revocation described in the preceding sentences is referred to herein as a "Revocation". Upon any Revocation, such Unrestricted Subsidiary shall constitute a member of the Group for all purposes of this agreement and shall comply with Clause 30 (Changes to the Obligors) if such Subsidiary is a Material Company following such Revocation. In addition, the designation of any Unrestricted Subsidiary as a member of the Group after the 2022 Effective Date shall constitute a return on any investment or loan made by the applicable Group Company in Unrestricted Subsidiaries pursuant to the preceding sentence in an amount equal to the fair market value at the date of such designation of such Group Companies' investment or loan in such subsidiary.

"US" means the United States of America;

"US Tax Obligor" means:

- (a) a Borrower which is resident for tax purposes in the US; or
- (b) an Obligor some or all of whose payments under the Finance Documents are from sources within the US for US federal income tax purposes;

"Utilisation" means a Loan or a Letter of Credit;

"Utilisation Date" means the date on which a Utilisation is made;

"Utilisation Request" means a notice substantially in the relevant form set out in part 1 of schedule 3 (Requests); and

"VAT" means value added tax as provided for in the Value Added Tax Act 1994, or any tax imposed in compliance with the Council Directive of 28 November 2006 on the common system of value added tax (EC Directive 2006/112) and any other tax of a similar nature.

1.2 Construction

- (a) Unless a contrary indication appears, a reference in this agreement to:
 - (i) any "Arranger", the "Facility Agent", any "Hedge Counterparty", any "Issuing Bank", any "Lender", any "Obligor", any "Party", any "Secured Party", the "Security Agent", any "Finance Party" or any other person shall be construed so as to include its successors in title, permitted assigns and permitted transferees and, in the case of the Security Agent, any person for the time being appointed as Security Agent or Security Agents in accordance with the Finance Documents;
 - (ii) "Affiliate" in relation to any member of the NatWest Group, the term "Affiliate" shall not include:
 - the UK government or any member or instrumentality thereof, including Her Majesty's Treasury and UK Financial Investments Limited (or any directors, officers, employees or entities thereof); or
 - (B) any persons or entities controlled by or under common control with the UK government or any member or instrumentality thereof (including Her Majesty's Treasury and UK Financial Investments Limited) and which are not part of NatWest Group plc and its subsidiaries or subsidiary undertakings;

- For the purposes of this definition, "NatWest Group" means NatWest Group plc and its subsidiaries and subsidiary undertakings.
- (iii) a document in "agreed form" is a document which is previously agreed in writing by or on behalf of Obligors' Agent and the Facility Agent or, if not so agreed, is in the form specified by the Facility Agent;
- (iv) "assets" includes present and future properties, revenues and rights of every description;
- (v) "guarantee" means (other than in clause 23 (Guarantee and Indemnity)) any guarantee, letter of credit, bond, indemnity or similar assurance against loss, or any obligation, direct or indirect, actual or contingent, to purchase or assume any indebtedness of any person or to make an investment in or loan to any person or to purchase assets of any person where, in each case, such obligation is assumed in order to maintain or assist the ability of such person to meet its indebtedness;
- (vi) "including" means including without limitation and "includes" and "included" shall be construed accordingly;
- (vii) "indebtedness" includes any obligation (whether incurred as principal or as surety) for the payment or repayment of money, whether present or future, actual or contingent;
- (viii) a Lender's "participation" in relation to a Letter of Credit, shall be construed as a reference to the relevant amount that is or may be payable by a Lender in relation to that Letter of Credit;
- (ix) a "person" or "Person" includes any person, firm, company, corporation, limited liability company, trust, association, government, state or agency of a state or any association, trust, partnership (whether or not having separate legal personality) or other entity;
- a "regulated stock exchange" shall include, without limitation, the NASDAQ Stock Exchange
 and any successor or other similar stock exchange and any other electronic securities
 exchange platform;
- (xi) a "regulation" includes any regulation, rule, official directive, request or guideline (whether
 or not having the force of law) of any governmental, intergovernmental or supranational body,
 agency, department or regulatory, self-regulatory or other authority or organisation;
- (xii) a "Finance Document" or a "Transaction Document" or any other agreement or instrument is a reference to that Finance Document or Transaction Document or other agreement or instrument as amended, novated, supplemented, extended or restated (however fundamentally) (excluding any amendment, novation, supplement, extension or restatement made contrary to any provision of the Finance Documents);
- (xiii) a provision of law is a reference to that provision as amended or re-enacted;
- (xiv) a time of day is a reference to London time;
- (xv) an Obligor of a particular nationality will be construed as a reference to an Obligor incorporated in the corresponding jurisdiction;
- (xvi) subject to Clause 1.4(f), a reference to the most recently reported Consolidated EBITDA at any time means Consolidated EBITDA for the then

- most recent 4 consecutive Financial Quarters (as defined in clause 26.1 (Financial Definitions) for which audited financial statements or Quarterly Information Packs have been provided pursuant to clause 25.1 (Financial Statements); and
- (xvii) any reference in this agreement to "ordinary course of business" shall be interpreted in accordance with the laws of the state of New York.
- (b) Clause and schedule headings are for ease of reference only.
- (c) Unless a contrary indication appears, a term used in any other Finance Document or in any notice given under or in connection with any Finance Document has the same meaning in that Finance Document or notice as in this agreement.
- (d) A Borrower providing "cash cover" for a Letter of Credit or an Ancillary Facility means a Borrower paying an amount in the currency of the Letter of Credit (or, as the case may be, Ancillary Facility) to an interest-bearing account in the name of the Borrower and the following conditions being met:
 - the account is with the Facility Agent (if the cash cover is to be provided for all the Lenders) or with a Lender or Ancillary Lender (if the cash cover is to be provided for that Lender or Ancillary Lender);
 - (ii) until no amount is or may be outstanding under that Letter of Credit or Ancillary Facility, withdrawals from the account may only be made to pay a Finance Party amounts due and payable to it under this agreement in respect of that Letter of Credit or Ancillary Facility; and
 - (iii) the Borrower has executed a security document over that account, in form and substance satisfactory to the Facility Agent or the Lender or Ancillary Lender with which that account is held, creating a first ranking security interest over that account.
- (e) A Default (other than an Event of Default) is "continuing" if it has not been remedied or waived and an Event of Default is "continuing" if it has not been remedied or waived.
- (f) Any lease entered into by a Group Company which in accordance with the Accounting Principles as applied as at the Sixth Restatement Date would be an operating lease shall not be construed to be a finance lease or capital lease for the purpose of the Finance Documents as a result of the change anticipated as of the date of the Sixth Restatement Date in IFRS to eliminate the distinction between operating and capital leases in the accounting treatment of leases.
- (g) A Borrower "repaying" or "prepaying" a Letter of Credit or Ancillary Outstandings means:
 - that Borrower providing cash cover for that Letter of Credit or in respect of the Ancillary Outstandings;
 - the maximum amount payable under the Letter of Credit or Ancillary Facility being reduced or cancelled in accordance with its terms; or
 - (iii) the Issuing Bank or Ancillary Lender being satisfied that it has no further liability under that Letter of Credit or Ancillary Facility,

and the amount by which a Letter of Credit is, or Ancillary Outstandings are, repaid or prepaid under paragraphs (i) and (ii) above is the amount of the relevant cash cover, reduction or cancellation.

- (h) An amount borrowed includes any amount utilised by way of Letter of Credit or under an Ancillary Facility.
- (i) A Lender funding its participation in a Utilisation includes a Lender participating in a Letter of Credit.
- (j) Amounts outstanding under this agreement include amounts outstanding under or in respect of any Letter of Credit.
- (k) An outstanding amount of a Letter of Credit at any time is the maximum amount that is or may be payable by the relevant Borrower in respect of that Letter of Credit at that time.
- (I) A Borrower's obligation on Utilisations becoming "due and payable" includes the Borrower repaying any Letter of Credit in accordance with paragraph (q) above.
- (m) Where any provisions of a Finance Document (other than clause 26 (Financial Definitions and Financial Covenant) or as otherwise specified) requires a Commitment or Loan not in the Base Currency to be aggregated or compared with a Commitment or Loan in the Base Currency, the non-Base Currency Commitment or Loan shall be notionally converted into the Base Currency at the Facility Agent's Spot Rate of Exchange on the date of calculation.
- (n) In this agreement where it relates to a Luxembourg Obligor, a reference to:
 - (i) a "liquidator", "receiver", "administrative receiver", "administrator", "compulsory manager" or similar officer includes any
 - (A) juge délégué, mandataire de justice or administrateur provisoire appointed under the law of 7 August 2023 on preservation of businesses and modernisation of bankruptcy law:
 - (B) juge-commissaire or insolvency receiver (curateur) appointed under the Luxembourg Commercial Code;
 - (C) liquidateur appointed under Articles 1100-1 to 1100-15 (inclusive) of the Luxembourg act dated 10 August 1915 on commercial companies, as amended; and
 - (D) *juge-commissaire* or *liquidateur* appointed under Articles 1200-1 and 1200-3 of the Luxembourg act dated 10 *August* 1915 on commercial companies, as amended

or similar officer pursuant to any insolvency or similar proceedings;

- (ii) a "winding-up", "administration", "bankruptcy", "suspension of payments", "moratorium of any indebtedness", "reorganisation" (by way of voluntary arrangement, scheme of arrangement or otherwise), "insolvency" or "dissolution" includes, without limitation, bankruptcy (faillite), liquidation, judicial reorganisation (réorganisation judiciaire) (including for the avoidance of doubt any sursis and reorganisation by amicable agreement (réorganisation par accord amiable)), administrative dissolution without liquidation (dissolution administrative sans liquidation), reprieve from payment (sursis de paiement) general settlement with creditors (including any amicable agreement (accord amiable)) and judicial winding-up or liquidation; and
- (iii) commencing negotiations with one or more of its creditors (excluding any Finance Party in its capacity as such) with a view to rescheduling any of its

- indebtedness, includes any negotiations conducted in order to reach an amicable agreement (accord amiable);
- (iv) a person being unable to pay its debts includes that person being in a state of suspension of payments (cessation de paiements).
- (o) Loans may be classified and referred to by Class (e.g., a "Revolving Facility Loan") or by Type (e.g., a "Term Benchmark Loan" or a "RFR Loan") or by Class and Type (e.g., a "Term Benchmark Revolving Facility Loan" or an "RFR Revolving Facility Loan").
- (p) In connection with any action being taken in connection with a Limited Condition Transaction (including any contemplated incurrence or assumption of Financial Indebtedness (including Incremental Facility or Additional Debt) in connection therewith), for purposes of:
 - determining compliance with any provision of this agreement which requires the calculation of any financial ratio (including the Net First Lien Secured Leverage, Net Total Secured Leverage or Net Total Leverage);
 - determining the accuracy of representations and warranties and/or whether a Default or Event
 of Default shall have occurred and be continuing (or any subset of Defaults or Events of
 Default); or
 - (iii) testing availability under baskets set forth in this agreement (including baskets measured as a percentage of Consolidated EBITDA);

in each case, at the option of Obligors' Agent (such election to exercise such option in connection with any Limited Condition Transaction, an "LCA Election"), the date of determination of whether any such action is permitted hereunder, shall be deemed to be the date the definitive agreements with respect to such Limited Condition Transaction are entered into or the date of delivery of the relevant notices, if any (the "LCA Test Date"), and if, on or after giving pro forma effect to the Limited Condition Transaction and the other transactions to be entered into in connection therewith (including any incurrence of Financial Indebtedness or Security and the use of proceeds thereof) as if they had occurred at the beginning of the most recent four Financial Quarters for which financial statements have been delivered pursuant to clause 25.1 (Financial Statements) ending prior to the LCA Test Date, the applicable Group Company could have taken such action on the relevant LCA Test Date in compliance with such ratio or basket, such ratio or basket shall be deemed to have been complied with.

For the avoidance of doubt, if the Obligors' Agent has made an LCA Election and any of the ratios or baskets for which compliance was determined or tested as of the LCA Test Date are exceeded as a result of fluctuations in any such ratio or basket, including due to fluctuations in exchange rates or in Consolidated EBITDA or the person subject to such Limited Condition Transaction, at or prior to the consummation of the relevant transaction or action, such baskets or ratios will not be deemed to have been exceeded as a result of such fluctuations; provided, however, if any ratios improve or baskets increase as a result of such fluctuations, such improved ratios or baskets may be utilized.

If Obligors' Agent has made an LCA Election for any Limited Condition Transaction, then, in connection with any subsequent calculation of the ratios or baskets on or following the relevant LCA Test Date and prior to the earlier of (i) the date on which such Limited Condition Transaction is consummated or (ii) the date that the definitive agreement for such Limited Condition Transaction is terminated or expires without consummation of such Limited Condition Transaction, any such ratio or basket shall

be calculated on a pro forma basis assuming such Limited Condition Transaction and other transactions in connection therewith (including any incurrence of Financial Indebtedness or Security and the use of proceeds thereof) have been consummated; provided that if such Limited Condition Transaction is terminated or expires without consummation of such Limited Condition Transaction, any action taken where compliance of such ratios or baskets was determined or tested assuming such Limited Condition Transaction and other transaction in connection therewith have been consummated will not be deemed to have been exceeded as a result of failure to consummate such Limited Condition Transaction and other transactions in connection therewith.

1.3 Third Party Rights

- (a) Unless expressly provided to the contrary in a Finance Document, a person who is not a Party has no right under the Contracts (Rights of Third Parties) Act 1999 (the **"Third Parties Act"**) to enforce or enjoy the benefit of any term of this agreement.
- (b) Notwithstanding any term of any Finance Document, the consent of any person who is not a Party is not required to rescind or vary this agreement at any time.

1.4 **Certain Calculations**

- (a) Notwithstanding anything in this agreement or any Finance Document to the contrary, with respect to any amounts incurred or transactions entered into (or consummated) in reliance on a provision of this agreement that does not require compliance with a financial ratio or test (including, without limitation, the Net First Lien Secured Leverage, Net Total Leverage and Net Total Secured Leverage) (any such amounts, the "Fixed Amounts"), substantially concurrently with any amounts incurred or transactions entered into (or consummated) in reliance on a provision of this agreement that requires compliance with any such financial ratio or test (any such amounts, the "Incurrence-Based Amounts"), it is understood and agreed that any Fixed Amount shall be disregarded in the calculation of the financial ratio or test applicable to the relevant Incurrence-Based Amount in connection with such substantially concurrent incurrence.
- For purposes of determining compliance at any time with Clause 27 (General Undertakings), in the event that any transaction meets the criteria of one or more of the categories of a Permitted Transaction, Permitted Acquisition, Permitted Loan, Permitted Guarantee, Permitted Financial Indebtedness, Permitted Joint Venture, Permitted Share Issue, Permitted Security, Permitted Disposal, Permitted Distribution, Permitted Payment, or Permitted Intercompany Debt Disposal, the Obligors' Agent, in its sole discretion, may, from time to time, classify or reclassify such transaction or item (or portion thereof) and will only be required to include the amount and type of such transaction (or portion thereof) in any such classified or reclassified category. It is understood and agreed that any Permitted Transaction, Permitted Acquisition, Permitted Loan, Permitted Guarantee, Permitted Financial Indebtedness, Permitted Joint Venture, Permitted Share Issue, Permitted Security, Permitted Disposal, Permitted Distribution, Permitted Payment, or Permitted Intercompany Debt Disposal need not be permitted solely by reference to one category of Permitted Transaction, Permitted Acquisition, Permitted Loan, Permitted Guarantee, Permitted Financial Indebtedness, Permitted Joint Venture, Permitted Share Issue, Permitted Security, Permitted Disposal, Permitted Distribution, Permitted Payment, or Permitted Intercompany Debt Disposal, respectively, but may instead be permitted in part under any combination thereof. For the avoidance of doubt, a transaction or item that is permitted by one category of Permitted Transaction, Permitted Acquisition, Permitted Loan, Permitted Guarantee, Permitted Financial Indebtedness, Permitted Joint Venture, Permitted Share Issue, Permitted Security, Permitted Disposal, Permitted Distribution, Permitted Payment, or

Permitted Intercompany Debt Disposal, respectively, shall not be counted against any other category of Permitted Transaction, Permitted Acquisition, Permitted Loan, Permitted Guarantee, Permitted Financial Indebtedness, Permitted Joint Venture, Permitted Share Issue, Permitted Security, Permitted Disposal, Permitted Distribution, Permitted Payment, or Permitted Intercompany Debt Disposal, respectively (i.e., each category shall be independent of each other category).

- Notwithstanding anything herein the contrary, the baskets set forth in Clause 27 (General (c) Undertakings) of this agreement (including any related definition and any related clauses, including, but not limited to, clause 2.2 (Incremental Facilities) and clause 2.3 (Additional Debt)) shall only be tested solely at the initial time of consummation of the relevant transaction or action utilizing any of such baskets and, for the avoidance of doubt, if any of such baskets (including any related definition and any related clauses) would subsequently be exceeded (including as a result of fluctuations to Consolidated EBITDA after the last time such baskets (including any related definition and any related clauses) were calculated or due to an increase in outstanding amounts following the consummation of the relevant transaction or action) for any purpose under Clause 27 (General Undertakings) of this agreement (including any related definition and any related clauses, including, but not limited to, clause 2.2 (Incremental Facilities) and clause 2.3 (Additional Debt)), such baskets (including any related definition and any related clauses, including, but not limited to, clause 2.2 (Incremental Facilities) and clause 2.3 (Additional Debt)) will not be deemed to have been exceeded (i.e., incurrence based negative covenants); provided that, at the election of the Obligors' Agent, the date of determination of whether any action made in connection with the Dorna Distribution (giving pro forma effect to such Dorna Distribution) is permitted under Clause 27 shall be deemed to be the 2024 Effective Date. For the avoidance of doubt, any reference to "outstanding" under Clause 27 (General Undertakings) of this agreement (including any related definition and any related clauses, including, but not limited to, clause 2.2 (Incremental Facilities) and clause 2.3 (Additional Debt)) shall not be implied to require the monitoring of any such amounts (i.e., no basket shall be maintenance based). If any Financial Indebtedness or Security or Quasi-Security securing Financial Indebtedness are incurred to refinance any existing Financial Indebtedness or Security or Quasi-Security securing Financial Indebtedness, in each case, initially incurred in reliance on a basket measured by reference to a percentage of Consolidated EBITDA at the time of incurrence, such refinancing would cause the percentage of Consolidated EBITDA restriction to be exceeded if calculated based on the Consolidated EBITDA at the time of incurrence of such refinancing, such percentage of Consolidated EBITDA restriction shall not be deemed to be exceeded so long as the principal amount of such Financial Indebtedness or such Financial Indebtedness secured by such Security or Quasi-Security, as applicable, does not exceed the principal amount (or accreted value or fair value, if applicable) of the relevant existing Financial Indebtedness or Financial Indebtedness secured by such Security or Quasi-Security, as applicable, being refinanced, plus Financial Indebtedness incurred to pay premiums, defeasance costs and fees and expenses in connection.
- (d) In connection with the incurrence of revolving loan Financial Indebtedness or any commitment or other transaction relating to the incurrence or issuance of Financial Indebtedness or the granting of any security to secure such Financial Indebtedness, the Obligors' Agent may designate such incurrence and the granting of any security therefor as having occurred in full on the date of first incurrence of such revolving loan Financial Indebtedness or commitment or intention to consummate such transaction (such date, the "Deemed Date"), and any related subsequent actual incurrence and granting of such security therefor will be deemed for all purposes under this agreement to have been incurred and granted on such Deemed Date; provided that, for purposes of calculating any ratio or usage of any baskets hereunder following the Deemed Date, only the outstanding portion of revolving loan

Financial Indebtedness shall be considered for purposes of calculating any leverage ratio or usage of any baskets hereunder (e.g., unused commitments under such revolving Financial Indebtedness shall not be considered outstanding for purposes of this agreement or any Finance Document).

- (e) Notwithstanding anything herein to the contrary, all usages of fixed dollar baskets under this agreement (including baskets measured as a percentage of Consolidated EBTIDA) prior to the 2024 Effective Date shall be disregarded as of the 2024 Effective Date and deemed permitted under this agreement and shall not be counted against any fixed dollar basket (including baskets measured as a percentage of Consolidated EBTIDA) that are set forth in this agreement on or after the 2024 Effective Date (i.e., all fixed dollar baskets are deemed refreshed on the 2024 Effective Date and any prior usages of such baskets are deemed to be zero).
- (f) With respect to determining availability under baskets set forth in this agreement that is determined by reference to a percentage of the most recently reported Consolidated EBITDA as of such time of determination (including on a "greater of" basis), the most recently reported Consolidated EBITDA shall be deemed to be the greater of (x) Consolidated EBITDA as of the applicable most recently ended Testing Period as of such time of determination and (y) the greatest Consolidated EBITDA for any trailing four fiscal quarter period ending prior to such Testing Period.

1.5 Cashless Roll.

Notwithstanding anything to the contrary contained in this agreement or in any other Finance Document, to the extent that any Lender extends the maturity date of, or replaces, renews or refinances, any of its then existing Loans with Incremental Term Facility Loans, Incremental RCF Commitments, Refinancing Term Loans, Replacement Revolving Loans, Extended Term Loans, Extended Revolving Loans or loans incurred under a new credit facility, in each case, to the extent such extension, replacement, renewal or refinancing is effected by means of a "cashless roll" by such Lender, such extension, replacement, renewal or refinancing shall be deemed to comply with any requirement hereunder or any other Finance Document that such payment be made "in Dollars", "in immediately available funds", "in cash" or any other similar requirement.

1.6 Interest Rates; Benchmark Notifications.

The interest rate on a Loan denominated in Dollars, euros or any other Optional Currency may be derived from an interest rate benchmark that may be discontinued or is, or may in the future become, the subject of regulatory reform. Upon the occurrence of a Benchmark Transition Event, Clauses 15 (Interest Periods) and 16 (Changes to the Calculation of Interest) provide a mechanism for determining an alternative rate of interest. The Facility Agent does not warrant or accept any responsibility for, and shall not have any liability with respect to, the administration, submission, performance or any other matter related to any interest rate used in this agreement, or with respect to any alternative or successor rate thereto, or replacement rate thereof, including without limitation, whether the composition or characteristics of any such alternative, successor or replacement reference rate will be similar to, or produce the same value or economic equivalence of, the existing interest rate being replaced or have the same volume or liquidity as did any existing interest rate prior to its discontinuance or unavailability. The Facility Agent and its affiliates and/or other related entities may engage in transactions that affect the calculation of any interest rate used in this agreement or any alternative, successor or alternative rate (including any Benchmark Replacement) and/or any relevant adjustments thereto, in each case, in a manner adverse to the Group. The Facility Agent may select information sources or services in its reasonable discretion to ascertain any interest rate used in this agreement, any component thereof, or rates referenced in the definition thereof, in each case pursuant to the terms of this agreement, and shall have no liability to the Group, any Lender or any other person or entity for damages of any kind, including direct or indirect, special, punitive,

incidental or consequential damages, costs, losses or expenses (whether in tort, contract or otherwise and whether at law or in equity), for any error or calculation of any such rate (or component thereof) provided by any such information source or service. Notwithstanding the provisions of the definition of Benchmark Replacement Conforming Changes herein, and Section 16.3 (Benchmark Replacement) hereunder, the Facility Agent shall be entitled to (but is not obliged to) request instructions from the Majority Lenders (or if this agreement or any other Finance Document stipulates the matter is a decision for any other Lender or group of Lenders) as to whether, and in what manner, it should make or refrain from making a Benchmark Replacement Conforming Change and the Facility Agent may refrain from acting (but shall not be obliged to refrain from acting) unless and until it received any such instructions that it has requested.

2. THE FACILITIES

2.1 The Facilities

- (a) Subject to the terms of this agreement, the Lenders make available:
 - a Base Currency term loan facility in an aggregate principal amount equal to the Total Facility A
 Commitments which, to the extent not originally already drawn on the 2022 Effective Date
 were made available by way of Loans on the 2022 Effective Date;
 - (ii) a Base Currency term loan facility in an aggregate principal amount equal to the Total Facility B Commitments and which were continued as and/or made available by way of Loans on the 2023 Effective Date as 2023 Continued Term Facility B Loan Participations on a cashless roll basis or 2023 Funded Term Facility B Loan Participations, as applicable, pursuant to the 2023 Term Facility B Refinancing Documents, and which were refinanced and made available by way of Loans on the 2024 Effective Date pursuant to the 2024 Incremental Facility Commitment Letter; and
 - (iii) a multicurrency revolving credit facility in an aggregate principal amount the Base Currency Amount of which is equal to the Total RCF Commitments which shall be available for drawing by each Borrower by way of Utilisations in an aggregate amount equal to the Total RCF Commitments from the 2022 Effective Date.
- (b) Each Term Facility will be available to (in respect of Term Facility A and Term Facility B) Delta 2 and (in respect of each other Term Facility) the Borrower(s) agreed between the Obligors' Agent and the Lenders in respect of that Term Facility. Each Revolving Facility will be available to any of the Borrowers.
- (c) No Group Company incorporated in Belgium or Switzerland is permitted to make Utilisations under any Facility.
- (d) Subject to the terms of this agreement and the Ancillary Documents, an Ancillary Lender may make available an Ancillary Facility to any of the Borrowers in place of all or part of its Commitment under a Revolving Facility.

2.2 Incremental Facilities

- (a) Subject to and in accordance with the provisions of this clause 2.2, the Obligors' Agent or any Borrower may:
 - (i) increase the commitments under an existing Term Facility and/or establish additional commitments for a new term facility (any such increased or additional commitments "Incremental Term Facility Commitments" and any such new term facility an "Incremental Term Facility"); and/or

- (ii) increase the commitments under an existing Revolving Facility and/or establish additional commitments for a new Revolving Facility (any such increased or additional commitments "Incremental RCF Commitments" and any such new Revolving Facility an "Incremental Revolving Facility"). Incremental Facility Commitments may be constituted in the Base Currency or any other currency or currencies agreed between the Parent or the relevant Borrower and the Lenders providing such commitments.
- (b) Without prejudice to paragraph (c) below, Incremental Facility Commitments may be constituted in an unlimited amount provided that, at the time such Incremental Facility Commitments are constituted, Net First Lien Secured Leverage, calculated on a pro forma basis for drawdown in full and use of the relevant Incremental Facility Commitments, acquisitions or disposals or prepayment of indebtedness in connection therewith and Certified Synergies and assuming full utilisation of any Revolving Facility Commitments and any Additional First Lien Debt which is a revolving facility to the extent such Revolving Facility Commitments and revolving Additional First Lien Debt exceed \$75,000,000 in aggregate (and only such excess shall be taken into account) would not exceed (x) 5.50:1 or (y) solely in the case of any such Incremental Facility Commitment being incurred to finance a Permitted Acquisition, the Net First Lien Secured Leverage immediately prior to the incurrence of such Incremental Facility Commitments (the "Incremental First Lien Ratio Builder").
- (c) Incremental Facility Commitments may be constituted regardless of whether the Net First Lien Secured Leverage requirements of paragraph (b) above can be met provided that the aggregate amount of Incremental Facility Commitments so constituted after the 2022 Effective Date (measured when incurred) shall not exceed (such amount, the "Incremental Fixed Amount"):
 - (i) the greater of (x) \$500,000,000 (or its equivalent in other currencies) and (y) 100% of the then most recently reported Consolidated EBITDA; plus
 - (ii) any unused capacity under clause (q) of the definition of "Permitted Financial Indebtedness" that Borrower elects to reallocate to the Incremental Fixed Amount; plus
 - (iii) the amount of any optional prepayment of any Loan (in the case of revolving Financial Indebtedness, accompanied by corresponding voluntary permanent reductions of revolving commitments) and any Incremental Facilities or Additional Debt, plus
 - (iv) the aggregate principal amount of any Loan reduction resulting from any assignment of such Loans to (and/or purchase of such Loans by) any Group Company (limited, in the case of purchases made at a discount to par value, to the actual purchase price of such Loan paid in cash) not financed with the proceeds of long-term Financial Indebtedness (other than any revolving indebtedness); minus
 - (v) the aggregate outstanding principal amount of all Incremental Facilities and/or Additional Debt incurred or issued after the 2022 Effective Date in reliance on the Incremental Fixed Amount.
- (d) Incremental Facility Commitments may be constituted at any time (x) regardless of whether the Net First Lien Secured Leverage requirements of paragraph (b) are not met and (y) regardless of whether any availability exists to incur Incremental Facilities pursuant to paragraph (c) so long as amounts funded by such commitments are to be applied directly or indirectly towards:
 - (i) refinancing or replacing any First Lien Liabilities; and/or

- (ii) the payment of principal, interest, fees, discounts, expenses, commissions, premia or other similar amounts payable under or in connection with such refinancing or replacement and/or fees, costs and expenses incurred in connection therewith.
- In relation to any broadly syndicated first lien Incremental Term Facility Commitments in USD (e) constituted within the first six months after the 2024 Effective Date (excluding broadly syndicated first lien Incremental Term Facility Commitments in an aggregate amount not exceeding \$850,000,000 established after the 2024 Effective Date), the Effective Yield in respect of the relevant Incremental Term Facility Commitments shall be no higher than the Effective Yield for the Term Facility B Loans plus 0.75 per cent per annum (the "Maximum Incremental Yield") unless the Effective Yield applicable to the Term Facility B Loans is increased (unless otherwise specified by the Parent, by way of Margin increase) by an amount equal to the amount by which the Effective Yield in respect of such Incremental Term Facility Commitments exceeds the Maximum Incremental Yield, provided that in the relevant Incremental Facility Commitment Letter the Obligors' Agent either confirms that the Effective Yield in respect of the relevant Incremental Term Facility Commitments is no higher than the Maximum Incremental Yield or otherwise provides details of any increase in Effective Yield applicable to the Term Facility B Loans required to be made pursuant to this paragraph and provided that no such increase shall result in the Effective Yield in respect of the Term Facility B Loans exceeding the Effective Yield in respect of the relevant Incremental Term Facility Commitments.
- (f) For the purposes of paragraph (e) above, "Effective Yield" means, in respect of Loans in any currency, the aggregate (in each case expressed to be a percentage) of:
 - (i) the Margin applicable to the relevant Loans on the date on which such Loans were made and ignoring any subsequent margin ratchet or step-down; and
 - (ii) any upfront fee or original issue discount applicable to the relevant Loans divided by 4 (and excluding, for the avoidance of doubt, any consent fees, commitment fees, underwriting, arrangement or structuring fees or other fees that are not customarily shared with the relevant lenders) and provided that any interest rate floor applicable to the relevant Incremental Term Facility Commitments and/or to the Term Facility B on the date of determination shall be equated to interest margin (at a rate equal to the rate of such floor) for determining the applicable Effective Yield.
- While any Term Facility B Loans remain outstanding the final maturity date for any Incremental Term (g) Facility Commitments in the form of a broadly syndicated term loan (for the avoidance of doubt, any Incremental Term Facility in the form of a term "a" facility or otherwise club credit facilities provided by relationship banks shall not be treated as broadly syndicated) may not fall before the latest Termination Date in relation to such Term Facility B Loans (as at the time the relevant Incremental Term Facility Commitments are originally constituted) and no scheduled repayment instalments may fall before such Termination Date (save for customary amortisation and any Incremental Term Facility that constitutes an increase of commitments to an existing tranche of Term Loans where the final $% \left(1\right) =\left(1\right) \left(1\right$ maturity remains the same); provided that any Incremental Term Facility Commitments in the form of a broadly syndicated term loan in an outstanding amount (measured solely when incurred) equal to the greater of (x) \$500,000,000 (or its equivalent in other currencies) and (y) 100% of the then mostly recently reported Consolidated EBITDA (as reduced by any outstanding (such amount measured solely when such Incremental Term Facility Commitments are incurred) (A) Additional Debt incurred under the Additional Debt Maturity Exclusion and (B) Refinancing Indebtedness incurred under the Refinancing

Maturity Exclusion) (such amount, the "**Incremental Term Facility B Maturity Exclusion**") may have a maturity date before the latest Termination Date of the Term Facility B Loans.

- (h) [Reserved]
- (i) Incremental Facility Commitments shall:
 - (i) rank pari passu in right of payment and security with the other Commitments; and
 - (ii) subject to the Agreed Security Principles be guaranteed and secured by the same guarantors and the same security as the other Commitments.
- (j) Incremental Facility Commitments may be provided by any Lender or any other person that the Obligors' Agent may select. Each entity which agrees to provide an Incremental Facility Commitment and which is selected by the Company to do so will be an Incremental Facility Lender subject to, in the case of any Further Lender, each such Further Lender having acceded to this agreement as a Lender and having acceded to the Intercreditor Agreement as a First Lien Lender by executing and delivering an Incremental Facility Commitment Letter. For the avoidance of doubt, no Lender is obliged to provide or participate with an Incremental Facility unless it is an Incremental Facility Lender in relation to such Incremental Facility.
- (k) Obligors' Agent may implement the proposed Incremental Facility Commitments by delivering to the Facility Agent an Incremental Facility Commitment Letter signed by the Obligors' Agent, the relevant Borrowers and the relevant Lenders and/or Further Lenders that have agreed to provide such Incremental Facility Commitments specifying:
 - (i) the amount of Incremental Facility Commitments to be made available by each such Lender and/or Further Lender;
 - (ii) whether such Incremental Facility Commitments are to increase the commitments under an existing Facility or to establish a new Facility;
 - (iii) the date on which such Incremental Facility Commitments are to become committed;
 - (iv) the identity and notice details of each such Lender and/or Further Lender;
 - (v) the Availability Period applicable to such Incremental Facility Commitments;
 - (vi) the identity of the Borrowers in respect of such Incremental Facility Commitments;
 - (vii) the interest periods (if applicable) and interest rates (and any applicable margin ratchet) applicable to such Incremental Facility Commitments;
 - (viii) the commitment fees payable in connection with such Incremental Facility Commitments;
 - (ix) the final maturity date applicable to such Incremental Facility Commitments and any other repayment dates;
 - the currency or currencies in which such Incremental Facility Commitments may be drawn, provided that such currencies constitute Incremental Facility Currencies;

- (xi) the permitted purposes for which such Incremental Facility Commitments may be used;
- (xii) any such conditions precedent to availability or drawdown or provisions which allow the drawdown of such Incremental Facility Commitments to be on a 'certain funds' basis as the Obligors' Agent and such Lenders and/or Further Lenders may wish to specify; and
- (xiii) such other terms as Obligors' Agent and such Lenders and/or Further Lenders may wish to specify which are applicable to such Incremental Facility.
- (I) Subject to any express restrictions set out in this clause 2.2 (Incremental Facilities), the principal amount, interest rate, interest periods (if any) and interest payment dates, use of proceeds, repayment schedule, availability period and fees and all other terms applicable to the relevant Incremental Facility Commitments (including provisions for the relevant Incremental Facility Commitments to be made available on a certain funds basis) shall be agreed by the relevant Borrower (or the Obligors' Agent on its behalf) and the lender(s) of those Incremental Facility Commitments without the need to obtain any further consent from any Finance Parties. Provided that an Incremental Facility Commitment Letter complies with the other provisions of this clause 2.2 (Incremental Facilities) and no Event of Default is continuing at the date of the relevant Incremental Facility Commitment Letter or would occur as a result of implementing the Incremental Facility Commitments and subject to completing any "know-your-customer" or other similar checks contemplated by paragraph (r) below, the Facility Agent shall (and is irrevocably authorised by the Finance Parties to) promptly counter-sign that Incremental Facility Commitment Letter. Upon countersignature of an Incremental Facility Commitment Letter by the Facility Agent:
 - (i) that Incremental Facility Commitment Letter shall constitute a "Finance Document" for the purposes of this agreement and the other Finance Documents;
 - the commitments set out in such Incremental Facility Commitment Letter shall constitute
 Incremental Facility Commitments as from the date specified therein, without prejudice to any
 conditions precedent specified therein;
 - (iii) the terms listed in paragraph (k) above and specified in that Incremental Facility Commitment Letter shall take effect and apply in respect of that Incremental Facility without prejudice to the other terms of this agreement which remain applicable to the other Facilities; and
 - (iv) that Incremental Facility Commitment Letter and this agreement shall be read and construed together as a single agreement and references in the Finance Documents to this agreement shall include that Incremental Facility Commitment Letter.
- (m) The Facility Agent shall notify all the other Lenders of the terms of an Incremental Facility Commitment Letter promptly following receipt and shall notify the other Lenders as and when an Incremental Facility Commitment Letter has been countersigned by the Facility Agent.
- (n) Each Obligor and each Finance Party agrees and acknowledges that:
 - (i) any Incremental Facility Commitments established pursuant to this clause 2.2 (Incremental Facilities) and any Utilisations in relation thereto are intended to form part of the "Indebtedness" or "Secured Obligations" or "Secured Liabilities" (or equivalent definitions) in the Transaction Security Documents

- entered into or amended and confirmed on or around the Sixth Restatement Date or after (the "Current Security Documents") to which that Obligor is party;
- (ii) to the extent provided by applicable law, the security granted pursuant to the Current Security Documents then in effect is intended to secure, without limitation, all of the obligations of the Obligors under the Finance Documents, including in respect of any Incremental Facility Commitments established pursuant to this clause 2.2 (Incremental Facilities).
- (o) Each Guarantor agrees and confirms that:
 - (i) the guarantee and indemnity given by it contained in clause 23 (Guarantee and Indemnity) is intended to continue in full force and effect in respect of the obligations of the Obligors under the Finance Documents notwithstanding the establishment of any Incremental Facility Commitments pursuant to this clause 2.2 (Incremental Facilities); and
 - (ii) the guarantees and indemnities given by it contained in clause 23 (Guarantee and Indemnity) are intended to apply and extend to all of the obligations of the Obligors under the Finance Documents including in respect of any Incremental Facility Commitments.
- (p) The confirmations given by the Obligors in paragraphs (n) and (o) above are given subject to any limitations on such security or guarantees and indemnities as recorded in the relevant Finance Documents including the Current Security Documents, in clause 23.12 (Guarantee Limitations) or any Accession Letter by which the relevant Obligor became a Borrower and/or a Guarantor.
- (q) At the time any Incremental Facility Commitments are established which increase Commitments under an existing Facility, the Facility Agent is authorised (but not obliged) to record the Incremental Facility Commitments as a separate sub-facility of the applicable facility on the terms agreed between the Obligors' Agent and the Facility Agent. The Facility Agent shall notify the Lenders of any such designation. Upon receiving an Incremental Facility Commitment Letter in respect of Incremental Facility Commitments which increase Commitments under an existing Facility, the Facility Agent may notify the relevant Borrowers and Lenders of any one-off interest periods and/or consolidation or division of Term Loans which is considered by it to be reasonably necessary for its administration of the Facility to which the Incremental Facility Commitments are to form part, whereupon such interest periods, consolidation or division shall take effect notwithstanding any provision to the contrary in this agreement or the relevant Incremental Facility Commitment Letter.
- (r) The creation of Incremental Facility Commitments will only be effective in relation to a person that is not already a Lender upon the performance by the Facility Agent of all necessary "know your customer" or other similar checks under all applicable laws and regulations, the completion of which the Facility Agent shall promptly notify to the Parent, such person and the Issuing Bank (if applicable). The Facility Agent shall promptly undertake all such "know your customer" and other similar checks as soon as reasonably practicable.
- (s) The Finance Parties and each Obligor shall (at the cost of the Obligors) enter into any amendment to the Finance Documents and/or replacement or additional documentation consequential on or required by applicable law to effectively create any Incremental Facility Commitments (or to conform the terms of this agreement to those in force as a result of the operation of paragraph (I) above) and/or to ensure that (subject to the Agreed Security Principles) the Lenders in respect of the Incremental Facilities have (and the other Finance Parties continue to have) the

benefit of the same guarantee and security interests under this agreement, any Accession Letter and the Transaction Security Documents as existed immediately before the establishment of any Incremental Facility Commitments pursuant to this clause 2.2 (Incremental Facilities), including:

- any changes to, the taking of, or the release coupled with the retaking of any Transaction Security;
- (ii) the entry into of any additional Transaction Security Documents; and/or
- (iii) the entry into of any supplemental agreements, confirmations or similar or equivalent documents.

For the avoidance of doubt, no such action of an Obligor described in this paragraph (s) shall be a condition precedent to the establishment of the relevant Incremental Facility Commitments (unless requested by Obligors' Agent or otherwise agreed between Obligors' Agent and either the Majority Lenders or the lenders in respect of the relevant Incremental Facility Commitments) but may be a condition subsequent to be delivered within a time after such establishment reasonably specified by the Security Agent (with respect to any documents or actions required pursuant to this paragraph (s) relating to the Transaction Security), or the Facility Agent (with respect to any other documents or actions required pursuant to this paragraph (s), and agreed by Obligors' Agent (acting reasonably).

- (t) The Facility Agent and the Security Agent are irrevocably authorised and instructed by each Finance Party to enter into any documentation referred to in the previous paragraph on behalf of any Finance Party and to take such other action as is reasonably required to give effect to the purposes set out in paragraph (s) above.
- The Obligors' Agent may elect to use the Incremental First Lien Ratio Builder prior to the Incremental Fixed Amount regardless of whether there is capacity under Incremental Fixed Amount, and if the Incremental Fixed Amount, on the one hand, and the Incremental First Lien Ratio Builder, on the other hand, are available and the Obligors' Agent does not make an election, the Obligors' Agent will be deemed to have elected to use the Incremental First Lien Ratio Builder to the extent compliant therewith. The Obligors' Agent may elect to use capacity under Incremental Fixed Amount on the same date that the Borrower uses capacity under the Incremental First Lien Ratio Builder, by first calculating the capacity under the Incremental First Lien Ratio Builder without regard to the any use of capacity under the Incremental Fixed Amount; provided, further, that the Obligors' Agent may redesignate all of any portion of any such Financial Indebtedness originally designated as incurred pursuant to the Incremental Fixed Amount and Incremental Fixed Amount as incurred pursuant to the Incremental Fixed Amount as incurred pursuant to the Incremental Fixed Amount and Incremental Fixed Amount Amo Builder if, at the time of such redesignation, the Obligors' Agent would be permitted to incur such Incremental Facility under the Incremental First Lien Ratio Builder (for purposes of clarity, with any such redesignation having the effect of increasing the ability to incur Financial Indebtedness under the Incremental Fixed Amount as of the date of such redesignation by the amount of such Financial Indebtedness so redesignated).

2.3 Additional Debt

- (a) Members of the Group may:
 - constitute commitments in respect of, incur or assume Financial Indebtedness which is secured
 on a pari passu basis with the Facilities ("Additional First Lien Debt");

- constitute commitments in respect of, incur or assume Financial Indebtedness which is secured on a junior basis to the Facilities ("Additional Second Lien Debt");
- (iii) constitute commitments in respect of, incur or assume Financial Indebtedness which is unsecured ("Additional Unsecured Debt"),

provided that:

- in respect of any Additional First Lien Debt, the agent or trustee in respect of such Financial Indebtedness becomes party to the Intercreditor Agreement as a First Lien Agent (and such Financial Indebtedness is designated as First Lien Liabilities);
- (B) in respect of any Additional Second Lien Debt, the agent or trustee in respect of such Financial Indebtedness becomes party to the Intercreditor Agreement as a Second Lien Agent (and such Financial Indebtedness is designated as Second Lien Liabilities);
- (C) the first scheduled repayment or redemption date (other than amortisation that is customary for such type of indebtedness) of any:
 - (aa) Additional First Lien Debt (other than Permitted Refinancing Indebtedness (which shall be subject to the maturity requirements set forth below for Permitted Refinancing Indebtedness) and any UKEF Facility) in the form of a broadly syndicated term loan (for the avoidance of doubt, any Additional First Lien Debt in the form of a term "a" facility or otherwise club credit facilities provided by relationship banks shall not be treated as broadly syndicated) is no earlier than the latest Termination Date applicable to the Term Facility B Loans (as at the time the relevant commitments and/or Financial Indebtedness constituting Additional First Lien Debt maturity Date"); and
 - (bb) Additional Second Lien Debt or Additional Unsecured Debt (other than Permitted Refinancing Indebtedness (which shall be subject to the maturity requirements set forth below for Permitted Refinancing Indebtedness) in the form of a broadly syndicated term loan (for the avoidance of doubt, any Additional Second Lien Debt in the form of a term "a" facility or otherwise club credit facilities provided by relationship banks shall not be treated as broadly syndicated) is no earlier than the date falling six months after the latest Termination Date applicable to the Term Facility B Loans (as at the time the relevant commitments and/or Financial Indebtedness constituting Additional Second Lien Debt or Additional Unsecured Debt are constituted and/or incurred) (such date, the "Additional Non-First Lien Debt Maturity Date"); and

in each case determining the final maturity of any relevant debt which is a bridge loan by reference to the notes or term loans into which such bridge loan is to be converted to or exchanged for at maturity, and excluding customary offers to repurchase or mandatory prepayments upon a change of control, asset sale, change in taxation or event of loss and customary acceleration rights after an event of default; provided that any Additional First Lien Debt, Additional Second Lien Debt and Additional Unsecured Debt, in each case, that is in the form

of a broadly syndicated term loan, in an outstanding amount (measured solely when incurred, created or assumed and not including any Permitted Refinancing Indebtedness or any UKEF Facility) equal to the greater of (x) \$500,000,000 (or its equivalent in other currencies) and (y) 100% of the then mostly recently reported Consolidated EBITDA (as reduced by any outstanding (such amount measured solely when such Additional First Lien Debt, Additional Second Lien Debt and Additional Unsecured Debt are incurred, created or assumed) (A) Incremental Term Facility Commitments incurred under the Incremental Term Facility B Maturity Exclusion and (B) Refinancing Indebtedness incurred under the Refinancing Maturity Exclusion) (such amount, the "Additional Debt Maturity Exclusion") may have a maturity date before the Additional First Lien Debt Maturity Date (in the case of Additional First Lien Debt that is in the form of a broadly syndicated term loan) or the Additional Unsecured Debt, in each case, that is in the form of a broadly syndicated term loan).

- (D) in respect of loan facilities, at the time the relevant commitments are constituted and, in respect of other Financial Indebtedness, at the time such Financial Indebtedness is incurred or assumed, either (A):
 - (1) in the case of Additional First Lien Debt (other than Permitted Refinancing Indebtedness incurred in respect of the refinancing of any then existing First Lien Liabilities) Net First Lien Secured Leverage, calculated on a pro forma basis for the incurrence in full and use of the relevant Financial Indebtedness, acquisitions or disposals or prepayment of indebtedness in connection therewith and Certified Synergies and assuming full utilisation of any Revolving Facility Commitments and any Additional First Lien Debt which is a revolving facility to the extent such Revolving Facility Commitments and Additional First Lien Debt exceed in aggregate \$75,000,000 (and only such excess shall be taken into account) is less than or equal to (x) 5.50:1 or (y) solely in the case of any such Additional First Lien Debt being incurred to finance a Permitted Acquisition, the Net First Lien Secured Leverage immediately prior to the incurrence of such Additional First Lien Debt (the "Additional First Lien Debt Ratio Test"); or
 - (2) in the case of Additional Second Lien Debt (other than Permitted Refinancing Indebtedness) Net Total Secured Leverage, calculated on a pro forma basis for the incurrence in full and use of the relevant Financial Indebtedness, acquisitions or disposals or prepayment of indebtedness in connection therewith and Certified Synergies, is less than or equal to (x) 8.00:1 or (y) solely in the case of any such Additional Second Lien Debt being incurred to finance a Permitted Acquisition, the Net Total Secured Leverage immediately prior to the incurrence of such Additional Second Lien Debt (the "Additional Second Lien Debt Ratio Test"); or
 - (3) in the case of Additional Unsecured Debt (other than Permitted Refinancing Indebtedness), either (x) Net Total Leverage, calculated on a pro forma basis for the incurrence in full and use of the relevant Financial Indebtedness, acquisitions or disposals or prepayment of indebtedness in connection therewith and

Certified Synergies, is less than or equal to (i) 8.25:1 or (ii) solely in the case of any such Additional Unsecured Debt being incurred to finance a Permitted Acquisition, the Net Total Leverage immediately prior to the incurrence of such Additional Unsecured Lien Debt or (y) Consolidated Interest Coverage Ratio, calculated on a pro forma basis for the incurrence in full and use of the relevant Financial Indebtedness, acquisitions or disposals or prepayment of indebtedness in connection therewith and Certified Synergies is not less than 2.00:1 (the "Additional Unsecured Debt Ratio Test" and, together with the Additional First Lien Debt Ratio Test and the Additional Second Lien Debt Ratio Test, the "Additional Debt Ratio Tests");

- or (B) the aggregate amount of Additional Debt, to be incurred, created or assumed does not exceed the then available Incremental Fixed Amount (measured at the time incurred, created or assumed);
- The Obligors' Agent may elect to use the Additional Debt Ratio Tests prior to the Incremental Fixed Amount regardless of whether there is capacity under Incremental Fixed Amount, and if the Incremental Fixed Amount, on the one hand, and the Additional Debt Ratio Tests, on the other hand, are available and the Obligors' Agent does not make an election, the Obligors' Agent will be deemed to have elected the applicable Additional Debt Ratio Test to the extent compliant therewith. The Obligors' Agent may elect to use capacity under Incremental Fixed Amount on the same date that the Obligors' Agent uses capacity under the Additional Debt Ratio Tests, by first calculating the capacity under the applicable Additional Debt Ratio Test without regard to the any use of capacity under the Incremental Fixed Amount; provided, further, that the Obligors' Agent may redesignate all of any portion of any such Financial Indebtedness originally designated as incurred pursuant to the Incremental Fixed Amount as incurred pursuant to the applicable Additional Debt Ratio Tests if, at the time of such redesignation, the Obligors' Agent would be permitted to incur such Incremental Facility under the applicable Additional Debt Ratio Tests (for purposes of clarity, with any such redesignation having the effect of increasing the ability to incur Financial Indebtedness under the Incremental Fixed Amount as of the date of such redesignation by the amount of such Financial Indebtedness so redesignated).
- (F) Subject to the Agreed Security Principles, the Additional Debt shall not be incurred, assumed or guaranteed by any Group Company that is not an Obligor and, to the extent secured, shall not be secured by any assets that do not secure the Commitments.
- (b) "Permitted Refinancing Indebtedness" means any Financial Indebtedness the proceeds of which are to be applied directly or indirectly towards:
 - (i) refinancing or replacing all or a portion of any First Lien Liabilities, Second Lien Liabilities, or Additional Unsecured Debt or any other outstanding Financial Indebtedness (including, for the avoidance of doubt, any outstanding principal amount of unutilized existing revolving commitments) (such Financial Indebtedness, the "Refinanced Indebtedness"); and/or
 - (ii) the payment of principal, interest, fees, discounts, expenses, commissions, premia or other similar amounts payable under or in connection with such

provided that (a) the principal amount (or accreted value or fair value, if applicable) of the Permitted Refinancing Indebtedness does not exceed the amounts set forth in clauses (i) and (ii) above, (b) the obligor of Permitted Refinancing Indebtedness is either the same obligor of the Refinanced Indebtedness or is an Obligor, (c) if the Permitted Refinancing Indebtedness was subordinated in right of payment or security to the Commitments, then such Permitted Refinancing Indebtedness, by its terms, is subordinate in right of payment and security to the Commitments at least to the same extent as the Refinanced Indebtedness, and (d) the first scheduled repayment or redemption date (other than amortisation that is customary for such type of indebtedness) of such Permitted Refinancing Indebtedness is either (i) no earlier than the Financial Indebtedness being repaid or refinanced or (ii) is no earlier than the latest Termination Date applicable to the Term Loans (or in the case of Permitted Refinancing Indebtedness being used to repay or refinance some but not all Term Loans, the latest Termination Date applicable to the Term Loans being repaid or refinanced) or the Termination Date applicable to the Facility RCF (such earlier date, the "Refinancing Maturity Date"), provided that any Permitted Refinancing Indebtedness in an outstanding amount (measured solely when incurred, created or assumed and not including any UKEF Facility) equal to the greater of (x) \$500,000,000 (or its equivalent in other currencies) and (y) 100% of the then mostly recently reported Consolidated EBITDA (as reduced by any outstanding (such amount measured solely when such Permitted Refinancing Indebtedness are incurred, created or assumed) with respect to (A) Incremental Term Facility Commitments incurred under the Incremental Term Facility B Maturity Exclusion and (B) Additional Debt incurred under the Additional Debt Maturity Exclusion) (such amount, the "Refinancing Maturity Exclusion") may have a maturity date before the Refinancing Maturity Date.

- (c) Save to the extent provided for in this clause 2.3 (Additional Debt), the principal amount, interest rate, interest periods and interest payment dates, use of proceeds, repayment schedule, availability period and fees and all other terms applicable to the relevant Additional Debt (including provisions for the relevant Additional Debt to be made available on a certain funds basis) shall be agreed by the relevant borrower (or Obligors' Agent on its behalf and the creditor(s) in respect of that Additional Debt without the need to obtain any further consent from any Finance Parties and included in the relevant Additional First Lien Finance Documents or Additional Second Lien Finance Documents (as applicable) (each as defined in the Intercreditor Agreement).
- (d) The Finance Parties and each Obligor shall (at the cost of the Obligors) enter into any amendment to the Finance Documents and/or replacement or additional documentation consequential on or required by applicable law to allow the Obligors to effectively constitute commitments in respect of, incur or assume Additional Debt from the relevant creditors and/or to ensure that (subject to the Agreed Security Principles) the creditors in respect of the Additional Debt have (and the Finance Parties continue to have) the benefit of the same guarantee and security interests under the Transaction Security Documents as existed immediately before the establishment of the relevant commitments and/or the incurrence or assumption of Additional Debt pursuant to this clause 2.3, including:
 - (i) any changes to, the taking of, or the release coupled with the retaking of any Transaction Security;
 - (ii) the entry into of any additional Transaction Security Documents; and/or
 - (iii) the entry into of any supplemental agreements, confirmations or similar or equivalent documents.

- (e) For the avoidance of doubt, no such action of an Obligor described above shall be a condition precedent to the incurrence or assumption of Additional Debt (unless agreed between Obligors' Agent and the Majority Lenders) but may be a condition subsequent to be delivered within a time reasonably specified by the Security Agent after such establishment and agreed by Obligors' Agent (acting reasonably).
- (f) The Facility Agent and the Security Agent are irrevocably authorised and instructed by each Finance Party to enter into any documentation referred to in the previous paragraph on behalf of any Finance Party and to take such other action as is reasonably required to give effect to the purposes set out in paragraph (d) above.

2.4 Finance Parties' Rights and Obligations

- (a) The obligations of each Finance Party under the Finance Documents are several. Failure by a Finance Party to perform its obligations under the Finance Documents does not affect the obligations of any other Party under the Finance Documents. No Finance Party is responsible for the obligations of any other Finance Party under the Finance Documents.
- (b) The rights of each Finance Party under or in connection with the Finance Documents are separate and independent rights and any debt arising under the Finance Documents to a Finance Party from an Obligor is a separate and independent debt in respect of which a Finance Party shall be entitled to enforce its rights in accordance with paragraph (c) below. The rights of each Finance Party include any debt owing to that Finance Party under the Finance Documents and, for the avoidance of doubt, any part of a Loan or any other amount owed by an Obligor which relates to a Finance Party's participation in a Facility or its role under a Finance Document (including any such amount payable to the Agent on its behalf) is a debt owing to that Finance Party by that Obligor.
- (c) A Finance Party may, except as specifically provided in the Finance Documents, separately enforce its rights under or in connection with the Finance Documents.

2.5 Obligors' Agent

- (a) Each Obligor (other than, if an Obligor, Obligors' Agent) by its execution of this agreement or an Accession Letter (or other accession hereto) irrevocably appoints Obligors' Agent to act on its behalf as its agent in relation to the Finance Documents and irrevocably authorises:
 - (i) Obligors' Agent on its behalf to supply all information concerning itself contemplated by this agreement to the Finance Parties and to give all notices and instructions (including, in the case of a Borrower, Utilisation Requests), to execute on its behalf any Accession Letter, to make such agreements and to effect the relevant amendments, supplements and variations (in each case, however fundamental) capable of being given, made or effected by any Obligor (notwithstanding that they may increase the Obligor's obligations or otherwise affect the Obligor) and to give confirmation as to continuation of surety obligations, without further reference to or the consent of that Obligor; and
 - (ii) each Finance Party to give any notice, demand or other communication to that Obligor pursuant to the Finance Documents to Obligors' Agent,

and in each case the Obligor shall be bound as though the Obligor itself had given the notices and instructions (including, without limitation, any Utilisation Requests) or executed or made the agreements or effected the amendments, supplements or variations, or received the relevant notice, demand or other communication.

(b) Every act, omission, agreement, undertaking, settlement, waiver, amendment, supplement, variation, notice or other communication given or made by the Obligors' Agent or given to the Obligors' Agent under any Finance Document on behalf of another Obligor or in connection with any Finance Document (whether or not known to any other Obligor and whether occurring before or after such other Obligor became an Obligor under any Finance Document) shall be binding for all purposes on that Obligor as if that Obligor had expressly made, given or concurred with it. In the event of any conflict between any notices or other communications of the Obligors' Agent and any other Obligor, those of the Obligors' Agent shall prevail.

2.6 Extension Offers

- Notwithstanding anything to the contrary in this agreement, pursuant to one or more offers made from time to time by Obligors' Agent to all Lenders of any participations in Term Loans with a like maturity date and/or Revolving Facility Commitments with a like maturity date on a pro rata basis, and on the same terms to each such Lender ("Pro Rata Extension Offers"), the Obligors' Agent is hereby permitted to consummate transactions with individual Lenders that agree to such transactions from time to time to extend the maturity date of such Lender's participations in Loans and/or Revolving Facility Commitments of such Class and to otherwise modify the terms of such Lender's participations in Loans and/or Revolving Facility Commitments of such Class pursuant to the terms of the relevant Pro Rata Extension Offer (including, without limitation, increasing the interest rate or fees payable in respect of such Lender's participations in Loans and/or Revolving Facility Commitments and/or modifying the amortization schedule in respect of such Lender's participations in Loans); provided that no Default shall have occurred and be continuing at the time a Pro Rata Extension Offer is delivered to the Lenders. For the avoidance of doubt, the reference to "on the same terms" in the preceding sentence shall mean, (i) in the case of an offer to the Lenders under any Class of Term Loans, that all of the participations in Term Loans of such Class are offered to be extended for the same amount of time and that the interest rate changes and fees payable with respect to such extension are the same and (ii) in the case of an offer to the Lenders under any Revolving Facility, that all of the Revolving Facility Commitments of such Facility are offered to be extended for the same amount of time and that the interest rate changes and fees payable with respect to such extension are the same. Any such extension (an "Extension") agreed to between the Obligors' Agent and any such Lender (an "Extending Lender") will be established under an Extension Amendment (any such extended participations in a Term Loan, a participation in an "Extended Term Loan"; any such extended Revolving Facility Commitment, an "Extended Revolving Commitment," and any Revolving Facility Loan made pursuant to such Extended Revolving Commitment, an "Extended Revolving Loan"). Each Pro Rata Extension Offer shall specify the date on which the Obligors' Agent proposes that the Extended Term Loan shall be made or the proposed Extended Revolving Commitment shall become effective, which shall be a date not earlier than five (5) Business Days after the date on which notice is delivered to the Facility Agent (or such shorter period agreed to by the Facility Agent in its reasonable discretion).
- (b) The Obligors' Agent and each Extending Lender shall execute and deliver to the Facility Agent an amendment to this agreement (an "Extension Amendment") and such other documentation as the Facility Agent shall reasonably specify to evidence the Extended Term Loans and/or Extended Revolving Facility Commitments of such Extending Lender. Each Extension Amendment shall specify the terms of the applicable Extended Term Loans and/or Extended Revolving Commitments; provided, that (i) except as to interest rates, fees and any other pricing terms, and amortization, final maturity date and participation in prepayments and commitment reductions (which shall, subject to clauses (ii) and (iii) of this proviso, be determined by the Obligors' Agent and set forth in the Pro Rata Extension Offer), the Extended Term

Loans shall have the same terms as the existing Class of Term Loans from which they are extended except for any terms which shall not apply until after the then latest Termination Date related to outstanding Term Loans, (ii) the final maturity date of any Extended Term Loans shall be no earlier than the maturity date of the Class of Term Loans subject to such Pro Rata Extension Offer, (iii) except as to interest rates, fees, any other pricing terms and final maturity (which shall be determined by the Obligors' Agent and set forth in the Pro Rata Extension Offer), any Extended Revolving Commitment shall have the same terms as the existing Class of Revolving Facility Commitments from which they are extended except for any terms which shall not apply until after the Termination Date related to outstanding Revolving Facility Commitments and, in respect of any other terms that would affect the rights or duties of any Issuing Bank, such terms as shall be reasonably satisfactory to such Issuing Bank, and (iv) any Extended Term Loans may require participation on a pro rata basis or a less than pro rata basis (but not a greater than pro rata basis) than the Term Loans in any mandatory prepayment hereunder. Upon the effectiveness of any Extension Amendment, this agreement shall be amended to the extent (but only to the extent) necessary to reflect the existence and terms of the Extended Term Loans and/or Extended Revolving Commitments evidenced thereby and shall not require the consent of any Lender (other than an Extending Lender). Any such deemed amendment may be memorialized in writing by the Facility Agent with the Obligors' Agent's consent (not to be unreasonably withheld) and furnished to the other parties hereto. If provided in any Extension Amendment with respect to any Extended Revolving Commitments, and with the consent of each Issuing Bank, participations in Letters of Credit shall be reallocated to lenders holding such Extended Revolving Commitments in the manner specified in such Extension Amendment, including upon effectiveness of such Extended Revolving Commitment or upon or prior to the maturity date for any Class of Revolving Facility Commitments.

- (c) Upon the effectiveness of any such Extension, the applicable Extending Lender's participations in a Term Loan will be automatically designated a participation in an Extended Term Loan and/or such Extending Lender's Revolving Facility Commitment will be automatically designated an Extended Revolving Commitment. For purposes of this agreement and the other Finance Documents, (i) if such Extending Lender is extending a participation in a Term Loan, such Extending Lender will be deemed to have a participation in an Extended Term Loan and (ii) if such Extending Lender is extending a Revolving Facility Commitment, such Extending Lender will be deemed to have an Extended Revolving Commitment.
- Notwithstanding anything to the contrary set forth in this agreement or any other Finance Document, (i) no Extended Term Loan or Extended Revolving Commitment is required to be in any minimum amount or any minimum increment, (ii) any Extending Lender may extend all or any portion of its participations in Term Loans and/or Revolving Facility Commitment pursuant to one or more Pro Rata Extension Offers (subject to applicable proration in the case of over participation) (including the extension of any Extended Term Loan and/or Extended Revolving Commitment), (iii) there shall be no condition to any Extension of any participations in a Loan or Revolving Facility Commitment at any time or from time to time other than notice to the Facility Agent of such Extension and the terms of the Extended Term Loan or Extended Revolving Commitment implemented thereby, (iv) all Extended Term Loans, Extended Revolving Commitments and all obligations in respect thereof shall be obligations of the relevant Obligors under this agreement and the other Finance Documents that rank equally and ratably in right of security with all other obligations of the Class being extended (and all other obligations secured by the same priority of security securing the participations in Term Loans or Revolving Facility Loans being extended), (v) no Issuing Bank shall be obligated to issue Letters of Credit under such Extended Revolving Commitments unless it shall have consented thereto and (vi) there shall be no borrower (other than the Borrowers) and no guarantors (other

than the Guarantors) in respect of any such Extended Term Loans or Extended Revolving Commitments.

(e) Each Extension shall be consummated pursuant to procedures set forth in the associated Pro Rata Extension Offer; provided, that the Borrower shall cooperate with the Facility Agent prior to making any Pro Rata Extension Offer to establish reasonable procedures with respect to mechanical provisions relating to such Extension, including, without limitation, timing, rounding and other adjustments.

PURPOSE

3.1 Purpose

- (a) Each Borrower shall apply amounts borrowed by it under each Revolving Facility and Term Facility, any Letter of Credit and any utilisation of any Ancillary Facility towards the general purposes (including, without limitation, for working capital purposes and any other purposes not prohibited by the Finance Documents) of the Group.
- (b) Unless otherwise agreed in the relevant Incremental Facility Commitment Letter, each Borrower shall apply any amounts borrowed by it under an Incremental Facility made available to it pursuant to clause 2.2 (Incremental Facilities) towards the general purposes of the Group.
- (c) No Borrower shall downstream or otherwise lend any amounts borrowed by it under this agreement to ASM.

3.2 Monitoring

No Finance Party is bound to monitor or verify the application of any amount borrowed pursuant to this agreement.

4. CONDITIONS OF UTILISATION

4.1 Initial Conditions Precedent

(a) The conditions to Lenders funding Loans on the 2022 Effective Date are as set out in the 2022 Amendment Request, 2022 Amendment Agreement, or the relevant Incremental Commitment Letter, as applicable.

4.2 Further Conditions Precedent

- (a) The Lenders will only be obliged to comply with clause 5.4 (Lenders' Participation) if on the date of the Utilisation Request and on the proposed Utilisation Date:
 - (i) in the case of a Rollover Loan, no winding up order has been made in respect of the relevant Borrower;
 - (ii) in the case of a Utilisation of an Incremental Facility, any further conditions precedent to Utilisation specified in the Incremental Facility Commitment Letter for that Incremental Facility are satisfied; and
 - (iii) in relation to any other Utilisation, the Repeating Representations, to be made by each Obligor are true in all material respects and no Default is continuing or would result from the proposed Utilisation
- (b) The Lenders will only be obliged to comply with clause 34.9 (Change of Currency) if, on the first day of an Interest Period, no Default is continuing or would result from the change of currency and the Repeating Representations to be made by each Obligor are true in all material respects.

4.3 Conditions Relating to Optional Currencies

- (a) A currency will constitute an Optional Currency in relation to a Revolving Facility Utilisation if:
 - (i) it is readily available in the amount required and freely convertible into the Base Currency in the Relevant Interbank Market at the Specified Time or, if later, on the date the Facility Agent receives the relevant Utilisation Request and the Utilisation Date for that Utilisation; and
 - (ii) it is euro, or has been approved by the Facility Agent (acting on the instructions of the RCF Lenders) on or prior to receipt by the Facility Agent of the relevant Utilisation Request for that Utilisation.
- (b) If the Facility Agent has received a written request from Obligors' Agent for a currency to be approved under paragraph (a)(ii) above, the Facility Agent will confirm to Obligors' Agent by the Specified Time:
 - (i) whether or not the Lenders have granted their approval; and
 - (ii) if approval has been granted, the minimum amount (and, if required, integral multiple) for any subsequent Utilisation in that currency.
- (c) Obligors' Agent and the Facility Agent shall negotiate any amendment required to any provision hereof (including applicable benchmark rate, giving due consideration to (i) any selection or recommendation of a benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (ii) any evolving or then-prevailing market convention for determining a benchmark rate for such Optional Currency for syndicated credit facilities denominated in the applicable Optional Currency at such time in the United States, and any adjustment) to facilitate that Loans be made in such Optional Currency in good faith and such amendment shall be deemed approved by the Lenders if the Majority RCF Lenders shall have received at least five Business Days' prior written notice of such notice to the Lenders with a Revolving Facility Commitment, a written notice from the Majority RCF Lenders stating that the Majority RCF Lenders object to such amendment.

4.4 Maximum Number of Utilisations

- (a) A Borrower (or Obligors' Agent) may not deliver a Utilisation Request or request that a Term Loan be divided if as a result of the proposed Utilisation or division (as the case may be):
 - (i) more than 10 Term Loans (excluding any Incremental Term Facility Loans) would be outstanding;
 - (ii) more than 16 Revolving Facility Utilisations would be outstanding.
- (b) Any Loan made by a single Lender under clause 8.2 (Unavailability of a Currency) shall not be taken into account in this clause 4.4.

4.5 Cancellation at end of Availability Period

At the end of the Availability Period in relation to a Facility, all unutilised Commitments under that Facility will automatically be cancelled.

5. UTILISATION - LOANS

5.1 Delivery of a Utilisation Request

A Borrower, in accordance with Section 2.5, may utilise a Facility by delivery to the Facility Agent of a duly completed Utilisation Request in the form of part 1A of schedule 3 (Requests) not later than the Specified Time (or in the case of an Incremental Facility as provided for in the relevant Incremental Facility Commitment Letter).

5.2 Completion of a Utilisation Request for Loans

- (a) Each Utilisation Request for a Loan is irrevocable (provided that a Utilisation Request may state that such notice is conditioned upon the occurrence or non-occurrence of any event specified therein, in which case such notice may be revoked by the Obligors' Agent (by notice to the Facility Agent on or prior to the specified effective date) if such condition is not satisfied) and will not be regarded as having been duly completed unless:
 - (i) it identifies the Facility to be utilised;
 - (ii) the proposed Utilisation Date is a Business Day within the Availability Period applicable to that Facility:
 - (iii) the currency and amount of the Utilisation comply with clause 5.3 (Currency and Amount);
 - (iv) it identifies whether such Loan will be a Term Benchmark Loan or a RFR Loan, provided that, subject to Clauses 16.2 (Alternate Rate of Interest) and 16.3 (Benchmark Replacement), all Loans shall be Term Benchmark Loans; and
 - (v) other than with respect to RFR Loans, the proposed Interest Period complies with clause 15 (Interest Periods) (or, in respect of an Incremental Facility Loan, the requirements stipulated for that Incremental Facility pursuant to clause 2.2 (Incremental Facilities)).
- (b) Multiple Utilisations may be requested in a Utilisation Request where the proposed Utilisation Date is the 2022 Effective Date. Only one Utilisation may be requested in each subsequent Utilisation Request.

5.3 Currency and Amount

- (a) The currency specified in a Utilisation Request must be:
 - (i) the Base Currency (in relation to a Term Facility denominated in the Base Currency);
 - (ii) Euro in relation to any Term Facility denominated in Euro; and
 - (iii) the Base Currency or an Optional Currency in relation to a Revolving Facility.
- (b) The amount of the proposed Loan must be:
 - (i) if the currency selected is the Base Currency, a minimum of:
 - (A) \$5,000,000 for Facility RCF; and
 - (B) unless otherwise agreed in the relevant Incremental Facility Commitment Letter, \$5,000,000 for an Incremental Revolving Facility,

- or in each case, if less, the Available Facility; or
- (ii) if the currency selected is Euro, a minimum of €5,000,000;
- (iii) if the currency selected is an Optional Currency, the minimum amount (and, if required, integral multiple) specified by the Facility Agent pursuant to clause 4.3(b)(ii) (Conditions Relating to Optional Currencies) or, if less, the Available Facility;
- (iv) in the case of an Incremental Facility Loan, a minimum amount as specified in the relevant Incremental Facility Commitment Letter.

5.4 Lenders' Participation

- (a) If the conditions set out in this agreement have been met, each Lender shall make its participation in each Loan available by the Utilisation Date through its Facility Office.
- (b) The amount of each Lender's participation in each Loan will be equal to the proportion borne by its Available Commitment (where applicable, under the relevant tranche thereof) to the Available Facility (where applicable, within the relevant tranche thereof) immediately prior to making the Loan.
- (c) The Facility Agent shall determine the Base Currency Amount of each Revolving Facility Loan which is to be made in an Optional Currency and notify each Lender of the amount, currency and the Base Currency Amount of each Loan and the amount of its participation in that Loan by the Specified Time.

5.5 **Limitations on Utilisations**

(a) The maximum aggregate Base Currency Amount of all Letters of Credit shall not exceed \$26,500,000 (or in the case of Letters of Credit issued under an Incremental Revolving Facility such other amount as is provided for in the relevant Incremental Facility Commitment Letter (it being understood that any increase in such amount set forth in the relevant Incremental Facility Commitment Letter shall require the consent of the applicable Issuing Bank)).

5.6 **Designated Entities**

- (a) A Lender (the "Related Lender") may designate an Affiliate or substitute Facility Office (a "Designated Entity") for the purpose of participating in any Utilisation to a Borrower in a particular jurisdiction.
- (b) An affiliate or Facility Office of a Lender may be designated for the purposes of paragraph (a) above by:
 - (i) appearing in the list of Designated Entities in schedule 12 (List of Designated Entities) of this agreement and signing this agreement as a Designated Entity; or
 - (ii) acceding as a Designated Entity by signing an accession agreement substantially in the form of schedule 13 (Form of Designated Entity Accession Agreement).
- (c) A Designated Entity does not have any Commitment and does not have any obligations under this agreement prior to such Designated Entity participating in any Utilisation.
- (d) When a Designated Entity participates in any Utilisation:

- subject to paragraph (e) below, it shall be entitled to all the rights of a Lender and have the corresponding obligations of a Lender, in each case under the Finance Documents relating to its participation in any such Utilisation; and
- (ii) the other parties to the Finance Documents shall treat the Designated Entity as a Lender for these purposes.
- (e) The Designated Entity is a party to this agreement for these purposes.
- (f) For the purposes only of voting in connection with any Finance Document, the participation of a Designated Entity in any outstanding Utilisation shall be deemed to be a participation of the Related Lender.
- (g) Any notice or communication to be made to a Designated Entity shall be served directly on the Designated Entity at the address supplied to the Facility Agent by the Related Lender where the Related Lender or Designated Entity reasonably requests or, if no such request has been made, shall be delivered to the Related Lender in accordance with this agreement.
- (h) A Designated Entity may assign or transfer any of its rights and obligations under this agreement in respect of its participation in any Utilisation (and the Related Lender may assign or transfer any corresponding Commitment) in accordance with clause 29.1 (Assignments and Transfers by the Lenders).

6. UTILISATION - LETTERS OF CREDIT

6.1 The Revolving Facilities

- (a) Each Revolving Facility may be utilised by way of Letters of Credit.
- (b) Other than clause 5.5 (Limitations on Utilisations), clause 5 (Utilisation Loans) does not apply to utilisations by way of Letters of Credit.

6.2 Delivery of a Utilisation Request for Letters of Credit

A Borrower (or Obligors' Agent on its behalf) may request a Letter of Credit to be issued by delivery to the Facility Agent of a duly completed Utilisation Request in the form of part 1B of schedule 3 (Requests) not later than the Specified Time.

6.3 Completion of a Utilisation Request for Letters of Credit

Each Utilisation Request for a Letter of Credit is irrevocable and will not be regarded as having been duly completed unless:

- (a) it identifies the Revolving Facility to be utilised;
- (b) it specifies that it is for a Letter of Credit;
- (c) it identifies the Borrower of the Letter of Credit;
- (d) it identifies the Issuing Bank which has agreed to issue the Letter of Credit;
- (e) the proposed Utilisation Date is a Business Day within the Availability Period applicable to the Revolving Facility to be utilised;
- (f) the currency and amount of the Letter of Credit comply with clause 6.4 (Currency and Amount);
- (g) the form of Letter of Credit is attached;

- the Expiry Date of the Letter of Credit falls on or before the Termination Date in relation to the Revolving Facility to be utilised;
- (i) the Term of the Letter of Credit is 12 Months or less; provided that any Letter of Credit may provide, subject to clause (h) above, for the automatic extension thereof for additional periods of lengths not to exceed 12 Months so long as any such Letter of Credit provides that the Issuing Bank has the right to terminate such automatic extension on an annual basis by notice to the beneficiary of such Letter of Credit;
- (j) the delivery instructions for the Letter of Credit are specified; and
- (k) it identifies the beneficiary of the Letter of Credit and such beneficiary is a person to whom such Letter of Credit can be issued without causing the Issuing Bank to be in breach of any Economic Sanctions Law.

6.4 Currency and Amount

- (a) The currency specified in a Utilisation Request must be the Base Currency or an Optional Currency.
- (b) Subject to clause 5.5(a) (Limitations on Utilisations), the amount of the proposed Letter of Credit must be an amount whose Base Currency Amount is not more than the Available Facility, unless cash cover has been provided in the amount by which the Available Facility is exceeded, and which is:
 - (i) if the currency selected is the Base Currency, a minimum of \$100,000 or, if less, the Available Facility; or
 - (ii) if the currency selected is an Optional Currency, the minimum amount specified by the Facility Agent pursuant to clause 4.3(b)(ii) (Conditions Relating to Optional Currencies) or, if less, the Available Facility.

6.5 **Issue of Letters of Credit**

- (a) If the conditions set out in this agreement have been met, the Issuing Bank shall issue the Letter of Credit on the Utilisation Date.
- (b) The Issuing Bank will only be obliged to comply with paragraph (a) above if on the date of the Utilisation Request or Renewal Request and on the proposed Utilisation Date:
 - (i) in the case of a Letter of Credit to be renewed in accordance with clause 6.6 (Renewal of a Letter of Credit) no winding up has occurred in respect of the relevant Borrower and, in the case of any other Utilisation, no Default is continuing or would result from the proposed Utilisation; and
 - (ii) in relation to any Utilisation by way of issue of a Letter of Credit, the Repeating Representations to be made by each Obligor are true in all material respects.
- (c) The amount of each Lender's participation in each Letter of Credit will be equal to the proportion borne by its Available Commitment to the Available Facility (in each case in relation to the relevant Revolving Facility) immediately prior to the issue of the Letter of Credit.
- (d) The Facility Agent shall determine the Base Currency Amount of each Letter of Credit which is to be issued in an Optional Currency and shall notify the Issuing Bank and each Lender of the details of the requested Letter of Credit and its participation in that Letter of Credit by the Specified Time.

6.6 Renewal of a Letter of Credit

- (a) A Borrower (or Obligors' Agent on its behalf) may request that any Letter of Credit issued on behalf of that Borrower be renewed by delivery to the Facility Agent of a Renewal Request in substantially similar form to a Utilisation Request for a Letter of Credit by the Specified Time.
- (b) The Finance Parties shall treat any Renewal Request in the same way as a Utilisation Request for a Letter of Credit except that the conditions set out in clause 6.3(g) (Completion of a Utilisation Request for Letters of Credit) shall not apply.
- (c) The terms of each renewed Letter of Credit shall be the same as those of the relevant Letter of Credit immediately prior to its renewal, except that:
 - (i) its amount may be less than the amount of the Letter of Credit immediately prior to its renewal; and
 - (ii) its Term shall start on the date which was the Expiry Date of the Letter of Credit immediately prior to its renewal, and shall end on the proposed Expiry Date specified in the Renewal Request.
- (d) If the conditions set out in this agreement have been met, the Issuing Bank shall amend and re-issue any Letter of Credit pursuant to a Renewal Request.

6.7 **Revaluation of Letters of Credit**

- (a) If any Letters of Credit are denominated in an Optional Currency, the Facility Agent shall, at six monthly intervals after the date of the Letter of Credit, recalculate the Base Currency Amount of each Letter of Credit by notionally converting into the Base Currency the outstanding amount of that Letter of Credit on the basis of the Facility Agent's Spot Rate of Exchange on the date of calculation.
- (b) Obligors' Agent shall, if requested by the Facility Agent within five days of any calculation under paragraph (a) above, ensure that, in relation to a Revolving Facility, within three Business Days sufficient Revolving Facility Utilisations are prepaid to prevent the Base Currency Amount of the Revolving Facility Utilisations exceeding the Total Revolving Facility Commitments (after deducting the total Ancillary Commitments) following any adjustment to a Base Currency Amount under paragraph (a) above.

7. **LETTERS OF CREDIT**

7.1 Immediately Payable

If a Letter of Credit or any amount outstanding under a Letter of Credit is expressed to be immediately payable, the Borrower that requested (or on behalf of which Obligors' Agent requested) the issue of that Letter of Credit shall repay or prepay that amount within 3 Business Days (subject to clause 7.6 (Conversion of Letter of Credit Liability)).

7.2 Claims under a Letter of Credit

- (a) Each Borrower and each Lender irrevocably and unconditionally authorises the Issuing Bank to pay any claim made or purported to be made under a Letter of Credit requested by that Borrower (or requested by Obligors' Agent on its behalf) and which appears on its face to be in order (in this clause 7, a "claim").
- (b) Each Borrower shall within three Business Days of demand pay to the Facility Agent for the Issuing Bank an amount equal to the amount of any claim (plus interest thereon at the Issuing Bank's cost of funds thereon from the date of payment by the

Issuing Bank to the date of its reimbursement (subject to clause 7.6 (Conversion of Letter of Credit Liability).

- (c) Each Borrower and each Lender acknowledges that the Issuing Bank:
 - is not obliged to carry out any investigation or seek any confirmation from any other person before paying a claim; and
 - (ii) deals in documents only and will not be concerned with the legality of a claim or any underlying transaction or any available set-off, counterclaim or other defence of any person.
- (d) The obligations of a Borrower and each Lender under this clause will not be affected by:
 - (i) the sufficiency, accuracy or genuineness of any claim or any other document; or
 - (ii) any incapacity of, or limitation on the powers of, any person signing a claim or other document.

7.3 Indemnities

- (a) Each Borrower shall within five Business Days indemnify the Issuing Bank against any cost, loss or liability incurred by the Issuing Bank (otherwise than by reason of the Issuing Bank's breach of this clause 7, gross negligence, bad faith or wilful misconduct as determined by a court of competent jurisdiction by final and nonappealable judgment) in acting as the Issuing Bank under any Letter of Credit requested by (or on behalf of) that Borrower.
- (b) Each Lender shall (according to its L/C Proportion) immediately on demand indemnify the Issuing Bank against any cost, loss or liability incurred by the Issuing Bank (otherwise than by reason of the Issuing Bank's gross negligence, bad faith or wilful misconduct as determined by a court of competent jurisdiction by final and nonappealable judgment) in acting as the Issuing Bank under any Letter of Credit (unless the Issuing Bank has been reimbursed by an Obligor pursuant to a Finance Document).
- (c) If any Lender is not permitted (by its constitutional documents or any applicable law) to comply with paragraph (b) above, then that Lender will not be obliged to comply with paragraph (b) and shall instead be deemed to have taken, on the date the Letter of Credit is issued (or if later, on the date the Lender's participation in the Letter of Credit is transferred or assigned to the Lender in accordance with the terms of this agreement), an undivided interest and participation in the Letter of Credit in an amount equal to its L/C Proportion of that Letter of Credit. On receipt of demand from the Facility Agent, that Lender shall pay to the Facility Agent (for the account of the Issuing Bank) an amount equal to its L/C Proportion of the amount demanded.
- (d) The Borrower which requested (or on behalf of which Obligors' Agent requested) a Letter of Credit shall immediately on demand reimburse any Lender for any payment it makes to the Issuing Bank under this clause 7.3 in respect of that Letter of Credit.
- (e) The obligations of each Borrower and each Lender under this clause are continuing obligations and will extend to the ultimate balance of sums payable by that Lender in respect of any Letter of Credit, regardless of any intermediate payment or discharge in whole or in part.
- (f) If a Borrower has provided cash cover in respect of a Lender's participation in a Letter of Credit, the Issuing Bank shall seek reimbursement from that cash cover before

- making a demand of that Lender under paragraph (b) above. Any recovery made by an Issuing Bank pursuant to that cash cover will reduce that Lender's liability under paragraph (b) above.
- (g) The obligations of any Lender or Borrower under this clause will not be affected by any act, omission, matter or thing which, but for this clause, would reduce, release or prejudice any of its obligations under this clause (including without limitation and whether or not known to it or any other person):
 - any time, waiver or consent granted to, or composition with, any Obligor, any beneficiary under a Letter of Credit or any other person;
 - (ii) the release of any other Obligor or any other person under the terms of any composition or arrangement with any creditor or any Group Company;
 - (iii) the taking, variation, compromise, exchange, renewal or release of, or refusal or neglect to perfect, take up or enforce, any rights against, or security over assets of, any Obligor, any beneficiary under a Letter of Credit or other person or any non-presentation or non-observance of any formality or other requirement in respect of any instrument or any failure to realise the full value of any security;
 - (iv) any incapacity or lack of power, authority or legal personality of or dissolution or change in the members or status of an Obligor, any beneficiary under a Letter of Credit or any other person;
 - any amendment (however fundamental) or replacement of a Finance Document, any Letter of Credit or any other document or security;
 - (vi) any unenforceability, illegality or invalidity of any obligation of any person under any Finance Document, any Letter of Credit or any other document or security; or
 - (vii) any insolvency or similar proceedings.

7.4 Cash Collateral by Non-Acceptable L/C Lender

- (a) If, at any time, a Lender under a Revolving Facility is a Non-Acceptable L/C Lender, the Issuing Bank may, by notice to that Lender, request that Lender to pay and that Lender shall pay, on or prior to the date falling five Business Days after the request by the Issuing Bank, an amount equal to that Lender's L/C Proportion of the outstanding amount of a Letter of Credit and in the currency of that Letter of Credit to an interest-bearing account held in the name of that Lender with the Issuing Bank.
- (b) The Non-Acceptable L/C Lender to whom a request has been made in accordance with paragraph (a) above shall enter into a security document or other form of collateral arrangement over the account, in form and substance satisfactory to the Issuing Bank, as collateral for any amounts due and payable under the Finance Documents by that Lender to the Issuing Bank in respect of that Letter of Credit.
- (c) Subject to paragraph (f) below, until no amount is or may be outstanding under that Letter of Credit, withdrawals from the account may only be made to pay to the Issuing Bank amounts due and payable to the Issuing Bank by the Non-Acceptable L/C Lender under the Finance Documents in respect of that Letter of Credit.
- (d) Each Lender under a Revolving Facility shall notify the Facility Agent and the Obligors' Agent:

- (i) on the 2022 Effective Date or on any later date on which it becomes such a Lender in accordance with clause 29 (Changes to the Lenders) whether it is a Non-Acceptable L/C Lender within paragraph (a) of the definition of "Non-Acceptable L/C Lender"; and
- (ii) as soon as practicable upon becoming aware of the same, that it has become a Non-Acceptable L/C Lender.

and an indication in schedule 1 (The Original Parties) or in a Transfer Certificate to that effect will constitute a notice under paragraph (d)(i) to the Facility Agent and, upon delivery in accordance with clause 29.6 (Copy of Transfer Certificate to Obligors' Agent), to Obligors' Agent.

- (e) Any notice received by the Facility Agent pursuant to paragraph (d) above shall constitute notice to the Issuing Bank of that Lender's status and the Facility Agent shall, upon receiving each such notice, promptly notify the Issuing Bank of that Lender's status as specified in that notice.
- (f) If a Lender who has provided cash collateral in accordance with this clause 7.4:
 - (i) ceases to be a Non-Acceptable L/C Lender; and
 - (ii) no amount is due and payable by that Lender in respect of a Letter of Credit,

that Lender may, at any time it is not a Non-Acceptable L/C Lender, by notice to the Issuing Bank request that an amount equal to the amount of the cash provided by it as collateral in respect of that Letter of Credit (together with any accrued interest) standing to the credit of the relevant account held with the Issuing Bank be returned to it and the Issuing Bank shall pay that amount to the Lender within three Business Days after the request from the Lender (and shall cooperate with the Lender in order to procure that the relevant security or collateral arrangement is released and discharged).

7.5 Rights of Contribution

No Obligor will be entitled to any right of contribution or indemnity from any Finance Party in respect of any payment it may make under this clause 7.

7.6 Conversion of Letter of Credit Liability

- (a) The relevant Issuing Bank shall as soon as reasonably practicable after receipt by it of any claim notify Obligors' Agent and the relevant Borrower for whose account the relevant Letter of Credit was issued specifying:
 - the latest date on which payment may be made in respect of the claim (the "Payment Date"); and
 - (ii) the amount of the claim (the "Claimed Amount").
- (b) The relevant Borrower may, but shall not be obliged to, without prejudice to paragraph (c) below and the Obligors' obligations under this agreement, not later than 9.30 a.m. on the Business Day preceding the Payment Date, pay or procure the payment to the Issuing Bank of the Claimed Amount.
- (c) Upon payment of any Claimed Amount by the Issuing Bank, the Borrower for whose account the relevant Letter of Credit was issued shall be deemed to have issued a Utilisation Request in respect of a Revolving Facility Loan in the currency of and an amount equal to that Claimed Amount (which shall be a permitted amount for the purposes of clause 5.3 (Currency and Amount)) and the Utilisation Date for such

Revolving Facility Loan shall be whichever is the later of the Specified Time referred to in clause 5.1 (Delivery of a Utilisation Request) for that Revolving Facility Loan and the relevant Payment Date and the Interest Period for such Revolving Facility Loan (if a Term Benchmark Loan) shall be one month.

8. **OPTIONAL CURRENCIES**

8.1 Selection of Currency

A Borrower (or Obligors' Agent on behalf of a Borrower) shall select the currency of a Utilisation of a Revolving Facility in a Utilisation Request.

8.2 Unavailability of a Currency

If before the Specified Time on any Quotation Day:

- (a) a Lender notifies the Facility Agent that the Optional Currency requested is not readily available to it in the amount required; or
- (b) a Lender notifies the Facility Agent that compliance with its obligation to participate in a Loan in the proposed Optional Currency would contravene a law or regulation applicable to it,

the Facility Agent will give notice to the relevant Borrower or Obligors' Agent to that effect by the Specified Time on that day. In this event, any Lender that gives notice pursuant to this clause 8.2 will be required to participate in the Loan in the Base Currency (in an amount equal to that Lender's proportion of the Base Currency Amount, or in respect of a Rollover Loan, an amount equal to that Lender's proportion of the Base Currency Amount of the Rollover Loan that is due to be made) and its participation will be treated as a separate Loan denominated in the Base Currency during that Interest Period (if applicable).

8.3 Facility Agent's Calculations

- (a) All calculations made by the Facility Agent pursuant to this clause 8 will take into account any repayment, prepayment, consolidation or division of Term Loans to be made on the first Interest Payment Date.
- (b) Each Lender's participation in a Loan will, subject to paragraph (a) above, be determined in accordance with clause 5.4(b) (Lenders' Participation).

9. **ANCILLARY FACILITIES**

9.1 Type of Facility

An Ancillary Facility may be by way of:

- (a) an overdraft facility;
- (b) a guarantee, bonding, documentary or stand-by letter of credit facility;
- (c) a short term loan facility;
- (d) a derivatives facility;
- (e) a foreign exchange facility; or
- (f) any other facility or accommodation required in connection with the business of the Group and which is agreed by Obligors' Agent with an Ancillary Lender.

9.2 Availability

- (a) If Obligors' Agent and a Lender agree and except as otherwise provided in this agreement, the Lender may provide an Ancillary Facility on a bilateral basis in place of all or part of that Lender's unutilised Revolving Facility Commitment.
- (b) An Ancillary Facility (other than the Existing Ancillary Facility which has been automatically deemed made available as an Ancillary Facility to Facility RCF pursuant to the 2022 Amendment Agreement) shall not be made available unless, not later than 5 Business Days prior to the Ancillary Commencement Date for an Ancillary Facility, the Facility Agent has received from Obligors' Agent:
 - (i) a notice in writing requesting the establishment of an Ancillary Facility and specifying:
 - (A) the Revolving Facility from which the Ancillary Facility will be established;
 - (B) the proposed Borrowers (or Affiliates of a Borrower) which may use the Ancillary Facility;
 - (C) the proposed Ancillary Commencement Date and expiry date of the Ancillary Facility;
 - (D) the proposed type of Ancillary Facility to be provided;
 - (E) the proposed Ancillary Lender;
 - (F) the proposed Ancillary Commitment, the maximum amount of the Ancillary Facility (if not denominated in the Base Currency) and, if the Ancillary Facility is an overdraft facility comprising more than one account its maximum gross amount (that amount being the "Designated Gross Amount") and its maximum net amount (that amount being the "Designated Net Amount"); and
 - (G) the proposed currency of the Ancillary Facility (if not denominated in the Base Currency);
 - (ii) a copy of the proposed Ancillary Document; and
 - (iii) any other information which the Facility Agent may reasonably request in connection with the Ancillary Facility.

The Facility Agent shall promptly notify Obligors' Agent, the Ancillary Lender and the other Lenders of the establishment of an Ancillary Facility.

No amendment or waiver of a term of any Ancillary Facility shall require the consent of any Finance Party other than the relevant Ancillary Lender unless such amendment or waiver itself relates to or gives rise to a matter which would require an amendment of or under this agreement (including, for the avoidance of doubt, under this clause). In such a case, the provisions of this agreement with regard to amendments and waivers will apply.

- (c) Subject to compliance with paragraph (b) above:
 - (i) the Lender concerned will become an Ancillary Lender; and
 - (ii) the Ancillary Facility will be available,

with effect from the date agreed by Obligors' Agent and the Ancillary Lender.

9.3 Terms of Ancillary Facilities

- (a) Except as provided below, the terms of any Ancillary Facility will be those agreed by the Ancillary Lender and Obligors' Agent.
- (b) However, those terms:
 - must be based upon normal commercial terms at that time (except as varied by this agreement);
 - (ii) may allow only Borrowers (or Affiliates of Borrowers nominated pursuant to clause 9.9 (Affiliates of Borrowers)) to use the Ancillary Facility;
 - (iii) may not allow the Ancillary Outstandings to exceed the Ancillary Commitment;
 - (iv) may not, otherwise than as a result solely of currency fluctuations, allow the Ancillary Commitment of a Lender to exceed the Available Commitment of that Lender under the relevant Revolving Facility (before taking into account the effect of the Ancillary Facility on that Available Commitment); and
 - (v) must require that the Ancillary Commitment is reduced to nil, and that all Ancillary Outstandings are repaid (or cash cover provided in respect of all the Ancillary Outstandings) not later than the Termination Date for the relevant Revolving Facility (or such earlier date as the applicable Revolving Facility Commitment of the relevant Ancillary Lender (or its Affiliate) is reduced to zero).
- (c) If there is any inconsistency between any term of an Ancillary Facility and any term of this agreement, this agreement shall prevail except for:
 - (i) clause 37.3 (Day Count Convention) which shall not prevail for the purposes of calculating fees, interest or commission relating to an Ancillary Facility; and
 - (ii) an Ancillary Facility comprising more than one account where the terms of the Ancillary Documents shall prevail.
- (d) Interest, commission and fees on Ancillary Facilities are dealt with in clause 17.5 (Interest, Commission and Fees on Ancillary Facilities).

9.4 Repayment of Ancillary Facility

- (a) An Ancillary Facility shall cease to be available on the Termination Date in relation to the applicable Revolving Facility or such earlier date on which its expiry date occurs or on which it is cancelled in accordance with the terms of this agreement.
- (b) If an Ancillary Facility expires in accordance with its terms the Ancillary Commitment of the Ancillary Lender shall be reduced to zero.
- (c) No Ancillary Lender may demand repayment or prepayment of any amounts or demand cash cover for any liabilities made available or incurred by it under its Ancillary Facility (except where the Ancillary Facility is provided on a net limit basis to the extent required to bring any gross outstandings down to the net limit) unless:
 - (i) the Total Revolving Facility Commitments under the relevant Revolving Facility have been cancelled in full, or all outstanding Utilisations under that Revolving Facility have become due and payable in accordance with the terms

- of this agreement, or the Facility Agent has declared all outstanding Utilisations under that Revolving Facility immediately due and payable, or the expiry date of the Ancillary Facility occurs; or
- it becomes unlawful in any applicable jurisdiction for the Ancillary Lender (or its Affiliate, if relevant) to perform any of its obligations as contemplated by this agreement or to fund, issue or maintain its participation in its Ancillary Facility; or
- (iii) the Ancillary Outstandings (if any) under that Ancillary Facility can be refinanced by a Revolving Facility Utilisation under the same Revolving Facility and the Ancillary Lender gives sufficient notice to enable a Utilisation of that Revolving Facility to be made to refinance those Ancillary Outstandings.
- (d) For the purposes of determining whether or not the Ancillary Outstandings under an Ancillary Facility mentioned in paragraph (c)(iii) above can be refinanced by a Utilisation of a Revolving Facility:
 - (i) the Revolving Facility Commitment of the Ancillary Lender under that Revolving Facility will be increased by the amount of its Ancillary Commitment; and
 - (ii) the Utilisation may (so long as paragraph (c)(i) above does not apply) be made irrespective of whether a Default is outstanding or any other applicable condition precedent is not satisfied (but only to the extent that the proceeds are applied in refinancing those Ancillary Outstandings) and irrespective of whether clause 4.4 (Maximum Number of Utilisations) or clause 5.2(a)(iii) (Completion of a Utilisation Request for Loans) applies.
- (e) On the making of a Utilisation of a Revolving Facility to refinance Ancillary Outstandings:
 - (i) each Lender will participate in that Utilisation in an amount (as determined by the Facility Agent) which will result as nearly as possible in the aggregate amount of its participation in Utilisations then outstanding bearing the same proportion to the aggregate amount of the Utilisations then outstanding as its Revolving Facility Commitment bears to the aggregate of the Revolving Facility Commitments, in each case, under that Revolving Facility; and
 - (ii) the relevant Ancillary Facility shall be cancelled.
- (f) In relation to an Ancillary Facility which comprises an overdraft facility where a Designated Net Amount has been established, the Ancillary Lender providing that Ancillary Facility shall only be obliged to take into account for the purposes of calculating compliance with the Designated Net Amount those credit balances which it is permitted to take into account by the then current law and regulations in relation to its reporting of exposures to applicable regulatory authorities as netted for capital adequacy purposes.
- (g) In relation to an Ancillary Facility which comprises either a single account or multiple account overdraft facility, any credit balances held by the relevant Ancillary Lender in relation to that Ancillary Facility will not be repayable to the extent that repayment would cause any Designated Net Amount to be exceeded.

9.5 **Ancillary Outstandings**

Each Borrower and each Ancillary Lender agrees with and for the benefit of each Lender that:

- (a) the Ancillary Outstandings under any Ancillary Facility provided by that Ancillary Lender shall not exceed the Ancillary Commitment applicable to that Ancillary Facility and where the Ancillary Facility is an overdraft facility comprising more than one account, Ancillary Outstandings under that Ancillary Facility shall not exceed the Designated Net Amount in respect of that Ancillary Facility; and
- (b) where all or part of the Ancillary Facility is an overdraft facility comprising more than one account, the Ancillary Outstandings (calculated on the basis that the words in brackets in paragraph (a) of the definition of that term were deleted) shall not exceed the Designated Gross Amount applicable to that Ancillary Facility.

9.6 Adjustment for Ancillary Facilities upon Acceleration

In this clause 9.6:

"Revolving Outstandings" means, in relation to a Lender and a Revolving Facility, the aggregate of the equivalent in the Base Currency of:

- (a) its participation in each Utilisation under that Revolving Facility then outstanding (together with the aggregate amount of all accrued interest, fees and commission owed to it as a Lender under that Revolving Facility); and
- (b) if the Lender is also an Ancillary Lender, the Ancillary Outstandings in respect of Ancillary Facilities provided by that Ancillary Lender under that Revolving Facility (together with the aggregate amount of all accrued interest, fees and commission owed to it as an Ancillary Lender in respect of the Ancillary Facility); and

"Total Revolving Outstandings" means the aggregate of all Revolving Outstandings under a Revolving Facility.

- (a) If a notice is served under clause 28.22 (Acceleration) (other than a notice declaring Utilisations to be due on demand), each Lender and each Ancillary Lender shall promptly adjust by corresponding transfers (to the extent necessary) their claims in respect of amounts outstanding to them under any Revolving Facility and each Ancillary Facility granted thereunder to ensure that after such transfers the Revolving Outstandings of each Lender bear the same proportion to the Total Revolving Outstandings as such Lender's Revolving Facility Commitment bears to the Total Revolving Facility Commitments, each as at the date the notice is served under clause 28.22 (Acceleration).
- (b) If an amount outstanding under an Ancillary Facility is a contingent liability and that contingent liability becomes an actual liability or is reduced to zero after the original adjustment is made under paragraph (a) above, then each Lender and Ancillary Lender will make a further adjustment by corresponding transfers (to the extent necessary) to put themselves in the position they would have been in had the original adjustment been determined by reference to the actual liability or, as the case may be, zero liability and not the contingent liability.
- (c) Any transfer of rights and obligations relating to Revolving Outstandings made pursuant to this Clause 9.6 shall be made for a purchase price in cash, payable at the time of transfer, in an amount equal to those Revolving Outstandings.
- (d) Prior to the application of the provisions of paragraph (a) above, an Ancillary Lender that has provided an overdraft comprising more than one account under an Ancillary Facility shall set off any liabilities owing to it under such overdraft facility against credit balances on any account comprised in such overdraft facility.
- (e) All calculations to be made pursuant to this clause 9.6 shall be made by the Facility Agent based on upon information provided to it by the Lenders and Ancillary Lenders.

9.7 Information

Each Borrower and each Ancillary Lender shall, promptly upon request by the Facility Agent, supply the Facility Agent with any information relating to the operation of an Ancillary Facility (including the Ancillary Outstandings) as the Facility Agent may reasonably request from time to time. Each Borrower consents to all such information being released to the Facility Agent and the other Finance Parties.

9.8 Affiliates of Lenders as Ancillary Lenders

- (a) Subject to the terms of this agreement, an Affiliate of a Lender may become an Ancillary Lender. In such case, the Lender and its Affiliate shall be treated as a single Lender having a Revolving Facility Commitment of the relevant Lender. For the purposes of calculating the Lender's Available Commitment with respect to a Revolving Facility, the Lender's Commitment shall be reduced to the extent of the aggregate of the Ancillary Commitments of its Affiliates in relation to that Revolving Facility.
- (b) Obligors' Agent shall specify any relevant Affiliate of a Lender in any notice delivered by Obligors' Agent to the Facility Agent pursuant to clause 9.2(b)(i) (Availability).
- (c) An Affiliate of a Lender which becomes an Ancillary Lender shall accede to this agreement and the Intercreditor Agreement by delivery to the Security Agent of a duly completed accession undertaking in the form scheduled to each Intercreditor Agreement.
- (d) If a Lender assigns all of its rights and benefits or transfers all of its rights and obligations to a New Lender (as defined in clause 29 (Changes to the Lenders)), its Affiliate shall cease to have any obligations under this agreement or any Ancillary Document.
- (e) Where this agreement or any other Finance Document imposes an obligation on an Ancillary Lender and the relevant Ancillary Lender is an Affiliate of a Lender which is not a party to that document, the relevant Lender shall ensure that the obligation is performed by its Affiliate.

9.9 Affiliates of Borrowers

- (a) Subject to the terms of this agreement, an Affiliate of a Borrower (other than any Affiliate incorporated in Belgium or Switzerland) may with the approval of the relevant Lender become a borrower with respect to an Ancillary Facility.
- (b) Obligors' Agent shall specify any relevant Affiliate of a Borrower in any notice delivered by Obligors' Agent to the Facility Agent pursuant to clause 9.2(b)(i) (Availability).
- (c) If a Borrower ceases to be a Borrower under this agreement in accordance with clause 30.3 (Resignation of a Borrower), its Affiliate shall cease to have any rights under this agreement or any Ancillary Document.
- (d) Where this agreement or any other Finance Document imposes an obligation on a Borrower under an Ancillary Facility and the relevant Borrower is an Affiliate of a Borrower which is not a party to that document, the relevant Borrower shall ensure that the obligation is performed by its Affiliate.
- (e) Any reference in this agreement or any other Finance Document to a Borrower being under no obligations (whether actual or contingent) as a Borrower under such Finance Document shall be construed to include a reference to any Affiliate of a

Borrower being under no obligations under any Finance Document or Ancillary Document.

9.10 Revolving Facility Commitment Amounts

Notwithstanding any other term of this agreement, in relation to each Revolving Facility, each Lender shall ensure that at all times its Revolving Facility Commitment is not less than:

- (a) its Ancillary Commitment; or
- (b) the Ancillary Commitment of its Affiliate.

10. **REPAYMENT**

10.1 Repayment of Term Loans

(a) The Borrower under the Term Facility A shall repay the Term Facility A Loans outstanding under the Term Facility A on the dates set forth below an amount equal to the product of the aggregate initial principal amount of Term Facility A Loans which have been made prior to such date multiplied by the amortization percentage set forth opposite such date (subject to adjustment for any prepayment made pursuant to the terms of this agreement and provided that any Term Facility A Loan made after any such date shall only be taken into account for repayment instalments after the date on which such Term Facility A loan was made):

<u>Date</u>	<u>Amount</u>
September 30, 2024	0.625%
December 31, 2024	0.625%
March 31, 2025	0.625%
June 30, 2025	0.625%
September 31, 2025	1.25%
December 31, 2025	1.25%
March 31, 2026	1.25%
June 30, 2026	1.25%
September 30, 2026	1.25%
December 31, 2026	1.25%
March 31, 2027	1.25%
June 30, 2027	1.25%
September 30, 2027	1.875%
December 30, 2027	1.875%
March 31, 2028	1.875%
June 30, 2028	1.875%
September 30, 2028	2.50%
December 31, 2028	2.50%
March 31, 2029	2.50%
June 30, 2029	2.50%
Termination Date for the Term Facility A Loan	Remaining outstanding amount

- (b) The Borrower under the Term Facility B shall repay the Term Facility B Loans on the Termination Date for the Term Facility B.
- (c) The Borrowers under each Incremental Term Facility shall repay each Incremental Term Facility Loan borrowed by it under that Incremental Term Facility in the amounts and on the dates agreed between the Borrower and the relevant Lenders thereof in accordance with clause 2.2 (Incremental Facilities).
- (d) The Borrowers may not reborrow any part of a Term Facility which is repaid.

10.2 Repayment of Revolving Facility Loans

- (a) Each Borrower which has drawn a Revolving Facility Loan shall repay that Loan on the last day of its Interest Period or in relation to a RFR Loan, its next Interest Payment Date.
- (b) Without prejudice to each Borrower's obligation under paragraph (a) above, if:
- (i) one or more Revolving Facility Loans are to be made available to a Borrower:
- (A) on the same day that a maturing Revolving Facility Loan is due to be repaid by that Borrower;
- (B) in the same currency as the maturing Revolving Facility Loan; and
- (C) in whole or in part for the purpose of refinancing the maturing Revolving Facility Loan; and
- (ii) the proportion borne by each Lender's participation in the maturing Revolving Facility Loan to the amount of that maturing Revolving Facility Loan is the same as the proportion borne by that Lender's participation in the new Revolving Facility Loans to the aggregate amount of those new Revolving Facility Loans, the aggregate amount of the new Revolving Facility Loans shall, unless the relevant Borrower notifies the Facility Agent to the contrary in the relevant Utilisation Request, be treated as if applied in or towards repayment of the maturing Revolving Facility Loan so that:
- (A) if the amount of the maturing Revolving Facility Loan exceeds the aggregate amount of the new Revolving Facility Loans:
- (1) the relevant Borrower will only be required to make a payment under Clause 34.1 (Payments to the Facility Agent) in an amount in the relevant currency equal to that excess; and
- (2) each Lender's participation in the new Revolving Facility Loans shall be treated as having been made available and applied by the Borrower in or towards repayment of that Lender's participation in the maturing Revolving Facility Loans and that Lender will not be required to make a payment under Clause 34.1 (Payments to the Facility Agent) in respect of its participation in the new Revolving Facility Loans; and
- (B) if the amount of the maturing Revolving Facility Loan is equal to or less than the aggregate amount of the new Revolving Facility Loans:
- (1) the relevant Borrower will not be required to make a payment under Clause 34.1 (Payments to the Facility Agent); and
- (2) each Lender will be required to make a payment under Clause 34.1 (Payments to the Facility Agent) in respect of its participation in the new Revolving Facility Loans only to the extent that its participation in the new Revolving Facility Loans exceeds that Lender's participation in the maturing Revolving Facility Loan and the remainder of that Lender's participation in the new Revolving Facility Loans shall be treated as having been made available and applied by the Borrower in or towards repayment of that Lender's participation in the maturing Revolving Facility Loan.

11. ILLEGALITY, VOLUNTARY PREPAYMENT AND CANCELLATION

11.1 Illegality

If it becomes unlawful in any applicable jurisdiction for a Lender (or its Affiliate which is an Ancillary Lender or Designated Entity, if relevant) to perform any of its obligations as

contemplated by this agreement or to fund, issue or maintain its participation in any Utilisation:

- (a) that Lender shall promptly notify the Facility Agent upon becoming aware of that event;
- (b) upon the Facility Agent notifying Obligors' Agent, the Commitment of that Lender will be immediately cancelled; and
- (c) each Borrower shall repay that Lender's participation in the Utilisations made to that Borrower on the last day of the Interest Period (or with respect to a RFR Loan, the next Interest Payment Date) for each Utilisation occurring after the Facility Agent has notified Obligors' Agent or, if earlier, the date specified by the Lender in the notice delivered to the Facility Agent (being no earlier than the last day of any applicable grace period permitted by law).

11.2 Illegality in Relation to Issuing Bank

If it becomes unlawful for an Issuing Bank to issue or leave outstanding any Letter of Credit, then:

- (a) that Issuing Bank shall promptly notify the Facility Agent upon becoming aware of that event;
- (b) upon the Facility Agent notifying Obligors' Agent, the Issuing Bank shall not be obliged to issue any Letter of Credit;
- (c) Obligors' Agent shall procure that the relevant Borrower shall use its best endeavours to procure the release of each Letter of Credit issued by that Issuing Bank and outstanding at such time; and
- (d) unless any other Lender has agreed to be an Issuing Bank pursuant to the terms of this agreement, the Revolving Facilities shall cease to be available for the issue of Letters of Credit.

11.3 Voluntary Cancellation

(a) Obligors' Agent may, if it gives the Facility Agent not less than three Business Days' (or such shorter period as the Majority Lenders may agree) prior notice, cancel the whole or any part (being a minimum amount of \$25,000,000) of an Available Facility; provided that a notice of cancellation may state that such notice is conditioned upon the occurrence or non-occurrence of any event specified therein (including the effectiveness of other credit facilities), in which case such notice may be revoked (by notice to the Facility Agent on or prior to the specified effective date) if such condition is not satisfied. Any cancellation under this clause 11.3 shall reduce the Commitments of the Lenders rateably under that Facility.

11.4 Voluntary Prepayment of Term Loans

(a) Subject to paragraphs (b) and (c) below, a Borrower to which a Term Loan has been made may, if it or Obligors' Agent gives the Facility Agent not less than five Business Days' (or such shorter period as the Majority Lenders may agree) prior notice, prepay the whole or any part of that Term Loan (but, if in part, being an amount that reduces the Base Currency Amount of that Term Loan by a minimum amount of \$10,000,000); provided that a notice of prepayment delivered by such Borrower may state that such notice is conditioned upon the occurrence or non-occurrence of any event specified therein (including the effectiveness of other credit facilities), in which case such notice may be revoked by such Borrower (by notice to the Facility Agent on or prior to the specified effective date) if such condition is not satisfied. Promptly

following receipt of any such notice relating to a Loan, the Facility Agent shall advise the applicable Lenders of the contents thereof.

- (b) A Term Loan may only be prepaid after the last day of the Availability Period applicable to the relevant Term Facility (or, if earlier, the day on which the applicable Available Facility is zero).
- (c) Obligors' Agent may select in its absolute discretion the Loan(s) against which prepayments made under this clause 11.4 shall be applied and may select which remaining instalments of the Term Loans under the applicable Class or Classes such prepayments under this clause 11.4 shall be applied against (if no selection is made then such prepayment shall be applied on remaining instalments in direct order).

11.5 Voluntary Prepayment of Revolving Facility Utilisations

A Borrower to which a Revolving Facility Utilisation has been made may, if it or Obligors' Agent gives the Facility Agent not less than five Business Days' (or such shorter period as the Majority Lenders may agree) prior notice, prepay the whole or any part of a Revolving Facility Utilisation (but if in part, being an amount that reduces the Base Currency Amount of the Revolving Facility Utilisation by a minimum amount of \$5,000,000); provided that a notice of prepayment delivered by such Borrower may state that such notice is conditioned upon the occurrence or non-occurrence of any event specified therein (including the effectiveness of other credit facilities), in which case such notice may be revoked by such Borrower (by notice to the Facility Agent on or prior to the specified effective date) if such condition is not satisfied.

11.6 Right of Cancellation and Repayment in Relation to a Single Lender or Issuing Bank

- (a) If:
 - (i) any sum payable to any Lender by an Obligor is required to be increased under clause 18.2(c) (Tax Gross-Up); or
 - (ii) any Lender or Issuing Bank claims indemnification from Obligors' Agent or an Obligor under clause 18.3 (Tax Indemnity) or clause 19.1 (Increased Costs),

Obligors' Agent may, whilst the circumstance giving rise to the requirement for indemnification continues, give the Facility Agent notice:

- (iii) (if such circumstances relate to a Lender) of cancellation of the Commitment of that Lender and its intention to procure the repayment of that Lender's participation in the Utilisations; or
- (iv) (if such circumstances relate to the Issuing Bank) of repayment of any outstanding Letter of Credit issued by it and cancellation of its appointment as an Issuing Bank under this agreement in relation to any Letters of Credit to be issued in the future,
- (v) or may, if such circumstances relate to a Lender, instead exercise its rights under clause 40.3 (Replacement of Lender).
- (b) On receipt of a notice referred to in paragraph (a) above in relation to a Lender, the Commitment of that Lender shall immediately be reduced to zero.
- (c) On the last day of each Interest Period (or with respect to a RFR Loan, the next Interest Payment Date) which ends after Obligors' Agent has given notice under paragraph (a) above in relation to a Lender (or, if earlier, the date specified by Obligors' Agent in that notice), each Borrower to which a Utilisation is outstanding

shall repay that Lender's participation in that Utilisation together with all interest and other amounts accrued under the Finance Documents.

12. MANDATORY PREPAYMENT

12.1 Exit

Upon the occurrence of the sale of all or substantially all of the assets of the Group by the Group to another person (other than a Group Company) whether in a single transaction or a series of related transactions, the Facilities will be cancelled and all outstanding Utilisations and Ancillary Outstandings, together with accrued interest, and all other amounts accrued under the Finance Documents, shall become immediately due and payable.

12.2 Disposal and Insurance Proceeds and Excess Cashflow

(a) For the purposes of this clause 12:

"Disposal" means a sale, lease, licence, transfer, loan or other disposal by a person of any asset, undertaking or business (whether by a voluntary or involuntary single transaction or series of transactions).

"Disposal Proceeds" means the cash receivable by any Group Company (including any amount receivable in repayment of intercompany debt) for any Disposal made by any Group Company received except for Excluded Disposal Proceeds and after deducting:

- (i) any reasonable expenses which are incurred by any Group Company with respect to that Disposal to persons who are not Group Companies (including legal expenses, commission payable to agents, bonus payments to which the management are contractually entitled in connection with the Disposal, auditors' fees and closure, survey costs, title insurance premiums, investment banking fees, restructuring and reorganisation costs incurred in preparation for or as a consequence of the relevant Disposal);
- (ii) any Tax incurred and required to be paid by a Group Company in connection with that Disposal including any Tax incurred in the transfer of the Disposal Proceeds intra-Group (as reasonably determined by the seller, on the basis of existing rates and taking account of any available credit, deduction or allowance);
- (iii) any Financial Indebtedness which is to be repaid to the undertaking or business the subject of the Disposal by a Group Company;
- (iv) any Financial Indebtedness which is secured on the assets the subject of the Disposal and which is to be prepaid from the proceeds of the relevant Disposal;
- any repayments of Additional First Lien Debt (limited to its proportionate share of such prepayment, based on the principal amount of such then outstanding debt as a percentage of the aggregate principal amount of all then outstanding Term Loans and Additional First Lien Debt); and
- (vi) any provisions in the accounts for liability in connection with the relevant Disposal.

 $\hbox{\bf "Excluded Disposal Proceeds"} \ \hbox{means the proceeds derived from any Disposal:}$

(i) arising as a result of a Permitted Intercompany Debt Disposal, Permitted Disposal identified in any of paragraphs (a), (b), (c), (f), (g), (h), (i), (l), (m),

- (q) and (s) of the definition of Permitted Disposal or arising as a result of a Permitted Share Issue; or
- (ii) which are reinvested, contractually committed to be reinvested or designated by the board of directors of Obligors' Agent for reinvestment in either:
 - (A) other assets;
 - (B) Permitted Acquisitions or Permitted Loans; or
 - (C) Capital Expenditure,

within 365 days (or 180 days following the end of such 365 day period if such cash proceeds have been contractually committed to be used within such 365 day period) after the Disposal occurs (it being understood that if any portion of such cash proceeds are not so used within such 365 day period or within 180 days following the end of such 365 day period shall constitute Insurance Proceeds as of such date).

"Excluded Insurance Proceeds" means any proceeds of an insurance claim which Obligors' Agent notifies the Facility Agent are, or are to be applied:

- (i) to meet a third party claim;
- (i) to the replacement, reinstatement and/or repair of the assets in respect of which the relevant insurance claim was made or
- (ii) to be reinvested, contractually committed to be reinvested or designated by the board of directors of Obligors' Agent for reinvestment in either:
 - (A) other assets;
 - (B) Permitted Acquisitions or Permitted Loans; or
 - (C) Capital Expenditure,

within 365 days (or 180 days following the end of such 365 day period if such cash proceeds have been contractually committed to be used within such 365 day period) after receipt (it being understood that if any portion of such cash proceeds are not so used within such 365 day period or within 180 days following the end of such 365 day period shall constitute Disposal Proceeds as of such date).

"Insurance Proceeds" means the cash proceeds of any insurance claim received by any Group Company in respect of the loss or destruction of assets (and not relating to third party liability, business interruption, loss of earnings or similar claims) except for Excluded Insurance Proceeds and after deducting:

- (i) any reasonable expenses in relation to that claim which are incurred by any Group Company to persons who are not Group Companies.
- (ii) any Tax incurred and required to be paid by a Group Company in connection with that claim;
- (iii) any Financial Indebtedness which is to be repaid to the undertaking or business the subject of the claim by a Group Company;
- (iv) any Financial Indebtedness which is secured on the assets the subject of the claim and which
 is to be prepaid from the proceeds of the relevant claim;

- (v) any repayments of Additional First Lien Debt (limited to its proportionate share of such prepayment, based on the principal amount of such then outstanding debt as a percentage of the aggregate principal amount of all then outstanding Term Loans and Additional First Lien Debt): and
- (vi) any provisions in the accounts for liability in connection with the relevant claim.
- (b) Subject to paragraph (c) below and clause 12.5, Obligors' Agent shall ensure that the Borrowers prepay Utilisations of the Term Facilities (or, with respect to clause (iii) below, the Term Facility B only) in the following amounts at the times and in the order of application contemplated by clause 12.3 (Application of Mandatory Prepayments) (each such prepayment, a "Prepayment Amount"):
 - (i) 100% (provided that such percentage shall be reduced to 50% or 0% if the Net First Lien Secured Leverage is equal to or less than 4.25:1.0 but greater than 4.0:1.0 or equal to or less than 4.0:1.0, respectively) of the cash proceeds actually received from any Disposal Proceeds during any Financial Year that are in excess of the greater of (i) \$26,500,000 (or its equivalent in other currencies) and (ii) 5.3% of the then most recently reported Consolidated EBITDA;
 - (ii) 100% (provided that such percentage shall be reduced to 50% or 0% if the Net First Lien Secured Leverage is equal to or less than 4.25:1.0 but greater than 4.0:1.0 or equal to or less than 4.0:1.0, respectively) of the cash proceeds actually received from any Insurance Proceeds during any Financial Year that are in excess of the greater of (i) \$26,500,000 (or its equivalent in other currencies) and (ii) 5.3% of the then most recently reported Consolidated EBITDA; and
 - (iii) in relation to each Financial Year ending on or after 31 December 2022, an amount equal to the Relevant Percentage of Excess Cashflow for that Financial Year, such Relevant Percentage to be determined by reference to the Net First Lien Secured Leverage for that Financial Year as set out in the table below:

Net First Lien Secured Leverage for that Financial Year	Relevant Percentage of Excess Cashflow
Greater than 5.50:1	50%
Less than or equal to 5.50:1	0%

(c) The Borrower shall notify the Facility Agent in writing of any mandatory prepayment of Term Loans required to be made pursuant to Clause 12.2(b) (Disposal and Insurance Proceeds and Excess Cash Flow). The Facility Agent will promptly notify each Lender with a participation in any Term Loans of the contents of any such prepayment notice and of such Lender's ratable portion of such prepayment (based on such Lender's pro rata share of each relevant Class of the Term Loans) at least 4 Business Days prior to the Prepayment Date. Lender may decline to accept all (but not less than all) of its share of any mandatory prepayment pursuant to Clause 12.2(b) (Disposal and Insurance Proceeds and Excess Cash Flow) (any such Lender, a "Declining Lender") by providing written notice to the Facility Agent no later than two Business Days after the date of such Lender's receipt of notice from the Facility Agent regarding such prepayment. If any Lender does not give a notice to the Facility Agent on or prior to such two Business Day time period informing the Facility Agent that it declines to accept the applicable prepayment, then such Lender will be deemed

to have accepted such prepayment. On any Prepayment Date, an amount equal to the Prepayment Amount minus the portion thereof allocable to Declining Lenders, in each case for such Prepayment Date, shall be paid to the Facility Agent by the Borrower and applied by the Facility Agent ratably to prepay Loans owing to Lenders (other than Declining Lenders) in the manner described in Clause 12.3 (Applicable of Mandatory Prepayments) for such prepayment. Any amounts that would otherwise have been applied to prepay Loans owing to Declining Lenders shall be retained by the Borrower (such amounts retained by the Borrower, "Declined Amounts") for any purposes not prohibited by this agreement.

12.3 Application of Mandatory Prepayments

- (a) Subject to the Intercreditor Agreement, a prepayment made under clause 12.2(b) (Disposal and Insurance Proceeds and Excess Cashflow) shall be applied pro rata against each Term Loan, provided that prepayments made under paragraph (iii) of clause 12.2(b) shall be applied against Term Facility B Loans and if applicable any Incremetal Term Facility Loans that the Obligors' Agent has designated as 'Incremental Term Facility B Loans'. In addition, such prepayments shall be applied to the remaining instalments of the Term Loans under the applicable Class or Classes as the relevant Borrower or the Obligors' Agent shall select (and if no selection is made then such prepayment shall be applied on remaining instalments in direct order).
- (b) Unless Obligors' Agent makes an election under paragraph (c) below, the Borrowers shall prepay Term Loans at the following times (each such date, a "**Prepayment Date**"):
 - (i) in the case of any prepayment relating to the amounts of Disposal Proceeds or Insurance Proceeds within 10 Business Days (or such longer period as may be reasonable acceptable to the Facility Agent) or, if Obligors' Agent elects to reinvest such amounts as permitted by clause 12.2 above, upon expiry of the applicable period for reinvestment and/or commitment to reinvest specified in clause 12.2 above; and
 - (ii) in the case of any prepayment relating to an amount of Excess Cashflow, on the last day of the first Interest Period (or with respect to a RFR Loan, the next Interest Payment Date) ending at least 10 Business Days after delivery pursuant to clause 25.1 (Financial Statements) of the Annual Financial Statements of the Parent.
- (c) Subject to paragraph (d) below, Obligors' Agent may elect that any prepayment under clause 12.2 (Disposal and Insurance Proceeds and Excess Cashflow) be applied in prepayment of a Term Loan that is a Term Benchmark Loan on the last day of the Interest Period (or with respect to a RFR Loan, the next Interest Payment Date) relating to that Term Loan. If Obligors' Agent makes that election, then a proportion of the Term Loan equal to the amount of the relevant prepayment will be due and payable on the last day of its Interest Period (or with respect to a RFR Loan, the next Interest Payment Date). For the avoidance of doubt, Term Facility Loans and Incremental Term Facility Loans shall be considered separate Term Loans for these purposes.
- (d) If Obligors' Agent has made an election under paragraph (c) above but a Default has occurred and is continuing, that election shall no longer apply and a proportion of the Term Loan in respect of which the election was made equal to the amount of the relevant prepayment shall be immediately due and payable (unless the Majority Lenders otherwise agree in writing).

12.4 Excluded Proceeds

Where Excluded Disposal Proceeds and Excluded Insurance Proceeds include amounts which are intended to be used for a specific purpose within a specified period (as set out in the relevant definition of Excluded Disposal Proceeds or Excluded Insurance Proceeds), Obligors' Agent shall ensure that those amounts are used for that purpose and, if requested to do so by the Facility Agent, shall promptly deliver a certificate to the Facility Agent at the time of such application and at the end of such period confirming the amount (if any) which has been so applied within the requisite time periods provided for in the relevant definition.

12.5 **Upstreaming Monies**

If:

- (a) monies are required to be applied in prepayment or repayment of the Facilities under this clause 12 (Mandatory Prepayments) but in order to be so applied need to be upstreamed or otherwise transferred from one Group Company to another Group Company to effect that prepayment or repayment; and
- (b) such monies cannot be so upstreamed or transferred or the relevant prepayment may not be made without breaching a financial assistance prohibition or without breaching some other legal restriction or without the Group incurring a material cost (whether as a result of paying additional Taxes or otherwise),

there will be no obligation to make that payment or prepayment until that impediment no longer applies. Each Obligor will (and Obligors' Agent will procure that each other Group Company will) use all reasonable endeavours to overcome that impediment.

12.6 Election

Subject to clause 12.3(a) (Application of Mandatory Prepayments) above, any prepayment of Loans required by this agreement shall (if there are two or more Borrowers of such Loans) be made by such Borrower or Borrowers and (if applicable) in such proportions as Obligors' Agent may specify.

13. **RESTRICTIONS**

13.1 Notices of Cancellation or Prepayment

Any notice of cancellation or prepayment, authorisation or other election given by any Party under clause 11 (Illegality, Voluntary Prepayment and Cancellation) or clause 12.3(c) (Application of Mandatory Prepayments) shall (subject to the terms of those clauses) be irrevocable and shall specify the date or dates upon which the relevant cancellation or prepayment is to be made and the amount of that cancellation or prepayment.

13.2 Interest and Other Amounts

Any prepayment under this agreement shall be made together with accrued interest on the amount prepaid and without premium or penalty other than Break Costs.

13.3 No Reborrowing of Term Facilities

Without prejudice to clause 2.2 (Incremental Facilities), no Borrower may reborrow any part of a Term Facility which is prepaid.

13.4 Reborrowing of Revolving Facility

Unless a contrary indication appears in this agreement, any part of a Revolving Facility which is prepaid may be reborrowed in accordance with the terms of this agreement.

13.5 Prepayment in accordance with Agreement

No Borrower shall repay or prepay all or any part of the Utilisations or cancel all or any part of the Commitments except at the times and in the manner expressly provided for in this agreement.

13.6 No Reinstatement of Commitments

Without prejudice to clause 2.2 (Incremental Facilities), no amount of the Total Commitments cancelled under this agreement may be subsequently reinstated.

13.7 Facility Agent's Receipt of Notices

If the Facility Agent receives a notice under clause 11 (Illegality, Voluntary Prepayment and Cancellation) or an election under clause 12.3(d) (Application of Mandatory Prepayments) it shall promptly forward a copy of that notice or election to either Obligors' Agent or the affected Lender, as appropriate.

13.8 Effect of repayment and prepayment on Commitments

If all or part of any Lender's participation in a Utilisation under a Facility is repaid or prepaid and is not available for redrawing (other than by operation of Clause 4.2 (Further conditions precedent)), an amount of that Lender's Commitment (equal to the Base Currency Amount of the amount of the participation which is repaid or prepaid) in respect of that Facility will be deemed to be cancelled on the date of repayment or prepayment.

14. INTEREST

14.1 Calculation of Interest

The rate of interest on each Loan is the percentage rate per annum which is the aggregate of the applicable:

- (a) Margin; and
- (b) (i) in relation to any Term Benchmark Loan denominated in Dollars, the Term SOFR Rate for the relevant Interest Period, (ii) in relation to any RFR Loan, the Daily Simple SOFR, (iii) in relation to any Term Benchmark Loan denominated in euro, EURIBOR for the relevant Interest Period or (iv) in relation to any ABR Loan, the Alternate Base Rate.

14.2 Payment of Interest

- (a) The Borrower to which a Loan has been made shall pay accrued interest on that Loan on each Interest Payment Date.
- (b) If the annual audited financial statements of the Group and related Compliance Certificate received by the Facility Agent show that a higher Margin should have applied during a certain period, then Obligors' Agent shall (or shall ensure the relevant Borrower shall) promptly pay to the Facility Agent any amounts necessary to put the Facility Agent and the Lenders in the position they would have been in had the appropriate rate of the Margin applied during such period.
- (c) If the annual audited financial statements of the Group and related Compliance Certificate received by the Facility Agent show that a lower Margin should have applied during a certain period, then the next payment of interest on the relevant Loans shall be reduced by an amount which is equal to the amount of interest paid by the Borrowers during the relevant periods less the amount which would have been payable if the Margin had been decreased to the correct level during the relevant periods as calculated by reference to the relevant annual audited financial statements of the Group.

14.3 **Default Interest**

- (a) If an Obligor fails to pay any amount payable by it under a Finance Document on its due date, interest shall accrue on the overdue amount from the due date up to the date of actual payment (both before and after judgment) at a rate which is one per cent higher than the rate otherwise applicable to such Loan. Any interest accruing under this clause 14.3 shall be immediately payable by the Obligor on demand by the Facility Agent.
- (b) [reserved].
- (c) Default interest (if unpaid) arising on an overdue amount will be compounded with the overdue amount at the end of each Interest Period applicable to that overdue amount but will remain immediately due and payable.

14.4 Notification of Rates of Interest

The Facility Agent shall promptly notify the Lenders and the relevant Borrower (or Obligors' Agent) of the determination of a rate of interest under this agreement.

14.5 Minimum Interest

- (a) When entering into this agreement, the parties hereto have assumed that the interest payable at the rates set out in this agreement and any other payments under the Finance Documents are not and will not become subject to Swiss Withholding Taxes. Therefore, each Obligor acknowledges and agrees that the interest rates set out in this agreement shall constitute minimum interest rates. Nevertheless, if a Tax Deduction is required by Swiss law (other than FATCA) to be made by any Obligor in respect of any interest or other payments under a Finance Document and should it be unlawful for such Obligor to comply with Clause 18.2(c) (Tax Gross-Up) for any reason then:
 - (i) the applicable interest, fee, commission rate in relation to that interest, fee or commission payment shall be:
 - (A) the interest, fee, commission rate which would have applied to that interest, fee or commission payment (as provided for in this agreement or any other Finance Document in the absence of this Clause 14.5), divided by
 - (B) one (1) minus the rate at which the relevant Tax Deduction is required to be made (where the rate at which the relevant Tax Deduction is required to be made is for this purpose expressed as a fraction of (1) rather than as a percentage)
 - (ii) the Obligor shall:
 - (A) pay the relevant interest, fee or commission at the adjusted rate in accordance with paragraph (i) above; and
 - (B) make the Tax Deduction on the interest, fee or commission so recalculated, and
 - (C) all references to a rate of interest, fee or commission in this agreement or any other Finance Document shall be construed accordingly.
- (b) To the extent that interest, fee or commission payable by an Obligor under a Finance Document becomes subject to Swiss Withholding Tax, each relevant Lender and that Obligor shall co-operate in completing any procedural formalities (including

submitting forms and documents required by the appropriate tax authority) to the extent possible and necessary for the relevant Obligor to obtain authorization (i) to make interest, fee or commission payments without them being subject to Swiss Withholding Tax, or (ii) to being subject to Swiss Withholding Tax at a rate reduced under applicable double taxation treaties.

(c) For the purposes of this clause 14.5, "the rate at which the relevant Tax Deduction is required to be made" shall mean Swiss withholding tax at the standard rate effective from time to time ((35% (thirty-five per cent) at the date of this agreement) unless a tax ruling obtained from the Swiss Federal Tax Administration confirms that the rate at which the relevant Tax Deduction is required to be made is, pursuant to any double-taxation treaty, at specified lower rates in relation to specified Lenders in which case such lower rates shall be applied in relation to the relevant Lenders.]

15. INTEREST PERIODS

15.1 Selection of Interest Periods and Terms

- (a) A Borrower (or Obligors' Agent on behalf of a Borrower) may select an Interest Period for a Term Benchmark Loan in the Utilisation Request for that Loan or (if the Term Benchmark Loan is a Term Loan and has already been borrowed) in a Selection Notice.
- (b) Each Selection Notice for a Term Benchmark Loan must be delivered to the Facility Agent by the Borrower (or Obligors' Agent on behalf of the Borrower) to which that Loan was made not later than the Specified Time.
- (c) If a Borrower (or Obligors' Agent) fails to deliver a Selection Notice to the Facility Agent in accordance with paragraph (b) above, the relevant Interest Period for Term Benchmark Loan will, subject to clause 15.2 (Changes to Interest Periods), be one Month.
- (d) Subject to this clause 15, a Borrower (or Obligors' Agent) may, in the case of a Term Benchmark Loan, select an Interest Period of one, three or six Months or any other period agreed between Obligors' Agent and the Facility Agent (acting on the instructions of all the Lenders under the same Class of Loans), in each case, subject to the availability for the Benchmark applicable to the relevant Loan; provided that, each Term Facility B Loan made or re-designated on the 2024 Effective Date shall have an Interest Period ending on 30 September 2024 (and shall bear interest at the interest rate applicable to a one-Month Interest Period ending on 30 September 2024) and thereafter shall have an Interest Period as selected by the Borrower (or Obligors' Agent on behalf of a Borrower) pursuant to this clause 15.1.
- (e) An Interest Period for a Term Benchmark Loan shall not extend beyond the Termination Date applicable to that Loan.
- (f) Each Interest Period for a Term Benchmark Loan shall start on the Utilisation Date or (if already made) on the last day of its preceding Interest Period.

15.2 [Reserved]

15.3 Non-Business Days

If an Interest Period would otherwise end on a day which is not a Business Day, that Interest Period will instead end on the next Business Day in that calendar month (if there is one) or the preceding Business Day (if there is not).

15.4 Consolidation and Division of Loans

- (a) Subject to paragraph (b) below, if two or more Interest Periods:
 - (i) relate to Term Benchmark Loans from the same Term Facility in the same currency;
 - (ii) end on the same date; and
 - (iii) are made to the same Borrower,

those Term Benchmark Loans will, unless that Borrower (or Obligors' Agent on its behalf) specifies to the contrary in the Selection Notice for the next Interest Period, be consolidated into, and treated as, a single Loan on the last day of the Interest Period.

(b) Subject to clause 4.4 (Maximum Number of Utilisations) and clause 5.3 (Currency and Amount), if a Borrower (or Obligors' Agent on its behalf) requests in a Selection Notice that a Term Benchmark Loan be divided into two or more Loans, that Loan will, on the last day of its Interest Period, be so divided with Base Currency Amounts specified in that Selection Notice, being an aggregate Base Currency Amount equal to the Base Currency Amount of the Loan immediately before its division.

16. CHANGES TO THE CALCULATION OF INTEREST

16.1 [Reserved]

16.2 Alternate Rate of Interest

- (a) Subject to Clause 16.3 (Benchmark Replacement), if:
- (i) the Facility Agent determines (which determination shall be conclusive absent manifest error) in consultation with the Obligors' Agent (A) prior to the commencement of any Interest Period for a Term Benchmark Loan, that adequate and reasonable means do not exist for ascertaining the Term SOFR Rate or the EURIBOR (including because the relevant Screen Rate is not available or published on a current basis), as applicable, for such Interest Period or (B) at any time, that adequate and reasonable means do not exist for ascertaining the Daily Simple SOFR; or
- (ii) the Facility Agent and the Obligors' Agent are advised by the Majority Lenders that (A) prior to the commencement of any Interest Period for a Term Benchmark Loan, the Term SOFR Rate or EURIBOR, as applicable, for such Interest Period will not adequately and fairly reflect the cost to such Lenders (or Lender) of making or maintaining their Loans (or its Loan) for such Interest Period or (B) at any time, Daily Simple SOFR will not adequately and fairly reflect the cost to such Lenders (or Lender) of making or maintaining their Loans (or its Loan); provided that a Lender shall not so advise the Facility Agent unless such Lender is generally making similar claims upon, or otherwise similarly enforcing its agreements with, similarly-situated borrowers (and provided further that no Lender shall have any obligation to disclose information about any other borrowers);

then the Facility Agent shall promptly give notice thereof to the relevant Borrower (and Obligors' Agent) and the Lenders by telephone, telecopy or electronic mail as promptly as practicable thereafter and, until (x) the Facility Agent notifies such Borrower (and Obligors' Agent) and the Lenders that the circumstances giving rise to such notice no longer exist with respect to the relevant Benchmark and (y) the relevant Borrower delivers a new Selection Notice or a new Utilisation Request, (A) for a Loan denominated in Dollars, (1) any Selection Notice that requests the continuation of any Loan as a Term Benchmark Loan and any Utilisation Request that

requests a Term Benchmark Loan shall instead be deemed to be an Selection Notice or a Utilisation Request, as applicable, for an RFR Loan so long as the Daily Simple SOFR is not also the subject of Clause 16.2(a)(i) or (ii) above and (2) any Utilisation Request that requests an RFR Loan that is subject to Clause 16.2(a)(i) or (ii) above shall instead be deemed to be a Utilisation Request, as applicable, for an ABR Loan and (B) for Loans denominated in euro, any Selection Notice that requests the continuation of any Loan as a Term Benchmark Loan and any Utilisation Request that requests a Term Benchmark Loan, in each case shall be ineffective; provided that if the circumstances giving rise to such notice affect only one Type of Loans, then all other Types of Loans shall be permitted.

Furthermore, if any Term Benchmark Loan or RFR Loan is outstanding on the date of the Borrower's (and Obligors' Agent's) receipt of the notice from the Facility Agent referred to in this Clause 16.2(a) with respect to a Benchmark applicable to such Term Benchmark Loan or RFR Loan, then until (x) the Facility Agent notifies the Borrower (and Obligors' Agent) and the Lenders that the circumstances giving rise to such notice no longer exist with respect to the relevant Benchmark and (y) the relevant Borrower (or Obligors' Agent) delivers a new Selection Notice or a new Utilisation Request, (A) for a Loan denominated in Dollars (1) any such Term Benchmark Loan shall on the last day of the Interest Period applicable to such Loan (or the next succeeding Business Day if such day is not a Business Day), be converted by the Facility Agent to, and shall constitute, an RFR Loan so long as the Daily Simple SOFR is not also the subject of Clause 16.2(a)(i) or (ii) above (y) an ABR Loan if the Daily Simple SOFR for Dollar Loans also is the subject of Clause 16.2(a)(i) or (ii) above, on such day, and (2) any RFR Loan that is subject to Clause 16.2(a)(i) or (ii) above shall on and from such day be converted by the Facility Agent to, and shall constitute an ABR Loan and (B) for a Loan denominated in euro, any Term Benchmark Loan shall, on the last day of the Interest Period applicable to such Loan (or the next succeeding Business Day if such day is not a Business Day) bear interest at the Central Bank Rate plus the Margin; provided that, if the Facility Agent determines (which determination shall be conclusive and binding absent manifest error) that the Central Bank Rate cannot be determined, any outstanding affected Term Benchmark Loans denominated in euro shall, at the Borrower's (or Obligors' Agent's) election prior to such day, (A) be prepaid by the relevant Borrower on such day or (B) solely for the purpose of calculating the interest rate applicable to such Term Benchmark Loan, such Term Benchmark Loan denominated in euro shall be deemed to be a Term Benchmark Loan denominated in Dollars and shall accrue interest at the same interest rate applicable to Term Benchmark Loans denominated in Dollars at such time.

16.3 **Benchmark Replacement**

(a) Notwithstanding anything to the contrary herein or in any other Finance Document (and any Hedging Agreement shall be deemed not to be a "Finance Document" for purposes of this Clause 16.3), if a Benchmark Transition Event and its related Benchmark Replacement Date have occurred prior to the Reference Time in respect of any setting of the then-current Benchmark, then (x) if a Benchmark Replacement is determined in accordance with clause (1) of the definition of "Benchmark Replacement" with respect to Dollars for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any Finance Document in respect of such Benchmark setting and subsequent Benchmark settings without any amendment to, or further action or consent of any other party to, this agreement or any other Finance Document and (y) if a Benchmark Replacement is determined in accordance with clause (2) of the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any Finance Document in respect of any Benchmark

setting at or after 5:00 p.m. (New York City time) on the fifth (5th) Business Day after the date notice of such Benchmark Replacement is provided to the Lenders without any amendment to, or further action or consent of any other party to, this agreement or any other Finance Document so long as the Facility Agent has not received, by such time, written notice of objection to such Benchmark Replacement from Lenders comprising the Majority Lenders (or, with respect to the Revolving Facility, the Majority RCF Lenders).

- (b) Notwithstanding anything to the contrary herein or in any other Finance Document, the Facility Agent will have the right to make Benchmark Replacement Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Finance Document, any amendments implementing such Benchmark Replacement Conforming Changes will become effective without any further action or consent of any other party to this agreement or any other Finance Document.
- The Facility Agent will promptly notify the Borrowers (and the Obligors' Agent) and the Lenders of (i) any occurrence of a Benchmark Transition Event, (ii) the implementation of any Benchmark Replacement, (iii) the effectiveness of any Benchmark Replacement Conforming Changes, (iv) the removal or reinstatement of any tenor of a Benchmark pursuant to clause (e) below and (v) the commencement or conclusion of any Benchmark Unavailability Period. Any determination, decision or election that may be made by the Facility Agent or, if applicable, any Lender (or group of Lenders) pursuant to Clause 16.2 (Alternate Rate of Interest) and this Clause 16.3, including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or refrain from taking any action or any selection, will be conclusive and binding absent manifest error and may be made in its or their sole discretion and without consent from any other party to this agreement or any other Finance Document, except, in each case, as expressly required pursuant to Clause 16.2 (Alternate Rate of Interest) and this Clause 16.3; provided, that the Facility Agent may seek clarification or direction from the Lenders prior to the exercise of any such instructed action and may refrain from acting or notifying until such clarification or direction has been provided.
- Notwithstanding anything to the contrary herein or in any other Finance Document, at any time (including in connection with the implementation of a Benchmark Replacement), (i) if the then-current Benchmark is a term rate (including the Term SOFR Rate or EURIBOR) and either (A) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by the Facility Agent in its reasonable discretion or (B) the regulatory supervisor for the administrator of such Benchmark has provided a public statement or publication of information announcing that any tenor for such Benchmark is or will be no longer representative, then the Facility Agent may modify (in consultation with the Borrower) the definition of "Interest Period", Clause 14 (Interest) and Clause 15 (Interest Periods) for any Benchmark settings at or after such time to remove such unavailable or non-representative tenor and (ii) if a tenor that was removed pursuant to clause (i) above either (A) is subsequently displayed on a screen or information service for a Benchmark (including a Benchmark Replacement) or (B) is not, or is no longer, subject to an announcement that it is or will no longer be representative for a Benchmark (including a Benchmark Replacement), then the Facility Agent may modify the definition of "Interest Period", Clause 14 (Interest) and Clause 15 (Interest Periods) for all Benchmark settings at or after such time to reinstate such previously removed tenor.

Benchmark Unavailability Period, the relevant Borrower (or Obligors' Agent) may revoke any request for a Term Benchmark Loan or RFR Loan, or continuation of Term Benchmark Loans, to be made or continued during any Benchmark Unavailability Period and, failing that, either (x) such Borrower will be deemed to have converted any request for (1) an Utilisation Request or Selection Notice for a Term Benchmark Loan denominated in Dollars (A) into a request for a Loan of or conversion to an RFR Loan so long as the Daily Simple SOFR is not the subject of a Benchmark Unavailability Period or (B) an ABR Loan if the Daily Simple SOFR for Dollar Loans is the subject of a Benchmark Unavailability Period or (y) any Utilisation Request for a Term Benchmark Loan denominated in euro shall be ineffective. Furthermore, if any Term Benchmark Loan or RFR Loan is outstanding on the date of the relevant Borrower's (and Obligors Agents') receipt of notice of the commencement of a Benchmark Unavailability Period with respect to a Benchmark applicable to such Term Benchmark Loan or RFR Loan, then until such time as a Benchmark Replacement is implemented pursuant to Clause 16.2 (Alternate Rate of Interest) and this Clause 16.3, (A) for a Loan denominated in Dollars (1) any Term Benchmark Loan shall on the last day of the Interest Period applicable to such Loan (or the next succeeding Business Day if such day is not a Business Day), be converted by the Facility Agent to, and shall constitute, (x) an RFR Loan so long as the Daily Simple SOFR is not the subject of a Benchmark Unavailability Period or (y) an ABR Loan if the Daily Simple SOFR for Dollar Loans is the subject of a Benchmark Unavailability Period, on such day and (2) any RFR Loan so long as the Daily Simple SOFR is subject of a Benchmark Unavailability Period shall on and from such day be converted by the Facility Agent to, and shall constitute an ABR Loan and (B) for Loans denominated in euro, any Term Benchmark Loan shall, on the last day of the Interest Period applicable to such Loan (or the next succeeding Business Day if such day is not a Business Day) bear interest at the Central Bank Rate plus the Margin; provided that, if the Facility Agent determines (which determination shall be conclusive and binding absent manifest error) that the Central Bank Rate for euro cannot be determined, any outstanding affected Term Benchmark Loans in euro shall, at the Borrower's (or Obligors' Agent's) election prior to such day: (A) be prepaid by the Borrower on such day or (B) solely for the purpose of calculating the interest rate applicable to such Term Benchmark Loan, such Term Benchmark Loan denominated in euro shall be deemed to be a Term Benchmark Loan denominated in Dollars and shall accrue interest at the same interest rate applicable to a Term Benchmark Loan denominated in Dollars at such time.

Upon the Borrowers' (and Obligors Agent's) receipt of notice of the commencement of a

16.4 Break Costs

- (a) With respect to a Term Benchmark Loan, each Borrower shall, within thirty Business Days of demand by a Finance Party, pay to that Finance Party its Break Costs attributable to all or any part of a Loan or Unpaid Sum being paid by that Borrower on a day other than the last day of an Interest Period for that Loan or Unpaid Sum.
- (b) Each Lender shall provide a certificate to the Borrowers (and Obligors' Agent) confirming the amount of its Break Costs for any Interest Period in which they accrue and each Lender waives its right to receive Break Costs if it fails to provide such certificate to the Borrowers (and Obligors' Agent) within 20 days after the event giving rise to a claim for Break Costs.

17. **FEES**

17.1 Commitment Fee

- (a) Obligors' Agent shall, or shall procure that another Obligor shall, pay to the Facility Agent (for the account of each Lender with a RCF Commitment) on the last Business Day of each March, June, September and December (commencing on December 31, 2022) and on the date on which is the Termination Date related to the Facility RCF, a commitment fee in the Base Currency computed at (x) prior to the first Adjustment Date occurring after the 2022 Effective Date, 0.25% per cent per annum and (y) on and after the first Adjustment Date occurring after the 2022 Effective Date, a per annum percentage determined in accordance with the Pricing Grid, in each case, on such Lender's daily amount of applicable Available Commitment under Facility RCF during the preceding quarter (or other period commencing with the 2022 Effective Date or ending with the date on which the RCF Commitments are terminated).
- (b) The Obligors' Agent shall, or shall procure that another Obligor shall, pay to the Facility Agent (for the amount of each Lender) a commitment fee in respect to any Incremental Revolving Facility, being such amount to be agreed between the Borrower and each relevant Lender under such Incremental Revolving Facility for the Availability Period applicable to that Incremental Revolving Facility.

17.2 Arrangement Fee

The Parent shall, or shall procure that another Obligor shall, pay to the Arrangers an arrangement fee in the amount and at the times agreed in a Fee Letter.

17.3 Agency Fee

The Obligors' Agent shall, or shall procure that another Obligor shall, pay to the Facility Agent (for its own account) an agency fee in the amount and at the times agreed in a Fee Letter.

17.4 Fees Payable in Respect of Letters of Credit

- (a) The Obligors' Agent or each Borrower shall pay to the Issuing Bank a fronting fee in an amount to be agreed with the Issuing Bank (but in no event greater than 0.125 per cent per annum) on the outstanding amount which is counter-indemnified by the other Lenders of each Letter of Credit requested by it and in respect of which the Borrower has not provided cash cover for the period from the issue of that Letter of Credit until its Expiry Date.
- (b) The Obligors' Agent or each Borrower shall pay to the Facility Agent (for the account of each Lender) a Letter of Credit fee in the Base Currency computed at the rate equal to:
 - (i) in respect of Letters of Credit which have a 50 per cent capital weighting for the relevant Lender, 50 per cent of the Margin applicable to a Revolving Facility Loan;
 - (ii) to the extent that any amount under a Letter of Credit is cash covered, 0.50 per cent per annum: and
 - (iii) in relation to all other Letters of Credit, the Margin applicable to a Revolving Facility Loan,

on the outstanding amount of each Letter of Credit requested by it for the period from the issue of that Letter of Credit until its Expiry Date. This fee shall be distributed according to each Lender's L/C Proportion of that Letter of Credit.

(c) The accrued fronting fee and Letter of Credit fee on a Letter of Credit shall be payable on the last day of each successive period of three Months (or such shorter period as shall end on the Expiry Date for that Letter of Credit) starting on the date of issue of

that Letter of Credit. The accrued fronting fee and Letter of Credit fee is also payable to the Facility Agent on the cancelled amount of any Lender's Revolving Facility Commitment at the time the cancellation is effective if that Commitment is cancelled in full and the Letter of Credit is prepaid or repaid in full.

17.5 Interest, Commission and Fees on Ancillary Facilities

The rate and time of payment of interest, commission, fees and any other remuneration in respect of each Ancillary Facility shall be determined by agreement between the relevant Ancillary Lender and the Borrower of that Ancillary Facility based upon normal market rates and terms.

17.6 Repricing Fee

- (a) In the event that, at any time in the period from but excluding the 2024 Effective Date to and including the day immediately prior to the date that is six (6) Months after the 2024 Effective Date any Borrower:
 - (i) prepays, refinances, substitutes or replaces any Term Facility B Loans in connection with a Repricing Transaction; or
 - (ii) effects any amendment of this agreement resulting in a Repricing Transaction,

the Parent shall procure the payment on the date that such Repricing Transaction takes effect (without double counting under any of the sub-paragraphs of this clause 17.6(a) in respect of that Repricing Transaction) of:

- (A) in the case of clause 17.6(a)(i) a prepayment premium to each Lender which provides a Term Facility B Loan which is prepaid, refinanced, substituted or replaced pursuant to clause 17.6(a)(i) of 1.00 per cent of the aggregate principal amount of that Lender's Facility B Loans so prepaid, refinanced, substituted or replaced; and
- (B) in the case of clause 17.6(a)(ii), a fee to each Lender which provides a Term Facility B Loan which is the subject of any amendment of this agreement resulting in a Repricing Transaction in respect of that Facility B Loan for an amount equal to 1.00 per cent of the aggregate principal amount of that Lender's affected Term Facility B Loans outstanding immediately prior to such amendment.

In this clause 17.6:

"Repricing Transaction" means the prepayment, refinancing, substitution or replacement of all or any part of the Term Facility B Loans after the 2024 Effective Date with the incurrence by any Group Company of any broadly syndicated first lien term loans having an effective interest cost or weighted average yield (with the comparative determinations to be made by the Parent consistent with generally accepted financial practices, after giving effect to, among other factors, margin, interest rate floors, upfront or similar fees or original issue discount shared with all providers of such financing, but excluding the effect of any arrangement, structuring, syndication or other fees payable in connection therewith that are not shared with all providers of such financing, and without taking into account any fluctuations in any Benchmark Rate) that is less than the effective interest cost or weighted average yield (as determined by the Parent on the same basis) of such Term Facility B Loans, including without limitation, as may be effected through any amendment to this agreement relating to the interest rate for, or weighted average yield of, such Term Facility B Loans, but only to the extent, in each case, that the primary purpose, as determined by the Obligors' Agent in good faith, of such prepayment, refinancing, substitution or replacement is to obtain a lower effective interest cost or weighted average yield on such Term Facility B Loans; provided, that in no event shall any such prepayment, repayment,

refinancing, substitution, replacement, amendment, waiver or other modification in connection with a Change of Control or Qualifying IPO or a Transformative Acquisition constitute a Repricing Transaction. For the avoidance of doubt and for purposes of the definition of "Repricing Transaction," any term loans provided pursuant to Term Facility A (or in the form of a term "a" facility or otherwise club credit facilities) or any UKEF Facility shall not be treated as broadly syndicated first lien term loans for the purpose of this Clause 17.6.

17.7 Incremental Facility Commitment Fees

The Obligors' Agent shall, or shall procure that another Obligor shall, pay to the relevant Incremental Facility Lender(s), or to the Facility Agent for the account of the relevant Incremental Facility Lender(s), any commitment fees relating to an Incremental Facility in the amount and at the times agreed in a Fee Letter relating to that Incremental Facility.

18. TAX GROSS-UP AND INDEMNITIES

18.1 **Definitions**

(a) In this agreement:

"Borrower DTTP Filing" means an HM Revenue & Customs' Form DTTP2 or HM Revenue & Customs' Form DTTP2A duly completed and filed by the relevant Borrower, which contains the scheme reference number and jurisdiction of tax residence stated in respect of a Treaty Lender in the documentation which:

- (i) where the Borrower is a Borrower as at the date on which that Treaty Lender becomes a Lender under such Facility, is filed with HM Revenue & Customs within 30 days of that date (or if later within 30 days of the 2019 Effective Date); or
- (ii) where the Borrower is not a Borrower as at the date on which that Treaty Lender becomes a Lender under such Facility, is filed with HM Revenue & Customs within 30 days of the date on which that Borrower becomes an Additional Borrower.

"Borrower's Tax Jurisdiction" means the jurisdiction in which a Borrower is resident for tax purposes, in the case of an Original Borrower, as at the date of this agreement and in the case of an Additional Borrower, as at the date which it becomes a Borrower;

"Effective Tax Rate" shall have the meaning prescribed in the Global Anti Base Erosion Rules.

"Excluded U.S. Tax Deduction" means (a) a Tax Deduction described in clause 18.2(m) without regard to (1) the parenthetical in clause (i)(x) thereof and (2) the exception in clause (i) thereof immediately following clause (i)(y) thereof (beginning with "except in each case"), and (b) a Tax Deduction under FATCA or a FATCA Agreement.

"Exempt Lender" means, a Lender which is:

- (i) a company resident in the United Kingdom for United Kingdom tax purposes;
- (i) a partnership each member of which is:
 - (A) a company so resident in the United Kingdom; or

- (B) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account in computing its chargeable profits (within the meaning of section 19 of the CTA) the whole of any share of interest payable in respect of that advance that falls to it by reason of Part 17 of the CTA; or
- (iii) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account interest payable in respect of that advance in computing the chargeable profits (within the meaning of section 19 of the CTA) of that company.
- **"FATCA Agreement"** means any agreement made by a person (or an affiliate of that person) with the US Internal Revenue Service of the United States of America, any governmental or taxation authority in any other jurisdiction or the United States government pursuant to, or related to, FATCA.
- "Minimum Rate" shall have the meaning prescribed in the Global Anti Base Erosion Rules.
- "Protected Party" means a Finance Party which is or will be subject to any liability, or required to make any payment, for or on account of Tax in relation to a sum received or receivable (or any sum deemed for the purposes of Tax to be received or receivable) under a Finance Document;
- "Qualifying Lender" means a Lender which is in relation to a Borrower incorporated in the United Kingdom:
- (i) an Exempt Lender; or
- (ii) a Treaty Lender; or
- (iii) a Qualifying UK Lender;
- "Qualifying UK Lender" means a Lender which is beneficially entitled to interest payable to that Lender in respect of an advance under a Finance Document and is a Lender:
- (i) which is a bank (as defined for the purpose of section 879 of the Income Tax Act 2007) making an advance under a Finance Document; or
- (ii) in respect of an advance made under a Finance Document by a person that was a bank (as defined for the purpose of section 879 of the Income Tax Act 2007) at the time that that advance was made,

and which is within the charge to United Kingdom corporation tax as respects any payments of interest made in respect of that advance;

"Tax Credit" means a credit against, relief or remission for, or repayment of, any Tax;

"Tax Deduction" means a deduction or withholding for or on account of Tax from a payment under a Finance Document;

"Tax Payment" means an increased payment made by an Obligor to a Finance Party under clause 18.2 (Tax Gross-Up) or a payment under clause 18.3 (Tax Indemnity);

"Treaty Lender" means, in relation to a Borrower, a Lender which is beneficially entitled to the interest payable to it and:

- (i) is treated as a resident of a Treaty State for the purposes of the Treaty and is entitled to the benefits of the full exemption under the Treaty from Tax imposed by the United Kingdom (assuming that all procedural formalities have been completed and provided that a Lender shall not be prevented from being a Treaty Lender if its failure to be entitled to such benefit is attributable to the status of, or any action or omission of, the Borrower or any Affiliate thereof or is attributable to any relationship between the Lender and the Borrower or any Affiliate thereof); and
- (ii) does not carry on a business in the United Kingdom through a permanent establishment with which that Lender's participation in the Loan is effectively connected; and

"Treaty State" means a jurisdiction having a double taxation agreement (a "Treaty") with the Borrower's Tax Jurisdiction (or the jurisdiction of source of the interest (if different)) which makes provision for full exemption from tax imposed by the Borrower's Tax Jurisdiction (or the jurisdiction of source of the interest (if different)) on interest; and

"US Lender" means any Lender, if any payment to such Lender may, in whole or in part, be treated as made to a "United States person" as defined in Section 7701(a)(30) of the IR Code for US federal tax purposes under any applicable section of the IR Code or the US Treasury regulations promulgated thereunder.

- (b) In this agreement any reference to a person being required by an applicable law to make a deduction or withholding for or on account of any Tax from any payment under a Finance Document shall be construed as including any circumstances in which a person is authorised under the Income Tax (Jersey) Law 1961 (as amended) to make such a deduction where a failure to allow such deduction would result in a fine being payable under Jersey law and the agreement under which the payment is made being void.
- (c) Unless a contrary indication appears, in this clause 18 a reference to "determines" or "determined" means a determination made in the absolute discretion of the person making the determination.

18.2 Tax Gross-Up

- (a) Each Obligor shall make all payments to be made by it without any Tax Deduction, unless a Tax Deduction is required by law or otherwise under FATCA or a FATCA Agreement.
- (b) Obligors' Agent shall promptly upon becoming aware that an Obligor must make a Tax Deduction (or that there is any change in the rate or the basis of a Tax Deduction) notify the Facility Agent accordingly. Similarly, a Finance Party shall notify the Facility Agent on becoming so aware in respect of a payment payable to that Finance Party. If the Facility Agent receives such notification from a Finance Party it shall notify Obligors' Agent and that Obligor.
- (c) If a Tax Deduction is required by law to be made by an Obligor or Facility Agent on a payment made by or on account of an obligation of an Obligor (and, in the case of a Tax Deduction required to be made by Facility Agent, Facility Agent complies with paragraph (h) below), the applicable Obligor shall increase the amount of such payment to an amount which (after making any such Tax Deduction) leaves an amount received by the applicable Finance Party equal to the payment which such Finance Party would have received if no such Tax Deduction had been required.

- (d) An Obligor is not required to make an increased payment to a Lender under paragraph (c) above for a Tax Deduction imposed by the United Kingdom from a payment of interest on a Loan or a payment as a Guarantor, if on the date on which the payment falls due:
 - (i) the payment could have been made to the relevant Lender without a Tax Deduction if it was a Qualifying Lender, but on that date that Lender is not, or has ceased to be such a Qualifying Lender other than as a result of any change after the date it became a Lender under this agreement in (or in the interpretation, administration, or application of) any law or Treaty, or any published practice or concession of any relevant taxing authority;
 - (ii) the relevant Lender is a Treaty Lender and the Obligor making the payment is able to demonstrate that the payment could have been made to the Lender without the Tax Deduction had that Lender complied with its obligations under paragraph (j) below; or
 - (iii) the Obligor is making the payment as a Guarantor in respect of the obligations of another Obligor (the "Guaranteed Obligor") and the Guaranteed Obligor would not have been liable pursuant to this clause 18.2(d) to make an increased payment had it made such payment itself.
- (e) An Obligor is not required to make an increased payment under paragraph (c) above from a payment in relation to this agreement for a Tax Deduction imposed by Luxembourg, if such a payment falls under the scope of the Luxembourg law of December 23, 2005 as amended, introducing a withholding tax on interest paid to resident individuals.
- (f) Notwithstanding paragraphs (d) and (e) above, an Obligor shall be obliged to make an increased payment to a Lender under paragraph (c) above in respect of a Tax Deduction required by law in the United Kingdom if such Lender is not a Qualifying Lender on the date on which such payment falls due and such Tax Deduction would in any event be required to be made by law at that time if it had been a Qualifying Lender.
- (g) A payment shall not be increased under clause 18.2(c) by reason of a Tax Deduction if on the date when the payment falls due the payment could have been made to the relevant Finance Party without a Tax Deduction if that Finance Party had been a FATCA Compliant Party.
- (h) If an Obligor or the Facility Agent is required to make a Tax Deduction, that Obligor or the Facility Agent, as applicable, shall make that Tax Deduction and any payment required in connection with that Tax Deduction within the time allowed and in the minimum amount required by law.
- (i) Within 30 days of making either a Tax Deduction or any payment required in connection with that Tax Deduction by an Obligor, the Obligor making that Tax Deduction shall deliver to the Facility Agent for the Finance Party entitled to the payment evidence reasonably satisfactory to that Finance Party that the Tax Deduction has been made or (as applicable) any appropriate payment paid to the relevant taxing authority. Within 30 days of making either a Tax Deduction or any payment required in connection with that Tax Deduction by the Facility Agent, the Facility Agent shall deliver to the Obligors' Agent evidence reasonably satisfactory to the Obligors' Agent that the Tax Deduction has been made or (as applicable) any appropriate payment paid to the relevant taxing authority.
- (j) A Treaty Lender and each Obligor which makes a payment to which that Treaty Lender is entitled and an Exempt Lender and each Obligor which makes a payment

to which that Exempt Lender is entitled shall co-operate in completing any procedural formalities necessary for that Obligor to obtain authorisation to make that payment without a Tax Deduction (including, without limitation, to the extent reasonably practicable, making and filing an appropriate application for relief and giving any appropriate confirmations provided that any Lender's compliance with the provisions of clause 18.2(k) below shall be treated as the Lender satisfying its obligations under this clause 18.2(j)).

- (k) A Treaty Lender which holds a passport under the HMRC DT Treaty Passport scheme, and which wishes that scheme to apply to this agreement, shall confirm its scheme reference number and its jurisdiction of tax residence in the documentation which it executes on becoming a Party as a Lender and, having done so, that Lender shall be under no obligation pursuant to paragraph (j) above.
- (I) If a Lender has confirmed its scheme reference number and its jurisdiction of tax residence in accordance with clause 18.2(k) above and:
 - a Borrower making a payment to that Lender has not made a Borrower DTTP Filing in respect of that Lender; or
 - (ii) a Borrower making a payment to that Lender has made a Borrower DTTP Filing in respect of that Lender but:
 - (A) that Borrower DTTP Filing has been rejected by HM Revenue & Customs; or
 - (B) HM Revenue & Customs has not given the Borrower authority to make payments to that Lender without a Tax Deduction within 75 days of the date of the Borrower DTTP Filing or such authorisation has ceased to apply,

and in each case, the Borrower has notified that Lender in writing, that Lender and the Borrower shall co-operate in completing any additional procedural formalities necessary for that Borrower to obtain authorisation to make that payment without a Tax Deduction

- (m) An Obligor is not required to make an increased payment under paragraph (c) above:
 - (i) for any Tax Deduction for or attributable to US federal withholding (or backup withholding) Taxes imposed on amounts payable by or on account of an obligation of a US Tax Obligor in relation to a Revolving Facility pursuant to a law in effect on the date on which (x) the relevant Lender acquires the relevant interest in a Revolving Facility Loan or Revolving Facility Commitment or the relevant Lender became a Lender (other than pursuant to an assignment request by an Obligor under Clause 40.3) or (y) the relevant Lender changes its lending office, except in each case to the extent that, pursuant to paragraph (c) above, amounts with respect to such Tax Deduction were payable either to such Lender's assignor immediately before such Lender became a party hereto or to such Lender immediately before it changed its lending office;
 - (ii) for any Tax Deduction imposed by the US to the extent that the relevant payment could have been made to the relevant Finance Party without the Tax Deduction had that Finance Party complied with its obligations under Clause 18.9 (Tax Documentation) and the representations and warranties therein had been accurate; or
 - (iii) with respect to any Finance Party (other than a Lender) or any US Lenders, for any Tax Deduction attributable to US backup withholding.

(n) Notwithstanding anything to the contrary in this agreement, (i) an Ancillary Lender (other than the Existing Ancillary Lender, with respect to the Existing Ancillary Facility, provided that the Existing Ancillary Lender complies with the requirements of clauses 18 and 19 and any other relevant provisions of this agreement) shall not be entitled to the benefits of clause 18.2(c), 18.3(a) or 19.1, or any other indemnity under this agreement with respect to Taxes in respect of the applicable Ancillary Documents or imposed with respect to payments made pursuant to the applicable Ancillary Documents, except to the extent provided in the applicable Ancillary Documents, and (ii) payments of fees pursuant to any fee letter executed in connection with or anticipation of the 2022 Amendment Request shall not be subject to clause 18.2(c), 18.3(a) or 19.1 or any other indemnity under this agreement with respect to Taxes (but instead shall be subject to the terms of the applicable fee letter).

18.3 Tax Indemnity

- (a) Obligors' Agent shall (within three Business Days of demand by the Facility Agent) pay to a Protected Party an amount equal to the loss, liability or cost which that Protected Party determines will be or has been (directly or indirectly) suffered for or on account of Tax by that Protected Party in respect of a Finance Document as a result of the introduction of, or a change in the interpretation or application of, any law or regulation relating to Tax occurring after the date of this agreement (or compliance by any Finance Party with any such law or regulation).
- (b) Paragraph (a) above shall not apply:
 - (i) with respect to any Tax assessed on a Finance Party:
 - under the law of the jurisdiction in which that Finance Party is incorporated or formed or, if different, the jurisdiction (or jurisdictions) in which that Finance Party is treated as resident for tax purposes; or
 - (B) under the law of the jurisdiction in which that Finance Party's Facility Office is located in respect of amounts received or receivable in that jurisdiction,

if that Tax is imposed on or calculated by reference to the net income received or receivable (but not any sum deemed to be received or receivable) by that Finance Party. For the avoidance of doubt, paragraph (a) shall not apply with respect to any such loss, liability, or cost arising to a Finance Party as a result of any Tax imposed solely as a result of that Finance Party or any of its Affiliates having an Effective Tax Rate below the Minimum Rate as defined in the Global Anti Base Erosion Rules.

- (ii) to the extent a loss, liability or cost:
 - is compensated for by an increased payment under clause 18.2 (Tax Gross-Up) or clause 14.5 (Minimum Interest);
 - (B) would have been compensated for by an increased payment under clause 18.2 (Tax Gross-Up) but was not so compensated solely because one of the exclusions in clause 18.2(d), (e), and (m) (Tax Gross-Up) applied;
 - (C) is related to FATCA;
- (c) A Protected Party making, or intending to make a claim under paragraph (a) above shall promptly notify the Facility Agent of the event which will give, or has given, rise to the claim, following which the Facility Agent shall notify Obligors' Agent.

(d) A Protected Party shall, on receiving a payment from an Obligor under this clause 18.3, notify the Facility Agent.

18.4 Tax Credit

- (a) If an Obligor makes a Tax Payment and the relevant Finance Party determines in good faith that:
 - (i) a Tax Credit is attributable to all or part of that Tax Payment; and
 - (ii) that Finance Party has obtained, utilised and retained that Tax Credit,

the Finance Party shall pay an amount to the Obligor which that Finance Party determines in good faith will leave it (after that payment) in the same after-Tax position as it would have been in had the Tax Payment not been made by the Obligor.

18.5 Stamp Taxes

The Parent shall pay and, within three Business Days of demand, indemnify each Finance Party against any cost, loss or liability that Finance Party incurs in relation to all stamp duty, registration and other similar Taxes payable in respect of any Finance Document provided that this clause 18.5 shall not apply in respect of:

- (a) in relation to Luxembourg registration duties, in the case of voluntary registration of this agreement or any document or agreement in relation thereto, by a Finance Party or any parties other than the Obligors where such voluntary registration is not necessary to enforce, maintain or preserve the rights of any party; and
- (b) any stamp duty, registration or other similar Taxes payable in respect of an assignment or transfer or sub-participation or sub-contract by a Lender of any of its rights or obligations under a Finance Document.

18.6 Value Added Tax

- (a) All consideration expressed to be payable under a Finance Document by any Party to a Finance Party shall be deemed to be exclusive of any VAT. Subject to paragraph (b) below, if VAT is chargeable on any supply made by any Finance Party to any Party in connection with a Finance Document, that Party shall pay to the Finance Party (in addition to and at the same time as paying the consideration) an amount equal to the amount of the VAT (and such Finance Party must promptly provide an appropriate VAT invoice to that Party).
- (b) If VAT is chargeable on any supply made by any Finance Party (the "Supplier") to any other Finance Party (the "Recipient") in connection with a Finance Document, and any Party is required by the terms of any Finance Document to pay an amount equal to the consideration for such supply to the Supplier, such Party shall also pay to the Supplier (in addition to and at the same time as paying such amount) an amount equal to the amount of such VAT.
- (c) Where a Finance Document requires any Party to reimburse a Finance Party for any costs or expenses, that Party shall also at the same time pay and indemnify the Finance Party against all VAT incurred by the Finance Party in respect of the costs or expenses to the extent that the Finance Party reasonably determines that it is not entitled to credit or repayment of the VAT.

18.7 Notification to Obligors' Agent and Facility Agent

Each Lender will promptly notify Obligors' Agent and the Facility Agent if it is not or ceases to be a Qualifying Lender.

18.8 FATCA Withholding

- (a) Notwithstanding anything to the contrary in this agreement, if a Facility Agent or Finance Party is required to make a Tax Deduction under FATCA or a FATCA Agreement then that Facility Agent or Finance Party may make such Tax Deduction.
- (b) The Facility Agent shall not be required to increase any payment made by it under or in connection with a Finance Document in the event the Facility Agent is required to make a Tax Deduction on the basis of FATCA or a FATCA Agreement in respect of such payment.
- (c) If the Facility Agent makes a payment to a Lender under clause 34.2 (Distributions by the Facility Agent) in respect of which it was (at any time) required under FATCA or a FATCA Agreement to make a FATCA Deduction, that Lender shall indemnify the Facility Agent against any liability that the Facility Agent may have to any governmental or taxation authority in relation to that payment.

18.9 Tax Documentation

- (a) If an Obligor or the Facility Agent reasonably believes that, in order to comply with FATCA or a FATCA Agreement, it requires information as to the status under FATCA or a FATCA Agreement of one or more Finance Parties, the Lender or, in the case of a request for information in respect of another Finance Party, such other Finance Party shall provide such Obligor or the Facility Agent with such information as it may reasonably request or alternatively advise such Obligor or the Facility Agent of its non-compliant status under FATCA. An Obligor or Facility Agent may only request information pursuant to this clause after 30 June 2013. The relevant Lender or, in the case of a request for information in respect of another Finance Party, such Finance Party shall within 20 Business Days of receipt of the request from the Obligor or the Facility Agent, as the case may be, either:
 - (i) provide the information requested; or
 - (ii) confirm to the party which requested the relevant information that it believes that its status under FATCA is such that payments made to it under the Finance Documents are or will be subject to a Tax Deduction on the basis of FATCA or a FATCA Agreement. A Finance Party shall only be required to comply with a request for information pursuant to this paragraph (a) to the extent that such compliance would not involve that Finance Party in any unlawful activity or any activity that is contrary to the policies of that Finance Party or any request, guidance or directive of any competent authority (whether or not having the force of law) and/or would not breach any contractual confidentiality or data protection obligations to which it is subject.
- (b) The Facility Agent may disclose any information received by it from a Finance Party pursuant to paragraph (a) above to the Company, provided that such disclosure would not involve the Facility Agent in any unlawful activity or any activity that is contrary to the policies of the Facility Agent or any request, guidance or directive of any competent authority (whether or not having the force of law) and would not breach any contractual confidentiality or data protection obligations it is subject to.
- (c) If a Lender or Facility Agent reasonably believes that in order to comply with FATCA or a FATCA Agreement, it requires information as to the status under FATCA of an Obligor then such Obligor shall provide that Lender or Facility Agent with such information as it may reasonably request. A Lender or Facility Agent may only request information pursuant to this clause after 30 June 2013. The relevant Obligor shall provide the information requested within 20 Business Days of receipt of the request from the relevant Lender or Facility Agent. An Obligor shall only be required

to comply with a request for information pursuant to this paragraph (c) to the extent that such compliance would not involve that Obligor in any unlawful activity or any activity that is contrary to any request, guidance or directive of any competent authority (whether or not having the force of law) and / or would not breach any contractual confidentiality or data protection obligations to which it is subject.

- (d) If an Obligor or the Facility Agent reasonably believes that its obligations under laws and regulations applicable to US withholding (including backup withholding) or information reporting require it, each Finance Party that is entitled to an exemption from or reduction of US withholding (including backup withholding) with respect to payments made under any Finance Document shall (and each Finance Party that has entered into a sub-participation or sub-contract of its rights and/or obligations under the Finance Documents shall procure its relevant sub-participants to) deliver to the Obligors' Agent and the Facility Agent, at the time or times reasonably requested by the Obligors' Agent or the Facility Agent:
 - (i) (A) a valid and properly executed Internal Revenue Service Form W-9 establishing an exemption from US backup withholding or (B) a valid and properly executed Internal Revenue Service Form W-8 (which, in the case of a Form W-8IMY, will be accompanied by Form W-8ECI, Form W-8BEN, Form W-8BEN-E or W-9 and/or other certification documents from each beneficial owner), as applicable (or, in each case, any successor form); and
 - (ii) such other documentation prescribed by applicable law or reasonably requested by the Obligors' Agent or the Facility Agent as will enable the Obligors' Agent or the Facility Agent to determine whether or not such Finance Party or sub-participant is subject to US backup withholding or information reporting requirements.

The Facility Agent shall not be required to increase any payment made by it under or in connection with a Finance Document in the event the Facility Agent is required to make a deduction due to US withholding (including backup withholding) in respect of such payment (including with respect to a Finance Party that is entitled to an exemption from or reduction of US backup withholding but which fails to comply with this clause 18.9(d)).

- (e) If an Obligor is a US Tax Obligor, any Finance Party that is entitled to an exemption from or reduction of US Tax (without duplication of paragraph (d) above) with respect to payments made under any Finance Document shall (and each Finance Party that has entered into a sub-participation or sub-contract of its rights and/or obligations under the Finance Documents shall procure its relevant sub-participants to), to the extent it is legally entitled to do so, deliver to the Obligors' Agent and the Facility Agent, at the time or times reasonably requested by the Obligors' Agent or the Facility Agent as will permit such payments to be made without such US withholding Tax or at a reduced rate of such US withholding Tax.
- (f) Without duplication of Lender's obligations described in the foregoing clauses (a) through (e), if the Facility Agent reasonably believes that its obligations under FATCA or any other applicable law or regulation require it, each Lender shall, within ten Business Days of the date of a request from the Facility Agent, supply (and each Lender that has entered into a sub-participation or sub-contract of its rights and/or obligations under the Finance Documents shall procure its relevant sub-participants to supply) to the Facility Agent:

- (i) a valid, properly completed and executed withholding certificate or Internal Revenue Service Form W-8 or Internal Revenue Service Form W-9 or any successor form; or
- (ii) any withholding statement or other document, authorisation or waiver as the Facility Agent may require to certify or establish the status of such Lender under FATCA or that other law or regulation.
- (g) If any withholding certificate, withholding statement, document, authorisation or waiver provided to the Facility Agent or the Obligors' Agent by a Finance Party or its sub-participant pursuant to this clause 18.9 is or becomes materially inaccurate or incomplete, or expires or becomes obsolete, that Finance Party shall (or shall procure the relevant sub-participant to) promptly update it and provide such updated withholding certificate, withholding statement, document, authorisation or waiver to the Facility Agent unless the Finance Party is legally ineligible to do so (in which case the Finance Party shall promptly notify the Facility Agent and the Facility Agent shall notify the Obligors' Agent). The Facility Agent shall provide any such updated withholding certificate, withholding statement, document, authorisation or waiver to the Obligors' Agent.
- (h) The Facility Agent shall provide any withholding certificate, withholding statement, document, authorisation, waiver, form or other information it receives from a Finance Party pursuant to paragraphs (a), (d), (e), (f) or (g) above, or paragraph (m) below, to the relevant Borrower.
- (i) The Facility Agent may rely on any withholding certificate, withholding statement, document, authorisation or waiver it receives from a Finance Party pursuant to paragraph (d), (e), (f) or (g) above without further verification. The Facility Agent shall not be liable for any losses resulting from any action taken by it under or in connection with paragraphs (d), (e), (f), (g) or (h) above.
- (j) Notwithstanding anything in this clause 18.9 to the contrary, the completion, execution and submission of such documentation (other than an Internal Revenue Service Form W-9 or applicable Internal Revenue Service Form W-8, or any successor form) shall not be required if in the Lender's (or its relevant sub-participants') reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender.
- On the 2022 Effective Date or as soon as reasonably practicable thereafter (and from time to time thereafter upon the reasonable request of the Obligors' Agent), the Facility Agent, each Arranger, the Security Agent, the Existing Ancillary Lender, and each Lender with an RCF Commitment shall provide to the Obligors' Agent a valid and properly executed Internal Revenue Service Form W-9 (establishing that such Person is exempt from US backup withholding) or, if such Person is not a United States person within the meaning of Section 7701(a)(30) of the IR Code, an Internal Revenue Service Form W-8, as applicable (or, in each case, any successor form). Any such Person that does not provide an Internal Revenue Service Form W-9 or W-8ECI (or successor form), and any other Finance Party that receives fees under any fee letter executed in connection with or anticipation of the 2022 Amendment Request or 2024 Amendment Request and does not provide an Internal Revenue Service Form W-9 or W-8ECI (or successor form), hereby represents and warrants that any fees under any Finance Document or related fee letter received by it are not for personal services performed in the United States (within the meaning of Section 861 of the IR Code or related US Treasury regulations promulgated under the IR Code). If the documentation provided to the Obligors' Agent pursuant to this paragraph (k) is or becomes materially inaccurate or incomplete, or expires or becomes obsolete, the Facility Agent, Arranger, Security Agent or Lender, as

applicable, shall promptly update it and provide such updated documentation to the Obligors' Agent, unless such Person is legally ineligible to do so (in which case such Person shall promptly notify the Obligors' Agent).

- (I) Without duplication of any Party's obligations described in the foregoing clauses (a) through (k), each Party shall, within ten Business Days of a reasonable request by another Party, supply to that other Party such forms, documentation and other information relating to its status as that other Party reasonably requests for the purposes of that other Party's compliance with any other law, regulation, or exchange of information regime provided always that this paragraph (I) shall not oblige any Party to do anything which would or might in its reasonable opinion constitute a breach of any law, regulation, fiduciary duty, or duty of confidentiality.
- (m) Upon reasonable written request by any Obligor or the Obligors' Agent to the Facility Agent, the Facility Agent shall reasonably cooperate with such Obligor or the Obligors' Agent to effectuate the provisions of this clause 18.9 (including by requesting forms, documents or other information under this clause 18.9 from other Finance Parties on behalf of such Obligor or the Obligor's Agent as reasonably requested by such Obligor or the Obligor's Agent) provided that the Facility Agent shall not be deemed to have breached its obligations under this Clause 18.9(m) in the event of any failure to effectuate the provisions of this clause 18.9.

19. INCREASED COSTS

19.1 Increased Costs

- (a) Subject to clause 19.3 (Exceptions) the Obligors shall, within thirty (30) Business Days of a demand by the Facility Agent, pay for the account of a Finance Party the amount of any Increased Costs incurred by that Finance Party or any of its Affiliates as a result of the introduction of or any change in (or in the interpretation, administration or application of) any law or regulation after the 2024 Effective Date or compliance with any law or regulation made after the 2024 Effective Date that are, in each case, being passed on by the relevant Finance Party to similar borrowers of similar facilities.
- (b) In this agreement:

"Basel III" means:

- (i) The agreements on capital requirements, a leverage ratio and liquidity standards container in "Basel III: A global regulatory framework for more resilient banks and banking systems", "Basel III: International framework for liquidity risk measurement, standards and monitoring" and "Guidance for national authorities operating the countercyclical capital buffer" published by the Basel Committee on Banking Supervision in December 2010, each as amended, supplemented or restated;
- (ii) The rules for global systemically important banks contained in "Global systemically important banks: assessment methodology and the additional loss absorbency requirement – Rules text" published by the Basel Committee on Banking Supervision in November 2011, as amended, supplemented or restated; and
- (iii) any further guidance or standards published by the Basel Committee on Banking Supervision relating to "Basel III".

"Increased Costs" means:

- (i) a reduction in the rate of return from a Facility or on a Finance Party's (or its Affiliate's) overall capital;
- (ii) an additional or increased cost; or
- (iii) a reduction of any amount due and payable under any Finance Document or Letter of Credit,

which is incurred or suffered by a Finance Party or any of its Affiliates to the extent that it is attributable to that Finance Party having entered into its Commitment or an Ancillary Commitment or funding or performing its obligations under any Finance Document or Letter of Credit.

19.2 Increased Cost Claims

- (a) A Finance Party intending to make a claim pursuant to clause 19.1 (Increased Costs) shall notify the Facility Agent of the event giving rise to the claim, following which the Facility Agent shall promptly notify Obligors' Agent.
- (b) Each Finance Party shall, as soon as practicable after a demand by the Facility Agent, provide a certificate confirming the amount of its Increased Costs.

19.3 Exceptions

- (a) Clause 19.1 (Increased Costs) does not apply to the extent that any Increased Cost is:
 - (i) attributable to a Tax Deduction required by law to be made;
 - (ii) compensated for by clause 18.3 (Tax Indemnity) (or would have been compensated for under clause 18.3 (Tax Indemnity) but was not so compensated solely because any of the exclusions in clause 18.3(b) (Tax Indemnity) applied);
 - (iii) attributable to the wilful breach by the relevant Finance Party or its Affiliates of any law or regulation;
 - (iv) attributable to the implementation or application of or compliance with the "International Convergence of Capital measurement and Capital Standards, a Revised Framework" published by the Basel Committee on Banking Supervision in June 2004 in the form existing on the date of this agreement (but excluding any amendment arising out of Basel III) ("Basel II") or any other law or regulation which implements Basel II (whether such implementation, application or compliance is by a government, regulator, Finance Party or any of its Affiliates).
 - (v) not notified to the Facility Agent within six months of the relevant Finance Party becoming aware thereof.
- (b) In this clause 19.3, a reference to a **"Tax Deduction"** has the same meaning given to the term in clause 18.1 (Definitions).
- (c) It is understood and agreed that the Dodd-Frank Wall Street Reform and Consumer Protection Act (Pub.L. 111-203, H.R. 4173) and the implementation or application of or compliance with Basel III, and all laws relating thereto, all interpretations and applications thereof and any compliance by a Lender with any request or directive relating thereto, shall, for the purposes of this agreement, be deemed to be adopted subsequent to the 2024 Effective Date.

20. OTHER INDEMNITIES

20.1 Currency Indemnity

- (a) If any sum due from an Obligor under the Finance Documents (a "Sum"), or any order, judgment or award given or made in relation to a Sum, has to be converted from the currency (the "First Currency") in which that Sum is payable into another currency (the "Second Currency") for the purpose of:
 - (i) making or filing a claim or proof against that Obligor; or
 - (ii) obtaining or enforcing an order, judgment or award in relation to any litigation or arbitration proceedings,

that Obligor shall as an independent obligation, within three Business Days of demand, indemnify the Arrangers and each other Secured Party to whom that Sum is due against any cost, loss or liability arising out of or as a result of the conversion including any discrepancy between:

- the rate of exchange used to convert that Sum from the First Currency into the Second Currency; and
- (B) the rate or rates of exchange available to that person at the time of its receipt of that Sum.
- (b) Each Obligor waives any right it may have in any jurisdiction to pay any amount under the Finance Documents in a currency or currency unit other than that in which it is expressed to be payable.

20.2 Other Indemnities

The Obligors' Agent shall (or shall procure that an Obligor will) indemnify each Finance Party against any cost, loss or liability incurred by it as a result of:

- (a) the occurrence of any Event of Default;
- a failure by an Obligor to pay any amount due under a Finance Document on its due date, including without limitation, any cost, loss or liability arising as a result of clause 33 (Sharing Among the Finance Parties);
- funding, or making arrangements to fund, its participation in a Utilisation requested by a Borrower in a Utilisation Request but not made in contravention of this agreement (other than by reason of default or negligence by that Finance Party alone);
- (d) issuing or making arrangements to issue a Letter of Credit requested by Obligors' Agent or a Borrower in a Utilisation Request but not issued in contravention of the provisions of this agreement;
- (e) a Utilisation (or part of a Utilisation) not being prepaid in accordance with a notice of prepayment in contravention of this agreement given by a Borrower or Obligors' Agent.

provided that such indemnity shall not, as to any Finance Party, be available to the extent that such cost, loss or liability (i) is determined by a court of competent jurisdiction by final and nonappealable judgment to have (x) resulted from the gross negligence, bad faith or wilful misconduct of such Finance Party or (y) arisen from a material breach of a Finance Document in bad faith by such Finance Party or (ii) result from a dispute solely among Finance Parties (other than any disputes

involving claims against any agent, arranger or bookrunner, in each case in their respective capacities as such) that did not involve actions or omissions of any Group Company.

20.3 Indemnity to the Facility Agent

The Obligors' Agent shall (or shall procure that an Obligor will) indemnify the Facility Agent against any cost, loss or liability incurred by the Facility Agent (acting reasonably) as a result of:

- (a) investigating any event which it reasonably believes is a Default;
- (b) entering into or performing any foreign exchange contract for the purposes of clause 34.9(b) (Change of Currency);
- (c) acting or relying on any notice, request or instruction which it reasonably believes to be genuine, correct and appropriately authorised under the Finance Documents;
- instructing lawyers, accountants, tax advisers, surveyors or other professional advisers or experts as permitted under this agreement and related to the Finance Documents (for the avoidance of doubt, subject to Clause 20.5 (Indemnity Cost Limitations)); or
- (e) any cost, loss or liability incurred by the Facility Agent in acting as Agent under the Finance Documents;

provided that such indemnity shall not, as to Facility Agent, be available to the extent that such cost, loss or liability (i) is determined by a court of competent jurisdiction by final and nonappealable judgment to have (x) resulted from the gross negligence, bad faith or wilful misconduct of the Facility Agent or (y) arisen from a material breach of a Finance Document in bad faith by such Facility Agent or (ii) result from a dispute solely among Finance Parties (other than any disputes involving claims against any agent, arranger or bookrunner, in each case in their respective capacities as such) that did not involve actions or omissions of any Group Company.

20.4 Indemnity to the Security Agent

- (a) Subject to clause 20.4(b), the Obligors' Agent shall (or shall procure that an Obligor will) indemnify the Security Agent and every Receiver and Delegate against any cost, loss or liability incurred by any of them as a result of:
 - (i) the taking, holding, protection or enforcement of the Transaction Security;
 - (ii) the exercise of any of the rights, powers, discretions and remedies vested in the Security Agent and each Receiver and Delegate by the Finance Documents or by law; and
 - (iii) any default by any Obligor in the performance of any of the obligations expressed to be assumed by it in the Finance Documents.
- (b) The indemnity granted by ASM pursuant to clause 20.4(a) above shall be limited to any cost, loss or liability incurred by the Security Agent and every Receiver and Delegate as a result of:
 - (i) the taking, holding, protection or enforcement of the Transaction Security granted by ASM;
 - (ii) the exercise of any of the rights, powers, discretions and remedies in relation to ASM vested in the Security Agent and each Receiver and Delegate by the Finance Documents or by law; and

- (iii) any default by ASM in the performance of any of the obligations expressed to be assumed by it in the Finance Documents.
- (c) Subject to clause 20.4(b), the Security Agent may, in priority to any payment to the Secured Parties, indemnify itself out of the Charged Property in respect of, and pay and retain, all sums necessary to give effect to the indemnity in this clause 20.4 and shall have a lien on the Transaction Security and the proceeds of the enforcement of the Transaction Security for all monies payable to it.
- (d) Notwithstanding anything in this Clause 20.4 to the contrary, such indemnity shall not, as to the Security Agent, any Receiver or any Delegate, be available to the extent that such cost, loss or liability (i) are determined by a court of competent jurisdiction by final and nonappealable judgment to have (x) resulted from the gross negligence, bad faith or wilful misconduct of the Security Agent, the Receiver or the Delegate or (y) arisen from a material breach of a Finance Document in bad faith by the Security Agent, the Receiver or the Delegate or (ii) result from a dispute solely among the Security Agent, the Receiver or the Delegate that did not involve actions or omissions of any Group Company.

20.5 Indemnity Cost Limitations

Notwithstanding anything in this Clause 20 (Other Indemnities) to the contrary, the Group Companies' indemnity obligations shall only cover those costs that are reasonable and documented and, in the case of counsel, shall be limited to not more than one counsel for all such parties subject to indemnity pursuant to this Clause 20 (Other Indemnities), taken as a whole, and, if necessary, a single local counsel in each applicable jurisdiction (and, in the case of an actual or potential conflict of interest where such party affected by such conflict informs the Parent of such conflict and thereafter retains its own counsel, of another firm of counsel for such affected party).

21. MITIGATION BY THE LENDERS

21.1 Mitigation

- (a) Each Finance Party shall, in consultation with Obligors' Agent, take all reasonable steps to mitigate any circumstances which arise and which would result in any amount becoming payable under or pursuant to, or cancelled pursuant to, any of clause 11.1 (Illegality) (or, in respect of the Issuing Bank, clause 11.2 (Illegality in Relation to Issuing Bank)), clause 14.5 (Minimum Interest), clause 18 (Tax Gross-Up and Indemnities) or clause 19 (Increased Costs) including (but not limited to) transferring its rights and obligations under the Finance Documents to another Affiliate or Facility Office.
- (b) Paragraph (a) above does not in any way limit the obligations of any Obligor under the Finance Documents.

21.2 Limitation of Liability

- (a) The Obligors shall indemnify each Finance Party for all documented costs and expenses reasonably incurred by that Finance Party as a result of steps taken by it under clause 21.1 (Mitigation).
- (b) A Finance Party is not obliged to take any steps under clause 21.1 (Mitigation) if, in the opinion of that Finance Party (acting reasonably), to do so might be prejudicial to it.

22. **COSTS AND EXPENSES**

22.1 Transaction Expenses

The Obligors' Agent shall or shall procure that another Obligor shall pay the Facility Agent, the Arrangers, the Issuing Bank and the Security Agent the amount of all costs and expenses (limited, in the case of legal fees, to the reasonable and documented legal fees of one primary counsel to all such parties and, if necessary, a single local counsel to all such parties in each applicable jurisdiction) reasonably incurred by any of them (and, in the case of the Security Agent, by any Receiver or Delegate) in connection with the negotiation, preparation, printing, execution, primary syndication and perfection of:

- (a) this agreement and any other documents referred to in this agreement and the Transaction Security;
- (b) any other Finance Documents (other than any Transfer Certificate prepared after the closing of primary syndication) executed after the date of this agreement.

22.2 Amendment Costs

If:

- (a) an Obligor requests an amendment, waiver or consent in relation to a Finance Document; or
- (b) an amendment is required pursuant to clause 34.9 (Change of Currency),

the Obligors' Agent shall, or shall procure that another Obligor shall, within a reasonable time after demand, reimburse each of the Facility Agent and the Security Agent for the amount of all documented costs and expenses (limited, in the case of legal fees, to the reasonable and documented legal fees of one primary counsel to all such parties and, if necessary, a single local counsel in each applicable jurisdiction to all such parties) reasonably incurred by the Facility Agent and the Security Agent (and, in the case of the Security Agent, by any Receiver or Delegate) in responding to, evaluating, negotiating or complying with that request or requirement.

22.3 Enforcement and Preservation Costs

The Obligors shall, within a reasonable time after demand, pay to the Arrangers and each other Secured Party the amount of all costs and expenses (limited, in the case of legal fees, to the reasonable and documented legal fees of one primary counsel to all such parties and, if necessary, a single local counsel in each applicable jurisdiction to all such parties) incurred by it in connection with the enforcement of or the preservation of any rights under any Finance Document and the Transaction Security and any proceedings instituted by or against the Security Agent as a consequence of taking or holding the Transaction Security or enforcing these rights.

23. **GUARANTEE AND INDEMNITY**

23.1 Guarantee and Indemnity

Subject to clauses 23.12 (Guarantee Limitations) and 23.13 (Limitations under Swiss Law), each Guarantor irrevocably and unconditionally jointly and severally:

- guarantees to each Finance Party punctual performance by each other Obligor of all that Obligor's obligations under the Finance Documents;
- (b) undertakes with each Finance Party that whenever another Obligor does not pay any amount when due under or in connection with any Finance Document, that Guarantor shall immediately on demand pay that amount as if it was the principal obligor; and
- (c) indemnifies each Finance Party immediately on demand against any cost, loss or liability suffered by that Finance Party if any obligation guaranteed by it is or becomes

unenforceable, invalid or illegal. The amount of the cost, loss or liability shall be equal to the amount which that Finance Party would otherwise have been entitled to recover.

23.2 Continuing Guarantee

This guarantee is a continuing guarantee and will extend to the ultimate balance of sums payable by any Obligor under the Finance Documents, regardless of any intermediate payment or discharge in whole or in part.

23.3 Reinstatement

If any payment by an Obligor or any discharge given by a Finance Party (whether in respect of the obligations of any Obligor or any security for those obligations or otherwise) is avoided or reduced as a result of insolvency or any similar event:

- (a) the liability of each Obligor shall continue as if the payment, discharge, avoidance or reduction had not occurred; and
- (b) each Finance Party shall be entitled to recover the value or amount of that security or payment from each Obligor, as if the payment, discharge, avoidance or reduction had not occurred.

23.4 Waiver of Defences

The obligations of each Guarantor under this clause 23 will not be affected by an act, omission, matter or thing which, but for this clause 23, would reduce, release or prejudice any of its obligations under this clause 23 (without limitation and whether or not known to it or any Finance Party) including:

- (a) any time, waiver or consent granted to, or composition with, any Obligor or other person;
- (b) the release of any other Obligor or any other person under the terms of any composition or arrangement with any creditor of any Group Company;
- (c) the taking, variation, compromise, exchange, renewal or release of, or refusal or neglect to perfect, take up or enforce, any rights against, or security over assets of, any Obligor or other person or any non-presentation or non-observance of any formality or other requirement in respect of any instrument or any failure to realise the full value of any security;
- (d) any incapacity or lack of power, authority or legal personality of or dissolution or change in the members or status of an Obligor or any other person;
- (e) any amendment, novation, supplement, extension (whether of maturity or otherwise) or restatement (in each case, however fundamental and of whatsoever nature) or replacement of a Finance Document or any other document or security;
- (f) any unenforceability, illegality or invalidity of any obligation of any person under any Finance Document or any other document or security; or
- (g) any insolvency or similar proceedings.

23.5 **Guarantor Intent**

Without prejudice to the generality of clause 23.4 (Waiver of Defences), each Guarantor expressly confirms that it intends that this guarantee shall extend from time to time to any (however fundamental) variation, increase, extension or addition of or to any of the Finance

Documents and/or any facility or amount made available under any of the Finance Documents for the purposes of or in connection with any of the following: acquisitions of any nature; increasing working capital; enabling investor distributions to be made; carrying out restructurings; refinancing existing facilities; refinancing any other indebtedness; making facilities available to new borrowers; any other variation or extension of the purposes for which any such facility or amount might be made available from time to time; and any fees, costs and/or expenses associated with any of the foregoing.

23.6 Immediate Recourse

Each Guarantor waives any right it may have of first requiring any Finance Party (or any trustee or agent on its behalf) to proceed against or enforce any other rights or security or claim payment from any person before claiming from that Guarantor under this clause 23. This waiver applies irrespective of any law or any provision of a Finance Document to the contrary.

23.7 Waiver of Droits

Each Guarantor irrevocably and unconditionally abandons and waives any right which it may have at any time under the existing or future laws of Jersey:

- (a) whether by virtue of the droit de discussion or otherwise to require that recourse be had by any Finance Party (or any trustee or agent on its behalf) to the assets of any other Guarantor or any other person before any claim is enforced against that Guarantor in respect of the obligations assumed by it under any of the Finance Documents; and
- (b) whether by virtue of the droit de division or otherwise to require that any liability under any of the Finance Documents be divided or apportioned with any other Guarantor or any other person or be reduced in any manner whatsoever.

23.8 Appropriations

Until all amounts which may be or become payable by the Obligors under or in connection with the Finance Documents have been irrevocably paid in full (other than contingent indemnification and expense reimbursement claims not then due), each Finance Party (or any trustee or agent on its behalf) may:

- (a) refrain from applying or enforcing any other moneys, security or rights held or received by that Finance Party (or any trustee or agent on its behalf) in respect of those amounts, or apply and enforce the same in such manner and order as it sees fit (whether against those amounts or otherwise) and no Guarantor shall be entitled to the benefit of the same; and
- (b) hold in an interest-bearing suspense account any moneys received from any Guarantor or on account of any Guarantor's liability under this clause 23.

23.9 **Deferral of Guarantors' Rights**

Until all amounts which may be or become payable by the Obligors under or in connection with the Finance Documents have been irrevocably paid in full (other than contingent indemnification and expense reimbursement claims not then due) and unless the Facility Agent otherwise directs, no Guarantor will exercise any rights which it may have by reason of performance by it of its obligations under the Finance Documents:

- (a) to be indemnified by an Obligor;
- (b) to claim any contribution from any other guarantor of any Obligor's obligations under the Finance Documents; and/or

- (c) to take the benefit (in whole or in part and whether by way of subrogation or otherwise) of any rights of the Finance Parties under the Finance Documents or of any other guarantee or security taken pursuant to, or in connection with, the Finance Documents by any Finance Party;
- to bring legal or other proceedings for an order requiring any Obligor to make any payment, or perform any obligation, in respect of which any Guarantor has given a guarantee, undertaking or indemnity under Clause 23.1 (Guarantee and indemnity);
- (e) to exercise any right of set-off against any Obligor; and/or
- (f) to claim or prove as a creditor of any Obligor in competition with any Finance Party.

If a Guarantor receives any benefit, payment or distribution in relation to such rights it shall hold that benefit, payment or distribution to the extent necessary to enable all amounts which may be or become payable to the Finance Parties by the Obligors under or in connection with the Finance Documents to be repaid in full on trust for the Finance Parties and shall promptly pay or transfer the same to the Facility Agent or as the Facility Agent may direct for application in accordance with Clause 34 (Payment Mechanics).

23.10 Release of Guarantors' Right of Contribution

If any Guarantor (a "Retiring Guarantor") ceases to be a Guarantor in accordance with the terms of the Finance Documents for the purpose of any sale or other disposal of that Retiring Guarantor then on the date such Retiring Guarantor automatically ceases to be a Guarantor:

- (a) that Retiring Guarantor is released by each other Guarantor from any liability (whether past, present or future and whether actual or contingent) to make a contribution to any other Guarantor arising by reason of the performance by any other Guarantor of its obligations under the Finance Documents; and
- (b) each other Guarantor:
 - (i) waives any rights it may have by reason of the performance of its obligations under the Finance Documents to take the benefit (in whole or in part and whether by way of subrogation or otherwise) of any rights of the Finance Parties under any Finance Document or of any other security taken pursuant to, or in connection with, any Finance Document where such rights or security are granted by or in relation to the assets of the Retiring Guarantor; and
 - (ii) confirms its obligations in this clause 23 will continue in respect of the remaining Obligors notwithstanding the release of the Retiring Guarantor.

23.11 Additional Security

This guarantee is in addition to and is not in any way prejudiced by any other guarantee or security now or subsequently held by any Finance Party.

23.12 **Guarantee Limitations**

(a) Guarantee Obligations

For the purpose of this clause 23.11:

(i) "Guarantee Obligations" means the obligations and liabilities of the relevant Obligor under (A) clause 23.1 (Guarantee and Indemnity) (B) clause 19 (Guarantee and Indemnity) of the Second Lien Facilities Agreement (C) clause 4.16 (Hedge Counterparty Guarantee) and schedule 5 of the

Intercreditor Agreement and (D) guarantees or indemnities contained in any Secured Debt Document (as defined in the Intercreditor Agreement) and given in respect of the obligations of persons other than the relevant Obligor; and

(ii) "Guarantee Demand Date" means the first date upon which a Finance Party makes written demand upon the relevant Obligor to make payment in respect of its Guarantee Obligations.

(b) Financial Assistance Limitations

- (i) Without limiting any specific exemptions set out below:
 - (A) no Obligor's Guarantee Obligations will extend to include any obligation or liability; and
 - (B) no Security granted by an Obligor will secure any Guarantee Obligation,

if to do so would be unlawful financial assistance in respect of the acquisition of shares in itself or its Parent Undertaking under the laws of its jurisdiction of incorporation.

(c) Luxembourg Obligors

- (i) The maximum amount payable by a Luxembourg Obligor in respect of its Guarantee Obligations under this agreement and any other Secured Debt Documents (as defined in the Intercreditor Agreement) will be limited to an amount equal to 90 per cent of that Luxembourg Obligor's net worth ("capitaux propres") as referred to in Annex I to the Grand Ducal regulation of 18 December 2015 determining the form and content of the layouts of the balance sheet and of the profit and loss account and implementing articles 34, 35, 46 and 47 of the amended law of 19 December 2002 on the register of commerce and companies and the accounting and annual accounts of understandings as reflected in its most recent annual accounts approved at a general meeting of its shareholders and published before the Guarantee Demand date.
- (ii) The limit in paragraph (i) above will not apply to the extent that the Guarantee Obligations of a Luxembourg Obligor relate to the indebtedness of any Subsidiary of such Luxembourg Obligor.
- (iii) The Guarantee Obligations of a Luxembourg Obligor will not extend to include any obligations or liabilities if this would constitute a misuse of corporate assets as defined under article 1500-11 of the Luxembourg law dated 10 August 1915 on commercial companies, as amended from time to time.
- (iv) If the limit in paragraph (i) above means that the relevant Luxembourg Obligor cannot discharge the Guarantee Obligations in full, those Guarantee Obligations will be discharged in the order of priority specified in the Intercreditor Agreement including (without limitation) the provisions of clause 2.1 (Creditor Liabilities) and clause 15 (Application of Proceeds).

23.13 Swiss Obligors

Notwithstanding anything to the contrary elsewhere in this clause 23 (Guarantee and Indemnity), clause 4.16 (Hedge Counterparty Guarantee) and Schedule 5 of the Intercreditor Agreement or any Secured Debt Document (as defined in the Intercreditor

Agreement), the amounts which may, as the case may be, become payable by ASM in respect of the Guarantee Obligations shall (from time to time) be limited as follows:

- (a) if and to the extent a payment in fulfilling the Guarantee Obligations would, at the time payment is due, under Swiss corporate law and practice prohibiting capital repayments or restricting profit distributions not be permitted, then such obligations and payment amount shall be limited to the amount permitted to be paid, provided that such limited amount shall at no time be less than the unrestricted equity capital surplus (including the unrestricted portion of general (legal) reserves, restricted services which may be converted into free reserves, other free reserves, retained earnings and current net profits) available (as the case may be, after conversion/re-qualification) for distribution to the shareholder(s) of ASM under Swiss corporate law and practice at the time or times payment under or pursuant to the relevant guarantee or indemnity is requested from the Guarantor (from time to time each the "Minimum Guarantee Amount"), and further provided that such limitation (as may apply from time to time or not) shall not (generally or definitively) free ASM from payment obligations hereunder in excess thereof, but merely postpone the payment date therefore until such times as payment is again permitted notwithstanding such limitation.
- (b) if and to the extent this is (at the time a payment in fulfilling the Guarantee Obligations is due) required under Swiss corporate law (restricting profit distributions), in order to allow the Security Agent (and the Finance Parties) to obtain a maximum benefit under and in respect of the Guarantee Obligations, ASM and its Parent Undertaking undertake to promptly implement all such measures and/or to promptly procure the fulfilment of all prerequisites allowing ASM to promptly make the (requested) payment(s) hereunder from time to time, including the following:
 - (i) preparation of an up-to-date audited balance sheet of ASM;
 - (ii) confirmation of the auditors of ASM that the relevant Minimum Guarantee Amount represents (the maximum of) freely distributable profits;
 - (iii) approval by a shareholder(s)' meeting of ASM of the (resulting) profit distribution in the amount of the Minimum Guarantee Amount; and
 - (iv) all such other measures necessary or useful to allow ASM to make the payments agreed hereunder with a minimum of limitations, including the conversion of unnecessary restricted reserves into distributable reserves.
- (c) ASM's Guarantee Obligations shall be limited to a guarantee of liabilities of Alpha D2 and any Subsidiary of the Parent.
- (d) If the limit in paragraph (i) above means that ASM cannot discharge the Guarantee Obligations in full, those Guarantee Obligations will be discharged in the order of priority specified in the Intercreditor Agreement including (without limitation) the provisions of clause 2.1 (Creditor Liabilities) and clause 15 (Application of Proceeds).
- (e) If so required under applicable law (including tax treaties) at the time it is required to make a payment under this agreement or any other Finance Document, ASM:
 - (i) shall use its best efforts (provided that each Finance Party shall co-operate as may be reasonably requested) to ensure that such payments can be made without deduction of Swiss Withholding Tax, or with deduction of Swiss Withholding Tax at a reduced rate, by discharging the liability to such tax by notification pursuant to applicable law (including tax treaties) rather than payment of the tax;

- (ii) shall deduct the Swiss Withholding Tax at such rate (being 35% on the date hereof) as in force from time to time if the notification procedure pursuant to paragraph (i) above does not apply, or shall deduct the Swiss Withholding Tax at the reduced rate resulting after discharge of part of such tax by notification if the notification procedure pursuant to paragraph (i) applies for a part of the Swiss Withholding Tax only, and shall pay within the time allowed any such taxes deducted to the Swiss Federal Tax Administration; and
- (iii) shall promptly notify the Facility Agent that such notification or, as the case may be, deduction has been made, and provide the Agent with evidence that such a notification of the Swiss Federal Tax Administration has been made or, as the case may be, such taxes deducted have been paid to the Swiss Federal Tax Administration.
- (f) In the case of a deduction of Swiss Withholding Tax, ASM shall use its best efforts to ensure that any person that is entitled to a full or partial refund of the Swiss Withholding Tax deducted from such payment under this agreement or any other Finance Document, will, as soon as possible after such deduction:
 - (i) request a refund of the Swiss Withholding Tax under applicable law (including tax treaties);
 - (ii) pay to the Facility Agent upon receipt any amount so refunded.
- (g) To the extent ASM is required to deduct Swiss Withholding Tax pursuant to this agreement or any other Finance Document, the Facility Agent shall be entitled to further enforce the guarantee granted by ASM under this agreement (and ASM shall withhold Swiss Withholding Tax on such additional amount in accordance with paragraph (e) above) so that after making any required deduction of Swiss Withholding Tax the aggregate net amount paid to the Agent is equal to the amount which would have been paid if no deduction of Swiss Withholding Tax had been required, subject always to the limitations set out in paragraph (a) above.

24. **REPRESENTATIONS**

24.1 General

Each Obligor makes the representations and warranties set out in this clause 24 to each Finance Party.

24.2 Status

- (a) It and each of its Subsidiaries is a limited liability corporation, duly incorporated and validly existing under the law of its jurisdiction of incorporation.
- (b) It and each of its Subsidiaries has the power to own its assets and carry on its business as it is being conducted.
- (c) In relation to each Obligor incorporated in Jersey,:
 - (i) it is not a "financial services company" or a "utility company" (as respectively defined in the Income Tax (Jersey) Law 1961);
 - (ii) it is exempt from the duty to hold a business licence pursuant to the Control of Housing and Work (Exemptions) (Jersey) Order 2013;
 - (iii) it does not conduct any unauthorised "financial services business" (as defined in the Financial Services (Jersey) Law 1998);

- it is not in breach of any approvals, authorisations, consents, licences, permits or registrations issued to it by any regulatory or governmental authority in Jersey and will not be in breach of the same as a result of entering into any of the Finance Documents;
- is not in the trade of importing into Jersey or supplying in Jersey hydrocarbon oil (as defined in Article 123(AA)(3) of the Income Tax Jersey Law 1961);
- (vi) is and will remain an international services entity (as defined in the Goods and Services Tax (Jersey) Law 2007; and
- (vii) it has not owned and does not own land in Jersey.

24.3 **Binding Obligations**

Subject to the Legal Reservations:

- (a) the obligations expressed to be assumed by it in each Finance Document to which it is a party are legal, valid, binding and enforceable obligations; and
- (b) (without limiting the generality of paragraph (a) above), each Transaction Security Document to which it is a party creates the security interests which that Transaction Security Document purports to create and those security interests are valid and effective.

24.4 Non-Conflict with Other Obligations

The entry into and performance by it of, and the transactions contemplated by, the Finance Documents and the granting of the Transaction Security do not and will not conflict with:

- (a) any law or regulation applicable to it in any material respect;
- (b) the constitutional documents of any Group Company in any material respect; or
- (c) any agreement or instrument binding upon it or any Group Company or any of its or any Group Company's assets which is reasonably likely to have a Material Adverse Effect.

24.5 Power and Authority

- (a) It has the power to enter into, perform and deliver, and has taken all necessary action to authorise its entry into, performance and delivery of, the Finance Documents to which it is or will be a party and the transactions contemplated by those Finance Documents.
- (b) No limit on its powers will be exceeded as a result of the borrowing, grant of security or giving of guarantees or indemnities contemplated by the Finance Documents to which it is a party.

24.6 Validity and Admissibility in Evidence

- (a) All Authorisations required or desirable:
 - (i) to enable it lawfully to enter into, exercise its rights and comply with its obligations in the Finance Documents to which it is a party; and
 - (ii) subject to the Legal Reservations, to make the Finance Documents to which it is a party admissible in evidence in its Relevant Jurisdictions,

have been obtained or effected and are in full force and effect except for those necessary to satisfy the Perfection Requirements.

(b) All Authorisations necessary for the conduct of the business, trade and ordinary activities of Group Companies have been obtained or effected and are in full force and effect if failure to obtain or effect those Authorisations has or is reasonably likely to have a Material Adverse Effect.

24.7 **Governing Law and Enforcement**

Subject to the Legal Reservations:

- (a) the choice of English law as the governing law of the Finance Documents that are expressed to be governed by English law will be recognised and enforced in its Relevant Jurisdictions; and
- (b) any judgment obtained in England in relation to a Finance Document in the jurisdiction of the governing law of that Finance Document will be recognised and enforced in its Relevant Jurisdictions.

24.8 Insolvency

No:

- (a) corporate action, legal proceeding or other procedure or step described in clause 28.7(a) (Insolvency Proceedings); or
- (b) creditors' process described in clause 28.8 (Creditors' Process),

has been taken or, to the knowledge of Obligors' Agent, threatened in relation to an Obligor; and none of the circumstances described in clauses 28.6(a) and 28.6(c) (Insolvency) applies to an Obligor.

24.9 **No Filing or Stamp Taxes**

Under the law of its jurisdiction of incorporation it is not necessary that the Finance Documents be filed, recorded or enrolled with any court or other authority in that jurisdiction or that any stamp, registration or similar Tax be paid on or in relation to the Finance Documents or the transactions contemplated by the Finance Documents except:

- (a) in respect of any assignment or transfer or sub-participation or sub-contract by a Lender of any of its rights or obligations under a Finance Document;
- (b) in the case of Luxembourg, if such Finance Document is either (i) attached as an annex to an act (annexés à un acte) that itself is subject to mandatory registration, or (ii) deposited in the minutes of a notary (déposés au rang des minutes d'un notaire) or (iii) voluntary registrered, upon which a fixed registration duty of EUR 12 (twelve euros) or an ad valorem duty (for instance, 0.24 (zero point twenty four) per cent of the amount of the payment obligation mentioned in the document so registered)) will be payable depending on the nature of the document so registered;
- (c) in respect of any Transaction Security created or purported to be created pursuant to the Security Interests (Jersey) Law 2012 (the "2012 Jersey Security Law"), the registration in the security interests register maintained by the Jersey registrar of companies in accordance with the 2012 Jersey Security Law and payment of fees in respect of such registration; and
- (d) any filing, recording or enrolling or the payment of any tax or fee payable which is referred to in any legal opinions delivered to the Finance Parties in connection with

the Finance Documents (or required in connection with any Perfection Requirements) with any court or other authority in the relevant jurisdictions.

24.10 **Deduction of Tax**

- (a) In relation to an Obligor incorporated or tax resident in the United Kingdom, it is not required to make any deduction for or on account of Tax (other than an Excluded U.S. Tax Deduction, if applicable) from any payment it may make under any Finance Document to a Lender which is a Qualifying Lender;
- (b) In relation to an Obligor incorporated in Luxembourg, subject to the condition that no Interest Payment is made to an individual resident in Luxembourg, it is not required to make any deduction for or on account of Tax (other than an Excluded U.S. Tax Deduction, if applicable) from any payment it may make under any Finance Document to a Lender;
- (c) In relation to an Obligor incorporated in a jurisdiction other than Luxembourg or the United Kingdom, it is not required to make any deduction for or on account of Tax from any payment it may make under any Finance Document to a Lender (other than an Excluded U.S. Tax Deduction, if applicable).

24.11 No Default

- (a) No Event of Default and, on the date of this agreement and the 2022 Effective Date, no Default is continuing or is reasonably likely to result from the making of any Utilisation or the entry into, the performance of, or any transaction contemplated by, any Finance Document.
- (b) No other event or circumstance is outstanding which constitutes (or, with the expiry of a grace period, the giving of notice, the making of any determination or any combination of any of the foregoing, would constitute) a default or termination event (however described) under any other agreement or instrument which is binding on it or any of its Subsidiaries or to which its (or any of its Subsidiaries') assets are subject which might reasonably be expected to have a Material Adverse Effect.

24.12 Tax Residence

It is resident for Tax purposes in its jurisdiction of incorporation, or, in the case of a Group Company which is not incorporated in the United Kingdom, the United Kingdom.

24.13 Financial Statements

The most recent financial statements delivered pursuant to clause 25.1 (Financial Statements) give in all material respects a true and fair view of (if audited) or fairly present in all material respects (if unaudited) the consolidated financial condition of the Group as at the end of, and the consolidated results of operations of the Group for, the period to which they relate and were prepared with the Accounting Principles consistently applied (to the extent appropriate in respect of the unaudited financial statements).

24.14 No Proceedings Pending or Threatened

No litigation, arbitration or administrative proceedings or investigations of, or before, any court, arbitral body or agency which are reasonably likely to be adversely determined and if adversely determined would be reasonably likely to have a Material Adverse Effect have been started or (to the best of its knowledge and belief) threatened against it or any of its Subsidiaries.

24.15 No Breach of Laws

- (a) It has not (and none of its Subsidiaries has) breached any law or regulation which breach has or is reasonably likely to have a Material Adverse Effect.
- (b) No labour disputes are current or, to the best of its knowledge and belief (having made due and careful enquiry), threatened against any Group Company which have or are reasonably likely to have a Material Adverse Effect.

24.16 Environmental Laws

- (a) To the best of its knowledge (having made due and careful enquiry) no circumstances have occurred which would prevent such compliance in a manner or to an extent which has or is reasonably likely to have a Material Adverse Effect.
- (b) No Environmental Claim has been commenced or (to the best of its knowledge (having made due and careful enquiry)) is threatened against any Group Company where that claim has or is reasonably likely, if determined against that Group Company, to have a Material Adverse Effect.

24.17 Taxation

- (a) It is not (and none of its Subsidiaries is) materially overdue in the filing of any Tax returns, and it is not (and none of its Subsidiaries is) overdue in the payment of any amount in respect of Tax which being overdue has had or could reasonably be expected to have a Material Adverse Effect.
- (b) No claims or investigations are being, or are reasonably likely to be, made or conducted against it (or any of its Subsidiaries) with respect to Taxes where any such claims or investigations have had or could reasonably be expected to have a Material Adverse Effect.

24.18 Security and Financial Indebtedness

- (a) No Security or Quasi-Security exists over all or any of the present or future assets of any Group Company other than as permitted by this agreement.
- (b) No Group Company has any Financial Indebtedness outstanding other than as permitted by this agreement.

24.19 Ranking

Subject to the Legal Reservations and the Perfection Requirements, the Transaction Security has or will have the ranking in priority which it is expressed to have in the Transaction Security Documents and it is not subject to any prior ranking or pari passu ranking Security other than Security created or evidenced by any Transaction Security Document.

24.20 Good Title to Assets

It and each of its Subsidiaries has a good, valid and marketable title to, or valid leases or licences of, and all appropriate Authorisations to use, all material assets necessary to carry on its business as presently conducted.

24.21 Legal and Beneficial Ownership

It and each of its Subsidiaries is the sole legal and beneficial owner of the respective assets over which it purports to grant Security.

24.22 **Shares**

The shares of any Group Company which are subject to the Transaction Security are fully paid and not subject to any option to purchase or similar rights. The constitutional documents of companies whose shares are subject to the Transaction Security do not and could not restrict or inhibit any transfer of those shares on creation or enforcement of the Transaction Security.

24.23 Intellectual Property

- (a) The Intellectual Property required for each Group Company to conduct its business as presently conducted:
 - (i) is legally and beneficially owned by or licensed to a Group Company (and where owned and registered by a Group Company and registered or the subject of an application for registration by a Group Company, a Group Company is the registered proprietor) free from any:
 - (A) licences or obligation to assign to third parties; and
 - (B) Security,

which in each case are (separately and collectively) materially prejudicial to the use of that Intellectual Property and will not be adversely affected by the transactions contemplated by the Finance Documents; and

- (ii) has not lapsed or been cancelled and all steps have been taken to protect and maintain that Intellectual Property, including paying renewal fees, where any such lapse, cancellation or failure to take steps (including paying renewal fees) could reasonably be expected to have a Material Adverse Effect.
- (b) Where the Intellectual Property required for each Group Company to conduct its business as presently conducted is subject to any right, permission to use or licence granted to or by any Group Company, such right, permission or licence has not been breached in any respect or terminated by any party, in each case to an extent that could reasonably be expected to have a Material Adverse Effect.
- (c) So far as it is aware no Group Company is infringing any intellectual property rights of any third party and the Intellectual Property is not being infringed by any third party, in each case to an extent that could reasonably be expected to have a Material Adverse Effect.

24.24 Anti-Terrorism

- (a) No Group Company nor, to the best of the Obligors' knowledge, any Affiliate of an Obligor:
 - (i) is, or is controlled by, a Restricted Party;
 - (ii) has received funds from, or made any payment under or in connection with the Finance Documents from funds or other property received from, a person who at the time of such receipt, to its knowledge, was a Restricted Party; or
 - (iii) is in breach of or is the subject of any action or investigation under any Anti-Terrorism Law.
- (b) Each Group Company and, to the best of the Obligors' knowledge, each Affiliate thereof has taken reasonable measures to ensure compliance with the Anti-Terrorism Laws.

(c) None of the Borrowers or any Subsidiary or, to the knowledge of the Borrowers, any directors, officers, employees, and agents of the Borrowers or any Subsidiary is a Sanctioned Person.

24.25 Times when Representations Made

- (a) All the representations and warranties in this clause 24 (other than clause 24.9) are made on the 2022 Effective Date by each Original Obligor party to this agreement on the 2022 Effective Date.
- (b) The Repeating Representations are deemed to be made by each Obligor on the date of each Utilisation Request, on each Utilisation Date and on the first day of each Interest Period (except that those contained in clause 24.13 (Financial Statements) will cease to be so made in respect of a set of financial statements once subsequent financial statements have been delivered under this agreement).
- (c) All representations and warranties in this clause 24 except clause 24.9 (No Filing or Stamp Taxes), clause 24.10 (Deduction of Tax) and clause 24.13 (Financial Statements) are deemed to be made by each Additional Obligor on the day on which it becomes (or it is proposed that it becomes) an Additional Obligor.
- (d) Each representation or warranty deemed to be made after the date of this agreement shall (except where the contrary is indicated) be deemed to be made by reference to the facts and circumstances existing at the date the representation or warranty is deemed to be made.

25. **INFORMATION UNDERTAKINGS**

The undertakings in this clause 25 remain in force from the date of this agreement for so long as any amount is outstanding under the Finance Documents or any Commitment is in force.

25.1 Financial Statements

The Parent (or Obligors' Agent) shall supply to the Facility Agent:

- (a) as soon as they are available, but in any event within 120 days after the end of each of its Financial Years, the Group's audited consolidated financial statements for that Financial Year (which may be satisfied with the Parent's audited consolidated financial statements with a customary schedule eliminating Unrestricted Subsidiaries, if any, and reconciling to such audited consolidated financial statements); and
- (b) as soon as they are available, but in any event within 60 days after the end of each Financial Quarter the management information pack for the Group (or the Parent with a customary schedule eliminating Unrestricted Subsidiaries, if any, and reconciling to such management information pack) for that Quarter (to include cumulative management accounts for the Financial Year to that Financial Quarter) (the "Quarterly Information Pack").

Notwithstanding the foregoing, the obligations referred to in this Clause 25.1 may be satisfied with respect to financial information of the Group by furnishing the financial statements of the Parent Reporter (with a customary schedule eliminating Unrestricted Subsidiaries, if any, and reconciling to such financial information) (such option, the "Parent Reporter Option"); provided that if and so long as the Parent Reporter has Independent Assets or Operations, such information is accompanied by consolidating information that reasonably shows the material differences between the information relating to the Parent Reporter and its Independent Assets or Operations, on the one hand, and the information relating to the Group on a standalone basis, on the other hand. For the avoidance of doubt, all financial ratio or tests hereunder shall be calculated with respect to the Group.

25.2 Provision and Contents of Compliance Certificate

- (a) The Parent (or Obligors' Agent) shall supply a Compliance Certificate to the Facility Agent with each set of the Group's audited consolidated Annual Financial Statements and with the Group's Quarterly Information Pack.
- (b) The Compliance Certificate shall, amongst other things set out (in reasonable detail) computations as to compliance with clause 26 (Financial Definitions and Financial Covenant) or confirmation that the financial covenant specified therein is not required to be tested in respect of that Testing Period pursuant to clause 26.2 (Financial Condition).
- (c) Each Compliance Certificate shall be signed by an authorised signatory of the Parent (or Obligors' Agent).

25.3 Requirements as to Financial Statements

- (a) The Parent (or Obligors' Agent) shall procure that each set of Annual Financial Statements and each management information pack delivered pursuant to clause 25.1(b) (Financial Statements) includes a balance sheet, profit and loss account (or equivalent income statement) and cashflow statement. In addition the Parent (or Obligors' Agent) shall procure that: (i) each set of Annual Financial Statements shall be audited by the Auditors and (ii) each management information pack delivered pursuant to clause 25.1(b) (Financial Statements) is accompanied by a statement by the directors of the Parent commenting on the performance of the Group for the quarter to which the management information pack relates (which requirement in this clause (ii) shall be deemed satisfied so long as such management information pack is generally consistent with the management information packs delivered to the Facility Agent (or its predecessors) prior to the 2022 Effective Date).
- (b) Each set of financial statements delivered pursuant to clause 25.1 (Financial Statements) (including those contained in each management information pack) shall be prepared in accordance with the Accounting Principles. If there is a change in the Accounting Principles after the date of this agreement (including, for the avoidance of doubt, any change between the application of UK generally accepted accounting principles, US generally accepted accounting principles or International Financial Reporting Standards):
 - (i) the Parent (or Obligors' Agent) shall promptly advise the Facility Agent;
 - (ii) following request by the Facility Agent, Obligors' Agent and the Facility Agent shall negotiate in good faith with a view to agreeing any amendments to this agreement which may be necessary to ensure that the change does not result in any material alteration in the commercial effect of those terms;
 - (iii) if amendments are agreed by Obligors' Agent and the Facility Agent within 30 days of the notification of such proposed amendments to the Lenders by the Facility Agent, those amendments shall take effect in accordance with the terms of the agreement so long as Majority Lender do not object to such amendments within such 30 day period; and
 - (iv) if amendments are objected to by the Majority Lenders within the 30 day period referenced in clause (iii) above then, within 15 days after the end of that 30 day period, Obligors' Agent shall either:
 - (A) deliver to the Facility Agent, in reasonable detail and in a form satisfactory to the Facility Agent (acting reasonably), details of any adjustments which need to be made to the relevant accounts in order

to bring them into line with the Accounting Principles as at the date of this agreement; or

(B) ensure that the relevant accounts are prepared in accordance with the Accounting Principles as at the date of this agreement.

25.4 Presentations

- (a) Only in the event that the Parent or any parent entity of the Parent is no longer publicly listed on any regulated stock exchange and the Facility Agent so requests, once in every Financial Year the Parent must give a presentation to the Finance Parties about the on-going business and financial performance of the Group on reasonable prior notice and at times reasonably convenient to such persons.
- (b) Nothing in paragraph (a) above shall require Parent to disclose or discuss any document, information or other matter (i) of a price sensitive nature that has not yet been publicly announced pursuant to the listing rules of a regulated stock exchange applicable to any direct or indirect shareholder in the Parent, (ii) that constitutes non-financial trade secrets or non-financial proprietary information of the Group Company and/or any of their respective customers and/or suppliers, (iii) in respect of which disclosure to any Finance Party (or any of their respective representatives or contractors) is prohibited by applicable law, (iv) that is subject to attorney-client or similar privilege or constitutes attorney work product or (v) in respect of which any Group Company owes confidentiality obligations to any third party (provided that such confidentiality obligations were not entered into solely in contemplation of this clause (v)).

25.5 Year-end

- (a) The Parent shall procure that each Financial Year-end of each Group Company falls on 31 December (or the equivalent year end accounting period).
- (b) The Parent shall procure that each quarterly accounting period and each Financial Quarter of each Group Company ends on an accounting date.

25.6 Information: Miscellaneous

- (a) The Parent (or the Obligors' Agent) shall supply to the Facility Agent (in sufficient copies for all the Lenders, if the Facility Agent so requests):
 - at the same time as they are dispatched, copies of all documents dispatched by the Parent to its creditors generally (or any class of them) (excluding for this purpose creditors which are Group Companies);
 - (ii) promptly upon becoming aware of them, the details of any litigation, arbitration or administrative proceedings, environmental claim or action or labour dispute relating to it or any of its subsidiaries which would result in a Material Adverse Effect;
 - (iii) promptly, such information as the Security Agent may reasonably require about the Charged Property and compliance of the Obligors with the terms of any Transaction Security Documents; and
 - (iv) promptly if the Facility Agent (acting reasonably) so requests, such further information regarding the financial condition and operations of the Group and/or any Group Company; and

- (v) promptly, the details of any material changes in the structure of the Group or in the identity of the Material Companies.
- (b) Nothing in paragraphs (i) to (v) above or clause 25.3 (Requirements as to Financial Statements) shall require any Group Member to disclose, permit the inspection, examination or making of copies of or taking abstracts from, or discuss any document, information or other matter (i) of a price sensitive nature that has not yet been publicly announced pursuant to the listing rules of a regulated stock exchange applicable to any direct or indirect shareholder in the Parent, (ii) that constitutes non-financial trade secrets or non-financial proprietary information of the Group Company and/or any of their respective customers and/or suppliers, (iii) in respect of which disclosure to any Finance Party (or any of their respective representatives or contractors) is prohibited by applicable law, (iv) that is subject to attorney-client or similar privilege or constitutes attorney work product or (v) in respect of which any Group Company owes confidentiality obligations to any third party (provided that such confidentiality obligations were not entered into solely in contemplation of this clause (v)).

25.7 Notification of Default

- (a) Each Obligor shall notify the Facility Agent of any Default (and the steps, if any, being taken to remedy it) promptly upon becoming aware of its occurrence (unless that Obligor is aware that a notification has already been provided by another Obligor).
- (b) Promptly upon a request by the Facility Agent made if the Facility Agent considers in good faith there may be a Default, Obligors' Agent shall supply to the Facility Agent a certificate signed by two of its directors or senior officers on its behalf certifying that no Default is continuing, or if a Default is continuing, specifying the Default and the steps, if any, being taken to remedy it).

25.8 "Know Your Customer" Checks

- (a) If:
 - the introduction of or any change in (or in the interpretation, administration or application of) any law or regulation made after the date of this agreement;
 - (ii) any change in the status of an Obligor or the composition of the shareholders of an Obligor after the date of this agreement; or
 - (iii) a proposed assignment or transfer by a Lender of any of its rights and/or obligations under this agreement to a party that is not a Lender prior to such assignment or transfer,

obliges the Facility Agent or any Lender (or, in the case of paragraph (a)(iii) above, any prospective new Lender) to comply with "know your customer" or similar identification procedures in circumstances where the necessary information is not already available to it, each Obligor shall promptly upon the request of the Facility Agent or any Lender supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Facility Agent (for itself or on behalf of any Lender) or any Lender (for itself or, in the case of the event described in paragraph (a) (iii) above, on behalf of any prospective new Lender) in order for the Facility Agent, such Lender or, in the case of the event described in paragraph (a)(iii) above, any prospective new Lender to carry out and be satisfied with the results of all necessary "know your customer" or other checks in relation to any relevant person pursuant to the transactions contemplated in the Finance Documents.

- (b) Each Lender shall promptly upon the request of the Facility Agent supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Facility Agent (for itself) in order for the Facility Agent to carry out and be satisfied with the results of all necessary "know your customer" or other checks on Lenders or prospective new Lenders pursuant to the transactions contemplated in the Finance Documents.
- (c) Obligors' Agent shall, by not less than ten Business Days' prior written notice to the Facility Agent, notify the Facility Agent (which shall promptly notify the Lenders) of its intention to request that one of its Subsidiaries becomes an Additional Obligor pursuant to clause 30 (Changes to the Obligors).
- (d) Following the giving of any notice pursuant to paragraph (c) above, if the accession of such Additional Obligor obliges the Facility Agent or any Lender to comply with "know your customer" or similar identification procedures in circumstances where the necessary information is not already available to it, the Parent (or Obligors' Agent) shall promptly upon the request of the Facility Agent or any Lender supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Facility Agent (for itself or on behalf of any Lender) or any Lender (for itself or on behalf of any prospective new Lender) in order for the Facility Agent or such Lender or any prospective new Lender to carry out and be satisfied it has complied with all necessary "know your customer" or other checks in relation to any relevant person pursuant to the accession of such Subsidiary to this agreement as an Additional Obligor.
- (e) Promptly following any request therefor, the Obligors shall provide information and documentation reasonably requested by the Facility Agent or any Lender for purposes of compliance with applicable "know your customer" requirements under the Patriot Act, 1 C.F.R. §1010.230. (the "Beneficial Ownership Regulation") or other applicable anti-money laundering laws.

25.9 Investigations

In the event that an Event of Default shall have occurred and be continuing under clause 28.1 (Non-Payment), clause 28.2 (Financial Covenant) or 28.6 (Insolvency) to 28.9 (Similar Events Elsewhere), the Facility Agent may (after consultation with Obligors' Agent as to the scope of the investigation) require that each Obligor shall (and Obligors' Agent shall ensure that each Obligor shall) permit the Facility Agent and/or accountants or other financial advisers appointed by the Facility Agent free access (in the presence of the Parent) at all reasonable times during normal business hours and on reasonable notice, at the expense of the Obligors, to:

- (a) inspect and take copies and extracts from the books, accounts and records of each Group Company (to the extent the Facility Agent (acting reasonably) considers such books, accounts or records to be relevant to the Event of Default which has occurred and is continuing); and
- (b) meet and discuss matters with the management of the Parent,

provided that all information obtained as a result of such access shall be subject to the confidentiality restrictions set out in this agreement.

Nothing in clause 25.9 shall require any Group Member to disclose, permit the inspection, examination or making of copies of or taking abstracts from, or discuss any document, information or other matter (i) of a price sensitive nature that has not yet been publicly announced pursuant to the listing rules of a regulated stock exchange applicable to any direct or indirect shareholder in the Parent, (ii) that constitutes non-financial trade secrets or non-financial proprietary information of the Group Company and/or any of their

respective customers and/or suppliers, (iii) in respect of which disclosure to any Finance Party (or any of their respective representatives or contractors) is prohibited by applicable law, (iv) that is subject to attorney-client or similar privilege or constitutes attorney work product or (v) in respect of which any Group Company owes confidentiality obligations to any third party (provided that such confidentiality obligations were not entered into solely in contemplation of this clause (v)).

26. FINANCIAL DEFINITIONS AND FINANCIAL COVENANT

26.1 Financial Definitions

In this clause 26 and where such terms are otherwise used in this agreement:

"Capital Expenditure" means, in respect of the Group and in respect of any period, any expenditure which should be treated as capital expenditure in the financial statements of the person incurring such expenditure in accordance with applicable Accounting Principles. For the avoidance of doubt, expenditure for acquisitions of businesses or expenditure arising from operating leases shall not constitute Capital Expenditure for the purposes of this definition;

"Cash" means cash credited to an account in the name of a Group Company with an Acceptable Bank;

"Consolidated EBIT" for any period, means the consolidated profits of the Group from ordinary activities for the relevant Testing Period before taxation (including before any distributions (including any anticipated or reserved tax distribution) described in clause (g) or clause (o) of the definition of Permitted Distributions):

- (a) before taking into account any interest (and amounts in the nature of interest including amounts in respect of any related hedging), commission, fees (including any amortized fees), discounts and other finance payments incurred or payable or owed by any member of the Group in respect of Financial Indebtedness;
- (b) after deducting the amount of any profit of any Group Company which is attributable to minority interests or the interests of any shareholder of or, as the case may be, partner in such member of the Group who is not a Group Company;
- (c) after taking into account any exchange gains and losses on operational trading and hedges relating to these items but not taking into account any other realised and unrealised exchange gains and losses (including those arising on translation of currency debt);
- (d) without double counting, before taking into account any gain or loss arising from an upward or downward revaluation of any asset or on the disposal of an asset;
- (e) plus any loss on any one-off or non-recurring items;
- (f) less any gain on any one-off or non-recurring items;
- (g) plus the proceeds from any business interruption insurance or similar insurance proceeds actually received by any Group Company during the relevant period or reasonably expected to be received in a subsequent period and within one year of the underlying loss; provided, that if not so received within such one year period, such amount shall be subtracted in the subsequent calculation period;
- (h) before deducting any one-off or non-recurring restructuring costs that are incurred;
- (i) before deducting any charge, loss, lost profit, expense or write-off that is actually reimbursed or reimbursable (and reasonably expected to be reimbursed within the

- next four Financial Quarters) by one or more third parties pursuant to indemnification or reimbursement provisions or similar agreements or insurance;
- (j) before deducting any Transaction Costs;
- (k) before deducting expenses relating to pensions including service costs and pension interest costs but after deducting cash pension costs;
- (I) plus any fees, costs, expenses or charges of a non-recurring nature related to any equity offering, investments, acquisitions or Financial Indebtedness (or any related hedging transactions) permitted under the Finance Documents (whether or not successful), including any amendment or other modification of this agreement and any other credit facilities or the offering of debt securities;
- (m) plus any management fees paid to the Investors and Parent's non-executive directors and Parent Undertaking costs where permitted to be paid under the Finance Documents;
- (n) before taking into account any charges for non-cash amortisation of one-off payments to any F1
 Teams which for accounting purposes are being spread over a multi-year term and for the avoidance
 of doubt any one-off payments to any F1 Team which are charged to the income statement in one
 single period (including any milestone or fixed payments) shall be taken account of if applicable to the
 relevant period);
- excluding fair value movements in derivative financial instruments but taking into account any actual close-out amounts to the extent included in Consolidated EBIT in respect of such derivative financial instruments;
- excluding any profit or loss recorded in the income statement in relation to payments or awards as part of the management long term incentive plan in force on the 2022 Effective Date or any similar arrangement supplementing or replacing such plan;
- (q) adding back amounts taken into account as deductions in calculating consolidated profits that are attributable to transactions with affiliates of the Group which are not paid in cash during the relevant Testing Period;
- (r) plus any non-cash charges, write-downs, expenses or losses (including any non-cash losses relating to mark-to-market or fair value accounting);
- (s) excluding any non-cash gains (including any non-cash gains relating to mark-to-market or fair value accounting);
- (t) before deducting any fee (including third party consultant fees and other similar fees), loss, charge, expense, cost, accrual or reserve of any kind relating to the closure or consolidation of any facility or location and/or discontinued operations (including but not limited to severance, rent termination costs, moving costs and legal costs);
- (u) after making such adjustments as any financial officer of the Group determines in good faith to be reasonable to reflect material timing differences in terms of recognition of race-related profits resulting from any change in race dates; and
- excluding project startup costs and other costs and expenses related to new office openings costs and the consolidation of offices and facilities (including the termination or discontinuance of activities constituting a business) (and proposals in connection therewith, whether or not successful);

- (w) plus the operating income of any non-wholly owned person up to the amount of cash or Cash Equivalent Investments (x) actually distributed by such person to any Group Company or (y) that could have been distributed as a dividend or other distribution or return on investment by such person to a Group Company;
- (x) after taking into account any Certified Synergies without double-counting for any synergies actually achieved or taken into account pursuant to clause 26.4 (Calculation Adjustments); provided that the amounts under this clause (x) for the relevant period shall not exceed 25% of Consolidated EBITDA (after any adjustments are made in respect of this paragraph (x)).

in each case, to the extent added, deducted or taken into account, as the case may be, for the purposes of determining profits of the Group from ordinary activities before taxation;

For the avoidance of doubt, it is understood and agreed that that the Group may elect not to apply any addback(s) or deduct(s) set forth in any clause or subclause set forth in the definition of Consolidated EBIT; provided that if any addback or deduct is not applied then the counter deduct or addback (if applicable) shall also not be applied.

"Consolidated EBITDA" for any period and without double counting means the Consolidated EBIT of the Group, plus the consolidated depreciation and amortisation and any impairment costs of the Group (each as defined by reference to the Annual Financial Statements of the Group) for the relevant Testing Period, for the avoidance of doubt, one-off payments to the F1 Teams in relation to the Concorde Agreement will not be deducted or will be excluded (as the case may be) from Consolidated EBITDA;

"Consolidated Interest Expense" means, without duplication, for the Group on a consolidated basis for any period, the total cash interest expense (including, without limitation, the interest component of capital lease obligations) on Financial Indebtedness minus total interest income of the Group during such period determined on a consolidated basis in accordance with Accounting Principles, plus all cash dividends paid or other recurring cash distributions paid (excluding items eliminated in consolidation) on any series of Disqualified Equity during such period. For purposes of this definition, interest on a capital lease obligation shall be deemed to accrue at an interest rate reasonably determined by the Parent (or Obligors' Agent) to be the rate of interest implicit in such capital lease obligation in accordance with Accounting Principles. For the avoidance of doubt, Consolidated Interest Expense shall not include (a) non-cash interest expense attributable to the movement of the mark-to-market valuation or fair value of obligations under hedging agreements or other derivative instruments, (b) any one-time cash costs associated with breakage for interest rates (including in respect of hedging agreements), (c) all non-recurring interest expense or "additional interest" for failure to timely comply with registration rights obligations, (d) any interest expense attributable to the exercise of appraisal rights and the settlement of any claims or actions (whether actual, contingent or potential) with respect thereto and with respect to any investment or acquisition, all as calculated on a consolidated basis in accordance with Accounting Principles, (e) any payments with respect to make-whole premiums or other breakage costs of any indebtedness, (f) penalties and interest relating to Taxes, (g) accretion or accrual of discounted liabilities not constituting Financial Indebtedness, (h) any interest expense attributable to a parent company resulting from push down accounting, (i) any expense resulting from the discounting of Financial Indebtedness in connection with the application of recapitalization or purchase accounting and (j) expensing of bridge, arrangement, structuring, commitment or other financing fees.

"Consolidated Net First Lien Secured Debt" means the principal amount of all Financial Indebtedness of the Group which constitutes First Lien Liabilities, less the aggregate amount at that time of Cash and Cash Equivalent Investments held by Group Companies, less the amount of any outstandings under any Ancillary Facilities (other than any overdraft or working capital facility or any letter of credit or other guarantee facility covering monies

borrowed by way any overdraft, working capital or term debt facility or other monies borrowed (in each case to the extent of such borrowings)) and after making such adjustments as any financial officer of the Group determines in good faith to be reasonable to reflect material timing differences in receipts of payments resulting from any change in race dates;

"Consolidated Total Net Secured Debt" means the principal amount of all Financial Indebtedness of the Group which constitutes First Lien Liabilities and Second Lien Liabilities less the aggregate amount at that time of Cash and Cash Equivalent Investments held by Group Companies, less the amount of any outstandings under any Ancillary Facilities (other than any overdraft or working capital facility or any letter of credit or other guarantee facility covering monies borrowed by way of term debt, any overdraft, working capital facility or other monies borrowed (in each case to the extent of such borrowings)) and after making such adjustments as any financial officer of the Group determines in good faith to be reasonable to reflect material timing differences in receipts of payments resulting from any change in race dates;

"Excess Cashflow" means, on the last day of each Financial Year, the amount by which the aggregate of the Group's Cash and Cash Equivalent Investments on such date (except to the extent to be applied for any scheduled or voluntary repayment of principal and scheduled or voluntary payment of interest or commitment fee to be made in respect of Financial Indebtedness) exceeds \$15,000,000; provided that for these purposes the following amounts shall be excluded from the Group's Cash and Cash Equivalent Investments (but without duplication):

- (i) any amounts which are required to ensure that the Group will be able to fund all of its liabilities to Taxes (including distributions described in clause (g) or (o) of the definition of Permitted Distributions), its ordinary course operating and financing costs (including expected Capital Expenditures and Restricted Payments), its liabilities to make payments to the F1 Teams, its liabilities to make payments to the FIA, and acquisitions and investments, in each case, committed or anticipated to be completed during the succeeding Financial Year, provided that if such amounts are not made in such succeeding Financial Year of the Group, they shall be added back as Cash and Cash Equivalents in the next calculation of Excess Cash Flow; and
- (ii) an amount equal to the Cash and Cash Equivalent Investments on the balance sheet of the Group on December 31, 2021;

"Financial Quarter" means the period of approximately 13 weeks ending on or about the last day of March, June, September and December in any Financial Year;

"Financial Year" means the period of 12 months ending on 31 December in each year;

"Net First Lien Secured Leverage" means, the ratio of Consolidated Net First Lien Secured Debt as at the relevant Testing Date to Consolidated EBITDA for the most recently ended Testing Period on or prior to such Testing Date;

"Net Total Secured Leverage" means the ratio of Consolidated Total Net Secured Debt as at the relevant Testing Date to Consolidated EBITDA for the most recently ended Testing Period on or prior to such Testing Date;

"New Shareholder Injections" means the subscription for shares in the Parent (or capital contributions made in Parent) or subscription for debt instruments in the Parent subordinated at least at the same extent as the Investor Liabilities (as defined in the Intercreditor Agreement);

"Testing Date" means, unless otherwise provided, the day which a particular financial ratio under this agreement is being tested or determined.

"Testing Period" means, subject to clause 26.4 (Calculation Adjustments), each period which corresponds to the annual accounting reference period of the Parent or four consecutive Financial Quarters, in each case, in respect of which financial statements have been delivered pursuant to the terms of this agreement; and

"Transaction Costs" means all fees, costs and expenses and Taxes incurred by Parent or any other member of the Group in connection with the negotiation, preparation, execution, notarisation and registration of the Transaction Documents or any investment and shareholders' agreements with the direct or indirect shareholders of any Group Company (or any supplement, amendment or restatement thereto) and otherwise in connection therewith including interest hedging costs and hedging costs incurred by way of one-off payments incurred in implementing the agreed hedging strategy.

26.2 Financial Condition

- (a) The Parent shall ensure that as of the last day of any Financial Quarter ending after the 2022 Effective Date (commencing with the first full Financial Quarter ending after the 2022 Effective Date), Net First Lien Secured Leverage shall not be more than 6.50:1.
- (b) Notwithstanding the foregoing, clause 26.2(a) shall be in effect (and shall only be in effect) as of the last day of any Financial Quarter when either (A) the Term Facility A Loans are outstanding as of such day or (B) the aggregate amount of Revolving Facility Utilisation as of such day (with respect to Letters of Credit, excluding (x) undrawn Letters of Credit of up to \$25,000,000 and (y) Letters of Credit then outstanding that have been cash collateralized pursuant to arrangements reasonably satisfactory to the Facility Agent) exceeds 35% of the aggregate amount of all Revolving Facility Commitments as of such day (it being understood that in all cases calculation of compliance with this clause 26.2 shall be determined as of the last day of such Financial Quarter).

26.3 Financial Testing

- (a) Subject to clause 26.2(b) (Financial Condition), the financial covenant set out in clause 26.2 (Financial Condition) shall be calculated in accordance with the Accounting Principles and tested by reference to each of the applicable financial statements and/or each Compliance Certificate delivered pursuant to clause 25.2 (Provision and Contents of Compliance Certificate).
- (b) If the Annual Financial Statements are not available when the covenant referred to in clause 26.2 (Financial Condition) is tested, but when those Annual Financial Statements become available, they show that the figures in any relevant Quarterly Financial Statements utilised for any such calculation cannot have been substantially accurate, the Facility Agent may require such adjustments to the calculations which it, in its discretion (acting reasonably), considers appropriate to rectify that inaccuracy and compliance with the covenant in clause 26.2 (Financial Condition) will be determined by reference to those adjusted figures provided always that any adjustment which indicates compliance with the covenant which had been breached when tested by reference to the relevant inaccurate figures will not have the effect of nullifying, waiving or otherwise curing that breach.
- (c) If:
 - the test of Net First Lien Secured Leverage at clause 26.2 (Financial Condition) would be breached; or

(ii) any Net First Lien Secured Leverage test or Net Total Secured Leverage test or Net Total Leverage test performed for the purpose of any provision of this agreement would exceed the relevant maximum ratio

if Consolidated Net First Lien Secured Debt or Consolidated Total Net Secured Debt or Consolidated Total Net Debt (as applicable) were to be calculated on the basis of the closing exchange rates as at the relevant Testing Date, but would not be breached or exceeded if Consolidated Net First Lien Secured Debt or Consolidated Total Net Secured Debt or Consolidated Total Net Debt (as applicable) were instead to be calculated on the basis of the average exchange rates over the relevant Testing Period, then Net First Lien Secured Leverage or Net Total Secured Leverage or Net Total Leverage (as applicable) for that Testing Date shall be calculated on the basis of those average exchange rates.

26.4 Calculation Adjustments

For the purposes of calculating Net Total Secured Leverage, Net First Lien Secured Leverage and Net Total Leverage if a Group Company acquires any company (having obtained any necessary consent under this agreement to do so), until the last day of the first Financial Quarter that ends more than 18 months after that company became a Subsidiary of the Parent, the results of that company will be deemed included with those of the rest of the Group for the full duration of the relevant Testing Period as if that company had become a Group Company at the commencement of the Testing Period. Any necessary aggregation of results will include any Certified Synergies in respect of such acquisition.

27. **GENERAL UNDERTAKINGS**

The undertakings in this clause 27 remain in force from the date of this agreement for so long as any amount is outstanding (other than contingent indemnification and expense reimbursement claims not then due) under the Finance Documents or any Commitment is in force.

27.1 Authorisations

Each Obligor shall promptly:

- (a) obtain, comply with and do all that is necessary to maintain in full force and effect; and
- (b) supply certified copies to the Facility Agent, on written request, of,

any Authorisation required under any law or regulation to:

- (i) enable it to perform its obligations under the Finance Documents;
- (ii) ensure, subject to the Legal Reservations, the legality, validity, enforceability or admissibility in evidence in its jurisdiction of incorporation of any Finance Document; and
- (iii) carry on its business as presently conducted where failure to do so could reasonably be expected to have a Material Adverse Effect.

27.2 Compliance with Laws

Each Obligor shall (and the Parent shall ensure that each Group Company will) comply in all respects with all laws to which it may be subject, if failure so to comply would materially impair its ability to perform its obligations under the Finance Documents.

27.3 Taxation

- (a) Each Obligor will, and will procure each of its Subsidiaries to, pay when due (or within any applicable time limit and in any event before material penalties are incurred if failure to do so could reasonably be expected to have a Material Adverse Effect), all Taxes imposed upon it or any of its assets, income or profits on any transactions undertaken or entered into by it except in relation to any bona fide tax dispute for which proper provision has been made in its accounts.
- (b) No Group Company may change its residence for Tax purposes except for any Permitted Tax Residency Change.

27.4 Merger

No Obligor shall (and the Parent shall ensure that no other Group Company will) enter into any amalgamation, demerger, merger, consolidation or corporate reconstruction other than a Permitted Transaction. Notwithstanding anything in this agreement to the contrary, any transaction by a Group Company the purpose of which is to redomesticate such entity in a United States jurisdiction (including any state or territory thereof or the District of Columbia) or any Permitted Jurisdiction shall be permitted under this agreement; provided that (i) for the avoidance of doubt, if such Obligor is a Borrower, the requirements of Clause 30.2 (other than Clauses 30.2(a)(i) and 30.2(a)(iii)) are satisfied as if such Borrower were a new Borrower and (ii) such Obligor must give equivalent security over its assets, to the extent required pursuant to the Agreed Security Principles and clause 30.8 (Further Security).

27.5 Change of Business

The Parent shall procure that:

- (a) the Group will continue the Business and use all reasonable endeavours to protect its rights thereto;and
- (b) the Group will not change its line of business from the lines of business conducted by the Group on the 2022 Effective Date (other than businesses incidental, complementary, similar related or ancillary thereto and reasonable extensions thereof).

27.6 Acquisitions and Investments

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no other Group Company will) acquire or invest (other than investments in the form of loans or guarantees, which are separately governed by Clause 27.13 (Loans or Credit) and Clause 27.14 (No Guarantees or Indemnities) and not by this Clause 27.6 (Acquisitions and Investments)) in a company or any shares or securities or a business (or, in each case, any interest in any of them);
- (b) Paragraph (a) above does not apply to an acquisition or investment which is:
 - (i) a Permitted Acquisition;
 - (ii) a Permitted Transaction;
 - (iii) a Permitted Share Issue;

27.7 **Joint Ventures**

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no other Group Company will):
 - (i) enter into, invest in or acquire (or agree to acquire) any shares, stocks, securities or other interest in any Joint Venture; or

- (ii) transfer any assets or lend to or guarantee or give an indemnity for or give Security for the obligations of a Joint Venture or maintain the solvency of or provide working capital to any Joint Venture (or agree to do any of the foregoing).
- (b) Paragraph (a) above does not apply to any acquisition (or agreement to acquire) any interest in a Joint Venture or transfer of assets (or agreement to transfer assets) to a Joint Venture or loan made to or guarantee given in respect of the obligations of a Joint Venture if:
 - (i) such transaction is a Permitted Joint Venture, Permitted Loans, Permitted Guarantees, Permitted Acquisition, Permitted Disposal or Permitted Security; and
 - (ii) no Event of Default is continuing at the time that the relevant Group Company makes the acquisition or transfer (or enters into the agreement to acquire or transfer) or other transaction referred to in paragraph (a)(i) or (a)(ii) above or the acquisition, transfer, agreement or other transaction is entered into pursuant to a binding contractual commitment made prior to the Event of Default.

27.8 [Reserved]

27.9 Pari Passu Ranking

Each Obligor shall ensure that any unsecured and unsubordinated claims of a Finance Party against it under the Finance Documents rank at least pari passu with the claims of all its other unsecured and unsubordinated creditors except those creditors whose claims are mandatorily preferred by laws of general application to companies.

27.10 Negative Pledge

In this clause 27.10, "Quasi-Security" means a transaction described in paragraph (a)(ii) below.

- (a) Except as permitted under paragraph (b) below:
 - (i) no Obligor shall (and the Parent shall ensure that no other Group Company will) create, incur, assume or permit to exist any Security over any of its assets.
 - (ii) no Obligor shall (and the Parent shall ensure that no other Group Company will):
 - sell, transfer or otherwise dispose of any of its assets on terms whereby they are or may be leased to or re-acquired by an Obligor or any other Group Company;
 - (B) sell, transfer or otherwise dispose of any of its receivables on recourse terms;

in circumstances where the arrangement or transaction is entered into primarily as a method of raising Financial Indebtedness or of financing the acquisition of an asset.

- (b) Paragraph (a) above does not apply to any Security or (as the case may be) Quasi-Security, which is:
 - (i) Permitted Security; or

- (ii) a Permitted Transaction.
- (c) For the avoidance of doubt, (x) this clause 27.10 shall only be tested solely at the initial time of consummation of the relevant transaction or action subject to clause 27.10(a) and (y) the references to "permit to exist" in this clause 27.10(a) is not intended to require any subsequent compliance with this clause 27.10 with respect to such relevant transaction or action (i.e., not a maintenance test).

27.11 Disposals

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no other Group Company will) enter into a transaction or a series of related transactions to sell, lease, transfer or otherwise dispose of any of its asset involving aggregate consideration in excess of the greater of (x) \$25,000,000 (or its equivalent in other currencies) and (y) 5% of the most recently reported Consolidated EBITDA;
- (b) Paragraph (a) above does not apply to any sale, lease, transfer or other disposal which is:
 - (i) a Permitted Disposal; or
 - (ii) a Permitted Transaction;

27.12 Arm's Length Basis

- (a) Except as permitted by paragraph (b) below, no Obligor shall (and the Parent shall ensure no other Group Company will) enter into any transaction with any Affiliate (excluding other Group Members) except on arm's length (or better for the Group) terms.
- (b) The following transactions shall not be a breach of this clause 27.12:
 - (i) intra-Group loans permitted under clause 27.13 (Loans or Credit);
 - (ii) fees, costs and expenses payable under or in connection with any Finance Document or under any other Transaction Document (or any investment and shareholders' agreement with the direct or indirect shareholders of any Group Company);
 - (iii) any transaction involving consideration less than the greater of (x) \$25,000,000 (or its equivalent in other currencies) and (y) 5% the most recently reported Consolidated EBITDA; and
 - (iv) any Permitted Transactions, Permitted Payments or Permitted Distributions or Permitted Intercompany Debt Disposals.

27.13 Loans or Credit

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no other Group Company will) be a creditor in respect of any Financial Indebtedness.
- (b) Paragraph (a) above does not apply to:
 - (i) a Permitted Loan; or
 - (ii) a Permitted Transaction.

27.14 No Guarantees or Indemnities

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no other Group Company will) incur or permit to exist any guarantee in respect of any Financial Indebtedness of any person.
- (b) Paragraph (a) does not apply to a guarantee which is:
 - (i) a Permitted Guarantee; or
 - (ii) a Permitted Transaction.
- (c) For the avoidance of doubt, (x) this clause 27.14 shall only be tested solely at the initial time of consummation of the relevant transaction or action subject to clause 27.14(a) and (y) the references to "permit to exist" in this clause 27.14(a) is not intended to require any subsequent compliance with this clause 27.14 with respect to such relevant transaction or action (i.e., not a maintenance test).

27.15 Dividends and Share Redemption

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no other Group Company will) (collectively, such restrictions, the "Restricted Payments"):
 - declare, make or pay any dividend, charge, fee or other distribution (or interest on any unpaid dividend, charge, fee or other distribution) (whether in cash or in kind) on or in respect of its share capital (or any class of its share capital);
 - (ii) repay or distribute any dividend or share premium reserve; or
 - (iii) [reserved]; or
 - (iv) redeem, repurchase, defease, retire or repay any of its share capital or resolve to do so.
- (b) Paragraph (a) above does not apply to:
 - (i) a Permitted Distribution;
 - (ii) a Permitted Transaction;
 - (iii) a Permitted Payment; or
 - (iv) a Permitted Intercompany Debt Disposal.

27.16 Investor Debt

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no other Group Company will):
 - repay or prepay any principal amount (or capitalised interest) outstanding in respect of the Investor Debt;
 - (ii) $\;\;$ pay any interest or any other amounts payable in connection with the Investor Debt; or
 - (iii) purchase, redeem, defease or discharge any amount outstanding with respect to the Investor Deht.

(b) Paragraph (a) does not apply to a payment, repayment, purchase, redemption defeasance or discharge which is a Permitted Payment or a Permitted Distribution.

27.17 Financial Indebtedness

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Parent shall ensure that no other Group Company will) incur, create, assume or permit to exist any Financial Indebtedness.
- (b) Paragraph (a) above does not apply to Financial Indebtedness which is:
 - (i) Permitted Financial Indebtedness; or
 - (ii) a Permitted Transaction.
- (c) It is understood and agreed that accrual of interest, accrual of dividends, the accretion of accreted value, increase in fair value, the accretion or amortization of original issue discount, the payment of interest in the form of additional Financial Indebtedness, the payment of dividends in the form of additional shares of preferred stock or Disqualified Equity or the reclassification of commitments or obligations not treated as Financial Indebtedness due to a change in Accounting Principles will not be deemed to be an incurrence of Financial Indebtedness for purposes of this Clause 27.17.
- (d) For the avoidance of doubt, (x) this clause 27.17 shall only be tested solely at the initial time of consummation of the relevant transaction or action subject to clause 27.17(a) and (y) the references to "permit to exist" in this clause 27.17(a) is not intended to require any subsequent compliance with this clause 27.17 with respect to such relevant transaction or action (i.e., not a maintenance test).

27.18 Share Capital

- (a) No Obligor shall (and the Parent shall ensure that no other Group Company will) issue any shares except pursuant to:
 - (i) a Permitted Share Issue; or
 - (ii) a Permitted Transaction.

27.19 Insurance

- (a) Each Obligor shall (and the Parent shall ensure that each other Group Company will) maintain insurances on and in relation to its business and assets against those risks and to the extent as is usual for companies carrying on the same or substantially similar business as may be reasonably available in the insurance market (including by way of insurance maintained with a "captive insurer").
- (b) All insurances must be with reputable independent (except in relation to a captive waiver) insurance companies or underwriters.

27.20 Cash Movement

(a) Except as permitted under paragraph (b) below, no Obligor will, and the Parent will procure that no other Group Company will, be a party to any contractual or similar restriction (except as set out in any Finance Document, in any cash pooling arrangement or in any other Secured Debt Document (as defined in the Intercreditor Agreement) or for any contractual or similar restriction existing in favour of or for the benefit of another Group Company, in each case to the extent such restriction could not reasonably be expected to prevent the Borrowers meeting debt service

obligations under the Finance Documents) by which any Group Company is prohibited from making loans or making any payment of dividends, distributions of income or other amounts.

- (b) Paragraph (b) above does not apply to:
 - (i) any agreement of any person that becomes a Group Company after the 2022 Effective Date that existed prior to the time such person became a Group Company; <u>provided</u> that such restrictions are not created in contemplation of or in connection with such acquisition, and any replacement or refinancing thereof so long as the restrictions are not materially more restrictive (as determined by the Borrower in good faith);
 - (ii) arising under customary provisions contained in any agreement relating to a Permitted Disposal pending the consummation of such Permitted Disposal;
 - (iii) restrictions in agreements that (as determined in good faith by the Parent) will not prevent the Obligors from satisfying their payment obligations hereunder; or
 - (iv) any agreements existing on the 2022 Effective Date and set forth in Schedule 14 part 8.

27.21 Intellectual Property

Each Obligor will, and will procure that each Group Company will:

- ensure that it legally and beneficially owns or has all necessary consents to use and, where applicable, is recorded as the registered proprietor of all the Intellectual Property that it requires in order to conduct its business;
- (b) observe and comply with all obligations and laws applicable to it in relation to material Intellectual Property;
- (c) maintain and protect material Intellectual Property, including maintaining and, where reasonable, pursuing registration,

in each case, where failure to do so could reasonably be expected to result in a Material Adverse Effect.

27.22 **[Reserved]**

27.23 Changes to Material Contracts

(a) No Obligor will, and the Parent will procure that no other Group Company will, agree to any amendment, variation or waiver of any term of any Material Contract (whether as part of a single transaction or as a series of transactions) which would materially adversely affect the interests and business of the Group as a whole without the prior written consent of the Majority Lenders (such consent not to be unreasonably withheld).

27.24 Financial Assistance and Distributions Undertaking

Each Obligor shall, and the Parent shall procure each member of the Group will, comply in all respects with Sections 678 and 679 of the Companies Act 2006, Part 17 of the Companies (Jersey) Law 1991 and any equivalent legislation in other jurisdictions (in each case to the extent applicable) including in relation to the execution of the Transaction Security Documents, transferring money directly or indirectly to the Parent and payment of amounts due under this agreement.

27.25 Treasury Transactions

No Obligor shall (and the Parent will procure that no other Group Company will) enter into any Treasury Transaction, other than:

- (a) interest rate or exchange rate hedging transactions entered into in respect of any Financial Indebtedness of the Group;
- (b) spot and forward delivery foreign exchange contracts entered into in the ordinary course of business and not for speculative purposes; and
- (c) any Treasury Transaction entered into for the hedging of actual or projected real exposures arising in the ordinary course of business activities of a Group Company and not for speculative purposes.

27.26 Guarantor Coverage

- (a) The Parent shall procure that the aggregate Consolidated EBITDA of the Borrowers and the Guarantors (determined for this purpose in respect of the relevant Group Company rather than the Group as a whole) represents not less than 80 per cent of Consolidated EBITDA of the Group;
- (b) The undertaking in paragraph (a) above shall be tested on the date falling 30 days after delivery to the Facility Agent of the Annual Financial Statements for the Group pursuant to clause 25.1 (Financial Statements) by reference to such Annual Financial Statements.

27.27 Further Assurance

- (a) Subject to the Agreed Security Principles, each Obligor shall (and the Parent shall procure that each other Group Company will) at its own expense promptly do all such acts or execute all such documents (including assignments, transfers, mortgages, charges, notices and instructions) as the Security Agent may reasonably specify (and in such form as the Security Agent may reasonably require in favour of the Security Agent or its nominee(s)):
 - (i) to perfect the Security created or intended to be created under or evidenced by the Transaction Security Documents (which may include the execution of a mortgage, charge, assignment or other Security over all or any of the assets which are, or are intended to be, the subject of the Transaction Security) or for the exercise of any rights, powers and remedies of the Security Agent or the Finance Parties provided by or pursuant to the Finance Documents or by law;
 - (ii) to confer on the Security Agent or confer on the Finance Parties Security over any property and assets of that Obligor located in any jurisdiction equivalent or similar to the Security intended to be conferred by or pursuant to the Transaction Security Documents; and/or
 - (iii) to facilitate the realisation of the assets which are, or are intended to be, the subject of the Transaction Security.
- (b) Subject to the Agreed Security Principles, each Obligor shall (and the Parent shall procure that each Group Company shall) take all such action as is available to it (including making all filings and registrations) as may be necessary for the purpose of the creation, perfection, protection or maintenance of any Security conferred or intended to be conferred on the Security Agent or the Finance Parties by or pursuant to the Finance Documents.

27.28 Centre of Main Interests

No Obligor (other than in the case of any Group Company other than a Borrower to the extent its centre of main interests changes to the United Kingdom or the US) shall, and each Obligor will procure that none of its Subsidiaries will, do anything to change the location of its centre of main interests, for the purposes of Regulation (EU) 2015/848 of the European Parliament and of the Council of 20 May 2015 on insolvency proceedings, where that change would be materially adverse to the interests of the Finance Parties.

27.29 [Reserved]

27.30 Termination Event Protection

The Parent acknowledges the importance to the Lenders and to the business of the Group of the continuation of the 100 Year Agreements and the ability of the relevant members of the Group to exploit the rights conferred upon the Group by the 100 Year Agreements. Accordingly, the Parent and all other Obligors undertake to the Lenders as follows:

- (a) to comply in all material respects with the terms of the 100 Year Agreements and not do (or omit to do) any act or thing which could reasonably be expected to result in a termination of the 100 Year Agreements or give the right to FIAS or FIAF to terminate the 100 Year Agreements;
- (b) to ensure that no Group Company agrees to amend, modify or restate any of the 100 Year Agreements or any agreement directly relating to or affecting the 100 Year Agreements which will have the effect of introducing any new or additional termination rights in favour of FIAS, FIAF or any third party if such introduction will:
 - (i) be detrimental in any material respect to any member of the Group; or
 - (ii) result in termination rights arising if any Finance Party demands payment from or accelerates any indebtedness of or takes any Enforcement Action (as defined in the Intercreditor Agreement) against any Parent Undertaking of SLEC (including enforcing any security over any shares in SLEC or any Parent Undertaking of SLEC); or
 - (iii) otherwise be adverse to the interests of the Finance Parties in any material respect;
- (c) to notify the Facility Agent promptly and keep the Facility Agent fully informed if there is any deterioration in the financial condition of any Subsidiary of SLEC or any other event which could result in such Subsidiary going into Insolvent Liquidation in the following 12 months (an "Insolvency Risk Notification");
- (d) following service of an Insolvency Risk Notification, at the request of the Facility Agent to take all reasonable steps to transfer the shares in the Subsidiary the subject of the Insolvency Risk Notification, such that the Subsidiary becomes a Subsidiary of a Parent Undertaking of SLEC and ceases to be a Subsidiary (direct or indirect) of SLEC provided that no such action may be required if it would itself be in breach of the 100 Year Agreements;
- (e) if a breach of the financial covenant set out in clause 26 (Financial Definitions and Financial Covenant) occurs and for so long as it is continuing, the Facility Agent may serve notice in writing on the Parent terminating the right of SLEC and any Subsidiary of SLEC to borrow (a "Restricted SLEC Company") under any Revolving Facility and under any Ancillary Facility (a "Revolver Termination Notice"). Forthwith upon service of a Revolver Termination Notice the rights of SLEC and any Subsidiary of SLEC to make any further borrowings or other Utilisations under any Revolving Facility and any Ancillary Facility will cease. Accordingly, on the next Interest

Payment Date applicable to the Revolving Facility Utilisations the existing Revolving Facility Utilisations by a SLEC Company must be refinanced by one or more Utilisations undertaken by another Subsidiary of the Parent (which is not a Restricted SLEC Company) reasonably acceptable to the Facility Agent. In addition, the Parent will use all reasonable endeavours to procure that any borrowings or other utilisations under any Ancillary Facility by a Restricted SLEC Company are refinanced as soon as practicable after receipt of a Revolver Termination Notice such that no Restricted SLEC Company has any utilisation outstanding in respect of such Ancillary Facility.

27.31 Basket Carry Forward

If in any Financial Year of the Parent the aggregate amount of the basket permitted pursuant to paragraph (o) of the definition of Permitted Disposal actually utilised is less than the relevant fixed dollar amount basket originally available for that Financial Year (the difference being referred to as the "Available Amount"), then the maximum fixed dollar amount basket for the immediately following Financial Year (the "Carry Forward Year") shall be increased by an amount equal to the Available Amount. In any Carry Forward Year, the original amount of that basket shall be treated as having been applied after any Available Amount carried forward into such Carried Forward Year. The basket for the next Financial Year may be carried back to the current financial year with a corresponding reduction for the next following Financial Year.

27.32 **Ratings**

On and from the 2022 Effective Date, the Parent shall use commercially reasonable efforts to maintain a public corporate family rating or analogous public rating (but in each case not a specific rating) from any two of S&P, Moody's or Fitch in respect of itself.

27.33 Sanctioned Territories

- (a) No Borrower shall directly or indirectly knowingly use the proceeds of any Loan or otherwise knowingly make available any such proceeds to any person for the purposes of financing the activities of, transactions with, or acquiring any interest in any person incorporated or operating in, what is at the time that such proceeds are applied a Sanctioned Territory or otherwise the subject of any Economics Sanctions Law or in any other manner that would result in a violation of Economic Sanctions Laws by any Party, except to the extent licensed or otherwise authorised by the competent authority responsible for implementation or enforcement of the applicable Economic Sanctions Law.
- (b) The Parent shall promptly deliver to the Facility Agent copies of any written communications it or any other member of the Group may have with OFAC, the United Nations or any other authority or government agency in the US, the United Kingdom or the European Union in relation to sanctions unless (i) it is of a price sensitive nature that has not yet been publicly announced pursuant to the listing rules of a regulated stock exchange applicable to any direct or indirect shareholder in the Parent, (ii) disclosure to any Finance Party (or any of their respective representatives or contractors) is prohibited by applicable law, (iii) it is subject to attorney-client or similar privilege or constitutes attorney work product or (iv) any Group Company owes confidentiality obligations to any third party that prohibits such disclosure (provided that such confidentiality obligations were not entered into solely in contemplation of this clause (iv)).

27.34 Foreign and Corrupt Practices Act

No part of the proceeds of the Loans will be used, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party,

candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended.

27.35 **[Reserved]**

27.36 Restrictions on Investor Debt

Notwithstanding any other provisions of this agreement, except with the prior consent of the Majority Lenders:

- (a) no Obligor shall (and the Parent shall ensure that no other Group Company will) make, agree to or permit any Permitted Intercompany Debt Disposal or Permitted Tax Residency Change if such Permitted Intercompany Debt Disposal or Permitted Tax Residency Change could reasonably be expected to result in a Material Adverse Effect;
- (b) the only Group Companies permitted to owe to a person other than a Group Company any Financial Indebtedness which is Investor Debt constituting Permitted Financial Indebtedness under paragraph (b) of the definition of Permitted Financial Indebtedness in Clause 1.1 (Definitions) and not otherwise Permitted Financial Indebtedness pursuant to any other provision of such definition (such Investor Debt outstanding from Group Companies being "Restricted Investor Debt" for the purposes of this clause 27.37), shall be the Group Companies which are Obligors (without prejudice to the permission of any Group Company to have any other Permitted Financial Indebtedness outstanding in accordance with the provisions herein); and
- (c) the Parent shall, or shall ensure that another Obligor shall, promptly notify the Facility Agent when any person who is not a Group Company (an "Investor Creditor" for the purposes of this clause 27.37) becomes the creditor of any Restricted Investor Debt and shall ensure that such Investor Creditor shall promptly take such steps as are reasonably required on its part to enter into a security document with the Security Agent in a form reasonably requested by the Security Agent creating security over the relevant Restricted Investor Debt (if such security has not already been granted by such Investor Creditor) and no other assets of the Investor Creditor provided that:
 - (i) the terms of such security document shall be substantially the same as the terms applicable to the Transaction Security created by Group Companies over intercompany receivables and no more onerous on the Investor Creditor and shall permit, provided that no Default has occurred, the transfer of the relevant Restricted Investor Debt to another person to whom such Restricted Investor Debt is permitted to be outstanding in accordance with this agreement without any requirement for consent provided that such person is party to such a security document (or another Transaction Security Document creating security over such Restricted Investor Debt);
 - (ii) Restricted Investor Debt may only be outstanding to an Investor Creditor which has complied with the requirements of this clause in relation to the relevant Restricted Investor Debt and from and including the 2022 Effective Date new Restricted Investor Debt shall only be advanced to an Obligor by an Investor Creditor which has prior to such advance taken such steps as are reasonably required on its part to enter into a security document with the Security Agent as provided for in this clause 27.37 and provided that for this purpose any Restricted Investor Debt transferred to an Investor Creditor pursuant to a Permitted Intercompany Debt Disposal shall not be "new" Restricted Investor Debt.

For the purposes of this clause 27.37 a transfer includes any sale, transfer, distribution, assignment or other disposal and the words "transferred" and "transfers" shall be construed accordingly."

28. EVENTS OF DEFAULT

Each of the events or circumstances set out in clause 28 is an Event of Default. References in this clause 28 to "Group Company" are deemed to exclude any company which is a Dormant Subsidiary.

28.1 Non-Payment

An Obligor does not pay on the due date any principal or any other amount payable pursuant to a Finance Document at the place and in the currency in which it is expressed to be payable unless:

- (a) in the case of interest, payment is made within 3 Business Days of its due date; and
- (b) in the case of any other amount (other than principal and interest):
 - (i) payment is made within five Business Days of its due date; or
 - (ii) if the failure is by reason of operational or administrative error, payment is made within ten Business Days of its due date.

28.2 Financial Covenant

- (a) Clause 26.2 (Financial Condition) is not satisfied provided that no breach or default under clause 26.2 (Financial Condition) will constitute a Default with respect to the Term Facility B unless the Majority TLA/RCF Lenders have accelerated any Loans outstanding under the Term Facility A and the Facility RCF as a result of such breach or default and such declaration has not be rescinded.
- (b) No Event of Default under paragraph (a) above will occur if New Shareholder Injections are made within ten Business Days of the delivery of the financial statements in accordance with clause 25.1 (Financial Statements) evidencing such failure to comply and if the financial covenant in clause 26.2 (Financial Condition) were re-tested for the relevant Testing Period, on the basis that such New Shareholder Injections are added to Consolidated EBITDA for that purpose only (and, for the avoidance of doubt, not taken into account as Cash or Cash Equivalent Investments in the calculation of Consolidated Net First Lien Secured Debt), such financial covenant would then be complied with. The minimum amount of any New Shareholder Injections provided pursuant to this paragraph required on such re-testing to ensure such financial covenant would then be complied with shall be taken into account in calculating Consolidated EBITDA for the purpose of clause 26.2 (Financial Condition) for each Testing Period (if tested) that ends on the last day of one of the next three Financial Quarters.
- (c) The right in clause 28.2(b) shall only be exercisable five times with not less than one clear Financial Ouarter between each such exercise.

28.3 Other Obligations

- (a) Any Obligor shall fail to comply with the provisions of clause 25.7 (Notification of Default).
- (b) An Obligor does not comply with any provision of the Finance Documents (other than those referred to in clause 28.1 (Non-Payment) and clause 28.2 (Financial Covenant)).

(c) No Event of Default under paragraph (b) above will occur if the failure to comply is capable of remedy and is remedied within 30 days of the Facility Agent giving notice to the Obligors' Agent or relevant Obligor.

28.4 Misrepresentation

- (a) Any representation or statement made or deemed to be made by an Obligor in the Finance Documents or any other document delivered by or on behalf of any Obligor under or in connection with any Finance Document is or proves to have been incorrect or misleading (where the relevant representation or warranty is not otherwise subject to a materiality qualification in any material respect) when made or deemed to be made.
- (b) No Event of Default under paragraph (a) above will occur if the event or circumstance giving rise to the breach is capable of remedy and is remedied within 30 days of the Facility Agent giving notice to the Obligors' Agent or relevant Obligor.

28.5 **Cross Default**

- (a) Any Financial Indebtedness of any Group Company is not paid when due nor within any originally applicable grace period;
- (b) Any Financial Indebtedness of any Group Company is declared to be or otherwise becomes due and payable prior to its specified maturity as a result of an event of default (however described);
- (c) Any commitment (other than a Commitment) for any Financial Indebtedness of any Group Company is cancelled or suspended by a creditor of any Group Company as a result of an event of default (however described);
- (d) Any creditor of any Group Company becomes entitled to declare any Financial Indebtedness of any Group Company due and payable prior to its specified maturity as a result of an event of default (however described)(other than arising as a result of a breach of this agreement); or
- (e) No Event of Default will occur under paragraphs (a) to (d) above if it arises as a result of a breach of this agreement or if the aggregate amount of any Financial Indebtedness or commitment for any Financial Indebtedness falling within paragraphs (a) to (d) above is less than the greater of (i) \$100,000,000 (or its equivalent in other currencies) and (ii) 20% of the most recently reported Consolidated EBITDA (or its equivalent in any other currency or currencies).

28.6 Insolvency

- (a) An Obligor is unable or admits inability to pay its debts as they fall due, suspends or threatens to suspend making payments on any of its debts or, by reason of actual or anticipated financial difficulties, commences negotiations with one or more of its creditors with a view to rescheduling any of its indebtedness.
- (b) A moratorium is declared in respect of any indebtedness of any Obligor. If a moratorium occurs, the ending of the moratorium will not remedy any Event of Default caused by that moratorium.
- (c) Any Obligor is declared "bankrupt" within the meaning of Article 8 (Meaning of Bankruptcy) of the Interpretation (Jersey) Law 1954.

28.7 Insolvency Proceedings

- (a) Any corporate action, legal proceedings or other procedure or step is taken in relation to:
 - the suspension of payments, a moratorium of any indebtedness, winding-up, dissolution, administration, examination or reorganisation (by way of voluntary arrangement, scheme of arrangement or otherwise) of any Group Company;
 - (ii) a composition, compromise, assignment or arrangement with any creditor of any Obligor;
 - (iii) the appointment of a liquidator, receiver, examiner, administrator, administrative receiver, compulsory manager or other similar officer in respect of any Obligor or any of its assets;
 - (iv) enforcement of any Security over any assets of any Obligor;
 - (v) any Obligor seeking or taking any action preliminary to it seeking, in relation to itself or its property, any grant, declaration, decision or winding-up to which reference is made in Article 8 (Meaning of Bankruptcy) of the Interpretation (Jersey) Law 1954; or
 - (vi) any proceedings being commenced or other steps being taken for any Obligor to be made subject to, or the subject of, any grant, declaration, decision or winding-up to which reference is made in Article 8 (Meaning of Bankruptcy) of the Interpretation (Jersey) Law 1954,

unless, in the case of a petition or action commenced by a third party, the relevant action is frivolous or vexatious, an abuse of insolvency process, or is dismissed at least two Business Days prior to the relevant return date.

- (b) Paragraph (a) shall not apply to:
 - any winding-up petition which is frivolous or vexatious, an abuse of insolvency process, or is dismissed at least two Business Days prior to the relevant return date;
 - (ii) the ASM Dissolution;
 - (iii) any step or procedure contemplated by the definition of Permitted Transaction;
 - (iv) any Permitted Group Company Disposal; or
 - (v) any transaction permitted by this agreement.

28.8 Creditors' Process

Any attachment, sequestration, distress or execution or any analogous process in any jurisdiction affects a material asset of the Group taken as a whole and is either not discharged within 60 days or is not being contested in good faith.

28.9 Similar Events Elsewhere

There occurs in relation to any Obligor or any of its assets in any country or territory in which it is incorporated or carries on business or to the jurisdiction of whose courts it or any of its assets is subject any event which appears to the Facility Agent to correspond in that country or territory with any of those mentioned in clauses 28.6 (Insolvency) to 28.8 (Creditors' Process) (inclusive) in the case of any Obligor (but, for the avoidance of doubt,

including any applicable thresholds, grace periods and qualifications in such clauses and other than solely as a result of balance sheet liabilities exceeding balance sheet assets).

28.10 Unlawfulness and Invalidity

- (a) (i) it is or becomes unlawful for an Obligor to perform any of its material obligations under the Finance Documents or (ii) any Transaction Security created or expressed to be created or evidenced by the Transaction Security Documents ceases to be effective or any subordination created under the Intercreditor Agreement is or becomes unlawful; other than by reason of (x) failure to maintain possession of any collateral actually delivered to the Security Agent or any sub-agent of the Security Agent (to the extent possession is required for perfection) or (y) failure to file Uniform Commercial Code (or equivalent) continuation statements.
- (b) Any material obligation or obligations of any Obligor under any Finance Document are not (subject to the Legal Reservations) or cease to be legal, valid, binding or enforceable and the cessation individually or cumulatively materially and adversely affects the interests of the Lenders under the Finance Documents.
- (c) Any Finance Document ceases to be in full force and effect except by reason of Legal Reservations or any Transaction Security or is alleged by a Group Company to be ineffective.

28.11 Intercreditor Agreement

Any party to the Intercreditor Agreement (other than a Finance Party) fails to comply with the provisions of, or does not perform its obligations under, that Intercreditor Agreement to a material extent.

28.12 Cessation of Business

The Group ceases to carry on (or threatens to cease to carry on) all or a substantial part of its business (taken as a whole).

28.13 **[Reserved]**

28.14 Audit Qualification

The Auditors of the Group (or the Parent Reporter, if the Parent Reporter Option is elected) qualify the audited annual consolidated financial statements delivered pursuant to clause 25.1(a) (Financial Statements) in a manner which is materially adverse to the interests of the Finance Parties in the context of the Finance Documents taken as a whole; <u>provided</u> that a "going concern" or like qualification, exception or explanatory paragraph that is due to (i) the impending maturity of any Facility or any other Financial Indebtedness within one year from the time such opinion is delivered or (ii) any actual or prospective inability to satisfy a financial maintenance covenant (including compliance with Clause 26.2 (Financial Condition)) shall not trigger an Event of Default pursuant to this Clause 28.13; provided further that no Event of Default under this Clause 28.13 above will occur if the failure to comply is capable of remedy and is remedied within 30 days of the Facility Agent giving notice to the Obligors' Agent or relevant Obligor of the failure to comply.

28.15 [Reserved]

28.16 Repudiation and Rescission of Agreements

(a) An Obligor rescinds or purports to rescind or repudiates or purports to repudiate a Finance Document or any Transaction Security or evidences an intention to rescind or repudiate a Finance Document or any Transaction Security. (b) Any party to an Intercreditor Agreement rescinds or purports to rescind or repudiates or purports to repudiate any of those agreements or instruments in whole or in part where to do so has or is likely to have a material adverse effect on the interests of the Lenders under the Finance Documents.

28.17 Litigation

Any litigation, arbitration, administrative, governmental, regulatory or other investigations, proceedings or disputes are commenced by or against any Group Company which is reasonably likely to be determined against the relevant Group Company and if so determined could be reasonably expected to have a Material Adverse Effect.

28.18 Material Contracts

- (a) Any Material Contract (other than the Concorde Agreement) is terminated, rescinded or otherwise ceases to be in full force and effect (other than on expiry under its terms, as in force at the 2022 Effective Date) and this could reasonably be expected to cause a Material Adverse Effect.
- (b) Except as set out in paragraph (e) below, the Concorde Agreement is terminated, rescinded or otherwise ceases to be in full force and effect (other than on expiry under its terms) and this could reasonably be expected to cause a Material Adverse Effect.
- (c) The 100 Year Agreements are terminated rescinded or otherwise cease to be in full force and effect (other than on expiry under any of their terms, as in force at the 2022 Effective Date).
- (d) Any Group Company repudiates any of its obligations under any Material Contract which could reasonably be expected to cause a Material Adverse Effect.
- (e) For the avoidance of doubt, the replacement from time to time of the then existing Concorde Agreement shall not constitute an Event of Default.

28.19 **[Reserved**]

28.20 Change of Control

- (a) Subject to paragraph (b) below, any event or circumstance occurs which results in a Change of Control.
- (b) No Event of Default under paragraph (a) will occur if:
 - (i) that Change of Control arises as a result of equity securities which are intended to form part, or the whole, of a Listing being transferred to any person: (A) which is to act as an intermediary as part of the proposed settlement process for that Listing; or (B) in the context of any reorganisation implemented prior to and for the purpose of facilitating the Listing (each being a "Pre-Listing Transfer"); and
 - (ii) if that Listing does not complete within the timeframe from the commencement of the settlement process, which is the generally accepted timeframe from completion of a Listing on the relevant regulated stock exchange or market the relevant equity securities are as soon as reasonably practicable (and in any event within 45 days of the Pre-Listing Transfer being effected) transferred back to such persons and in such amounts as is necessary to remedy the Change of Control that would otherwise have arisen if not for this paragraph (b).

28.21 [Reserved]

28.22 Acceleration

- (a) On and at any time after the occurrence of an Event of Default which is continuing (excluding an Event of Default under clause 28.2(a) (Financial Covenant) in respect of which an aggregate amount equal to, or in excess of, the dollar amount threshold specified in clause 28.5(e) (Cross Default) above has not occurred and is not continuing), the Facility Agent may, and shall if so directed by the Majority Lenders, by notice to the Obligors' Agent:
 - (i) cancel the Total Commitments and/or Ancillary Commitments at which time they shall immediately be cancelled;
 - (ii) declare that all or part of the Utilisations, together with accrued interest, and all other amounts accrued or outstanding under the Finance Documents be immediately due and payable, at which time they shall become immediately due and payable;
 - (iii) declare that all or part of the Utilisations be payable on demand, at which time they shall immediately become payable on demand by the Facility Agent on the instructions of the Majority Lenders;
 - (iv) declare that cash cover in respect of each Letter of Credit is immediately due and payable at which time it shall become immediately due and payable;
 - declare that cash cover in respect of each Letter of Credit is payable on demand at which time
 it shall immediately become due and payable on demand by the Facility Agent on the
 instructions of the Majority Lenders;
 - (vi) declare all or any part of the amounts (or cash cover in relation to those amounts) outstanding under the Ancillary Facilities to be immediately due and payable, at which time they shall become immediately due and payable;
 - (vii) declare that all or any part of the amounts (or cash cover in relation to those amounts) outstanding under the Ancillary Facilities be payable on demand, at which time they shall immediately become payable on demand by the Facility Agent on the instructions of the Majority Lenders; and/or
 - (viii) exercise or direct the Security Agent to exercise any or all of its rights, remedies, powers or discretions under the Finance Documents.
- (b) Subject to the terms of the Intercreditor Agreement, on and at any time after the occurrence of an Event of Default under clause 28.2(a) (Financial Covenant) which is continuing, the Facility Agent may, and shall if so directed by the Majority TLA/RCF Lenders, by notice to the Parent:
 - cancel the Total Revolving Facility Commitments and/or Ancillary Commitments at which time they shall immediately be cancelled;
 - (ii) declare that all or part of the Revolving Facility Utilisations and Term Facility A (if outstanding), together with accrued interest, and all other amounts accrued or outstanding under any Revolving Facility and any Term Facility A be immediately due and payable, at which time they shall become immediately due and payable;
 - (iii) declare that all or part of the Revolving Facility Utilisations and Term Facility A Loans (if outstanding) be payable on demand, at which time they shall

- immediately become payable on demand by the Facility Agent on the instructions of the Majority TLA/RCF Lenders;
- (iv) declare that cash cover in respect of each Letter of Credit is immediately due and payable at which time it shall become immediately due and payable;
- (v) declare that cash cover in respect of each Letter of Credit is payable on demand at which time
 it shall immediately become due and payable on demand by the Facility Agent on the
 instructions of the Majority TLA/RCF Lenders;
- (vi) declare all or any part of the amounts (or cash cover in relation to those amounts) outstanding under the Ancillary Facilities to be immediately due and payable, at which time they shall become immediately due and payable; and
- (vii) declare that all or any part of the amounts (or cash cover in relation to those amounts) outstanding under the Ancillary Facilities be payable on demand, at which time they shall immediately become payable on demand by the Facility Agent on the instructions of the Majority TLA/RCF Lenders.

29. CHANGES TO THE LENDERS

29.1 Assignments and Transfers by the Lenders

Subject to this clause 29, a Lender (the "Existing Lender") may:

- (a) assign any of its rights; or
- (b) transfer any of its rights and obligations,

under any Finance Document to another person (the "New Lender"), other than any Disqualified Lender.

29.2 Conditions of Assignment or Transfer

- (a) An Existing Lender must obtain the consent of Obligors' Agent (not to be unreasonably withheld (it being understood that, without limitation to other reasonable basis, the investment objectives history of any proposed assignee and strategic efforts of such assignee shall be a reasonable basis for the Obligors' Agent to withhold such consent) and which will be deemed to be given if, ten Business Days after such consent has been requested of Obligors' Agent, no objection to the relevant assignment or transfer has been made) before it may make an assignment or transfer in accordance with clause 29.1 (Assignments and Transfers by the Lenders) unless the assignment or transfer is:
 - (i) to another Lender or an Affiliate of a Lender;
 - (ii) if the Existing Lender is a fund, to a fund which is a Related Fund of the Existing Lender; or
 - (iii) made at a time when an Event of Default is continuing.
- (b) The consent of the Issuing Bank is required for any assignment or transfer by an Existing Lender of any of its rights and/or obligations under a Revolving Facility in respect of which that Issuing Bank is an Issuing Bank (such consent not to be unreasonably withheld or delayed).
- (c) Each assignment or transfer of any Lender's participation (when aggregated with, for this purpose, the participation of its Affiliates or Related Funds) shall be in a

minimum amount of \$1,000,000 in relation to participations in each Term Facility and shall be on terms such that no Lender (when aggregated with, for this purpose, Affiliates or, Related Funds) at any time holds an aggregate participation of less than:

- (i) in the case of a bank (when aggregated with, for this purpose, Affiliates or, Related Funds), \$1,000,000 with respect to each such Facility; or
- (ii) in the case of any other Lender (when aggregated with, for this purpose, Affiliates or, Related Funds), \$1,000,000,

unless, in each case, the assignment or transfer is of the whole of that Lender's participation under this agreement.

- (d) An assignment of rights will only be effective on:
 - receipt by the Facility Agent of written confirmation from the New Lender (in form and substance satisfactory to the Facility Agent) that the New Lender will assume the same obligations to the other Finance Parties and the other Secured Parties as it would have been under if it was an Original Lender;
 - (ii) the New Lender entering into the documentation required for it to accede as a party to the Intercreditor Agreement; and
 - (iii) the performance by the Facility Agent of all necessary "know your customer" or other similar checks under all applicable laws and regulations in relation to such assignment to a New Lender, the completion of which the Facility Agent shall promptly notify to the Existing Lender and the New Lender.
- (e) A transfer of rights and obligations will only be effective if the New Lender enters into the documentation required for it to accede as a party to the Intercreditor Agreement and if the procedure set out in clause 29.5 (Procedure for Transfer) is complied with.
- (f) If:
 - a Lender assigns or transfers any of its rights or obligations under the Finance Documents or changes its Facility Office (including, any designation of a Designated Entity pursuant to clause 5.6 (Designated Entities); and
 - (ii) as a result of circumstances existing at the date the assignment, transfer or change occurs, an Obligor would be obliged to make a payment to the New Lender, Designated Entity or Lender acting through its new Facility Office under clause 18 (Tax Gross-Up and Indemnities) or clause 19 (Increased Costs),

then the New Lender, Designated Entity or Lender acting through its new Facility Office is only entitled to receive payment under those clauses to the same extent as the Existing Lender or Lender acting through its previous Facility Office would have been if the assignment, transfer or change had not occurred.

- (g) Subject to paragraph (h) below, nothing in this agreement will restrict the ability of a Lender to subparticipate or sub-contract any of its obligations under any Finance Document.
- (h) No Lender may enter into a sub-participation or sub-contract under which voting rights (other than voting in respect of waivers and amendments that require the consent of all the Lenders pursuant to clause 40.2 (Exceptions)) are transferred or are reasonably likely to be transferred unless the relevant Lender has obtained the prior consent of Obligors' Agent (not to be unreasonably withheld and which will be

deemed to be given if, ten Business Days after such consent has been requested of Obligors' Agent, no objection to the relevant sub-participation or sub-contract has been made). A Lender shall notify Obligors' Agent of any such sub participation or sub contracting by it. Notwithstanding anything herein to the contrary, no sub-participation or sub-contract can be made to a Disqualified Lender (but only to the extent a list of Disqualified Lenders has been made available to any Lender that requests such list).

- (i) If a decision of the Lenders (or certain of the Lenders) is required, then:
 - (i) each Lender which has Commitments in respect of which it is entitled to cast a vote or votes ("Eligible Commitments") may elect to vote all or only part of those Eligible Commitments and may also vote part of its Eligible Commitments in one way whilst voting another part of its Eligible Commitments in another way; and
 - (ii) any New Lender or sub-participant which, in contravention of any provision of this clause, has been assigned or transferred Commitments, or has otherwise required rights in respect of Commitments, in respect of which it is entitled to cast a vote or votes shall have no right to veto a decision of the Lenders, Super Majority Lenders or Majority Lenders (as the case may be) without the prior written consent of the Obligors' Agent or the Parent.
- (j) The Facility Agent, acting solely for this purpose as a non-fiduciary agent of the Borrowers, shall maintain at one of its offices in the US a copy of each Transfer Certificate delivered to it and a register for the recordation of the names and addresses of the Lenders, and holders of voting subparticipations and the Commitments of, and principal amounts (and stated interest) of the Loans owing to, each Lender and voting sub-participant pursuant to the terms of this agreement from time to time (the "Register"). The entries in the Register shall be conclusive absent manifest error, and each Borrower, the Facility Agent and the Lenders shall treat each person whose name is recorded in the Register pursuant to the terms hereof as a Lender or holder of a sub-participation for all purposes of this agreement. The Register shall be available for inspection by the Obligors and any Lender, at any reasonable time and from time to time upon reasonable prior notice. Notwithstanding anything to the contrary in any Finance Document, no transfer of a Loan or a sub-participation shall be valid unless and until recorded it is in the Register.

29.3 Assignment or Transfer Fee

Unless the Facility Agent otherwise agrees and excluding an assignment or transfer:

- (a) to an Affiliate of a Lender;
- (b) to a Related Fund; or
- (c) made within the later of (x) 30 days of the date of the Dorna Acquisition and (y) the period separately provided in the master consent agreement between the 2024 Incremental Term Lender, the Facility Agent and the Obligors' Agent dated the 2024 Effective Date,

the New Lender shall, on the date upon which an assignment or transfer takes effect, pay to the Facility Agent (for its own account) a fee of \$3,500.

29.4 Limitation of Responsibility of Existing Lenders

(a) Unless expressly agreed to the contrary, an Existing Lender makes no representation or warranty and assumes no responsibility to a New Lender for:

- the legality, validity, effectiveness, adequacy or enforceability of the Transaction Documents, the Transaction Security or any other documents;
- (ii) the financial condition of any Obligor;
- (iii) the performance and observance by any Obligor of its obligations under the Transaction Documents or any other documents; or
- (iv) the accuracy of any statements (whether written or oral) made in or in connection with any Transaction Document or any other document,

and any representations or warranties implied by law are excluded.

- (b) Each New Lender confirms to the Existing Lender, the other Finance Parties and the Secured Parties that it:
 - (i) has made (and shall continue to make) its own independent investigation and assessment of the financial condition and affairs of each Obligor and its related entities in connection with its participation in this agreement and has not relied exclusively on any information provided to it by the Existing Lender or any other Finance Party in connection with any Transaction Document or the Transaction Security;
 - (ii) will continue to make its own independent appraisal of the creditworthiness of each Obligor and its related entities whilst any amount is or may be outstanding under the Finance Documents or any Commitment is in force; and
 - (iii) if all or any Loans or other rights transferred are rescheduled or renegotiated, the New Lender and not the Existing Lender will be subject to the rescheduled or renegotiated terms.
- (c) Nothing in any Finance Document obliges an Existing Lender to:
 - accept a re-transfer or re-assignment from a New Lender of any of the rights and obligations assigned or transferred under this clause 29; or
 - (ii) support any losses directly or indirectly incurred by the New Lender by reason of the nonperformance by any Obligor of its obligations under the Transaction Documents or otherwise.

29.5 **Procedure for Transfer**

- (a) Subject to the conditions set out in clause 29.2 (Conditions of Assignment or Transfer) a transfer is effected in accordance with paragraph (c) below when the Facility Agent executes an otherwise duly completed Transfer Certificate delivered to it by the Existing Lender and the New Lender. The Facility Agent shall, subject to paragraph (b) below, as soon as reasonably practicable after receipt by it of a duly completed Transfer Certificate appearing on its face to comply with the terms of this agreement and delivered in accordance with the terms of this agreement, execute that Transfer Certificate.
- (b) The Facility Agent shall only be obliged to execute a Transfer Certificate delivered to it by the Existing Lender and the New Lender upon its completion of all "know your customer" or similar checks relating to any person that it is required to carry out in relation to the transfer to such New Lender.
- (c) On the Transfer Date:

- to the extent the Transfer Certificate records a transfer of Loans which are outstanding on the Transfer Date, the transfer will take effect in relation to those Loans and all related rights under the Intercreditor Agreement and other Finance Documents by way of assignment;
- (ii) to the extent that the Transfer Certificate records a transfer of rights, benefits and obligations under the Finance Documents and in respect of the Transaction Security (in the case of rights, to the extent not assigned as contemplated by paragraph (i) above) each of the Obligors and the Existing Lender shall be released from further obligations towards one another under the Finance Documents and in respect of the Transaction Security and their respective rights against one another under the Finance Documents and in respect of the Transaction Security shall be cancelled (being the "Discharged Rights and Obligations");
- (iii) each of the Obligors and the New Lender shall assume obligations towards one another and/or acquire rights and benefits against one another which differ from the Discharged Rights and Obligations only insofar as that Obligor or other Group Company and the New Lender have assumed and/or acquired the same in place of that Obligor and the Existing Lender;
- (iv) the Facility Agent, the Arrangers, the Security Agent, the New Lender, the other Lenders, the Issuing Bank and any relevant Ancillary Lender shall acquire the same rights and assume the same obligations between themselves and in respect of the Transaction Security as they would have acquired and assumed had the New Lender been an Original Lender with the rights, and/or obligations acquired or assumed by it as a result of the transfer and to that extent the Facility Agent, the Arranger, the Security Agent, the Issuing Bank and any relevant Ancillary Lender and the Existing Lender shall each be released from further obligations to each other under the Finance Documents; and
- (v) the New Lender shall become a Party as a "Lender".
- (d) The provisions of paragraph (c)(i) above will not apply if:
 - (i) the transfer of Loans recorded in a Transfer Certificate would give rise to a stamp duty or other transfer tax; or
 - (ii) the New Lender elects by deleting paragraphs 3(b) and 9 from the Transfer Certificate and records in the Transfer Certificate that clause 29.5(c)(i) is not to apply to the transfer of Loans recorded in the Transfer Certificate,
 - (iii) and accordingly, the transfer of rights and entitlements in respect of Loans recorded in the Transfer Certificate will be made pursuant to paragraphs (c)(ii) to (v) (inclusive) above by way of novation.

29.6 Copy of Transfer Certificate to Obligors' Agent

The Facility Agent shall, either prior to executing a Transfer Certificate or as soon as reasonably practicable after it has executed a Transfer Certificate, send to Obligors' Agent a copy of that Transfer Certificate which Obligors' Agent shall promptly sign (if approved by the Obligors' Agent) by way of acknowledgement on behalf of all of the Obligors and return to the Facility Agent.

29.7 **Disclosure of Information**

(a) Each of the Finance Party agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its and its

Affiliates' directors, officers, employees and agents, including accountants, legal counsel and other advisors (it being understood that the persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential and the disclosing party will be responsible for any disclosure by such persons), (b) to the extent requested by any regulatory or self-regulatory authority have jurisdiction over such Finance Party, (c) to the extent required by applicable laws or regulations or by any subpoena or similar legal process, (d) to any other party to this agreement (so long as such person remains party to this agreement), (e) in connection with the exercise of any remedies hereunder or any other Finance Document or any suit, action or proceeding relating to this agreement or any other Finance Document or the enforcement of rights hereunder or thereunder, (f) subject to an agreement containing provisions substantially the same as those of this Clause, to (i) any assignee of or participant in, or any prospective assignee of or participant in, any of its rights or obligations under this agreement or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction, or to any credit insurance provider, in each case, relating to the Group and its obligations, (g) with the consent of the Obligors' Agent, (h) to the extent such Information (i) becomes publicly available other than as a result of a breach of this Clause or an agreement described in clause (f) hereof or (ii) becomes available to a Finance Party on a non-confidential basis from a source other than the Group Companies (other than a source actually known by such disclosing person to be bound by confidentiality obligations with respect thereto) or (i) to data service providers, including league table providers, that serve the lending industry (but, in the case of this clause (i), solely to the extent that (x) such Information is information about the terms of the financing routinely provided by arrangers to data services providers and (y) such Information is provided to such data service providers no earlier than the fifth Business Day after the 2022 Effective Date). In addition, each Finance Party shall comply with each of its other confidentiality arrangements applicable to any particular Information covered therein and furnished by such Finance Party in its capacity as a Finance Party in connection with this agreement (whether such confidentiality arrangements are agreed to before or after the 2022 Effective Date) unless the Parent otherwise consent to the disclosure of such Information to any person in connection with such person becoming a Lender or participant hereunder (such consent not to be unreasonably conditioned or withheld in the event that such person is willing to agree to enter into substantially similar confidentiality arrangements). "Information" means all information received from the Group Companies or their affiliates relating to the Group Companies, their Subsidiaries or their businesses. Notwithstanding anything herein to the contrary, in no event shall any Information be disclosed to any Disqualified Lender.

- (b) Each Finance Party acknowledges that information furnished to it pursuant to this agreement or the other Finance Documents may include material non-public information concerning the Group Companies and their affiliates and their related parties or their respective securities, and confirms that it has developed compliance procedures regarding the use of material non-public information and that it will handle such material non-public information in accordance with those procedures and applicable law, including United States Federal and state securities laws.
- (c) All information, including requests for waivers and amendments, furnished by the Group or the Facility Agent pursuant to, or in the course of administering, this agreement or the other Finance Documents will be syndicate-level information, which may contain material non-public information about the Group and their affiliates and their related parties or their respective securities. Accordingly, each Finance Party represents to the Group and the Facility Agent that either (x) it has identified in its administrative questionnaire a credit contact who may receive information that may contain material non-public information in accordance with its compliance procedures and applicable law, including United States Federal and state securities laws or (y)

that it does not want to receive any material non-public information about the Group and their affiliates and their related parties or their respective securities; provided that if such Financial Party elects not to receive any material non-public information referred to in clause (y), then such Finance Party agrees that the failure to receive such material non-public information shall not be the basis of any Default or Event of Default under this agreement or the Finance Documents

- For certain of the Lenders (each, a "Public Lender") that have personnel who do not wish to receive material non-public information with respect to the Group or its Affiliates, or the respective securities of any of the foregoing, the Obligors' Agent, at the request of the Facility Agent, hereby agrees that (w) it will, with respect to certain materials and/or information provided by or on behalf of the Group hereunder (the "Borrower Materials") that are to be distributed to the Public Lenders, clearly and conspicuously mark such Borrower Materials that do not contain material non-public information as "PUBLIC" which, at a minimum, shall mean that the word "PUBLIC" shall appear prominently on the first page thereof; (x) by marking Borrower Materials "PUBLIC," the Obligors' Agent shall be deemed to have authorized the Facility Agent, the Arrangers, the Issuing Banks and the Lenders to treat such Borrower Materials as not containing any material non-public information (although it may be sensitive and proprietary) with respect to the Group, its affiliates or any of their respective securities for purposes of United States Federal and state securities laws; (y) all Borrower Materials marked "PUBLIC" are permitted to be made available to Public Lenders in any electronic platform designated "Public Side Information"; and (z) the Facility Agent and the Arrangers shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" as only being suitable for non-Public Lenders. For the avoidance of doubt, whether any Borrower Materials contains material non-public information with respect to the Group or its Affiliates, or the respective securities of any of the foregoing, shall be solely determined by the Obligors' Agent in good faith; provided that the Finance Parties hereto agree that each Compliance Certificate to be delivered to the Facility Agent pursuant to clause 25.2 (Provision and Contents of Compliance Certificate) shall be considered material non-public information.
- (e) Each Lender, each Issuing Bank, the Facility Agent (for itself and not on behalf of any Lender) and the Security Agent hereby notifies each Obligor that, pursuant to the requirements of the USA PATRIOT Act, it is required to obtain, verify and record information that identifies such Obligor, which information includes the name and address of such Obligor and other information that will allow such Lender, such Issuing Bank, the Facility Agent or the Security Agent, as applicable, to identify such Obligor in accordance with the USA PATRIOT Act, and each Obligor agrees to provide such information from time to time as required by such Lender, such Issuing Bank, the Facility Agent and the Security Agent, as applicable, in order to comply with its continuing obligations under the USA PATRIOT Act.

29.8 Security over Lenders' Rights

In addition to the other rights provided to Lenders under this clause 29, each Lender may without consulting with or obtaining consent from any Obligor, at any time charge, assign or otherwise create Security in or over (whether by way of collateral or otherwise) all or any of its rights under any Finance Document to secure obligations of that Lender including, without limitation:

- (a) any charge, assignment or other Security to secure obligations to a federal reserve or central bank;and
- (b) in the case of any Lender which is a fund, any charge, assignment or other Security granted to any holders (or trustee or representatives of holders) of obligations owed, or securities issued, by that Lender as security for those obligations or securities,

except that no such charge, assignment or Security shall:

- release a Lender from any of its obligations under the Finance Documents or substitute the beneficiary of the relevant charge, assignment or Security for the Lender as a party to any of the Finance Documents; or
- (ii) require any payments to be made by an Obligor or grant to any person any more extensive rights than those required to be made or granted to the relevant Lender under the Finance Documents

29.9 Continuation of Security

- (a) Each Obligor consents to the assignments and transfers of rights and obligations permitted under and made in accordance with this clause 29. Each Obligor agrees and confirms that its guarantee and indemnity obligations under the Finance Documents and any Transaction Security granted by it in support of its own borrowing obligations or its guarantee or indemnity obligations under the Finance Documents will continue notwithstanding any transfer under this clause 29 and will extend to cover and support obligations owed to New Lenders and to continuing Finance Parties.
- (b) For the purposes of Article 1278 of the Luxembourg Civil Code, it is expressly agreed that the security created or evidenced by the Transaction Security Documents shall be preserved for the benefit of each New Lender and each Finance Party.
- (c) Obligors' Agent (for itself and as agent for the Obligors) will (at its own cost) promptly execute such documents and take such other actions as are necessary to effect or perfect a transfer of rights and/or obligations to a New Lender under the Finance Documents. Such action will include:
 - (i) promptly countersigning any approved Transfer Certificates (although any delay or failure by Obligors' Agent to so countersign a Transfer Certificate will not invalidate its operation); and
 - (ii) taking such steps as the Facility Agent or the Security Agent may request (including reexecution of Transaction Security Documents) for the purpose of ensuring that the New Lender has (and the other Finance Parties continue to have) the benefit of the same security interests under the Transaction Security Documents as existed immediately before the relevant transfer.

29.10 List of Lenders

At any time upon the request of Obligors' Agent, the Facility Agent shall provide a list of all of the Lenders.

29.11 Further Lenders

A person which is to be a Further Lender shall only become a party to this agreement as a Lender if it has (i) executed and delivered to the Facility Agent an Incremental Facility Commitment Letter and the Facility Agent has counter-signed the same which the Facility Agent agrees to do promptly, subject to completing any "know your customers" or similar checks under all applicable laws and regulations as contemplated in clause 2.2(r) (Incremental Facilities), which it is acknowledged may take longer than the three Business Days minimum period referred to in clause 2.2(k)(iii) (Incremental Facilities) above and (ii) acceded to the terms of the Intercreditor Agreement in accordance with the terms thereof.

29.12 Permitted Debt Purchase Transactions

- Notwithstanding anything in this Clause 29 (Changes to the Lenders) to the contrary, Revolving Facility Commitments, participations in Revolving Facility Loans and participations in Term Loans may be assigned to LMC or any of its subsidiaries (collectively, the "Affiliated Lenders") as long as (i) no Default or Event of Default shall have then occurred and be continuing, (ii) no proceeds from Revolving Loans are used, directly or indirectly, to fund any assignments of Term Facility A Loans, (iii) solely with respect to RCF Commitments, Facility RCF and participations in Term Facility A Loans, each of the Lenders that hold participations in Loans or commitments in the applicable such Facility as the participations in Loans or commitments being assigned has been offered the chance to participate in such assignment on a pro rata basis (in proportion to the participations in Loans or commitments in such Facility) pursuant to procedures reasonably satisfactory to the Facility Agent at the same price (based on a percentage of par); provided that, for the avoidance of doubt, with respect to any participations in a Term Facility B Loan or participations in any Incremental Term Loans, any Lender under such Facility may assign all or a portion of its rights and obligations with respect to the participations in Term Facility B Loans or Incremental Term Loans, as applicable, under this agreement on a pro-rata or non-pro rata basis (but to the extent effected through "Dutch auctions", so long as any offer to purchase or take by assignment by such Affiliated Lender shall have been made to all Term Lenders under the Term Facility), (iv) in the case of any assignment of RCF Commitments and/or participations in RCF Loans, such RCF Commitments and participations in RCF Loans shall be contributed to the Borrowers and permanently extinguished within one Business Day of such assignment and (v) no assignment may result in the Affiliated Lenders holding an outstanding principal amount of participations in Term Loans exceeding 25% of the principal amount of all Term Loans outstanding under this agreement at the time of such assignment, unless such excess is contributed to the relevant Borrower and permanently extinguished within one Business Day of such assignment. Affiliated Lenders will not be entitled to receive information provided solely to Lenders by the Facility Agent or any Lender, and may not attend or participate in Lender only meetings. Any participations in Loans or commitments acquired by any Affiliated Lender may (but shall not be required to) be contributed to the Group for purposes of cancellation; provided that upon such cancellation, the aggregate outstanding principal amount of the Loans or commitments shall be deemed reduced, as of the date of such contribution, by the full par value of the aggregate principal amount of the participations in Loans or commitments so contributed and cancelled, and each principal repayment instalment with respect to the Loans (if applicable) shall be reduced pro rata by the full par value of the aggregate principal amount of participations in Loans so contributed and cancelled.
- Each Affiliated Lender hereby agrees that, if a proceeding under any bankruptcy law shall be (b) commenced by or against any Group Company at a time when such Lender is an Affiliated Lender, such Affiliated Lender irrevocably authorizes and empowers the Facility Agent to vote on behalf of such Affiliated Lender with respect to the participations in Loans held by such Affiliated Lender in any manner in the Facility Agent's sole discretion, unless the Facility Agent instructs such Affiliated Lender to vote, in which case such Affiliated Lender shall vote with respect to the participations in Loans held by it as the Facility Agent directs; provided that (a) such Affiliated Lender shall be entitled to vote in accordance with its sole discretion (and not in accordance with the direction of the Facility Agent) and (b) the Facility Agent shall not be entitled to vote on behalf of such Affiliated Lender, in each case, in connection with any matter to the extent any such matter proposes to treat any obligation held by such Affiliated Lender in a manner that is different than the proposed treatment of similar obligations held by Lenders that are not affiliates of the Parent. Each Affiliated Lender hereby irrevocably appoints the Facility Agent (such appointment being coupled with an interest) as such Affiliated Lender's attorney-in-fact, with full authority in the place and stead of such Affiliated Lender and in the name of such Affiliated Lender (solely in respect of Loans and participations therein and not in

respect of any other claim or status such Affiliated Lender may otherwise have), from time to time in the Facility Agent's discretion to take any action and to execute any instrument that the Facility Agent may deem reasonably necessary to carry out the provisions of (but subject to the limitations set forth in) this paragraph.

(c) The Revolving Facility Commitments and any participations in Term Loans held by the Affiliated Lenders shall not be included in determining whether the Majority Lenders have taken or may take any action hereunder (including any consent to any amendment, waiver or other modification); provided that the provisions of this sentence shall not apply to any action requiring the consent of all Lenders or each affected Lender.

29.13 Certain Matters Related to Disqualified Lenders

The Facility Agent shall not be responsible or have any liability for, or have any duty to ascertain, inquire into, monitor or enforce, compliance with the provisions hereof relating to Disqualified Lenders. Without limiting the generality of the foregoing, the Facility Agent shall not (a) be obligated to ascertain, monitor or inquire as to whether any Lender or participant or prospective Lender or participant is a Disqualified Lender, Affiliated Lender or Net Short Lender or (b) have any liability with respect to or arising out of any assignment or participation of Loans or Commitments, or disclosure of confidential information, to any Disqualified Lender.

Notwithstanding anything in Clause 11 (Illegality, Voluntary Prepayment and Cancellation) to the contrary, if any assignment or sub-participation or sub-contract is made to a Disqualified Lender, the Borrowers shall have the right to (A) in the case of any outstanding Revolving Facility Commitments, terminate any Revolving Facility Commitment of such Disqualified Lender on a non-pro rata basis and repay all obligations of the Borrowers owing to such Disqualified Lender on a non-pro rata basis in connection with such Revolving Facility Commitment, (B) in the case of participations in outstanding Term Loans held by Disqualified Lenders, prepay such participations on a non-pro rata basis by paying the lesser of (x) the principal amount thereof and (y) the amount that such Disqualified Lender paid to acquire such Term Loans, in each case plus accrued interest, accrued fees and all other amounts (other than principal amounts) payable to it hereunder and under the other Finance Documents and/or (C) require such Disqualified Lender to assign and delegate, without recourse, all of its interest, rights and obligations under this agreement and related Finance Documents to a New Lender that shall assume such obligations at the lesser of (x) the principal amount thereof and (y) the amount that such Disqualified Lender paid to acquire such interests, rights and obligations, in each case plus accrued interest, accrued fees and all other amounts (other than principal amounts) payable to it hereunder and the other Finance Documents.

29.14 Disclosure to CUSIP Service Bureau

- (a) The Facility Agent may disclose to the CUSIP Service Bureau or any similar agency in connection with the issuance and monitoring of CUSIP numbers with respect to the Facilities evidenced by this agreement the following information:
 - (i) names of Obligors;
 - (ii) country of domicile of Obligors;
 - (iii) place of incorporation of Obligors;
 - (iv) date of this agreement;
 - (v) Clause 42 (Governing law);
 - (vi) the names of the Facility Agent and the Arranger;

- (vii) date of each amendment and restatement of this agreement;
- (viii) amounts of, and names of, the Facilities (and any tranches);
- (ix) amount of Total Commitments;
- (x) currencies of the Facilities;
- (xi) type of Facilities;
- (xii) ranking of Facilities;
- (xiii) Termination Date for the Facilities;
- (xiv) changes to any of the information previously supplied pursuant to paragraphs (i) to (xiii) above; and
- (xv) such other information agreed between the Facility Agent and the Parent.
- (b) The Parties acknowledge and agree that CUSIP numbers assigned to this agreement, the Facilities and/or one or more Obligors by the CUSIP Service Bureau or any similar agency and the information associated with each such CUSIP number may be disclosed to users of its services in accordance with the standard terms and conditions of the CUSIP Service Bureau or any similar agency.

30. CHANGES TO THE OBLIGORS

30.1 Assignment and Transfers by Obligors

No Obligor or any other Group Company may assign any of its rights or transfer any of its rights or obligations under the Finance Documents unless permitted by the terms of this agreement.

30.2 Additional Borrowers

- (a) Subject to compliance with the provisions of clause 25.8(c) and (d) ("Know Your Customer" Checks), Obligors' Agent may request that any wholly owned Subsidiary of the Parent which is not a Dormant Subsidiary becomes a Borrower. That Subsidiary shall become a Borrower if:
 - unless the proposed new Borrower is incorporated in and borrows from the same jurisdiction as an existing Borrower:
 - (A) in respect of a Term Facility, all the Lenders approve the addition of that Subsidiary; or
 - (B) in respect of a Revolving Facility, the proposed new Borrower is incorporated in a jurisdiction within OECD, it would not be unlawful for any Lender in respect of the relevant Revolving Facility to lend to a borrower in that jurisdiction and the RCF Lenders in respect of the relevant Revolving Facility approve the addition of that Subsidiary; or
 - (C) if either of the above conditions specified in sub-paragraph (B) above as to jurisdiction of incorporation or illegality, are not satisfied in respect of a Revolving Facility, all the Lenders in respect of the relevant Revolving Facility approve the addition of that Subsidiary:
 - (ii) Obligors' Agent and that Subsidiary deliver to the Facility Agent a duly completed and executed Accession Letter;

- (iii) the Subsidiary is (or becomes) a Guarantor prior to becoming a Borrower;
- (iv) Obligors' Agent confirms that no Default is continuing or would occur as a result of that Subsidiary becoming an Additional Borrower;
- (v) the proposed new Borrower is not a subsidiary of ASM; and
- (vi) the Facility Agent has received all of the documents and other evidence listed in part 3 of schedule 2 (Conditions Precedent) in relation to that Additional Borrower, each in form and substance satisfactory to the Facility Agent.
- (b) The Facility Agent shall notify the Obligors' Agent and the Lenders promptly upon being satisfied that it has received (in form and substance satisfactory to it) all the documents and other evidence listed in part 3 of schedule 2 (Conditions Precedent).

30.3 Resignation of a Borrower

- (a) In this clause 30.3, clause 30.5 (Resignation of a Guarantor) and clause 30.7 (Resignation and Release of Security on Disposal), "Third Party Disposal" means the disposal of an Obligor to a person which is not a Group Company where that disposal is permitted under clause 27.11 (Disposals) or made with the approval of the Majority Lenders.
- (b) If a Borrower is the subject of a Third Party Disposal or Permitted Group Company Disposal, Obligors' Agent may request that such Borrower ceases to be a Borrower by delivering to the Facility Agent a Resignation Letter.
- (c) The Facility Agent shall accept a Resignation Letter and notify Obligors' Agent and the other Finance Parties of its acceptance if:
 - (i) Obligors' Agent has confirmed that no Default is continuing or would result from the acceptance of the Resignation Letter;
 - (ii) the Borrower is under no actual or contingent obligations as a Borrower under any Finance Document;
 - (iii) where the Borrower is also a Guarantor (unless its resignation has been accepted in accordance with clause 30.5 (Resignation of a Guarantor)), its obligations in its capacity as Guarantor continue to be legal, valid, binding and enforceable and in full force and effect (subject to the Legal Reservations) and the amount guaranteed by it as a Guarantor is not decreased; and
 - (iv) the Obligors' Agent has confirmed that it shall ensure that any relevant Disposal Proceeds will be applied in accordance with clause 12.3 (Application of Mandatory Prepayments).
- (d) Upon notification by the Facility Agent to the Obligor Agents' of its acceptance of the resignation of a Borrower, that company shall cease to be a Borrower and shall have no further rights or obligations under the Finance Documents as a Borrower except that the resignation shall not take effect (and the Borrower will continue to have rights and obligations under the Finance Documents) until the date on which the Third Party Disposal or Permitted Group Company Disposal, as applicable takes effect.
- (e) The Facility Agent may, at the cost and expense of the Obligors, require a legal opinion from counsel to the Facility Agent confirming the matters set out in paragraph (c)(iii) above and the Facility Agent shall be under no obligation to accept a Resignation Letter until it has obtained such opinion in form and substance satisfactory to it.

30.4 Additional Guarantors

- (a) Subject to compliance with the provisions of clause 25.8(c) and 25.8(d) ("Know Your Customer" Checks), Obligors' Agent may request that any Subsidiary of the Parent becomes a Guarantor.
- (b) Subject to the Agreed Security Principles, the Parent shall procure that any other wholly owned Subsidiary of the Parent which is a Material Company shall, as soon as possible after becoming a Material Company, become an Additional Guarantor and, subject to the Agreed Security Principles, grant any Transaction Security as the Facility Agent may require and shall accede to the Intercreditor Agreement.
- (c) A Group Company shall become an Additional Guarantor if:
 - (i) it is not a subsidiary of ASM.:
 - the Obligors' Parent and the proposed Additional Guarantor deliver to the Facility Agent a duly completed and executed Accession Letter; and
 - (iii) the Facility Agent has received all of the documents and other evidence listed in part 3 of schedule 2 (Conditions Precedent) in relation to that Additional Guarantor, each in form and substance satisfactory to the Facility Agent.
- (d) The Facility Agent shall notify Obligors' Agent and the Lenders promptly upon being satisfied that it has received (in form and substance satisfactory to it) all the documents and other evidence listed in part 3 of schedule 2 (Conditions Precedent).

30.5 Resignation of a Guarantor

- (a) Obligors' Agent may request that a Guarantor (other than the Parent) ceases to be a Guarantor by delivering to the Facility Agent a Resignation Letter if:
 - that Guarantor is being disposed of by way of a Third Party Disposal (as defined in clause 30.3 (Resignation of a Borrower));
 - (ii) in the case only of a request that ASM resign as a Guarantor, Obligors' Agent delivers to the Facility Agent the minutes of an ASM shareholder meeting deciding the ASM Dissolution;
 - (iii) the Super Majority Lenders have consented to the resignation of that Guarantor; or
 - (iv) the Guarantor is proposed to be subject to a Permitted Group Company Disposal.
- (b) The Facility Agent shall accept a Resignation Letter and notify Obligors' Agent and the Lenders of its acceptance if:
 - (i) Obligors' Agent has confirmed that no Default is continuing or would result from the acceptance of the Resignation Letter;
 - (ii) no payment is due from the Guarantor under clause 23.1 (Guarantee and Indemnity);
 - (iii) where the Guarantor is also a Borrower, it is under no actual or contingent obligations as a Borrower and has resigned and ceased to be a Borrower under clause 30.3 (Resignation of a Borrower); and

- (iv) Obligors' Agent has confirmed that it shall ensure that the Disposal Proceeds (if any) will be applied in accordance with clause 12.3 (Application of Mandatory Prepayments).
- (c) The resignation of that Guarantor shall not be effective until:
 - (i) the date of the relevant Third Party Disposal;
 - (ii) in the case of the resignation of ASM in connection with the ASM Dissolution, the amendment of ASM's purposes clause (to the effect that ASM is "in liquidation") is published in the Swiss Official Gazette of Commerce; or
 - (iii) in the case of the resignation of a Group Company in connection with the Permitted Group Company Disposal, on the date that the relevant Resignation Letter is delivered to the Facility Agent provided that the conditions in paragraphs (a) and (b) above are satisfied,

at which time that company shall cease to be a Guarantor and shall have no further rights or obligations under the Finance Documents as a Guarantor.

30.6 Repetition of Representations

Delivery of an Accession Letter constitutes confirmation by the relevant Subsidiary that the representations and warranties referred to in clause 24.25(c) (Times when Representations Made) are true and correct in relation to it as at the date of delivery as if made by reference to the facts and circumstances then existing.

30.7 Resignation and Release of Security on Disposal

- (a) If an Obligor is or is proposed to be the subject of a Third Party Disposal then:
 - (i) where that Obligor created Transaction Security over any of its assets or business in favour of the Security Agent, or Transaction Security in favour of the Security Agent was created over the shares (or equivalent) of that Obligor, the Security Agent will, at the cost and request of the Obligors, release those assets, business or shares (or equivalent);
 - (ii) the resignation of that Obligor and related release of Transaction Security referred to in paragraph (a) above shall not become effective until the date of that disposal; and
 - (iii) if the disposal of that Obligor is not made, the Resignation Letter of that Obligor and the related release of Transaction Security referred to in paragraph (a) above shall have no effect and the obligations of that Obligor and the Transaction Security created or intended to be created by or over that Obligor shall continue in full force and effect.
- (b) If a Group Company disposes of any other asset to a person which is not a Group Company where that disposal is permitted under clause 27.11 (Disposals) or with the approval of the Majority Lenders the Security Agent will at the request and cost of the Obligors' Agent, release those assets from the Transaction Security and issue any certificate of non-crystallisation of any floating charge that may, in the absolute discretion of the Security Agent, be considered necessary or desirable.
- (c) If any Group Company is, or is proposed to be, subject to a Permitted Group Company Disposal:
 - (i) where that Group Company created Transaction Security over any of its assets or business in favour of the Security Agent, or Transaction Security in favour

- of the Security Agent was created over the shares (or equivalent) of that Group Company, the Security Agent will, at the cost and request of the Obligors' Agent, release those assets, business or shares (or equivalent); and
- (ii) the release of Transaction Security referred to in paragraph (i) above shall not become effective until immediately prior to the relevant Permitted Group Company Disposal (or any other prior date agreed between the Obligors' Agent and the Security Agent).

30.8 Further Security

- (a) Subject to the Agreed Security Principles, the Parent and Obligors' Agent shall procure that any Obligor which has not entered into a Transaction Security Document over any material assets which are of a type subject to Transaction Security granted by that Obligor in the relevant jurisdiction as at the Sixth Restatement Date (the **"Relevant Assets"**) in accordance with this agreement shall within thirty (30) Business Days after being required to do so by the Facility Agent execute a Transaction Security Document, subject always to the Agreed Security Principles over the Relevant Assets.
- (b) Without prejudice to the obligations of the Parent or Obligors' Agent under paragraph (a) above but subject to the Agreed Security Principles:
 - (i) if FIA confirms that all or any contracts with promoters, broadcasters or advertisers do not fall within the definition of Transfer Rights for the purpose of the 100 Year Agreements (in each case the "Excluded Contracts") the Parent will (subject always to the Agreed Security Principles and to the extent that such Group Companies have not already granted a floating charge that covers such rights) procure that the relevant Group Companies which have the benefit of the rights under such Excluded Contracts grant a floating charge to the Security Agent pursuant to a Transaction Security Document as soon as practicable (and in any event within 30 Business Days after it is established that such contract or contracts is or are Excluded Contracts);
 - (ii) if FIA confirms that the issued shares in the share capitals of each of FOA and FOL are not prohibited under the 100 Year Agreements from being subject to Security in favour of the Security Agent, the Parent will procure that the relevant Group Companies that own those shares shall charge them by way of fixed charge pursuant to a Transaction Security Document as soon as practicable (and in any event within 30 Business Days after it is established that such shares may be so charged).
- (c) When a Transaction Security Document is entered into under clause 30.8(a), the Obligors' Agent shall deliver to the Security Agent:
 - (i) the original Transaction Security Document executed by the relevant Group Company and the Security Agent and all other documents required to perfect the security in conformity with the Agreed Security Principles; and
 - (ii) where the relevant Group Company is not already an Obligor, all of the documents and other evidence listed in part 3 of schedule 2 (Conditions Precedent) in relation to the relevant Group Company,

each in form and substance satisfactory to the Security Agent (acting reasonably).

30.9 Permitted Transaction Releases

The Security Agent shall at the cost and expense of the Obligors, promptly take such actions as are reasonably requested by any Obligor to implement any Permitted Transaction Release.

31. ROLE OF THE FACILITY AGENT, THE SECURITY AGENT, THE ARRANGERS, THE ISSUING BANK AND OTHERS

31.1 Appointment of the Facility Agent and the Security Agent

- (a) Each of the Finance Parties appoints J.P. Morgan SE as the Facility Agent and Natwest Markets PLC as the Security Agent under and in connection with the Finance Documents.
- (b) Each of the Finance Parties authorises the Facility Agent and the Security Agent to exercise the rights, powers, authorities and discretions specifically given to the Facility Agent or the Security Agent (as applicable) under or in connection with the Finance Documents together with any other incidental rights, powers, authorities and discretions.
- (c) In relation to any Swiss law governed Transaction Security Documents under which security of an accessory (*akzessorische*) nature is granted each present and future Finance Party hereby authorises the Security Agent to act as its direct representative (*direkter Stellvertreter*).
- (d) Each Lender and the Issuing Bank exempts the Facility Agent from the restrictions pursuant to Section 181 Civil Code (Bürgerliches Gesetzbuch) and similar restrictions applicable to it pursuant to any other applicable law, in each case to the extent legally possible to such Lender and the Issuing Bank. Any Lender and the Issuing Bank which cannot grant such exemption shall notify the Facility Agent accordingly and, upon request of the Facility Agent, either act in accordance with the terms of this agreement and/or any other Finance Document as required pursuant to this agreement and/or such other Finance Document or grant a special power of attorney to a party acting on its behalf, in a manner that is not prohibited pursuant to Section 181 of the German Civil Code (Bürgerliches Gesetzbuch) and/or any other applicable laws.

31.2 Duties of the Facility Agent

- (a) The Facility Agent shall promptly forward to a Party the original or a copy of any document which is delivered to the Facility Agent for that Party by any other Party.
- (b) Except where a Finance Document specifically provides otherwise, the Facility Agent is not obliged to review or check the adequacy, accuracy or completeness of any document it forwards to another Party.
- (c) If the Facility Agent receives notice from a Party referring to this agreement, describing a Default and stating that the circumstance described is a Default, it shall promptly notify the other Finance Parties.
- (d) If the Facility Agent is aware of the non-payment of any principal, interest, commitment fee or other fee payable to a Finance Party (other than the Facility Agent, the Arrangers or the Security Agent) under this agreement it shall promptly notify the other Finance Parties.
- (e) The Facility Agent's duties under the Finance Documents are solely mechanical and administrative in nature
- (f) The Facility Agent shall notify the Parent if any amount due under this agreement is not identified as having been received by it.

31.3 Role of the Arranger

Except as specifically provided in the Finance Documents, the Arrangers have no obligations of any kind to any other Party under or in connection with any Finance Document.

31.4 No Fiduciary Duties

- (a) Nothing in this agreement constitutes the Facility Agent, the Arrangers and/or the Issuing Bank as a trustee or fiduciary of any other person.
- (b) None of the Facility Agent, the Security Agent, the Arrangers, the Issuing Bank or any Ancillary Lender shall be bound to account to any Lender for any sum or the profit element of any sum received by it for its own account.

31.5 Business with the Group

The Facility Agent, the Security Agent, the Arrangers, the Issuing Bank and each Ancillary Lender may accept deposits from, lend money to and generally engage in any kind of banking or other business with any Group Company.

31.6 Rights and Discretions

- (a) The Facility Agent and the Issuing Bank may rely on:
 - any representation, notice or document believed by it to be genuine, correct and appropriately authorised; and
 - (ii) any statement made by a director, authorised signatory or employee of any person regarding any matters which may reasonably be assumed to be within his knowledge or within his power to verify.
- (b) The Facility Agent may assume (unless it has received notice to the contrary in its capacity as agent for the Lenders) that:
 - no Default has occurred (unless it has actual knowledge of a Default arising under clause 28.1 (Non-Payment));
 - (ii) any right, power, authority or discretion vested in any Party or the Majority Lenders has not been exercised;
 - (iii) any notice or request made by Obligors' Agent (other than a Utilisation Request or Selection Notice) is made on behalf of and with the consent and knowledge of all the Obligors; and
 - (iv) any instructions received by it from the Majority Lenders, the Super Majority Lenders, any Lenders or any group of Lenders are duly given in accordance with the terms of the Finance Documents;
- (c) The Facility Agent may engage, pay for and rely on the advice or services of any lawyers, tax advisers, accountants, surveyors or other professional advisers or experts.
- (d) The Facility Agent may act in relation to the Finance Documents through its personnel and agents including its Affiliates provided that no greater amount shall be required to be paid by any Party as a result of any such action than would have been otherwise payable and each such Affiliate shall comply with Clause 18.9(k) and (l) as though it were the Facility Agent; provided further that the Facility Agent shall be primarily liable for any breach of the Finance Documents by (and any other acts performed by) its personnel and agents including its Affiliates.

- (e) The Facility Agent may disclose to any other Party any information it reasonably believes it has received as agent under this agreement.
- (f) Notwithstanding any other provision of any Finance Document to the contrary, none of the Facility Agent, the Arrangers or the Issuing Bank is obliged to do or omit to do anything if it would or might in its reasonable opinion constitute a breach of any law or regulation or a breach of a fiduciary duty or duty of confidentiality.
- (g) Without prejudice to the generality of paragraph (c) above, the Facility Agent may at any time engage and pay for the services of any lawyers to act as independent counsel to the Facility Agent (and so separate from any lawyers instructed by the Lenders) if the Facility Agent in its reasonable opinion deems this to be desirable.
- (h) The Facility Agent may rely on the advice or services of any lawyers, accountants, tax advisers, surveyors or other professional advisers or experts (whether obtained by the Facility Agent or by any other Party) and shall not be liable for any damages, costs or losses to any person, any diminution in value or any liability whatsoever arising as a result of its so relying unless directly caused by its gross negligence, bad faith or wilful misconduct as determined by a court of competent jurisdiction by final and nonappealable judgment.
- (i) Notwithstanding any provision of any Finance Document to the contrary, the Facility Agent is not obliged to expend or risk its own funds or otherwise incur any financial liability in the performance of its duties, obligations or responsibilities or the exercise of any right, power, authority or discretion if it has grounds for believing the repayment of such funds or adequate indemnity against, or security for, such risk or liability is not reasonably assured to it.

31.7 Majority Lenders' Instructions

- (a) Unless a contrary indication appears in a Finance Document, the Facility Agent shall:
 - exercise any right, power, authority or discretion vested in it as Facility Agent in accordance
 with any instructions given to it by the Majority Lenders (or, if so instructed by the Majority
 Lenders, refrain from exercising any right, power, authority or discretion vested in it as Facility
 Agent); and
 - (ii) not be liable for any act (or omission) if it acts (or refrains from taking any action) in accordance with an instruction of the Majority Lenders.
- (b) Unless a contrary indication appears in a Finance Document, any instructions given by the Majority Lenders will be binding on all the Finance Parties other than the Security Agent.
- (c) The Facility Agent may refrain from acting in accordance with the instructions of the Majority Lenders (or, if appropriate, the Lenders) until it has received such security as it may require for any cost, loss or liability (together with any associated VAT) which it may incur in complying with the instructions.
- (d) In the absence of instructions from the Majority Lenders, (or, if appropriate, the Lenders) the Facility Agent may act (or refrain from taking action) as it considers to be in the best interest of the Lenders.
- (e) The Facility Agent is not authorised to act on behalf of a Lender (without first obtaining that Lender's consent) in any legal or arbitration proceedings relating to any Finance Document. This paragraph (e) shall not apply to any legal or arbitration proceeding relating to the perfection, preservation or protection of rights under the Transaction Security Documents or enforcement of the Transaction Security or Transaction Security Documents.

(f) The participation of any Lender not responding to a request for instructions within the time period specified by the Facility Agent will be disregarded in assessing whether the Majority Lenders have given the relevant instruction.

31.8 Responsibility for Documentation

None of the Facility Agent, the Arrangers, the Issuing Bank or any Ancillary Lender:

- (a) is responsible or liable for the adequacy, accuracy and/or completeness of any information (whether oral or written) supplied by the Facility Agent, the Arrangers, the Issuing Bank, an Ancillary Lender, an Obligor or any other person given in or in connection with any Finance Document or the information memoranda or the reports prepared and delivered in connection therewith or the transactions contemplated in the Finance Documents, unless directly caused by its gross negligence, bad faith or wilful misconduct as determined by a court of competent jurisdiction by final and nonappealable judgment;
- (b) is responsible or liable for any determination as to whether any information provided or to be provided to any Finance Party is non-public information the use of which may be regulated or prohibited by applicable law or regulation relating to insider dealing or otherwise; or
- (c) is responsible for the legality, validity, effectiveness, adequacy or enforceability of any Finance Document or the Transaction Security or any other agreement, arrangement or document entered into, made or executed in anticipation of or in connection with any Finance Document or the Transaction Security.

31.9 Exclusion of Liability

- (a) Without limiting paragraph (b) below, none of the Facility Agent, the Issuing Bank, or any Ancillary Lender will be liable for:
 - (i) any damages, costs or losses to any person, any diminution in value, or any liability whatsoever arising as a result of taking or not taking any action taken by it under or in connection with any Finance Document or the Transaction Security, unless directly caused by its gross negligence, bad faith or wilful misconduct as determined by a court of competent jurisdiction by final and nonappealable judgment;
 - (ii) exercising, or not exercising, any right, power, authority or discretion given to it by, or in connection with, any Finance Document, the Transaction Security or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with, any Finance Document or the Transaction Security, unless directly caused by its gross negligence, bad faith or wilful misconduct as determined by a court of competent jurisdiction by final and nonappealable judgment;
 - (iii) without prejudice to the generality of paragraphs (i) and (ii) above, any damages, costs or losses to any person, any diminution in value or any liability whatsoever arising (but not including any claim based on the fraud of the Facility Agent, Issuing Bank or any Ancillary Lender, as applicable) as a result of:
 - (A) any act, event or circumstance not reasonably within its control; or
 - (B) the general risks of investment in, or the holding of assets in, any jurisdiction,

including (in each case and without limitation) such damages, costs, losses, diminution in value or liability arising as a result of: nationalisation, expropriation or other governmental actions; any regulation, currency restriction, devaluation or fluctuation; market conditions affecting the execution or settlement of transactions or the value of assets (including any Disruption Event); breakdown, failure or malfunction of any third party transport, telecommunications, computer services or systems; natural disasters or acts of God; war, terrorism, insurrection or revolution; or strikes or industrial action.

- (b) No Finance Party (other than the Facility Agent, the Issuing Bank or an Ancillary Lender (as applicable)) may take any proceedings against any officer, employee or agent of the Facility Agent, the Issuing Bank or any Ancillary Lender, in respect of any claim it might have against the Facility Agent, the Issuing Bank or an Ancillary Lender or in respect of any act or omission of any kind by that officer, employee or agent in relation to any Finance Document or any Transaction Document and any officer, employee or agent of the Facility Agent, the Issuing Bank, the Security Agent or any Ancillary Lender may rely on this clause subject to clause 1.3 (Third Party Rights) and the provisions of the Third Parties Act.
- (c) The Facility Agent will not be liable for any delay (or any related consequences) in crediting an account with an amount required under the Finance Documents to be paid by the Facility Agent if the Facility Agent has taken all necessary steps as soon as reasonably practicable to comply with the regulations or operating procedures of any recognised clearing or settlement system used by the Facility Agent for that purpose.
- (d) Nothing in this agreement shall oblige the Facility Agent or the Arrangers to carry out any "know your customer" or other checks in relation to any person on behalf of any Lender and each Lender confirms to the Facility Agent and the Arrangers that it is solely responsible for any such checks it is required to carry out and that it may not rely on any statement in relation to such checks made by the Facility Agent or the Arrangers.
- (e) Without prejudice to any provision of any Finance Document excluding or limiting the Facility Agent's liability, any liability of the Facility Agent arising under or in connection with any Finance Document or the Transaction Security shall be limited to the amount of actual loss which has been finally judicially determined to have been suffered (as determined by reference to the date of default of the Facility Agent or, if later, the date on which the loss arises as a result of such default) but without reference to any special conditions or circumstances known to the Facility Agent at any time which increase the amount of that loss. In no event shall the Facility Agent be liable for any loss of profits, goodwill, reputation, business opportunity or anticipated saving, or for special, punitive, indirect or consequential damages, whether or not the Facility Agent has been advised of the possibility of such loss or damages.

31.10 Lenders' Indemnity to the Facility Agent

Each Lender shall (in proportion to its share of the Total Commitments or, if the Total Commitments are then zero, to its share of the Total Commitments immediately prior to their reduction to zero) indemnify the Facility Agent, within three Business Days of demand, against any cost, loss or liability (including, without limitation, for negligence or any other category of liability whatsoever) incurred by the Facility Agent (otherwise than by reason of the Facility Agent's gross negligence, bad faith or wilful misconduct as determined by a court of competent jurisdiction by final and nonappealable judgment).

- (b) Subject to paragraphs (c) and (d) below and solely to the extent required pursuant to clause 20.2 (Other Indemnities), the Obligors' Agent shall (or shall procure that an Obligor will) reimburse any Lender for any payment that Lender makes to the Facility Agent pursuant to paragraph (a) above.
- (c) Paragraph (b) above shall not apply to the extent that (x) the indemnity payment in respect of which the Lender claims reimbursement relates to a liability of the Facility Agent to an Obligor or (y) arise from a dispute solely between a Lender and the Facility Agent.
- (d) Notwithstanding anything in this Clause 31.10 to the contrary, the Group Companies' indemnity obligations under this Clause 31.10(b) shall only cover those costs that are reasonable and documented and shall be limited by Clause 20.5 (Indemnity Costs Limitations).

31.11 Resignation of the Facility Agent

- (a) The Facility Agent may resign and appoint one of its Affiliates acting through an office in the European Union, the United Kingdom or the United States as successor by giving notice to the Lenders and the Parent.
- (b) Alternatively the Facility Agent may resign by giving notice to the Lenders and the Parent, in which case the Majority Lenders (after consultation with the Parent) may appoint a successor Facility Agent.
- (c) If the Majority Lenders have not appointed a successor Facility Agent in accordance with paragraph (b) above within 30 days after notice of resignation was given, the Facility Agent (after consultation with the Parent) may appoint a successor Facility Agent (acting through an office in the United Kingdom) provided that if the Facility Agent, as applicable, shall notify the Borrower and the Lenders that no successor agent has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice:
 - (i) the resigning Facility Agent shall be discharged from its duties and obligations hereunder and under the other Finance Documents; and
 - (ii) all payments, communications and determinations provided to be made by, to or through the Facility Agent shall instead be made by or to each Finance Party directly, until such time as the Majority Lenders appoint a successor Facility Agent as provided for in this clause 31; a Party which has made a payment in accordance with paragraph (c)(ii) shall be discharged of the relevant payment obligation under the Finance Documents.
- (d) The retiring Facility Agent shall, at its own cost, make available to the successor Facility Agent such documents and records and provide such assistance as the successor Facility Agent may reasonably request for the purposes of performing its functions as Facility Agent under the Finance Documents.
- (e) Upon the appointment of a successor or a resignation taking effect pursuant to paragraph (c) above, the retiring Facility Agent shall be discharged from any further obligation in respect of the Finance Documents but shall remain entitled to the benefit of this clause 31. Its successor and each of the other Parties shall have the same rights and obligations amongst themselves as they would have had if such successor had been an original Party.
- (f) After consultation with the Obligors' Agent, the Majority Lenders may, by notice to the Facility Agent, require it to resign in accordance with paragraph (b) above. In this event, the Facility Agent shall resign in accordance with paragraph (b) above.

- (g) The Facility Agent shall resign in accordance with paragraph (b) above (and, to the extent applicable, shall appoint a successor agent pursuant to paragraph (c) above) if on or after the date which is six months before the FATCA Application Date relating to any payment to the Facility Agent:
 - the Facility Agent fails to respond to a request under clause 18.9 (Tax Documentation) and the Company or a Lender reasonably believes that the Facility Agent will not be a FATCA Compliant Party on or at any material time after that FATCA Application Date;
 - (ii) the information supplied by the Facility Agent pursuant to clause 18.9 (Tax Documentation) indicates that the Facility Agent will not be a FATCA Compliant Party on or at any material time after that FATCA Application Date; or
 - (iii) the Facility Agent notifies the Company or a Lender that, as a result of the Facility Agent's status under FATCA, the Facility Agent will not be a FATCA Compliant Party on or at any material time after that FATCA Application Date,

and the Obligors' Agent or that Lender, by notice to the Facility Agent, requires it to resign.

31.12 Confidentiality

- (a) In acting as agent for the Finance Parties, the Facility Agent shall be regarded as acting through its agency division which shall be treated as a separate entity from any other of its divisions or departments.
- (b) If information is received by another division or department of the Facility Agent, it may be treated as confidential to that division or department and the Facility Agent shall not be deemed to have notice of it.
- (c) Notwithstanding any other provision of any Finance Document to the contrary, neither the Facility Agent nor the Arrangers are obliged to disclose to any other person:
 - (i) any confidential information; or
 - (ii) any other information

if the disclosure would or might in its reasonable opinion constitute a breach of any law or a breach of a fiduciary duty.

31.13 Relationship with the Lenders

- (a) The Facility Agent may treat each Lender as a Lender, entitled to payments under this agreement and acting through its Facility Office unless it has received not less than five Business Days prior notice from that Lender to the contrary in accordance with the terms of this agreement.
- (b) Each Lender shall supply the Facility Agent with any information that the Security Agent may reasonably specify (through the Facility Agent) as being necessary or desirable to enable the Security Agent to perform its functions as Security Agent. Each Lender shall deal with the Security Agent exclusively through the Facility Agent and shall not deal directly with the Security Agent.

31.14 Credit Appraisal by the Lenders, Issuing Bank and Ancillary Lenders

Without affecting the responsibility of any Obligor for information supplied by it or on its behalf in connection with any Finance Document, each Lender, Issuing Bank and Ancillary Lender confirms to the Facility Agent, the Arrangers, the Issuing Bank and each Ancillary Lender that it has been, and will continue to be, solely responsible for making its own independent appraisal and investigation of all risks arising under or in connection with any Finance Document including but not limited to:

- (a) the financial condition, status and nature of each Group Company;
- (b) the legality, validity, effectiveness, adequacy or enforceability of any Finance Document and the Transaction Security and any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document or the Transaction Security;
- (c) whether that Secured Party has recourse, and the nature and extent of that recourse, against any Party or any of its respective assets under or in connection with any Finance Document, the Transaction Security, the transactions contemplated by the Finance Documents or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document;
- (d) the adequacy, accuracy and/or completeness of any information provided by the Facility Agent, any Party or by any other person under or in connection with any Finance Document, the transactions contemplated by the Finance Documents or any information memoranda or reports prepared and delivered in connection therewith or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document; and
- (e) the right or title of any person in or to, or the value or sufficiency of any part of the Charged Property, the priority of any of the Transaction Security or the existence of any Security affecting the Charged Property.

31.15 Amounts paid in error

- (a) If the Facility Agent pays an amount to another Finance Party and the Facility Agent notifies that Finance Party that such payment was an Erroneous Payment then the Finance Party to whom that amount was paid by the Facility Agent shall on demand refund the same to the Facility Agent together with interest on that amount from the date of payment to the date of receipt by the Facility Agent, calculated by the Facility Agent to reflect its cost of funds.
- (b) Neither:
 - (i) the obligations of any Lender to the Facility Agent; nor
 - (ii) the remedies of the Facility Agent,

(whether arising under this Clause 31.15 or otherwise) which relate to an Erroneous Payment will be affected by any act, omission, matter or thing which, but for this paragraph (b), would reduce, release or prejudice any such obligation or remedy (whether or not known by the Facility Agent or any Lender).

(c) All payments to be made by a Lender to the Facility Agent (whether made pursuant to this Clause 31.15 or otherwise) which relate to an Erroneous Payment shall be calculated and be made without (and free and clear of any deduction for) set-off or counterclaim.

(d) In this agreement, "Erroneous Payment" means a payment of an amount by the Facility Agent to the Lender which the Facility Agent determines (in its sole discretion) was made in error.

31.16 Deduction from Amounts Payable by the Facility Agent

If any Finance Party owes an amount to the Facility Agent under the Finance Documents the Facility Agent may, after giving notice to that Finance Party, deduct an amount not exceeding that amount from any payment to that Finance Party which the Facility Agent would otherwise be obliged to make under the Finance Documents and apply the amount deducted in or towards satisfaction of the amount owed. For the purposes of the Finance Documents that Finance Party shall be regarded as having received any amount so deducted.

31.17 Reliance and Engagement Letters

Each Finance Party and Secured Party confirms that each of the Arrangers and the Facility Agent has authority to accept on its behalf (and ratifies the acceptance on its behalf of any letters or reports already accepted by the Arrangers or Facility Agent) the terms of any reliance letter or engagement letters relating to any reports or letters provided by accountants in connection with the Finance Documents or the transactions contemplated in the Finance Documents (including any net asset letter in connection with the financial assistance procedures) and to bind it in respect of those reports or letters and to sign such letters on its behalf and further confirms that it accepts the terms and qualifications set out in such letters.

31.18 Intercreditor Agreement

Each Lender that receives an assignment or transfer of any rights or obligations pursuant to clause 29 (Changes to the Lenders) shall, to the extent it has not already done so, accede to the Intercreditor Agreement by delivery to the Security Agent of a duly completed accession undertaking in the form required under the Intercreditor Agreement.

31.19 No duty to monitor

The Facility Agent shall not be bound to enquire:

- (a) whether or not any Default has occurred;
- (b) as to the performance, default or any breach by any Party of its obligations under any Finance Document: or
- (c) whether any other event specified in any Finance Document has occurred.

32. CONDUCT OF BUSINESS BY THE FINANCE PARTIES

No provision of this agreement will:

- (a) interfere with the right of any Finance Party to arrange its affairs (tax or otherwise) in whatever manner it thinks fit;
- (b) oblige any Finance Party to investigate or claim any credit, relief, remission or repayment available to it or the extent, order and manner of any claim; or
- (c) oblige any Finance Party to disclose any information relating to its affairs (tax or otherwise) or any computations in respect of Tax.

33. SHARING AMONG THE FINANCE PARTIES

33.1 Payments to Finance Parties

If a Finance Party (a "Recovering Finance Party") receives or recovers any amount from an Obligor other than in accordance with clause 34 (Payment Mechanics) and applies that amount to a payment due under the Finance Documents then:

- (a) the Recovering Finance Party shall, within three Business Days, notify details of the receipt or recovery, to the Facility Agent;
- (b) the Facility Agent shall determine whether the receipt or recovery is in excess of the amount the Recovering Finance Party would have been paid had the receipt or recovery been received or made by the Facility Agent and distributed in accordance with clause 34 (Payment Mechanics), without taking account of any Tax which would be imposed on the Facility Agent in relation to the receipt, recovery or distribution; and
- (c) the Recovering Finance Party shall, within three Business Days of demand by the Facility Agent, pay to the Facility Agent an amount (the "Sharing Payment") equal to such receipt or recovery less any amount which the Facility Agent determines may be retained by the Recovering Finance Party as its share of any payment to be made, in accordance with clause 34.5 (Partial Payments).

33.2 Redistribution of Payments

The Facility Agent shall treat the Sharing Payment as if it had been paid by the relevant Obligor and distribute it between the Finance Parties (other than the Recovering Finance Party) in accordance with clause 34.5 (Partial Payments).

33.3 Recovering Finance Party's Rights

- (a) On a distribution by the Facility Agent under clause 33.2 (Redistribution of Payments), the Recovering Finance Party will be subrogated to the rights of the Finance Parties which have shared in the redistribution.
- (b) If and to the extent that the Recovering Finance Party is not able to rely on its rights under paragraph (a) above, the relevant Obligor shall be liable to the Recovering Finance Party for a debt equal to the Sharing Payment which is immediately due and payable.

33.4 Reversal of Redistribution

If any part of the Sharing Payment received or recovered by a Recovering Finance Party becomes repayable and is repaid by that Recovering Finance Party, then:

- (a) each Finance Party which has received a share of the relevant Sharing Payment pursuant to clause 33.2 (Redistribution of Payments) shall, upon request of the Facility Agent, pay to the Facility Agent for the account of that Recovering Finance Party an amount equal to the appropriate part of its share of the Sharing Payment (together with an amount as is necessary to reimburse that Recovering Finance Party for its proportion of any interest on the Sharing Payment which that Recovering Finance Party is required to pay); and
- (b) that Recovering Finance Party's rights of subrogation in respect of any reimbursement shall be cancelled and the relevant Obligor will be liable to the reimbursing Finance Party for the amount so reimbursed.

33.5 Exceptions

- (a) This clause 32 shall not apply to the extent that the Recovering Finance Party would not, after making any payment pursuant to this clause, have a valid and enforceable claim against the relevant Obligor.
- (b) A Recovering Finance Party is not obliged to share with any other Finance Party any amount which the Recovering Finance Party has received or recovered as a result of taking legal or arbitration proceedings, if:
 - (i) it notified the other Finance Party of the legal or arbitration proceedings; and
 - (ii) the other Finance Party had an opportunity to participate in those legal or arbitration proceedings but did not do so as soon as reasonably practicable having received notice and did not take separate legal or arbitration proceedings.

33.6 Ancillary Lenders

- (a) This clause 33 shall not apply to any receipt or recovery by a Lender in its capacity as an Ancillary Lender at any time prior to service of notice under clause 28.22 (Acceleration).
- (b) Following service of notice under clause 28.22 (Acceleration), this clause 33 shall apply to all receipts or recoveries by Ancillary Lenders except to the extent that the receipt or recovery represents a reduction from the Designated Gross Amount for an Ancillary Facility to its Designated Net Amount.

34. PAYMENT MECHANICS

34.1 Payments to the Facility Agent

- (a) On each date on which an Obligor or a Lender is required to make a payment under a Finance Document excluding a payment under the terms of an Ancillary Document, that Obligor or Lender shall make the same available to the Facility Agent (unless a contrary indication appears in a Finance Document) for value on the due date at the time and in such funds specified by the Facility Agent as being customary at the time for settlement of transactions in the relevant currency in the place of payment.
- (b) Payment shall be made to such account in the principal financial centre of the country of that currency (or, in relation to euro, in a principal financial centre in a Participating Member State or London) with such bank as the Facility Agent specifies.

34.2 **Distributions by the Facility Agent**

Each payment received by the Facility Agent under the Finance Documents for another Party shall, subject to clause 34.3 (Distributions to an Obligor) and clause 34.4 (Clawback) be made available by the Facility Agent as soon as practicable after receipt to the Party entitled to receive payment in accordance with this agreement (in the case of a Lender, for the account of its Facility Office), to such account as that Party may notify to the Facility Agent by not less than five Business Days' notice with a bank in the principal financial centre of the country of that currency (or, in relation to euro, in the principal financial centre of a Participating Member State or London).

34.3 **Distributions to an Obligor**

The Facility Agent may (with the consent of the Obligor or in accordance with clause 35 (Set-Off)) apply any amount received by it for that Obligor in or towards payment (on the date and in the currency and funds of receipt) of any amount due from that Obligor under the Finance Documents or in or towards purchase of any amount of any currency to be so applied.

34.4 Clawback

- (a) Where a sum is to be paid to the Facility Agent under the Finance Documents for another Party, the Facility Agent is not obliged to pay that sum to that other Party (or to enter into or perform any related exchange contract) until it has been able to establish to its satisfaction that it has actually received that sum.
- (b) If the Facility Agent pays an amount to another Party and it proves to be the case that the Facility Agent had not actually received that amount, then the Party to whom that amount (or the proceeds of any related exchange contract) was paid by the Facility Agent shall on demand refund the same to the Facility Agent together with interest on that amount from the date of payment to the date of receipt by the Facility Agent, calculated by the Facility Agent to reflect its reasonable cost of funds.

34.5 Partial Payments

- (a) If the Facility Agent receives a payment for application against amounts due in respect of any Finance Documents that is insufficient to discharge all the amounts then due and payable by an Obligor under those Finance Documents, the Facility Agent shall apply that payment towards the obligations of that Obligor under those Finance Documents in the following order:
 - (i) first, in or towards payment pro rata of any unpaid fees, costs and expenses of the Facility Agent, the Issuing Bank and the Security Agent under those Finance Documents;
 - (ii) **secondly**, in or towards payment pro rata of any accrued interest, fee or commission due but unpaid under those Finance Documents;
 - (iii) thirdly, in or towards payment pro rata of any principal due but unpaid under those Finance Documents and any amount due but unpaid under clause 7.2 (Claims under a Letter of Credit) and clause 7.3 (Indemnities); and
 - (iv) fourthly, in or towards payment pro rata of any other sum due but unpaid under the Finance Documents.
- (b) Paragraph (a) above will override any appropriation made by an Obligor.

34.6 No Set-Off by Obligors

All payments to be made by an Obligor under the Finance Documents shall be calculated and be made without (and free and clear of any deduction for) set-off or counterclaim.

34.7 **Business Days**

- (a) Any payment which is due to be made on a day that is not a Business Day shall be made on the next Business Day in the same calendar month (if there is one) or the preceding Business Day (if there is not).
- (b) During any extension of the due date for payment of any principal or Unpaid Sum under this agreement interest is payable on the principal or Unpaid Sum at the rate payable on the original due date.

34.8 Currency of Account

(a) Subject to paragraphs (b) to (e) below, the Base Currency is the currency of account and payment for any sum due from an Obligor under any Finance Document.

- (b) A repayment or prepayment of a Utilisation or Unpaid Sum or a part of a Utilisation or Unpaid Sum shall be made in the currency in which that Utilisation or Unpaid Sum is denominated on its due date.
- (c) Each payment of interest shall be made in the currency in which the sum in respect of which the interest is payable was denominated when that interest accrued.
- (d) Each payment in respect of costs, expenses or Taxes shall be made in the currency in which the costs, expenses or Taxes are incurred.
- (e) Any amount expressed to be payable in a currency other than the Base Currency shall be paid in that other currency.

34.9 Change of Currency

- (a) Unless otherwise prohibited by law, if more than one currency or currency unit are at the same time recognised by the central bank of any country as the lawful currency of that country, then:
 - any reference in the Finance Documents to, and any obligations arising under the Finance Documents in, the currency of that country shall be translated into, or paid in, the currency or currency unit of that country designated by the Facility Agent (after consultation with the Obligors' Agent); and
 - (ii) any translation from one currency or currency unit to another shall be at the official rate of exchange recognised by the central bank for the conversion of that currency or currency unit into the other, rounded up or down by the Facility Agent (acting reasonably).
- (b) If a change in any currency of a country occurs, this agreement will, to the extent the Facility Agent (acting reasonably and after consultation with the Obligors' Agent) specifies to be necessary, be amended to comply with any generally accepted conventions and market practice in the Relevant Interbank Market and otherwise to reflect the change in currency.

35. **SET-OFF**

- 35.1 A Finance Party may set off any matured obligation due from an Obligor under the Finance Documents (to the extent beneficially owned by that Finance Party) against any matured obligation owed by that Finance Party to that Obligor, regardless of the place of payment, booking branch or currency of either obligation. If the obligations are in different currencies, the Finance Party may convert either obligation at a market rate of exchange in its usual course of business for the purpose of the set-off.
- 35.2 Any credit balances taken into account by an Ancillary Lender when operating a net limit in respect of any overdraft under an Ancillary Facility shall on enforcement of the Finance Documents be applied first in reduction of the overdraft provided under that Ancillary Facility in accordance with its terms.

36. **NOTICES**

36.1 Communications in Writing

Any communication to be made under or in connection with the Finance Documents shall be made in writing and, unless otherwise stated, may be made by fax or letter.

36.2 Addresses

The address, fax number and electronic address (and the department or officer, if any, for whose attention the communication is to be made) of each Party for any communication or document to be made or delivered under or in connection with the Finance Documents is:

- (a) in the case of Obligors' Agent or each Group Company, the contact information set forth on Schedule 15 part 1;
- (b) in the case of each Lender, the Issuing Bank, each Ancillary Lender or any other Obligor, that notified in writing to the Facility Agent on or prior to the date on which it becomes a Party; and
- (c) in the case of the Facility Agent or the Security Agent, that contact information set forth on Schedule 15 part 2,

or any substitute address, fax number, electronic communication or department or officer as the Party may notify to the Facility Agent (or the Facility Agent may notify to the other Parties, if a change is made by the Facility Agent) by not less than five Business Days' notice.

36.3 **Delivery**

- (a) Any communication or document made or delivered by one person to another under or in connection with the Finance Documents will only be effective:
 - (i) if by way of fax, when received in legible form;
 - (ii) if by way of letter, when it has been left at the relevant address or five Business Days after being deposited in the post postage prepaid in an envelope addressed to it at that address; or
 - (iii) if by electronic communications, upon confirmation of receipt (which may be means of a read receipt),

and, if a particular department or officer is specified as part of its address details provided under clause 36.2 (Addresses), if addressed to that department or officer.

- (b) Any communication or document to be made or delivered to the Facility Agent or the Security Agent will be effective only when actually received by the Facility Agent or Security Agent during its normal business (or, if received other than during its normal business hours, upon the commencement of its normal business hours) and then only if it is expressly marked for the attention of the department or officer identified with the Facility Agent's or Security Agent's signature below (or any substitute department or officer as the Facility Agent or Security Agent shall specify for this purpose).
- (c) All notices from or to an Obligor shall be sent through the Facility Agent.
- (d) Any communication or document made or delivered to Obligors' Agent in accordance with this clause 36.3 will be deemed to have been made or delivered to each of the Obligors.

36.4 Notification of Address and Fax Number

Promptly upon receipt of notification of an address, fax number or electronic address or change of address or change of electronic address or fax number pursuant to clause 36.2 (Addresses) or changing its own address or fax number or electronic address, the Facility Agent shall notify the other Parties.

36.5 Electronic Communication

- (a) Any communication to be made between the Facility Agent or the Security Agent and a Lender under or in connection with the Finance Documents may be made by electronic mail or other electronic means, if the Facility Agent, the Security Agent and the relevant Lender:
 - (i) agree that, unless and until notified to the contrary, this is to be an accepted form of communication;
 - (ii) notify each other in writing of their electronic mail address and/or any other information required to enable the sending and receipt of information by that means; and
- (b) Any electronic communication made between the Facility Agent and a Lender or the Security Agent will be effective only when actually received during its normal business (or, if received other than during its normal business hours, upon the commencement of its normal business hours) in readable form and in the case of any electronic communication made by a Lender to the Facility Agent or the Security Agent only if it is addressed in such a manner as the Facility Agent or Security Agent shall specify for this purpose.

36.6 Use of Websites

- (a) The Parent or Obligors' Agent may satisfy their obligations under this agreement to deliver any information in relation to those Lenders (the "Website Lenders") who accept this method of communication by posting this information onto an electronic website designated by the Parent and the Facility Agent (the "Designated Website") if:
 - the Facility Agent expressly agrees (after consultation with each of the Lenders) that it will accept communication of the information by this method;
 - (ii) both the Parent or Obligors' Agent (as the case may be) and the Facility Agent are aware of the address of and any relevant password specifications for the Designated Website; and
 - (iii) the information is in a format previously agreed between the Parent or Obligors' Agent and the Facility Agent.

If any Lender (a **"Paper Form Lender"**) does not agree to the delivery of information electronically then the Facility Agent shall notify the Parent or Obligors' Agent (as the case may be) accordingly and the Parent or Obligors' Agent (as the case may be) shall at its own cost supply the information to the Facility Agent in paper form.

- (b) The Facility Agent shall supply each Website Lender with the address of and any relevant password specifications for the Designated Website following designation of that website by the Parent or Obligors' Agent and the Facility Agent.
- (c) The Parent or Obligors' Agent shall promptly upon becoming aware of its occurrence notify the Facility Agent if:
 - (i) the Designated Website cannot be accessed due to technical failure;
 - (ii) the password specifications for the Designated Website change;

- any new information which is required to be provided under this agreement is posted onto the Designated Website;
- (iv) any existing information which has been provided under this agreement and posted onto the Designated Website is amended; or
- (v) the Parent or Obligors' Agent becomes aware that the Designated Website or any information posted onto the Designated Website is or has been infected by any electronic virus or similar software.

If the Parent or Obligors' Agent notifies the Facility Agent under paragraph (c)(i) or paragraph (c)(v) above, all information to be provided by the Parent or Obligors' Agent under this agreement after the date of that notice shall be supplied in paper form unless and until the Facility Agent and each Website Lender is satisfied that the circumstances giving rise to the notification are no longer continuing.

(d) Any Website Lender may request, through the Facility Agent, one paper copy of any information required to be provided under this agreement which is posted onto the Designated Website. The Parent or Obligors' Agent (as the case may be) shall at its own cost comply with any such request within 30 Business Days.

36.7 English Language

- (a) Any notice given under or in connection with any Finance Document must be in English.
- (b) All other documents provided under or in connection with any Finance Document must be:
 - (i) in English; or
 - (ii) if not in English, and if so reasonably required by the Facility Agent, accompanied by a certified English translation and, in this case, the English translation will prevail unless the document is a constitutional, statutory or other official document.

36.8 [Reserved]

37. CALCULATIONS AND CERTIFICATES

37.1 Accounts

In any litigation or arbitration proceedings arising out of or in connection with a Finance Document, the entries made in the accounts maintained by a Finance Party are prima facie evidence of the matters to which they relate.

37.2 Certificates and Determinations

Any certification or determination by a Finance Party of a rate or amount under any Finance Document is, in the absence of manifest error, conclusive evidence of the matters to which it relates.

37.3 Day Count Convention

Any interest, commission or fee accruing under a Finance Document will accrue from day to day and is calculated on the basis of the actual number of days elapsed and a year of 360 days or, in any case where the practice in the Relevant Interbank Market differs, in accordance with that market practice; provided that interest computed by reference to the

Alternate Base Rate at times when the Alternate Base Rate is based on the Prime Rate shall be computed on the basis of a year of 365 days (or 366 days in a leap year).

37.4 Receipt of Payments

The Facility Agent shall notify Obligors' Agent if any payment due under this agreement is not identified as having been received as soon as reasonably practicable after becoming aware of the relevant circumstances.

38. PARTIAL INVALIDITY

If, at any time, any provision of the Finance Documents is or becomes illegal, invalid or unenforceable in any respect under any law of any jurisdiction, neither the legality, validity or enforceability of the remaining provisions nor the legality, validity or enforceability of such provision under the law of any other jurisdiction will in any way be affected or impaired.

39. **REMEDIES AND WAIVERS**

No failure to exercise, nor any delay in exercising, on the part of any Finance Party or Secured Party, any right or remedy under the Finance Documents shall operate as a waiver, nor shall any single or partial exercise of any right or remedy prevent any further or other exercise or the exercise of any other right or remedy. The rights and remedies provided in this agreement are cumulative and not exclusive of any rights or remedies provided by law.

40. **AMENDMENTS AND WAIVERS**

40.1 Required Consents

- (a) Subject to clause 40.2 (Exceptions), clause 2.2 (Incremental Facilities), clause 2.3 (Additional Debt), clause 2.6 (Extension Offers), clause 16.3 (Benchmark Replacement) and the provisions of the Intercreditor Agreement, any term of the Finance Documents may be amended or waived only with the consent of the Majority Lenders and Obligors' Agent and any such amendment or waiver will be binding on all Parties; provided that, at Obligors' Agent's option, any Commitments that will be terminated or Loans that will be repaid in connection with such amendment or waiver may be disregarded for purposes of determining whether the consent of the Majority Lenders has been obtained.
- (b) The Facility Agent may effect, on behalf of any Finance Party, any amendment or waiver permitted by this clause 40.
- (c) Each Obligor agrees to any such amendment or waiver permitted by this clause 40 which is agreed to by Obligors' Agent. This includes any amendment or waiver which would, but for this paragraph (c), require the consent of all of the Guarantors.
- (d) The requirements of clause 40.2(d) (Exceptions) are additional to the requirements of clause 40.1(a) (Required Consents). The requirements of clauses 40.2(a), 40.2(b), 40.2(c), 40.2(e), 40.2(h), 40.2(i) and 40.2(j) (Exceptions) override and replace the requirement of clause 40.1(a) (Required Consents) in respect of the matters covered by those clauses.
- (e) For the avoidance of doubt, no further consent of a Finance Party shall be required for any amendment or waiver in respect of a Permitted Debt Change under or in respect of any Finance Document.

40.2 Exceptions

(a) An amendment or waiver that has the effect of changing or which relates to:

- the definitions of "Majority Lenders" or "Super Majority Lenders" in clause 1.1 (Definitions);
- (ii) an increase in any Commitment or the Total Commitments;
- (iii) a change to the Borrowers or Guarantors other than in accordance with clause 30 (Changes to the Obligors);
- (iv) any amendment to the order of priority or subordination under the Intercreditor Agreement;
- (v) clause 34.5 (Partial Payments);
- (vi) clause 2.4 (Finance Parties' Rights and Obligations), clause 29 (Changes to the Lenders), clause 33 (Sharing Among the Finance Parties) or this clause 40;
- (vii) any provision which expressly requires the consent of all the Lenders,

may be made only with the prior consent of all the Lenders and Obligors' Agent and any such amendment or waiver will be binding on the Parties. If, however, the amendment or waiver is a Structural Change or is consequential on or incidental to or required to implement or reflect a Structural Change, the provisions of clause 40.2(e) (Exceptions) will apply provided that in relation to paragraph (vi) above the Structural Change does not result in a new facility or class of indebtedness being created which is to rank for priority ahead of one or more of the existing Facilities. This paragraph (a) is also subject to paragraph (j) below in respect of any changes which would fall within the scope of that paragraph.

- (b) An amendment or waiver that has the effect of being or which relates to:
 - (i) an extension of any Commitment or the Total Commitments;
 - (ii) an extension to the date of payment of any amount under the Finance Documents (other than an extension which results from an amendment or waiver to clause 12 (Mandatory Prepayment);
 - (iii) a reduction in the Margin or a reduction in the amount of any payment of principal, interest, fees or commission payable under the Finance Documents,

may only be made with the prior consent of any Lender in respect of whose Loan(s) and/or commitments the relevant extension of Commitments, extension of payment date or reduction of Margin or amounts referred to in sub-paragraph (iii) above relates.

- (c) An amendment or waiver (made other than as a result of or consequential on or incidental to or required to implement or reflect a Structural Change or a Permitted Debt Change) that has the effect of changing or which relates to:
 - (i) the nature or scope of the Charged Property or the manner in which the proceeds of enforcement of the Transaction Security are distributed (except in so far as it relates to a sale or disposal of an asset which is the subject of Transaction Security where such sale or disposal is expressly permitted under this agreement or any other Finance Document); or
 - (ii) the release of any Transaction Security unless permitted under this agreement or any other Finance Document or relating to a sale or disposal of an asset which is the subject of the Transaction Security where such sale or disposal is expressly permitted under this agreement or any other Finance Document,

may be made only with the prior consent of the Super Majority Lenders and Obligors' Agent and any such amendment or waiver will be binding on the Parties; provided that, at Obligors' Agent's option, any Commitments that will be terminated or Loans that will be repaid in connection with such amendment or waiver may be disregarded for purposes of determining whether the consent of the Super Majority Lenders has been obtained.

- (d) An amendment or waiver which relates to the rights or obligations of the Facility Agent, the Arrangers, the Issuing Bank, the Security Agent or any Ancillary Lender may not be effected without the consent of the Facility Agent, the Arrangers, the Issuing Bank, the Security Agent or that Ancillary Lender.
- (e) An amendment or waiver required to implement a Structural Change (including any modifications to the Finance Documents which are consequential on or incidental to or require to implement or reflect the Structural Change) will only require:
 - (i) the consent of each Lender that elects to participate in the relevant additional tranche or facility or in any increase in, extension of maturity or availability of or redenomination of commitment or extension of that date for payment, or redenomination or reduction of amount due to it involved in the relevant Structural Change.

For the avoidance of doubt, this paragraph (e) is without prejudice to the provisions of clause 2.2 (Incremental Facilities) in respect of Incremental Facilities, clause 2.3 (Additional Debt) in respect of Additional Debt, clause 2.6 (Extension Offers) and clauses 40.2(b) and (j) (Exceptions).

- (f) For the purposes of this clause 40.2 **"Structural Change"** means any of the following modifications to the Finance Documents:
 - the introduction of any additional tranche or facility under the Finance Documents (including a tranche of facility that is Permitted Refinancing Indebtedness to refinance in part or in full an existing Facility); or
 - (ii) any increase in or addition of any commitment, any extension of a commitment's maturity or availability, the redenomination of a commitment into another currency or any extension of the date for payment of or redenomination of, or a reduction of, any amount owing under the Finance Documents.
- (g) The Finance Parties shall (and in respect of the Facility Agent and Security Agent are irrevocably authorised by the other Finance Parties to) enter into any amendments or additional documentation consequential on or required by applicable law to effectively implement a Permitted Debt Change without requiring the consent of any Finance Party.
- (h) Any manifest error or ambiguity in the Finance Documents may be amended by agreement between the Facility Agent and Obligors' Agent and any such amendment will be binding on all Parties.
- (i) Any amendment or waiver that has the effect of changing or which relates to:
 - (i) clause 26.2 (Financial Condition);
 - clause 26.3 (Financial Testing) (except sub-paragraph (c)(ii) of clause 26.3), clause 26.4 (Calculation Adjustments) or the definition of "Net First Lien Secured Leverage" or any component definition thereof solely as such clause or definition is used for the purposes of clause 26.2 (Financial Condition);

- (iii) clauses 28.2(a), 28.2(b) and 28.2(c) (Financial Covenant);
- (iv) clause 28.22(b) (Acceleration);
- (v) the definitions of "Majority TLA/RCF Lenders" set out in clause 1.1 (Definitions),

shall not be effected without (and may be effected with) the consent of the Majority TLA/RCF Lenders (and the consent of no other Finance Party is required); provided that, at Obligors' Agent's option, any Commitments that will be terminated or Loans that will be repaid in connection with such amendment or waiver may be disregarded for purposes of determining whether the consent of the Majority TLA/RCF Lenders has been obtained.

- (j) Any amendment or waiver which:
 - relates only to the rights or obligations applicable to a particular Utilisation, Facility or class of Lender; and
 - (ii) does not materially and adversely affect the rights or interests of Lenders in respect of any other Utilisation or Facility or another class of Lender.

may be made in accordance with the other relevant provisions dealing with amendments and waivers whose consent would, but for this provision, be required for that amendment or waiver as if references to the specified proportion of Lenders (including, for the avoidance of doubt, all Lenders) were to that proportion of the Lenders participating in that particular Utilisation or Facility or forming part of that particular class.

40.3 Replacement of Lender

- (a) If at any time:
 - (i) any Lender becomes a Non-Consenting Lender (as defined in paragraph (c) below) or is subject to a Disruption Event;
 - (ii) an Obligor becomes obliged to repay any amount in accordance with clause 11.1 (Illegality) or to pay additional amounts pursuant to clause 14.5 (Minimum Interest) or clause 19.1 (Increased Costs) or clause 18.2 (Tax Gross-Up) or clause 18.3 (Tax Indemnity) to any Lender in excess of amounts payable to the other Lenders generally;
 - (iii) any Lender becomes insolvent and its assets become subject to a receiver, liquidator, trustee, custodian or other person having similar powers or any winding-up, dissolution or administration; or
 - (iv) Obligors' Agent requests that a Subsidiary becomes a new Borrower under a Revolving Facility and it is unlawful for a Revolving Facility Lender in respect of that Revolving Facility to lend to a borrower in the jurisdiction of incorporation of the proposed new Borrower and the Majority RCF Lenders in respect of the relevant Revolving Facility have otherwise approved the addition of a Subsidiary in such jurisdiction as an Additional Borrower pursuant to clause 30.2 (Additional Borrowers),

then Obligors' Agent may with prior written notice to the Facility Agent and that Lender:

(A) replace that Lender by requiring that Lender to (and that Lender shall) transfer pursuant to clause 29 (Changes to the Lenders) all (and not part only) of its rights and obligations under this agreement to a Lender or other bank, financial institution, trust, fund or other entity or person nominated by Obligors' Agent and/or the Investors (a "Replacement Lender") which in each case is acceptable to the Facility Agent (acting reasonably) and (in the case of any transfer of a Revolving Facility Commitment to the extent it is acting as Issuing Bank in respect of the relevant Revolving Facility Commitment) the Issuing Bank, which confirms its willingness to assume and does assume all the obligations of the transferring Lender (including the assumption of the transferring Lender's participations on the same basis as the transferring Lender) for a purchase price in cash payable at the time of transfer equal to the outstanding principal amount of such Lender's participation in the outstanding Utilisations and all accrued interest and/or Letter of Credit fees, Break Costs and other amounts payable in relation thereto under the Finance Documents; or

- (B) prepay such Lender's participation in the outstanding Utilisations and all approved interest and/or Letter of Credit fees, Break Costs and other amounts payable in relation thereto under the Finance Documents.
- (b) The replacement of a Lender pursuant to this clause shall be subject to the following conditions:
 - (i) the Obligors' Agent shall have no right to replace the Facility Agent or Security Agent;
 - (ii) neither the Facility Agent nor the Lender shall have any obligation to Obligors' Agent to find a Replacement Lender;
 - (iii) in the event of a replacement or prepayment of a Non-Consenting Lender such replacement or prepayment must take place no later than 30 days (or such later date as may be reasonably agreed by the Facility Agent) after the date the Non-Consenting Lender notifies Obligors' Agent and the Facility Agent of its failure or refusal to agree to any consent, waiver or amendment to the Finance Documents requested by Obligors' Agent; and
 - (iv) in no event shall the Lender replaced under this paragraph (b) be required to pay or surrender to such Replacement Lender any of the fees received by such Lender pursuant to the Finance Documents
- (c) If Obligors' Agent or the Facility Agent (at the request of Obligors' Agent) has requested the Lenders to consent to a waiver or amendment of any provisions of the Finance Documents and:
 - (i) the waiver or amendment in question requires the consent of the Super Majority Lenders or all the Lenders or all of the affected Lenders or is a Structural Change which requires a particular Lender's consent (as the case may be) and the Majority Lenders have consented to such waiver or amendment; or
 - (ii) the waiver or amendment in question is to be made pursuant to clauses 40.2(b), 40.2(i) or 40.2(j) (Exceptions), is requested from all the Lenders under a particular Facility or participating in a particular Loan or Utilisation and the Majority Lenders under the relevant Facility or in respect of that Loan or Utilisation have consented to such waiver or amendment,

then any Lender who does not and continues not to agree to such waiver or amendment shall be deemed a "Non-Consenting Lender".

(d) If a Lender does not accept or reject a waiver on request, its Commitment and/or participations shall not be included for the purposes of calculating the Total Commitments or participations under the relevant Facility when ascertaining whether a certain percentage of Total Commitment and/or participations has been obtained to approve an amendment or waiver provided that this shall not apply if a Lender explicitly abstains from accepting or rejecting a request.

40.4 Technical Amendments

Notwithstanding the other provisions of this clause 40:

- (a) the Facility Agent and Obligors' Agent may agree (without the further consent of any other Finance Party) amendments to any Finance Document which in their opinion it would be proper to make provided that any such amendment is not (in the opinion of the Facility Agent) materially prejudicial to the interests of any Lender and is of a formal, minor or technical nature or made to correct a manifest error; and
- (b) the Parties agree to negotiate in good faith any amendments to any Finance Documents reasonably requested by Obligors' Agent in connection with any Permitted Acquisition (whether during a clean-up period or prior to such transaction taking effect) in order to correct any inconsistencies, ambiguities, defects or operational or technical issues arising in the Finance Documents as a result of that transaction, provided that any such amendment is not materially prejudicial to the interests of the Lenders and provided further that nothing in this paragraph (b) shall itself deem a transaction which is not otherwise a Permitted Acquisition to be so permitted.

41. **COUNTERPARTS**

Each Finance Document may be executed in any number of counterparts, and this has the same effect as if the signatures on the counterparts were on a single copy of the Finance Document.

42. **GOVERNING LAW**

This agreement and any non-contractual obligations arising out of or in connection with it are governed by English law.

43. **ENFORCEMENT**

43.1 Jurisdiction of English Courts

- (a) The courts of England, sitting in London have exclusive jurisdiction to settle any dispute arising out of or in connection with this agreement (including a dispute regarding the existence, validity or termination of this agreement) (a "Dispute").
- (b) The Parties agree that the courts of England, sitting in London are the most appropriate and convenient courts to settle Disputes and accordingly no Party will argue to the contrary.

43.2 Service of Process

(a) Without prejudice to any other mode of service allowed under any relevant law, each Obligor (other than an Obligor incorporated in England and Wales):

- irrevocably appoints Obligors' Agent as its agent for service of process in relation to any proceedings before the English courts in connection with any Finance Document (and Obligors' Agent by its execution of this agreement, accepts that appointment); and
- (ii) agrees that failure by an agent for service of process to notify the relevant Obligor of the process will not invalidate the proceedings concerned.
- (b) If any person appointed as an agent for service of process is unable for any reason to act as agent for service of process, Obligors' Agent (on behalf of all the Obligors) must immediately (and in any event within five days of such event taking place) appoint another agent on terms acceptable to the Facility Agent. Failing this, the Facility Agent may appoint another agent for this purpose.
- (c) Obligors' Agent expressly agrees and consents to the provisions of this clause 43 and clause 42 (Governing Law).

44. **NET SHORT LENDERS**

Notwithstanding anything to the contrary herein, in connection with any determination as to whether the Majority Lenders or the Majority RCF Lenders, as applicable, have (A) delivered (or not delivered) a notice of a Default or an Event of Default or a notice of acceleration under any Finance Document, (B) otherwise acted on any matter related to a Default, an Event of Default or an acceleration under any Finance Document, or (C) directed or required the Facility Agent to deliver (or not deliver) a notice of a Default or an Event of Default or a notice of acceleration under any Finance Document, any Lender (other than (x) any Lender that is a Regulated Bank, (y) any RCF Lender as of the 2022 Effective Date and (z) any Affiliate of the foregoing (or as otherwise agreed among the Parent, the applicable RCF Lender and the Facility Agent)) that, as a result of its (or its Affiliates) interest in any total return swap, total rate of return swap, credit default swap or other derivative contract (other than any such total return swap, total rate of return swap, credit default swap or other derivative contract entered into pursuant to bona fide market making activities), has a net short position with respect to the Loans and/or Commitments (each, a "Net Short Lender"), without the consent of the Parent, shall have no right to take any action pursuant to Clause 28 (Events of Default), be counted as a Majority Lender or Majority RCF Lender with respect any action pursuant to Clause 28 (Events of Default) or vote any of its Loans and Commitments with respect to any action pursuant to Clause 28 (Events of Default) (and, with respect to any votes with respect to any action pursuant to Clause 28 (Events of Default), shall be deemed to have voted its interest as a Lender without discretion in the same proportion as the allocation of voting with respect to such matter by Lenders who are not Net Short Lenders). For purposes of determining whether a Lender (alone or together with its Affiliates) has a "net short position" on any date of determination: (i) derivative contracts with respect to the Loans and Commitments and such contracts that are the functional equivalent thereof shall be counted at the notional amount thereof in Dollars, (ii) notional amounts in other currencies shall be converted to the dollar equivalent thereof by such Lender in a commercially reasonable manner consistent with generally accepted financial practices and based on the prevailing conversion rate (determined on a mid-market basis) on the date of determination, (iii) derivative contracts in respect of an index that includes any of the Group Company, or any parent company of the Parent or other Obligors or any instrument issued or guaranteed by any Group Company, parent company of Parent or other Obligors shall not be deemed to create a short position with respect to the Loans and/or Commitments, so long as (x) such index is not created, designed, administered or requested by such Lender or its Affiliates and (y) the Group Company, or any parent company of the Parent or other Obligors and any instrument issued or guaranteed by any of Group Company, or any parent company of the Parent or other Obligors, collectively, shall represent less than 5% of the components of such index, (iv) derivative transactions that are documented using either the 2014 ISDA Credit Derivatives Definitions or the 2003 ISDA Credit Derivatives Definitions (collectively, the "ISDA CDS Definitions") shall be

deemed to create a short position with respect to the Loans and/or Commitments if such Lender or its Affiliates is a protection buyer or the equivalent thereof for such derivative transaction and (x) the Loans or the Commitments are a "Reference Obligation" under the terms of such derivative transaction (whether specified by name in the related documentation, included as a "Standard Reference Obligation" on the most recent list published by Markit, if "Standard Reference Obligation" is specified as applicable in the relevant documentation or in any other manner), (y) the Loans or the Commitments would be a "Deliverable Obligation" under the terms of such derivative transaction or (z) any of the Group Company, or any parent company of the Parent or other Obligors (or their successor) are designated as a "Reference Entity" under the terms of such derivative transactions, and (v) credit derivative transactions or other derivatives transactions not documented using the ISDA CDS Definitions shall be deemed to create a short position with respect to the Loans and/or Commitments if such transactions are functionally equivalent to a transaction that offers the Lender or its Affiliates protection in respect of the Loans or the Commitments, or as to the credit quality of any of the Group Company, or any parent company of the Parent or other Obligors other than, in each case, as part of an index so long as (x) such index is not created, designed, administered or requested by such Lender or its Affiliates and (y) the Group Company, or any parent company of the Parent or other Obligors and any instrument issued or guaranteed by any of the Group Company, or any parent company of the Parent or other Obligors, collectively, shall represent less than 5% of the components of such index. In connection with any such determination, each Lender shall promptly notify the Facility Agent in writing that it is a Net Short Lender, or, in the absence of any such written notification to the Facility Agent prior to the date of such determination, shall otherwise be deemed to have represented and warranted to the Parent and the Facility Agent that it is not a Net Short Lender (it being understood and agreed that the Parent and the Facility Agent shall be entitled to rely conclusively on each such representation and deemed representation) (the "Net Short Representation"). The Facility Agent shall be entitled to rely conclusively on any Net Short Representation delivered, provided or made (or deemed delivered, provided or made) to it in accordance with this agreement, shall have no duty to inquire as to or investigate the accuracy of any Net Short Representation, verify any statements in any officer's certificate delivered to it, or otherwise make any calculations, investigations or determinations with respect to any interest in any total return swap, total rate of return swap, credit default swap or other derivative contract or "net short positions" or any person.

45. **BAIL-IN**

45.1 **Bail-In Definitions**

In this Clause 45:

"Article 55 BRRD" means Article 55 of Directive 2014/59/EU establishing a framework for the recovery and resolution of credit institutions and investment firms.

"Bail-In Action" means the exercise of any Write-down and Conversion Powers.

"Bail-In Legislation" means:

- (a) in relation to an EEA Member Country which has implemented, or which at any time implements, Article 55 BRRD, the relevant implementing law or regulation as described in the EU Bail-In Legislation Schedule from time to time;
- (b) in relation to the United Kingdom, the UK Bail-In Legislation; and
- (c) in relation to any state other than such an EEA Member Country and the United Kingdom, any analogous law or regulation from time to time which requires contractual recognition of any Writedown and Conversion Powers contained in that law or regulation.

"EEA Member Country" means any member state of the European Union, Iceland, Liechtenstein and Norway.

"EU Bail-In Legislation Schedule" means the document described as such and published by the Loan Market Association (or any successor person) from time to time.

"Resolution Authority" means any body which has authority to exercise any Write-down and Conversion Powers.

"**UK Bail-In Legislation**" means Part I of the United Kingdom Banking Act 2009 and any other law or regulation applicable in the United Kingdom relating to the resolution of unsound or failing banks, investment firms or other financial institutions or their affiliates (otherwise than through liquidation, administration or other insolvency proceedings).

"Write-down and Conversion Powers" means:

- (a) in relation to any Bail-In Legislation described in the EU Bail-In Legislation Schedule from time to time, the powers described as such in relation to that Bail-In Legislation in the EU Bail-In Legislation Schedule;
- (b) in relation to the UK Bail-In Legislation, any powers under that UK Bail-In Legislation to cancel, transfer or dilute shares issued by a person that is a bank or investment firm or other financial institution or affiliate of a bank, investment firm or other financial institution, to cancel, reduce, modify or change the form of a liability of such a person or any contract or instrument under which that liability arises, to convert all or part of that liability into shares, securities or obligations of that person or any other person, to provide that any such contract or instrument is to have effect as if a right had been exercised under it or to suspend any obligation in respect of that liability or any of the powers under that UK Bail-In Legislation that are related to or ancillary to any of those powers; and
- (c) in relation to any other applicable Bail-In Legislation:
 - (i) any powers under that Bail-In Legislation to cancel, transfer or dilute shares issued by a person that is a bank or investment firm or other financial institution or affiliate of a bank, investment firm or other financial institution, to cancel, reduce, modify or change the form of a liability of such a person or any contract or instrument under which that liability arises, to convert all or part of that liability into shares, securities or obligations of that person or any other person, to provide that any such contract or instrument is to have effect as if a right had been exercised under it or to suspend any obligation in respect of that liability or any of the powers under that Bail-In Legislation that are related to or ancillary to any of those powers; and
 - (ii) any similar or analogous powers under that Bail-In Legislation.

45.2 Contractual recognition of bail-in

Notwithstanding any other term of any Finance Document or any other agreement, arrangement or understanding between the Parties, each Party acknowledges and accepts that any liability of any Party to any other Party under or in connection with the Finance Documents may be subject to Bail-In Action by the relevant Resolution Authority and acknowledges and accepts to be bound by the effect of:

- (a) any Bail-In Action in relation to any such liability, including (without limitation):
 - (i) a reduction, in full or in part, in the principal amount, or outstanding amount due (including any accrued but unpaid interest) in respect of any such liability;

- (ii) a conversion of all, or part of, any such liability into shares or other instruments of ownership that may be issued to, or conferred on, it; and
- (iii) a cancellation of any such liability; and
- (b) a variation of any term of any Finance Document to the extent necessary to give effect to any Bail-In Action in relation to any such liability.

THIS AGREEMENT has been entered into on the date stated at the beginning of this agreement.

SCHEDULE 1

The Original Parties

Part 1 - The Original Obligors

Name of Original Borrower/registration number (or equivalent, if any) and jurisdiction of incorporation

Term Loan borrower

Delta 2 (Lux) Sàrl	B 122 129	Luxembourg
RCF borrowers		
Delta 2 (Lux) Sàrl	B 122 129	Luxembourg
Formula One Hospitality and Event Services Limited	8445563	England and Wales
Formula One Management Limited	1545332	England and Wales
Formula One Marketing Limited	5731193	England and Wales
Formula One World Championship Limited	4174493	England and Wales
Formula One Digital Media Limited	08915039	England and Wales

Name of Original Guarantor/registration number (or equivalent, if any) and jurisdiction of incorporation

Allsport Management S.A.	CHE-103.211.396	Switzerland
Alpha Prema UK Limited	5613176	England and Wales
Formula One Research, Engineering and Development Limited	5821941	England and Wales
Beta Holdings Limited	5729365	England and Wales
Delta 2 (Lux) Sàrl	B 122 129	Luxembourg
Delta 3 (UK) Limited	5988544	England and Wales
Formula One Hospitality and Event Services Limited	8445563	England and Wales
Formula One Management Limited	1545332	England and Wales
Formula One Marketing Limited	5731193	England and Wales
Formula One Marketing II Limited	7714752	England and Wales
Formula One World Championship Limited	4174493	England and Wales

SLEC Holdings Limited	68316	Jersey
Formula One Digital Media Limited	08915039	England and Wales
Delta Topco Limited	95136	Jersey

Part 2 - Total Commitments

Total Commitments (as of 2022 Effective Date)

Total Facility A Commitments	Total Facility B Commitments	Total RCF Commitments
\$725,000,000	\$1,700,000,000	\$500,000,000

Following the modification to Commitments arising on or about the 2022 Effective Date

Total Commitments (as of 2024 Effective Date)

Total Facility A Commitments	Total Facility B Commitments	Total RCF Commitments
\$688,750,000	\$1,700,000,000	\$500,000,000

Following the modification to Commitments arising on or about the 2024 Effective Date

Part 3 - RCF Lenders

RCF Lenders	RCF Commitment (as of 2022 Effective Date)	RCF Commitment (as of 2024 Effective Date)	DTTP Scheme Reference Number	Jurisdiction of Tax Residence
Goldman Sachs Lending Partners LLC	\$63,000,000	\$63,000,000	13/G/356209/DTTP	United States
Citibank, N.A.	\$47,500,000	\$47,500,000	13/C/62301/DTTP	United States
Société Générale	\$47,500,000	\$47,500,000	005/S/70085	France
Bank of America, N.A., London Branch	\$44,500,000	\$44,500,000	N/A	United States
BNP Paribas	\$43,500,000	\$43,500,000	5/B/255139/DTTP	France
The Toronto-Dominion Bank, New York Branch	\$36,000,000	\$36,000,000	3/T/80000/DTTP	Canada
Truist Bank	\$36,000,000	\$36,000,000	13/T/357522	United States
Credit Suisse AG, New York Branch	\$35,000,000	\$0	N/A	Switzerland
JPMorgan Chase Bank, N.A., London Branch	\$30,000,000	\$30,000,000	13/M/0268710/DTTP	United States
CIBC Capital Markets (Europe) S.A.	\$18,000,000	\$28,000,000	N/A	Luxembourg
Credit Agricole Corporate and Investment Bank	\$27,000,000	\$27,000,000	005/C/0222082/DTTP	France
Morgan Stanley Senior Funding, Inc.	\$27,000,000	\$27,000,000	13/M/227953/DTTP	United States
Royal Bank of Canada	N/A	\$25,000,000	3/R/70780/DTTP	Canada
MUFG Bank, Ltd.	\$22,000,000	\$22,000,000	43/B/322072/DTTP	Japan
Mizuho Bank, Ltd.	\$14,000,000	\$14,000,000	43/M/274822/DTTP	Japan
National Westminster Bank plc	\$9,000,000	\$9,000,000	N/A	United Kingdom

Part 4 - Term Facility A Lenders

Term Facility A Lenders	Term Facility A Commitment (as of 2022 Effective Date)	Term Facility A Commitment (as of 2024 Effective Date)	DTTP Scheme Reference Number	Jurisdiction of Tax Residence
National Westminster Bank plc	\$80,000,000	\$76,000,000	N/A	United Kingdom
JPMorgan Chase Bank, N.A.	\$78,000,000	\$74,100,000	13/M/0268710/DTTP	United States
Mizuho Bank, Ltd.	\$75,000,000	\$71,250,000	43/M/274822/DTTP	Japan
BNP Paribas	\$64,500,000	\$61,275,000	5/B/255139/DTTP	France
The Toronto-Dominion Bank, New York Branch	\$53,000,000	\$50,350,000	3/T/80000/DTTP	Canada
Truist Bank	\$53,000,000	\$50,350,000	13/T/357522	United States
Goldman Sachs Bank USA	\$45,000,000	\$42,750,000	13/G/351779/DTTP	United States
MUFG Bank, Ltd.	\$45,000,000	\$42,750,000	43/B/322072/DTTP	Japan
Bank of America, N.A., London Branch	\$44,500,000	\$42,275,000	N/A	United States
Société Générale	\$41,500,000	\$39,425,000	005/S/70085	France
Credit Agricole Corporate and Investment Bank	\$40,000,000	\$38,000,000	005/C/0222082/DTTP	France
Morgan Stanley Senior Funding, Inc.	\$40,000,000	\$38,000,000	13/M/227953/DTTP	United States
CIBC Capital Markets (Europe) S.A.	\$27,000,000	\$35,150,000	N/A	Luxembourg
Citibank, N.A.	\$28,500,000	\$27,075,000	13/C/62301/DTTP	United States
Credit Suisse AG, New York Branch	\$10,000,000	\$0	N/A	Switzerland

Part 5 - 2024 Incremental Term Facility Lenders

2024 Incremental Term	Term Facility B	DTTP Scheme Reference	Jurisdiction of Tax
Lenders	Commitment	Number	Residence
Goldman Sachs Bank USA	\$1,700,000,000	13/G/351779/DTTP	United States

SCHEDULE 2

Conditions Precedent

Part 1A – Conditions Precedent to Signing of the Agreement

[The conditions precedent to signing recorded below were satisfied]

Part 1B - Conditions Precedent to Initial Utilisation

[The conditions precedent originally recorded in this agreement have been satisfied]

Part 2A - Conditions Subsequent

[The conditions subsequent originally recorded in this agreement have been satisfied]

Part 2B - Conditions Subsequent

[The conditions subsequent originally recorded in this agreement have been satisfied]

Part 2C - Conditions Subsequent

[The conditions subsequent originally recorded in this agreement have been satisfied]

Part 3 - Conditions Precedent Required to be Delivered by an Additional Obligor

- 1. An Accession Letter executed by the Additional Obligor and the Parent.
- 2. A copy of the constitutional documents of the Additional Obligor.
- 3. A copy of a resolution of the board of directors of the Additional Obligor:
 - (a) approving the terms of, and the transactions contemplated by, the Accession Letter and the Finance
 Documents and resolving that it execute, deliver and perform the Accession Letter and any other
 Finance Document to which it is party;
 - (b) authorising a specified person or persons to execute the Accession Letter and other Finance Documents on its behalf;
 - (c) authorising a specified person or persons, on its behalf, to sign and/or despatch all other documents and notices (including, in relation to an Additional Borrower, any Utilisation Request or Selection Notice) to be signed and/or despatched by it under or in connection with the Finance Documents to which it is a party; and
 - (d) authorising Obligors' Agent to act as its agent in connection with the Finance Documents.
- 4. A specimen of the signature of each person authorised by the resolution referred to in paragraph 3 above which has signed a document in relation to that Additional Obligor referred to in this Part 3.
- 5. If required by the Facility Agent, a copy of a resolution signed by all the holders of the issued shares of the Additional Guarantor, approving the terms of, and the transactions contemplated by, the Finance Documents to which the Additional Guarantor is a party.
- 6. A certificate of the Additional Obligor (signed by a director) confirming that borrowing or guaranteeing or securing, as appropriate, the Total Commitments would not cause any borrowing, guarantee, security or similar limit binding on it to be exceeded.
- 7. A certificate of the Additional Obligor (signed by a director) certifying that each copy document listed in this part 3 of schedule 2 is correct, complete and in full force and effect and has not been amended or superseded as at a date no earlier than the date of the Accession Letter.
- 8. If available, the latest audited financial statements of the Additional Obligor.
- 9. The following legal opinions, each addressed to the Facility Agent, the Security Agent and the Lenders (in each case only to the extent required by the Facility Agent):
 - (a) a legal opinion of the legal advisers to the Facility Agent in England, as to English law in the form distributed to the Lenders prior to signing the Accession Letter;
 - (b) if the Additional Obligor is incorporated in or has its "centre of main interest" or "establishment" (as referred to in clause 27.28 (Centre of Main Interests)) in a jurisdiction other than the United Kingdom or the US or is executing a Finance Document which is governed by a law other than English law:
 - a legal opinion of the legal advisers to the Facility Agent in the jurisdiction of its incorporation,
 "centre of main interest" or "establishment" (as applicable) or, as the case may be, the jurisdiction of the governing law of that Finance

Document (the "Applicable Jurisdiction") as to the law of the Applicable Jurisdiction; and

(ii) a legal opinion of the legal advisers to the Parent in relation to matters customary for a borrower's lawyers to give an opinion on in the Applicable Jurisdiction,

and each in the form distributed to the Lenders prior to signing the Accession Letter.

- 10. If the proposed Additional Obligor is incorporated in a jurisdiction other than England and Wales, evidence that the process agent specified in clause 43.2 (Service of Process), if not an Original Obligor, has accepted its appointment in relation to the proposed Additional Obligor.
- 11. Any security documents which, subject to the Agreed Security Principles, are required by the Facility Agent to be executed by the proposed Additional Obligor.
- 12. Any notices or documents required to be given or executed under the terms of those security documents.
- 13. A certified extract of the register of members of each Group Company, the shares of which are subject to or expressed to be subject to the relevant security documents (if such Group Company has a register of members).
- 14. To the extent required by the Facility Agent only, evidence that each Group Company has done all that is necessary under any relevant laws relating to financial assistance or analogous processes (including without limitation for Group Companies incorporated in England and Wales, Scotland or Northern Ireland reregistering as a private company) in order to enable all relevant Obligors to enter into the Finance Documents and perform their obligations under the Finance Documents.
- 15. (a) In relation to each Additional Obligor (in each case certified as a true copy by a director and only to the extent available in relation to such Group Company):
 - a copy of its certificate of incorporation or equivalent confirming legal form, legal and trading name, address, statutory seat and registration number with trade register; and
 - (ii) a copy of its share register confirming the identify of its shareholders.
 - (b) In relation to each authorised signatory of each Additional Obligor, if required by the Facility Agent, (in each case certified as a true copy by the legal advisors to the Parent):
 - a copy of his or her current passport or national identity card including a photograph (to evidence name and date of birth); and
 - (ii) a copy of any one of the following (to evidence address): bank statement issued within the last three months, current driving licence or utility bill issued within the last three months.
 - (c) Any other evidence required by the Facility Agent under clause 25.8(d) ("Know Your Customer" Checks).
- 16. Any other Authorisation, document, opinion or assurance which the Facility Agent notifies the Parent is reasonably necessary or desirable in connection with the entry of the Additional Obligor into and performance of the Finance Documents or for the validity and enforceability of any Finance Document.

Part 4A – transaction security documents

Existing Transaction Security Documents	Name of applicable Obligor(s)	Supplemental Transaction Security Documents
ENGLISH LAW GOVERNED SECURI	TY DOCUMENTS	
Debenture dated 24 November 2006 (the "2006 Debenture") charging the assets specified therein by way of fixed and/or floating charge in favour of the NatWest Markets plc (formerly known as The Royal Bank of Scotland plc) ("RBS") as Security Agent.	Alpha Prema UK Limited, Formula One Research Engineering and Development Limited (formerly known as Beta D3 Limited), Beta Holdings Limited, Formula One Marketing Limited (formerly known as Beta (Prema) UK Limited), Delta 3 (UK) Limited, Formula One World Championship Limited, Formula One Management Limited, Formula One Marketing II Limited (formerly known as Formula One Marketing (India) Limited and Beta Prema 2 Limited) (the "English Obligors"); and Beta Topco 1 Limited, SLEC Holdings Limited, Delta Topco Limited, Delta 2 (Lux) Sàrl, Allsport Management SA (the "A3 Charging Companies").	
Share Charge dated 24 November 2006 (the "2006 Shares Charge") charging in favour of RBS as Security Agent with full title guarantee all the Subsidiary Shares and Investments and corresponding Distribution Rights as first legal mortgage or as a fixed charge as set out therein.	SLEC Holdings Limited and Delta 2 (Lux) Sàrl	
Debenture dated 27 April 2012 charging the assets specified therein by fixed and/or floating charge in favour of RBS as the Security Agent subject to the Existing Debenture and, as further security, including a Security Assignment assigning all rights, title and interest in	The English Obligors and the A3 Charging Companies.	

the Assigned Agreements as defined therein (the "2012 Debenture")		
Share Charge dated 27 April 2012 (the "2012 Shares Charge") charging in favour of RBS as the Security Agent with full title guarantee (subject to the 2006 Shares Charge) all the Subsidiary Shares and Investments and corresponding Distribution Rights as first legal mortgage or as a fixed charge as set out therein.	SLEC Holdings Limited and Delta 2 (Lux) Sàrl	
Supplemental Debenture dated 25 June 2013 charging the assets specified therein by fixed and/or floating charge in favour of RBS as the Security Agent subject to the Existing Debenture and the 2012 Debenture and, as further security, including a Security Assignment assigning all rights, title and interest in the Assigned Agreements as defined therein (the "Supplemental Debenture").	The English Obligors and the A3 Charging Companies	
Supplemental Shares Charge dated 25 June 2013 charging in favour of RBS as the Security Agent with full title guarantee (subject to the 2006 Shares Charge and the 2012 Shares Charge) all the Subsidiary Shares and Investments and corresponding Distribution Rights as first legal mortgage or as a fixed charge as set out therein (the "Supplemental Shares Charge").	SLEC Holdings Limited and Delta 2 (Lux) Sàrl	
SWISS LAW GOVERNED SECURITY DOCUMENTS		
Share Pledge agreement dated 24 November 2006 granting first ranking pledge over the existing shares and all other or additional shares in Allsport Management SA in favour of	Formula One Marketing Limited (formerly known as Beta Prema UK Limited)	

RBS as Security Agent and Trustee		
Fiduciary Assignment Agreement & Declaration of Assignment dated 24 November 2006 Assigning all rights and claims in respect of bank accounts, all rights and claims in respect of intra group receivables and all other rights and benefits relating thereto to RBS as Facility Agent and Trustee	Allsport Management SA	

JERSEY LAW GOVERNED SECURITY	Y DOCUMENTS	
A Security Agreement (Possession of Certificates of Title to and Assignment of Securities) dated 24 November 2006 over shares in SLEC Holdings Limited	Alpha Prema UK Limited	A Junior Security Agreement (Possession of Certificates of Title to and Assignment of Securities) dated 27 April 2012_over shares in SLEC Holdings Limited A Junior Security Agreement (Possession of Certificates of Title to and Assignment of Securities) dated 25 June 2013 over shares in SLEC Holdings Limited

LUXEMBOURG LAW GOVERNED SECURITY DOCUMENTS				
Pledge agreement over receivables dated 24 November 2006 granted by Delta 2 (Lux) Sàrl as pledgor in favour of RBS as security agent.	Delta 2 (Lux) Sàrl	The existing pledge agreement was amended and confirmed on 6 July 2007, 27 April 2012, and 25 June 2013 and 8 August 2014 pursuant to Luxembourg law amendment and confirmation agreements		
A Share Pledge Agreement dated 24 December 2023 between Delta Topco Limited as pledgor, Natwest Markets PLC as pledgee and security agent and Delta 2 (Lux) Sàrl as company.	Delta 2 (Lux) Sàrl			

Part 4B - Sixth Restatement Agreement Transaction Security Documents

1. ENGLISH LAW GOVERNED TRANSACTION SECURITY DOCUMENTS

- 1.1 A debenture dated the Sixth Restatement Date executed by amongst other Formula One Research, Engineering and Development Limited (formerly known as Beta D3 Limited), Beta Holdings Limited, Alpha Prema UK Limited, Formula One Marketing Limited, Formula One Marketing II Limited, Delta 3 (UK) Limited, Formula One Management Limited, Formula World Championship Limited, SLEC Holdings Limited, Delta 2 (Lux) Sàrl, Allsport Management SA, and Formula One Hospitality and Event Services Limited substantially in the form of the English law debenture dated 27 April 2012 (except that fixed security over bank accounts is not granted).
- 1.2 A share charge dated the Sixth Restatement Date granted by SLEC Holdings Limited and Delta 2 (Lux) Sàrl over the issued share capital of Formula One Asset Management Limited (in respect of an ordinary shares only), Formula One World Championship Limited, Delta 3 (UK) Limited and Formula One Research, Engineering and Development Limited (formerly known as Beta D3 Limited) as applicable.

2. JERSEY LAW GOVERNED TRANSACTION SECURITY DOCUMENTS

2.1 Specific Security Agreement dated the Sixth Restatement Date executed by Alpha Prema UK Limited over all of its shares in SLEC Holdings Limited.

3. LUXEMBOURG LAW GOVERNED TRANSACTION SECURITY DOCUMENTS

3.1 A confirmation and amendment agreement dated the Sixth Restatement Date in relation to the pledge agreement over Delta 2 (Lux) Sàrl's bank accounts and receivables, between Delta 2 (Lux) Sàrl and RBS, dated 24 November 2006 (confirming the release of the pledge over the Bank Account Balances (as defined therein).

4. SWISS LAW GOVERNED TRANSACTION SECURITY DOCUMENTS

4.1 An amendment and confirmation agreement dated the Sixth Restatement Date in relation to (i) the pledge agreement dated as of 24 November 2006 between, inter alia, Formula One Marketing Limited (formerly known as Beta Prema UK Limited) as Pledgor and RBS as Pledgee and as Original Pledgee in relation to all shares in Allsport Management SA and (ii) a fiduciary assignment agreement & declaration of assignment dated as of 24 November 2006 between, inter alia, Allsport Management SA and RBS as Assignee and as Original Assignee (except that security over bank accounts is not granted).

SCHEDULE 3

Requests

Part 1A - Utilisation Request Loans

From: [Borro	wer] [FORMULA ONE MANA	AGEMENT LIMITED]		
То:	[Facility Agent]			
Dated:	[]			
Dear Si	rs				
	emen	t dated 23 November 2022	d 21 November 2006 as amended and restated by an amendment as amended and restated by an amendment agreement dated 19 er amended from time to time (the "Facilities Agreement")		
1.	We refer to the Facilities Agreement. This is a Utilisation Request. Terms defined in the Facilities Agreement have the same meaning in this Utilisation Request unless given a different meaning in this Utilisation Request.				
2.	We v	vish to borrow a Loan on the	following terms:		
	(a)	Borrower:	П		
	(b)	Proposed Utilisation Date:	[] (or, if that is not a Business Day, the next Business Day)		
	(c)	Facility to be utilised:	П		
	(d)	Currency of Loan:			
	(e)	Amount:	[] or, if less, the Available Facility		
	(f)	Type of Loan:	[Term Benchmark Loan/RFR Loan]		
	(g)	Interest Period (if a Term Benchmark Loan):	D .		
3.	We confirm that each applicable condition specified in clause 4.2 (Further Conditions Precedent) is satisfied on the date of this Utilisation Request.				
4.	[The proceeds of this Loan should be credited to [account]].				
5.	-	Sutilisation Request is irrevoor rrence of []]	cable]/[This Utilisation Request is conditioned upon the occurrence or non-		
			Yours faithfully		
			authorised signatory for		
[F	ormul	a One Management Limited o	n behalf of [insert name of relevant Borrower]]/ [insert name of Borrower]		

Part 1B - Utilisation Request Letters of Credit

From: [Borrower] [Formula One Management Limited]					
To:	[Facility Agent]			
Dated:	[]			
Dear Si	rs				
First Lien Facilities Agreement dated 21 November 2006 as amended and restated by an amendment agreement dated 23 November 2022 as amended and restated by an amendment agreement dated 19 September 2024 and as further amended from time to time (the "Facilities Agreement")					
1.	We refer to the Facilities Agreement. This is a Utilisation Request. Terms defined in the Facilities Agreement have the same meaning in this Utilisation Request unless given a different meaning in this Utilisation Request.				
2.		vish to arrange for a Letter of Coso) on the following terms:	redit to be issued by the Issuing Bank specified below (which has agreed		
	(a)	Borrower:	0		
	(b)	Issuing Bank:	0		
	(c)	Proposed Utilisation Date:	[] (or, if that is not a Business Day, the next Business Day)		
	(d)	Facility to be utilised:	1		
	(e)	Currency of Letter of Credit:	[]		
	(f)	Amount:	[] or, if less, the Available Facility in relation to []		
	(g)	Term:	[]		
3.	We confirm that each condition specified in of clause 6.5(c) (Issue of Letters of Credit) is satisfied on the date of this Utilisation Request.				
4.	We attach a copy of the proposed Letter of Credit.				
5.	The purpose of this proposed Letter of Credit is [].				
6.	This	Utilisation Request is irrevocabl	e.		
			authorised signatory for		
[Formula One Management Limited on behalf of] [insert name of relevant Borrower]]/[insert name of Relevant Borrower]					

Part 2 - Selection Notice

From: [Borrower] [Formula One Management Limited]

To:	[Facility Agent]			
Dated:	D .			
Dear Si	rs			
First Lien Facilities Agreement dated 21 November 2006 as amended and restated by an amendment agreement dated 23 November 2022 as amended and restated by an amendment agreement dated 19 September 2024 and as further amended from time to time (the "Facilities Agreement")				
1.	We refer to the Facilities Agreement. This is a Selection Notice. Terms defined in the Facilities Agreement have the same meaning in this Selection Notice unless given a different meaning in this Selection Notice.			
2.	We refer to the following Facility [] Loan[s] with [an Interest Period ending on] / [an Interest Payment Date of] [].			
3.	[We request that the above Facility [] Loan[s] be divided into [] Facility [] Loans with the following Base Currency Amounts and Interest Periods:]			
or				
[We request that the next Interest Period for the above Facility [] Loan[s] is []].				
4.	This Selection Notice is irrevocable.			
Yours fa	aithfully			
	authorised signatory for			
	[Formula One Management Limited on behalf of] [insert name of relevant Borrower]			
	230			

SCHEDULE 4

Form of Transfer Certificate

WARNING

[Execution and delivery of this transfer certificate may not of itself be sufficient to transfer the underlying security. Advice should be taken from lawyers in the jurisdictions where the security providers are incorporated.]

[Each of the existing lender and new lender should ensure that all regulatory requirements are satisfied in connection with this transfer certificate - including those necessary to remove the transferred assets from the regulatory balance sheet of the existing lender.]

To: [●] as Facility Agent

From: [The Existing Lender] (the "Existing Lender") and [The New Lender] (the "New Lender")

Dated: []

First Lien Facilities Agreement dated 21 November 2006 as amended and restated by an amendment agreement 23 November 2022 as amended and restated by an amendment agreement dated 19 September 2024 and as further amended from time to time (the "Facilities Agreement")

1. **Defined Terms**

We refer to the Facilities Agreement. This is a Transfer Certificate. Terms defined in the Facilities Agreement have the same meaning in this Transfer Certificate unless given a different meaning in this Transfer Certificate.

2. **Existing Lender Commitments**

The Existing Lender confirms that the details which appear in part 1 of the schedule to this certificate accurately record the amount of the Existing Lender's Commitments at the date of this certificate.

Transfer

- (a) The Existing Lender, the New Lender and the Facility Agent agree to the transfer of the Existing Lender's Commitments, rights and obligations specified in part 2 of the schedule to this certificate to the New Lender in accordance with clause 29.5 (Procedure for Transfer).
- (b) To the extent this certificate transfers participations in Loans outstanding on the Transfer Date, that transfer will take effect in relation to those participations in Loans [and all related rights under the Facilities Agreement] by way of assignment (the "Assigned Rights").]
- (c) The proposed Transfer Date is [].

4. New Lender Undertaking and Acknowledgement

(a) The New Lender undertakes with the Existing Lender and each of the other parties to the Facilities Agreement that it will perform all those obligations which, by the terms of the Facilities Agreement, will be assumed by it following execution of this certificate by the Facility Agent.

- (b) The New Lender acknowledges and agrees that it enters into this certificate subject to the terms of clause 29.4 (Limitation of Responsibility of Existing Lenders).
- (c) The New Lender undertakes with the Existing Lender and each of the other parties to the Facilities Agreement represents and warrants that it is not a Disqualified Lender or a Net Short Lender.
- (d) The New Lender expressly confirms that it [can/cannot] exempt the Facility Agent from the restrictions pursuant to section 181 of the German Civil Code (Bürgerliches Gesetzbuch) and similar restrictions applicable to it pursuant to any other applicable law as provided for in Clause 31.1 (Appointment of the Facility Agent and the Security Agent).

5. Revolving Commitments

To the extent that this certificate operates to transfer Revolving Commitments, each Issuing Lender has consented to that transfer in accordance with clause 29.2 (Conditions of Assignment or Transfer).

6. Luxembourg Security

For the purposes of Article 1278 of the Luxembourg Civil Code (to the extent applicable), the New Lender and the Existing Lender agree that, upon any transfer in whole or in part of any rights of an Existing Lender to a New Lender, the security created by the Transaction Security Documents will be preserved for the benefit of the New Lender and the Security Agent.

7. No Set-off Rights/Ownership

The Existing Lender represents and warrants that:

- (a) the Assigned Rights are assigned free of any rights of set-off in favour of any Obligor and free of any lien, security interest or other encumbrance; and
- (b) immediately prior to the Transfer Date, the Existing Lender is the beneficial owner of the Assigned Rights.

8. [L/C Lender

The New Lender confirms that it [is][is not] a Non-Acceptable L/C Lender.]

9. Intercreditor Agreement

The New Lender (if not already a party in this capacity) agrees to become a First Lien Lender for the purpose of the Intercreditor Agreement and agrees to be bound by the Intercreditor Agreement in that capacity.

10. Counterparts

This certificate may be executed in any number of counterparts and all of those counterparts taken together shall be deemed to constitute one and the same instrument.

11. **Law**

This certificate and any non-contractual obligations arising out of or in connection with it (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this certificate or any non-contractual obligations arising out of or in connection with it) shall be governed by and construed in accordance with English law. The

parties submit to the jurisdiction of the English courts in accordance with clause 43 (Enforcement) of the Facilities Agreement.

IN WITNESS of which the parties to this certificate have duly executed this certificate on the date which appears at the end of this certificate.

Schedule to Transfer Certificate

Part 1

Existing Le	nder's existing	g Commitments				
[Details to b	[Details to be inserted]					
			ı	Part 2		
Commitme	nts transferre	d				
Details to b	e inserted]					
	-					
Participation	n in Term Loar	ıs transferred:				
Borrower				Interest I	Period	
[]	Participation [\$ [\$]	[]	
L	J	L⊅	J	L	J	
Participation	n in Revolving	Loans transferr	ed:			
Borrower		Participation		Interest P	eriod	
[]	[]	[]	[]

Borrower	Beneficia	ry	Face Amount	Participation	Expiry Date
[]	[]	[]	[]	
[]	[]	[]	[]	[]
			Part 3		
Particulars relati	ng to the New	Lender			
Facility Office:					
Contact Name:					
Account for Payme	ents:				
Address for Notice	es:				
Telephone:					
Facsimile:					
		Sign	atories to Transfer	Certificate	
[Existing Lender]				Lender] cuted as a deed by]	
Ву:					
Date:	••		Date:		
[Facility Agent]					
By:					
Date:					
We acknowledge t	he above and in	n narticul:	ar the transfer of oh	ligations to the New	· Lender recorded in the ab
certificate.	above and n	. particult	a. a.s dansier of ob	gat.ono to the New	22 recorded in the di
By:					

Form of Accession Letter

[] as Facility Agent

To:

From:	[Subsidiary] and [Formula One Management Limited]
Dated:	
Dear Si	S
agreen	en Facilities Agreement dated 21 November 2006 as amended and restated by an amendment ent dated 23 November 2022 as amended and restated by an amendment agreement dated 19 ber 2024 and as further amended from time to time (the "Facilities Agreement")
1.	We refer to the Facilities Agreement. This is an Accession Letter. Terms defined in the Facilities Agreement have the same meaning in this Accession Letter unless given a different meaning in this Accession Letter.
2.	[Subsidiary] agrees to become an Additional [Borrower]/[Guarantor] and to be bound by the terms of the Facilities Agreement, the Intercreditor Agreement and the other Finance Documents as an Additional [Borrower]/[Guarantor] pursuant to clause [30.2 (Additional Borrowers)]/[clause 30.4 (Additional Guarantors)] of the Facilities Agreement and as an [Obligor] pursuant to clause [] of the Intercreditor Agreement. [Subsidiary] is a company duly incorporated under the laws of [name of relevant jurisdiction] and is a limited liability company with registered number [].
3.	[Subsidiary's] administrative details are as follows:
	Address:
	Fax No.:
	Attention:
4.	[Add any applicable guarantee limitation language, following the Agreed Security Principles]
5.	This Accession Letter and any non-contractual obligations arising out of or in connection with it are governed by English law.
	[This Guarantor Accession Letter is entered into by deed.]
	[Formula One Management Limited] [Subsidiary]
	236

Form of Resignation Letter

To:		[] as Facility Agent	
From:		[resigning Obligor] and [Formula 0	One Management Limited]
Dated:	:	[]	
Dear Si	rs		
agreen	nent da	ated 23 November 2022 as amende	ber 2006 as amended and restated by an amendment d and restated by an amendment agreement dated 19 ne to time (the "Facilities Agreement")
1.			Resignation Letter. Terms defined in the Facilities Agreement ter unless given a different meaning in this Resignation Letter.
2.	reques		Borrower)]/[clause 30.5 (Resignation of a Guarantor)], we d from its obligations as a [Borrower]/[Guarantor] under the ent and the Finance Documents.
3.	We co	nfirm that:	
	(a)	no Default is continuing or would result	from the acceptance of this request; and
	(b)	[this request is given in relation to a Th	ird Party Disposal of [resigning Obligor];
	(c)	[this request is given in relation to the A	ASM Dissolution];
	(d)	[this request is given in relation to a Per	rmitted Group Company Disposal];
	(e)	the Disposal Proceeds (if any) have been and Insurance Proceeds and Excess Case	en or will be applied in accordance with clause 12.2 (Disposal $hflow$).
4.	This le law.	etter and any non-contractual obligations	arising out of or in connection with it are governed by English
	[For	mula One Management Limited]	resigning Obligor]
	Ву:	I	Ву:

		Form of C	Compliance Certificate
To:		J.P. Morgan SE as Facility Agent	
From:		[Delta Topco Limited] / [FOM]	
Dated	:	[]	
Dear Si	irs		
agreen	ment dat	ed 23 November 2022 as amer	ember 2006 as amended and restated by an amendment added and restated by an amendment agreement dated 19 time to time (the "Facilities Agreement")
1.	Agreem		his is a Compliance Certificate. Terms defined in the Facilities in used in this Compliance Certificate unless given a different
2.	We conf	irm that, in respect of the Testing P	eriod ending []:
			ified] / [the financial covenant specified in clause 26 (Financial acilities Agreement was not required to be tested.]
	-	firm that [Net First Lien Secured Le ould be [] per cent.]	verage] is []:1 and that, therefore, the Margin relating to [Facility
3.	[We cor	firm that no Default is continuing.]	
4.	_	nfirm that the following companies ent: [].]	constitute Material Companies for the purposes of the Facilities
	Covena	nt)] / of the Borrower and the Gu	DA, as defined in clause 26 (Financial Definitions and Financial arantors exceeds 80 per cent of the [Consolidated EBITDA, as nd Financial Covenants)] of the Group].
Signed	d	Authorised Signatory Of [Parent]	Authorised Signatory Of [Parent]
[insert	auditor	's report language]	
		£ 0.5	
ior and	on behal	1 01	

[name of Auditors of the Parent]

Agreed Security Principles

1. SECURITY PRINCIPLES

- (i) The guarantees and security to be provided after the Sixth Restatement Date will be given in accordance with the agreed security principles set out in this schedule. This schedule addresses the manner in which the agreed security principles will impact on the guarantees and security proposed to be taken in relation to this transaction.
- (ii) The Agreed Security Principles embody recognition by all parties that there may be certain legal and practical difficulties in obtaining guarantees and security from all Obligors in every jurisdiction in which Obligors are located. In particular:
 - (a) general statutory limitations, financial assistance, corporate benefit, fraudulent preference, "thin capitalisation" rules, retention of title claims and similar principles may limit the ability of a member of the Group to provide a guarantee or security or may require that the guarantee or security be limited by an amount or otherwise; the Parent will use reasonable endeavours to assist in demonstrating that adequate corporate benefit accrues to each Obligor;
 - (b) the security and extent of its perfection will be agreed taking into account the cost to the Group of providing security so as to ensure that it is proportionate (in the opinion of the Security Agent, acting reasonably) to the benefit accruing to the Lenders (in particular no security will be granted by any member of the Group which is not a Guarantor or Borrower);
 - (c) any assets subject to third party arrangements which are permitted by this agreement and which prevent those assets from being charged will be excluded from the fixed charge (or equivalent local security) in any relevant security document **provided that** reasonable endeavours to obtain consent to charging any such assets shall be used by the Obligors if the relevant asset is material;
 - (d) members of the Group not incorporated in the UK will not be required to give guarantees or enter into security documents if it would conflict with the fiduciary duties of their directors or contravene any legal or regulatory prohibition or result in a risk of personal or criminal liability on the part of any officer **provided that** the relevant Group member shall use reasonable endeavours to overcome any such obstacle;
 - (e) perfection of security, when required, and other legal formalities will be completed as soon as practicable and, in any event, within the time periods specified in the Finance Documents therefor or (if earlier or to the extent no such time periods are specified in the Finance Documents) within the time periods specified by applicable law in order to ensure due perfection. The perfection of security granted will not be required if it would have a material adverse effect on the ability of the relevant Obligor to conduct its operations and business in the ordinary course or as otherwise permitted by the Finance Documents (including, without limitation, notification of receivables security to third party debtors until an Event of Default has occurred and is continuing and the Facility Agent has given notice of intention to enforce in accordance with the terms of the relevant credit agreement). The registration of security interests in intellectual property will not be required;

- (f) no perfection action will be required in jurisdictions where Obligors are not located but perfection action may be required in the jurisdiction of one Obligor in relation to security granted by another Obligor located in a different jurisdiction;
- (h) other than a general security agreement and related filing, no perfection action will be required with respect to assets of a type not owned by members of the Group;
- (j) the Security Agent will hold one set of security for the Lenders; and
- (k) each of the Lenders shall be able to enforce the security constituted by the security documents without any restriction from either (i) the constitutional documents of the relevant Obligor or (ii) any company which is or whose assets are the subject of such security document (but subject to any inalienable statutory rights which the company may have to challenge such enforcement) or (iii) any shareholders of the foregoing not party to the relevant security document.
- (iii) The Security Agent or the Finance Parties, as the case may be, shall promptly discharge any guarantees and release any Security which is or are subject to any legal or regulatory prohibition as is referred to in paragraph (ii)(d) above.
- (iv) The Group will not grant security contrary to the 100 Year Agreements. The Obligors will grant security over shares and intercompany loans but a floating charge (in the case of Obligors who are incorporated in England or another jurisdiction where floating charges are a recognised and valid form of security) only will be granted over the other assets. The floating charge will be stated to take effect only to the extent it does not contravene the 100 Year Agreements; and the Security Agent will agree not to enforce the security in a manner that contravenes the 100 Year Agreements and to release security to the extent necessary (in connection with any transfer of "Transfer Rights" (as defined in the 100 Year Agreements) to FOAM to comply with the provisions of the FOAM articles of association. It is acknowledged that the terms of the Sixth Restatement English Law Debenture comply with the requirements of this paragraph (iv).

2. GUARANTORS AND SECURITY

- (i) Each guarantee will be an upstream, cross-stream and downstream guarantee and for all liabilities of the Obligors in respect of the Secured Obligations (as defined in the Intercreditor Agreement) in accordance with, and subject to, the requirements of the Agreed Security Principles in each relevant jurisdiction. Transaction Security Documents will secure the guarantee obligations of the relevant security provider or, if such security is provided on a third party basis, all liabilities of the Obligors in respect of the Secured Obligations (as defined in the Intercreditor Agreement), in each case in accordance with, and subject to, the requirements of the Agreed Security Principles in each relevant jurisdiction.
- (ii) Where an Obligor pledges shares, the security document will be governed by the laws of the company whose shares are being pledged and not by the law of the country of the pledgor. Subject to these principles the shares owned by Group Companies in each guarantor company (other than the Parent) shall be secured.
- (iii) To the extent legally effective, all security shall be given in favour of the Security Agent and not the Finance Parties individually. "Parallel debt" provisions will be used where necessary; such provisions will be contained in the Intercreditor Agreement and not the individual Transaction Security Documents unless required under local laws. To the extent possible, there should be no action required to be taken in

- relation to the guarantees or security when any Lender assigns or transfers any of its participation in the Facilities to a new Lender.
- (iv) The Guarantors will not be required to pay the cost of any re-execution, notarisation, re-registration, amendment or other perfection requirement for any security on any assignment or transfer to a new Lender. The cost or fee shall be for the account of the transferee Lender.
- (v) Where an Obligor acquires assets of material value or significance (in the opinion of the Majority Lenders, acting reasonably) after the date on which it initially grants security, such Obligor shall enter into security in accordance with the Agreed Security Principles in respect of such assets to the extent that such assets are of the same type as assets already subject to Transaction Security and such assets are not subject to the existing security created by such Obligor.

3. TERMS OF TRANSACTION SECURITY DOCUMENTS

The following principles will be reflected in the terms of any security taken as part of this transaction except to the extent that local counsel to the Facility Agent in any relevant jurisdiction advises that such principles are inconsistent with valid, binding and enforceable security over the assets which are intended to be the subject of the relevant security:

- (i) the security will be first ranking, to the extent possible;
- security will not be enforceable until an Event of Default has occurred which is continuing and notice
 of acceleration in connection thereof has been given by the relevant Facility Agent in accordance with
 the terms of the relevant credit agreement;
- (iii) the Lenders and Hedge Counterparties shall only be able to exercise a power of attorney following the occurrence of an Event of Default which is continuing or if the relevant Obligor has failed to comply with a further assurance or perfection obligation;
- (iv) in the Transaction Security Documents there will be no repetition or extension of clauses set out in this agreement or the Intercreditor Agreement such as those relating to notices, cost and expenses, indemnities, tax gross up, distribution of proceeds and release of security other than as required in connection with the perfection of such security; representations and undertakings shall be included in the security documents only to the extent required by local law in order to create or perfect the security interest expressed to be created thereby;
- (v) information, such as lists of assets, will be provided if, and only to the extent, required by local law to be provided to perfect or register the security and, when required, shall be provided annually (unless required more frequently under local law) or, following an Event of Default which is outstanding, on the Security Agent's reasonable request; and
- (vi) security will, where possible and practical, automatically create security over future assets of the same type as those already secured; where local law requires supplemental pledges to be delivered in respect of future acquired assets in order for effective security to be created over that class of asset, such supplemental pledges shall be provided at intervals required under local law).

4. INTERCOMPANY RECEIVABLES

Each Obligor will grant security over its material intercompany receivables, provided that it shall be free to deal with those receivables in the course of its business until an Event of Default has occurred which is continuing and notice of acceleration in connection thereof

has been given by the relevant Facility Agent in accordance with the terms of the relevant credit agreement.

If required by local law to perfect the security, notice of the security will be served on the relevant debtor promptly upon the security being granted and the Obligor shall use its reasonable endeavours to obtain an acknowledgement of that notice within 20 Business Days of service. Irrespective of whether notice of the security is required for perfection if the service of notice would prevent the Obligor from dealing with an intercompany receivable in the course of its business no notice of security shall be served until an Event of Default has occurred which is continuing and notice of acceleration in connection thereof has been given by the relevant Facility Agent in accordance with the terms of the relevant credit agreement.

If required under local law security over intercompany receivables will be registered subject to the general principles set out in these Agreed Security Principles.

Where any intercompany receivables being secured are governed by the laws of England and Wales, to the extent practicable such security will be granted by way of accession by the charging company to the Sixth Restatement English Law Debenture or otherwise pursuant to another security document the terms of which shall be consistent with the Sixth Restatement English Law Debenture.

5. SHARES

The security document will be governed by the laws of the company whose shares are being secured and not by the law of the country of the Obligor granting the security.

Until an Event of Default has occurred which is continuing and notice of acceleration in connection therewith has been given by the relevant Facility Agent in accordance with the terms of the relevant credit agreement, the charging Obligor will be permitted to retain and to exercise voting rights appertaining to any shares charged by it and the company whose shares have been charged will be permitted to pay dividends upstream on pledged shares to the extent permitted under the Finance Documents with the proceeds to be available to the Parent and its subsidiaries.

Where customary, immediately upon execution of the share charge, the share certificate and a stock transfer form executed in blank will be provided to the Security Agent and where customary the share certificate or shareholders' register will be endorsed or written up and the endorsed share certificate or a copy of the written up register provided to the Security Agent.

Unless the restriction is required by law or regulation, the constitutional documents of the company whose shares have been charged will be amended to remove any restriction on the transfer or the registration of the transfer of the shares on enforcement of the security granted over them.

Where the company whose shares are being secured is incorporated in England and Wales, to the extent practicable such security will be granted by way of accession by the charging company to the Sixth Restatement English Law Debenture or the Sixth Restatement English Law Share Charge (where the charging company is not incorporated under the laws of England and Wales) or otherwise pursuant to another security document the terms of which shall be consistent with the Sixth Restatement English Law Share Charge.

6. OTHER TYPES OF ASSETS

No security over bank accounts, fixed assets, insurance policies, intellectual property, trade receivables or real estate shall be required other than to the extent provided under a floating charge over all or substantially all of the relevant Obligor's assets (in the case of Obligors

who are incorporated in England or another jurisdiction where floating charges are a recognised and valid form of security).

7. ENGLISH LAW DEBENTURE AND ENGLISH LAW SECURITY

Any Additional Obligor incorporated in England and Wales shall accede to the Sixth Restatement English Law Debenture (or otherwise accede or enter into another security document the terms of which are consistent with the Sixth Restatement English law Debenture) which is acknowledged to be compliant with any references to a "floating charge" in these security principles.

8. RELEASE OF SECURITY

Unless required by local law the circumstances in which the security shall be released should not be dealt with in individual security documents but, if so required, shall, except to the extent required by local law, be the same as those set out in the Intercreditor Agreement or in this agreement.

Timetables

Part 1 - Loans

	Loans in euro	Loans in Dollars	Loans in other currencies
Approval as an Optional Currency, if required (clause 4.3 (Conditions Relating to Optional Currencies))			
Facility Agent notifies Formula One Management Limited if a currency is approved as an Optional Currency in accordance with clause 4.3 (Conditions Relating to Optional Currencies)	-	-	U-4
Delivery of a duly completed Utilisation	U-3	U-3	U-3
Request (clause 5.1 (Delivery of a Utilisation Request) or a Selection Notice (clause 15.1 (Selection of Interest Periods and Terms)	9.30 a.m.	9.30 a.m.	9.30 a.m.
Facility Agent determines (in relation to a	U-3	U-3	U-3
Utilisation) the Base Currency Amount of the Loan, if required under clause 5.4 (Lenders' Participation)	Noon	Noon	Noon
Facility Agent notifies the Lenders of the Loan in accordance with clause 5.4	U-3	U-3	U-3
(Lenders' Participation)	3.00 p.m.	3.00 p.m.	3.00 p.m.
Facility Agent receives a notification from a Lender under clause 8.2 (Unavailability of	-	Q	Q
a Currency)		3.00 p.m.	3.00 p.m.
Facility Agent gives notice in accordance	-	Q	Q
with clause 8.2 (Unavailability of a Currency)		5.00 p.m.	5.00 p.m.
Term SOFR Rate or EURIBOR is fixed	Quotation Day as of 11.00 a.m. (Brussels time) in respect of EURIBOR	Quotation Day as of 5.00 a.m. (Chicago time) in respect of Term SOFR	Quotation Day as of 11.00 a.m.

[&]quot;Q" = Quotation Day

[&]quot;U" = date of utilisation, or in the case of a Term Loan that has already been borrowed, the first day of the relevant Interest Period for that Term Loan (or, if it is a RFR Loan, the Interest Payment Date for that RFR Loan)

"U - X" $\,=\,$ X Business Days prior to date of utilisation, or in the case of a Term Loan that has already been borrowed, the first day of the relevant Interest Period for that Term Loan

Part 2 - Letters of Credit

Letters of Credit

Delivery of a duly completed Utilisation Request (clause U - 5 6.2 (Delivery of a Utilisation Request for Letters of Credit)

9.30 a.m.

Facility Agent determines (in relation to a Utilisation) $\,$ U - 1 the Base Currency Amount of the Letter of Credit if required under clause 6.5(d) (Issue of Letters of Credit) noon and notifies the Issuing Bank and Lenders of the Letter of Credit in accordance with clause 6.5(d) (Issue of Letters of Credit).

Delivery of duly completed Renewal Request

"U" = date of utilisation

"U-X" = Business Days prior to date of utilisation

Form of Letter of Credit

To: [Beneficiary](the "Beneficiary")

Date

Irrevocable Standby Letter of Credit no. []

At the request of [], [Issuing Bank] (the "Issuing Bank") issues this irrevocable standby Letter of Credit ("Letter of Credit") in your favour on the following terms and conditions:

1. **Definitions**

In this Letter of Credit:

"Business Day" means a day (other than a Saturday or a Sunday) on which banks are open for general business in [London].

"Demand" means a demand for a payment under this Letter of Credit in the form of the schedule to this Letter of Credit.

"Expiry Date" means [].

"Total L/C Amount" means [].

2. Issuing Bank's Agreement

- (a) The Beneficiary may request a drawing or drawings under this Letter of Credit by giving to the Issuing Bank a duly completed Demand. A Demand must be received by the Issuing Bank by no later than [] p.m. ([London] time) on the Expiry Date.
- (b) Subject to the terms of this Letter of Credit, the Issuing Bank unconditionally and irrevocably undertakes to the Beneficiary that, within [ten] Business Days of receipt by it of a Demand, it must pay to the Beneficiary the amount demanded in that Demand.
- (c) The Issuing Bank will not be obliged to make a payment under this Letter of Credit if as a result the aggregate of all payments made by it under this Letter of Credit would exceed the Total L/C Amount.

Expiry

- (a) The Issuing Bank will be released from its obligations under this Letter of Credit on the date (if any) notified by the Beneficiary to the Issuing Bank as the date upon which the obligations of the Issuing Bank under this Letter of Credit are released.
- (b) Unless previously released under paragraph (a) above, on [] p.m.([London] time) on the Expiry Date the obligations of the Issuing Bank under this Letter of Credit will cease with no further liability on the part of the Issuing Bank except for any Demand validly presented under the Letter of Credit that remains unpaid.
- (c) When the Issuing Bank is no longer under any further obligations under this Letter of Credit, the Beneficiary must return the original of this Letter of Credit to the Issuing Bank.

4. Payments

All payments under this Letter of Credit shall be made in [] and for value on the due date to the account of the Beneficiary specified in the Demand.

5. **Delivery of Demand**

Each Demand shall be in writing, and, unless otherwise stated, may be made by letter, fax or telex and must be received in legible form by the Issuing Bank at its address and by the particular department or office (if any) as follows:

[

]

6. **Assignment**

The Beneficiary's rights under this Letter of Credit may not assigned or transferred.

7. **ISF**

Except to the extent it is inconsistent with the express terms of this Letter of Credit, this Letter of Credit is subject to the International Standby Practices (ISP 98), International Chamber of Commerce Publication No. 590.

8. Governing Law

This Letter of Credit and any non-contractual obligations arising out of or in connection with it are governed by English law.

9. Jurisdiction

The [courts of England, sitting in London] have exclusive jurisdiction to settle any dispute arising out of or in connection with this Letter of Credit.

Yours faithfully

[Issuing Bank]

By:

SCHEDULE Form of Demand

To:	[ISSUING	BANK ¹

[Date]

Dears Sirs

Standby Letter of Credit no. [] issued in favour of [BENEFICIARY] (the "Letter of Credit")

We refer to the Letter of Credit. Terms defined in the Letter of Credit have the same meaning when used in this Demand.

- 1. We certify that the sum of [] is due [and has remained unpaid for at least [] Business Days] [under [set out underlying contract or agreement]]. We therefore demand payment of the sum of [].
- 2. Payment should be made to the following account:

Name

Account Number:

Bank:

3. The date of this Demand is not later than the Expiry Date.

Yours faithfully

(Authorised Signatory)

(Authorised Signatory)

For

[Beneficiary]

Incremental Facility Commitment Letter

To:	[] as Facility Agent
From:	[The Lender] (the "Lender")

Dated:

First Lien Facilities Agreement dated 21 November 2006 as amended and restated by an amendment agreement dated 23 November 2022 as amended and restated by an amendment agreement dated 19 September 2024 and as further amended from time to time (the "Facilities Agreement")

- 1. We refer to the Facilities Agreement. This letter (the **"Letter"**) shall take effect as an Incremental Facility Commitment Letter for the purpose of the Facilities Agreement. Terms defined in the Facilities Agreement have the same meaning in this Letter unless given a different meaning in this Letter.
- 2. We hereby agree to [become a Lender and] assume a New Incremental Facility Commitment in an amount of [] upon the following terms.

[Insert details/terms pursuant to clause 2.2(k)].

- 3. The proposed date on which our Incremental Facility Commitment is to take effect (the "Increase Date") is [].
- 4. We refer to clause 20.8 (New First Lien Funders and Second Lien Funders) of the Intercreditor Agreement and in consideration of our being accepted as a First Lien Lender for the purposes of the Intercreditor Agreement, we confirm that, as from Increase Date, we intend to be party to the Intercreditor Agreement as a First Lien Lender and we undertake to perform all the obligations expressed in the Intercreditor Agreement to be assumed by a First Lien Lender and agree that we shall be bound by all the provisions of the Intercreditor Agreement, as if we had been an original party to the Intercreditor Agreement.
- 5. Our Facility Office and address, fax number and attention details for notices to the Increase Lender for the purposes of clause 36.2 (Addresses) as follows:

[Facility Office address, fax number and attention details for notices and accounts details for payments]]

- 6. [On the Increase Date, we agree to become:
 - (a) party to the Facilities Agreement as a Lender; and
 - (b) party to the Intercreditor Agreement as a First Lien Lender.]
- 7. [We confirm, for the benefit of the Facility Agent and without liability to any Obligor, that we are:
 - (a) [not a Qualifying Lender];
 - (b) [a Qualifying Lender].]
- 8. This Letter and any non-contractual obligations arising out of or in connection with it are governed by English law.

Note: The execution of this Incremental Facility Commitment Letter may not entitle the Lender to a proportionate share of the Transaction Security in all jurisdictions. It is the responsibility of the Lender to ascertain whether any other documents or other formalities are required in any jurisdiction and, if so, to arrange for execution of those documents and completion of those formalities.

SIGNED

For and on behalf of [LENDER]

List of Designated Entities

Jurisdictions in relation to which the Designated Entity will participate in any Utilisation **Designated Entity Related Lender**

Bank of America Merrill Lynch Bank of America, N.A.

International Limited

United States of America

Form of Designated Entity Accession Agreement

To:	[FACILITY AGENT] as Facility Agent
From:	[DESIGNATED ENTITY] and [RELATED LENDER]
Date:	
agree	Lien Facilities Agreement dated 21 November 2006 as amended and restated by an amendment ment dated 23 November 2022 as amended and restated by an amendment agreement dated 19 mber 2024 and as further amended from time to time (the "Facilities Agreement")
Words	and expressions defined in the Agreement have the same meaning in this accession agreement.
1.	We refer to clause 5.6 (Designated Entities) of the Facilities Agreement. This is an accession agreement.
2.	The Related Lender designates the Designated Entity as its Facility Office for the purpose of participating in Utilisations to Borrowers in [JURISDICTION].
3.	[Name of Designated Entity] agrees to become a party to and to be bound by the terms of the Facilities Agreement as a Designated Entity.
4.	For the purposes of clause 36 (Notices) of the Facilities Agreement, the Designated Entity's address for notices is:
5.	This accession agreement and any non-contractual obligations arising in connection with it are governed by English law.
[DESIG	GNATED ENTITY]
Ву:	

Part 1 - Permitted Acquisition

Not used.	
Not used.	Part 2 - Permitted Disposal
Not used.	Part 3 - Permitted Financial Indebtedness
Not used.	Part 4 - Permitted Guarantees
Not used.	
	Part 5 – Permitted Joint Ventures
Not used.	
	Part 6 - Permitted Loans
Not used.	
	Part 7 – Permitted Security
Not used.	
	Part 8 – Cash Movement
Not used.	
	254

Notices

Part 1 - Obligors' Agent

Obligors' Agent:

Address: No. 2 St. James's Market, London, United Kingdom, SW1Y 4AH

Attention: The directors, Adam Babiker and David Thomsen

Group Companies:

(i) To Allsport Management S.A.:

Address: 10 route de l'Aeroport, 1215 Geneva 15, Switzerland Attention: The directors, Adam Babiker and David Thomsen

(ii) To Delta 2 (Lux) S.à r.l:

Address: 9, rue de Bitbourg, L-1273 Luxembourg, Grand Duchy of Luxembourg, R.C.S. Luxembourg:

B 122.129

Attention: The managers, Adam Babiker and David Thomsen

(iii) To Alpha Prema UK Limited, Beta Holdings Limited, Delta 3 (UK) Limited, Delta Topco Limited, Formula One Hospitality and Event Services Limited, Formula One Management Limited, Formula One Marketing Limited, Formula One Marketing II Limited, Formula One Digital Media Limited, Formula One Research, Engineering and Development Limited, Formula One World Championship Limited and SLEC Holdings Limited:

Address: No. 2 St. James's Market, London, United Kingdom, SW1Y 4AH

Attention: The directors, Adam Babiker and David Thomsen

Part 2 - Facility Agent and Security Agent

Facility Agent:

J.P. Morgan SE

Loans Agency Group

25 Bank Street, Canary Wharf, London E14 5JP

Telephone Number: +44 (0) 20 7742 1000

Fax Number: +44 (0)20 7777 2360

E-Fax: 12016395145@tls.ldsprod.com

Email: emea.london.agency@jpmorgan.com

Security Agent:

Manuel Caseiro

250 Bishopsgate, London EC2M 4AA, 4^{th} Floor,

Email: manuel.caseiro@natwest.com

SIGNATORIES TO THE SENIOR FACILITIES AGREEMENT

[Signatures not restated]¹

 $^{\scriptsize 1}$ Notice details not amended for this purpose but latest provided to Facility Agent will apply.

CERTIFICATION

- I, Gregory B. Maffei, certify that:
 - 1. I have reviewed this quarterly report on Form 10-Q of Liberty Media Corporation;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements and other financial information included in this quarterly report fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and we have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
 - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by this quarterly report based on such evaluation; and
 - d) disclosed in this quarterly report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent function):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date:	November 12, 2024
/s/ GREGO	RY B. MAFFEI
Gregory B. Maffei	
President and Chief Executive Officer	

CERTIFICATION

- I, Brian J. Wendling, certify that:
 - 1. I have reviewed this quarterly report on Form 10-Q of Liberty Media Corporation;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements and other financial information included in this quarterly report fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and we have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
 - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by this quarterly report based on such evaluation; and
 - d) disclosed in this quarterly report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent function):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date:	November 12, 2024		
/s/ BRIAN J. WENDLING			
Brian J. Wendling			
Chief Accounting Officer and Principal Financial Officer			

Certification

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code)

Pursuant to section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of section 1350, chapter 63 of title 18, United States Code), each of the undersigned officers of Liberty Media Corporation, a Delaware corporation (the "Company"), does hereby certify, to such officer's knowledge, that:

The Quarterly Report on Form 10-Q for the period ended September 30, 2024 (the "Form 10-Q") of the Company fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934 and information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: November 12, 2024	/s/ GREGORY B. MAFFEI
	Gregory B. Maffei
	President and Chief Executive Officer
Dated: November 12, 2024	/s/ BRIAN J. WENDLING
	Brian J. Wendling
	Chief Accounting Officer and Principal Financial Officer

The foregoing certification is being furnished solely pursuant to section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of section 1350, chapter 63 of title 18, United States Code) and is not being filed as part of the Form 10-Q or as a separate disclosure document.

Unaudited Attributed Financial Information for Tracking Stock Groups

The following tables present our assets and liabilities as of September 30, 2024 and revenue and expenses for the three and nine months ended September 30, 2024 and 2023 and cash flows for the nine months ended September 30, 2024 and 2023. The tables further present our assets, liabilities, revenue, expenses and cash flows that are intended to be attributed to the Liberty Formula One Group (the "Formula One Group"), the Liberty Live Group, the Liberty SiriusXM Group (prior to the Liberty Sirius XM Holdings Split-Off, as defined in note 1) and the Liberty Braves Group (the "Braves Group") (prior to the Atlanta Braves Holdings Split-Off, as defined in note 1), respectively. The Reclassification, as described in note 1, is reflected in the attributed financial statements on a prospective basis from August 3, 2023. The financial information should be read in conjunction with our condensed consolidated financial statements for the nine months ended September 30, 2024 included in this Quarterly Report on Form 10-Q.

Notwithstanding the following attribution of assets, liabilities, revenue, expenses and cash flows to the Liberty SiriusXM Group (prior to the Liberty Sirius XM Holdings Split-Off, as defined in note 1), the Formula One Group, the Liberty Live Group and the Braves Group (prior to the Atlanta Braves Holdings Split-Off, as defined in note 1), our tracking stock capital structure does not affect the ownership or the respective legal title to our assets or responsibility for our liabilities. We and our subsidiaries are each responsible for our respective liabilities. Holders of Liberty Formula One common stock and Liberty Live common stock are holders of our common stock and are subject to risks associated with an investment in our company and all of our businesses, assets and liabilities. The issuance of Liberty Formula One common stock and Liberty Live common stock does not affect the rights of our creditors.

SUMMARY ATTRIBUTED FINANCIAL DATA

Formula One Group

Summary Balance Sheet Data:

	S	September 30, 2024	December 31, 2023	
		amounts in mi	llions	
Cash and cash equivalents	\$	2,666	1,408	
Investments in affiliates, accounted for using the equity method	\$	35	41	
Goodwill	\$	4,192	3,956	
Intangible assets subject to amortization, net	\$	2,757	2,858	
Total assets	\$	11,980	10,267	
Long-term debt, including current portion	\$	2,928	2,906	
Attributed net assets	\$	7,642	6,419	

Summary Statement of Operations Data:

	Three months ended		Nine months ended			
	September 30,		30,	Septembe	r 30,	
		2024	2023	2024	2023	
			amounts in mil	lions		
Revenue	\$	911	887	2,486	1,992	
Cost of Formula 1 revenue	\$	(554)	(615)	(1,472)	(1,340)	
Selling, general and administrative expense (1)	\$	(113)	(79)	(295)	(223)	
Operating income (loss)	\$	110	107	264	175	
Interest expense	\$	(54)	(56)	(162)	(161)	
Share of earnings (losses) of affiliates, net	\$	(1)	2	(6)	(1)	
Realized and unrealized gains (losses) on financial instruments,						
net	\$	39	66	86	83	
Unrealized gains (losses) on intergroup interest	\$	_	16	_	15	
Income tax (expense) benefit	\$	2	(31)	(20)	(35)	
Earnings (loss) attributable to Liberty stockholders	\$	117	118	218	125	

⁽¹⁾ Includes stock-based compensation of \$6 million and \$4 million for the three months ended September 30, 2024 and 2023, respectively, and \$24 million and \$14 million for the nine months ended September 30, 2024 and 2023, respectively.

Liberty Live Group

Summary Balance Sheet Data:

	 September 30, 2024	December 31, 2023
	amounts in mill	lions
Cash and cash equivalents	\$ 388	305
Investments in affiliates, accounted for using the equity method	\$ 448	333
Total assets	\$ 1,255	1,162
Long-term debt, including current portion	\$ 1,437	1,317
Attributed net assets	\$ (206)	(188)

Summary Statement of Operations Data:

		Three months ended		Nine month	s ended
		September 30,		Septembe	er 30,
	2	2024	2023	2024	2023
Selling, general and administrative expense (1)	\$	(3)	(5)	(7)	(5)
Operating income (loss)	\$	(3)	(5)	(7)	(5)
Share of earnings (losses) of affiliates, net	\$	117	90	181	90
Income tax (expense) benefit	\$	(5)	5	(21)	5
Earnings (loss) attributable to Liberty stockholders	\$	15	(19)	76	(19)

⁽¹⁾ Includes stock-based compensation expense of \$1 million for each of the three months ended September 30, 2024 and 2023, and \$3 million and \$1 million for the nine months ended September 30, 2024 and 2023, respectively.

BALANCE SHEET INFORMATION September 30, 2024 (unaudited)

		Attributed (note 1)				
	For	rmula One Group	Liberty Live Group amounts in 1	Inter-Group Eliminations	Consolidated Liberty	
Assets			amounts in i	millions		
Current assets:						
Cash and cash equivalents	\$	2,666	388	_	3,054	
Trade and other receivables, net		122	_	_	122	
Other current assets		503			503	
Total current assets		3,291	388	_	3,679	
Investments in affiliates, accounted for using the equity method (note 1)		35	448	_	483	
Property and equipment, at cost		1,010	_	_	1,010	
Accumulated depreciation		(183)	_	_	(183)	
·		827			827	
Goodwill		4,192	_	_	4,192	
Intangible assets subject to amortization, net		2,757	_	_	2,757	
Other assets		878	419	(27)	1,270	
Total assets	\$	11,980	1,255	(27)	13,208	
Liabilities and Equity						
Current liabilities:						
Accounts payable and accrued liabilities	\$	404	_	_	404	
Current portion of debt		21	55	_	76	
Deferred revenue		756	_	_	756	
Other current liabilities		37	_	_	37	
Total current liabilities	-	1,218	55		1,273	
Long-term debt (note 1)		2,907	1,382	_	4,289	
Deferred income tax liabilities		27	_	(27)	_	
Other liabilities		186	1		187	
Total liabilities		4,338	1,438	(27)	5,749	
Equity / Attributed net assets		7,642	(206)		7,436	
Noncontrolling interests in equity of subsidiaries			23	_	23	
Total liabilities and equity	\$	11,980	1,255	(27)	13,208	

STATEMENT OF OPERATIONS INFORMATION Three months ended September 30, 2024 (unaudited)

		Attributed (note 1)					
			Liberty	Liberty			
	Fori	nula One	Live	SiriusXM	Consolidated		
	(Group	Group	Group	Liberty		
			amounts in mi	llions			
Revenue:							
Formula 1 revenue	\$	848	_	_	848		
Other revenue		63			63		
Total revenue		911		_	911		
Operating costs and expenses:							
Cost of Formula 1 revenue (exclusive of							
depreciation shown separately below)		554	_	_	554		
Other cost of sales		41	_	_	41		
Other operating expenses		2	_	_	2		
Selling, general and administrative, including							
stock-based compensation (note 2)		113	3	_	116		
Impairment and acquisition costs		3	_	_	3		
Depreciation and amortization		88	_	_	88		
		801	3		804		
Operating income (loss)		110	(3)		107		
Other income (expense):			,				
Interest expense		(54)	(8)	_	(62)		
Share of earnings (losses) of affiliates, net		(1)	117	_	116		
Realized and unrealized gains (losses) on							
financial instruments, net		39	(94)	_	(55)		
Other, net		21	8	_	29		
		5	23		28		
Earnings (loss) from continuing operations before							
ncome taxes		115	20	_	135		
Income tax (expense) benefit		2	(5)	_	(3)		
Net earnings (loss) from continuing operations		117	15		132		
Net earnings (loss) from discontinued operations			_	(3,002)	(3,002)		
Net earnings (loss)		117	15	(3,002)	(2,870)		
Less net earnings (loss) attributable to the				(- ,)	(,,,,,)		
noncontrolling interests		_	_	(502)	(502)		
Net earnings (loss) attributable to Liberty							
stockholders	\$	117	15	(2,500)	(2,368)		

STATEMENT OF OPERATIONS INFORMATION Nine months ended September 30, 2024 (unaudited)

	'		Liberty	Liberty	_
	For	mula One	Live	SiriusXM	Consolidated
		Group	Group	Group	Liberty
			amounts in m	illions	
Revenue:					
Formula 1 revenue	\$	2,251	_	_	2,251
Other revenue		235	_	_	235
Total revenue		2,486	_	_	2,486
Operating costs and expenses:					
Cost of Formula 1 revenue (exclusive of					
depreciation shown separately below)		1,472	_	_	1,472
Other cost of sales		161	_	_	161
Other operating expenses		8	_	_	8
Selling, general and administrative, including					
stock-based compensation (note 2)		295	7	_	302
Impairment and acquisition costs		23	_	_	23
Depreciation and amortization		263	_	_	263
		2,222	7		2,229
Operating income (loss)		264	(7)		257
Other income (expense):			· · ·		
Interest expense		(162)	(22)	_	(184)
Share of earnings (losses) of affiliates, net		(6)	181	_	175
Realized and unrealized gains (losses) on					
financial instruments, net		86	(75)	_	11
Other, net		56	20	_	76
		(26)	104		78
Earnings (loss) from continuing operations before		<u> </u>			
ncome taxes		238	97	_	335
Income tax (expense) benefit		(20)	(21)	_	(41)
Net earnings (loss) from continuing operations	_	218	76		294
Net earnings (loss) from discontinued operations		_	_	(2,412)	(2,412)
Net earnings (loss)		218	76	(2,412)	(2,118)
Less net earnings (loss) attributable to the					
noncontrolling interests		_	_	(410)	(410)
Net earnings (loss) attributable to Liberty					
stockholders	\$	218	76	(2,002)	(1,708)

STATEMENT OF OPERATIONS INFORMATION Three months ended September 30, 2023 (unaudited)

			Attributed (n	ote 1)		
			Liberty	Liberty		
	Formula (One	Live	SiriusXM	Braves	Consolidated
	Group		Group	Group	Group	Liberty
			amoi	unts in millions		
Revenue:						
Formula 1 revenue	\$	887	_	_	_	887
Other revenue		_	_	_	49	49
Total revenue	,	887		_	49	936
Operating costs and expenses:						
Cost of Formula 1 revenue (exclusive of depreciation						
shown separately below)		615	_	_	_	615
Other operating expenses		_	_	_	38	38
Selling, general and administrative, including stock-						
based compensation (note 2)		79	5	_	9	93
Depreciation and amortization		86	_	_	3	89
		780	5		50	835
Operating income (loss)		107	(5)		(1)	101
Other income (expense):						
Interest expense		(56)	(2)	(1)	(3)	(62)
Share of earnings (losses) of affiliates, net		2	90	50	1	143
Realized and unrealized gains (losses) on financial						
instruments, net		66	(72)	45	_	39
Unrealized gains (losses) on intergroup interests (note						
1)		16	_	_	(20)	(4)
Other, net		14	(35)	1	1	(19)
		42	(19)	95	(21)	97
Earnings (loss) from continuing operations before income						
taxes		149	(24)	95	(22)	198
Income tax (expense) benefit		(31)	5	(21)	1	(46)
Net earnings (loss) from continuing operations		118	(19)	74	(21)	152
Net earnings (loss) from discontinued operations		_	<u>`</u>	291		291
Net earnings (loss)		118	(19)	365	(21)	443
Less net earnings (loss) attributable to the						
noncontrolling interests		_	_	58	_	58
Net earnings (loss) attributable to Liberty stockholders	\$	118	(19)	307	(21)	385
5 ()					<u> </u>	

STATEMENT OF OPERATIONS INFORMATION Nine months ended September 30, 2023 (unaudited)

Revenue 1 Signature Liberty Live Liberty SiriusXM Braves Group Consolidated SiriusXM Revenue: Formula I revenue \$ 1,992 — — 1,992 Other revenue — — — 350 350 Total revenue — — — — 350 2,342 Operating costs and expenses: — — — — 350 2,342 Operating costs and expenses: — — — — 350 2,342 Operating costs and expenses: — — — — — 350 2,342 Operating costs and expenses: — — — — — — — — 1,340 Other operating expenses —
Group Group Group Group Liberty Revenue: Formula 1 revenue \$ 1,992 — — — 1,992 Other revenue — — — 350 350 Total revenue — — — 350 2,342 Operating costs and expenses: — — — — 350 2,342 Operating costs and expenses: — — — — 350 2,342 Operating costs and expenses: — — — — — 1,340 Other operating expenses — — — — — — 1,340 Other operating expenses —
Revenue: Samounts in millions Samounts
Revenue: Formula 1 revenue \$ 1,992 — — — 1,992 Other revenue — — — — 350 350 Total revenue 1,992 — — 350 2,342 Operating costs and expenses: — — — — 350 2,342 Operating costs and expenses: — — — — — 1,340 Shown separately below) 1,340 — — — — 1,340 Other operating expenses — — — — — — 1,340 Other operating expenses — — — — — — — — — — 1,340 Other operating expenses — — — — — — — — — — 274 274 274 Section of Expension of Exp
Formula I revenue \$ 1,992 — — — 1,992 Other revenue — — — 350 350 Total revenue 1,992 — — 350 2,342 Operating costs and expenses: — — — — — 1,340 Other operating expenses — — — — — 1,340 Other operating expenses — — — — — — 1,340 Other operating expenses — — — — — — — — — 1,340 Other operating expenses — — — — — — — — — — 274 274 274 Selling, general and administrative, including stock-based compensation (note 2) 223 5 — 69 297 Impairment and acquisition costs — — — 1 1 1 1 1 1 1 1
Other revenue — — — 350 350 Total revenue 1,992 — — 350 2,342 Operating costs and expenses: Cost of Formula 1 revenue (exclusive of depreciation shown separately below) Other operating expenses — — — — — 1,340 Other operating expenses — — — — 274 274 Selling, general and administrative, including stock-based compensation (note 2) 223 5 — 69 297 Impairment and acquisition costs — — — 1 1 Depreciation and amortization 254 — — 37 291 Operating income (loss) 175 (5) — (31) 139 Other income (expense): Interest expense (161) (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial
Total revenue 1,992 — — 350 2,342 Operating costs and expenses: Cost of Formula 1 revenue (exclusive of depreciation shown separately below) 1,340 — — — 1,340 Other operating expenses — — — 274 274 Selling, general and administrative, including stock-based compensation (note 2) 223 5 — 69 297 Impairment and acquisition costs — — — — 1 1 Depreciation and amortization 254 — — 37 291 Operating income (loss) 175 5 — 381 2,203 Other income (expense): Interest expense (161) (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial — — — — — 350 — — — — — — — <
Operating costs and expenses: Cost of Formula 1 revenue (exclusive of depreciation shown separately below) 1,340 — — — 1,340 Other operating expenses — — — 274 274 Selling, general and administrative, including stock-based compensation (note 2) 223 5 — 69 297 Impairment and acquisition costs — — — 1 1 Depreciation and amortization 254 — — 37 291 Operating income (loss) 175 5 — 381 2,203 Other income (expense): 0 175 (5) — (31) 139 Other of earnings (losses) of affiliates, net (161) (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial
Cost of Formula 1 revenue (exclusive of depreciation shown separately below) 1,340 — — — — 1,340 Other operating expenses — — — 274 274 Selling, general and administrative, including stock-based compensation (note 2) 223 5 — 69 297 Impairment and acquisition costs — — — 1 1 Depreciation and amortization 254 — — 37 291 Operating income (loss) 175 5 — 381 2,203 Other income (expense): 0 175 0 — (31) 139 Other income (expense): 0 1 (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial 0 1 </td
shown separately below) 1,340 — — — 1,340 Other operating expenses — — — 274 274 Selling, general and administrative, including stock-based compensation (note 2) 223 5 — 69 297 Impairment and acquisition costs — — — 1 1 Depreciation and amortization 254 — — 37 291 Operating income (loss) 175 5 — 381 2,203 Other income (expense): — (161) (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial — — — — 1,340
Other operating expenses — — — 274 274 Selling, general and administrative, including stock-based compensation (note 2) 223 5 — 69 297 Impairment and acquisition costs — — — 1 1 Depreciation and amortization 254 — — 37 291 Operating income (loss) 175 5 — 381 2,203 Other income (expense): 175 (5) — (31) 139 Other income (expense): 161 (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial 1 <
Selling, general and administrative, including stock-based compensation (note 2) 223 5 — 69 297 Impairment and acquisition costs — — — 1 1 Depreciation and amortization 254 — — 37 291 1,817 5 — 381 2,203 Operating income (loss) 175 (5) — (31) 139 Other income (expense): Interest expense (161) (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial
based compensation (note 2) 223 5 — 69 297 Impairment and acquisition costs — — — — 1 1 Depreciation and amortization 254 — — 37 291 Sequence of the composition
Impairment and acquisition costs — — — — 1 1 Depreciation and amortization 254 — — 37 291 1,817 5 — 381 2,203 Operating income (loss) 175 (5) — (31) 139 Other income (expense): — — (161) (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial
Depreciation and amortization 254 — — 37 291 1,817 5 — 381 2,203 Operating income (loss) 175 (5) — (31) 139 Other income (expense): Therest expense (161) (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial — — 37 291
1,817 5 — 381 2,203 Operating income (loss) 175 (5) — (31) 139 Other income (expense): Interest expense (161) (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial 10
Operating income (loss) 175 (5) — (31) 139 Other income (expense): Interest expense (161) (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial (1) <td< td=""></td<>
Other income (expense): Interest expense (161) (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial
Interest expense (161) (2) (4) (21) (188) Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial
Share of earnings (losses) of affiliates, net (1) 90 127 12 228 Realized and unrealized gains (losses) on financial
Realized and unrealized gains (losses) on financial
instruments, net 83 (72) (59) 3 (45)
Unrealized gains (losses) on intergroup interests (note
1) — — (83) — (68)
Other, net 48 (35) (6) 5 12
(16) (19) 58 (84) (61)
Earnings (loss) from continuing operations before
income taxes 159 (24) 58 (115) 78
Income tax (expense) benefit (35) 5 (13) 6 (37)
Net earnings (loss) from continuing operations 124 (19) 45 (109) 41
Net earnings (loss) from discontinued operations — 757 — 757
Net earnings (loss) 124 (19) 802 (109) 798
Less net earnings (loss) attributable to the
noncontrolling interests (1) — 146 — 145
Net earnings (loss) attributable to Liberty stockholders \$ 125 (19) 656 (109) 653

STATEMENT OF CASH FLOWS INFORMATION Nine months ended September 30, 2024 (unaudited)

		Attri	buted (note 1)		
			Liberty	Liberty	
	Formula O	ne	Live	SiriusXM	Consolidated
	Group		Group	Group	Liberty
			amounts in m		
Cash flows from operating activities:					
Net earnings (loss)	\$	218	76	(2,412)	(2,118)
Adjustments to reconcile net earnings (loss) to net cash provided by operating					
activities:					
Net (earnings) loss from discontinued operations		_	_	2,412	2,412
Depreciation and amortization		263	_	_	263
Stock-based compensation		24	3	_	27
Share of (earnings) loss of affiliates, net		6	(181)	_	(175)
Realized and unrealized (gains) losses on financial instruments, net		(86)	75	_	(11)
Deferred income tax expense (benefit)		22	22	_	44
Intergroup tax allocation		(97)	(1)		(98)
Intergroup tax (payments) receipts		128	3	_	131
Other, net		25	(4)	_	21
Changes in operating assets and liabilities					
Current and other assets		(71)	_	_	(71)
Payables and other liabilities		155	(4)	_	151
Net cash provided (used) by operating activities		587	(11)		576
Cash flows from investing activities:					
Investments in equity method affiliates and debt and equity securities		(7)	(3)	_	(10)
Cash proceeds from dispositions		_	107	_	107
Cash (paid) received for acquisitions, net of cash acquired		(205)	_	_	(205)
Capital expended for property and equipment, including internal-use		` ′			` ` `
software and website development		(52)	_	_	(52)
Other investing activities, net		(13)	1	_	(12)
Net cash provided (used) by investing activities		(277)	105		(172)
Cash flows from financing activities:		()			
Borrowings of debt		644	_	_	644
Repayments of debt		(671)	(12)	_	(683)
Issuance of Series C Liberty Formula One common stock		939	_	_	939
Other financing activities, net		42	1	_	43
Net cash provided (used) by financing activities		954	(11)		943
Effect of foreign exchange rate changes on cash, cash equivalents and		751	(11)		
restricted cash		1	_		1
Net cash provided (used) by discontinued operations:		•			1
Cash provided (used) by operating activities		_	_	879	879
Cash provided (used) by investing activities		_	_	(709)	(709)
Cash provided (used) by financing activities			_	(485)	(485)
Net cash provided (used) by discontinued operations				(315)	(315)
Net increase (decrease) in cash, cash equivalents and restricted cash		.265	83	(315)	1,033
Cash, cash equivalents and restricted cash at beginning of period		,408	305	315	2,028
		_		313	
Cash, cash equivalents and restricted cash at end of period	\$ 2	2,673	388		3,061

STATEMENT OF CASH FLOWS INFORMATION Nine months ended September 30, 2023 (unaudited)

		Attributed (no	te 1)			
		Liberty	Liberty			
	Formula One	Live	SiriusXM	Braves	Consolidated	
	Group	Group	Group	Group	Liberty	
	отопр		ints in millions	Отопр		
Cash flows from operating activities:						
Net earnings (loss)	\$ 124	(19)	802	(109)	798	
Adjustments to reconcile net earnings (loss) to net cash						
provided by operating activities:						
Net (earnings) loss from discontinued operations	_	_	(757)	_	(757)	
Depreciation and amortization	254	_	_	37	291	
Stock-based compensation	14	1	_	7	22	
Share of (earnings) loss of affiliates, net	1	(90)	(127)	(12)	(228)	
Realized and unrealized (gains) losses on financial						
instruments, net	(83)	72	59	(3)	45	
Unrealized (gains) losses on intergroup interests, net	(15)	_	_	83	68	
Deferred income tax expense (benefit)	19	(4)	14	(5)	24	
Intergroup tax allocation	(135)	(1)	(1)	_	(137)	
Intergroup tax (payments) receipts	91	_	_	(1)	90	
Other, net	6	34	6	4	50	
Changes in operating assets and liabilities						
Current and other assets	(122)	(10)	_	(34)	(166)	
Payables and other liabilities	400	14	1	65	480	
Net cash provided (used) by operating activities	554	(3)	(3)	32	580	
Cash flows from investing activities:						
Investments in equity method affiliates and debt and equity						
securities	(173)	_	_	_	(173)	
Cash proceeds from dispositions	110	1	_	_	111	
Capital expended for property and equipment, including						
internal-use software and website development	(308)	_	_	(35)	(343)	
Other investing activities, net	(20)	_	_	_	(20)	
Net cash provided (used) by investing activities	(391)	1		(35)	(425)	
Cash flows from financing activities:						
Borrowings of debt	_	1.135	_	30	1,165	
Repayments of debt	(64)	(918)	_	(20)	(1,002)	
Settlement of intergroup interests	(273)	_	_	_	(273)	
Atlanta Braves Holdings, Inc. Split-Off	_	_	_	(188)	(188)	
Reclassification	(100)	100	_	_	_	
Other financing activities, net	11	_	3	8	22	
Net cash provided (used) by financing activities	(426)	317	3	(170)	(276)	
Net cash provided (used) by discontinued operations:	(120)			(170)	(270)	
Cash provided (used) by operating activities	_	_	1,301	_	1,301	
Cash provided (used) by operating activities Cash provided (used) by investing activities	_		(565)		(565)	
Cash provided (used) by financing activities Cash provided (used) by financing activities			(771)		(771)	
Net cash provided (used) by discontinued operations			(35)		(35)	
Net increase (decrease) in cash, cash equivalents and			(33)		(33)	
restricted cash	(263)	315	(35)	(173)	(156)	
Cash, cash equivalents and restricted cash at beginning	(203)	313	(33)	(1/3)	(130)	
of period	1 722	NA	370	173	2,276	
	1,733	INA	3/0	1/3	2,2/0	
Cash, cash equivalents and restricted cash at end of period	\$ 1,470	315	335		2,120	
periou	1,770	515	333		2,120	

Notes to Attributed Financial Information (Continued) (unaudited)

 A tracking stock is a type of common stock that the issuing company intends to reflect or "track" the economic performance of a particular business or "group," rather than the economic performance of the company as a whole.

On July 18, 2023, the Company completed the split-off (the "Atlanta Braves Holdings Split-Off") of its wholly owned subsidiary, Atlanta Braves Holdings, Inc ("Atlanta Braves Holdings"). The Atlanta Braves Holdings Split-Off was accomplished by a redemption by the Company of each outstanding share of Liberty Braves common stock in exchange for one share of the corresponding series of Atlanta Braves Holdings common stock. Atlanta Braves Holdings is comprised of the businesses, assets and liabilities attributed to the Braves Group immediately prior to the Atlanta Braves Holdings Split-Off, except for the intergroup interests in the Braves Group attributed to the Liberty SiriusXM Group and the Formula One Group, which were settled and extinguished in connection with the Atlanta Braves Holdings Split-Off.

On August 3, 2023, the Company reclassified its then-outstanding shares of common stock into three new tracking stocks—Liberty SiriusXM common stock, Liberty Formula One common stock and Liberty Live common stock, and, in connection therewith, provided for the attribution of the businesses, assets and liabilities of the Company's remaining tracking stock groups among its newly created Liberty SiriusXM Group, Formula One Group and Liberty Live Group (the "Reclassification"). As a result of the Reclassification, each then-outstanding share of Liberty SiriusXM common stock was reclassified into one share of the corresponding series of new Liberty SiriusXM common stock and 0.2500 of a share of the corresponding series of Liberty Live common stock and each outstanding share of Liberty Formula One common stock was reclassified into one share of the corresponding series of new Liberty Formula One common stock and 0.0428 of a share of the corresponding series of Liberty Live common stock.

Each of the Atlanta Braves Holdings Split-Off and the Reclassification were intended to be tax-free to stockholders of the Company, except with respect to the receipt of cash in lieu of fractional shares. In July 2024, the IRS completed its review of the Reclassification and notified the Company that it agreed with the nontaxable characterization of the transaction. In September 2024, the IRS completed its review of the Atlanta Braves Holdings Split-Off and notified the Company that it agreed with the nontaxable characterization of the transaction. The Atlanta Braves Holdings Split-Off and the Reclassification are reflected in the Company's consolidated financial statements and these attributed financial statements on a prospective basis.

On September 9, 2024, Liberty completed the split-off (the "Liberty Sirius XM Holdings Split-Off") of its wholly owned subsidiary, Liberty Sirius XM Holdings Inc. ("Liberty Sirius XM Holdings"). The Liberty Sirius XM Holdings Split-Off was accomplished through the redemption by the Company of each outstanding share of Liberty SiriusXM common stock in exchange for 0.8375 of a share of Liberty Sirius XM Holdings common stock, with cash paid in lieu of fractional shares. Liberty Sirius XM Holdings is comprised of the businesses, assets and liabilities attributed to the Liberty SiriusXM Group immediately prior to the Liberty Sirius XM Holdings Split-Off. The Liberty Sirius XM Holdings Split-Off was intended to be tax-free to holders of Liberty SiriusXM common stock (except with respect to cash received in lieu of fractional shares).

Liberty Sirius XM Holdings is presented as a discontinued operation in the Company's condensed consolidated financial statements as the Liberty Sirius XM Holdings Split-Off represents a strategic shift that had a major effect on the Company's operations and financial results.

While the Formula One Group and Liberty Live Group have separate collections of businesses, assets and liabilities attributed to them, no group is a separate legal entity and therefore cannot own assets, issue securities or enter into legally binding agreements. Holders of tracking stock have no direct claim to the group's stock or assets and therefore, do not own, by virtue of their ownership of a Liberty tracking stock, any equity or voting interest in a public company, such as Live Nation Entertainment, Inc. ("Live Nation"), in which Liberty holds an interest and that is attributed to a Liberty tracking stock group, the Liberty Live Group. Holders of tracking stock are also not represented by separate boards of directors. Instead, holders of tracking stock are stockholders of the parent corporation, with a single board of directors and subject to all of the risks and liabilities of the parent corporation.

Notes to Attributed Financial Information (Continued) (unaudited)

As of September 30, 2024, the Formula One Group was primarily comprised of Liberty's interests in Formula 1 and QuintEvents, LLC, cash and Liberty's 2.25% Convertible Senior Notes due 2027. As of September 30, 2024, the Formula One Group had cash and cash equivalents of approximately \$2,666 million, which included \$1,449 million of subsidiary cash.

As of September 30, 2024, the Liberty Live Group is primarily comprised of Liberty's interest in Live Nation, cash, other minority investments, Liberty's 0.5% Exchangeable Senior Debentures due 2050, Liberty's 2.375% Exchangeable Senior Debentures due 2053 and an undrawn margin loan.

Prior to the Liberty Sirius XM Holdings Split-Off, the Liberty SiriusXM common stock was intended to track and reflect the separate economic performance of the businesses, assets and liabilities attributed to the Liberty SiriusXM Group. At the time of the Liberty Sirius XM Holdings Split-Off, the Liberty SiriusXM Group was comprised of Liberty's interest in Sirius XM Holdings, corporate cash, Liberty's 3.75% Convertible Senior Notes due 2028, Liberty's 2.75% Exchangeable Senior Debentures due 2049 and a margin loan obligation incurred by a wholly-owned special purpose subsidiary of Liberty. As previously disclosed, Liberty Sirius XM Holdings is presented as a discontinued operation in the Company's condensed consolidated financial statements. Prior to the Reclassification, Liberty's interest in Live Nation, Liberty's 0.5% Exchangeable Senior Debentures due 2050 and a margin loan secured by shares of Live Nation were attributed to the Liberty SiriusXM Group and are presented as continuing operations in the Company's condensed consolidated financial statements.

Prior to the Atlanta Braves Holdings Split-Off, the Braves Group was primarily comprised of Braves Holdings, which indirectly owns the Atlanta Braves Major League Baseball Club ("ANLBC") and certain assets and liabilities associated with ANLBC's stadium and mixed-use development and cash.

As of December 31, 2022, 6,792,903 notional shares represented an 11.0% intergroup interest in the Braves Group previously held by the Formula One Group, 1,811,066 notional shares represented a 2.9% intergroup interest in the Braves Group previously held by the Liberty SiriusXM Group and 4,165,288 notional shares represented a 1.7% intergroup interest in the Formula One Group previously held by the Liberty SiriusXM Group.

The intergroup interests represented quasi-equity interests which were not represented by outstanding shares of common stock; rather, the Formula One Group and Liberty SiriusXM Group had attributed interests in the Braves Group, which were generally stated in terms of a number of shares of Liberty Braves common stock, and the Liberty SiriusXM Group also had an attributed interest in the Formula One Group, which was generally stated in terms of a number of shares of Liberty Formula One common stock. Each reporting period, the notional shares representing the intergroup interests were marked to fair value. The changes in fair value were recorded in the Unrealized gain (loss) on intergroup interests line item in the unaudited attributed condensed consolidated statements of operations.

During March 2023, the Formula One Group paid approximately \$202 million to the Liberty SiriusXM Group to settle a portion of the intergroup interest in the Formula One Group held by the Liberty SiriusXM Group, as a result of the repurchase of a portion of Liberty's 1.375% Cash Convertible Notes due 2023. On July 12, 2023, the Formula One Group paid approximately \$71 million to the Liberty SiriusXM Group to settle and extinguish the remaining intergroup interest in the Formula One Group held by the Liberty SiriusXM Group.

In connection with the Split-Off, the intergroup interests in the Braves Group attributed to the Liberty SiriusXM Group and Formula One Group were settled and extinguished through the attribution, to the respective tracking stock group, of Atlanta Braves Holdings Series C common stock on a one-for-one basis equal to the number of notional shares representing the intergroup interest. On July 19, 2023, the shares of Atlanta Braves Holdings Series C common stock attributed to the Formula One Group to settle and extinguish the intergroup interest in connection with the Split-Off were distributed on a pro rata basis to holders of Liberty Formula One common stock. During November 2023, Liberty exchanged the shares of Atlanta Braves Holdings Series C common stock attributed to the Liberty SiriusXM Group with a third party to satisfy certain debt obligations attributed to the Liberty SiriusXM Group.

Notes to Attributed Financial Information (Continued) (unaudited)

For information relating to investments in affiliates accounted for using the equity method and debt, see notes 7 and 8, respectively, of the accompanying condensed consolidated financial statements.

(2) Cash compensation expense for our corporate employees is allocated among the Formula One Group and Liberty Live Group and was allocated to the Braves Group prior to the Atlanta Braves Holdings Split-Off and the Liberty Sirius XM Group prior to the Liberty Sirius XM Holdings Split-Off, based on the estimated percentage of time spent providing services for each group. On an annual basis, estimated time spent is determined through an interview process and a review of personnel duties unless transactions significantly change the composition of companies and investments in either respective group which would require a timelier reevaluation of estimated time spent. Other general and administrative expenses are charged directly to the groups whenever possible and are otherwise allocated based on estimated usage or some other reasonably determined methodology. Stock compensation related to each tracking stock is calculated based on actual awards outstanding.

While we believe that this allocation method is reasonable and fair to each group, we may elect to change the allocation methodology or percentages used to allocate general and administrative expenses in the future.

- (3) The intergroup balance at September 30, 2024 and December 31, 2023 is primarily a result of timing of tax benefits.
- (4) The Liberty Formula One common stock and Liberty Live common stock have voting and conversion rights under our restated charter. Following is a summary of those rights. Holders of Series A common stock of each group are entitled to one vote per share, and holders of Series B common stock of each group are entitled to ten votes per share. Holders of Series C common stock of each group are entitled to 1/100th of a vote per share in certain limited cases and will otherwise not be entitled to vote. In general, holders of Series A and Series B common stock vote as a single class. In certain limited circumstances, the board may elect to seek the approval of the holders of only Series A and Series B Liberty Formula One common stock, or only Series A and Series B Liberty Live common stock.

At the option of the holder, each share of Series B common stock of each group will be converted into one share of Series A common stock of the same group. At the discretion of our board, the common stock related to one group may be converted into common stock of the same series that is related to another other group.